Triplecrown Acquisition Corp. Form PREM14A September 08, 2009

0

# SCHEDULE 14A (Rule 14a-101)

### INFORMATION REQUIRED IN PROXY STATEMENT

### **SCHEDULE 14A INFORMATION**

# Proxy Statement Pursuant to Section 14(a) of the Securities Exchange Act of 1934 (Amendment No. )

Filed by the Registrant x
Filed by a Party other than the Registrant o
Check the appropriate box:

x Preliminary Proxy Statement

Confidential, For Use of the Commission Only (as permitted by Rule 14a-6(e)(2))

o Definitive Proxy Statement
o Definitive Additional Materials
o Soliciting Material Under Rule 14a-12

### TRIPLECROWN ACQUISITION CORP.

(Name of Registrant As Specified in Its Charter)

(Name of Person(s) Filing Proxy Statement, if Other than the Registrant)

Payment of Filing Fee (Check the appropriate box):

o No fee required.

Fee computed on table below per Exchange Act Rules 14a-6(i)(1) and 0-11.

(1) Title of each class of securities to which transaction applies: *Common Stock and Warrants* 

(2) Aggregate number of securities to which transaction applies:

217,455,000

(3) Per unit price or other underlying value of transaction computed pursuant to Exchange Act Rule 0-11 (set forth the amount on which the filing fee is calculated and state how it was determined): \$9.68 and \$0.18, respectively (4) Proposed maximum aggregate value of transaction: \$1,353,564,400 (5) Total fee paid: \$75,528.81 Fee paid previously with preliminary materials: o Check box if any part of the fee is offset as provided by Exchange Act Rule 0-11(a)(2) and identify the filing for x which the offsetting fee was paid previously. Identify the previous filing by registration statement number, or the form or schedule and the date of its filing. (1) Amount previously paid: \$75,528.81 (2) Form, Schedule or Registration Statement No.: Form S-4, Registration Statement File No. 333-161773 (3) Filing Party: Cullen Agricultural Holding Corp. Date Filed: (4)

September 4, 2009

## TRIPLECROWN ACQUISITION CORP. 970 WEST BROADWAY, PMB 402 JACKSON, WYOMING 83001

# NOTICE OF SPECIAL MEETING OF STOCKHOLDER OF TRIPLECROWN ACQUISITION CORP. TO BE HELD ON , 2009

To the Stockholders of Triplecrown Acquisition Corp.:

NOTICE IS HEREBY GIVEN that the special meeting of stockholders of Triplecrown Acquisition Corp. (Triplecrown), a Delaware corporation, will be held at 10:00 a.m. eastern time, or2009, at the offices of Graubard Miller, Triplecrown s counsel, at The Chrysler Building, 405 Lexington Avenue, 19th Floor, New York, New York 10174. You are cordially invited to attend the meeting, which will be held for the following purposes:

to consider and vote upon separate proposals to amend Triplecrown s amended and restated certificate of incorporation to allow Triplecrown to complete the merger with Cullen Agricultural Holding Corp. (CAH), a wholly-owned subsidiary of Triplecrown, notwithstanding that: (i) Cullen Agricultural Technologies Inc. (Cullen Agritech), a Georgia corporation that will be CAH s operating business following the proposed business combination, is owned by Cullen Inc. Holdings Ltd. (Cullen Holdings), an affiliate of Eric J. Watson, Triplecrown s

- chairman of the board and treasurer, (ii) Triplecrown will ultimately be acquired by CAH and therefore

  Triplecrown will not acquire at least 50% of the voting securities of the target business, (iii) Cullen Agritech is not
  an operating business in the financial services industry and (iv) the fair market value of Cullen Agritech on the date
  of the transaction is less than 80% of the balance of Triplecrown s trust account we refer to these proposals
  collectively as the initial charter proposals;
  - to consider and vote upon a proposal to (i) adopt the Agreement and Plan of Reorganization, dated as of September 4, 2009, among Triplecrown, CAH, CAT Merger Sub, Inc. ( Triplecrown Merger Sub ), a wholly-owned subsidiary of CAH, Cullen Agritech and Cullen Holdings, which, among other things, provides for (a) the merger of
- (2) Triplecrown with and into CAH, with CAH being the surviving entity and (b) the merger of Triplecrown Merger Sub with and into Cullen Agritech, with Cullen Agritech being the surviving entity and becoming a wholly owned subsidiary of CAH, and (ii) approve the business combination contemplated by such Agreement and Plan of Reorganization we refer to this proposal as the merger proposal;
- (3) to consider and vote upon separate proposals to approve the following differences between the amended and restated certificate of incorporation of CAH to be in effect following the merger and Triplecrown s current amended and restated certificate of incorporation: (i) the name of the new public entity will be Cullen Agricultural Holding Corp. as opposed to Triplecrown Acquisition Corp.; (ii) CAH will have 200,000,000 authorized shares of common stock and 1,000,000 authorized shares of preferred stock and may increase or decrease such amounts without stockholder approval, as opposed to Triplecrown having 160,000,000 authorized shares of common stock and 1,000,000 authorized shares of preferred stock and may not increase or decrease such amounts without stockholder approval; (iii) CAH s corporate existence will be perpetual as opposed to Triplecrown s corporate existence terminating on October 22, 2009; and (iv) CAH s amended and restated certificate of incorporation will not include the various provisions applicable only to specified purpose acquisition corporations that Triplecrown s amended and restated certificate of incorporation contains we refer to these proposals collectively as the secondary charter

proposals; and

to consider and vote upon a proposal to adjourn the special meeting to a later date or dates, if necessary, to permit (4) further solicitation and vote of proxies if, based upon the tabulated vote at the time of the special meeting, the required proposals have not been approved we refer to this proposal as the stockholder adjournment proposal.

#### **TABLE OF CONTENTS**

These items of business are described in the attached proxy statement/prospectus, which you are encouraged to read in its entirety before voting. Only holders of record of Triplecrown common stock at the close of business on , 2009 are entitled to notice of the special meeting and to vote and have their votes counted at the special meeting and any adjournments or postponements of the special meeting.

After careful consideration, Triplecrown s board of directors has determined that the proposals are fair to and in the best interests of Triplecrown and its stockholders and recommends that you vote or give instruction to vote FOR the approval of all of the proposals.

All Triplecrown stockholders are cordially invited to attend the special meeting in person. To ensure your representation at the special meeting, however, you are urged to complete, sign, date and return the enclosed proxy card as soon as possible. If you are a stockholder of record of Triplecrown common stock, you may also cast your vote in person at the special meeting. If your shares are held in an account at a brokerage firm or bank, you must instruct your broker or bank on how to vote your shares or, if you wish to attend the meeting and vote in person, obtain a proxy from your broker or bank. If you do not submit your proxy or vote in person at the special meeting or, if you hold your shares through a broker or bank, if you do not instruct your broker how to vote your shares or obtain a proxy from your broker or bank to vote in person at the special meeting, it will have the same effect as a vote against the approval of the initial charter proposals, the merger proposal and the secondary charter proposals.

The officers, directors and stockholders of Triplecrown prior to its initial public offering have agreed to vote the shares of Triplecrown common stock they received before such initial public offering in accordance with the majority of the votes cast at the special meeting of stockholders with respect to the merger proposal by the holders of shares sold in Triplecrown s initial public offering.

A complete list of Triplecrown stockholders of record entitled to vote at the special meeting will be available for ten days before the special meeting at the principal executive offices of Triplecrown for inspection by stockholders during ordinary business hours for any purpose germane to the special meeting.

Your vote is important regardless of the number of shares you own. Whether you plan to attend the special meeting or not, please sign, date and return the enclosed proxy card as soon as possible in the envelope provided. If your shares are held in street name or are in a margin or similar account, you should contact your broker to ensure that votes related to the shares you beneficially own are properly counted.

Thank you for your participation. We look forward to your continued support.

, 2009

By Order of the Board of Directors Jonathan J. Ledecky

President, Secretary and Director

IF YOU RETURN YOUR PROXY CARD WITHOUT AN INDICATION OF HOW YOU WISH TO VOTE, YOUR SHARES WILL BE VOTED IN FAVOR OF EACH OF THE PROPOSALS AND YOU WILL NOT BE ELIGIBLE TO HAVE YOUR SHARES CONVERTED INTO A PRO RATA PORTION OF THE TRUST ACCOUNT IN WHICH A SUBSTANTIAL PORTION OF THE NET PROCEEDS OF TRIPLECROWN S INITIAL PUBLIC OFFERING ( IPO ) ARE HELD. YOU MUST AFFIRMATIVELY VOTE AGAINST THE MERGER PROPOSAL AND DEMAND THAT TRIPLECROWN CONVERT YOUR SHARES INTO CASH NO LATER THAN THE CLOSE OF THE VOTE ON THE MERGER PROPOSAL TO EXERCISE YOUR CONVERSION RIGHTS. IN ORDER TO CONVERT YOUR SHARES, YOU MUST TENDER YOUR

STOCK TO TRIPLECROWN S STOCK TRANSFER AGENT PRIOR TO THE SPECIAL MEETING OF TRIPLECROWN STOCKHOLDERS. YOU MAY TENDER YOUR STOCK BY EITHER DELIVERING YOUR STOCK CERTIFICATE TO THE TRANSFER AGENT OR BY DELIVERING YOUR SHARES ELECTRONICALLY THROUGH THE DEPOSITORY TRUST. IF THE MERGER IS NOT COMPLETED, THEN THESE SHARES WILL NOT BE CONVERTED INTO CASH. IF YOU HOLD THE SHARES IN STREET NAME, YOU WILL NEED TO INSTRUCT THE ACCOUNT EXECUTIVE AT YOUR BANK OR BROKER TO WITHDRAW THE SHARES FROM YOUR ACCOUNT IN ORDER TO EXERCISE YOUR CONVERSION RIGHTS. SEE SPECIAL MEETING OF TRIPLECROWN STOCKHOLDERS AND WARRANTHOLDERS- CONVERSION RIGHTS FOR MORE SPECIFIC INSTRUCTIONS.

## TRIPLECROWN ACQUISITION CORP. 970 WEST BROADWAY, PMB 402 **JACKSON, WYOMING 83001**

## NOTICE OF SPECIAL MEETING OF **WARRANTHOLDERS** OF TRIPLECROWN ACQUISITION CORP. **TO BE HELD ON , 2009**

To the Warrantholders of Triplecrown Acquisition Corp.:

NOTICE IS HEREBY GIVEN that the special meeting of warrantholders of Triplecrown Acquisition Corp. (Triplecrown), a Delaware corporation, will be held at 10:00 a.m. eastern time, on2009, at the offices of Graubard Miller, Triplecrown s counsel, at The Chrysler Building, 405 Lexington Avenue, 19th Floor, New York, New York 10174. You are cordially invited to attend the meeting, which will be held for the following purposes:

to consider and vote upon separate proposals to amend certain terms of the Warrant Agreement, dated as of October 22, 2007, between Triplecrown and Continental Stock Transfer & Trust Company which governs the terms of Triplecrown s outstanding warrants, in connection with the consummation of the transactions contemplated by the Agreement and Plan of Reorganization, dated as of September 4, 2009, among Triplecrown, Cullen Agricultural Holding Corp. ( CAH ), a wholly-owned subsidiary of Triplecrown, CAT Merger Sub, Inc. ( Triplecrown Merger Sub ), a wholly-owned subsidiary of CAH, Cullen Agricultural Technologies Inc. ( Cullen Agritech ), a Georgia corporation, and Cullen Inc. Holdings Ltd. ( Cullen Holdings ), the sole stockholder of Cullen

- (1) Agritech, which, among other things, provides for (a) the merger of Triplecrown with and into CAH, with CAH being the surviving entity and (b) the merger of Triplecrown Merger Sub with and into Cullen Agritech, with Cullen Agritech being the surviving entity and becoming a wholly owned subsidiary of CAH. The amendments to the Warrant Agreement will provide that (i) the exercise price of Triplecrown s warrants will be increased from \$7.50 per share to \$12.00 per share, (ii) the expiration date of the warrants will be extended from October 21, 2012 to October 21, 2013 and (iii) the price at which the stock must trade for the warrants to be called for redemption will be increased from \$13.75 per share to \$17.00 per share we refer to these proposals collectively as the warrant amendment proposals;
  - to consider and vote upon a proposal to adjourn the special meeting to a later date or dates, if necessary, to permit
- (2) further solicitation and vote of proxies if, based upon the tabulated vote at the time of the special meeting, Triplecrown is not authorized to consummate any of the warrant amendment proposals we refer to this proposal as the warrantholder adjournment proposal.
- These items of business are described in the attached proxy statement/prospectus, which you are encouraged to read in its entirety before voting. Only holders of record of Triplecrown warrants at the close of business on , 2009 are entitled to notice of the special meeting and to vote and have their votes counted at the special meeting and any adjournments or postponements of the special meeting.

The approval of the warrant amendment proposals is a condition to the consummation of the merger discussed above. Triplecrown s officers, directors and stockholders prior to Triplecrown s initial public offering, as well as certain other

warrantholders, representing % of Triplecrown s outstanding warrants, have executed lockup agreements whereby such parties have agreed to vote in favor of the warrant amendment proposals at the special meeting.

After careful consideration, Triplecrown s board of directors has determined that the proposals are fair to and in the best interests of Triplecrown and its warrantholders and recommends that you vote or give instruction to vote FOR the approval of all of the proposals.

All Triplecrown warrantholders are cordially invited to attend the special meeting in person. To ensure your representation at the special meeting, however, you are urged to complete, sign, date and return the enclosed proxy card as soon as possible. If you are a warrantholder of record of Triplecrown, you may also cast your vote in person at the special meeting. If your warrants are held in an account at a brokerage firm or

#### TABLE OF CONTENTS

bank, you must instruct your broker or bank on how to vote your warrants or, if you wish to attend the meeting and vote in person, obtain a proxy from your broker or bank. If you do not vote or do not instruct your broker or bank how to vote, it will have the same effect as voting against the warrant amendment proposals.

A complete list of Triplecrown warrantholders of record entitled to vote at the special meeting will be available for ten days before the special meeting at the principal executive offices of Triplecrown for inspection by warrantholders during ordinary business hours for any purpose germane to the special meeting.

Your vote is important regardless of the number of warrants you own. Whether you plan to attend the special meeting or not, please sign, date and return the enclosed proxy card as soon as possible in the envelope provided. If your warrants are held in street name or are in a margin or similar account, you should contact your broker to ensure that votes related to the warrants you beneficially own are properly counted.

Thank you for your participation. We look forward to your continued support.

, 2009

By Order of the Board of Directors Jonathan J. Ledecky

President, Secretary and Director

IF YOU RETURN YOUR PROXY CARD WITHOUT AN INDICATION OF HOW YOU WISH TO VOTE, YOUR WARRANTS WILL BE VOTED IN FAVOR OF EACH OF THE PROPOSALS. IF THE MERGER IS NOT COMPLETED AND TRIPLECROWN DOES NOT COMPLETE AN INITIAL BUSINESS COMBINATION PRIOR TO OCTOBER 22, 2009, THEN THE WARRANTS WILL EXPIRE WORTHLESS.

The information in this proxy statement/prospectus is not complete and may be changed. We may not issue these securities until the registration statement filed with the Securities and Exchange Commission is effective. This proxy statement/prospectus is not an offer to sell these securities and it is not soliciting an offer to buy these securities in any state where the offer or sale is not permitted.

### SUBJECT TO AMENDMENT AND COMPLETION, DATED SEPTEMBER 4, 2009

# PROXY STATEMENT FOR SPECIAL MEETINGS OF STOCKHOLDERS AND WARRANTHOLDERS

OF

TRIPLECROWN ACQUISITION CORP.

PROSPECTUS FOR UP TO

69,455,000 SHARES OF COMMON STOCK

**AND** 

74,000,000 WARRANTS

**AND** 

74,000,000 SHARES OF COMMON STOCK UNDERLYING SUCH WARRANTS

OF

### CULLEN AGRICULTURAL HOLDING CORP.

Triplecrown Acquisition Corp. ( Triplecrown ) is pleased to report that its board of directors has approved an Agreement and Plan of Reorganization, dated as of September 4, 2009, among Triplecrown, Cullen Agricultural Holding Corp. ( CAH ), a wholly-owned subsidiary of Triplecrown, CAT Merger Sub, Inc. ( Triplecrown Merger Sub ), a wholly-owned subsidiary of CAH, Cullen Agricultural Technologies Inc. ( Cullen Agritech ), a Georgia corporation, and Cullen Inc. Holdings Ltd. ( Cullen Holdings ), the sole stockholder of Cullen Agritech, pursuant to which (i) Triplecrown will merge with and into CAH with CAH surviving the merger and becoming the new publicly-traded corporation of which the present holders of Triplecrown securities will be security holders and (ii) Triplecrown Merger Sub will merge with and into Cullen Agritech with Cullen Agritech surviving the merger and becoming a wholly-owned subsidiary of CAH. We refer to the Agreement and Plan of Reorganization throughout this proxy statement/prospectus as the merger agreement.

CAH is a newly-formed Delaware corporation that will commence operations as a holding company of Cullen Agritech upon completion of the transactions described in this proxy statement/prospectus. Cullen Agritech is a newly formed company committed to the development and commercialization of advanced agricultural technologies. Cullen Agritech will provide advisory services associated with the implementation of efficient farming techniques and promote a methodology that incorporates components of New Zealand s pasture-based farming system. Cullen Agritech s principle focus will be to improve agricultural yields through forage and animal sciences.

Upon completion of the transactions described in this proxy statement/prospectus, the current holders of common stock of Triplecrown will own 57,740,000 shares of CAH stock (assuming no holders of shares of Triplecrown common stock sold in its initial public offering elect to convert their shares into a portion of Triplecrown s trust account and assuming the officers, directors and founders of Triplecrown cancel an aggregate of 11,260,000 shares they will receive upon exchange of their Triplecrown common stock in connection with the merger as described in more detail in this proxy statement/prospectus) and the current holder of common stock of Cullen Agritech will own 15,881,148 shares of CAH common stock. Of the shares of CAH common stock to be owned by the holder of Cullen Agritech common stock, 10% of the shares to be issued in the merger (1,588,115 shares) will be placed in escrow to provide a fund to satisfy indemnification obligations under the merger agreement. See the section entitled *The Merger Proposal Structure of the Merger*.

In connection with the merger, Triplecrown is seeking to amend the terms of the warrant agreement governing its outstanding warrants exercisable for shares of Triplecrown common stock. The amendments to the warrant agreement will provide that (i) the exercise price of Triplecrown s warrants will be increased from \$7.50 per share to \$12.00 per share, (ii) the expiration date of the warrants will be extended from October 21, 2012 to October 21, 2013 and (iii) the price at which the stock must trade for the warrants to be called for redemption will be increased from \$13.75 per share to \$17.00 per share. The approval of the warrant amendment proposals are a condition to the consummation of the merger.

Proposals to approve the merger agreement and the other matters discussed in this proxy statement/prospectus will be presented at the special meetings of stockholders and warrantholders of Triplecrown scheduled to be held on , 2009.

Triplecrown s common stock, units and warrants are currently listed on the NYSE Amex under the symbols TCW, TCW.U and TCW.WS, respectively. Triplecrown s units, common stock and warrants will no longer be traded following consummation of the merger. The parties intend to seek to have the common stock and warrants of CAH listed on the New York Stock Exchange or NASDAQ following consummation of the merger under the symbols and .WS, respectively. However, there is no assurance that the common stock and warrants will be listed on any exchange following consummation of the merger.

This proxy statement/prospectus provides you with detailed information about the merger and other matters to be considered by the Triplecrown stockholders and warrantholders. You are encouraged to carefully read the entire document and the documents

#### **TABLE OF CONTENTS**

incorporated by reference. IN PARTICULAR, YOU SHOULD CAREFULLY CONSIDER THE MATTERS DISCUSSED UNDER RISK FACTORS BEGINNING ON PAGE 14.

**Your vote is very important.** Whether or not you expect to attend the special meetings, the details of which are described on the following pages, please complete, date, sign and promptly return the accompanying proxy in the enclosed envelope.

Neither the Securities and Exchange Commission nor any state securities commission has approved or disapproved of these securities or passed upon the adequacy or accuracy of this proxy statement/prospectus.

Any representation to the contrary is a criminal offense.

The proxy statement/prospectus statement is dated, 2009, and is first being mailed on or about, 2009.

Triplecrown consummated its initial public offering ( IPO ) on October 25, 2007. Citigroup Global Markets Inc. ( Citigroup ) acted as lead manager for the IPO and Jefferies & Company, Ladenburg Thalmann & Co. Inc. and Broadband Capital Management LLC acted as co-managers for the IPO. Upon consummation of the merger, the underwriters in Triplecrown s IPO will be entitled to receive \$ of deferred underwriting commissions, which represents a reduction from the amount of the deferred underwriting commissions they were entitled to under the underwriting agreement executed in connection with the IPO. If the merger is not consummated and Triplecrown is required to be liquidated, the underwriters will not receive any of such funds and such funds will be returned to stockholders upon Triplecrown s liquidation.

# CULLEN AGRICULTURAL HOLDING CORP. TABLE OF CONTENTS

	Page
Summary of the Material Terms of the Merger	1
Questions and Answers for Triplecrown Stockholders and Warrantholders About the Proposals	<u>4</u>
Selected Historical Consolidated Financial Information	<u>11</u>
Selected Unaudited Pro Forma Condensed Combined Financial Information	<u>13</u>
Risk Factors	<u>14</u>
Forward-Looking Statements	<u>25</u>
Special Meeting of Triplecrown Stockholders and Warrantholders	<u> 26</u>
Proposals to be Considered by the Triplecrown Stockholders	<u>31</u>
The Initial Charter Proposals	<u>31</u>
The Merger Proposal	<u>34</u>
The Merger Agreement	<u>59</u>
U.S. Federal Income Tax Considerations to Triplecrown Stockholders	<u>66</u>
<u>Unaudited Pro Forma Condensed Combined Financial Information</u>	<u>67</u>
The Secondary Charter Proposals	<u>75</u>
The Stockholder Adjournment Proposal	<u>77</u>
Proposals to Be Considered by the Triplecrown Warrantholders	<u>78</u>
The Warrant Amendment Proposals	<u>78</u>
The Warrantholder Adjournment Proposal	<u>79</u>
Other Information Related to Triplecrown	<u>80</u>
Business of Cullen Agritech	<u>90</u>
Management of CAH Following the Merger	<u>113</u>
Cullen Agritech s Managements Discussion and Analysis of Financial Condition and Results of	107
<u>Operations</u>	
Beneficial Ownership of Securities	<u>122</u>
Certain Relationships and Related Transactions	<u>125</u>
<u>Description of Securities</u>	<u>128</u>
Price Range of Securities and Dividends	<u>132</u>
Stockholder Proposals	<u>133</u>
<u>Legal Matters</u>	<u>133</u>
<u>Experts</u>	<u>133</u>
<u>Delivery of Documents to Stockholders</u>	<u>134</u>
Where You Can Find More Information	<u>134</u>
<u>Index to Financial Statements</u>	<u>F-1</u>

TABLE OF CONTENTS 14

#### TABLE OF CONTENTS

	Page
ANNEXES	
Agreement and Plan of Reorganization	<u>A-1</u>
CAH Amended and Restated Certificate of Incorporation	<u>B-1</u>
CAH Bylaws	<u>C-1</u>
Employment Agreement	<u>D-1</u>
Form of Escrow Agreement	<u>E-1</u>
Tax Opinion of Graubard Miller	<u>F-0</u>
Triplecrown Amended and Restated Certificate of Incorporation	<u>G-1</u>
2009 Long-Term Incentive Equity Plan	<u>H-1</u>
Opinion of Richards, Layton & Finger, P.A.	<u>I-1</u>
Lock-Up Agreement	<u>J-1</u>
Fairness Opinion of Duff & Phelps	<u>K-1</u>
Section 262 of the Delaware General Corporation Law	<u>L-1</u>
*	

ii

TABLE OF CONTENTS 15

## SUMMARY OF THE MATERIAL TERMS OF THE MERGER

Pursuant to Triplecrown s amended and restated certificate of incorporation and as disclosed in the prospectus included in the registration statement for Triplecrown s IPO, Triplecrown undertook to consummate an initial business combination in which it acquired an operating business in the financial services industry with a fair market value equal to at least 80% of the balance in Triplecrown s trust account (excluding deferred underwriting discounts and commissions) and resulting in the ownership by Triplecrown of not less than 50% of the voting securities of the target business, Additionally, Triplecrown was prohibited from consummating a business combination with a business affiliated with any of Triplecrown s officers, directors and stockholders prior to the IPO ( Triplecrown Founders ). In the proposed merger, (i) the sole stockholder of Cullen Agritech, the entity that will become CAH s operating business following the proposed transaction, is Cullen Holdings, an entity controlled by Eric J. Watson, Triplecrown s chairman of the board and treasurer, (ii) Triplecrown will ultimately be acquired by CAH and therefore not acquire at least 50% of the voting securities of the target business, (iii) Cullen Agritech is not an operating business in the financial services industry and (iv) the fair market value of Cullen Agritech on the date of the transaction is less than 80% of the balance of the trust account. For a more detailed description of the interests of the Triplecrown Founders and other persons having an interest in the transaction, see the section entitled The Merger Proposal Interests of Triplecrown s Directors and Officers and Others in the Merger. Accordingly, the proposed merger does not satisfy the requirements set forth in Triplecrown s amended and restated certificate of incorporation or IPO prospectus. Notwithstanding the foregoing, Triplecrown considered and analyzed numerous companies and acquisition opportunities in its search for an attractive business combination candidate, none of which were believed to be as attractive to public stockholders as the proposed merger. Accordingly, Triplecrown is proposing to amend the terms of its amended and restated certificate of incorporation to allow for the consummation of the proposed transaction. See the section entitled The Initial Charter Proposals.

If the initial charter proposals are approved, stockholders will be asked to adopt the Agreement and Plan of Reorganization and approve the business combination contemplated by such Agreement and Plan of Reorganization. The parties to the Agreement and Plan of Reorganization are Triplecrown, CAH, Triplecrown Merger Sub, Cullen Agritech and Cullen Holdings. Pursuant to the agreement, (i) Triplecrown will merge with and into CAH with CAH surviving the merger and becoming the new publicly-traded corporation of which the present holders of Triplecrown securities will be security holders and (ii) Triplecrown Merger Sub will merge with and into Cullen Agritech with Cullen Agritech surviving the merger and becoming a wholly-owned subsidiary of CAH.

Cullen Agritech is a newly formed company committed to the development and commercialization of advanced agricultural technologies. Cullen Agritech will provide advisory services associated with the implementation of efficient farming techniques and promote a methodology that incorporates components of New Zealand s pasture-based farming system. Cullen Agritech s principle focus will be to improve agricultural yields through forage and animal sciences. See the section entitled *Business of Cullen Agritech*.

Upon completion of the transactions contemplated by this proxy statement/prospectus, the current holders of common stock of Triplecrown will own 57,740,000 shares of CAH stock (assuming no holders of shares of Triplecrown common stock sold in its initial public offering elect to convert their shares into a portion of Triplecrown s trust account and assuming the founders of Triplecrown cancel an aggregate of 11,260,000 shares they will receive upon exchange of their Triplecrown common stock as described in more detail below) and the current holder of common stock of Cullen Agritech will own 15,881,148 shares of CAH common stock. Cullen Holdings has agreed that it will not sell any of the shares it receives in the merger until the 12-month anniversary of the consummation of the merger. To provide a fund for payment to Triplecrown with respect to its post-closing rights to indemnification under the merger agreement, at the closing of the merger, the current holder of common stock of Cullen Agritech will place in escrow (with an independent escrow agent) an aggregate of 1,588,115 shares of CAH common stock it will receive in

the merger, representing 10% of the initial shares to be issued to it in the merger. See the section entitled *The Merger Proposal Indemnification of Triplecrown*.

Prior to Triplecrown s IPO, the Triplecrown Founders acquired 13,800,000 units of Triplecrown (Founders Units), representing 13,800,000 shares of common stock of Triplecrown (Founders Shares) and 13,800,000 warrants to purchase shares of common stock of Triplecrown (Founders Warrants). All of such securities are held in escrow under an escrow agreement with Continental Stock Transfer & Trust Company, pursuant to which such securities will not be released from escrow until one year after the consummation of Triplecrown s initial business combination (except in certain limited situations). In connection with the merger, the Triplecrown Founders have agreed to have cancelled an aggregate of 11,260,000 shares of common stock of CAH they will receive in exchange for 11,260,000 Founders Shares upon consummation of the merger. The remaining 2,540,000 shares, as well as all of the warrants of CAH that the Triplecrown Founders will receive in exchange for their Founders Warrants, will continue to be held in escrow pursuant to the original terms of the escrow agreement and will be released one year after the consummation of the merger.

In connection with the proposed business combination with CAH, Triplecrown and an affiliate of Mr. Watson entered into a contract to purchase a certain piece of land to be used by CAH following consummation of the merger. The total purchase price of the land is \$8,662,500 on which Triplecrown paid a deposit of approximately \$1.8 million. If Triplecrown is unable to complete the proposed business combination by the date it is required to liquidate, Mr. Watson and his affiliate have jointly and severally agreed to assume Triplecrown s remaining obligations under the purchase contract (approximately \$6.9 million) and Triplecrown will have no further obligations or liability whatsoever under the purchase contract.

The merger and the transactions contemplated by the merger agreement are not subject to any additional federal or state regulatory requirement or approval, except for Hart-Scott-Rodino Antitrust Improvements Act of 1976 filings, filings with the State of Delaware necessary to effectuate the merger and the registration statement, of which this proxy statement/prospectus forms a part, shall have become effective and no stop order suspending its effectiveness, or proceeding to that effect, shall have been implemented by the SEC.

Triplecrown has received an opinion from its counsel, Graubard Miller, relating to the tax treatment of the proposed transaction and the effect on Triplecrown's stockholders. Graubard Miller has consented to the use of its opinions in this proxy statement/prospectus. For a detailed description of the material U.S. federal income tax consequences of the merger and warrant amendment, see the section entitled U.S. Federal Income Tax Considerations.

It is possible that the present holders of 30.0% or more of the Public Shares will vote against the merger and seek conversion of their Public Shares into cash in accordance with Triplecrown's amended and restated certificate of incorporation. If such event were to occur, the merger could not be completed. To preclude such possibility Triplecrown, the Triplecrown Founders, CAH, Cullen Agritech and their respective affiliates may enter into arrangements to provide for the purchase of the Public Shares from holders thereof who indicate their intention to vote against the merger and seek conversion or otherwise wish to sell their Public Shares or other arrangements that would induce holders of Public Shares not to vote against the merger proposal. Definitive arrangements have not yet been determined but some possible methods are described in the section titled The Merger Proposal Actions That May Be Taken to Secure Approval of Triplecrown's Stockholders.

At the closing of the merger, the funds in Triplecrown s trust account will be released to pay transaction fees and expenses (including the balance of the purchase price for the land to be used by CAH following consummation of the merger), deferred underwriting discounts and commissions, tax liabilities, if any, and reimbursement of expenses of the Triplecrown Founders and to make purchases of Public Shares, if any. The balance of the funds will be released to CAH to pay Triplecrown stockholders who properly exercise their conversion rights and for working capital and general corporate purposes of CAH and Cullen Agritech. Additionally, if CAH has access to more than \$150 million after closing of the merger, CAH may use such funds, depending on market conditions, to repurchase shares of common stock.

After the merger, the directors of CAH will be Eric J. Watson, , , and , who are designees of Cullen Holdings, and , and , who are designees of Triplecrown. , , and will be considered independent directors under applicable regulatory rules. The officers of CAH will be Eric J. Watson, , and .

In addition to voting on the initial charter proposals and the merger proposal, the stockholders of Triplecrown will vote on proposals to:

approve the following differences between the amended and restated certificate of incorporation of CAH to be in effect following the merger and Triplecrown s amended and restated certificate of incorporation: (i) the name of the new public entity will be Cullen Agricultural Holding Corp. as opposed to Triplecrown Acquisition Corp.; (ii) CAH will have 200,000,000 authorized shares of common stock and 1,000,000 authorized shares of preferred stock and may increase or decrease such amounts without stockholder approval, as opposed to Triplecrown having 160,000,000 authorized shares of preferred stock and may not increase or decrease such amounts without stockholder approval; (iii) CAH s corporate existence will be perpetual as opposed to Triplecrown s corporate existence terminating on October 22, 2009; and (iv) CAH s amended and restated certificate of incorporation will not include the various provisions applicable only to specified purpose acquisition corporations that Triplecrown s amended and restated certificate of incorporation contains. See the section entitled *The Secondary Charter Proposals*.

adjourn the meeting, if necessary. It is possible for Triplecrown to obtain sufficient votes to approve the stockholder adjournment proposal but not receive sufficient votes to approve the other proposals. In such a situation, Triplecrown could adjourn the meeting and attempt to solicit additional votes in favor of such proposals. See the section entitled *The Stockholder Adjournment Proposal*.

Triplecrown is also seeking the approval from the holders of its warrants to amend the terms of the warrant agreement to (i) increase the exercise price of Triplecrown s warrants from \$7.50 per share to \$12.00 per share, (ii) extend the expiration date of the warrants from October 21, 2012 to October 21, 2013 and (iii) increase the price at which the stock must trade for the warrants to be called for redemption from \$13.75 per share to \$17.00 per share. The approval of each of the warrant amendment proposals is a condition to the merger being consummated. The amendments will be effective immediately upon consummation of the merger. The Triplecrown Founders, as well as certain other warrantholders, representing % of Triplecrown s outstanding warrants, have executed lockup agreements whereby such parties have agreed to vote in favor of the warrant amendment proposals at the special meeting. See the section entitled *The Warrant Amendment Proposals*.

In evaluating the proposals described above, you should carefully read this proxy statement/prospectus and especially consider the factors discussed in the section entitled *Risk Factors*.

# QUESTIONS AND ANSWERS FOR TRIPLECROWN STOCKHOLDERS AND WARRANTHOLDERS ABOUT THE PROPOSALS

The following questions and answers briefly address some commonly asked questions about Triplecrown s Special Meetings of Stockholders and Warrantholders. The following questions and answers may not include all the information that is important to stockholders and warrantholders of Triplecrown. We urge stockholders and warrantholders to read carefully this entire proxy statement/prospectus, including the annexes and the other documents referred to herein.

#### A.

Triplecrown has agreed to a business combination under the terms of the Agreement and Plan of Reorganization that is described in this proxy statement/prospectus. A copy of the merger agreement is attached to this proxy statement/prospectus as Annex A, which you are encouraged to read.

Q.

Why am I receiving this proxy statement/ prospectus?

Stockholders are being asked to consider and vote upon proposals entitled *Initial Charter Proposals*, *The Merger Proposal*, *The Secondary Charter Proposals* and *The Stockholder Adjournment Proposal*, all as described in more detail in this proxy statement/prospectus. Warrantholders are being asked to consider and vote upon proposals entitled *The Warrant Amendment Proposals* and *The Warrantholder Adjournment Proposal*, all as described in more detail in this proxy statement/prospectus.

The approval of each of the initial charter proposals, the merger proposal and the warrant amendment proposals is a condition to the consummation of the merger. If any of the initial charter proposals, the merger proposal or the warrant amendment proposals is not approved, the other proposals will not be presented to stockholders and warrantholders for a vote and the merger will not be consummated. The approval of the secondary charter proposals are not a condition to the consummation of the merger and the vote on such proposals will not impact whether the merger is consummated.

This proxy statement/prospectus contains important information about the proposed merger and the other matters to be acted upon at the special meetings. You should read it carefully.

Your vote is important. You are encouraged to vote as soon as possible after carefully reviewing this proxy statement/prospectus.

A.

Q.

Do I have conversion rights?

If you are a holder of Public Shares, you have the right to vote against the merger proposal and demand that Triplecrown convert such shares into a pro rata portion of the trust account in which a substantial portion of the net proceeds of Triplecrown s IPO are held. These rights to vote against the merger and demand conversion of the Public Shares into a pro rata portion of the trust account are sometimes referred to herein as conversion rights.

A.

Q.

How do I exercise my conversion rights?

If you are a holder of Public Shares and wish to exercise your conversion rights, you must (i) vote against the merger proposal, (ii) demand that Triplecrown convert your shares into cash, and (iii) deliver your stock to Triplecrown s transfer agent physically or electronically through the Depository Trust Company prior to the vote at the meeting.

Any action that does not include an affirmative vote against the merger will prevent you from exercising your conversion rights. Your vote on any proposal other than the merger proposal will have no impact on your right to convert.

You may exercise your conversion rights either by checking the box on the proxy card or by submitting your request in writing to Mark Zimkind of Continental Stock Transfer & Trust Company, Triplecrown s transfer agent, at the address listed at the end of this section. If you (i) initially vote for the merger proposal but then wish to vote against it and exercise your conversion rights or (ii) initially vote against the merger proposal and wish to exercise your conversion rights but do not check the box on the proxy card providing for the exercise of your conversion rights or do not send a written request to Triplecrown to exercise your conversion rights, or (iii) initially vote against the merger but later wish to vote for it, you may request Triplecrown to send you another proxy card on which you may indicate your intended vote. You may make such request by contacting Triplecrown at the phone number or address listed at the end of this section.

Any request for conversion, once made, may be withdrawn at any time up to the vote taken with respect to the merger proposal. If you delivered your shares for conversion to Triplecrown s transfer agent and decide prior to the special meeting not to elect conversion, you may request that Triplecrown s transfer agent return the shares (physically or electronically). You may make such request by contacting Triplecrown s transfer agent at the phone number or address listed at the end of this section.

Any corrected or changed proxy card must be received by Triplecrown s secretary prior to the special meeting. No demand for conversion will be honored unless the holder s stock has been delivered (either physically or electronically) to the transfer agent prior to the vote at the meeting.

If the merger is completed, then, if you have also properly exercised your conversion rights, you will be entitled to receive a pro rata portion of the trust account, including any interest earned thereon, calculated as of two business days prior to the date of the consummation of the merger. As of September 4, 2009, there was \$538,796,312 in the trust account, which would amount to approximately \$9.76 per Public Share upon conversion. If you exercise your conversion rights, then you will be exchanging your shares of Triplecrown common stock for cash and will no longer own these shares.

Exercise of your conversion rights does not result in either the exercise or loss of any Triplecrown warrants that you may hold. Your warrants will continue to be outstanding following a conversion of your common stock, will be automatically converted into warrants to purchase shares of CAH s common stock that will have terms that are substantially similar in all material respects to those of the

Triplecrown warrants (subject to the amendments to the warrants contemplated by the warrant amendment proposals) and will become exercisable upon consummation of the merger. A registration statement must be in effect to allow you to exercise any warrants you may hold or to allow CAH to call the warrants for redemption if the redemption conditions are satisfied. If the merger is not consummated and Triplecrown does not complete a different business combination prior to October 22, 2009, the warrants will not become exercisable and will be worthless upon dissolution of Triplecrown in accordance with its charter.

Α

Q.

Do I have appraisal rights if I object to the proposed merger?

mergers contemplated by this proxy statement/prospectus. If appraisal rights are available, holders of shares of Triplecrown common stock who do not vote in favor of the merger proposal and who properly demand appraisal of their shares will be entitled to appraisal rights in connection with the mergers under Section 262 of the DGCL. Holders of Public Shares electing to exercise conversion rights will not be entitled to appraisal rights. For additional information, including the procedures for properly demanding appraisal, see The Merger Proposal Appraisal Rights.

Triplecrown stockholders may have appraisal rights in connection with the

A.

Q.

What happens to the funds deposited in the trust account after consummation of the merger?

Q.

Since Triplecrown s amended and restated certificate of incorporation and IPO prospectus contained certain differences in what is being proposed at the meeting, what are my legal rights?

At the closing of the merger, the funds in Triplecrown s trust account will be released to pay transaction fees and expenses (including the balance of the purchase price for the land to be used by CAH following consummation of the merger), deferred underwriting discounts and commissions, tax liabilities, if any, and reimbursement of expenses of the Triplecrown Founders and to make purchases of Public Shares, if any. The balance of the funds will be released to CAH to pay Triplecrown stockholders who properly exercise their conversion rights and for working capital and general corporate purposes of CAH and Cullen Agritech. Additionally, if CAH has access to more than \$150 million after closing of the merger, CAH may use such funds, depending on market conditions, to repurchase shares of common stock.

Α.

You should be aware that Triplecrown s amended and restated certificate of incorporation and IPO prospectus require Triplecrown to complete a business combination in which it acquires a target business in the financial services industry having a fair market value equal to at least 80% of Triplecrown s trust account balance (excluding deferred underwriting discounts and commissions) and resulting in the ownership by Triplecrown of not less than 50% of the voting securities of the target business and restricts Triplecrown from completing a business combination with a target business that is affiliated with any of the Triplecrown Founders. Furthermore, Triplecrown s IPO prospectus did not disclose that funds in its trust account might be used, directly or indirectly, to purchase Public Shares from holders who have indicated their intention to vote against the merger and seek conversion of their shares to cash (as Triplecrown may contemplate doing). Also, Triplecrown s IPO prospectus stated that specific provisions in Triplecrown s amended and restated certificate of incorporation may not be amended prior to the consummation of an initial business combination but that Triplecrown had been advised that such provision limiting its ability to amend its amended and restated certificate of incorporation may not be enforceable under Delaware law. Accordingly, each person who purchased Public Shares in the IPO and still held such shares upon learning of these facts may have securities law claims against Triplecrown for rescission (under which a successful claimant has the right to receive the total amount paid for his or her

securities pursuant to an allegedly deficient prospectus, plus interest and less any income earned on the securities, in exchange for surrender of the securities) or damages (compensation for loss on an investment caused by alleged material misrepresentations or omissions in the sale of a security).

Such claims may entitle stockholders asserting them to as much as \$10.00 or more per share, based on the initial offering price of the IPO units comprised of stock and warrants, less any amount received from sale of the original warrants, plus interest from the date of Triplecrown s IPO (which, in the case of holders of Public Shares, may be more than the pro rata share of the trust account to which they are entitled on conversion or liquidation). See *The Merger Proposal Rescission Rights*.

Α.

If the merger is not consummated by the date on which Triplecrown must liquidate, either party may terminate the merger agreement. If Triplecrown is unable to complete the merger or another business combination by October 22, 2009, its amended and restated certificate of incorporation provides that it must liquidate. In any liquidation of Triplecrown, the funds deposited in the trust account, plus any interest earned thereon, less claims requiring payment from the trust account by creditors who have not waived their rights against the trust account, if any, will be distributed pro rata to the holders of Triplecrown s Public Shares. Holders of Triplecrown common stock issued prior to the IPO, including all of Triplecrown s officers and directors, have waived any right to any liquidation distribution with respect to those shares. Eric J. Watson, Triplecrown s chairman and treasurer, and Jonathan J. Ledecky, Triplecrown s president and secretary, have agreed to be personally liable under certain circumstances to ensure that the proceeds in the trust account are not reduced by the claims of prospective target businesses and vendors or other entities that are owed money by Triplecrown for services rendered or products sold to it, but only to the extent such entities have not signed a waiver. Triplecrown cannot assure you that Messrs. Watson and Ledecky will be able to satisfy those obligations. See the section entitled Other Information Related to Triplecrown Liquidation If No Business Combination for additional information.

Q.

What happens if the merger is not consummated or the merger agreement is terminated?

Q.

When do you expect the merger to be completed?

Q.

What do I need to do now?

Q. **A.** 

It is currently anticipated that the merger will be consummated promptly following the Triplecrown special meetings on , 2009. For a description of the conditions for the completion of the merger, see the section entitled *The Merger Agreement Conditions to Closing of the Merger*.

A.

Α.

Triplecrown urges you to read carefully and consider the information contained in this proxy statement/prospectus, including the annexes, and to consider how the merger, warrant amendment and other proposals will affect you as a stockholder or warrantholder of Triplecrown. You should then vote as soon as possible in accordance with the instructions provided in this proxy statement/prospectus and on the enclosed proxy card.

How do I vote?

If you are a holder of record of Triplecrown common stock or warrants, you may vote in person at the special meetings or by submitting a proxy for the special meetings. You may submit your proxy by completing, signing, dating and returning the enclosed proxy card in the accompanying pre-addressed postage paid envelope. If you hold your shares or warrants in street name, which means your shares or warrants are held of record by a broker, bank or nominee, you should contact your broker to ensure that votes related to the shares or warrants you beneficially own are properly counted. In this regard, you must provide the record holder of your shares or warrants with instructions on how to vote your shares or warrants or, if you wish to attend the meetings and vote in person, obtain a proxy from your broker, bank or nominee.

instructions to their banks, brokers or other nominees.

A.

Q.

What will happen if I sign and return my proxy card without indicating how I wish to vote?

Signed and dated proxies received by Triplecrown without an indication of how the stockholder or warrantholder intends to vote on a proposal will be voted in favor of each proposal presented to the stockholders or warrantholders, as the case may be.

Stockholders will not be entitled to exercise their conversion rights if such

stockholders return proxy cards to Triplecrown without an indication of how they desire to vote with respect to the merger proposal or, for stockholders holding their shares in street name, if such stockholders fail to provide voting

Q.

If my shares or warrants are held in street name, will my broker, bank or nominee automatically vote my shares or warrants for me? A.

street name, will my No. Your broker, bank or nominee cannot vote your shares or warrants unless you provide instructions on how to vote in accordance with the information and procedures provided to you by your broker, bank or nominee.

A.

Q.

May I change my vote after I have mailed my signed proxy card?

Yes. Send a later-dated, signed proxy card to Triplecrown s secretary at the address set forth below so that it is received by Triplecrown s secretary prior to the special meetings or attend the special meetings in person and vote. You also may revoke your proxy by either sending a notice of revocation to Triplecrown s secretary, which must be received by Triplecrown s secretary prior to the special meetings, or attending the special meetings and revoking your proxy and voting in person.

A.

Upon consummation of the merger, Triplecrown s units will automatically separate and no longer be traded as a separate security.

Q.

What should I do with my stock, warrant and unit certificates?

If you are not electing conversion in connection with your vote on the merger proposal, the merger is approved and consummated, and you hold your securities in Triplecrown in certificate form, as opposed to holding your securities through your broker, you do not need to exchange your existing certificates for certificates issued by CAH. Your current Triplecrown certificates will automatically represent your rights in CAH s securities. You may, however, exchange your certificates if you choose, by contacting CAH s transfer agent, Continental Stock Transfer & Trust Company (Reorganization Department), after the consummation of the merger and following their requirements for reissuance.

Triplecrown stockholders who affirmatively vote against the merger and exercise their conversion rights must deliver their shares to Triplecrown s transfer agent (either physically or electronically) as instructed by Triplecrown or Triplecrown s transfer agent prior to the vote at the meeting.

A.

Q.

What should I do if I receive more than one set of voting materials?

You may receive more than one set of voting materials, including multiple copies of this proxy statement/prospectus and multiple proxy cards or voting instruction cards. For example, if you hold your shares or warrants in more than one brokerage account, you will receive a separate voting instruction card for each brokerage account in which you hold shares or warrants. If you are a holder of record and your shares or warrants are registered in more than one name, you will receive more than one proxy card. Please complete, sign, date and return each proxy card and voting instruction card that you receive in order to cast a vote with respect to all of your Triplecrown shares or warrants.

O. **A.** 

Who can help answer my questions?

If you have questions about the merger or if you need additional copies of the proxy statement/prospectus or the enclosed proxy card you should contact:

Mr. Jonathan J. Ledecky Secretary Triplecrown Acquisition Corp. c/o Paul Vassilakos 590 Madison Avenue, 21st Floor New York, New York 10022 Tel: (212) 521-5396

Tel: (212) 521-5396 Fax: (212) 521-4389

or

Morrow & Co., LLC 470 West Avenue Stamford, Connecticut Tel: (800) 607-0088

You may also obtain additional information about Triplecrown from documents filed with the Securities and Exchange Commission (SEC) by following the instructions in the section entitled *Where You Can Find More Information*.

If you intend to affirmatively vote against the merger and seek conversion of your shares, you will need to deliver your stock (either physically or electronically) to Triplecrown s transfer agent prior to the vote at the meeting.

If you have questions regarding the certification of your position or delivery of your stock, please contact:

Mr. Mark Zimkind Continental Stock Transfer & Trust Company 17 Battery Place New York, New York 10004 Tel: (212) 845-3287

Fax: (212) 616-7616

# SELECTED HISTORICAL CONSOLIDATED FINANCIAL INFORMATION

Triplecrown and CAH are providing the following selected historical financial information to assist you in your analysis of the financial aspects of the merger.

CAH s balance sheet data, statement of financial condition and cash flows data for the period from August 27, 2009 (inception) to August 31, 2009 are derived from CAH s audited financial statements, which are included elsewhere in this proxy statement/prospectus.

Cullen Agritech s balance sheet data, statement of financial condition and cash flows data for the period from June 3, 2009 (inception) to August 18, 2009 are derived from Cullen Agritech s audited financial statements, which are included elsewhere in this proxy statement/prospectus.

Triplecrown s balance sheet data as of December 31, 2008 and December 31, 2007 and statements of income data for the year ended December 31, 2008, and for the period from June 8, 2007 (inception) through December 31, 2007 and 2008 are derived from Triplecrown s audited financial statements, which are included elsewhere in this proxy statement/prospectus. Triplecrown s balance sheet data as of June 30, 2009 and statements of income data for the six months ended June 30, 2009 and 2008 are derived from Triplecrown s unaudited financial statements, which are included elsewhere in this proxy statement/prospectus.

The information is only a summary and should be read in conjunction with each of Triplecrown s, CAH s and Cullen
Agritech s historical financial statements and related notes and Other Information Related to
Triplecrown Triplecrown s Management s Discussion and Analysis of Financial Condition and Results of Operations
and Cullen Agritech s Management s Discussion and Analysis of Financial Condition and Results of Operations
contained elsewhere herein. The historical results included below and elsewhere in this proxy statement/prospectus are
not indicative of the future performance of CAH or Cullen Agritech.

# SELECTED HISTORICAL CONSOLIDATED FINANCIAL INFORMATION CAH (AUDITED)

August 31, 2009

(22,163)

Balance Sheet:

Stockholder s equity

Total assets

Total liabilities 1,000 Stockholder s equity 1,000

# SELECTED HISTORICAL CONSOLIDATED FINANCIAL INFORMATION CULLEN AGRICULTURAL TECHNOLOGIES INC. (AUDITED)

August 18, 2009

Balance Sheet:
Total assets \$ 22,567

Total liabilities 44,730

# SELECTED HISTORICAL FINANCIAL INFORMATION TRIPLECROWN

	For the Six M June 30, (Unaudited)	Ionths Ended	For the Year Ended December 31, 2008 (Audited)	For the Period from June 8, 2007 (Inception) Through December 31, 2007
Income Statement Data:	2009	2008		(Audited)
Revenue Loss from operations Interest and dividend income Net (loss) income attributable to common stockholders Basic and diluted net (loss) income per share	(1,267,364) 172,901 (869,886)	5,506,806	(1,484,138) 8,443,326 4,074,811	(180,350 ) 3,573,186 1,824,397
	(0.02)		0.08	0.07
Weighted average shares outstanding excluding shares subject to possible conversion basic and diluted	52,440,001	52,440,001	52,440,001	26,554,952
Balance Sheet Data:  Working capital Trust account, restricted Total assets Total liabilities Value of common stock which may be rede Stockholders equity	emed for cash	June 30, 2009 3,319,032 539,328,529 543,418,160 770,599 161,798,549 380,849,012	December 31, 2008 4,505,229 538,876,656 543,786,567 404,682 161,662,987 381,718,898	December 31, 2007 1,793,077 536,930,000 540,426,916 1,703,839 161,078,990 377,644,087
	For the Six M June 30,	Months Ended	For the Year Ended December 31,	For the Period from June 8, 2007 (Inception) Through
Cash Flow Data:	2009	2008	2008	December 31, 2007
Net cash (used in) provided by operating activities	(1,383,383)	1,464,793	3,289,954	3,506,866
Net cash (used in) provided by investing activities	766,203	(1,760,678)	(3,475,261)	(540,108,633)
Net cash provided by financing activities	505,574			536,898,680

Net (decrease) increase in cash

(111,606 ) (295,885 ) (185,307 ) 296,913

# SELECTED UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION (USD Except Earning per Share)

	Pro Forma No Conversion	Pro Forma Maximum Conversion
	USD \$	USD \$
Consolidated Pro Forma Income Statement Data for Six months ended June		
30, 2009:		
Net sales	\$	\$
Net loss available to common stockholders	(864,486 )	(916,336)
Basic and diluted net income per share	(0.01)	(0.02)
Shares Outstanding	74,076,148	57,516,149
Weighted average number of shares	74,076,148	57,516,149
Consolidated Pro Forma Income Statement Data for Year ended		
December 31, 2008:		
Net sales	\$	\$
Net loss available to common stockholders	4,152,718	1,611,167
Basic and diluted net income per share	(0.06)	0.03
Shares Outstanding	74,076,148	57,516,149
Weighted average number of shares	74,076,148	57,516,149
Consolidated Pro Forma Balance Sheet Data at June 30, 2009:		
Total assets	\$518,615,154	\$356,952,167
Total liabilities	309,755	309,755
Long-term debt		
Total stockholders' equity	518,305,399	356,642,412

### **RISK FACTORS**

You should carefully consider the following risk factors, together with all of the other information included in this proxy statement/prospectus, before you decide whether to vote or instruct your vote to be cast to approve the proposals described in this proxy statement/prospectus. The risks and uncertainties described below are not the only risks and uncertainties which CAH or Cullen Agritech will face in the future. Additional risks and uncertainties not presently known or that are currently considered to be immaterial may materially and adversely affect the business operations or stock price of CAH or Cullen Agritech following the transactions described in this proxy statement/prospectus. If any of the following risks or uncertainties occurs, CAH s or Cullen Agritech s business, financial condition or operating results could materially suffer. In that event, the trading price of your securities could decline and you may lose all or part of your investment.

## Risks Related to Cullen Agritech s Business and Operations Following the Merger

The value of your investment in CAH following consummation of the merger will be subject to the significant risks affecting CAH and its business, including that of Cullen Agritech. If any of the events described below occur, CAH s post-acquisition business, financial condition, liquidity and/or results of operations could be adversely affected in a material way. This could cause the price of its common stock to decline, perhaps significantly, and you therefore may lose all or part of your investment.

### Risks Related to Cullen Agritech s Business

Cullen Agritech has no operating history and may not be able to successfully operate its business or generate sufficient revenue to make or sustain distributions to its stockholders.

Cullen Agritech was incorporated in June 2009, is a development stage company and has no operating history. Cullen Agritech has no significant assets and will commence operations only upon consummation of the transactions described in this proxy statement/prospectus. Cullen Agritech cannot assure you that it will be able to operate its business successfully or implement its policies and strategies as described in this proxy statement/prospectus. Additionally, the past performance of key personell of Cullen Agritech or its subsidiaries should not be viewed as an indication of the future performance of CAH and Cullen Agritech.

Cullen Agritech s independent registered public accounting firm s report contains an explanatory paragraph that expresses substantial doubt about Cullen Agritech s ability to continue as a going concern.

Cullen Agritech had a net loss of \$22,362 for the period from June 3, 2009 (inception) through August 18, 2009, and has no cash as of August 18, 2009. To date, Cullen Agritech s operations have been funded by Cullen Holdings. Although Cullen Holdings plans to continue to fund the operations as deemed necessary, these factors, among others, raise substantial doubt about Cullen Agritech s ability to continue as a going concern. Cullen Agritech s ability to continue as a going concern is dependent on its ability to finance operations through debt and equity financing. Cullen

Agritech may have difficulty in obtaining financing on favorable terms, if at all. The financial statements contained elsewhere in this prospectus do not include any adjustments that might result from our inability to consummate this offering or our inability to continue as a going concern.

### The recent disruptions in the overall economy and the financial markets may adversely impact Cullen Agritech s business and results of operations.

The dairy industry is sensitive to changes in general economic conditions, both nationally and locally. Recent disruptions in global financial markets and banking systems have made it more difficult for companies to access credit and capital markets. The economic crisis may adversely affect Cullen Agritech in a variety of ways. Access to lines of credit or the capital markets may be severely restricted, which may preclude Cullen Agritech from raising funds required for operations and to fund continued expansion. It may be more difficult for Cullen Agritech to complete strategic transactions with third parties. Continuing volatility in the credit and capital markets could potentially impair Cullen Agritech s customers ability to access these markets and increase associated costs, and Cullen Agritech may be materially affected by these financial market disruptions as economic events and circumstances continue to evolve.

The financial and credit market turmoil could also negatively impact Cullen Agritech s potential suppliers and customers, which could decrease Cullen Agritech s ability to source, produce and distribute Culen Agritech s products and could decrease demand for its products.

While it is not possible to predict with certainty the duration or severity of the current disruption in financial and credit markets, if economic conditions continue to worsen, it is possible these factors could significantly impact Cullen Agritech s financial condition and ability to implement its strategic growth plan.

Any negative public perception regarding Cullen Agritech s products or industry, or any ill effects of product liability claims, could harm Cullen Agritech s reputation, damage its brand, result in costly and damaging recalls and expose Cullen Agritech to government investigations and sanctions, which would materially and adversely affect its results of operations.

Cullen Agritech will sell products for human consumption, which involves a number of risks. Product contamination, spoilage or other adulteration could result in the inability to sell Cullen Agritech s products. Cullen Agritech also may be subject to liability if its products or operations violate applicable laws or regulations or in the event its products cause injury, illness or death. A significant product liability or other legal judgment against Cullen Agritech or a widespread product recall may negatively impact Cullen Agritech s profitability. Even if a product liability or consumer fraud claim is unsuccessful or is not merited, the negative publicity surrounding such assertions regarding Cullen Agritech s products or processes could materially and adversely affect Cullen Agritech s reputation, brand image and results of operations. Finally, serious product quality concerns could result in governmental action against Cullen Agritech, which, among other things, could result in the suspension of production or distribution of Cullen Agritech s products, loss of certain licenses, or other governmental penalties, including possible criminal liability.

If Cullen Agritech is unable to purchase certain assets material to the implementation of its business in a timely manner or at all, such inability would materially adversely affect Cullen Agritech s business and results of operations.

Cullen Agritech is a development stage company with no real operating history, assets or revenues. The implementation of Cullen Agritech s business plan will rely on its ability to purchase land, livestock and other material assets crucial to its business plan. If Cullen Agritech is unable to obtain such assets in a timely manner or at all, such inability will materially adversely affect Cullen Agritech s ability to implement its strategic plan and negatively affect its operating results.

### Cullen Agritech may not realize anticipated benefits from its strategic growth plan.

Cullen Agritech will implement a strategic growth plan, which includes a number of initiatives, that it believes are necessary in order to position Cullen Agritech s business for future success and growth. Over the next several years, these initiatives will require investments in people, systems, tools and facilities. Cullen Agritech s success and earnings growth depends in part on Cullen Agritech s ability to maintain budgeted costs and efficiencies. If Cullen Agritech is unable to successfully implement these initiatives, or fails to implement them as timely as anticipated, Cullen Agritech s results of operations could be adversely impacted.

### Cullen Agritech s business will be subject to various environmental laws, which may increase Cullen Agritech s compliance costs.

Cullen Agritech s business operations will be subject to various environmental and governmental regulations. These laws and regulations cover the discharge of pollutants, wastewater, and hazardous materials into the environment. In addition, various laws and regulations addressing climate change are being considered or implemented at the federal and state levels. New legislation, as well as current federal and other state regulatory initiatives, relating to these environmental matters could require Cullen Agritech to replace equipment, install additional pollution controls, purchase various emission allowances or curtail operations. These costs could adversely affect CAH s results of operations and financial condition.

## Cullen Agritech s operations will be subject to numerous laws and regulations, exposing it to potential claims and compliance costs that could adversely affect its business.

Cullen Agritech will be subject to Federal, state and local laws and regulations relating to the manufacturing, labeling, packaging, health and safety, sanitation, quality control, fair trade practices, and other aspects of our business. In addition, zoning, construction and operating permits are required from governmental

agencies which focus on issues such as land use, environmental protection, waste management, and the movement of animals across state lines. These laws and regulations may, in certain instances, affect our ability to develop and market new products and to utilize technological innovations in our business. In addition, changes in these rules might increase the cost of operating Cullen Agritech s facilities or conducting its business which would adversely affect its finances

Cullen Agritech s dairy business will be affected by Federal price support programs and federal and state pooling and pricing programs to support the prices of certain products we sell. Federal and certain state regulations help ensure that the supply of raw milk flows in priority to fluid milk and soft cream producers before producers of hard products such as cheese and butter. If any of these programs was no longer available to us, the prices we pay for milk could increase and reduce our profitability.

Several states also have laws that restrict the ability of corporations to engage in farming activities. These regulations may require us to alter or restrict our operations or cause us to incur additional costs in order to comply with the regulations.

### Inability to protect Cullen Agritech s trademarks and other proprietary rights could damage its competitive position.

Cullen Agritech will acquire certain intellectual property pursuant to the Deed of Assignment upon completion of the Merger. Cullen Agritech will rely on patents, copyrights, trademarks, trade secrets, know-how confidentiality provisions and licensing arrangements to establish and protect our intellectual property. Any infringement or misappropriation of Cullen Agritech s intellectual property could damage its value and could limit its ability to compete. Cullen Agritech may have to engage in litigation to protect its rights to intellectual property, which could result in significant litigation costs and require a significant amount of management s time.

Cullen Agritech believes that the know-how associated with Cullen Agritech's farming systems for the production of raw milk are trade secrets. In addition, Cullen Agritech has amassed a large body of knowledge regarding animal nutrition and pasture-based farming which it believes to be proprietary. Because most of this proprietary information is not patented, it may be more difficult to protect. Cullen Agritech relies on security procedures and confidentiality agreements to protect this proprietary information, however such agreements and security procedures may be insufficient to keep others from acquiring this information. Any such dissemination or misappropriation of this information could deprive Cullen Agritech of the value of its proprietary information and negatively affect its results.

### Cullen Agritech s proprietary farming system could be replicated creating additional competition in the grass-fed dairy industry.

Despite what Cullen Agritech believes as its first mover advantage and substantial amount of research and development that it believes would be required to replicate its farming system, it is possible that other producers could replicate Cullen Agritech s model with a certain degree of success. This could put Cullen Agritech s market share and competitive advantages at risk.

#### The efficiencies of Cullen Agritech s farming system may not be scaleable.

Although Cullen Agritech believes that additional scale will provide further operating efficiencies to Cullen Agritech s farming system and has no reason to believe that its system is not scalable, it is currently only proved on a farm which

Cullen Agritech s operations will be subject to numerous laws and regulations, exposing it to potential claims and co

is smaller than those farms it is expecting to roll-out in the future. If its system was not as efficient on a larger scale, this could impair Cullen Agritech s ability to implement its strategic plan and negatively affects its operating results.

### Key assets such as land, livestock and infrastructure could increase in price, reducing the ability to roll-out farms under the current capital requirements.

Despite Cullen Agritech s forecasts based on what it believes to be conservative capital budgets, an increase in the cost of Cullen Agritech s key capital items such as land, livestock and infrastructure could reduce Cullen Agritech s ability to roll-out farms with its planned amount of capital. Although Cullen Agritech

does not believe this will occur and it will work with its suppliers to minimize Cullen Agritech s capital costs, key assets may increase substantially in price and additional capital may not be available to Cullen Agritech on acceptable terms when needed.

### The price of land could decrease, reducing the underlying asset value of the business.

Cullen Agritech s current business plan involves buying land assets. If these assets were to be acquired and then the value of these assets decrease, this could reduce the strength of Cullen Agritech s balance sheet in the future and affect its ability to obtain additional capital and implement its business plan.

### Cullen Agritech may establish and maintain relationships with only a small number of co-operatives for the collection and processing of its raw milk.

The dairy processing industry is made up of a number of co-operatives that collect and process all raw milk produced at farms. Cullen Agritech s business plan anticipates that it will establish and maintain relationships with co-operatives for the collection and processing of its raw milk. It is anticipated that it will not initially, if at all, establish contracts with a large number of different co-operatives, which could expose Cullen Agritech to a customer concentration risk.

### Milk and corn price volatility could reduce revenues and negatively affect Cullen Agritech s results of operations.

Although Cullen Agritech will seek to minimize the risk of milk price volatility by employing a milk price hedging strategy (which would include purchase and sale of milk futures as well as forward contracting of Cullen Agritech s milk sales), if the price of milk decreased to that which is substantially lower than expected, this could result in a material reduction in Cullen Agritech s revenues and negatively affect Cullen Agritech s results of operations.

Although a large proportion of Cullen Agritech s livestock s diet will be forage-based, Cullen Agritech s feeding strategy will utilize a certain proportion of other feedstocks, some of which are corn-based, the price of which fluctuates according to the price of corn. Cullen Agritech intends to employ a breeding strategy and a corn price hedging strategy, which would include purchase and sale of corn futures, to reduce Cullen Agritech s exposure to fluctuations in corn prices. However, if corn prices were to rise significantly, Cullen Agritech could experience a material reduction in its operating margins.

## Raw milk production is influenced by a number of factors that are beyond Cullen Agritech s control, such factors may have a material adverse effect on Cullen Agritech s business.

Raw milk production is influenced by a number of factors that are beyond Cullen Agritech s control, including, not limited to, the following:

Seasonal Factors: dairy cows generally produce more milk in temperate weather than in cold or hot weather and extended unseasonably cold or hot weather could lead to lower than expected production;

*Environmental Factors*: the volume and quality of milk produced by dairy cows is closely linked to the quality of the nourishment provided by the environment around them, and, therefore, if environmental

Key assets such as land, livestock and infrastructure could increase in price, reducing the ability to roll-out#arms un

factors cause the quality of nourishment to decline, our milk production could decline; and *Governmental Agricultural and Environmental Policy*: declines in government grants, subsidies, provision of land, technical assistance and other changes in agricultural and environmental policies may have a negative effect on the viability of our farms, and the numbers of dairy cows and quantities of milk they are able to produce.

Such factors could have a material adverse effect on Cullen Agritech s business.

### The milk business is highly competitive and, therefore, Cullen Agritech faces substantial competition in connection with the sale of its products.

Cullen Agritech faces competition from other milk producers across the U.S. Most of Cullen Agritech's competitors are well established, have greater financial, marketing, personnel and other resources, have been

in business for longer periods of time than Cullen Agritech has, and have products that have gained wide customer acceptance in the marketplace. Cullen Agritech may be unable to compete successfully or its competitors may develop products which have superior qualities or gain wider market acceptance than Cullen Agritech s.

### Large-scale disease could harm a significant portion of Cullen Agritech s livestock, reducing its ability to produce revenue.

Although all of Cullen Agritech s livestock will be insured against significant herd loss due to disease and all Cullen Agritech s farms will be run under strict biosecurity regulations to minimize any risk of infection, Cullen Agritech s ability to produce revenue will be dependent on the continual survival and health of Cullen Agritech s livestock. If a significant number of Cullen Agritech s livestock died or were infected with a disease, Cullen Agritech s ability to produce milk would be reduced.

### Cullen Agritech s results of operations fluctuate by season and are affected by weather conditions.

Although Cullen Agritech s farm management strategies can be adapted according to varying weather conditions and all of Cullen Agritech s farms will be purchased in locations in which the prevailing climate is appropriate for Cullen Agritech s business model, any adverse or major deviations from the typical weather conditions expected in a region could negatively impact Cullen Agritech s ability to produce revenue under Cullen Agritech s current strategy. In addition, severe weather conditions and natural disasters, such as floods, droughts, frosts or earthquakes, or adverse growing conditions, diseases and insect-infestation problems may reduce the quantity and quality of its milk production. For example, dairy cows produce less milk when subjected to extreme weather conditions, including hot and cold temperatures. A significant reduction in the quantity or quality of milk produced due to adverse weather conditions, disease, insect problems or other factors could result in increased processing costs and decreased production, with adverse financial consequences to Cullen Agritech.

### A change in the water availability may negatively impact the efficiency of the business model.

Although all Cullen Agritech s land will be acquired in areas where water is readily available, the success of Cullen Agritech s farming system is dependent on the availability of water to successfully grow forage. If there was a reduction in water availability on a farm subsequent to acquiring and converting that property, due to draught, contamination or otherwise, Cullen Agritech s ability to produce milk on that farm could be negatively affected.

#### Grass fed milk might not be valued at a premium in the marketplace.

Although Cullen Agritech currently assumes no premium to be received for its milk and no such premiums are built into the budgets, Cullen Agritech s management believes its milk will ultimately sell for a premium in the marketplace due to higher milk solid content, quality of the milk and compliance with applicable food safety standards. However, trends in organic milk sales show that a product which was previously valued at a premium could lose a significant portion of its market share due to recessionary economic cycle, higher levels of unemployment, higher consumer debt levels, or other unfavorable economic factors. Cullen Agritech cannot provide any assurances that a premium will be received.

## Risks Related to Triplecrown and the Merger and the Securities of CAH Following the Merger

If Triplecrown is unable to effect a business combination and is forced to liquidate, its warrants will expire worthless.

If Triplecrown does not complete the merger or another business combination by October 22, 2009, its amended and restated certificate of incorporation provides that its corporate existence will automatically terminate and it will distribute to all holders of Public Shares, in proportion to the number of Public Shares held by them, an aggregate sum equal to the amount in the trust fund, inclusive of any interest, plus any remaining net assets. In such event, there will be no distribution with respect to Triplecrown s outstanding warrants. Accordingly, the warrants will expire worthless.

## If Triplecrown is forced to liquidate, Triplecrown s stockholders may be held liable for claims by third parties against Triplecrown to the extent of distributions received by them.

Triplecrown s amended and restated certificate of incorporation provides that Triplecrown will continue in existence only until October 22, 2009. If Triplecrown has not completed a business combination by such date and amended this provision in connection thereto, pursuant to the Delaware General Corporation Law (DGCL), its corporate existence will cease except for the purposes of winding up its affairs and liquidating. Under Sections 280 through 282 of the DGCL, stockholders may be held liable for claims by third parties against a corporation to the extent of distributions received by them in a dissolution. If the corporation complies with certain procedures set forth in Section 280 of the DGCL intended to ensure that it makes reasonable provision for all claims against it, including a 60-day notice period during which any third-party claims can be brought against the corporation, a 90-day period during which the corporation may reject any claims brought, and an additional 150-day waiting period before any liquidating distributions are made to stockholders, any liability of stockholders with respect to a liquidating distribution is limited to the lesser of such stockholder s pro rata share of the claim or the amount distributed to the stockholder, and any liability of the stockholder would be barred after the third anniversary of the dissolution. However, it is Triplecrown s intention to make liquidating distributions to its stockholders as soon as reasonably possible after October 22, 2009 and, therefore, it does not intend to comply with those procedures.

Because Triplecrown will not be complying with those procedures, it is required, pursuant to Section 281 of the DGCL, to adopt a plan that will provide for its payment, based on facts known to it at such time, of (i) all existing claims, (ii) all pending claims and (iii) all claims that may be potentially brought against it within the subsequent 10 years. Accordingly, Triplecrown would be required to provide for any creditors known to it at that time or those that it believes could be potentially brought against it within the subsequent 10 years prior to distributing the funds held in the trust to stockholders. Triplecrown cannot assure you that it will properly assess all claims that may be potentially brought against it. As such, Triplecrown s stockholders could potentially be liable for any claims to the extent of distributions received by them (but no more) and any liability of Triplecrown s stockholders may extend well beyond the third anniversary of such date. Accordingly, there can be no assurance that third parties will not seek to recover from Triplecrown s stockholders amounts owed to them by Triplecrown.

If Triplecrown is forced to file a bankruptcy case or an involuntary bankruptcy case is filed against it which is not dismissed, any distributions received by stockholders could be viewed under applicable debtor/creditor and/or bankruptcy laws as either a preferential transfer or a fraudulent conveyance. As a result, a bankruptcy court could seek to recover all amounts received by Triplecrown s stockholders. Furthermore, because Triplecrown intends to distribute the proceeds held in the trust fund to its public stockholders promptly after October 22, 2009 if it has not completed a business combination by such date, this may be viewed or interpreted as giving preference to Triplecrown s public stockholders over any potential creditors with respect to access to or distributions from Triplecrown s assets.

Furthermore, Triplecrown s board may be viewed as having breached their fiduciary duties to Triplecrown s creditors and/or may have acted in bad faith; thereby exposing itself and Triplecrown to claims of punitive damages, by paying public stockholders from the trust fund prior to addressing the claims of creditors. There can be no assurance that claims will not be brought against Triplecrown for these reasons.

### You will not be able to exercise your warrants if an effective registration statement is not in place when you desire to do so.

None of the warrants issued by CAH to the public holders of Triplecrown warrants will be exercisable and CAH will

If Triplecrown is unable to effect a business combination and is forced toliquidate, its warrants will expire \(\psi\_6\)rthless.

not be obligated to issue shares of common stock unless, at the time a holder seeks to exercise such public warrant, a prospectus relating to the common stock issuable upon exercise of the warrant is current. Under the terms of the warrant agreement, CAH will be required to use its best efforts to meet these conditions and to maintain a current prospectus relating to the shares of common stock issuable upon exercise of the warrants until the expiration of the warrants. However, there can be no assurance that CAH will be able to do so, and if it does not maintain a current prospectus related to the shares of common stock issuable upon exercise of the warrants, holders will be unable to exercise their warrants. Additionally, CAH will have no obligation to settle the warrants for cash or net cash settle any warrant exercise. Accordingly, if the prospectus relating to the common stock issuable upon the exercise of the warrants is not current, the warrants

may have no value, the market for the warrants may be limited and the warrants may expire worthless. If the warrants expire worthless, this would mean that a person who paid \$10.00 for a unit in Triplecrown s IPO and who did not sell the warrant included in the unit would have effectively paid \$10.00 for one share of CAH common stock.

# An investor will only be able to exercise a warrant if the issuance of CAH shares of common stock upon such exercise has been registered or qualified or is deemed exempt under the securities laws of the state of residence of the holder of the warrants.

None of the warrants issued by CAH to the holders of Triplecrown warrants will be exercisable and CAH will not be obligated to issue shares of common stock unless the shares of common stock issuable upon such exercise have been registered or qualified or deemed to be exempt under the securities laws of the state of residence of the holder of the warrants. At the time that the warrants become exercisable (following completion of the merger), CAH expects to become listed on the New York Stock Exchange or NASDAQ, which would provide an exemption from registration in every state. Accordingly, CAH believes holders in every state will be able to exercise their warrants as long as its prospectus relating to the shares of common stock issuable upon exercise of the warrants is current. However, there can be no assurance of this fact. If a warrant holder is unable to exercise his warrants in a particular state, he may be forced to sell his warrant and therefore lose the benefit of purchasing CAH stock. Furthermore, the price he receives for his warrant may not equal the difference between the exercise price and the stock price.

### CAH s warrants may be exercised in the future, which would increase the number of shares eligible for future resale in the public market.

Following consummation of the merger, outstanding warrants to purchase an aggregate of 55,200,000 shares of CAH common stock (issued in exchange for Triplecrown warrants issued in the IPO), warrants to purchase an aggregate of 13,800,000 shares of common stock (issued in exchange for the Founders Warrants) and 5,000,000 shares of common stock (issued in exchange for the warrants sold to the Triplecrown Founders simultaneously with the consummation of the IPO (Sponsors Warrants)) will become exercisable after the consummation of the merger. To the extent such warrants are exercised, additional shares of CAH common stock will be issued, which would dilute the ownership of existing stockholders.

## Triplecrown s stockholders will experience immediate dilution as a consequence of the issuance of common stock as consideration in the merger.

CAH will issue 15,881,148 shares of common stock at the closing of the merger to Cullen Holdings. As a result, holders of Triplecrown common stock will experience immediate dilution as a consequence. The ability of the former Triplecrown stockholders following the merger to influence management of CAH through the election of directors will be reduced as a result of this dilution.

If Triplecrown stockholders fail to vote against the merger proposal and fail to deliver their shares in accordance with the conversion requirements specified in this proxy statement/prospectus, they will not be entitled to convert their shares of common stock of Triplecrown into a pro rata portion of the trust

#### account.

Triplecrown stockholders holding Public Shares who affirmatively vote against the merger proposal may demand that Triplecrown convert their shares into a pro rata portion of the trust account, calculated as of two business days prior to the anticipated consummation of the merger. Triplecrown stockholders who seek to exercise this conversion right must affirmatively vote against the merger and deliver their stock (either physically or electronically) to Triplecrown s transfer agent prior to the vote at the meeting. Any Triplecrown stockholder who fails to vote against the merger proposal and who fails to deliver his or her stock will not be entitled to convert his or her shares into a pro rata portion of the trust account for conversion of his or her shares. See the section entitled *The Merger Proposal Conversion Rights* for the procedures to be followed if you wish to convert your shares to cash.

Triplecrown s ability to request indemnification from Cullen Holdings for damages arising out of the merger is limited to those claims where damages exceed \$300,000 and is also limited to the shares of CAH s common stock placed in escrow.

To provide a fund to secure the indemnification obligations of Cullen Holdings to Triplecrown against losses that the surviving entity of the merger may sustain as a result of (i) the inaccuracy or breach of any

representation or warranty made by Cullen Agritech in the merger agreement or any schedule or certificate delivered by it in connection with the merger agreement and (ii) the non-fulfillment or breach of any covenant or agreement made by Cullen Agritech in the merger agreement, an aggregate of 10% of the initial shares to be issued in the merger (1,588,115 shares of CAH common stock) will be placed in escrow (with an independent escrow agent), which will be canceled to the extent that Triplecrown has damages for which it is entitled to indemnification. The escrow will be the sole remedy for Triplecrown for its rights to indemnification pursuant to the merger agreement. Claims for indemnification may be asserted against the escrow by Triplecrown once its damages exceed a deductible and will be reimbursable to the full extent of the damages in excess of such amount up to a maximum of the escrowed shares. Triplecrown may assert claims for indemnification until the 30th day after the date CAH has filed with the SEC its Annual Report on Form 10-K for the year ending December 31, 2010. As a consequence of these limitations, Triplecrown may not be able to be entirely compensated for indemnifiable damages that it may sustain.

## The New York Stock Exchange or NASDAQ may not list CAH s securities on its exchange, which could limit investors ability to make transactions in CAH s securities and subject CAH to additional trading restrictions.

CAH will seek listing of its common stock and warrants on the New York Stock Exchange or NASDAQ as soon as practicable in connection with the merger. CAH will be required to meet the initial listing requirements to be listed. CAH may not be able to meet those initial listing requirements. Even if such application is accepted and CAH s securities are so listed, CAH may be unable to maintain the listing of its securities in the future.

If either the New York Stock Exchange or NASDAQdo not list CAH s securities for trading on its exchange, CAH could face significant material adverse consequences, including:

a limited availability of market quotations for its securities; reduced liquidity with respect to its securities;

a determination that its shares of common stock are penny stock, which will require brokers trading in its shares of common stock to adhere to more stringent rules, possibly resulting in a reduced level of trading activity in the secondary trading market for the shares of common stock;

a limited amount of news and analyst coverage for CAH; and a decreased ability to issue additional securities or obtain additional financing in the future.

### CAH s stock price could fluctuate and could cause you to lose a significant part of your investment.

Following consummation of the merger, the market price of CAH s securities may be influenced by many factors, some of which are beyond its control, including those described above and the following:

changes in financial estimates by analysts;

if the benefits of the merger do not meet the expectations of financial and marketing analysts; fluctuations in its quarterly financial results or the quarterly financial results of companies perceived to be similar to it:

general economic conditions; changes in market valuations of similar companies;

terrorist acts;

changes in its capital structure, such as future issuances of securities or the incurrence of additional debt; future sales of its common stock;

Triplecrown s ability to request indemnification from Cullen Holdings fordamages arising out of the merge 10s limited

regulatory developments in the United States, foreign countries or both; litigation involving CAH, its subsidiaries or its general industry; and

additions or departures of key personnel.

### Future sales of CAH s common stock may cause the market price of its securities to drop significantly, even if its business is doing well.

In accordance with the lock up agreement to be executed in connection with the consummation of the merger, Cullen Holdings will be able to sell the shares of common stock of CAH it receives in the merger beginning on the first anniversary of the consummation of the merger. In addition, the Triplecrown Founders will be entitled to demand that CAH register the resale of its securities at any time generally commencing nine months after the consummation of the merger. In accordance with SEC regulations, the Triplecrown Founders will not be able to sell any of the CAH securities they receive in exchange for their Triplecrown securities until the first anniversary of the consummation of the merger, subject to certain exceptions. The presence of these additional securities trading in the public market may have an adverse effect on the market price of CAH s securities. The sale by any of the foregoing, or entities they control or their permitted transferees, could cause the market price of CAH s securities to decline.

## CAH may apply the net proceeds released from the trust account in a manner that does not improve its results of operations or increase the value of your investment.

At the closing of the merger, the funds in Triplecrown s trust account will be released to pay transaction fees and expenses (including the balance of the purchase price for the land to be used by CAH following consummation of the merger), deferred underwriting discounts and commissions, tax liabilities, if any, and reimbursement of expenses of the Triplecrown Founders and to make purchases of Public Shares, if any. The balance of the funds will be released to CAH to pay Triplecrown stockholders who properly exercise their conversion rights and for working capital and general corporate purposes of CAH and Cullen Agritech. Additionally, if CAH has access to more than \$150 million after closing of the merger, CAH may use such funds, depending on market conditions, to repurchase shares of common stock. Other than these uses, CAH does not have specific plans for the funds and will have broad discretion regarding how it uses such funds. These funds could be used in a manner with which you may not agree or applied in ways that do not improve CAH s results of operations or increase the value of your investment.

# Triplecrown s current directors and executive officers own securities of Triplecrown that will be worthless if the merger is not approved. Such interests may have influenced their decision to approve the business combination with CAH.

Triplecrown s Founders beneficially own Founders Units that they purchased prior to its IPO and 5,000,000 Sponsors Warrants they purchased in a private placement that occurred simultaneously with Triplecrown s IPO. Such persons are not entitled to receive any of the cash proceeds that may be distributed upon Triplecrown s liquidation with respect to shares they acquired prior to its IPO. Therefore, if the merger is not approved and Triplecrown does not consummate another business combination by October 22, 2009 and is forced to liquidate, such Founders Units and Sponsors Warrants held by such persons will be worthless. As of , 2009, the record date for the special meeting, Triplecrown s Founders held \$ in units (based on a market price of \$) and \$ in warrants (based on a market price of \$). The Triplecrown Founders have agreed to contribute an aggregate of 11,260,000 of the 13,800,000 the Founders Shares to Triplecrown for cancellation. Furthermore, Eric J. Watson is the chief executive officer of Cullen Agritech and of Cullen Holdings. As a result, he will benefit to the extent of the 15,881,148 shares being received by Cullen Holdings in the merger. See the section entitled *The Merger Proposal Interests of Triplecrown s Directors and* 

Officers and Others in the Merger.

These financial interests of Triplecrown s Founders may have influenced their decision to approve Triplecrown s merger with CAH and to continue to pursue the merger. In considering the recommendations of Triplecrown s board of directors to vote for the merger proposal and other proposals, you should consider these interests.

# Triplecrown s chairman and president are liable to ensure that proceeds of the trust are not reduced by vendor claims in the event the merger is not consummated. Such liability may have influenced their decision to approve the merger with CAH.

If Triplecrown liquidates prior to the consummation of a business combination, Eric J. Watson, Triplecrown s chairman and treasurer, and Jonathan J. Ledecky, Triplecrown s president and secretary, will be personally liable to ensure that the proceeds in the trust account are not reduced by the claims of target businesses or claims of vendors or other entities that are owed money by Triplecrown for services rendered or contracted for or products sold to Triplecrown, but only if such entities did not execute a waiver. On the other hand, if Triplecrown consummates the merger, CAH will be liable for all such claims. Neither Triplecrown nor Messrs. Watson and Ledecky has any reason to believe that Messrs. Watson and Ledecky will not be able to fulfill their indemnity obligations to Triplecrown if required to do so.

Additionally, if Triplecrown is required to be liquidated and there are no funds remaining to pay the costs associated with the implementation and completion of such liquidation, Messrs. Watson and Ledecky have agreed to advance Triplecrown the funds necessary to pay such costs and complete such liquidation (currently anticipated to be no more than approximately \$15,000) and not to seek repayment for such expenses. If Triplecrown consummates the merger, Messrs. Watson and Ledecky will no longer be responsible for such expenses. See the section entitled *Other Information Related to Triplecrown Triplecrown s Management s Discussion and Analysis of Financial Condition and Results of Operations* for further information.

These personal obligations may have influenced Messrs. Watson s and Ledecky s decision to approve Triplecrown s merger with CAH and to continue to pursue the merger. In considering the recommendations of Triplecrown s board of directors to vote for the merger proposal and other proposals, you should consider these interests.

## The fairness opinion Triplecrown received assumed that Cullen Agritech had access to the capital it needed to implement its business plan and therefore Cullen Agritech may not be worth as much as such valuation supposes.

In connection with its determination to approve the transactions contemplated by this proxy statement/prospectus,

Triplecrown s board of directors engaged Duff & Phelps to provide it with a fairness opinion on which the board
relied. The purpose of the fairness opinion was to determine whether the merger consideration to be paid by

Triplecrown is fair, from a financial point of view, to Triplecrown. In coming to a valuation range for Cullen Agritech,
Duff & Phelps assumed that Cullen Agritech would have access to the capital it needed to implement its business
plan. As a result of purchases of Public Shares and holders of Public Shares exercising their conversion rights, Cullen
Agritech may not have access to all the capital it needs from the consummation of our transaction. Accordingly, if
Cullen Agritech does not have access to capital other than through this transaction, Cullen Agritech may not be worth
the value that Duff & Phelps placed on it.

The exercise of Triplecrown s directors and officers discretion in agreeing to changes or waivers in the terms of the merger may result in a conflict of interest when determining whether such changes to the terms of the merger or waivers of conditions are appropriate and in Triplecrown s stockholders

#### best interest.

In the period leading up to the closing of the merger, events may occur that, pursuant to the merger agreement, would cause Triplecrown to agree to amend the merger agreement, to consent to certain actions taken by CAH or to waive rights that Triplecrown is entitled to under the merger agreement. Such events could arise because of a request by CAH to undertake actions that would otherwise be prohibited by the terms of the merger agreement or the occurrence of other events that would have a material adverse effect on CAH s business and would entitle Triplecrown to terminate the merger agreement. In any of such circumstances, it would be discretionary on Triplecrown, acting through its board of directors, to grant its consent or waive its rights. The existence of the financial and personal interests of the directors described in the preceding risk factors may result in a conflict of interest on the part of one or more of the directors between what he may believe is best for Triplecrown and what he may believe is best for himself in determining whether or not to take the requested action. As of the date of this proxy statement/prospectus,

Triplecrown does not believe

there will be any changes or waivers that its directors and officers would be likely to make after stockholder approval of the merger proposal has been obtained. Although certain changes could be made without further stockholder approval, Triplecrown will circulate a new or amended proxy statement/prospectus and resolicit its stockholders if changes to the terms of the transaction that would have a material impact on its stockholders are required prior to the stockholder vote on the merger proposal.

If the merger is completed, a large portion of the funds in the trust account established by Triplecrown in connection with its IPO for the benefit of the holders of the Public Shares may be used to pay converting stockholders or for the purchase, directly or indirectly, of Public Shares. As a consequence, if the merger is completed, such funds will not be available to CAH for working capital and general corporate purposes and the number of beneficial holders of Triplecrown s and CAH s securities may be reduced to a number that may preclude the quotation, trading or listing of CAH s securities other than on the Over-the-Counter Bulletin Board.

Pursuant to Triplecrown s amended and restated certificate of incorporation, holders of Public Shares may vote against the merger proposal and demand that Triplecrown convert their shares, calculated as of two business days prior to the anticipated consummation of the merger, into a pro rata share of the trust account where a substantial portion of the net proceeds of the IPO are held. Triplecrown will not consummate the merger if holders of 16,559,999 or more Public Shares exercise these conversion rights. Furthermore, a large portion of the funds in the trust account may be used to acquire Public Shares, either from holders thereof who vote against the merger proposal and elect to convert their shares into cash or from holders thereof who have indicated their intention to do so but first sell their shares to CAH, Triplecrown or their affiliates so that such shares will be voted in favor of the merger proposal. As a consequence of such purchases,

the funds in Triplecrown s trust account that are so used will not be available to CAH after the merger and the actual amount of such funds that CAH may retain for its own use will be diminished; and the public float of CAH s common stock may be reduced and the number of beneficial holders of Triplecrown s and CAH s securities may be reduced, which may make it difficult to obtain the quotation, listing or trading of CAH s securities on the New York Stock Exchange or any other national securities exchange.

Persons who purchased Public Shares in the IPO may have rights to rescind their purchases and assert a claim for damages therefor against CAH, its directors and officers and the former directors and officers of Triplecrown.

Triplecrown s amended and restated certificate of incorporation and IPO prospectus require Triplecrown to complete a business combination in which it acquires a target business operating in the financial services industry having a fair market value equal to at least 80% of Triplecrown s trust account balance (excluding deferred underwriting discounts and commissions) and resulting in the ownership of 50% or more of the voting securities of the target business and prohibit Triplecrown from consummating a business combination with a target business affiliated with any of the Triplecrown Founders. Furthermore, Triplecrown s IPO prospectus did not disclose that funds in its trust account might be used, directly or indirectly, to purchase Public Shares from holders who have indicated their intention to vote against the merger and seek conversion of their shares to cash (as Triplecrown may contemplate doing). Also, Triplecrown s IPO prospectus stated that specific provisions in Triplecrown s amended and restated certificate of

incorporation may not be amended prior to the consummation of an initial business combination but that Triplecrown had been advised that such provision limiting its ability to amend its amended and restated certificate of incorporation may not be enforceable under Delaware law. Consequently, any person who purchased Public Shares in the IPO and still held such shares upon learning of these facts may seek rescission of the purchase of the units he acquired in the IPO (under which a successful claimant has the right to receive the total amount paid for his or her securities pursuant to an allegedly deficient prospectus, plus interest and less any income earned on the securities, in exchange for surrender of the securities) or bring an action for damages (compensation for loss on an investment caused by alleged material misrepresentations or omissions in the sale of a security). Any holder of Public Shares who votes against the merger proposal and converts Public Shares into cash upon consummation of the merger will no longer own any stock in Triplecrown and therefore will not have any claim for rescission.

#### FORWARD-LOOKING STATEMENTS

Triplecrown, CAH and Cullen Agritech believe that some of the information in this proxy statement/prospectus constitutes forward-looking statements within the definition of the Private Securities Litigation Reform Act of 1995. You can identify these statements by forward-looking words such as may, expect, anticipate, contemplate, belief estimate, intends, and continue or similar words. You should read statements that contain these words carefully because they:

discuss future expectations; contain projections of future results of operations or financial condition; or state other forward-looking information.

Triplecrown, CAH and Cullen Agritech believe it is important to communicate their expectations to their stockholders. However, there may be events in the future that they are not able to predict accurately or over which they have no control. The risk factors and cautionary language discussed in this proxy statement/prospectus provide examples of risks, uncertainties and events that may cause actual results to differ materially from the expectations described by Triplecrown, CAH or Cullen Agritech in such forward-looking statements, including among other things:

Triplecrown s ability to complete its initial business combinaiton within the specified time limits; officers and directors allocating their time to other businesses and potentially having conflicts of interest with Triplecrown s business or in approving the merger;

the number and percentage of Triplecrown s stockholders voting against the merger proposal and seeking conversion; changes adversely affecting the business in which CAH and Cullen Agritech will be engaged; success in retaining or recruiting, or changes required in, the Company s officers, key employees or directors following the merger;

management of growth; general economic conditions; CAH s and Cullen Agritech s business strategy and plans;

the result of future financing efforts;

delisting of Triplecrown s securities from the NYSE Amex or the ability to have CAH s securities listed on the New York Stock Exchange or NASDAQ or another exchange following the merger; and

the potential liquidity and trading of Triplecrown s and CAH s public securities.

Should one or more of these risks or uncertainties materialize, or should any of the underlying assumptions prove incorrect, actual results may vary in material respects from those expressed or implied by these forward-looking statements.

You are cautioned not to place undue reliance on these forward-looking statements, which speak only as of the date of this proxy statement/prospectus.

All forward-looking statements included herein attributable to any of Triplecrown, CAH, Cullen Agritech or any person acting on such parties behalf are expressly qualified in their entirety by the cautionary statements contained or referred to in this section. Except to the extent required by applicable laws and regulations, Triplecrown and CAH undertake no obligations to update these forward-looking statements to reflect events or circumstances after the date of this proxy statement/prospectus or to reflect the occurrence of unanticipated events.

Before you grant your proxy or instruct how your vote should be cast or vote on the merger proposal or any of the other proposals, you should be aware that the occurrence of the events described in the *Risk Factors* section and elsewhere in this proxy statement/prospectus may adversely affect Triplecrown, CAH and/or Cullen Agritech.

## SPECIAL MEETINGS OF TRIPLECROWN STOCKHOLDERS AND WARRANTHOLDERS

#### General

Triplecrown is furnishing this proxy statement/prospectus to its stockholders and warrantholders as part of the solicitation of proxies by its board of directors for use at the special meetings of Triplecrown stockholders and warrantholders to be held on , 2009, and at any adjournment or postponement thereof. This proxy statement/prospectus is first being furnished to Triplecrown stockholders and warrantholders on or about , 2009 in connection with the vote on the proposals described herein. This proxy statement/prospectus provides you with information you need to know to be able to vote or instruct your vote to be cast at the special meetings.

#### **Date, Time and Place**

The special meetings of stockholders and warrantholders will be held on , 2009, at 10:00 a.m., eastern time, at the offices of Graubard Miller, Triplecrown s counsel, at The Chrysler Building, 405 Lexington Avenue, 19th Floor, New York, New York 10174, or such other date, time and place to which such meetings may be adjourned or postponed.

#### **Purpose of the Triplecrown Special Meetings**

At the special meeting of stockholders, Triplecrown will ask holders of its common stock to:

consider and vote upon separate proposals to amend Triplecrown s amended and restated certificate of incorporation to allow Triplecrown to complete the merger with CAH notwithstanding that: (i) Cullen Agritech, the entity that will be CAH s operating business following the proposed business combination, is owned by Cullen Holdings, an affiliate of Eric J. Watson, Triplecrown s chairman of the board and treasurer, (ii) Triplecrown will ultimately be acquired by CAH and therefore not acquire at least 50% of the voting securities of the target business, (iii) Cullen Agritech is not an operating business in the financial services industry and (iv) the fair market value of Cullen Agritech on the date of the transaction is less than 80% of the balance of Triplecrown s trust account (the initial charter proposals); consider and vote upon a proposal to (i) adopt the Agreement and Plan of Reorganization, dated as of September 4, 2009, among Triplecrown, CAH, Triplecrown Merger Sub, Cullen Agritech and Cullen Holdings which, among other things, provides for (a) the merger of Triplecrown with and into CAH, with CAH being the surviving entity and (b) the merger of Triplecrown Merger Sub with and into Cullen Agritech, with Cullen Agritech being the surviving entity and becoming a wholly owned subsidiary of CAH, and (ii) approve the business combination contemplated by such Agreement and Plan of Reorganization (the merger proposal);

consider and vote upon separate proposals to approve the following differences between the amended and restated certificate of incorporation of CAH to be in effect following the merger and Triplecrown s current amended and restated certificate of incorporation: (i) the name of the new public entity will be Cullen Agricultural Holding Corp. as opposed to Triplecrown Acquisition Corp.; (ii) CAH will have 200,000,000 authorized shares of common stock and 1,000,000 authorized shares of preferred stock and may increase or decrease such amounts without stockholder

approval, as opposed to Triplecrown having 160,000,000 authorized shares of common stock and 1,000,000 authorized shares of preferred stock and may not increase or decrease such amounts without stockholder approval; (iii) CAH s corporate existence will be perpetual as opposed to Triplecrown s corporate existence terminating on October 22, 2009; and (iv) CAH s amended and restated certificate of incorporation will not include the various provisions applicable only to specified purpose acquisition corporations that Triplecrown s amended and restated certificate of incorporation contains (the secondary charter proposals); and

consider and vote upon a proposal to adjourn the special meeting to a later date or dates, if necessary, to permit further solicitation and vote of proxies if, based upon the tabulated vote at the time of the special meeting, Triplecrown is not authorized to consummate the merger (the stockholder adjournment proposal).

At the special meeting of warrantholders, Triplecrown will ask holders of its warrants to:

in connection with the transactions contemplated by the Agreement and Plan of Reorganization, consider and vote upon separate proposals to amend the Warrant Agreement, dated as of October 22, 2007, between Triplecrown and Continental Stock Transfer & Trust Company which governs the terms of Triplecrown s outstanding warrants to (i) increase the exercise price of Triplecrown s warrants from \$7.50 per share to \$12.00 per share, (ii) extend the expiration date of the warrants from October 21, 2012 to October 21, 2013 and (iii) increase the price at which the stock must trade for the warrants to be called for redemption from \$13.75 per share to \$17.00 per share (the warrant amendment proposals); and

consider and vote upon a proposal to adjourn the special meeting to a later date or dates, if necessary, to permit further solicitation and vote of proxies if, based upon the tabulated vote at the time of the special meeting, Triplecrown is not authorized to consummate the warrant amendment proposals (the warrantholder adjournment proposal).

#### **Recommendation of Triplecrown Board of Directors**

Triplecrown s board of directors:

has determined that each of the proposals is fair to and in the best interests of Triplecrown and its stockholders and warrantholders:

has approved each of the proposals;

recommends that Triplecrown s common stockholders vote FOR the initial charter proposals; recommends that Triplecrown s common stockholders vote FOR the merger proposal; recommends that Triplecrown s common stockholders vote FOR the secondary charter proposals; recommends that Triplecrown s common stockholders vote FOR the stockholder adjournment proposal; recommends that Triplecrown s warrantholders vote FOR the warrant amendment proposals; and recommends that Triplecrown s warrantholders vote FOR the warrantholder adjournment proposal.

#### Record Date; Who is Entitled to Vote

Triplecrown has fixed the close of business on , 2009, as the record date for determining Triplecrown stockholders and warrantholders entitled to notice of and to attend and vote at its special meetings. As of the close of business on , 2009, there were 69,000,000 shares of Triplecrown s common stock outstanding and entitled to vote and 74,000,000 warrants outstanding and entitled to vote. Each share of Triplecrown s common stock is entitled to one vote per share at the special meeting of stockholders and each warrant is entitled to one vote per warrant at the special meeting of warrantholders.

Pursuant to agreements with Triplecrown, the 13,800,000 Founders Shares held by the Triplecrown Founders will be voted on the merger proposal in accordance with the majority of the votes cast at the special meeting of stockholders on such proposal by the holders of the Public Shares. Accordingly, the vote of such shares will not affect the outcome of the vote on the merger proposal.

The Triplecrown Founders, as well as certain other warrantholders, representing % of Triplecrown s outstanding warrants, have executed lockup agreements whereby such parties have agreed to vote in favor of the warrant amendment proposals at the special meeting.

The Triplecrown Founders have indicated that they intend to vote their Founders Shares, Founders Warrants and Sponsors Warrants in favor of all of the other proposals being presented at the meeting.

#### Quorum

A quorum of stockholders and warrantholders is necessary to hold a valid stockholders meeting and warrantholders meeting, respectively. The presence, in person or by proxy, of a majority of all the outstanding shares of common stock entitled to vote constitutes a quorum at the special meeting of stockholders. The presence, in person or by proxy, of a majority Triplecrown s then outstanding warrants constitutes a quorum at the special meeting of warrantholders.

#### **Abstentions and Broker Non-Votes**

Proxies that are marked abstain and proxies relating to street name shares or warrants that are returned to Triplecrown but marked by brokers as not voted will be treated as shares or warrants present for purposes of determining the presence of a quorum on all matters. The latter will not be treated as shares or warrants entitled to vote on the matter as to which authority to vote is withheld from the broker. If you do not give the broker voting instructions, under applicable self-regulatory organization rules, your broker may not vote your shares or warrants on non-routine proposals, such as the initial charter proposals, the merger proposal, the secondary charter proposals and the warrant amendment proposals. Since a stockholder must affirmatively vote against the merger proposal to have conversion rights, individuals who fail to vote or who abstain from voting may not exercise their conversion rights. See the information set forth in *The Merger Proposal Conversion Rights*.

#### Vote of Triplecrown s Stockholders Required

The approval of the merger proposal will require (i) the affirmative vote of the holders of a majority of Triplecrown common stock outstanding on the record date and (ii) the affirmative vote for the proposal by the holders of a majority of the Public Shares present (in person or represented by proxy) and entitled to vote on the proposal at the meeting. Additionally, the merger will not be consummated if the holders of 30% or more of the Public Shares vote against the merger proposal and properly demand that Triplecrown convert their Public Shares into their pro rata share of the trust account. Abstentions and broker non-votes will have the same effect as a vote AGAINST the merger proposal. You cannot seek conversion unless you affirmatively vote against the merger proposal.

Each of the initial charter proposals and secondary charter proposals will require the affirmative vote of the holders of a majority of Triplecrown common stock outstanding on the record date. Because these proposals require the affirmative vote of a majority of the shares of common stock outstanding for approval, abstentions and shares not entitled to vote because of a broker non-vote will have the same effect as a vote against these proposals.

The approval of the stockholder adjournment proposal will require the affirmative vote of the holders of a majority of Triplecrown s common stock represented and entitled to vote thereon at the meeting. Abstentions are deemed entitled to vote on such proposal. Therefore, they have the same effect as a vote against the proposal. Broker non-votes are not deemed entitled to vote on such proposals and, therefore, they will have no effect on the vote on such proposal.

The approval of the initial charter proposals, the merger proposal and the warrant amendment proposals is a condition to the consummation of the merger. If any of the initial charter proposals, the merger proposal or warrant amendment proposals is not approved, the other proposals will not be presented to stockholders and/or warrantholders for a vote and the merger will not be consummated.

Quorum 64

#### Vote of Triplecrown s Warrantholders Required

The approval of the warrant amendment proposals will require the affirmative vote of the holders of a majority of Triplecrown s warrants outstanding on the record date. Because these proposals require the affirmative vote of a majority of the warrants outstanding for approval, abstentions and warrants not entitled to vote because of a broker non-vote will have the same effect as a vote against these proposals.

The approval of the warrantholder adjournment proposal will require the affirmative vote of the holders of a majority of Triplecrown s warrants represented and entitled to vote thereon at the meeting. Abstentions

are deemed entitled to vote on such proposal. Therefore, they have the same effect as a vote against this proposal. Broker non-votes are not deemed entitled to vote on such proposal and, therefore, they will have no effect on the vote on such proposal.

The approval of the initial charter proposals, the merger proposal and the warrant amendment proposals is a condition to the consummation of the merger. If either the initial charter proposals or the merger proposal is not approved, none of the proposals will be presented to warrantholders for a vote and the merger will not be consummated.

#### **Voting Your Shares or Warrants**

Each share of Triplecrown common stock or warrant of Triplecrown that you own in your name entitles you to one vote at the special meetings of stockholders and warrantholders, respectively. Your proxy card shows the number of shares of Triplecrown s common stock or warrants that you own. If your shares or warrants are held in street name or are in a margin or similar account, you should contact your broker to ensure that votes related to the shares or warrants you beneficially own are properly counted.

There are two ways to vote your shares of Triplecrown common stock or warrants at the special meetings:

You can Vote by Signing and Returning the Enclosed Proxy Card. If you vote by proxy card, your proxy, whose name is listed on the proxy card, will vote your shares or warrants as you instruct on the proxy card. If you sign and return the proxy card but do not give instructions on how to vote your shares or warrants, your shares or warrants will be voted as recommended by Triplecrown s board FOR the all of the proposals. Votes received after a matter has been voted upon at the special meetings will not be counted.

You can Attend the Special Meetings and Vote in Person. Triplecrown will give you a ballot when you arrive. However, if your shares or warrants are held in the name of your broker, bank or another nominee, you must get a proxy from the broker, bank or other nominee. That is the only way Triplecrown can be sure that the broker, bank or nominee has not already voted your shares or warrants.

#### **Revoking Your Proxy**

If you give a proxy, you may revoke it at any time before it is exercised by doing any one of the following:

you may send another proxy card with a later date;

you may notify Jonathan J. Ledecky, Triplecrown s president and secretary, in writing before the special meetings that you have revoked your proxy; or

you may attend the special meetings, revoke your proxy, and vote in person, as indicated above.

### Who Can Answer Your Questions About Voting Your Shares or Warrants

If you have any questions about how to vote or direct a vote in respect of your shares of Triplecrown s common stock or warrants, you may call Morrow & Co., LLC, Triplecrown s proxy solicitor, at (800) 607-0088, or Jonathan J. Ledecky, Triplecrown s secretary, at (212) 521-5396.

#### **Proxy Solicitation Costs**

Triplecrown is soliciting proxies on behalf of its board of directors. All solicitation costs will be paid by Triplecrown. This solicitation is being made by mail but also may be made by telephone or in person. Triplecrown and its directors, officers and employees may also solicit proxies in person, by telephone or by other electronic means, including email and facsimile.

Triplecrown has hired Morrow & Co., LLC to assist in the proxy solicitation process. It will pay that firm a fee of \$ plus disbursements. Such payments will be made from non-trust account funds. If the merger is successfully closed, Triplecrown will pay Morrow & Co., LLC an additional contingent fee of \$.

Triplecrown will ask banks, brokers and other institutions, nominees and fiduciaries to forward its proxy materials to their principals and to obtain their authority to execute proxies and voting instructions. Triplecrown will reimburse them for their reasonable expenses.

Triplecrown, CAH, Cullen Agritech and their respective directors and executive officers may be deemed to be participants in the solicitation of proxies.

#### **Triplecrown Founders**

As of , 2009, the record date for the Triplecrown special meetings, the Triplecrown Founders beneficially owned and were entitled to vote 13,800,000 Founders Shares and 13,800,000 Founders Warrants. In connection with Triplecrown s IPO, Triplecrown and Citigroup entered into agreements with each of the Triplecrown Founders pursuant to which each Triplecrown Founder agreed to vote his, her or its Founders Shares on the merger proposal in accordance with the majority of the votes cast by the holders of Public Shares. The Triplecrown Founders have indicated that they intend to vote their Founders Shares, Founders Warrants and Sponsors Warrants in favor of all of the other proposals being presented at the meetings. The Founders Shares have no liquidation rights and will be worthless if no business combination is effected by Triplecrown. In connection with the IPO, the Triplecrown Founders placed their Founders Shares and Founders Warrants in escrow with Continental Stock Transfer & Trust Company and agreed that they would not sell such securities until the earlier of twelve months after a business combination or Triplecrown s liquidation, subject to earlier release within such twelve month period if (i) Triplecrown s common stock has a last sales price equal to or exceeding \$13.75 per share for any 20 trading days within any 30-trading day period commencing 90 days after the successful consummation of a business combination or (ii) Triplecrown consummates a subsequent liquidation, merger, stock exchange or other similar transaction that results in all of Triplecrown s stockholders having the right to exchange their shares for cash, securities or other property. If the merger is consummated, the Triplecrown Founders have agreed to cancel 11,260,000 of the Founders Shares.

From the consummation of the IPO to, 2009, no Triplecrown Founder has purchased any shares of Triplecrown common stock or warrants in the open market. If the Triplecrown Founders believe it would be desirable for them or their affiliates to purchase shares or warrants in advance of the special meetings, such determination would be based on factors such as the likelihood of approval or disapproval of the proposals, the number of shares for which conversion may be requested and the financial resources available to such prospective purchasers. Any additional shares or warrants purchased by the Triplecrown Founders will be voted by them in favor of the merger and the other proposals.

30

Triplecrown Founders

## PROPOSALS TO BE CONSIDERED BY THE TRIPLECROWN STOCKHOLDERS

#### THE INITIAL CHARTER PROPOSALS

Triplecrown is proposing to amend its amended and restated certificate of incorporation to revise the definition of a business combination and to remove the prohibition on Triplecrown consummating a business combination with an entity affiliated with any of the Triplecrown Founders. A business combination is defined in Triplecrown s amended and restated certificate of incorporation as follows:

A Business Combination shall mean the initial acquisition by the Corporation, whether by merger, capital stock exchange, asset or stock acquisition or other similar type of transaction, of one or more operating businesses in the financial services industry ( Target Businesses ), whose collective fair market value is equal to at least 80% of the balance in the Trust Fund (as defined below), excluding deferred underwriting discounts and commissions, and resulting in the ownership by the Corporation of not less than 50% of the voting securities of the Target Business or Businesses.

Fair market value for purposes of this Article SEVENTH shall be determined by the Board of Directors of the Corporation based upon financial standards generally accepted by the financial community, such as actual and potential gross margins, the values of comparable businesses, earnings and cash flow, and book value. If the Corporation s Board of Directors is not able to determine independently that the Target Business or Businesses has a sufficient fair market value to meet the threshold criterion, it will obtain an opinion in that regard from an unaffiliated, independent investment banking firm that is a member of the Financial Industry Regulatory Authority and any successor organization thereto. The Corporation is not required to obtain an opinion from an investment banking firm as to the fair market value of the Target Business or Businesses if its Board of Directors independently determines that the Target Business or Businesses have sufficient fair market value to meet the threshold criterion.

Additionally, the prohibition on Triplecrown consummating a business combination with an entity affiliated with any of the Triplecrown Founders contained in Triplecrown s amended and restated certificate of incorporation is stated as follows:

Unless and until the Corporation has consummated a Business Combination as permitted under this Article SEVENTH, the Corporation may not consummate any other business combination, whether by merger, capital stock exchange, asset acquisition, stock purchase, reorganization or other similar business combination or transaction or otherwise. The Corporation will not enter into a Business Combination with any entity which is affiliated with any of its officers, directors, initial stockholders or sponsors or any entity that has received a material financial investment from its initial stockholders or sponsors or any entity affiliated with our officers, directors, initial stockholders or sponsors.

Because (i) the sole stockholder of CAH is controlled by Eric J. Watson, Triplecrown s chairman of the board and treasurer, (ii) Triplecrown will ultimately be acquired by CAH and therefore not acquire at least 50% of the voting securities of the target business, (iii) Cullen Agritech is not an operating business in the financial services industry and (iv) the fair market value of Cullen Agritech on the date of the transaction is less than 80% of the balance of the trust account, the proposed transaction does not meet the requirements as set forth above. Accordingly, Triplecrown must amend its amended and restated certificate of incorporation immediately prior to consummation of the merger in order

to allow for Triplecrown to complete the proposed merger.

Triplecrown s amended and restated certificate of incorporation purports to prohibit amendment to certain of its provisions, including the requirement to obtain approval of disinterested directors and a fairness opinion, prior to consummation of an initial business combination. However, Triplecrown believes that the proposed merger is an attractive opportunity in the current market environment and therefore, public stockholders should be given the opportunity to consider the business combination. In considering the initial charter proposals, Triplecrown s board of directors came to the conclusion that the potential benefits of the proposed merger with CAH to Triplecrown and its stockholders outweighed the possibility of any liability described below as a result of the initial charter proposals being approved. Moreover, Triplecrown is still offering holders of Public Shares the right to affirmatively vote their Public Shares against the merger proposal and demand that such shares be converted into a pro rata portion of the trust account.

Triplecrown has received an opinion from special Delaware counsel, Richards, Layton & Finger, P.A., concerning the validity of the initial charter proposals. Triplecrown did not request Richards, Layton & Finger to opine on whether the clause currently contained in Article Seventh of its amended and restated certificate of incorporation prohibiting amendment of Article Seventh prior to consummation of a business combination was valid when adopted and does not intend on seeking advice of counsel on that question from any other source. Richards, Layton & Finger concluded in its opinion, based upon the analysis set forth therein and its examination of Delaware law, and subject to the assumptions, qualifications, limitations and exceptions set forth in its opinion, that the initial charter amendment, if duly adopted by the board of directors of Triplecrown (by vote of the majority of the directors present at a meeting at which a quorum is present or, alternatively, by unanimous written consent) and duly approved by the holders of a majority of the outstanding stock of the Company entitled to vote thereon, all in accordance with Section 242(b) of the DGCL, would be valid and effective when filed with the Secretary of State in accordance with Sections 103 and 242 of the DGCL. A copy of Richards, Layton & Finger s opinion is included as Annex I to this proxy statement/prospectus, and stockholders are urged to review it in its entirety.

Because Triplecrown s amended and restated certificate of incorporation in its current form does not allow for Triplecrown to complete the proposed merger, each person who purchased his or her Public Shares in the IPO and still held such shares upon learning of the facts set forth above may have securities law claims against Triplecrown for rescission (under which a successful claimant has the right to receive the total amount paid for his or her securities pursuant to an allegedly deficient prospectus, plus interest and less any income earned on the securities, in exchange for surrender of the securities) or damages (compensation for loss on an investment caused by alleged material misrepresentations or omissions in the sale of a security).

Such claims may entitle stockholders asserting them to as much as \$10.00 or more per share, based on the initial offering price of the IPO units comprised of stock and warrants, less any amount received from sale of the original warrants purchased with them, plus interest from the date of Triplecrown s IPO (which, in the case of holders of Public Shares, may be more than the pro rata share of the trust account to which they are entitled on conversion or liquidation).

In general, a person who purchased shares pursuant to a defective prospectus or other representation must make a claim for rescission within the applicable statute of limitations period, which, for claims made under Section 12 of the Securities Act and some state statutes, is one year from the time the claimant discovered or reasonably should have discovered the facts giving rise to the claim, but not more than three years from the occurrence of the event giving rise to the claim. A successful claimant for damages under federal or state law could be awarded an amount to compensate for the decrease in value of his or her shares caused by the alleged violation (including, possibly, punitive damages), together with interest, while retaining the shares. Claims under the anti-fraud provisions of the federal securities laws must generally be brought within two years of discovery, but not more than five years after occurrence. Rescission and damages claims would not necessarily be finally adjudicated by the time the merger with CAH may be completed, and such claims would not be extinguished by consummation of that transaction.

Even if you do not pursue such claims, others, who may include all holders of Public Shares, may. Neither Triplecrown nor CAH can predict whether stockholders will bring such claims, how many might bring them or the extent to which they might be successful.

The approval of each of the initial charter proposals requires the affirmative vote of the holders of a majority of the outstanding shares of Triplecrown common stock on the record date.

A copy of Triplecrown s proposed second amended and restated certificate of incorporation is attached as Annex G to this proxy statement/prospectus. If the initial charter proposals are approved, Triplecrown will present the other proposals to stockholders and warrantholders for their approval. If all of such proposals are approved, the following will occur:

Triplecrown will file an amended and restated certificate of incorporation with the Secretary of State of the State of Delaware to amend Article Seventh to revise the definition of a business combination as set forth below, to delete all references to fair market value and to delete the second sentence of Section E of Article Seventh relating to Triplecrown being prohibited from consummating a business combination with an entity affiliated with any Triplecrown Founder:

A Business Combination shall mean any merger, capital stock exchange, asset, stock purchase, reorganization or other similar business combination between the Corporation and one or more entities or assets ( Target Business or Target Businesses ).

Immediately after the filing of such amended and restated certificate of incorporation, Triplecrown will be authorized to complete the proposed transaction. Thereafter, Triplecrown will look to satisfy all necessary conditions to closing the merger.

Once all conditions to closing the transaction are satisfied, Triplecrown will file all necessary documents with the Secretary of State of the State of Delaware to effectuate such transaction.

If any of the initial charter proposals are not approved, the remaining proposals will not be submitted to stockholders and warrantholders for their approval.

TRIPLECROWN S BOARD OF DIRECTORS RECOMMENDS THAT TRIPLECROWN S STOCKHOLDERS VOTE FOR THE APPROVAL OF EACH OF THE INITIAL CHARTER PROPOSALS.

## THE MERGER PROPOSAL

The discussion in this proxy statement/prospectus of the merger and the principal terms of the merger agreement by and among Triplecrown, CAH, Triplecrown Merger Sub, Cullen Agritech and Cullen Holdings is subject to, and is qualified in its entirety by reference to, the merger agreement. A copy of the merger agreement is attached as Annex A to this proxy statement/prospectus and is incorporated into this proxy statement/prospectus by reference.

## **The Parties**

## **Triplecrown**

Triplecrown Acquisition Corp. is a specified purpose acquisition company (SPAC), formed on June 8, 2007 as a vehicle to effect a merger, capital stock exchange, asset acquisition or other similar business combination with an operating business in the financial services industry.

On October 25, 2007, Triplecrown closed its IPO of 55,200,000 units, including 7,200,000 units subject to the underwriters over-allotment option, with each unit consisting of one share of its common stock and one warrant, each to purchase one share of its common stock at an exercise price of \$7.50 per share. The units from the IPO (including the over-allotment option) were sold at an offering price of \$10.00 per unit, generating total gross proceeds of \$552,000,000. Simultaneously with the consummation of the IPO, Triplecrown consummated the private sale of 5,000,000 Sponsors Warrants at \$1.00 per warrant to Eric J. Watson and Jonathan J. Ledecky for an aggregate purchase price of \$5,000,000. After deducting the underwriting discounts and commissions and the offering expenses, \$536,930,000 was deposited into the trust account and the remaining proceeds became available to be used to provide for business, legal and accounting due diligence on prospective business combinations and continuing general and administrative expenses.

If the merger is consummated, Triplecrown intends to use the funds held in the trust account to pay transaction fees and expenses (including the balance of the purchase price for the land to be used by CAH following consummation of the merger), deferred underwriting discounts and commissions, tax liabilities, if any, and reimbursement of expenses of the Triplecrown Founders and to make purchases of Public Shares, if any. The balance of the funds will be released to CAH to pay Triplecrown stockholders who properly exercise their conversion rights and for working capital and general corporate purposes of CAH and Cullen Agritech. Additionally, if CAH has access to more than \$150 million after closing of the merger, CAH may use such funds, depending on market conditions, to repurchase shares of common stock. It is possible that the present holders of 30.0% or more of the Public Shares will affirmatively vote against the merger and seek conversion of their Public Shares into cash in accordance with Triplecrown s amended and restated certificate of incorporation. If such event were to occur, the merger could not be completed. To preclude such possibility, Triplecrown, the Triplecrown Founders, CAH, Cullen Agritech and their respective affiliates may enter into arrangements to provide for the purchase of the Public Shares from holders thereof who indicate their intention to vote against the merger and seek conversion or who otherwise wish to sell their Public Shares or other arrangements that would induce holders of Public Shares not to vote against the merger proposal. It is likely that such arrangements would involve the purchase by Triplecrown, after the merger, of the Public Shares that are held by the persons or entities who enter into such arrangements using funds from Triplecrown s trust account. As a consequence, it is likely that the amount of funds available to CAH and Cullen Agritech for working capital and general corporate purposes from the trust account would be diminished. Definitive arrangements have not yet been determined but some possible methods are described in the section titled The Merger Proposal Actions That May Be Taken to Secure Approval of Triplecrown s Stockholders. Regardless of the specific arrangements that are made to purchase Public Shares, there

will be sufficient funds from the trust account funds transferred to Triplecrown to pay the holders of all Public Shares that are properly converted and Triplecrown will use such funds for such purpose.

If the merger is not consummated by the date on which Triplecrown is required to be liquidated, either party may terminate the merger agreement. If Triplecrown is unable to complete the merger or another business combination by October 22, 2009, its amended and restated certificate of incorporation provides that its corporate existence will automatically terminate and it will liquidate and promptly distribute to its public stockholders the amount in its trust account plus any remaining non-trust account funds after payment of its

34

Triplecrown 75

liabilities. Accordingly, a holder would have to wait until Triplecrown liquidates in connection with its dissolution to receive liquidation proceeds, which liquidation could take 60 days or more to complete.

Triplecrown s common stock, units and warrants are currently listed on the NYSE Amex under the symbols TCW, TCW.U and TCW.WS, respectively. Triplecrown s common stock, units and warrants will cease trading upon consummation of the merger.

The mailing address of Triplecrown s principal executive office is 970 West Broadway, PMB 402, Jackson, Wyoming 83001. Its telephone number is (307) 633-2831.

### CAH

Cullen Agricultural Holding Corp. is a Delaware corporation wholly owned by Triplecrown that was incorporated on August 27, 2009 solely for the purpose of effecting the merger with Triplecrown described herein and becoming the publicly traded company thereafter. CAH owns no material assets and does not operate any business.

CAH s principal executive office is currently located at 970 West Broadway, PMB 402, Jackson, Wyoming 83001 and its telephone number is (307) 633-2831. Following the merger, CAH s principal executive office will be located at 22 Barnett Shoals Road, Watkinsville, Georgia 30677 and its telephone number will be .

## **Triplecrown Merger Sub**

CAT Merger Sub, Inc. is a Delaware corporation wholly owned by CAH that was incorporated on August 31, 2009 solely for the purpose of effecting the merger with Cullen Agritech described herein. Triplecrown Merger Sub owns no material assets and does not operate any business.

The mailing address of Triplecrown Merger Sub s principal executive office is 970 West Broadway, PMB 402, Jackson, Wyoming 83001. Its telephone number is (307) 633-2831.

## **Cullen Agritech**

Cullen Agricultural Technologies Inc. is a Georgia corporation that was organized in June 2009. Cullen Agritech is wholly owned by Cullen Inc Holdings Ltd. Cullen Agritech is a newly formed company committed to the development and commercialization of advanced agricultural technologies. Cullen Agritech will provide advisory services associated with the implementation of efficient farming techniques and promote a methodology that incorporates components of New Zealand s pasture-based farming system. Cullen Agritech s principle focus will be to improve agricultural yields through forage and animal sciences.

Cullen Agritech s principal executive office is currently located at 22 Barnett Shoals Road, Watkinsville, Georgia 30677 and its telephone number is .

## Structure of the Merger

The merger agreement provides for (i) Triplecrown to merge with and into CAH with CAH surviving the merger and becoming the new publicly-traded corporation of which the present holders of Triplecrown securities will be security holders and (ii) Triplecrown Merger Sub to merge with and into Cullen Agritech with Cullen Agritech surviving the

CAH 76

merger and becoming a wholly-owned subsidiary of CAH.

Upon completion of the transactions contemplated by this proxy statement/prospectus, assuming no holders of the Public Shares elect to convert their shares into a portion of Triplecrown s trust account, the current holders of common stock of Triplecrown will own 57,740,000 shares of CAH stock (assuming the officers, directors and founders of Triplecrown cancel an aggregate of 11,260,000 shares they will receive upon exchange of their Triplecrown common stock in connection with the merger as described in more detail in this proxy statement/prospectus), representing 78.6% of CAH s outstanding common stock, and the current holder of common stock of Cullen Agritech will own 15,881,148 shares of CAH common stock, representing 21.4% of CAH s outstanding common stock. Assuming that holders of approximately 29.99% of the Public Shares elect to convert their shares into a portion of Triplecrown s trust account, the current holders of common stock of Triplecrown will own 41,180,000 shares of CAH stock (assuming the officers, directors and founders of Triplecrown cancel an aggregate of 11,260,000 shares they will receive upon exchange of their

Triplecrown common stock in connection with the merger as described in more detail in this proxy statement/prospectus), representing 72.4% of CAH s outstanding common stock, and the current holder of common stock of Cullen Agritech will own 15,881,148 shares of CAH common stock, representing 27.6% of CAH s outstanding common stock. Cullen Holdings has agreed that it will not sell any of the shares it receives in the merger until the 12-month anniversary of the consummation of the merger. Assuming CAH purchases shares after the closing so that \$150 million is left available to operate CAH s business, the holders of Triplcrown common stock will own 56.8% and Cullen Holdings will own 43.2% of CAH s outstanding common stock.

Prior to Triplecrown s IPO, the Triplecrown Founders acquired 13,800,000 Founders. Units, representing 13,800,000 Founders. Shares and 13,800,000 Founders. Warrants. All of such securities are held in escrow under an escrow agreement with Continental Stock Transfer & Trust Company, pursuant to which such securities will not be released from escrow until one year after the consummation of Triplecrown s initial business combination (except in certain limited situations). In connection with the merger, the Triplecrown Founders have agreed to have cancelled an aggregate of 11,260,000 shares of common stock of CAH they will receive in exchange for 11,260,000 Founders. Shares upon consummation of the merger. The remaining 2,540,000 shares, as well as all of the warrants of CAH that the Triplecrown Founders will receive in exchange for their Founders. Warrants, will continue to be held in escrow pursuant to the original terms of the escrow agreement and will be released one year after the consummation of the merger.

## Name; Headquarters; Stock Symbols

After completion of the merger:

the name of the publicly-traded holding company will be Cullen Agricultural Holding Corp.; the corporate headquarters and principal executive offices of CAH will be located at 22 Barnett Shoals Road, Watkinsville, Georgia 30677; and

the parties intend to apply to have CAH s common stock and warrants listed for trading on the New York Stock Exchange or NASDAQ under the symbols and .WS, respectively, after consummation of the merger.

## Indemnification

To provide a fund to secure the indemnification obligations of Cullen Holdings to Triplecrown against losses that the surviving entity of the merger may sustain as a result of (i) the inaccuracy or breach of any representation or warranty made by Cullen Agritech in the merger agreement or any schedule or certificate delivered by it in connection with the merger agreement and (ii) the non-fulfillment or breach of any covenant or agreement made by Cullen Agritech in the merger agreement, an aggregate of 10% of the initial shares to be issued in the merger (1,588,115 shares of Triplecrown common stock) will be placed in escrow (with an independent escrow agent), which will be canceled to the extent that Triplecrown has damages for which it is entitled to indemnification. The escrow will be the sole remedy for Triplecrown for its rights to indemnification pursuant to the merger agreement. Claims for indemnification may be asserted against the escrow by Triplecrown once its damages exceed a deductible and will be reimbursable to the full extent of the damages in excess of such amount up to a maximum of the escrowed shares. Claims for indemnification may be asserted until the 30th day after the date CAH has filed with the SEC its Annual Report on Form 10-K for the year ending December 31, 2010. As a consequence of these limitations, Triplecrown may not be able to be entirely compensated for indemnifiable damages that it may sustain.

## **Employment Agreement**

Dr. Richard Watson has entered into an employment agreement with Natural Dairy. The employment agreement is for a 3-year term subject to earlier termination in certain circumstances, and may be extended by mutual agreement of the executive and Cullen Agritech.

## Lock-Up

Cullen Holdings has agreed that it will not sell any shares of CAH common stock it receives in the merger until one year after the consummation of the merger. A copy of the lock-up agreement that will be executed by Cullen Holdings is attached to this proxy statement/prospectus as Annex J.

## **Background of the Merger**

The following is a brief discussion of the background of these negotiations, the merger agreement and related transactions.

Triplecrown was formed on June 8, 2007 to effect a merger, capital stock exchange, asset acquisition or other similar business combination with an operating business. Triplecrown s amended and restated certificate of incorporation provides that Triplecrown must liquidate unless it has consummated a business combination by October 22, 2009. As of June 30, 2009, \$539,328,529 was held in deposit in the trust account.

Promptly following the IPO of Triplecrown, Triplecrown contacted several investment bankers, private equity firms, consulting firms, legal and accounting firms, as well as numerous other business relationships. Through these and further efforts, Triplecrown identified and reviewed information with respect to approximately 250 potential target companies.

The majority of these situations were in the financial services and related sectors. Triplecrown held over 200 preliminary, and in some cases, multiple detailed due diligence meetings with respect to these target companies drawn from commercial banking firms, asset management companies, hedge funds, investment banks, life insurance, life insurance brokerage and life settlement companies, credit card companies and credit card processors, Small Business Investment Companies, Business Development Companies, Real Estate Investment Trusts, debt collection companies, and pension management firms. These target companies were situated both domestically and internationally. Triplecrown delivered and negotiated over two dozen letters of intent to those prospective target companies in which it believed a suitable transaction could be structured. However, none of these letters of intent were ever executed by the targets. The majority of these target companies ultimately rejected these letters of intent due to the perceived inherent shortcomings of Triplecrown s operating structure, dilution and warrant overhang.

In June 2009, Eric J. Watson indicated to Jonathan J. Ledecky that Cullen Agritech, an entity owned by Cullen Holdings, which Mr. Watson controlled, might be an attractive opportunity for Triplecrown to pursue. The parties then commenced discussions regarding a potential transaction involving Cullen Agritech and Triplecrown. In connection with such potential business combination, Triplecrown paid deposits in the amount of approximately \$1.8 million for a certain piece of land that would be utilized in connection with such transaction.

On August 7, 2009, a meeting of the Triplecrown board of directors was held. All directors attended, as did, by invitation, Brian L. Ross and Jeffrey M. Gallant of Graubard Miller. Prior to the meeting, information concerning the proposed transaction with Cullen Agritech was delivered to the directors. Triplecrown s management discussed its efforts to find a suitable candidate for a business combination, the benefits of selecting Cullen Agritech as well as its risks as set forth below. The board also discussed the necessary deposits to be paid by Triplecrown to secure the piece of land to be used by CAH following consummation of the merger. After considerable discussion, it was agreed that Triplecrown should continue to negotiate the potential transaction with Cullen Agritech and ratified Triplecrown to paying the deposits for the land provided that the agreement calling for such payment would require Cullen Agritech or an affiliate to reimburse Triplerown for the deposits if the parties were not able to execute the merger agreement.

During the week of August 24, 2009 through August 28, 2009, the parties negotiated various terms and conditions of the proposed transaction telephonically and continued their due diligence work. On August 28, 2009, drafts of a merger agreement and proxy statement were circulated by Triplecrown counsel to Cullen Agritech and its counsel.

Lock-Up 80

During that same time period, Triplecrown entered into negotiations with the underwriters in its IPO regarding the reduction of their deferred underwriting commissions payable at the consummation of the merger. Negotiations continue to be ongoing. Additionally, because (i) the sole stockholder of CAH is controlled by Eric J. Watson, Triplecrown s chairman of the board and treasurer, (ii) Triplecrown will ultimately be acquired by CAH and therefore not acquire at least 50% of the voting securities of the target business, (iii) Cullen Agritech is not an operating business in the financial services industry and (iv) the fair market value of Cullen Agritech on the date of the transaction is less than 80% of the balance of the trust account,

the proposed transaction does not meet the IPO and amended and restated certificate of incorporation requirements discussed elsewhere in this proxy statement/prospectus. Therefore, Triplecrown also entered into discussions with the underwriters to obtain their consent to any necessary amendments to the agreements entered into in connection with the IPO in order to consummate the transactions described herein. If Triplecrown is unable to obtain such consent but still determines to proceed with the transaction, Triplecrown will be exposing itself to a claim by the underwriters that it proceeded without their consent. Although the parties do not believe such a claim would be viable, because it is unclear what damages the underwriters would be able to claim, we cannot assure you of this fact.

On August 31, 2009, another meeting of the board of directors of Triplecrown was held to consider approval of the transactions. All directors attended, as did David Alan Miller, Brian L. Ross and Jeffrey M. Gallant of Graubard Miller. Also attending were Philip Wisler and Abhijeet Shekdar from Duff & Phelps, financial advisors to Triplecrown. At this meeting, Messrs. Wisler and Shekdar of Duff & Phelps made a presentation regarding the fairness of the consideration to be paid in the transaction. Messrs. Wisler and Shekdar advised the board that it was the opinion of Duff & Phelps that the consideration to be paid in the merger was fair to Triplecrown from a financial point of view. Messrs. Wisler and Shekdar then detailed for the board the analysis performed by Duff & Phelps and made a presentation concerning how Duff & Phelps had arrived at its opinion. Messrs. Wisler and Shekdar discussed at length with Triplecrown s board the different analyses used to determine whether the merger consideration to be paid was fair from a financial point of view to Triplecrown. For a more detailed description of the Duff & Phelps fairness opinion, see the section entitled *Fairness Opinion* below. After considerable review and discussion, the merger agreement and related documents were approved (with Mr. Watson abstaining from the vote as a result of his relationship with Cullen Agritech), subject to final negotiations and modifications, and the board determined to recommend the approval of the merger agreement.

The merger agreement was signed on September 4, 2009. Prior to the market open on September, 2009, Triplecrown issued a press release and subsequently filed a Current Report on Form 8-K on the same day announcing the execution of the merger agreement and discussing the terms of the merger agreement.

# Triplecrown s Board of Directors Reasons for the Approval of the Merger

Triplecrown s board of directors carefully evaluated the agreements relating to the proposed merger and reviewed industry and financial data in order to determine that the transaction terms were reasonable and that the merger was in the best interests of Triplecrown s stockholders.

# Key Cullen Agritech personnel have extensive experience in improving yields through applying forage based farming techniques

Cullen Agritech will rely on in-house scientific experts, along with an adjunct Advisory Board made up of some of the world s leading scientists, policy analysts and business leaders in the agricultural arena. Additionally, Cullen Agritech s Chief Scientific Officer, Dr. Richard H. Watson, has an extensive background in pastoral science and technology, from the lab bench through to commercialization and industry application of technologies.

Cullen Agritech also has quantitative geneticists that have the ability to develop lines of dairy cattle designed to produce milk efficiently on pasture. Cullen Agritech has exclusive access to embryo, semen and genetic screening technologies (SNP-chip) to accelerate genetic improvement and deployment of these lines in the United States dairy industry. Only Cullen Agritech and its customers will have access to these grazing lines.

## Proven Model through Research Farms developed by Key Cullen Agritech Personnel to provide key cost advantages

Key personnel now with Cullen Agritech developed Cullen Agritech s farming system and has been proven on research farms in Girard, Georgia, the first of which was established in late 2007. These farms were established to develop and test Cullen Agritech s proprietary grazing system that will be rolled-out in the Southeast United States through Natural Dairy. The first research farm began producing milk in March 2008. During 2008, it was used to refine and develop the Cullen Agritech s farming system, the know-how of which makes up Cullen Agritech s intellectual property. This research was focused on the development of a forage

crop system to maximize the production and utilization of grown pasture. Breeding and calving season trials have been conducted to optimize the relationship between feed grown on farm and the feed demand of the herd. During 2009, Cullen Agritech has implemented its refined strategy and has been able to achieve favorable production cost results, proving the efficiency of its system.

## Forage based system provides Lower Cost per CWT of Dairy Production:

The cost of producing milk will vary greatly depending on the region, the exact management practices and the quality of farmers. For a significant majority of dairy farmers, the high dependence on the use of corn-based concentrate as a feedstock results in an extremely challenging cost base. The use of forage as a replacement for corn-based concentrate in Cullen Agritech s model reduces this expense substantially, creating a much more economically-sustainable cost structure. Cullen Agritech s model is also less labor intensive and has reduced animal health costs due to healthier and less confined conditions.

# Natural Dairy is strategically located in a region of significant demand coupled with a substantial shortage in supply

Natural Dairy s roll-out is focused on the Southeastern United States where there is currently a significant shortage in the supply of fresh liquid milk. In addition, the Eastern Seaboard represents the largest fresh liquid milk market in the world. Natural Dairy is strategically positioned to help fill that supply gap and produce milk for this market, which is currently starved of supply. Management estimates that \$2 billion of capital would be needed to be deployed to satisfy the current shortage.

### Cullen Agritech s systems can achieve a high production per cow

Cullen Agritech utilizes forage production systems that optimize seasonal qualitative attributes of the pasture to best match the energy demands of the cow. Under these systems, the cow is provided sufficient nutrients to meet her needs for body maintenance and milk production. Cullen Agritech uses energy balances to ensure that the cows at pasture are fed to achieve the highest possible production targets in the most cost effective manner. This level of feed management sets Cullen Agritech apart from other grazing operations in the USA.

#### Cullen Agritech s system results in healthier livestock and invreased longevity

The common United States dairy industry cow is the US Holstein. Cullen Agritech s model is more suited to smaller framed livestock such as Jersey/Holstein cross-breeds or a Friesian Holstein. These breeds of livestock generally have longer productive lives than a typical US Holstein, which is further lengthened by the healthier conditions associated with Cullen Agritech s pasture-based farming system. This longevity will result in reduced livestock culling rates resulting in additional revenue from surplus livestock sales. Livestock managed under Cullen Agritech s system are also likely to have fewer health issues due to increased exercise and exposure to cleaner, less confined living conditions. This maintains a healthier animal and reduces the speed at which infection can spread throughout a herd. This results in increased longevity as well as reduced health-related operating expenses on the farms.

## **Reduced labor costs**

Cullen Agritech will utilize state of the art milking systems which will be custom made to maximize efficiency and minimize labor costs. This technology, combined with a unique and efficient farm design and management strategy will result in reduced labor costs, further reducing the cost of production under Cullen Agritech s unique system.

## Potential to achieve higher pricing in the future

The demand for naturally produced animal products is increasing as the population s concern with how their food is produced increases. General awareness of the animal ethics and human health benefits of the grazing-based production system have also grown a USDA survey showed that 48% of United States consumers now recognize Grass-fed as a brand.

Products from animals fed on a pasture dominant diet have been found to contain higher levels of a number of naturally occurring metabolites that have proven human health benefits. The fermentation of the forage diet in grazing animals by rumen bacteria create higher levels of conjugated linoleic acid (CLA), omega-3 and 6 fatty acids and vitamins A&E in the milk. Production of these in confinement cows is reduced by the heavy-starch grain diet, which reduces the formation of these beneficial fermentation products.

Currently, there is a small but rapidly growing market for grass-fed or pasture-fed beef products. However, the grass-fed milk market is virtually non-existent due to dominance of the confinement model and the lack of producers who have the technical knowledge to produce milk on pasture year round.

Natural Dairy has the ability to produce grass-fed milk year round and its production levels may in the future cause separate processing, with Natural Dairy s milk as a grass-fed product being processed separate to other milk. Therefore, Natural Dairy milk will be initially sold as standard milk along with milk from confinement production. This means it will not initially receive any premium pricing on the basis of its grass-fed Product. Accordingly, Natural Dairy has not included any premium pricing due to the grass-fed nature of its product in its financial projections.

## Strategic agreements and relationships allow for efficient large scale rollout of pasture based system

Cullen Agritech entered into a strategic cooperation agreement with New Zealand Agritech Inc (NZ Agritech) to promote the interests of NZ Agritech and its members. Cullen Agritech will assist members of NZ Agritech to mitigate barriers of market entry and provide the opportunity to realize the potential growth in various markets. This alliance records an important connection to participants of New Zealand s agricultural technology industry and enables Cullen Agritech to offer its customers the benefit of advanced technologies.

## **Adverse Factors Considered by Triplecrown**

Triplecrown s board also evaluated several adverse factors in its consideration of the acquisition of Cullen Agritech.

These included:

## Lack of operating history

Cullen Agritech was incorporated in June 2009, is a development stage company and has no operating history. Cullen Agritech has no significant assets and will commence operations only upon consummation of the transactions described in this proxy statement/prospectus. Accordingly, it may not be able to operate its business successfully or implement its policies and strategies as described in this proxy statement/prospectus.

### Adverse general economic conditions

In its evaluation of Cullen Agritech, Triplecrown s board of directors considered the current adverse economic conditions and the impact such conditions could have on Cullen Agritech s business. It was the board s belief that the trends evidenced in the overall dairy market and Cullen Agritech s strategy outweighed concerns about general economic conditions.

The board of Triplecrown was cognizant of Triplecrown s liquidation date of October 22, 2009, but ultimately evaluated the potential business combination with Cullen Agritech strictly on the quantitative and qualitative information regarding Cullen Agritech and its business that was available. Since completion of Triplecrown s IPO, the board has been regularly kept apprised of potential business combination targets and management s discussions and evaluation of such targets. As discussed above, Triplecrown engaged in an ongoing and systematic search for potential business combination candidates, deciding on its own accord in various situations to terminate discussions with potential candidates when determined by management that such candidates did not ultimately represent the investment opportunity that Triplecrown wanted to present to its stockholders.

## **Fairness Opinion**

Triplecrown engaged Duff & Phelps, LLC ( Duff & Phelps ) to render an opinion to its board of directors as to the fairness, from a financial point of view, to the Company of the valuation of Cullen Agritech negotiated between Triplecrown and Cullen Holdings. Triplecrown selected Duff & Phelps because Duff & Phelps is a leading independent financial advisory firm, offering a broad range of valuation, investment banking and consulting services, including fairness and solvency opinions, mergers and acquisitions advisory, mergers and acquisitions due diligence services, financial reporting and tax valuation, fixed asset and real estate consulting, ESOP and ERISA advisory services, legal business solutions and dispute consulting. Duff & Phelps is regularly engaged in the valuation of businesses and securities and the preparation of fairness opinions in connection with mergers, acquisitions and other strategic transactions. Furthermore, Duff & Phelps has extensive industry expertise within the dairy sector.

40

#### **TABLE OF CONTENTS**

On August 31, 2009, Duff & Phelps rendered its oral opinion to Triplecrown s board of directors, which was subsequently confirmed in a written opinion dated August 31, 2009, that, subject to the limitations, exceptions, assumptions and qualifications set forth therein, as of August 31, 2009, the valuation of Cullen Agritech negotiated between the parties in connection with the merger is fair, from a financial point of view, to the Company.

The full text of the written opinion of Duff & Phelps, which sets forth, among other things, assumptions made, procedures followed, matters considered and qualifications and limitations of the review undertaken in rendering the opinion, is attached as Annex K to this proxy statement. Stockholders are urged to read the opinion carefully and in its entirety.

The Duff & Phelps opinion is directed to Triplecrown s board of directors and addresses only the fairness to Triplecrown, from a financial point of view, of the valuation of Cullen Agritech negotiated between the parties to the merger. The Duff & Phelps opinion is not a recommendation as to how any stockholder should vote or act with respect to any matters relating to the merger (including, without limitation, with respect to the exercise of rights to convert Public Shares into cash). Further, the Duff & Phelps opinion does not in any manner address Triplecrown s underlying business decision to engage in the merger or the relative merits of the merger as compared to any alternative business transaction or strategy (including, without limitation, a liquidation of Triplecrown after not completing a business combination transaction within the allotted time). The decision as to whether to approve the merger or any related transaction may depend on an assessment of factors unrelated to the financial analysis on which the Duff & Phelps opinion is based.

Triplecrown has advised Duff & Phelps that it intends to issue 15,881,148 shares of CAH common stock at an agreed value of \$9.76 per share (as to which value Duff & Phelps expressed no opinion) as provided in the merger agreement. Based on the aggregate merger consideration of 15,881,148 shares of CAH common stock and an assumed value of CAH common stock of \$9.76 per share, which is Triplecrown s prevailing trust account cash balance on a per share basis, Duff & Phelps noted that the aggregate merger consideration implied a total value of Cullen Agritech of \$155 million.

Triplecrown has also advised Duff & Phelps that, upon consummation of the merger, CAH intends to deploy Cullen Agritech s intellectual property in concert with CAH s own capital to make additional investments in land, infrastructure, livestock and working capital in the furtherance of its U.S. based dairy business plan involving the establishment of 40 new dairy farms operating within the Southeast United States (the U.S. Dairy Business Plan ).

The following is a summary of the material analyses performed by Duff & Phelps in connection with rendering its opinion. Duff & Phelps noted that the basis and methodology for the opinion have been designed specifically for this purpose and may not translate to any other purposes. While this summary describes the analysis and factors that Duff & Phelps deemed material in its presentation and opinion to the Triplecrown board of directors, it does not purport to be a comprehensive description of all analyses and factors considered by Duff & Phelps. The opinion is based on the comprehensive consideration of the various analyses performed. This summary is qualified in its entirety by reference to the full text of the opinion.

In arriving at its opinion, Duff & Phelps did not attribute any particular weight to any particular analysis or factor considered by it, but rather made qualitative judgments as to the significance and relevance of each analysis and factor. Several analytical methodologies were employed by Duff & Phelps in its analyses, and no one single method of analysis should be regarded as critical to the overall conclusion reached by Duff & Phelps. Each analytical technique has inherent strengths and weaknesses, and the nature of the available information may further affect the value of particular techniques. Accordingly, Duff & Phelps believes that its analyses must be considered as a whole and that selecting portions of its analyses and of the factors considered by it, without considering all analyses and

factors in their entirety, could create a misleading or incomplete view of the evaluation process underlying its opinion. The conclusion reached by Duff & Phelps, therefore, is based on the application of Duff & Phelps own experience and judgment to all analyses and factors considered by Duff & Phelps, taken as a whole.

41

In connection with preparing the opinion, Duff & Phelps made such reviews, analyses and inquiries as Duff & Phelps deemed necessary and appropriate under the circumstances, including, but not limited to, the following:

Discussed the operations, financial conditions future prospects and projected operations and performance of Triplecrown and Cullen Agritech with the managements of Triplecrown and Cullen Agritech;

Discussed the proposed merger with the managements of Triplecrown and Cullen Agritech; Reviewed certain publicly available financial statements and other business and financial information of Triplecrown and proprietary information regarding Cullen Agritech and publicly available data for certain operating companies within the industry sector in which Cullen Agritech operates or will operate;

Reviewed certain operating information concerning Triplecrown and Cullen Agritech, which each of Triplecrown and Cullen Agritech have identified as being the most current financial statements available;

Reviewed unaudited financial statements for the two operating test farms in Georgia provided by management of Cullen Agritech;

Reviewed certain financial forecasts prepared by the managements of Triplecrown and Cullen Agritech, respectively, reflecting base case forecasts for financial and operating results under the U.S. Dairy Business Plan;

Reviewed the Cullen Agritech investor presentation dated August 2009;

Reviewed the Cullen Agritech presentation entitled Reconciliation of Actual CAT Research Farm Performance to Forecasts ;

Reviewed historical and forecasted domestic (U.S.) milk price information;

Reviewed U.S. milk supply and demand data prepared by the U.S. Department of Agriculture;

Reviewed farm operating (e.g., herd stocking rates) and operating cost data for various farming formats (e.g., confinement, pastoral) for farms based domestically and abroad;

Reviewed a draft of the merger agreement dated August 28, 2009;

Reviewed the historical trading price and trading volume of the publicly traded securities of certain other companies that it deemed relevant:

Compared the financial performance of Cullen Agritech with financial performance metrics of certain other publicly traded companies that Duff & Phelps deemed relevant;

Compared certain financial terms of the proposed merger to financial terms, to the extent publicly available, of certain other business combination transactions that it deemed relevant; and

Conducted such other analyses and considered such other factors as it deemed appropriate.

In its review and analysis, and in arriving at its opinion, Duff & Phelps, with Triplecrown s consent:

Relied upon the accuracy, completeness, and fair presentation of all information, data, advice, opinions and representations obtained from public sources or provided to it from private sources, including the managements of Triplecrown and Cullen Agritech, and did not independently verify such information;

Relied upon managements characterization of the proprietary nature of Cullen Agritech s intellectual property and the fact that while such technology is not currently patent protected, the trade practices 42

#### TABLE OF CONTENTS

developed by Cullen Agritech in the areas of forage sciences, animal genetics and health, farm management, and systems and training, have been uniquely adapted for conditions in the target market in Georgia and could not be readily imitated or adopted by current or future market participants;

Assumed that any estimates, evaluations, forecasts and projections furnished to Duff & Phelps were reasonably prepared and based upon the best currently available information and good faith judgment of the person furnishing the same:

Assumed that the final versions of all documents reviewed by Duff & Phelps in draft form conform in all material respects to the drafts reviewed;

Assumed that information supplied to Duff & Phelps and representations and warranties made in the merger agreement are substantially accurate;

Assumed that all of the conditions required to implement the proposed merger will be satisfied and that the merger will be completed in accordance with the merger agreement without any amendments thereto or any waivers of any terms or conditions thereof;

Assumed that all of the capital required to implement the proposed U.S. Dairy Business Plan will be available in the amounts required as and when needed as reflected in the U.S. Dairy Business Plan;

Relied upon the fact that Triplecrown and its board of directors have been advised by counsel as to all legal matters with respect to the proposed merger, including whether all procedures required by law to be taken in connection with the merger have been duly, validly and timely taken; and

Assumed that all governmental, regulatory or other consents and approvals necessary for the consummation of the proposed merger will be obtained without any adverse effect on Triplecrown, Cullen Agritech, the U.S. Dairy Business Plan or the contemplated benefits expected to be derived in the merger.

In its analysis and in connection with the preparation of its opinion, Duff & Phelps made numerous assumptions with respect to industry performance, regulatory and environmental (e.g., weather), general business, market and economic conditions and other matters, many of which are beyond the control of any party involved in the merger. To the extent that any of the foregoing assumptions or any of the facts on which the Duff & Phelps opinion is based proves to be untrue in any material respect, Duff & Phelps has advised Triplecrown s board of directors that the Duff & Phelps opinion cannot and should not be relied upon.

Duff & Phelps did not make any independent evaluation, appraisal or physical inspection of Triplecrown s solvency or of any of its specific assets or liabilities (contingent or otherwise). The Duff & Phelps opinion should not be construed as a valuation opinion, credit rating, solvency opinion, or analysis of Triplecrown s creditworthiness, as tax advice or as accounting advice. Duff & Phelps has not made, and assumes no responsibility to make, any representation, or render any opinion, as to any legal matter.

Duff & Phelps has not been requested to, and did not, (a) initiate any discussions with, or solicit any indications of interest from, third parties with respect to the proposed merger, Cullen Agritech, the assets, businesses or operations of Triplecrown, or any alternatives to the proposed merger, (b) negotiate the terms of the proposed merger, and therefore, Duff & Phelps has assumed that such terms are the most beneficial terms, from Triplecrown s perspective, that could, under the circumstances, be negotiated among the parties to the merger agreement and the merger, or (c) advise Triplecrown s board of directors or any other party with respect to alternatives to the proposed merger. In addition, Duff & Phelps is not expressing any opinion as to the market price or value of the Shares or Share equivalents either as part of the proposed merger or after announcement of the merger.

Duff & Phelps prepared its opinion as of August 31, 2009. The opinion was necessarily based upon market, economic, financial and other conditions as they existed and could be evaluated as of such date, and Duff & Phelps disclaims any undertaking or obligation to advise any person of any change in any fact or matter affecting its opinion coming or brought to the attention of Duff & Phelps after the date of the Duff & Phelps opinion or otherwise to update, revise or reaffirm its opinion. Notwithstanding and without limiting the

foregoing, in the event that there is any change in any fact or matter affecting the Duff & Phelps opinion after the date of the opinion and prior to the completion of the merger, Duff & Phelps reserves the right to change, modify or withdraw the opinion.

## **Summary of Financial Analyses by Duff & Phelps**

For the purpose of determining whether the valuation of Cullen Agritech negotiated between the parties to the merger agreement was fair, from a financial point of view, to Triplecrown, Duff & Phelps relied on several generally accepted valuation techniques, including a calculation based on discounted cash flow, and valuation multiples determined based on an analysis of selected public companies and M&A transactions. Given Cullen Agritech s greenfield business model, Duff & Phelps also focused its analysis on determining value indications for the primary asset of Cullen Agritech its intellectual property.

The following is a summary of the material financial analyses used by Duff & Phelps in connection with providing its opinion to Triplecrown s board of directors. The financial analyses summarized below include information presented in tabular format. In order to fully understand the financial analyses used by Duff & Phelps, the tables must be read together with the text of each summary. The tables alone do not constitute a complete description of the financial analyses. Rather, the analyses listed in the tables and described below must be considered as a whole; considering any portion of such analyses and of the factors considered, without considering all analyses and factors, could create a misleading or incomplete view of the process underlying Duff & Phelps opinion.

## **Discounted Cash Flow Analysis**

A discounted cash flow analysis is a traditional valuation methodology used to derive a valuation of an asset by calculating the present value of estimated future cash flows of the asset. Present value refers to the current value of future cash flows or amounts and is obtained by discounting those future cash flows or amounts by a discount rate that takes into account macro-economic assumptions and estimates of risk, the opportunity cost of capital, expected returns and other appropriate factors.

Duff & Phelps performed its discounted cash flow analysis by adding (1) the present value of projected free cash flows for Cullen Agritech for the fiscal years 2009 through 2018 to (2) the present value of the terminal value for Cullen Agritech as of 2018. Free cash flow is defined as cash that is available to either reinvest or to distribute to securityholders and terminal value refers to the value of all future cash flows from an asset at a particular point in time.

The methodology used by Duff & Phelps to determine the present value of the free cash flows of Cullen Agritech is summarized below:

## **Discounted Cash Flow Analysis Methodology**

Net operating profit after tax for the U.S. dairy business of Cullen Agritech from 2009 through 2018<sup>(1)</sup>
Adjusted for

Non-cash items (i.e., depreciation expense) Less

Investments in tangible assets and working capital<sup>(2)</sup>

## Debt-free cash flow attributable to Cullen Agritech Multiplied by

Based on adjusted version of pro	pjections provided by Triplecrown s management, as described below, with net								
Based on adjusted version of projections provided by Triplecrown s management, as described below, with net operating losses accumulated and valued separately									
(2)	Includes structures, livestock, land and equipment								
44									

Present value factor Equals

Present value of debt-free cash flow attributable to Cullen Agritech

The projected financial information that Duff & Phelps used in its discounted cash flow analysis is based on financial forecasts and estimates provided by Triplecrown s management as adjusted to reflect inflationary growth in costs, industry estimates of forward milk prices and a sustainable earnings before interest, taxes, depreciation and amortization (EBITDA) margin. Note that these projections involve numerous and significant subjective determinations which may or may not prove to be correct or complete. No representation or warranty, express or implied, is made as to the accuracy or completeness of such projections and none of these projections should be relied upon as a representation, warranty or guaranty, whether as to the past, present or future.

The projected revenues, EBITDA and free cash flow figures used by Duff & Phelps in its discounted cash flow analysis for Cullen Agritech are presented below. EBITDA is not a presentation made in accordance with U.S. generally accepted accounting principles. Duff & Phelps used EBITDA because it believes EBITDA is a measure that is generally accepted by the financial community in the valuation of assets.

Cullen Agritech Projected Financial Information (\$ in Millions)

	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018
Revenues	\$0	\$46	\$150	\$184	\$189	\$193	\$204	\$215	\$218	\$ 222
EBITDA	(0)	9	48	78	83	89	100	113	116	121
Free Cash Flow <sup>(1)</sup>	0	(281)	(7)	57	63	66	69	69	72	74

(1) Free cash flows are net of capital investments in land, infrastructure, livestock and working capital required to execute the U.S. Dairy Business Plan.

Duff & Phelps calculated the projected free cash flows for Cullen Agritech for fiscal years 2009 through 2018 based on its projected net operating profit after tax, plus depreciation, less increases in net working capital, less capital expenditures. Duff & Phelps also calculated a terminal value for Cullen Agritech as of 2018 by applying a commonly used method of assigning an exit multiple to the sustainable level of EBITDA achieved by the end of the discrete projection period. The EBITDA multiple used for this purpose was 6.0x. Duff & Phelps believes that the multiple used in the determination of terminal value is supported by various benchmarks and the growth rates exhibited by the publicly traded companies that Duff & Phelps selected for purposes of its analysis.

Finally, Duff &Phelps discounted the projected free cash flows and the terminal value for Cullen Agritech to their respective present value equivalents using cost of capital assumptions ranging from 11.5% to 13.5%.

The foregoing discounted cash flow analysis indicated a range of values for Cullen Agritech of \$110 million to \$160 million.

## Selected Public Company/Selected M&A Transaction Analysis

The selected public company analysis is based upon a comparison of the subject company to publicly held companies whose stocks are actively traded and whose market multiples may be used to provide indications of value for

comparable companies. As part of this analysis, Duff & Phelps identified six publicly-traded companies that share many of the same operating characteristics and are affected by many of the same economic forces as Cullen Agritech, and which Duff & Phelps considered to be reasonably comparable to Cullen Agritech in terms of investment risks and attributes, as well as products provided and markets served.

Duff & Phelps also identified six recent merger and acquisition transactions involving target companies operating in the dairy and dairy products industries, and derived implied valuation multiples for the targets in those transactions.

None of the companies utilized for comparative purposes in this analysis are identical to Cullen Agritech. Accordingly, a complete valuation analysis cannot be limited to a quantitative review of the selected companies and involves complex considerations and judgments concerning differences in financial and operating characteristics of such companies, as well as other factors that could affect their value relative to that of Cullen Agritech.

The specific steps applied by Duff & Phelps in its selected public company/selected M&A transaction analysis are summarized below:

## Selected Public Company/Selected M&A Transaction Analysis Methodology

U.S. Dairy Business Plan EBITDA for 2012 (stabilized)<sup>(1)</sup>
Multiplied by

Selected market valuation multiples derived from selected public company/selected M&A transactions analysis Equals

Future asset value of stabilized U.S. dairy business of Cullen Agritech Multiplied by

Present value factor

Equals

Present value of U.S. dairy business (adjusted for net operating losses)

Less

Present value of Net Investment<sup>(2)</sup> Equals

Present value of Cullen Agritech

Relevant market data for the six public companies used in the selected public company portion of the analysis is summarized in the table below:

Company Information	Market I (\$ in Mi Except p Share Pr	llions, per		et Multi prise Va	•	ı Multiple	of:	
	Danitra		FY	FY	FY	FY	FY	FY
Company Name	Equity Value	EV	2009	2010	2011	LTM	2010	2011
	value		Revs	Revs	Revs	EBITD.	A EBITD	AEBITDA
Bongrain SA	\$639	\$1,134	0.3x	0.3x	0.2x	6.2x	5.3x	5.0x
Glanbia plc	747	1,207	0.6x	0.6x	0.5x	7.4x	7.3x	6.3x
Saputo, Inc.	5,435	6,056	1.0x	0.9x	N/A	10.9x	8.9x	8.4x
Dairy Crest Group plc	426	878	0.5x	0.5x	0.5x	6.5x	6.1x	5.9x
Warrnambool Cheese And Butter								
Factory Company Holdings	87	164	0.4x	0.4x	0.4x	9.4x	5.5x	4.7x
Limited								

China Excluded Highest

\$5,435 \$6,056 1.0x 0.9x 0.5x 10.9x 8.9x 8.4x

- (1) Based on adjusted version of projections provided by Triplecrown s management used in connection with discounted cash flow analysis
- (2) Includes all capital expenditures and investment in working capital and the return on these assets during the investment period through 2012

Company Information	Market I (\$ in Mil Except p Share Pri	lions, er		t Multip rise Valu		Iultiple of:		
Company Name	Equity Value	EV	FY 2009 Revs	FY 2010 Revs	FY 2011 Revs	FY LTM EBITDA	FY 2010 EBITDA	FY 2011 EBITDA
Lowest	<b>\$87</b>	<b>\$164</b>	0.3x	0.3x	0.2x	6.2x	5.3x	4.7x
Mean	<b>\$1,467</b>	\$1,888	0.6x	0.6x	0.4x	8.1x	6.6x	6.1x
Median	\$639	\$1,134	0.5x	0.5x	0.5x	7.4x	6.1x	5.9x

Relevant market data for the six target companies in the M&A transactions used in the selected M&A transactions portion of the analysis is summarized in the table below:

Company Information	Growth	ı Ra	tes						Margin A	Anal	ysis					
Company Name	LTM Revs		2009 Revs		2010 Revs		2011 Revs		LTM EBITDA		2009 EBIT	DΑ	2010 EBITI	DΑ	2011 EBIT	DΑ
Bongrain SA	14.0	%	-3.1	%	2.8	%	1.4	%	5.1	%	5.5	%	6.1	%	3.7	%
Glanbia plc	1.2	%	-6.5	%	3.0	%	4.6	%	7.3	%	7.4	%	7.7	%	8.5	%
Saputo, Inc.	13.2	%	8.2	%	6.1	%	N/A		9.5	%	9.8	%	10.6	%	N/A	
Dairy Crest Group plc	11.2	%	0.0	%	0.2	%	-0.4	%	8.2	%	8.6	%	8.8	%	9.1	%
Warrnambool Cheese And Butter Factory Company Holdings Limited	20.6	%	-23.1	%	-5.3	%	8.5	%	3.4	%	2.9	%	7.9	%	8.6	%
China Mengniu Dairy Co.	19.2	%	0.5	%	15.3	%	18.6	%	-102.3	%	7.0	%	7.5	%	7.8	%
Ltd. <sup>(4)</sup>				70						70					7.0	70
Highest	20.6%	, D	8.2%		15.3%		18.6%		9.5%		9.8%		10.69		9.1%	
Lowest	1.2%		-23.1%	,	-5.3%	)	-0.4%	,	-102.3%	o	2.9%		6.1%		3.7%	
Mean	13.2%	,	-4.0%		3.7%		6.5%		-11.5%		6.9%	o o	8.1%		7.5%	Ó
Median	13.6%	,	-1.5%		2.9%		4.6%		6.2%		7.2%	o o	<b>7.8</b> %		8.5	%

Date Announced	Acquiror Name	Target Name	Enterpris Value	seLTM Revenue	LTM EBITD	EBITDA AMargin		EV/ ueEBITDA
6/23/2009	Saputo, Inc.	F&A Dairy of California, Inc.	\$44.50	\$140.00	N/A	N/A	0.3x	N/A
6/15/2009	KKR & Co.	Ma Anshan Modern Farming Co	750.0	N/A	N/A	N/A	N/A	N/A
6/15/2009	Dean Foods Co.	Alpro NV William	325.0	260.0	N/A	N/A	1.3x	N/A
10/22/2008	Saputo, Inc.	Neilson, Ltd.	465.0	600.0	50	8.33 %	0.8x	9.3x
1/29/2008	Saputo Cheese USA, Inc.	Alto Dairy Cooperative	160.0	378.0	19.6	5.19 %	0.4x	8.2x

10/16/2007	Wimm-Bill-Dann Foods OJSC		5.0	2.0	N/A	N/A	2.5x	N/A
		Limited						

Duff & Phelps used the selected public company and selected M&A transaction multiples shown in the tables above as data points in determining the appropriate valuation multiples to use in its analysis. Duff & Phelps noted that the selected M&A transaction multiples were in line with the selected public company multiples.

Duff & Phelps considered several key business and financial factors in the selection of the valuation multiples:

Cullen Agritech owns and operates one test farm and one model farm utilizing the intellectual property in Southeast U.S.

According to the management of Cullen Agritech, the existing test farm is break-even in the recent economic climate.

Cullen Agritech is not expected to generate revenues until 2010.

Cullen Agritech projects normalized earnings to occur starting in 2012.

Duff & Phelps noted that Cullen Agritech s revenue growth is at the high end of the peer group because it is a start-up company unlike most of the companies to which it is being compared. In addition, Duff & Phelps noted that there are no public dairy farms, and none of the selected comparable companies are identical to Cullen Agritech and its greenfield business model. Based on the foregoing, Duff & Phelps concluded that it would be appropriate to use an EBITDA multiple for the analysis in line with the dairy industry benchmarks.

Based on the foregoing. Duff & Phelps relied on the following determinations in deriving indications of value from its selected public company/selected M&A transaction analysis:

Duff & Phelps utilized a current (latest twelve months) EBITDA multiple range of 7.5x to 8.5x and applied this range of multiples to Cullen Agritech s projected EBITDA for 2012 as reflected in the adjusted management case projections used in the discounted cash flow analysis.

Duff & Phelps converted this 2012 future asset value to present value using the same cost of capital assumptions applied in the discounted cash flow analysis.

Duff & Phelps subtracted from this value the investment in tangible assets and working capital required to achieve steady state in the business model leaving a residual present value indication solely attributable to Cullen Agritech.

The foregoing selected public company/selected M&A transaction analysis indicated a range of values for Cullen Agritech of \$90 million to \$162 million.

## **Summary of Analyses**

The range of values for Cullen Agritech that Duff & Phelps derived from its discounted cash flow analysis was \$110 million to \$160 million, and the range of values for Cullen Agritech that Duff & Phelps derived from its selected public company/selected M&A transaction analysis was \$90 million to \$162 million. Duff & Phelps noted that the valuation of Cullen Agritech of \$155 million negotiated by the parties to the merger falls near the top of each of the ranges of value indications resulting from the discounted cash flow analysis and the selected public company/selected M&A transaction analysis.

Duff & Phelps opinion and financial analyses were only one of the many factors considered by Triplecrown s board of directors in its evaluation of the merger and should not be viewed as determinative of the views of its board of directors.

## **Fees and Expenses**

The Duff & Phelps engagement letter with Triplecrown, dated August 17, 2009, provides that, for its services, Duff & Phelps is entitled to receive a fee of \$150,000 from Triplecrown, which is due and payable as follows: \$85,000 non-refundable retainer upon execution of the engagement letter and \$65,000 upon Triplecrown formally requesting Duff & Phelps to issue its opinion. The engagement letter also provides that Duff & Phelps will be paid additional fees at its standard hourly rates for any time incurred should Duff & Phelps be called upon to support its findings subsequent to the delivery or the opinion, or assist in the preparation of or review of relevant sections of required SEC

Summary of Analyses 101

disclosures, proxy materials, or other documents associated with the merger. In addition, Triplecrown has agreed to reimburse Duff & Phelps for its reasonable out-of-pocket expenses and to indemnify Duff & Phelps and certain related persons against liabilities arising out of Duff & Phelps service as a financial advisor to the board of directors of Triplecrown.

48

Fees and Expenses 102

Other than the preparation of the opinion in connection with the merger, during the two years preceding the date of its opinion, Duff & Phelps has not had any material relationship with any party to the merger for which compensation has been received or is intended to be received, nor is any such material relationship or related compensation mutually understood to be contemplated. As of the date of the Duff & Phelps opinion, Cullen Agritech is considering utilizing the services of Duff & Phelps Securities, LLC, an affiliate of Duff & Phelps, LLC in connection with capital raising efforts Cullen Agritech may undertake if the proposed merger does not occur. Duff & Phelps Securities, LLC is qualified to perform such services. Duff & Phelps has advised Triplecrown that the professional fees Duff & Phelps Securities, LLC would receive if it is awarded this additional project, some of which may be contingent upon the amount and type of capital raised, would not be material to the financial and operating results of Duff & Phelps.

# Interests of Triplecrown s Directors and Officers and Others in the Merger

In considering the recommendation of the board of directors of Triplecrown to vote for the proposal to approve the merger proposal, you should be aware that Triplecrown s directors and officers have agreements or arrangements that provide them with interests in the merger that differ from, or are in addition to, those of Triplecrown stockholders generally. In particular:

If the merger is not consummated by October 22, 2009, Triplecrown's amended and restated certificate of incorporation provides that it will automatically be liquidated. In such event, the 13,800,000 Founders—Units held by Triplecrown's directors and officers that were acquired before the IPO for an aggregate purchase price of \$25,000 would be worthless because Triplecrown's directors and officers are not entitled to receive any of the liquidation proceeds with respect to such securities. The Founders—Warrants included in the Founders—Units are identical to the public warrants except that they will become exercisable after the merger if and when the last sales price of CAH is common stock exceeds \$13.75 per share for any 20 trading days within any 30-trading day period beginning 90 days after the consummation of the merger and the Founders—Warrants will be exercisable on a cashless basis and will not be redeemable by CAH, in each case, as long as they are held by the Founders or their permitted transferees. The Founders—Units had an aggregate market value of \$ based upon the unit is closing bid price of \$ on the NYSE Amex on , 2009, the record date for the Triplecrown special meeting. If the merger is consummated, the Triplecrown Founders have agreed that 11,260,000 shares of common stock of CAH they will receive in exchange for 11,260,000 Founders Shares upon consummation of the merger will be cancelled.

Eric J. Watson and Jonathan J. Ledecky also purchased 5,000,000 Sponsors Warrants, for an aggregate purchase price of \$5,000,000 (or \$1.00 per warrant), pursuant to agreements with Triplecrown and Citigroup that were entered into in connection with Triplecrown s IPO. These purchases took place on a private placement basis simultaneously with the consummation of Triplecrown s IPO. All of the proceeds Triplecrown received from these purchases were placed in Triplecrown s trust account. The Sponsors Warrants are identical to the Triplecrown warrants except that (i) the warrants will not be transferable or salable by holders (except in certain limited circumstances such as to relatives and trusts for estate planning purposes, provided the transferee agrees to be bound by the transfer restrictions) until Triplecrown completes a business combination, (ii) they will be exercisable on a cashless basis and (iii) if Triplecrown calls the warrants for redemption, the Sponsors Warrants will not be redeemable so long as such warrants are held by Messrs. Watson or Ledecky or their affiliates, including any permitted transferees. All of the Sponsors Warrants will become worthless if the merger is not consummated and Triplecrown is liquidated (as will the public warrants). Such Sponsors Warrants had an aggregate market value of \$, based on the warrants closing bid price of \$ on the NYSE Amex on , 2009, the record date for the Triplecrown special meeting.

Eric J. Watson, chairman of the board and treasurer of Triplecrown, controls Cullen Holdings, the sole stockholder of Cullen Agritech. Accordingly, he will benefit to the extent Cullen Holdings receives shares of CAH following the merger. Additionally, Mr. Watson will serve as chief executive officer and a director of CAH following the merger

and receive compensation in connection therewith.

In connection with the proposed business combination, Triplecrown and an affiliate of Mr. Watson entered into a contract to purchase a certain piece of land to be used by CAH following consummation of the merger. The total purchase price of the land is \$8,662,500 on which Triplecrown paid a deposit of approximately \$1.8 million. If Triplecrown is unable to complete the proposed business combination by the date it is required to liquidate, Mr. Watson and his affiliate have jointly and severally agreed to assume Triplecrown s remaining obligations under the purchase contract (approximately \$6.9 million) and Triplecrown will have no further obligations or liability whatsoever under the purchase contract.

If Triplecrown liquidates prior to the consummation of a business combination, Eric J. Watson, Triplecrown s chairman and treasurer, and Jonathan J. Ledecky, Triplecrown s president and secretary, will be personally liable to pay debts and obligations to vendors and other entities that are owed money by Triplecrown for services rendered or products sold to Triplecrown, or to any target business, to the extent such creditors bring claims that would otherwise require payment from monies in the trust account, but only if such entities did not execute a waiver. Based on Triplecrown s estimated debts and obligations, it is not currently expected that Messrs. Watson and Ledecky will have any exposure under this arrangement in the event of a liquidation will not seek recourse against the trust account notwithstanding such agreements. However, based on Triplecrown s resources outside of the trust account, it is not anticipated that Messrs. Watson or Ledecky will have any exposure under this arrangement.

If Triplecrown is required to be liquidated and there are no funds remaining to pay the costs associated with the implementation and completion of such liquidation, Eric J. Watson and Jonathan J. Ledecky have agreed to advance Triplecrown the funds necessary to pay such costs and complete such liquidation (currently anticipated to be no more than approximately \$15,000) and not to seek repayment for such expenses.

Upon consummation of the merger, the underwriters in Triplecrown s IPO will be entitled to receive \$19,320,000 of deferred underwriting commissions. Triplecrown is currently in negotiations with its underwriters to reduce the fee payable to them upon consummation of the merger. Additionally, two consultants for Triplecrown will receive an aggregate of 455,000 shares of CAH common stock upon closing of the merger. Such consultants have executed lockup agreements pursuant to which they have agreed that they will not sell any of such shares for a period of two years from the consummation of the transaction. Neither the underwriters nor the consultants will receive any of this consideration if the merger is not consummated.

## **Recommendation of Triplecrown s Board of Directors**

After careful consideration of the matters described above, Triplecrown s board of directors determined that the merger proposal is fair to and in the best interests of Triplecrown and its stockholders and that the merger agreement is no more favorable to Cullen Agritech than it would be to an unaffiliated third party. Triplecrown s board of directors has approved and declared advisable and recommend that you vote or give instructions to vote FOR the merger proposal.

The foregoing discussion of the information and factors considered by the Triplecrown board of directors is not meant to be exhaustive, but includes the material information and factors considered by the Triplecrown board of directors.

## **Conversion Rights**

Any of Triplecrown s stockholders holding Public Shares as of the record date who affirmatively vote their Public Shares against the merger proposal may also demand that such shares be converted into a pro rata portion of the trust account, calculated as of two business days prior to the consummation of the merger. If demand is properly made and the merger is consummated, these shares will be converted into a pro rata portion of funds deposited in the trust account plus interest, calculated as of such date.

Triplecrown stockholders who seek to exercise this conversion right ( converting stockholders ) must affirmatively vote against the merger proposal. Abstentions and broker non-votes do not satisfy this requirement. Additionally, holders demanding conversion must deliver their shares (either physically or electronically

50

Conversion Rights 106

through the Depository Trust Company) to Triplecrown s transfer agent up to the vote at the meeting. Accordingly, a holder will have at least fourteen days from the date notice of the meeting is mailed to stockholders to obtain a certificate if they intend to comply with the conversion requirements by physically delivering their shares to Triplecrown s transfer agent. As the delivery process can be accomplished by the stockholder in a matter of hours by simply contacting the transfer agent or his broker and requesting delivery of his shares through the DWAC System, it is believed that this time period is sufficient for an average investor. If you hold the shares in street name, you will have to coordinate with your broker to have your shares certificated or delivered electronically. Shares that have not been tendered (either physically or electronically) in accordance with these procedures will not be converted into cash. There is a nominal cost associated with the above-referenced tendering process and the act of certificating the shares or delivering them through the Depository Trust Company. The transfer agent will typically charge the tendering broker \$35 per transaction and it would be up to the broker whether or not to pass this cost on to the converting holder. This fee may discourage stockholders from seeking conversion rights and may make it more beneficial for such stockholders to try to sell their shares in the open market.

If the holders of at least 16,559,999 or more Public Shares (an amount equal to 30.0% or more of the Public Shares) vote against the merger proposal and properly demand conversion of their shares, Triplecrown will not be able to consummate the merger.

The closing bid price of Triplecrown s common stock on , 2009 (the record date for the Triplecrown special meeting) was \$. The cash held in the trust account on , 2009 was approximately \$ (\$ per Public Share). Prior to exercising conversion rights, stockholders should verify the market price of Triplecrown s common stock as they may receive higher proceeds from the sale of their common stock in the public market than from exercising their conversion rights if the market price per share is higher than the conversion price. Triplecrown cannot assure its stockholders that they will be able to sell their shares of Triplecrown common stock in the open market, even if the market price per share is higher than the conversion price stated above, as there may not be sufficient liquidity in Triplecrown s securities when Triplecrown s stockholders wish to sell their shares.

If you exercise your conversion rights, then you will be exchanging your shares of Triplecrown common stock for cash and will no longer own those shares. You will be entitled to receive cash for these shares only if you affirmatively vote against the merger proposal, properly demand conversion, and deliver your stock certificate (either physically or electronically) to Triplecrown s transfer agent up to the vote at the meeting.

If the merger is not consummated by the date Triplecrown must liquidate, either party may terminate the merger agreement. If Triplecrown is unable to complete the merger or another business combination by October 22, 2009, its amended and restated certificate of incorporation provides that its corporate existence will terminate on that date and, upon its resulting liquidation, the holders of Public Shares will receive an amount equal to the amount of funds in the trust account, inclusive of interest not previously released to Triplecrown, as well as any remaining net assets outside of the trust account, at the time of the liquidation distribution divided by the number of Public Shares. If the transaction is not consummated, a holder would have to wait until Triplecrown liquidates in connection with its dissolution to receive liquidation proceeds, which liquidation could take 60 days or more to complete. The amount a holder of Public Shares would receive at liquidation may be more or less than the amount such a holder would have received had it sought conversion of its shares in connection with the merger because (i) there will be greater earned interest in the trust account at the time of a liquidation distribution since it would occur at a later date than a conversion and (ii) Triplecrown may incur expenses it otherwise would not incur if Triplecrown consummates the merger, including, potentially, claims requiring payment from the trust account by creditors who have not waived their rights against the trust account. Eric J. Watson, Triplecrown s chairman and treasurer, and Jonathan J. Ledecky, Triplecrown s president and secretary, will be personally liable under certain circumstances (for example, if a vendor successfully makes a claim against funds in the trust account) to ensure that the proceeds in the trust account are not

Conversion Rights 107

reduced by the claims of prospective target businesses and vendors or other entities that are owed money by Triplecrown for services rendered or products sold to it, but only if such target business, vendor or other entity does not execute a waiver. While Triplecrown has no reason to believe that Messrs. Watson and Ledecky will not be able to satisfy those obligations, there cannot be any assurance to that effect. See the section entitled *Other Information Related to Triplecrown Liquidation If No Business Combination* for additional information.

51

Conversion Rights 108

# Actions that May Be Taken to Secure Approval of Triplecrown s Stockholders

Based on recently completed business combinations by other similarly structured blank check companies, it is believed by Triplecrown that the present holders of 30.0% or more of the Public Shares may have the intention to vote against the merger and seek conversion of their Public Shares into cash in accordance with Triplecrown's amended and restated certificate of incorporation. If such event were to occur, the merger could not be completed. To preclude such possibility, Triplecrown, the Triplecrown Founders, Cullen Agritech and their respective affiliates may negotiate arrangements to provide for the purchase of the Public Shares from holders who indicate their intention to vote against the merger and seek conversion or who otherwise wish to sell their Public Shares. The maximum cash purchase price that will be offered to the holders of Public Shares by Triplecrown, the Triplecrown Founders, Cullen Agritech and their respective affiliates for their shares will be the per-share conversion price at the time of the business combination. Although holders of Public Shares that enter into these types of arrangements with Triplecrown will not receive a higher purchase price than a holder that properly seeks conversion of his shares, entering into such arrangements (and agreeing to vote in favor of the merger) provides the holder with greater certainty that the transaction will be consummated, in which event such holder will receive his conversion proceeds promptly. If the transaction is not consummated, a holder would have to wait until Triplecrown liquidates in connection with its dissolution to receive liquidation proceeds, which liquidation could take 60 days or more to complete.

Triplecrown, the Triplecrown Founders, Cullen Agritech and their respective affiliates may also enter into transactions with potential investors or existing holders of Public Shares in order to induce them to purchase Public Shares and/or vote in favor of the merger proposal with respect to currently owned Public Shares and, in each case, to remain a stockholder of Two Harbors following consummation of the merger. These transactions would not result in any of Triplecrown, the Triplecrown Founders, Cullen Agritech or their respective affiliates purchasing any Public Shares and there would be no limit on the consideration that may be paid pursuant to these arrangements. The consideration paid to the potential investors or existing holders of Public Shares could be cash or other non-cash consideration (such as the transfer of shares or warrants held by the Triplecrown Founders).

It is anticipated that Triplecrown and/or Cullen Agritech would approach a limited number of large holders of Triplecrown that have voted against the merger proposal and demanded conversion of their shares, or that have indicated an intention to do so, and engage in direct negotiations for the purchase of such holders—positions. All holders approached in this manner would be institutional or sophisticated holders. Arrangements of such nature would only be entered into and effected in accordance with applicable law at a time when Triplecrown, the Triplecrown Founders, Cullen Agritech and/or their respective affiliates are not aware of any material nonpublic information regarding Triplecrown and its securities or pursuant to agreements between the buyer and seller of such shares in a form that would not violate insider trading rules. Definitive arrangements have not yet been determined but might include:

Agreements between Triplecrown and the holders of Public Shares pursuant to which Triplecrown would agree to purchase Public Shares from such holders immediately after the closing of the merger for the price and fees specified in the arrangements.

Agreements with third parties to be identified pursuant to which the third parties would purchase Public Shares during the period beginning on the date that the registration statement of which this prospectus/proxy statement/prospectus is a part is declared effective. Such arrangements would also provide for Triplecrown, immediately after the closing of the merger, to purchase from the third parties all of the Public Shares purchased by them for the price and fees specified in the arrangements.

Agreements with third parties pursuant to which the Triplecrown Founders, Cullen Agritech and their respective affiliates would borrow funds to make purchases of Public Shares for its own account. Triplecrown would repay such borrowings with funds transferred to it from Triplecrown s trust account upon closing of the merger.

As a result of the purchases that may be effected through such arrangements, it is likely that the number of shares of common stock of Triplecrown in its public float will be reduced and that the number of beneficial

holders of Triplecrown s and Triplecrown s securities also will be reduced. This may make it difficult to obtain the quotation, listing or trading of Triplecrown s securities on the New York Stock Exchange or any other national securities exchange.

The purpose of such arrangements would be to increase the likelihood of satisfaction of the requirements that the holders of a majority of Triplecrown common stock outstanding vote in favor of the merger proposal, the merger proposal is approved by the necessary vote of the holders of the Public Shares and that holders of fewer than 30.0% of the Public Shares vote against the merger proposal and demand conversion of their Public Shares into cash where it appears that such requirements would otherwise not be met. The maximum cash purchase price that will be offered by Triplecrown, the Triplecrown Founders, Cullen Agritech and their respective affiliates to holders of public shares for their shares will be the per-share conversion price at the time of the business combination. However, if holders refuse to enter into arrangements with Triplecrown to sell their Public Shares, Triplecrown may determine to engage a third party aggregator to buy shares prior to the meeting from such holders that have already indicated an intention to sell their shares and/or vote against the merger proposal. In such a case, the aggregator would purchase the shares from the original holder and then subsequently sell such shares to Triplecrown. The maximum purchase price that will be offered by such aggregators to holders of public shares for their shares will be the per-share conversion price at the time of the business combination. Triplecrown would, in addition to paying the purchase price of such shares (which would be the per-share conversion price) to this aggregator, pay it a fee. Such fee would typically be a small percentage of the aggregator s total purchase price for such shares. Although the parties do not anticipate needing to engage the services of such an aggregator, if one is needed, the parties believe it will be in the best interests of stockholders that are voting in favor of the transaction since the retention of the aggregator can help ensure that the transaction will be completed and the additional fee payable to the aggregator is not significant. All shares purchased pursuant to such arrangements would remain outstanding until the closing of the transaction and would be voted in favor of the merger proposal. Any agreement between the parties will provide for the holder to withdraw or revoke any exercise of its conversion exercise and grant a proxy to Triplecrown s designees to vote such shares in favor of the merger proposal at the meeting. Accordingly, this will effectively render the 30.0% threshold established in Triplecrowns IPO prospectus and amended and restated certificate of incorporation ineffective and make it easier for the parties to complete the transaction because such purchased shares would no longer be counted towards the 30.0% threshold. If, for some reason, the merger is not closed despite such purchases, the purchasers would be entitled to participate in liquidation distributions from Triplecrown s trust account with respect to such shares.

Triplecrown will file a Current Report on Form 8-K to disclose arrangements entered into or significant purchases or transfers made by any of the aforementioned persons, including aggregators, that would affect the vote on the merger proposal or the conversion threshold. Any such report will include descriptions of any arrangements entered into or significant purchases or transfers by any of the aforementioned persons. If members of Triplecrown s board of directors or officers make purchases or transfer shares pursuant to such arrangements, they will be required to report these purchases or transfers on beneficial ownership reports filed with the SEC.

Purchases pursuant to such arrangements ultimately paid for with funds in Triplecrown s trust account would diminish the funds available to CAH after the merger for working capital and general corporate purposes. In all events there will be sufficient funds available to Triplecrown from the trust account to pay the holders of all Public Shares that are properly converted.

It is possible that the special meetings could be adjourned to provide time to seek out and negotiate such transactions if, at the time of the meetings, it appears that the requisite vote will not be obtained or that the limitation on conversion will be exceeded, assuming that the stockholder adjournment proposal is approved. Also, under Delaware law, the board of directors may postpone the meetings at any time prior to it being called to order in order to provide time to seek out and negotiate such transactions.

# **Rescission Rights**

Triplecrown s amended and restated certificate of incorporation and the prospectus issued by Triplecrown in its IPO (i) states that Triplecrown is required to complete a business combination in which it acquired a

53

Rescission Rights 112

target business in the financial services industry having a fair market value equal to at least 80% of Triplecrown s trust account balance (excluding deferred underwriting discounts and commissions) and resulting in the ownership by Triplecrown of not less than 50% of the voting securities of the target business and prohibits Triplecrown from completing a business combination with a business affiliated with any of the Triplecrown Founders, (ii) did not disclose that funds in the trust account might be used to purchase Public Shares from holders thereof who have indicated their intention to vote against the merger and convert their shares into cash, and (iii) stated that specific provisions in Triplecrown s amended and restated certificate of incorporation may not be amended prior to the consummation of an initial business combination but that Triplecrown had been advised that such provision limiting its ability to amend its amended and restated certificate of incorporation may not be enforceable under Delaware law. Accordingly, each person who purchased Public Shares in the IPO and still held such shares upon learning of these facts may have securities law claims against Triplecrown for rescission (under which a successful claimant has the right to receive the total amount paid for his or her securities pursuant to an allegedly deficient prospectus, plus interest and less any income earned on the securities, in exchange for surrender of the securities) or damages (compensation for loss on an investment caused by alleged material misrepresentations or omissions in the sale of a security).

Such claims may entitle stockholders asserting them to as much as \$10.00 or more per share, based on the initial offering price of the IPO units comprised of stock and warrants, less any amount received from sale of the original warrants purchased with them, plus interest from the date of Triplecrown s IPO (which, in the case of holders of Public Shares, may be more than the pro rata share of the trust account to which they are entitled on conversion or liquidation).

In general, a person who purchased shares pursuant to a defective prospectus or other representation must make a claim for rescission within the applicable statute of limitations period, which, for claims made under Section 12 of the Securities Act and some state statutes, is one year from the time the claimant discovered or reasonably should have discovered the facts giving rise to the claim, but not more than three years from the occurrence of the event giving rise to the claim. A successful claimant for damages under federal or state law could be awarded an amount to compensate for the decrease in value of his or her shares caused by the alleged violation (including, possibly, punitive damages), together with interest, while retaining the shares. Claims under the anti-fraud provisions of the federal securities laws must generally be brought within two years of discovery, but not more than five years after occurrence. Rescission and damages claims would not necessarily be finally adjudicated by the time the merger may be completed, and such claims would not be extinguished by consummation of that transaction.

Even if you do not pursue such claims, others, who may include all holders of Public Shares, may. Neither Triplecrown nor CAH can predict whether Triplecrown stockholders will bring such claims, how many might bring them or the extent to which they might be successful.

# **Appraisal Rights**

In the event CAH s securities are not listed on a national securities exchange at the time the merger of CAH and Triplecrown is consummated, appraisal rights will be available to all Triplecrown stockholders pursuant to Section 262 of the DGCL. Appraisal rights are not available to holders of Triplecrown warrants. If appraisal rights are available, holders of shares of Triplecrown common stock who do not vote in favor of the merger proposal and who properly demand appraisal of their shares will be entitled to appraisal rights in connection with the merger under Section 262 of the DGCL. If the common stock of CAH is listed on a national securities exchange at the time the merger of CAH and Triplecrown is consummated, Triplecrown stockholders will not be entitled to assert appraisal rights under Section 262.

Appraisal Rights 113

Holders of Public Shares electing to exercise conversion rights will not be entitled to appraisal rights.

The following discussion is not a complete statement of the law pertaining to appraisal rights under the DGCL and is qualified in its entirety by the full text of Section 262, which is attached to this proxy statement/prospectus as Annex L. The following summary does not constitute any legal or other advice nor does it constitute a recommendation that stockholders exercise their appraisal rights, if any, under Section 262. All references in Section 262 and in this summary to a stockholder are to the record holder of the shares of common stock of Triplecrown as to which appraisal rights are asserted. A person having a beneficial interest in shares of common stock of Triplecrown held of record in the name of another person, such as a broker,

54

Appraisal Rights 114

fiduciary, depositary or other nominee, must act promptly to cause the record holder to follow the steps summarized below properly and in a timely manner to perfect appraisal rights, if available.

In the event that appraisal rights are available, under Section 262, holders of shares of common stock of Triplecrown who do not vote in favor of the merger proposal and who otherwise follow the procedures set forth in Section 262 will be entitled to have their shares appraised by the Delaware Court of Chancery and to receive payment in cash of the fair value of the shares, exclusive of any element of value arising from the accomplishment or expectation of the merger of CAH and Triplecrown, together with a fair rate of interest, if any, as determined by the court.

Under Section 262, where a merger or consolidation agreement is to be submitted for adoption at a meeting of stockholders, the corporation, not less than 20 days prior to the meeting, must notify each of its stockholders entitled to appraisal rights that appraisal rights are available and include in the notice a copy of Section 262. To the extent appraisal rights are available in connection with the merger of CAH and Triplecrown, this proxy statement/prospectus shall constitute the notice, and the full text of Section 262 is attached to this proxy statement as Annex L. In the event appraisal rights are available in connection with the merger of CAH and Triplecrown, any holder of common stock of Triplecrown who wishes to exercise appraisal rights, or who wishes to preserve such holder s right to do so, should review the following discussion and Annex L carefully because failure to timely and properly comply with the procedures specified will result in the loss of appraisal rights. Moreover, because of the complexity of the procedures for exercising the right to seek appraisal of shares of common stock, Triplecrown believes that if a stockholder considers exercising such rights, such stockholder should seek the advice of legal counsel.

# **Filing Written Demand**

If appraisal rights are available in connection with the merger of CAH and Triplecrown, any holder of common stock of Triplecrown wishing to exercise appraisal rights must deliver to Triplecrown, before the vote on the merger proposal, a written demand for the appraisal of the stockholder s shares, and that stockholder must not vote in favor of the adoption of the merger agreement. A holder of shares of Triplecrown common stock wishing to exercise appraisal rights must hold of record the shares on the date the written demand for appraisal is made and must continue to hold the shares of record through the effective time of the merger of CAH and Triplecrown. The stockholder must not vote in favor of the merger proposal. A proxy that is submitted and does not contain voting instructions will, unless revoked, be voted in favor of the merger proposal, and it will constitute a waiver of the stockholder s right of appraisal and will nullify any previously delivered written demand for appraisal. Therefore, a stockholder who submits a proxy and who wishes to exercise appraisal rights must submit a proxy containing instructions to vote against the merger proposal or abstain from voting on the adoption of the merger agreement. Neither voting against the adoption of the merger agreement nor abstaining from voting or failing to vote on the merger proposal will, in and of itself, constitute a written demand for appraisal satisfying the requirements of Section 262. The written demand for appraisal must be in addition to and separate from any proxy or vote on the merger proposal. The demand must reasonably inform Triplecrown of the identity of the holder, as well as the intention of the holder to demand an appraisal of the fair value of the shares held by the holder. A stockholder s failure to make the written demand prior to the taking of the vote on the merger proposal will constitute a waiver of appraisal rights.

If appraisal rights are available in connection with the merger of CAH and Triplecrown, only a holder of record of shares of Triplecrown common stock is entitled to assert appraisal rights for the shares registered in that holder s name. A demand for appraisal in respect of shares of common stock of Triplecrown should be executed by or on behalf of the holder of record, fully and correctly, as the holder s name appears on the holder s stock certificates, should specify the holder s name and mailing address and the number of shares registered in the holder s name and must state that the person intends thereby to demand appraisal of the holder s shares in connection with the merger of CAH and

Triplecrown. If the shares are owned of record in a fiduciary capacity, such as by a trustee, guardian or custodian, execution of the demand should be made in that capacity, and if the shares are owned of record by more than one person, as in a joint tenancy and tenancy in common, the demand should be executed by or on behalf of all joint owners. An authorized agent, including an agent for two or more joint owners, may execute a demand for appraisal on behalf of a holder of record; however, the agent must identify the record owner or owners and expressly disclose that, in executing the demand, the agent is acting as agent for the record owner or owners. If the shares are held in street

name by a broker, bank or nominee, the broker, bank or nominee may exercise appraisal rights with respect to the shares held for one or more beneficial owners while not exercising the rights with respect to the shares held for other beneficial owners; in such case, however, the written demand should set forth the number of shares as to which appraisal is sought, and where no number of shares is expressly mentioned, the demand will be presumed to cover all shares of common stock of Triplecrown held in the name of the record owner. Stockholders who hold their shares in brokerage accounts or other nominee forms and who wish to exercise appraisal rights are urged to consult with their brokers to determine the appropriate procedures for the making of a demand for appraisal by such a nominee.

All written demands for appraisal pursuant to Section 262 should be sent or delivered to Triplecrown Acquisition Corp. at 590 Madison Avenue, 21st Floor, New York, New York 10022, Attention Paul Vassilakos.

Any holder of common stock of Triplecrown may withdraw his, her or its demand for appraisal and accept the consideration offered pursuant to the merger agreement by delivering to CAH as the surviving entity of the nerger, a written withdrawal of the demand for appraisal. However, any such attempt to withdraw the demand made more than 60 days after the effective date of the merger will require written approval of the surviving corporation. No appraisal proceeding in the Delaware Court of Chancery will be dismissed without the approval of the Delaware Court of Chancery, and such approval may be conditioned upon such terms as the Court deems just.

# **Notice by the Surviving Corporation**

If appraisal rights are available in connection with the merger of CAH and Triplecrown, within 10 days after the effective time of the merger, CAH, as the surviving corporation, must notify each holder of common stock of Triplecrown who has made a written demand for appraisal pursuant to Section 262, and who has not voted in favor of the merger proposal, that the merger has become effective.

# Filing a Petition for Appraisal

Within 120 days after the effective time of the merger of CAH and Triplecrown, but not thereafter, CAH, as the surviving entity of the merger, or any holder of common stock of Triplecrown who has so complied with Section 262 and is entitled to appraisal rights under Section 262 may file a petition in the Delaware Court of Chancery demanding a determination of the fair value of the shares held by all dissenting holders. CAH, as the surviving entity is under no obligation to and has no present intention to file a petition, and holders should not assume that CAH will file a petition. Accordingly, it is the obligation of the holders of common stock of Triplecrown to initiate all necessary action to perfect their appraisal rights in respect of shares of common stock of Triplecrown within the time prescribed in Section 262.

Within 120 days after the effective time of the merger, any holder of common stock of Triplecrown who has complied with the requirements for exercise of appraisal rights will be entitled, upon written request, to receive from CAH a statement setting forth the aggregate number of shares not voted in favor of the merger proposal and with respect to which demands for appraisal have been received and the aggregate number of holders of such shares. The statement must be mailed within 10 days after a written request therefor has been received by the surviving corporation.

If a petition for an appraisal is timely filed by a holder of shares of common stock of Triplecrown and a copy thereof is served upon the surviving corporation, the surviving corporation will then be obligated within 20 days to file with the Delaware Register in Chancery a duly verified list containing the names and addresses of all stockholders who have demanded an appraisal of their shares and with whom agreements as to the value of their shares have not been reached. After notice to the stockholders as required by the court, the Delaware Court of Chancery is empowered to

conduct a hearing on the petition to determine those stockholders who have complied with Section 262 and who have become entitled to appraisal rights thereunder. The Delaware Court of Chancery may require the stockholders who demanded payment for their shares to submit their stock certificates to the Register in Chancery for notation thereon of the pendency of the appraisal proceeding, and if any stockholder fails to comply with the direction, the Court of Chancery may dismiss the proceedings as to such stockholder.

# **Determination of Fair Value**

After determining the holders of common stock of Triplecrown entitled to appraisal, the Delaware Court of Chancery will appraise the fair value of their shares, exclusive of any element of value arising from the accomplishment or expectation of the merger of CAH and Triplecrown, together with a fair rate of interest, if any, to be paid upon the amount determined to be the fair value. In determining fair value and, if applicable, a fair rate of interest, the Delaware Court of Chancery will take into account all relevant factors. In Weinberger v. UOP, Inc., the Supreme Court of Delaware discussed the factors that could be considered in determining fair value in an appraisal proceeding, stating that proof of value by any techniques or methods that are generally considered acceptable in the financial community and otherwise admissible in court should be considered, and that fair price obviously requires consideration of all relevant factors involving the value of a company. The Delaware Supreme Court stated that, in making this determination of fair value, the court must consider market value, asset value, dividends, earnings prospects, the nature of the enterprise and any other facts that could be ascertained as of the date of the Acquisition that throw any light on future prospects of the merged corporation. Section 262 provides that fair value is to be exclusive of any element of value arising from the accomplishment or expectation of the merger. In Cede & Co. v. Technicolor, Inc., the Delaware Supreme Court stated that such exclusion is a narrow exclusion that does not encompass known elements of value, but which rather applies only to the speculative elements of value arising from such accomplishment or expectation. In Weinberger, the Supreme Court of Delaware also stated that elements of future value, including the nature of the enterprise, which are known or susceptible of proof as of the date of the merger and not the product of speculation, may be considered.

Stockholders considering seeking appraisal should be aware that the fair value of their shares as so determined could be more than, the same as or less than the consideration they would receive pursuant to the Acquisition if they did not seek appraisal of their shares and that an investment banking opinion as to the fairness from a financial point of view of the consideration payable in a merger is not an opinion as to fair value under Section 262. Although Triplecrown believes that the exchange of Triplecrown common stock for CAH common stock is fair, no representation is made as to the outcome of the appraisal of fair value as determined by the Delaware Court of Chancery, and stockholders should recognize that such an appraisal could result in a determination of a value higher or lower than, or the same as, this consideration. Neither Triplecrown nor CAH anticipate offering more than the applicable shares of common stock of CAH to any stockholder of Triplecrown exercising appraisal rights, and each of Triplecrown and CAH reserves the right to assert, in any appraisal proceeding, that for purposes of Section 262, the fair value of a share of common stock of Triplecrown is less than the applicable shares of common stock of CAH, and that the methods which are generally considered acceptable in the financial community and otherwise admissible in court should be considered in the appraisal proceedings. In addition, Delaware courts have decided that the statutory appraisal remedy, depending on factual circumstances, may or may not be a dissenter s exclusive remedy. The Delaware Court of Chancery will also determine the amount of interest, if any, to be paid upon the amounts to be received by persons whose shares of common stock of Triplecrown have been appraised. If a petition for appraisal is not timely filed, then the right to an appraisal will cease. The costs of the action (which do not include attorneys fees or the fees and expenses of experts) may be determined by the Court and taxed upon the parties as the Court deems equitable under the circumstances. The Court may also order that all or a portion of the expenses incurred by a stockholder in connection with an appraisal, including, without limitation, reasonable attorneys fees and the fees and expenses of experts utilized in the appraisal proceeding, be charged pro rata against the value of all the shares entitled to be appraised.

If any stockholder who demands appraisal of shares of common stock of Triplecrown under Section 262 fails to perfect, or successfully withdraws or loses, such holder s right to appraisal, the stockholder s shares of common stock of Triplecrown will be deemed to have been converted at the effective time of the merger into the right to receive common stock of CAH. A stockholder will fail to perfect, or lose or withdraw, the holder s right to appraisal if no

petition for appraisal is filed within 120 days after the effective time of the merger or if the stockholder delivers to the surviving corporation a written withdrawal of the holder s demand for appraisal and an acceptance of the common stock of CAH in accordance with Section 262.

From and after the effective time of the merger of CAH and Triplecrown, no dissenting stockholder shall have any rights of a stockholder of Triplecrown with respect to that holder s shares for any purpose, except to

receive payment of fair value and to receive payment of dividends or other distributions on the holder s shares of common stock of Triplecrown, if any, payable to stockholders of Triplecrown of record as of a time prior to the effective time of the merger; provided, however, that if a dissenting stockholder delivers to the surviving company a written withdrawal of the demand for an appraisal within 60 days after the effective time of the merger, or subsequently with the written approval of the surviving company, then the right of that dissenting stockholder to an appraisal will cease and the dissenting stockholder will be entitled to receive only the merger consideration in accordance with the terms of the merger agreement. Once a petition for appraisal is filed with the Delaware court, however, the appraisal proceeding may not be dismissed as to any stockholder of Triplecrown without the approval of the court.

Failure to comply strictly with all of the procedures set forth in Section 262 of the DGCL may result in the loss of a stockholder s statutory appraisal rights. Consequently, any stockholder wishing to exercise appraisal rights is urged to consult legal counsel before attempting to exercise those rights.

# **Anticipated Accounting Treatment**

As a result of the proposed merger, Triplecrown will own all of the outstanding Cullen Agritech shares. Since, Cullen Agritech s Chief Executive Officer is the Chairman of the Board of Triplecrown, Cullen Agritech is deemed to be a related party. Consequently, Triplecrown s accounting for the purchase of Cullen Agritech will include consolidating Cullen Agritech s assets and liabilities at carrying value with those of Triplecrown. The cost of the purchase will be based on the par value of the Triplecrown common stock issued to the Cullen Agritech stockholder.

# **Regulatory Matters**

The merger and the transactions contemplated by the merger agreement are not subject to any additional federal or state regulatory requirement or approval, except for Hart-Scott-Rodino Antitrust Improvements Act of 1976 filings, filings with the State of Delaware necessary to effectuate the merger and the registration statement, of which this proxy statement/prospectus forms a part, shall have become effective and no stop order suspending its effectiveness, or proceeding to that effect, shall have been implemented by the SEC.

# **Required Vote**

The approval of the merger proposal will require (i) the affirmative vote of the holders of a majority of Triplecrown common stock outstanding on the record date and (ii) the affirmative vote of the holders of a majority of the Public Shares present (in person or represented by proxy) and entitled to vote at the Triplecrown special meeting. Additionally, the merger will not be consummated if the holders of 30% or more of the Public Shares vote against the merger proposal and properly demand that Triplecrown convert their Public Shares into their pro rata share of the trust account.

THE TRIPLECROWN BOARD OF DIRECTORS RECOMMENDS THAT THE TRIPLECROWN STOCKHOLDERS VOTE FOR THE APPROVAL OF THE MERGER PROPOSAL.

# THE MERGER AGREEMENT

For a discussion of the merger structure, merger consideration and indemnification provisions of the merger agreement, see the section entitled *The Merger Proposal*. Such discussion and the following summary of other material provisions of the merger agreement is qualified by reference to the complete text of the merger agreement, a copy of which is attached as Annex A to this proxy statement/prospectus and is incorporated herein by reference. All stockholders are encouraged to read the merger agreement in its entirety for a more complete description of the terms and conditions of the merger. The merger agreement has been included as an annex to this proxy statement/prospectus to provide investors and security holders with information regarding its terms. It is not intended to provide any other factual information about Triplecrown, Cullen Agritech or their respective subsidiaries and affiliates. The representations, warranties and covenants contained in the merger agreement were made only for purposes of that agreement and as of specific dates, were solely for the benefit of the parties to the merger agreement, may be subject to limitations agreed upon by the contracting parties, including being qualified by confidential disclosures made for the purposes of allocating contractual risk between the parties to the merger agreement instead of establishing these matters as facts, and may be subject to standards of materiality applicable to the contracting parties that differ from those applicable to investors. Stockholders are not third-party beneficiaries under the merger agreement and should not rely on the representations, warranties and covenants or any descriptions thereof as characterizations of the actual state of facts or condition of Triplecrown or Cullen Agritech or any of their respective subsidiaries or affiliates. Moreover, information concerning the subject matter of the representations, warranties and covenants may change after the date of the merger agreement, which subsequent information may or may not be fully reflected in public disclosures by CAH and Triplecrown.

# **Structure of the Merger**

The merger agreement provides for (i) the merger of Triplecrown with and into CAH with CAH surviving the merger and becoming the new publicly-traded corporation of which the present holders of Triplecrown securities will be security holders and (ii) the merger of Triplecrown Merger Sub with and into Cullen Agritech with Cullen Agritech surviving the merger and becoming a wholly-owned subsidiary of CAH.

# Closing and Effective Time of the Merger

The closing of the merger will take place promptly following the satisfaction of the conditions described below under the subsection entitled *Conditions to Closing of the Merger*, unless Triplecrown and Cullen Agritech agree in writing to another time. The merger is expected to be consummated promptly after the special meetings of Triplecrown s stockholders and warrantholders described in this proxy statement/prospectus.

# Indemnification

To provide a fund to secure the indemnification obligations of Cullen Holdings to Triplecrown against losses that the surviving entity of the merger may sustain as a result of (i) the inaccuracy or breach of any representation or warranty made by Cullen Agritech in the merger agreement or any schedule or certificate delivered by it in connection with the merger agreement and (ii) the non-fulfillment or breach of any covenant or agreement made by Cullen Agritech in the merger agreement, an aggregate of approximately 10% of the initial shares to be issued to Cullen Holdings in the merger (1,588,114 shares of Triplecrown common stock) will be placed in escrow (with an independent escrow agent), which will be canceled to the extent that Triplecrown has damages for which it is entitled to indemnification.

The escrow will be the sole remedy for Triplecrown for its rights to indemnification pursuant to the merger agreement. Claims for indemnification may be asserted against the escrow by Triplecrown once its damages exceed a deductible and will be reimbursable to the full extent of the damages in excess of such amount up to a maximum of the escrowed shares. Claims for indemnification may be asserted until the 30th day after the date CAH has filed with the SEC its Annual Report on Form 10-K for the year ending December 31, 2010. (but in any event no later than April 16, 2011). As a consequence of these limitations, Triplecrown may not be able to be entirely compensated for indemnifiable damages that it may sustain. A copy of this escrow agreement is attached to this proxy statement/prospectus as Annex E.

59

Indemnification 123

# **Representations and Warranties**

The merger agreement contains representations and warranties of each of Triplecrown, CAH, Triplecrown Merger Sub, Cullen Agritech and Cullen Holdings relating, among other things, to:

proper organization and similar limited liability and corporate matters; capital structure of each constituent company; the authorization, performance and enforceability of the merger agreement; licenses and permits;

taxes:

financial statements, information and absence of undisclosed liabilities; holding of leases and ownership of other properties, including intellectual property; contracts;

title to, and condition of, properties and assets and environmental and other conditions thereof; absence of certain changes;

SEC reports, financial statements and Sarbanes-Oxley Act; employee matters; compliance with laws; litigation; and regulatory matters and compliance.

# Covenants

The parties have each agreed to use commercially reasonable efforts to take such actions as are necessary, proper or advisable to consummate the merger. Triplecrown, CAH and Cullen Agritech have each also agreed to continue to operate their respective businesses in the ordinary course or in a commercially reasonable manner prior to the closing and, unless otherwise required or permitted under the merger agreement, not to take the following actions, among others, without the prior written consent of the other party:

waive any stock repurchase rights, accelerate, amend or (except as specifically provided for in the merger agreement) change the period of exercisability of options or restricted stock, or reprice options granted under any employee, consultant, director or other stock plans or authorize cash payments in exchange for any options granted under any of such plans;

grant any severance or termination pay to any officer or employee except pursuant to applicable law, written agreements outstanding, or policies existing on the date hereof and as previously or concurrently disclosed in writing or made available to the other party, or adopt any new severance plan, or amend or modify or alter in any manner any severance plan, agreement or arrangement existing on the date of the merger agreement;

except as indicated in the merger agreement, transfer or license to any person or otherwise extend, amend or modify any material rights to any intellectual property of Cullen Agritech or Triplecrown, as applicable, or enter into grants to transfer or license to any person future patent rights, other than in a commercially reasonable manner provided that in no event shall Cullen Agritech or Triplecrown license on an exclusive basis or sell any intellectual property of Cullen Agritech or Triplecrown as applicable;

declare, set aside or pay any dividends on or make any other distributions (whether in cash, stock, equity securities or property) in respect of any capital stock or split, combine or reclassify any capital stock or issue or authorize the issuance of any other securities in respect of, in lieu of or in substitution for any capital stock;

60

Covenants 125

except with respect to Triplecrown, purchase, redeem or otherwise acquire, directly or indirectly, any shares of capital stock or other equity securities or ownership interests, including repurchases of unvested shares at cost in connection with the termination of the relationship with any employee or consultant pursuant to agreements in effect on the date of the merger agreement;

except for the offering on commercially reasonable terms by Cullen Agritech of securities, debt or other equity of Cullen Agritech to raise capital in furtherance of its business plan ( Cullen Capital Raise ) issue, deliver, sell, authorize, pledge or otherwise encumber, or agree to any of the foregoing with respect to, any shares of capital stock or other equity securities or ownership interests or any securities convertible into or exchangeable for shares of capital stock or other equity securities or ownership interests, or subscriptions, rights, warrants or options to acquire any shares of capital stock or other equity securities or ownership interests or any securities convertible into or exchangeable for shares of capital stock or other equity securities or other ownership interests, or enter into other agreements or commitments of any character obligating it to issue any such shares, equity securities or other ownership interests or convertible or exchangeable securities;

amend its charter documents (except with respect to Triplecrown);

except with respect to the land being purchased in connection with the transactions contemplated under the merger agreement, acquire or agree to acquire by merging or consolidating with, or by purchasing any equity interest in or a portion of the assets of, or by any other manner, any business or any corporation, partnership, association or other business organization or division thereof, or otherwise acquire or agree to acquire any assets which are material, individually or in the aggregate, to the business of Triplecrown or Cullen Agritech as applicable, or enter into any joint ventures, strategic partnerships or alliances or other arrangements that provide for exclusivity of territory or otherwise restrict such party—s ability to compete or to offer or sell any products or services;

except as set forth in the merger agreement, sell, lease, license, encumber or otherwise dispose of any properties or assets, except (A) sales of inventory in a commercially reasonable manner, and (B) the sale, lease or disposition (other than through licensing) of property or assets that are not material, individually or in the aggregate, to the business of such party;

except with respect to Cullen Agritech s capital raising efforts, incur any indebtedness for borrowed money in excess of \$200,000 in the aggregate or guarantee any such indebtedness of another person or persons in excess of \$200,000 in the aggregate, issue or sell any debt securities or options, warrants, calls or other rights to acquire any debt securities of Triplecrown or Cullen Agritech, as applicable, enter into any keep well or other agreement to maintain any financial statement condition or enter into any arrangement having the economic effect of any of the foregoing; adopt or amend any employee benefit plan, policy or arrangement, any employee stock purchase or employee stock option plan, or enter into any employment contract or collective bargaining agreement (other than offer letters and letter agreements entered into in a commercially reasonable manner with employees who are terminable at will), pay any special bonus or special remuneration to any director or employee, or increase the salaries or wage rates or fringe benefits (including rights to severance or indemnification) of its directors, officers, employees or consultants, except in the ordinary course of business consistent with past practices;

except with respect to the land being purchased in connection with the transactions contemplated under the merger agreement, pay, discharge, settle or satisfy any claims, liabilities or obligations (absolute, accrued, asserted or unasserted, contingent or otherwise), or litigation (whether or not commenced prior to the date of this Agreement) other than the payment, discharge, settlement or satisfaction, in a commercially reasonable manner or in accordance with their terms, or liabilities recognized or disclosed in such party—s audited financial statements or in the most recent financial statements included in Triplecrown—s reports filed with the SEC, as applicable, or incurred since the date of such financial statements, or waive the benefits of, agree to modify in any manner, terminate,

61

Covenants 126

release any person from or knowingly fail to enforce any confidentiality or similar agreement to which Cullen Agritech is a party or of which Cullen Agritech is a beneficiary or to which Triplecrown is a party or of which Parent is a beneficiary, as applicable;

except in a commercially reasonable manner, modify, amend or terminate any material contract or waive, delay the exercise of, release or assign any material rights or claims thereunder;

except as required by U.S. GAAP, revalue any of Triplecrown s or Cullen Agritech s assets or make any change in accounting methods, principles or practices;

excet with respect to the land being purchased in connection with the transactions contemplated under the merger agreement and except in a commercially reasonable manner, incur or enter into any agreement, contract or commitment requiring such party to pay in excess of \$200,000 in any 12 month period;

engage in any action that could reasonably be expected to cause the mergers contemplated by the merger agreement to fail to qualify as a tax-free transactions pursuant to Section 351, Section 368(a)(1)(A) and Section 368(a)(2)(E) of the Code:

settle any litigation where the consideration given is other than monetary or to which an officer, director, Stockholder or holder of derivative securities of Triplecrown or Cullen Agritech is a party;

make or rescind any tax elections that, individually or in the aggregate, could be reasonably likely to adversely affect in any material respect the tax liability or tax attributes of such party, settle or compromise any material income tax liability or, except as required by applicable law, materially change any method of accounting for tax purposes or prepare or file any return in a manner inconsistent with past practice;

form, establish or acquire any subsidiary except as contemplated by the merger agreement; permit any person to exercise any of its discretionary rights under any plan to provide for the automatic acceleration of any outstanding options, the termination of any outstanding repurchase rights or the termination of any cancellation rights issued pursuant to such plans;

excet with respect to the land being purchased in connection with the transactions contemplated under the merger agreement, make capital expenditures except in accordance with prudent business and operational practices;

make or omit to take any action which would be reasonably anticipated to have a material adverse effect; enter into any transaction with or distribute or advance any assets or property to any of its officers, directors, partners, stockholders, managers, members or other affiliates other than the payment of salary and benefits in a commercially reasonable manner; or

agree, resolve or commit to do any of the foregoing.

Additionally, Cullen Holdings has agreed not to do any of the foregoing with respect to itself to the extent it prevents the consummation of the transactions contemplated by the merger agreement. No Cullen Capital Raise shall be consummated during the period commencing on the date of effectiveness of the Proxy Statement/Prospectus and ending on the earlier of the date of consummation of the Mergers or the termination of this Agreement.

The merger agreement also contains additional covenants of the parties, including covenants providing for:

the protection of confidential information of the parties and, subject to the confidentiality requirements, the provision of reasonable access to information;

Triplecrown and CAH to prepare and file a proxy statement and registration statement, which shall contain this proxy statement/prospectus, to solicit proxies from the Triplecrown stockholders and 62

Covenants 127

warrantholders to vote on the proposals that will be presented for consideration at the special meeting and to register, under the Securities Act, the CAH shares and warrants that will be issued to the securityholders of Triplecrown pursuant to the merger agreement;

CAH and Cullen Agritech to waive their respective rights to make claims against Triplecrown to collect from the trust fund established for the benefit of the holders of the Public Shares for any monies that may be owed to either by Triplecrown; and

Triplecrown and Cullen Agritech to use their commercially reasonable best efforts to cause the CAH securities issued in the merger to be listed on the NYSE or NASDAQ, subject to official notice of issuance, as of or prior to the effective time of the merger.

# **Conditions to Closing of the Merger**

# **General Conditions**

Consummation of the merger is conditioned on (i) the holders of (a) a majority of the Public Shares present and entitled to vote at a meeting called for this and other related purposes, approving the merger and (b) a majority of Triplecrown s outstanding common stock on the record date, at a meeting called for this and other related purposes, approving the merger proposal, (ii) the holders of fewer than 30.0% of the Public Shares voting against the merger and properly demanding that their Public Shares be converted into a pro-rata portion of the trust account, calculated as of two business days prior to the anticipated consummation of the merger, (iii) the holders of a majority of Triplecrown s common stock outstanding on the record date approving the initial charter proposals and the subsequent filing of Triplecrown s second amended and restated certificate of incorporation and (iv) the holders of a majority of Triplecrown s warrants outstanding on the record date approving the warrant amendment proposals.

In addition, the consummation of the transactions contemplated by the merger agreement is conditioned upon, among other things:

no statute, rule, ruling, regulation, judgment, decision, order, injunction, writ or decree shall have been enacted, entered, ordered, promulgated, issued or enforced by any court or other governmental authority that is in effect and prohibits, enjoins or restricts the consummation of the transactions;

the execution by and delivery to each party of each of the various transaction documents; the delivery by each party to the other party of a certificate to the effect that the representations and warranties of each party are true and correct as of the closing, and all covenants contained in the merger agreement have been materially complied with by each party;

the receipt of all material consents and approvals by third parties and the completion of necessary proceedings in compliance with the rules and regulations of each jurisdiction having jurisdiction over the subject matters; the amendment of Triplecrown s amended and restated certificate of incorporation to provide for the initial charter amendment:

the cancellation of a certain number of shares of Triplecrown common stock by the Triplecrown Founders; the receipt of legal opinions as may be mutually agreed upon by Cullen Agritech and Triplecrown; the lock-up agreements, the escrow agreement and other necessary transaction documents shall have been executed and delivered by the parties; and

the registration statement, of which this proxy statement/prospectus forms a part, shall have become effective and no stop order suspending its effectiveness, or proceeding to that effect, shall have been implemented by the SEC.

# Cullen Agritech s and Cullen Holdings Conditions to Closing

The obligations of Cullen Agritech and Cullen Holdings to consummate the transactions contemplated by the merger agreement also are conditioned upon, among other things:

there being no material adverse effect on Triplecrown since the date of the merger agreement; the initial charter, secondary charter and warrant amendment proposals have been approved; specified officers and directors of Triplecrown shall have resigned from their positions.

Although a condition to closing is the warrant amendment proposals being approved, such condition may be waived by Cullen Agritech or Cullen Holdings as indicated below under the section titled *Waiver*. Accordingly, in determining how to vote on the transaction, holders should understand that the transaction may be consummated without the warrants being amended if such condition is waived.

# Triplecrown s Conditions to Closing

The obligations of Triplecrown to consummate the transactions contemplated by the merger agreement also are conditioned upon each of the following, among other things,

there shall have been no material adverse effect on CAH since the date of the merger agreement; employment agreement between Natural Dairy, Inc. and Dr. Richard Watson shall have been executed; the transfer of certain intellectual property to Cullen Agritech; and specified officers and directors of Cullen Agritech shall have resigned from their positions.

# Waiver

If permitted under applicable law, either Triplecrown or Cullen Agritech may waive any inaccuracies in the representations and warranties made to such party contained in the merger agreement or in any document delivered pursuant to the merger agreement and waive compliance with any agreements or conditions (including but not limited to the condition that the warrant amendment proposals are approved) for the benefit of itself or such party contained in the merger agreement or in any document delivered pursuant to the merger agreement. The condition requiring that the holders of fewer than 30.0% of the Public Shares affirmatively vote against the merger proposal and demand conversion of their shares into cash may not be waived. Triplecrown would file a Current Report on Form 8-K and issue a press release to disclose any waiver of any representation, warranty or condition to the merger agreement. If such waiver is material to investors, a proxy statement/prospectus supplement would also be sent to holders of Public Shares as promptly as practicable. There can be no assurance that all of the conditions will be satisfied or waived.

At any time prior to the closing, either Triplecrown or Cullen Agritech may, in writing, to the extent legally allowed, extend the time for the performance of any of the obligations or other acts of the other parties to the merger agreement.

The existence of the financial and personal interests of the directors may result in a conflict of interest on the part of one or more of them between what he may believe is best for Triplecrown and what he may believe is best for himself in determining whether or not to grant a waiver in a specific situation.

# **Termination**

The merger agreement may be terminated at any time, but not later than the closing, as follows:

by mutual written agreement of Triplecrown and Cullen Agritech; by either Triplecrown or Cullen Agritech if the merger is not consummated by the date Triplecrown is required to liquidate, provided that such termination is not available to a party whose action or failure to act has been a principal cause of or resulted in the failure of the merger to be consummated before such date and such action or failure to act is

a breach of the merger agreement;

by either Triplecrown or Cullen Agritech if a governmental entity shall have issued an order, decree or ruling or taken any other action, in any case having the effect of permanently restraining, enjoining or otherwise prohibiting the merger, which order, decree, judgment, ruling or other action is final and nonappealable; 64

Termination 131

by either Triplecrown or Cullen Agritech if the other party has breached any of its covenants or representations and warranties in any material respect and has not cured its breach within thirty days of the notice of an intent to terminate, provided that the terminating party is itself not in breach; and

by either Triplecrown or Cullen Agritech if, at the Triplecrown stockholder meeting, the merger agreement shall fail to be approved by the affirmative vote of the holders of a majority of the Public Shares present (in person or represented by proxy) and entitled to vote at the meeting or the holders of 30% or more of the Public Shares exercise conversion rights.

# **Effect of Termination**

In the event of proper termination by either Triplecrown or CAH, the merger agreement will become void and have no effect, without any liability or obligation on the part of Triplecrown or CAH, except that:

The confidentiality obligations set forth in the merger agreement will survive; The waiver by Cullen Agritech and Cullen Holdings of all rights against Triplecrown to collect from the trust account any monies that may be owed to it by Triplecrown for any reason whatsoever, including but not limited to a breach of the merger agreement, and the acknowledgement that Cullen Agritech and Cullen Holdings will not seek recourse against the trust account for any reason whatsoever, will survive;

the rights of the parties to bring actions against each other for breach of the merger agreement will survive; and the fees and expenses incurred in connection with the merger agreement and the transactions contemplated thereby will be paid by the party incurring such expenses.

# **Fees and Expenses**

All fees and expenses incurred in connection with the merger agreement and the transactions contemplated thereby will be paid by the party incurring such expenses whether or not the merger is consummated, except that Triplecrown and CAH have agreed to reimburse Cullen Holdings, Cullen Agritech and their subsidiaries for costs incurred in connection with the transactions in the event such transactions are consummated.

# **Confidentiality; Access to Information**

Triplecrown and Cullen Agritech will afford to the other party and its financial advisors, accountants, counsel and other representatives prior to the completion of the merger reasonable access during normal business hours, upon reasonable notice, to all of their respective properties, books, records and personnel to obtain all information concerning the business, including the status of product development efforts, properties, results of operations and personnel, as each party may reasonably request. Triplecrown and Cullen Agritech will maintain in confidence any non-public information received from the other party, and use such non-public information only for purposes of consummating the transactions contemplated by the merger agreement, subject to customary exceptions.

# **Amendments**

The merger agreement may be amended by the parties thereto at any time by execution of an instrument in writing signed on behalf of each of the parties. Triplecrown would file a Current Report on Form 8-K and issue a press release to disclose any amendment to the merger agreement entered into by the parties. If such amendment is material to investors, a proxy statement/prospectus supplement would also be sent to holders of Public Shares as promptly as practicable.

Effect of Termination 132

# **Public Announcements**

The parties have agreed that until closing or termination of the merger agreement, the parties will:

cooperate in good faith to jointly prepare all press releases and public announcements pertaining to the merger agreement and the transactions governed by it; and

not issue or otherwise make any public announcement or communication pertaining to the merger agreement or the transaction without the prior consent of the other party, which shall not be unreasonably withheld by the other party, except as may be required by applicable law or court process.

65

Public Announcements 133

# U.S. FEDERAL INCOME TAX CONSIDERATIONS TO TRIPLECROWN STOCKHOLDERS

The following section is a summary of the opinion of Graubard Miller, counsel to Triplecrown, regarding material United States federal income tax consequences of the business combination and the merger to holders of Triplecrown common stock. This discussion addresses only those Triplecrown security holders that hold their securities as a capital asset within the meaning of Section 1221 of the Internal Revenue Code of 1986, as amended (the Code), and does not address all the United States federal income tax consequences that may be relevant to particular holders in light of their individual circumstances or to holders that are subject to special rules, such as:

financial institutions; investors in pass-through entities; tax-exempt organizations; dealers in securities or currencies;

traders in securities that elect to use a mark-to-market method of accounting; persons that hold Triplecrown common stock as part of a straddle, hedge, constructive sale or conversion transaction; and

persons who are not citizens or residents of the United States.

The Graubard Miller opinion is based upon the Code, applicable treasury regulations thereunder, published rulings and court decisions, all as currently in effect as of the date hereof, and all of which are subject to change, possibly with retroactive effect. Tax considerations under state, local and foreign laws, or federal laws other than those pertaining to the income tax, are not addressed.

Neither Triplecrown nor CAH intends to request any ruling from the Internal Revenue Service as to the U.S. federal income tax consequences of the business combination and the mergers.

It is the opinion of Graubard Miller the business combination and the merger of Triplecrown and CAH will constitute a tax-free transaction pursuant to Section 351 and a reorganization within the meaning of Section 368(a) of the Internal Revenue Code and that no gain or loss will be recognized by Triplecrown or by the stockholders of Triplecrown if their conversion rights are not exercised.

The U.S. federal tax basis of the stock of CAH received by the holder of Triplecrown common stock in the merger will be the same as the adjusted tax basis of the Triplecrown common stock surrendered in exchange therefore. The holding period of the stock of CAH received in the merger by a holder of Triplecrown common stock will include the period during which such Triplecrown common stock was held as a capital asset on the date of the merger.

It is also the opinion of Graubard Miller that a stockholder of Triplecrown who exercises conversion rights and effects a termination of the stockholder s interest in Triplecrown will be required to recognize gain or loss upon the exchange of that stockholder s shares of common stock of Triplecrown for cash. Such gain or loss will be measured by the difference between the amount of cash received and the tax basis of that stockholder s shares of Triplecrown common stock. This gain or loss will be a capital gain or loss if such shares were held as a capital asset on the date of the business combination and will be a long-term capital gain or loss if the holding period for the share of Triplecrown common stock is more than one year.

The conclusions expressed above are based on current law. Future legislative, administrative or judicial changes or interpretations, which can apply retroactively, could affect the accuracy of those conclusions. The tax opinion issued

to Triplecrown by Graubard Miller, its counsel, is attached to this proxy statement/prospectus as Annex F. Graubard Miller has consented to the use of its opinion in this proxy statement/prospectus.

This discussion is intended to provide only a summary of the material United States federal income tax consequences of the merger. It does not address tax consequences that may vary with, or are contingent on, your individual circumstances. In addition, the discussion does not address any non-income tax or any foreign, state or local tax consequences of the business combination. Accordingly, you are strongly urged to consult with your tax advisor to determine the particular United States federal, state, local or foreign income or other tax consequences to you of the business combination.

# UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION (IN DOLLARS)

Cullen Agritech and Triplecrown are providing the following unaudited pro forma condensed combined financial information to aid you in your analysis of the financial aspects of the merger.

The following unaudited pro forma condensed combined statement of operations for the six months ended June 30, 2009, statement of operations for the year ended December 31, 2008 and balance sheet at June 30, 2009 are based on the historical financial statements of Triplecrown and Cullen Agritech after giving effect to the merger.

The unaudited condensed combined pro forma statements of operations for the six months ended June 30, 2009 and the year ended December 31, 2008 give pro forma effect to the merger as if it had occurred on January 1, 2008. The unaudited pro forma condensed combined balance sheet at June 30, 2009 assumes that the merger was effective on June 30, 2009.

The unaudited condensed combined pro forma statement of operations for the six months ended June 30, 2009 was derived from Triplecrown s unaudited condensed financial statements for the six months ended June 30, 2009, and the unaudited condensed combined pro forma statement of operations for the year ended December 31, 2008 was derived from Triplecrown s audited financial statements for the year ended December 31, 2008. The unaudited pro forma condensed combined balance sheet at June 30, 2009 was derived from Triplecrown s unaudited condensed financial statements and Cullen Agritech audited financial statements as of June 30, 2009 and August 18, 2009, respectively.

As a result of the proposed merger, Triplecrown will own all of the outstanding Cullen Agritech shares. Since, Cullen Agritech s Chief Executive Officer is the Chairman of the Board of Triplecrown, Cullen Agritech is deemed to be a related party. Consequently, Triplecrown s accounting for the purchase of Cullen Agritech will include consolidating Cullen Agritech s assets and liabilities at carrying value with those of Triplecrown. The cost of the purchase will be based on the par value of the Triplecrown common stock issued to the Cullen Agritech stockholder.

Consummation of the merger is conditioned on (i) the holders of (a) a majority of the Public Shares present and entitled to vote at a meeting called for this and other related purposes, approving the merger and (b) a majority of Triplecrown s outstanding common stock on the record date, at a meeting called for this and other related purposes, approving the merger proposal, (ii) the holders of fewer than 30.0% of the Public Shares voting against the merger and properly demanding that their Public Shares be converted into a pro-rata portion of the trust account, calculated as of two business days prior to the anticipated consummation of the merger, (iii) the holders of a majority of Triplecrown s common stock outstanding on the record date approving the initial charter proposals and the subsequent filing of Triplecrown s second amended and restated certificate of incorporation and (iv) the holders of a majority of Triplecrown s warrants outstanding on the record date approving the warrant amendment proposals.

assuming no conversions this presentation assumes that no stockholders of Triplecrown seek to convert their shares into a pro rata share of the trust account;

assuming maximum conversions this presentation assumes stockholders of Triplecrown owning 29.99% of the stock sold in Triplecrown s initial public offering seek conversion.

The unaudited pro forma condensed combined financial statements are provided for informational purposes only and are subject to a number of uncertainties and assumptions and do not purport to represent what the companies actual

performance or financial position would have been had the transaction occurred on the dates indicated and does not purport to indicate the financial position or results of operations as of any future date or for any future period. Please refer to the following information in conjunction with the accompanying notes to these pro forma financial statements and the historical financial statements and the accompanying notes thereto and the sections entitled Unaudited Pro Forma Condensed Combined Financial Data, Triplecrown s Management s Discussion and Analysis of Financial Condition and Results of Operations, and Cullen Agritech Management s Discussion and Analysis of Financial Condition and Results of Operations Cullen Agritech Investment Corp and Subsidiaries Unaudited Condensed Combined Pro Forma Statement of Operations For the Year Ended December 31, 2008.

# **UNAUDITED PRO FORMA CONDENSED COMBINED** STATEMENT OF OPERATIONS **DECEMBER 31, 2008** (in dollars)

	Triplecrown Acquisition Corp. USD \$ Note 1	Cullen Agricultura Technologic Inc. USD \$ Note 2	l No es	Pro Forma Adjustments te No Conversion USD \$	S	Pro Forma No Conversion USD \$	Pro Forma Adjustments Maximum Conversion USD \$	No	Pro Forma Combined Maximum Conversion USD \$
Net sales	\$	\$		\$		\$	\$		\$
Cost of goods sold Gross profit Salling, general and									
Selling, general and administrative	1,484,138		3	100,000		1,584,138			1,584,138
Organizational Costs Professional Fees		6,200 15,013				6,200 15,013			6,200 15,013
Travel		753				753			- ,
Website Costs		396				396			396
Income (loss) from operations Interest and other (income) expense	(1,484,138)	(22,362)		(100,000	)	(1,606,500)			(1,606,500)
Interest expense	(0.442.226)		1	202.720		(0.050.500.)	0.541.551	_	(5.510.047.)
Dividend and Interest income	(8,443,326) (8,443,326)		1	383,728 383,728		(8,059,598) (8,059,598)	2,541,551 2,541,551	5	(5,518,047) (5,518,047)
Income (loss) before income taxes	6,959,188	(22,362)		(483,728	)	6,453,098	(2,541,551)		3,911,547
Income tax provision (benefit) Net income (loss)	2,300,380 4,658,808	(22,362)		(483,728	)	2,300,380 4,152,718	(2,541,551)		2,300,380 1,611,166
Accretion of trust account, relating to Common Stock subject to possible conversion Change in fair value of	583,997		2	(583,997	)				
currency option									
Net income (loss) available to common stockholders	\$4,074,811	\$(22,362)		\$100,268		\$4,152,718	\$(2,541,551)		\$1,611,167
Shares Outstanding	69,000,000			5,076,148		74,076,148	(16,559,999)		57,516,149
Weighted average number of shares	52,440,001		4	21,636,147	7	74,076,148	(16,599,999)	7	57,516,149
Basic and diluted net income per share	\$0.08	\$				\$0.06			\$0.03



# NOTES TO UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL STATEMENTS DECEMBER 31, 2008 (in dollars)

# Pro Forma Income Statement Adjustments

# Assuming no conversions:

Adjustment reflects the reduction of interest income due to the lower balance of cash and cash equivalents following payment of (i) approximately \$19,320,000 for the deferred underwriting fee related to Triplecrown's initial public offering and (ii) \$5,000,000 of estimated legal and other advisory fees and expenses incurred in connection with the merger. The calculation assumes an average rate of return of 1.58%, approximately the rate of interest earned by Triplecrown on its investments during the year ended December 31, 2008.

Decrease interest income 383,728

Note 2 Adjustment reflects the elimination of the accretion related to the trust account, relating to common stock

Decrease accretion of trust account, relating to common stock

583,997

Note Adjustment in selling, general and administrative due to additional employment agreements entered into as part of the acquisition agreement

Increase in selling, general and administrative expense

100,000

Note To record issuance of shares 15,881,148 shares of Triplecrown common stock to holders of common stock and preferred stock of CAT in connection with the merger.

Existing Triplecrown Shares	69,000,000
Decrease cash held in trust account	455,000
Increase cash and cash equivalents	(11,260,000)
Shares Repurchased	(37,339,548)
Shares issued to CAT stockholders in connection with merger	15,881,148
Total	74,076,148
Increase	

74,076,148

Weighted average shares outstanding-basic

# **Assuming maximum conversions**

Adjustment reflects the reduction of interest income due to the lower balance of cash and cash equivalents following payment of approximately \$161,078,990 assuming the holders of 29.99% of the Triplecrown common stock sold in its initial public offering exercise their right to convert their stock into a pro rata share of the trust account. The calculation assumes an average rate of return of 1.50%, approximately the rate of interest earned by Triplecrown on its investments during the year ended December 31, 2008.

Decrease interest income

2,541,551

Note o

e Adjustment reflects the reduction in shares of common stock outstanding resulting from the holders of 29.99% of the Triplecrown common stock sold in its initial public offering exercising their right to convert their stock into a pro rata share of the trust account

Decrease

16,559,999

Maximum number of shares to be converted

Total

57,516,149

Weighted average shares outstanding, basic and diluted

Common stock equivalents consisting of warrants were not included in the calculation of the diluted per share because it is not certain that these securities would be in the money subsequent to the merger. Potentially dilutive securities of 55,200,000 warrants (included within the units sold in the IPO), 5,000,000 incentive warrants purchased by the founders and 13,800,000 warrants as part of the Founder units have been excluded from the computation of diluted net income (loss) per share, because the effect would be anti-dilutive.

# **UNAUDITED PRO FORMA CONDENSED COMBINED BALANCE SHEET** June 30, 2009

1								
	Triplecrown Acquisition Corp. USD \$ Note 1	Cullen Agricultura Technologic Inc. USD \$ Note 2	Note es	Pro Forma Adjustments No Conversion USD \$	Pro Forma No Conversion USD \$	Pro Forma Adjustments Maximum Conversion USD \$	Note	Pro Forma Combined Maximum Conversio USD \$
ts ent Assets								ĺ
and cash equivalents unts receivable	\$2,856,870	\$	3,8,10	\$514,433,993	\$517,290,863	\$(161,662,987)	11	\$355,627,
aid expenses and other nt assets ntories, net	1,232,761	19,241			1,252,002			1,252,00
r current assets								•
current assets	4,089,631	19,241		514,433,993	518,542,865	(161,662,987)		356,879,
held in Trust Account	539,259,567		3	(539,259,567)				60.06 <b>0</b>
ued Interest	68,962	2 227			68,962 3,327			68,962 3 327
gible assets, net assets	\$543,418,160	3,327 \$22,568		\$(24,825,574)		\$(161,662,987)		3,327 \$356,952,
lities and stockholders' y	Ф <i>Э</i> ТЭ,Т10,100	φ <i>22,50</i> 0		Ψ( <del>Δ1,023,311</del> )	φυ10,010,10-1	φ(101,002,701)		φ <i>υυ</i> ,,,,,,,,
ent liabilities								ĺ
overdraft	\$505,574	\$		\$(505,574)	\$	\$		\$
unts payable and Accrued nses	265,025	44,730			309,755			309,755
current liabilities t-term debt, net of current	770,599	44,730		(505,574)	309,755			309,755
on I liabilities	770,599	44,730		(505,574)	309,755			309,755
mon stock, subject to		•	<b>7</b> 0		307,733			307,100
ble conversion cholders' equity	161,798,549		5,9	(161,798,549)				
mon stock, no par value,	5 244	200	£.	200	5 750			5 750
mon stock sury Stock	5,244	200	6	308	5,752			5,752
tional paid-in capital mulated other	375,094,887		5,6,7,8,9	137,455,879	512,550,766	(161,662,987)	11	350,887
rehensive income (loss) ned earnings	5,748,881	(22,362)	7	22,362	5,748,882			5,748,88
illed carnings	3,740,001	(22,302)	/	127,470,540	5,740,002	(1.61.660.007)		3,740,00

380,849,012 (22,162)

stockholders' equity

356,642.

137,478,549 518,305,399 (161,662,987)

liabilities and holders' equity 70

\$543,418,160 \$22,568

\$(24,825,574) \$518,615,154 \$(161,662,987)

\$356,952,

# UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF OPERATIONS June 30, 2009

	Triplecrown Acquisition Corp. USD \$ Note 1	Cullen Agricultura Technologic Inc. USD \$ Note 2	l Note es	Pro Forma Adjustments No Conversion USD \$	3	Pro Forma No Conversion USD \$	Pro Forma Adjustments Maximum Conversion USD \$	Not	Pro Forma Combined Maximum Conversion USD \$
Net sales	\$	\$		\$		\$	\$		\$
Cost of goods sold Gross profit									
Selling, general and administrative	1,267,364		14	100,000		1,367,364			1,367,364
Organizational Costs Professional Fees Travel		6,200 15,013 753				6,200 15,013 753			6,200 15,013 753
Website Costs Income (loss) from operations Interest and other (income) expense Interest expense	(1,267,364)	396 (22,362)		(100,000	)	396 (1,389,726)			396 (1,389,726)
Other (income) expense Dividend and Interest income	(172,901 ) (172,901 )		12	7,800 7,800		(165,101 ) (165,101 )	51,850 51,850	16	(113,251 ) (113,251 )
Income (loss) before income taxes	(1,094,463)	(22,362)		(107,800	)	(1,244,625)	(51,850		(1,276,475)
Income tax provision (benefit) Net income (loss) Accretion of trust account,	(360,139 ) (734,324 )	(22,362)		(107,800	)	(360,139 ) (864,486 )	(51,850		(360,139 ) (916,336 )
relating to Common Stock subject to possible conversion Change in fair value of currency option	135,562		13	(135,562	)				
Net income (loss) available to common stockholders	\$(869,886)	\$(22,362)		\$27,761		\$(864,486)	\$(51,850)		\$(916,336 )
Shares Outstanding	69,000,000			5,076,148		74,076,148	(16,559,999)		57,516,149
Weighted average number of shares	52,440,001		15	21,636,147	7	74,076,148	(16,559,999)	17	57,516,149
Basic and diluted net income per share 71	\$(0.02)	\$				\$(0.01)			\$(0.02)



# NOTES TO UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL STATEMENTS June 30, 2009

Note 1 Derived from the audited financial statements of Triplecrown Acquisition Corp. as of December 31, 2008. Note 2 Derived from the audited consolidated financial statements of CAT Corporation as of August 18, 2009.

## **Pro Forma Balance Sheet Adjustments**

Note Reclassification to reflect the release of \$539,259,567 of restricted cash held in trust and the transfer of the balance to cash and cash equivalents.

Decrease cash held in trust account 539,259,567 Increase cash and cash equivalents 539,259,567

Note 4 Intentionally left blank

Note 5 Reflects the reclassification of the Triplecrown common stock subject to conversion to stockholders equity assuming no holders of Triplecrown common stock sold in its initial public offering exercise their right to convert their common stock into a pro rata share of cash in trust.

Decrease common stock subject to possible conversion 161,662,987 Increase additional paid in capital 161,662,987

Record 15,881,148 shares of Triplecrown common stock issued to holders of common stock of CAT and new Note shares of 455,000 issued to Consultants as well as a reduction due of 11,260,000 shares cancelled by the

founders. A deemed distribution has been recorded due to a related party transaction whereby the carrying value of the assets acquired was zero.

Shares Issued to CAT	15,881,148
New shares issued to Consultants	455,000
Shares cancelled by Founders and Board	(11,260,000)
Net Shares Issued at closing	5,076,148
Increase common stock issued	508
Decrease common stock related to CAT	200
Decrease additional paid in capital	308

Note 7 Adjustment reflects the elimination CAT's historical retained earnings as a result of the merger

Increase additional paid in capital (22,362 )
Decrease retained earnings (22,362 )

Note Record payment of \$19,320,000 in deferred underwriting fees related to Triplecrown's initial public offering and \$5,000,000 of estimated legal and other advisory fees and expenses incurred in connection with the merger

Decrease cash and cash equivalents

24,320,000

Note	Decrease additional paid in capital Adjustment reflects the reversal of Triplecrown's accretion related to the trustock	24,320,000 ast account, relating to common
72	Decrease common stock, subject to possible conversion Increase additional paid in capital	135,562 135,562

#### TABLE OF CONTENTS

Note 10 Reclassification of cash overdraft shich is offset by cash in trust.

Decrease cash overdraft 505,574
Decrease cash and cash equivalents 505,574

Assuming maximum conversions

Note 11 The adjustment reflects the payment of cash to the holders of 29.99% of the Triplecrown common stock sold in its initial public offering upon exercising their right to convert their common stock into a pro rata share of the trust account (which reflects the maximum number of shares eligible for such conversion election).

Decrease additional paid in capital 161,662,987
Decrease cash and cash equivalents 161,662,987

## **Pro Forma Income Statement Adjustments**

Adjustment reflects the reduction of interest income due to the lower balance of cash and cash equivalents following payment of (i) approximately \$19,320,000 for the deferred underwriting fee related to Triplecrown's initial public offering and (ii) \$5,000,000 of estimated legal and other advisory fees and expenses incurred in connection with the merger. The calculation assumes an average rate of return of 0.03%, approximately the rate of interest earned by Triplecrown on its investments during the six months ended June 30, 2009.

Decrease interest income 7.800

Note 13 Adjustment reflects the elimination of the accretion related to the trust account, relating to common stock.

Decrease accretion of trust account, relating to common stock 135,562

Note Adjustment in selling, general and administrative due to additional employment agreements entered into as part of the acquisition agreement

Increase in selling, general and administrative expense 100,000

Note To record issuance of shares 15,881,148 shares of Triplecrown common stock to holders of common stock and preferred stock of CAT in connection with the merger.

Existing Triplecrown Shares	69,000,000
Decrease cash held in trust account	455,000
Increase cash and cash equivalents	(11,260,000)
Shares issued to CAT stockholders in connection with merger	15,881,148
Total	74,076,148

Increase

74,076,148

Weighted average shares outstanding-basic

Assuming maximum conversions

Note 16 Adjustment reflects the reduction of interest income due to the lower balance of cash and cash equivalents following payment of approximately \$161,662,987 assuming the holders of 29.99% of the Triplecrown common stock sold in its initial public offering exercise their right to convert their stock into a pro rata share of the trust account. The calculation assumes an average rate of return of 0.03%, approximately the rate of interest earned by Triplecrown on its investments during the six months ended June 30, 2009.

Decrease interest income

51,850

Note 17

Adjustment reflects the reduction in shares of common stock outstanding resulting from the holders of 29.99% of the Triplecrown common stock sold in its initial public offering exercising their right to convert their stock into a pro rata share of the trust account.

Decrease

16,559,999

Maximum number of shares to be converted Total

57,516,149

Weighted average shares outstanding, basic and diluted

Common stock equivalents consisting of warrants were not included in the calculation of the diluted per share because it is not certain that these securities would be in the money subsequent to the merger. Potentially dilutive securities of 55,200,000 warrants (included within the units sold in the IPO), 5,000,000 incentive warrants purchased by the founders and 13,800,000 warrants as part of the Founder units have been excluded from the computation of diluted net income (loss) per share, because the effect would be anti-dilutive.

## THE SECONDARY CHARTER PROPOSALS

The secondary charter proposals, if approved, will approve the following differences between the amended and restated certificate of incorporation of CAH to be in effect following the merger and Triplecrown s current amended and restated certificate of incorporation:

the name of the new public entity will be Cullen Agricultural Holding Corp. as opposed to Triplecrown Acquisition Corp.;

CAH will have 200,000,000 authorized shares of common stock and 1,000,000 authorized shares of preferred stock and may increase or decrease such amounts without stockholder approval, as opposed to Triplecrown having 160,000,000 authorized shares of common stock and 1,000,000 authorized shares of preferred stock and may not increase or decrease such amounts without stockholder approval;

CAH s corporate existence will be perpetual as opposed to Triplecrown s corporate existence terminating on October 22, 2009; and

CAH s amended and restated certificate of incorporation will not include the various provisions applicable only to specified purpose acquisition corporations that Triplecrown s amended and restated certificate of incorporation contains (Article Seventh).

As noted above, the provisions of Article Seventh of Triplecrown s amended and restated certificate of incorporation will not be included in CAH s certificate of incorporation, which (other than Section K) by the terms of the preamble, apply only during the period that will terminate upon the consummation of the business combination that will be effected by the merger. Section A requires that the business combination be submitted to Triplecrown s stockholders for approval under the DGCL and is authorized by the vote of a majority of the Public Shares, provided that the business combination shall not be consummated if the holders of 30.0% or more of the Public Shares exercise their conversion rights. Section B specifies the procedures for exercising conversion rights. Section C provides that, if a business combination is not consummated by the Termination Date (October 22, 2009), only the holders of the Public Shares will be entitled to receive liquidating distributions. Section D provides that holders of Public Shares are entitled to receive distributions from Triplecrown s trust account established in connection with its IPO only in the event of Triplecrown s liquidation or by demanding conversion in accordance with Section B. Section E provides that no other business combination may be consummated until Triplecrown s initial business combination meeting all of the requirements set forth in its amended and restated certificate of incorporation is consummated and prohibits Triplecrown from consummating a business combination with an entity affiliated with any Triplecrown Founder. Section F provides what amounts may be released from Triplecrown s trust account. Sections G and H provide that the audit committee must approve certain actions until consummation by Triplecrown of its initial business combination. Section I provides that Triplecrown may not issue any additional securities that share in the trust account. Section J provides for Triplecrown s board to be staggered in classes.

In the judgment of Triplecrown s board of directors, the secondary charter proposals are desirable for the following reasons:

The name of the new public entity is desirable to reflect the transaction with CAH. The greater number of authorized number of shares of capital stock is desirable for CAH to have sufficient stock to issue to the holders of common stock to complete the merger and have additional authorized shares of common and preferred stock for financing its business, for acquiring other businesses, for forming strategic partnerships and alliances and for stock dividends and stock splits.

Although the present amended and restated certificate of incorporation provides that Triplecrown s corporate existence will terminate on October 22, 2009, perpetual existence is the usual period of existence for corporations and Triplecrown s board of directors believes it is the most appropriate period for CAH.

The preamble and Article Seventh (other than Section K providing for a classified board) relate to the operation of Triplecrown as a blank check company prior to the consummation of its initial business combination and would not be applicable to CAH after consummation of the merger. Accordingly, they would serve no further purpose. Notwithstanding the foregoing, authorized but unissued shares of common stock may enable CAH s board of directors to render it more difficult or to discourage an attempt to obtain control of CAH and thereby protect continuity of or entrench its management, which may adversely affect the market price of CAH s common stock. If, in the due exercise of its fiduciary obligations, for example, CAH s board of directors were to determine that a takeover proposal were not in the best interests of CAH, such shares could be issued by the board of directors without stockholder approval in one or more private placements or other transactions that might prevent or render more difficult or make more costly the completion of any attempted takeover transaction by diluting voting or other rights of the proposed acquirer or insurgent stockholder group, by creating a substantial voting block in institutional or other hands that might support the position of the incumbent board of directors, by effect effecting an acquisition that might complicate or preclude the takeover, or otherwise. The authorization of additional shares of common stock will also enable CAH to have the flexibility to authorize the issuance of shares of common stock in the future for financing its business, for acquiring other businesses, for forming strategic partnerships and alliances and for stock dividends and stock splits. CAH currently has no such plans, proposals, or arrangements, written or otherwise, to issue any of the additional authorized shares of common stock for such purposes.

If the initial charter proposals or the merger proposal is not approved, the secondary charter proposals will not be presented at the meeting.

The approval of each secondary charter proposal will require the affirmative vote of the holders of a majority of the outstanding shares of Triplecrown common stock on the record date.

Under the merger agreement, the approval of the secondary charter proposals are not a condition to the consummation of the merger and the vote on such proposals will not impact whether the merger is consummated.

A copy of CAH s amended and restated certificate of incorporation, as it will be in effect assuming approval of all of the secondary charter proposals and upon consummation of the merger and filing in the office of the Secretary of State of the State of Delaware, is attached to this proxy statement as Annex B.

TRIPLECROWN S BOARD OF DIRECTORS RECOMMENDS THAT STOCKHOLDERS VOTE FOR THE APPROVAL OF EACH OF THE SECONDARY CHARTER PROPOSALS.

## THE STOCKHOLDER ADJOURNMENT PROPOSAL

The stockholder adjournment proposal, if adopted, will allow Triplecrown s board of directors to adjourn the special meeting of stockholders to a later date or dates to permit further solicitation of proxies in the event, based on the tabulated votes, there are not sufficient votes at the time of the special meeting to approve the consummation of the merger. In no event will Triplecrown adjourn the special meeting or consummate the merger beyond the date by which it may properly do so under its amended and restated certificate of incorporation and Delaware law. The purpose of the stockholder adjournment proposal is to provide more time for Triplecrown, the Triplecrown Founders, Cullen Agritech and/or their respective affiliates to solicit additional votes and make purchases of Public Shares or other arrangements that would increase the likelihood of obtaining a favorable vote on the proposals being presented to stockholders and to meet the requirement that the holders of fewer than 30.0% of the Public Shares vote against the merger proposal and demand that their Public Shares be converted into cash. See the section entitled *The Merger Proposal Actions That May Be Taken to Secure Approval of Triplecrown s Stockholders*.

In addition to an adjournment of the special meeting upon approval of the stockholder adjournment proposal, the board of directors of Triplecrown is empowered under Delaware law to postpone the meeting at any time prior to the meeting being called to order. In such event, Triplecrown will issue a press release and take such other steps as it believes are necessary and practical in the circumstances to inform its stockholders of the postponement.

## Consequences if the Stockholder Adjournment Proposal is Not Approved

If the stockholder adjournment proposal is not approved by the stockholders, Triplecrown s board of directors may not be able to adjourn the special meeting to a later date in the event, based on the tabulated votes, there are not sufficient votes at the time of the special meeting to approve the consummation of the merger (because the merger proposal is not approved or because the holders of 30.0% or more of the Public Shares vote against the merger proposal and demand conversion of their Public Shares into cash) or the other required proposals are not approved. In such event, the merger would not be completed and, unless Triplecrown were able to consummate a business combination with another party no later than October 22, 2009, it would be required to liquidate.

## **Required Vote**

Adoption of the stockholder adjournment proposal requires the affirmative vote of a majority of the issued and outstanding shares of Triplecrown s common stock represented in person or by proxy at the meeting and entitled to vote thereon. Adoption of the stockholder adjournment proposal is not conditioned upon the adoption of any of the other proposals.

TRIPLECROWN S BOARD OF DIRECTORS RECOMMENDS THAT TRIPLECROWN S STOCKHOLDERS VOTE FOR THE APPROVAL OF THE STOCKHOLDER ADJOURNMENT PROPOSAL.

## PROPOSALS TO BE CONSIDERED BY THE TRIPLECROWN WARRANTHOLDERS

## THE WARRANT AMENDMENT PROPOSALS

In connection with the proposed merger, Triplecrown is proposing amendments to the warrant agreement governing its outstanding warrants to (i) increase the exercise price of the warrants from \$7.50 per share to \$12.00 per share, (ii) extend the expiration date of the warrants from October 21, 2012 to October 21, 2013 and (iii) increase the price at which the stock must trade for the warrants to be called for redemption from \$13.75 per share to \$17.00 per share. The warrant agreement will also be amended to make certain other immaterial changes to ensure that the warrants of CAH that will be received by the holders of warrants of Triplecrown after the merger will be governed by the warrant agreement and that CAH will assume all of the rights and obligations of Triplecrown under the warrant agreement after the merger. Pursuant to the Warrant Agreement, dated as of October 22, 2007, by and between Triplecrown and Continental Stock Transfer & Trust Company, as warrant agent, the parties may amend any provision of the warrant agreement with the consent of the holders of a majority of the then outstanding warrants. Approval of each of the warrant amendment proposals is a condition to consummation of the merger.

Triplecrown believes the amendments to the warrants is appropriate given the change in structure of Triplecrown following completion of the merger. Additionally, if the merger is not consummated and Triplecrown does not complete a different business combination by October 22, 2009, the warrants will expire worthless. If the warrant amendment proposals are approved, all other material terms of Triplecrown s warrants will remain the same.

## **Required Vote**

Approval of each of the warrant amendment proposals requires the affirmative vote of the holders of a majority of Triplecrown s then outstanding warrants. The Triplecrown Founders, as well as certain other warrantholders, representing % of Triplecrown s outstanding warrants, have executed lockup agreements whereby such parties have agreed to vote in favor of the warrant amendment proposals at the special meeting.

## Recommendation

THE BOARD OF DIRECTORS RECOMMENDS THAT THE WARRANTHOLDERS VOTE FOR THE APPROVAL OF EACH OF THE WARRANT AMENDMENT PROPOSALS.

## THE WARRANTHOLDER ADJOURNMENT PROPOSAL

The warrantholder adjournment proposal, if adopted, will allow Triplecrown s board of directors to adjourn the special meeting of warrantholders to a later date or dates to permit further solicitation of proxies in the event, based on the tabulated votes, there are not sufficient votes at the time of the special meeting to approve the warrant amendment proposals. In no event will Triplecrown adjourn the special meeting or consummate the warrant amendment proposals beyond the date by which it may properly do so under its amended and restated certificate of incorporation and Delaware law. The purpose of the warrantholder adjournment proposal is to provide more time for Triplecrown and the Triplecrown Founders to make purchases of warrants or other arrangements that would increase the likelihood of obtaining a favorable vote on the warrant amendment proposals.

In addition to an adjournment of the special meeting upon approval of the warrantholder adjournment proposal, the board of directors of Triplecrown is empowered under Delaware law to postpone the meeting at any time prior to the meeting being called to order. In such event, Triplecrown will issue a press release and take such other steps as it believes are necessary and practical in the circumstances to inform its stockholders of the postponement.

## Consequences if the Warrantholder Adjournment Proposal is Not Approved

If the warrantholder adjournment proposal is not approved by the warrantholders, Triplecrown s board of directors may not be able to adjourn the special meeting to a later date in the event, based on the tabulated votes, there are not sufficient votes at the time of the special meeting to approve the warrant amendment proposals. In such event, the warrant amendment would not be completed.

## **Required Vote**

Adoption of the warrantholder adjournment proposal requires the affirmative vote of a majority of Triplecrown s then outstanding warrants represented in person or by proxy at the meeting and entitled to vote thereon. Adoption of the warrantholder adjournment proposal is not conditioned upon the adoption of any of the other proposals.

TRIPLECROWN S BOARD OF DIRECTORS RECOMMENDS THAT TRIPLECROWN S WARRANTHOLDERS VOTE FOR THE APPROVAL OF THE WARRANTHOLDER ADJOURNMENT PROPOSAL.

## OTHER INFORMATION RELATED TO TRIPLECROWN

## **Business of Triplecrown**

Triplecrown was formed on June 8, 2007 to effect a merger, capital stock exchange, asset acquisition or other similar business combination with an operating business in the financial services industry. Prior to executing the merger agreement with CAH, Triplecrown s efforts were limited to organizational activities, completion of its IPO and the evaluation of possible business combinations.

## Offering Proceeds Held in Trust

On October 25, 2007, Triplecrown consummated its IPO of 55,200,000 units, including 7,200,000 units which were subject to the underwriters—over-allotment option. The net proceeds of the offering, including proceeds from the over-allotment option and from the private sale of 5,000,000 Sponsors—Warrants at a price of \$1.00 per warrant and after deducting the underwriting discounts and commissions and the offering expenses, were \$536,873,680. Of that amount, \$536,930,000 was deposited into the trust account and invested in government securities. The remaining proceeds have been used by Triplecrown in its pursuit of a business combination.

In addition, Triplecrown is entitled to draw for use of working capital up to \$6,000,000 of interest earned on the trust account, as well as any amounts necessary to pay its tax obligations. Through , 2009, Triplecrown has drawn from the trust account \$ for working capital and \$ for tax obligations and may draw up to an additional \$ for working capital needs in addition to any future tax payments. Except as set forth above, no funds in the trust account have been released and only the remaining interest income that Triplecrown may use for working capital requirements and amounts necessary for its tax obligations will be released until the earlier of the consummation of a business combination or the liquidation of Triplecrown. The trust account contained \$538,796,312 as of September 4, 2009.

The funds in Triplecrown s trust account will be released to pay transaction fees and expenses (including the balance of the purchase price for the land to be used by CAH following consummation of the merger), deferred underwriting discounts and commissions, tax liabilities, if any, and reimbursement of expenses of the Triplecrown Founders and to make purchases of Public Shares, if any. The balance of the funds will be released to CAH to pay Triplecrown stockholders who properly exercise their conversion rights and for working capital and general corporate purposes of CAH and Cullen Agritech. Additionally, if CAH has access to more than \$150 million after closing of the merger, CAH may use such funds, depending on market conditions, to repurchase shares of common stock.

The holders of Public Shares will be entitled to receive funds from the trust account only in the event of Triplecrown s liquidation or if they seek to convert their shares into cash and the merger is actually completed. In no other circumstances will a stockholder have any right or interest of any kind to or in the trust account.

## **Stockholder Approval of Business Combination**

Triplecrown will proceed with the merger only if a majority of the Public Shares present and entitled to vote on the merger proposal at the special meeting is voted in favor of the merger proposal. The Triplecrown Founders have agreed to vote their common stock issued prior to the IPO on the merger proposal in accordance with the vote of holders of a majority of the Public Shares present (in person or represented by proxy) and entitled to vote at the special meeting. If the holders of 30% or more of the Public Shares vote against the merger proposal and properly

demand that Triplecrown convert their Public Shares into their pro rata share of the trust account, Triplecrown will not consummate the merger. In this case, Triplecrown will be forced to liquidate.

## **Liquidation if No Business Combination**

Triplecrown s amended and restated certificate of incorporation provides for the automatic termination of Triplecrown s corporate existence and mandatory liquidation of Triplecrown if Triplecrown does not consummate a business combination by October 22, 2009. The amended and restated certificate of incorporation provides that if Triplecrown has not completed a business combination by such date, its corporate existence

will cease except for the purposes of winding up its affairs liquidating, pursuant to Section 278 of the DGCL. This has the same effect as if Triplecrown s board of directors and stockholders had formally voted to approve Triplecrown s dissolution pursuant to Section 275 of the DGCL. Accordingly, limiting Triplecrown s corporate existence to a specified date as permitted by Section 102(b)(5) of the DGCL removes the necessity to comply with the formal procedures set forth in Section 275 (which would have required Triplecrown s board of directors and stockholders to formally vote to approve its dissolution and liquidation and to have filed a certificate of dissolution with the Delaware Secretary of State).

In connection with its liquidation, Triplecrown will distribute to the holders of its Public Shares, in proportion to their respective amounts of Public Shares, an aggregate sum equal to the amount in the trust account, inclusive of any interest thereon, plus remaining net assets (subject to Triplecrown s obligations under Delaware law to provide for claims of creditors as described below). The Triplecrown Founders have waived their rights to participate in any liquidation distribution with respect to their Founders Common Stock. As a consequence of such waivers, a liquidating distribution will be made only with respect to the Public Shares. There will be no distribution from the trust account with respect to Triplecrown s warrants, which will expire worthless.

The proceeds deposited in the trust account could become subject to the claims of Triplecrown s creditors (which could be prior to the claims of the holders of the Public Shares and could include vendors and service providers Triplecrown has engaged to assist it in connection with its search for a target business and that are owed money by it, as well as target businesses themselves). If Triplecrown liquidates prior to the consummation of a business combination, Messrs. Watson and Ledecky have agreed that they will be personally liable to pay debts and obligations to target businesses or vendors or other entities that are owed money by Triplecrown for services rendered or contracted for or products sold to Triplecrown in excess of the net proceeds of Triplecrown s IPO not held in the trust account, but only if such a target business or vendor or other entity did not execute such a waiver. Accordingly, if a claim brought by a target business or vendor or other entity did not exceed the amount of funds available to Triplecrown outside of the trust account or available to be released to Triplecrown from interest earned on the trust account balance or if such an entity executed a waiver, Messrs. Watson and Ledecky would not have any personal obligation to indemnify such claims as they would be paid from such available funds. However, if a claim exceeded such amounts and such entity did not execute a waiver, there is no exception to the obligations of Messrs. Watson and Ledecky to pay such claim. There is no assurance, however, that they would be able to satisfy those obligations.

Under the DGCL, stockholders may be held liable for claims by third parties against a corporation to the extent of distributions received by them in a dissolution. If the corporation complies with certain procedures set forth in Section 280 of the DGCL intended to ensure that it makes reasonable provision for all claims against it, including a 60-day notice period during which any third-party claims can be brought against the corporation, a 90-day period during which the corporation may reject any claims brought, and an additional 150-day waiting period before any liquidating distributions are made to stockholders, any liability of stockholders with respect to a liquidating distribution is limited to the lesser of such stockholder s pro rata share of the claim or the amount distributed to the stockholder, and any liability of the stockholder would be barred after the third anniversary of the dissolution. However, as stated above, it is Triplecrown s intention to make liquidating distributions to its stockholders as soon as reasonably possible after October 22, 2009 and, therefore, Triplecrown does not intend to comply with those procedures. As such, Triplecrown s stockholders could potentially be liable for any claims to the extent of distributions received by them and any liability of Triplecrown s stockholders may extend well beyond the third anniversary of such date. Because Triplecrown will not be complying with Section 280, Section 281(b) of the DGCL requires Triplecrown s to adopt a plan that will provide for payment, based on facts known to it at such time, of (i) all existing claims, (ii) all pending claims and (iii) all claims that may be potentially brought against it within the subsequent 10 years. Accordingly, Triplecrown would be required to provide for any claims of creditors known to it at that time or those that it believes could be potentially brought against it within the subsequent 10 years prior to it distributing the funds in the trust account to its public

stockholders. Triplecrown cannot make any assurance as to when such plan will be completed and when liquidation distributions will be made. As a result, liquidation distributions could take 60 days or more to be completed.

Because Triplecrown is a blank check company, rather than an operating company, and its operations have been limited to searching for prospective target businesses to acquire, the only likely claims to arise would be from potential target businesses, many of whom have given Triplecrown agreements waiving any right, title, interest or claim of any kind they may have in or to any monies held in the trust account, or Triplecrown s vendors (such as accountants, lawyers, investment bankers, etc.). As a result, the claims that could be made against Triplecrown are significantly limited and the likelihood that any claim that would result in any liability extending to the trust is remote. Nevertheless, such agreements may not be enforceable. Accordingly, Triplecrown cannot assure you that third parties will not seek to recover from Triplecrown s stockholders amounts owed to them by Triplecrown.

If there are no funds remaining to pay the costs associated with the implementation and completion of the liquidation and distribution, Eric J. Watson and Jonathan J. Ledecky have agreed to advance Triplecrown the funds necessary to pay such costs and complete such liquidation (currently anticipated to be no more than approximately \$15,000) and not to seek repayment for such expenses.

If Triplecrown is forced to file a bankruptcy case or an involuntary bankruptcy case is filed against it that is not dismissed, the proceeds held in the trust account could be subject to applicable bankruptcy law, and may be included in Triplecrown s bankruptcy estate and subject to the claims of third parties with priority over the claims of Triplecrown s stockholders. Also, in any such case, any distributions received by stockholders could be viewed under applicable debtor/creditor and/or bankruptcy laws as either a preferential transfer or a fraudulent conveyance. As a result, a bankruptcy court could seek to recover all amounts received by Triplecrown s stockholders. Furthermore, because, in the event of a liquidation, Triplecrown intends to distribute the proceeds held in the trust account to its public stockholders promptly after October 22, 2009, this may be viewed or interpreted as giving preference to Triplecrown s public stockholders over any potential creditors with respect to access to or distributions from Triplecrown s assets. In addition, Triplecrown s board may be viewed as having breached their fiduciary duties to Triplecrown s creditors and/or may have acted in bad faith, and thereby exposing itself and Triplecrown to claims of punitive damages, by paying public stockholders from the trust account prior to addressing the claims of creditors and/or complying with certain provisions of the DGCL with respect to Triplecrown s liquidation. Triplecrown cannot assure you that claims will not be brought against it for these reasons. To the extent any bankruptcy or other claims deplete the trust account, Triplecrown cannot assure you it will be able to return to its public stockholders at least \$9.76 per share.

## **Facilities**

Triplecrown maintains executive offices at 970 West Broadway, PMB 402, Jackson, Wyoming 83001. Jonathan J. Ledecky is providing this space to Triplecrown at no charge.

## **Employees**

Triplecrown has two executive officers who receive no compensation from Triplecrown for such services. These individuals are not obligated to contribute any specific number of hours per week and devote only as much time as they deem necessary to Triplecrown s affairs. Triplecrown does not intend to have any full time employees prior to the consummation of the merger.

## **Directors and Executive Officers**

Triplecrown s current directors and executive officers are as follows:

Facilities 161

Name	Age	Position
Eric J. Watson	50	Chairman of the Board and Treasurer
Jonathan J. Ledecky	51	President, Secretary and Director
Robert B. Hersov	48	Director
Edward J. Mathias	67	Director
Kerry Kennedy	49	Director
Richard Y. Roberts	57	Director
Jimmie Lee Solomon, Jr.	53	Director
Jay H. Nussbaum	65	Director
Richard A. Stein	49	Director
Edward Hanson	33	Director
Jim Gray	49	Director

Eric J. Watson has been Triplecrown s chairman of the board and treasurer since its inception. He has also been the chief executive officer of Cullen Agritech since its inception in June 2009 and chief executive officer, secretary, treasurer and sole director of CAH since its inception in August 2009. From July 2005 until December 2007, Mr. Watson served as the chairman of the board and treasurer of Endeavor Acquisition Corp., an NYSE Amex listed blank check company formed to acquire an operating business. Endeavor Acquisition Corp. consummated its business combination with American Apparel, Inc. on December 12, 2007. From January 2007 to April 2009, Mr. Watson was the chairman of the board and treasurer of Victory Acquisition Corp., a blank check company formed for the purpose of effecting a merger, capital stock exchange, asset acquisition or other similar business combination with an operating business in any industry other than the franchising, financial services or healthcare industries. Victory Acquisition Corp. failed to consummate a business combination and liquidated as a result thereof. Since January 1995, Mr. Watson has been the executive chairman of, and interests associated with him own, Cullen Investments Limited, an international private investment company which has its origins in a start up founded by Mr. Watson through which he has actively invested his own capital in a range of successful mergers and acquisitions. Mr. Watson and his associated interests have a substantial portfolio comprising interests in the fashion retail, financial services, real estate, sports and entertainment sectors. Cullen Investments interests include ownership of Bendon, an international manufacturer and retailer of women s lingerie whose prestige brands include the licensed Elle Macpherson Intimates and Stella McCartney labels. Another major investment held by interests associated with Mr. Watson is a 50% ownership of the Hanover Group, one of the largest privately owned financial services business in New Zealand with operations extending to the United States and Australia. In 1998, Logan Corporation, an entity owned by Cullen Investments, invested in publicly listed Wall Group (formerly Pacific Retail Group or PRG). PRG was listed on the New Zealand Stock Exchange until 2006, and operated several consumer focused companies. These included Noel Leeming Group, a New Zealand specialty appliance retail chain, Pacific Retail Finance Limited, a New Zealand consumer finance business, and Bendon. PRG acquired PRG PowerHouse Limited ( PowerHouse ), a specialty appliance retail chain in the United Kingdom in September 2003, at which time Logan Corporation was a majority stockholder in publicly listed PRG. Mr. Watson was then and remains a director of PRG, but has not at any time been an executive officer of PRG. In August 2006, PowerHouse, as a result of adverse market conditions and increasing losses, was placed in administration under United Kingdom law, a process similar to a United States bankruptcy proceeding. The administrator determined that the best course of action with respect to PowerHouse was to close its stores and realize the assets for the benefit of its creditors. Subsequently, at the end of April 2007, PowerHouse was placed into liquidation by the administrator. In addition, PRG had provided guarantees in respect of certain debts owed by PowerHouse to its creditors. Due to an unforeseen material deterioration in the outcome of the Powerhouse administration, PRG is no longer in a position to satisfy, in full, the claims of its creditors.

Accordingly, PRG s creditors approved a compromise with PRG in September 2007 in respect of their debts. PRG was privatized by Logan Corporation in 2006. Prior to founding Cullen Investments, Mr. Watson was the founding chairman and largest stockholder of Blue Star Group, a retail and distribution group he founded in January 1992. In 1996, Blue Star Group was sold to U.S. Office Products, a diversified supplier of a broad range of office products and business services to corporate customers. Until August 1999, Mr. Watson continued as executive chairman of Blue Star Group, a wholly-owned subsidiary of U.S. Office Products after the acquisition. Following the acquisition of Blue Star Group by U.S. Office Products, Mr. Watson served as a director of McCollam Printers from July 1997 to June 1998. Prior to serving with U.S. Office Products, Mr. Watson held several positions with Xerox Corporation, an office products company, including president of operations for Australasia. Mr. Watson received a diploma of general management from Auckland University.

*Jonathan J. Ledecky* has been Triplecrown s president, secretary and a member of its board of directors since its inception. From July 2005 to December 2007, Mr. Ledecky served as president, secretary and a director of Endeavor Acquisition Corp. From January 2007 to April 2009, Mr. Ledecky has served as president, secretary and a director of Victory Acquisition Corp. Since June 1999, Mr. Ledecky has served as chairman of the Ledecky Foundation, a

philanthropic organization which contributes funds to programs for the education of disadvantaged inner city youth in Washington, D.C., New York and Boston. Since March 1999, Mr. Ledecky has also served as chairman of Ironbound Partners Fund LLC, a private investment management fund that oversees the Ledecky Foundation and other Ledecky family investments. In October 1994, Mr. Ledecky founded U.S. Office Products and served as its chief executive officer until November 1997 and chairman until June 1998. During his tenure, U.S. Office Products completed over 260 acquisitions, and grew

to a Fortune 500 company with over \$2.6 billion in revenues. In June 1998, U.S. Office Products completed a comprehensive restructuring plan whereby four separate entities were spun off to stockholders and U.S. Office Products underwent a leveraged recapitalization. In connection with these transactions, Mr. Ledecky resigned from his position as chairman of U.S. Office Products and became a director of each of the four spin-off entities. In February 1997, Mr. Ledecky founded Building One Services Corporation (originally Consolidation Capital Corporation), an entity formed to identify attractive consolidation opportunities which ultimately focused on the facilities management industry. In November 1997, Building One raised \$552 million in an initial public offering. Mr. Ledecky served as Building One s chief executive officer from November 1997 through February 1999 and as its chairman from inception through its February 2000 merger with Group Maintenance America Corporation. During his tenure with Building One, it completed 46 acquisitions and grew to over \$1.5 billion in revenues. From July 1999 to July 2001, Mr. Ledecky was vice chairman of Lincoln Holdings, owners of the Washington sports franchises in the NBA, NHL and WNBA. Since June 1998, Mr. Ledecky has served as a director of School Specialty, a Nasdaq Global Market listed education company that provides products, programs and services that enhance student achievement and development. School Specialty spun out of U.S. Office Products in June 1998. Since 1994, Mr. Ledecky has been involved with numerous other companies in director positions. Mr. Ledecky was a trustee of George Washington University, served as a director of the U.S. Chamber of Commerce and served as commissioner on the National Commission on Entrepreneurship. In addition, in 2004, Mr. Ledecky was elected the Chief Marshal of the 2004 Harvard University Commencement, a singular honor bestowed by his alumni peers for a 25th reunion graduate deemed to have made exceptional contributions to Harvard and the greater society while achieving outstanding professional success. Mr. Ledecky received a B.A. (cum laude) from Harvard University and a M.B.A from Harvard Business School.

Robert B. Hersov has been a member of Triplecrown s board of directors since its inception. Since January 2004, Mr. Hersov has been the vice chairman of NetJets Europe Ltd., a subsidiary of NetJets, Inc., a private aviation and fractional jet ownership company which was acquired by Berkshire Hathaway Inc. in 1998. Mr. Hersov founded and, from December 2002 to April 2004, served as the chief executive officer of Marquis Jet Europe, a private aviation company which was acquired by NetJets, Inc. in 2004. Since September 2007, Mr. Hersov has served as a non-executive director of Australian privately-owned company Global Aviation Leasing Group. Mr. Hersov is also chairman of Sapinda Limited, a UK private company, which is the main shareholder of Vatas GmbH, a private German investment company. Mr. Hersov also founded and, from October 1998 to December 2002, served as the chairman of Sportal Ltd., a company that operates an Internet site that offers sports-related games and videos. From October 1996 to September 1998, he served as the executive director of Enic plc, a holding company listed on the London Stock Exchange that invests primarily in the sports and media sectors. From September 1995 to September 1997, Mr. Hersov was the chief executive officer of Telepiu PayTV in Milan, Italy, a pay TV and digital satellite company. From March 1993 to August 1995, Mr. Hersov served as an executive director of Richemont, a tobacco, luxury and media conglomerate listed on the SWX Swiss Exchange. Since June 2005, Mr. Hersov has been a member of the board of directors of Shine Media Acquisition Corp., a blank check company that was formed to acquire a direct or indirect interest in an operating business in the media and advertising industry in the People s Republic of China. He was a director of Endeavor Acquisition Corp. from July 2005 to December 2007 and a director of Victory Acquisition Corp. from January 2007 to April 2009. Mr. Hersov received a B.B.S. from the University of Cape Town and a M.B.A. from the Harvard Business School.

Edward J. Mathias has been a member of Triplecrown s board of directors since its inception. Mr. Mathias was involved with the founding of The Carlyle Group, a global private equity firm headquartered in Washington, DC. He has been a managing director since January 1994 and presently serves as an Investment Committee member for a number of Carlyle s partnerships. Previously, Mr. Mathias served on the management committee and board of directors of T. Rowe Price Associates, Inc., an investment management organization where he was employed from 1971 to December 1993. He was a director of Endeavor Acquisition Corp. from July 2005 to December 2007 and a

director of Victory Acquisition Corp. from January 2007 to April 2009. He has also been a director of NexCen Brands, formerly Aether Systems, since June 2002 and Allied Capital Corp. since January 2009. Mr. Mathias also serves on The Howard Hughes Institute s Investment Advisory Committee. Mr. Mathias received an M.B.A. from The Harvard Business School where he is

on The Board of Dean s Advisors and a B.A. from The University of Pennsylvania where he is currently a trustee and member of The Penn Investment Board which oversees the University s endowment.

Kerry Kennedy has been a member of Triplecrown s board of directors since its inception. In April 1988, she established the Robert F. Kennedy Memorial Center for Human Rights and acted as its executive director until January 1995 working on diverse human rights issues. Ms. Kennedy has been the Chair of the Amnesty International Leadership Council since January 1996. She was a director of Endeavor Acquisition Corp. from July 2005 to December 2007 and a director of Victory Acquisition Corp. from January 2007 to April 2009. She also serves on the board of directors of the International Center for Ethics and Justice and Public Life at Brandeis University. Ms. Kennedy received a B.A. from Brown University and an LLM from Boston College Law School.

**Richard Y. Roberts** has been a member of Triplecrown s board of directors since its inception. In February 2006, Mr. Roberts co-founded a regulatory/legislative consulting firm, Roberts, Raheb & Gradler LLC. He was a partner with Thelen Reid & Priest LLP, a national law firm, from January 1997 to February 2006. From August 1995 to January 1997, Mr. Roberts was a consultant at Princeton Venture Research, Inc., a private consulting firm. From 1990 to 1995, Mr. Roberts was a commissioner of the Securities and Exchange Commission, and, in this capacity, was actively involved in, has written about or has testified on, a wide range of subjects affecting the capital markets. Since leaving the Commission, Mr. Roberts has been a frequent media commentator and writer on various securities public policy issues and has assisted the Governments of Romania and Ukraine in the development of a securities market. Since September 2005, Mr. Roberts has served as a member of the board of directors of Nyfix, Inc., a Nasdaq Global Market listed provider of industry interconnectivity networks, electronic trade communication technologies, trading workstations and middle-office trade automation technologies. He was a director of Endeavor Acquisition Corp. from July 2005 to December 2007 and a director of Victory Acquisition Corp. from January 2007 to April 2009. From 1987 to 1990, he was the chief of staff for Senator Richard Shelby. He is a member of the Alabama Bar and the District of Columbia Bar. Mr. Roberts is a member of the Advisory Board of Securities Regulation & Law Reports, of the Advisory Board of the International Journal of Disclosure and Governance, and of the Editorial Board of the Municipal Finance Journal. Mr. Roberts also previously served as a member of the District 10 Regional Consultative Committee of the Financial Industry Regulatory Authority, the Market Regulation Advisory Board of the FINRA, and the Legal Advisory Board of the FINRA, Mr. Roberts received a B.E.E. from Auburn University, a J.D. from the University of Alabama School of Law, and a Master of Laws from the George Washington University Law Center.

Jimmie Lee Solomon, Jr. has been a member of Triplecrown s board of directors since its inception. Mr. Solomon has been affiliated with Major League Baseball since July 1991 as executive director of minor league operations, senior vice president of baseball operations and most recently since June 2005 as executive vice president of baseball operations. From June 1981 to June 1991, he was a lawyer at Baker & Hostetler, LLP where he counseled a variety of clients including Major League Baseball, the National Football League Management Council and the United States Football League front office personnel. Mr. Solomon was a director of Victory Acquisition Corp. from January 2007 to April 2009. Mr. Solomon is also on the board of directors for the NAACP Legal Defense and Education Fund, Inc. He has previously served on the board of directors for the Greater Washington Boys and Girls Club and has served as an advisory board member for Elementary Baseball, Inc. He has also served as a member of the Diversity Task Force for Women s Sports Foundation. Mr. Solomon received a B.A. from Dartmouth College and a J.D. from Harvard Law School.

Jay H. Nussbaum has been a member of Triplecrown s board of directors since its inception. He is the founder and Chief Operating Officer of Agilex Technologies, a company formed in 2006 to offer leading technology solutions to government and commercial clients in the United States. From May 2004 to July 2006, Mr. Nussbaum served as the global head of sales, marketing and business development for Citigroup(r) Global Transaction Services, a division of Citigroup Global Markets Inc. which handles cash management, trade, securities services and fund services. He was a

director of Endeavor Acquisition Corp. from July 2005 to December 2007 and a director of Victory Acquisition Corp. from January 2007 to April 2009. From January 2002 to April 2004, Mr. Nussbaum was affiliated with BearingPoint, Inc. (formerly KPMG Consulting), a consulting company, where he served most recently as head of worldwide sales. From 1991 to January 2002, Mr. Nussbaum was affiliated with Oracle Corporation, a Nasdaq Global Market listed

enterprise software company, where he most recently served as executive vice president. Prior to joining Oracle Corporation, Mr. Nussbaum was affiliated with Xerox Corporation for 24 years where he most recently served as president of integrated systems operations. Mr. Nussbaum received a B.A. from the University of Maryland.

Jim Gray has been a member of Triplecrown s board of directors since its inception. Mr. Gray has been a network television and national cable sportscaster since 1983. During this time, he has worked with CBS Sports, NBC Sports, and currently has long-term relationships with ESPN, ABC-TV, Showtime and Westwood One Radio. His career highlights include coverage of 7 Olympic Games, 18 NFL Super Bowls, 9 MLB World Series, 18 NBA Finals, 9 NCAA Final Fours, 18 Masters Golf Tournaments, and more than 300 World Championship Boxing matches. Gray has also broadcast on numerous occasions the Ryder Cup, the Presidents Cup, NBA and MLB All-Star Games, National Football League AFC and NFC Championship Games, and Major League Baseball American and National League Championship Series Games. Mr. Gray has won 11 national Emmy Awards for journalism and reporting. He was named Sportscaster of the Year in 1998 and 1999 as Sports Reporter of the Year by his member peers of the American Sportscasters Association. In 1997, he won the prestigious Broadcast of the Year Award presented by the National Sportswriters and Sportscasters Association for his interview with Mike Tyson following Tyson vs. Holyfield II. Mr. Gray has been named Sports Reporter of the Year twelve times by the USA Today, and that same publication in April 2005 named Mr. Gray the country s best sports reporter of the past quarter century. Mr. Gray received a B.S. from the University of Colorado.

Richard A. Stein has been a member of Triplecrown s board of directors since its inception. Mr. Stein has served as chief executive officer and a director of Powerride Motorsports, Inc., a powersports retailer of which Mr. Ledecky is the controlling shareholder, since November 2001. From January 2001 to November 2001, Mr. Stein was managing his personal investments. From 1987 to January 2001, Mr. Stein was the co-founder and served as the executive vice president and chief financial officer of Hanger Orthopedic Group, Inc., a provider of orthotic and prosthetic services and products. From 1982 to 1987, Mr. Stein served as a supervisor in the emerging business services division of PriceWaterhouse Coopers, an accounting firm. Mr. Stein received a B.S. from the University of Florida. Mr. Stein is a Certified Public Accountant.

Edward Hanson has been a member of Triplecrown s board of directors since its inception. Mr. Hanson is a Director of Babcock & Brown (UK) Limited. Babcock & Brown is a principal investment firm headquartered in Sydney and Mr. Hanson has worked in the London office since 1997. He also runs the private equity fund, Babcock & Brown Global Partners, which he raised in July 2005. Babcock & Brown invest in asset backed businesses around the world. From 1996 to 1997, Mr. Hanson worked at Cavill White Securities, an investment bank in New Zealand. Mr. Hanson is a member of the board of directors of BGP Investment S.à r.l., a European real estate joint venture that pursues a range of property related activities including the acquisition and management of new assets and selected development projects. Mr. Hanson is also on the board of Corvus Capital, an AIM listed investment company. Mr. Hanson received a Bachelor of Commerce from the University of Auckland in New Zealand.

## Periodic Reporting and Audited Financial Statements

Triplecrown has registered its securities under the Exchange Act and has reporting obligations, including the requirement to file annual and quarterly reports with the SEC. In accordance with the requirements of the Exchange Act, Triplecrown s annual reports contain financial statements audited and reported on by Triplecrown s independent accountants. Triplecrown has filed with the SEC its Annual Reports on Form 10-K covering the fiscal years ended December 31, 2008 and 2007 and its Quarterly Reports on Form 10-Q covering the quarters ended March 31, 2008, June 30, 2008, September 30, 2008, March 31, 2009 and June 30, 2009.

## **Legal Proceedings**

There are no legal proceedings pending against Triplecrown.

## Triplecrown s Management s Discussion and Analysis of Financial Condition and Results of Operations

The following discussion should be read in conjunction with Triplecrown s financial statements and related notes thereto included elsewhere in this proxy statement/prospectus.

86

Legal Proceedings 170

#### TABLE OF CONTENTS

Triplecrown is a blank check company formed on June 8, 2007 to effect a merger, capital stock exchange, asset acquisition or other similar business combination with an operating business in the financial services industry.

For the three months ended June 30, 2009 and 2008, Triplecrown had a net (loss) income of \$(356,667) and \$1,089,208 after a (benefit) provision for income taxes of (\$165,636) and \$567,744; for the six months ended June 30, 2009 and 2008, Triplecrown had a net (loss) income of \$(734,324) and \$3,255,003 after a (benefit) provision for income taxes of (\$360,139) and \$1,569,864, and for the period from June 8, 2007 (inception) through June 30, 2009, Triplecrown had net income of \$5,748,881 after a provision for income taxes of \$3,508,680. Triplecrown incurred formation and operating costs for the three months ended June 30, 2009 and 2008 of \$643,309 and \$319,209, for the six months ended June 30, 2009 and 2008 of \$1,267,364 and \$681,939, and for the period from June 8, 2007 (inception) through June 30, 2009, of \$2,931,852. These costs consisted primarily of professional and consulting fees, and Delaware franchise taxes.

For the year ended December 31, 2008, Triplecrown had a net income of \$4,658,808 consisting of \$8,443,326 of dividend and interest income offset by \$1,484,138 of formation and operating costs and \$2,300,380 of provisions for income tax.

For the period from June 8, 2007 (inception) through December 31, 2007, Triplecrown had a net income of \$1,824,397 consisting of \$3,573,186 of dividend and interest income offset by \$180,350 of formation and operating costs and \$1,568,439 of provisions for income tax.

Triplecrown s activity from June 8, 2007 (inception) through October 25, 2007 was to prepare for its IPO. Since November 15, 2007, Triplecrown s efforts have been devoted to identifying an acquisition candidate. Triplecrown believes that it has sufficient funds available to complete its efforts to effect a business combination by October 22, 2009.

Triplecrown entered into an underwriting agreement with the underwriters in its IPO. The underwriting agreement required Triplecrown to pay 3.5% of the gross proceeds of the IPO as an underwriting discount upon consummation of the IPO plus an additional 3.5% of the gross proceeds only upon consummation of a business combination. Triplecrown paid an underwriting discount of 3.5% of the gross proceeds (\$19,320,000) in connection with the consummation of the IPO. The deferred 3.5% of the gross proceeds (\$19,320,000) was included as part of the funds deposited in its trust account. The underwriters waived their right to receive payment of the 3.5% of the gross proceeds upon Triplecrown s liquidation if it is unable to complete a business combination. Triplecrown is currently in negotiations with its underwriters to reduce the fee payable to them upon consummation of the merger.

## Going Concern and Management s Plan and Intentions

Triplecrown s only source of income to enable it to continue to fund its search for an acquisition candidate is the interest and dividends Triplecrown earns on its cash held in the trust account. These funds may not be sufficient to maintain Triplecrown until a business combination is consummated. Pursuant to Triplecrown s amended and restated certificate of incorporation, if Triplecrown is unable to consummate a timely business combination prior to October 22, 2009, Triplecrown would have to liquidate and return the funds held in the trust account to the holders of Public Shares. There can be no assurance that Triplecrown will consummate its business combination prior to October 22, 2009. These factors raise substantial doubt about Triplecrown s ability to continue as a going concern. Triplecrown s financial statements do not include any adjustment that might result from the outcome of these uncertainties.

## **Off-Balance Sheet Arrangements**

Triplecrown does not have any obligations, assets or liabilities that would be considered off-balance sheet arrangements as defined in Item 303(a)(4)(ii) of Regulation S-K. Triplecrown does not participate in transactions that create relationships with unconsolidated entities or financial partnerships, often referred to as variable interest entities, which would have been established for the purpose of facilitating off-balance sheet arrangements.

Triplecrown has not entered into any off-balance sheet financing arrangements, established any special purpose entities, guaranteed any debt or commitments of other entities, or acquired any non-financial assets.

## **Contractual Obligations**

Triplecrown does not have any long-term debt, capital lease obligations, operating lease obligations or long-term liabilities.

## **Critical Accounting Policies**

Triplecrown s significant accounting policies are more fully described in its condensed financial statements. However, certain accounting policies are particularly important to the portrayal of financial position and results of operations and require the application of significant judgments by management. In applying those policies, management used its judgment to determine the appropriate assumptions to be used in determination of certain estimates. Triplecrown s accounting policy is to use estimates based on its historical experience, terms of existing contracts, observance of trends in the industry and information available from outside sources, as appropriate.

#### Quantitative and Qualitative Disclosures about Market Risk

As of June 30, 2009, Triplecrown s efforts were limited to organizational activities, activities relating to its IPO and the search for an acquisition candidate. Triplecrown had neither engaged in any operations nor generated any revenues.

Market risk is a broad term for the risk of economic loss due to adverse changes in the fair value of a financial instrument. These changes may be the result of various factors, including interest rates, foreign exchange rates, commodity prices and/or equity prices. Due to the level of risk associated with certain investment securities, it is at least reasonably possible that changes in the values of investment securities will occur in the near term and that such changes will have a severe affect on the results of operations. As of June 30, 2009, assets in the Trust account were invested in United States Government Treasury-Bills and earned interest at an effective annual rate of 0.0002% for the quarter ended June 30, 2009.

## **Independent Auditors** Fees

The firm of Marcum LLP ( Marcum ) acts as Triplecrown s independent registered public accounting firm. The following is a summary of fees paid or to be paid to Marcum for services rendered to Triplecrown.

## **Audit Fees**

During the year ended December 31, 2008, fees for Triplecrown s registered public accounting firm were \$85,000 for the services they performed in connection with its Annual Report for the fiscal year ended December 31, 2007 and for the three Quarterly Reports for the fiscal quarters ended March 31, 2008, June 30, 2008 and September 30, 2008.

During the period from June 8, 2007 (inception) through December 31, 2007, fees for Triplecrown s independent registered public accounting firm were \$70,000 for the services they performed in connection with its IPO, including the financial statements included in the Form 8-K filed with the Securities and Exchange Commission on October 29, 2007.

## **Audit-Related Fees**

During 2008 and 2007 there were no fees billed for audit-related services provided by Triplecrown s independent registered public accounting firm other than those set forth above.

## **Tax Fees**

During 2008, Triplecrown was billed \$3,900 for tax services, principally the preparation of income tax returns. For 2007, Triplecrown s independent registered public accounting firm did not render any services to it for tax compliance, tax advice and tax planning.

## **All Other Fees**

During 2008 and 2007, there were no fees billed for products and services provided by Triplecrown s independent registered public accounting firm other than those set forth above.

88

Audit-Related Fees 174

## **Audit Committee Pre-Approval Policies and Procedures**

Since Triplecrown s audit committee was not formed until October 2007, the audit committee did not pre-approve all of the foregoing services although any services rendered prior to the formation of Triplecrown s audit committee were approved by its board of directors. However, all services render since October 2007 were pre-approved by Triplecrown s audit committee. In addition, in accordance with Section 10A(i) of the Securities Exchange Act of 1934, before Triplecrown engages its independent accountant to render audit or non-audit services on a going-forward basis, the engagement will be approved by its audit committee.

In accordance with Section 10A(i) of the Securities Exchange Act of 1934, before CAH engages its independent accountant to render audit or non-audit services on a going-forward basis, the engagement will be approved by its audit committee.

#### Code of Ethics

In October 2007, Triplecrown s board of directors adopted a code of ethics that applies to Triplecrown s directors, officers and employees as well as those of its subsidiaries. A copy of Triplecrown s code of ethics may be obtained free of charge by submitting a request in writing to Triplecrown Acquisition Corp., 970 West Broadway, PMB 402, Jackson, Wyoming 83001.

Upon the consummation of the merger, CAH will adopt a similar code of ethics that will apply to CAH s directors, officers and employees as well as those of its subsidiaries.

89

All Other Fees 175

## **BUSINESS OF CULLEN AGRITECH**

### **Business Overview**

Cullen Agritech is a newly formed company formed in June 2009 committed to the development and commercialization of advanced agricultural technologies. Cullen Agritech will offer advisory services associated with the development and implementation of efficient farming techniques. This will promote a methodology that incorporates components of New Zealand s pasture-based farming system, scientifically adapted for use in local markets. Cullen Agritech s principle focus will be to improve agricultural yields through pasture and animal sciences.

Upon completion of the merger Cullen Agritech will acquire rights with respect to a proprietary, pasture-based, farming system for the production of raw milk, primarily in the Southeastern United States. This farming system has been proven through application on research farms in the state of Georgia in the US. The resulting know-how associated with its farming system, including the constituent components incorporated within the holistic system, is proprietary information owned by Cullen Agritech. Natural Dairy Inc. ( Natural Dairy or NDI ) a wholly owned subsidiary of Cullen Agritech has been formed to rollout a long term scalable dairy farming operation utilizing the Cullen Agritech farming system in the Southeastern United States.

Natural Dairy intends to acquire farmland in the Southeastern United States for the implementation of Cullen Agritech's farming systems. That region has been strategically selected as being suitable for Cullen Agritech's pasture-based farming system.

## **Industry Overview**

## **Global Farming and Agricultural Technologies**

An increasing global population and an increase in the standard of living in some of the world slargest developing nations will continue to put demand on food production in the medium to long term. According to The Food and Agricuture Organization of the United Nations, the current worldwide demand for food is predicted to double by 2050, requiring a vast improvement in the productivity of our arable and grassland resources.

According to the report of World Resources Earth Trends Institute, as a proportion of the world s landmass, arable land makes up less than 11% (1.5 billion ha) and competition for this resource has more than doubled between 1950 and 2000 as the amount of arable land per capita fell from 0.52ha/person to 0.25ha/person (due to the increasing population). In addition, regulatory and social pressures suggest productivity gains must be achieved without jeopardizing environmental or economic sustainability and in recognition of the demand for the ethical treatment of food animals.

## Chart 1: World Population Growth and Arable Farmland Per Person

## Source: Missouri Grazing Conference 2009, Grazing Around the World

A paradigm shift may be required in many of the world s food production systems to compensate for these challenges. Approximately 30% of the arable land in the world has become unproductive over the last 40 years through destructive cropping practices that have depleted the soil resource (Food, Land, Population, and the US Economy, Pimertal and Giampietro, Nov 1994). In addition, the common techniques currently used are inefficient confinement animal production systems, where the feed (grain crops) and fuel inputs require ten times the energy than is contained in the food they produce (The Tightening Conflict; Population, Energy Use, and the Ecology of Agriculture, Permental and Giampietro, November 1994).

Despite its small size, New Zealand has risen as one of the world's agricultural leaders. A limited land base, an export-dependant economy and removal of all government intervention drove New Zealand's innovation in agriculture. The New Zealand Government removed all agricultural subsidies, tariffs and quotas in 1984. New Zealand now exports dairy, sheep and beef products all over the world and can supply the populous European and North American markets with product at a fraction of the cost of domestically produced items (Comparative Energy and Greenhouse Gas Emissions of New Zealand's and the UK's Dairy Industry, Saunders and Barber, 2007). New Zealand's production model is based on the use of renewable pasture resources that enable both economic (due to low input costs) and environmental (lower energy requirements) sustainability.

Government research and development organizations and private agricultural technology companies are focused on achieving further improvements in crop yields to meet growing food demand, while reducing the environmental impacts of agriculture. These two goals are often inversely related so research and development efforts have looked at improving the efficiency of production (production per unit of input) in developed countries and applying existing crop technologies to increase production from poorly or underutilized land in developing countries. The larger global biotechnology companies develop genetic modification techniques to improve crop efficiency (primarily through fertilizer and water), though the impact of such technologies are currently limited due to required development time and public resistance to Genetically Modified Organisms (GMO) technologies. The most immediate advances in global agricultural production are being made with

the adoption of modern farming technologies in developing countries to fully unlock the production potential of previously underutilized land. This has been achieved: (i) by foreign investment of capital and technology into the developing markets with the establishment of corporate farms, and/or (ii) through education and funding of the local farmers through government programs and incentives.

Generally, the foreign investment model has been most successful at improving food production in developing nations in comparison to the education and funding model because local farmers that they are often small subsistence producers who are unable to or unwilling to learn and implement new technologies.

Almost all technology movements around the world have been focused on production of grain crops for either human or animal consumption. To date, there are very few agricultural technology companies that seek to deploy new grazing technologies despite the vast potential to do so in many of the world s largest consumer markets. Fonterra, New Zealand s largest milk cooperative, has developed and implemented dairy farms in Brazil and China using some of New Zealand s advanced pasture technologies and New Zealand Farming Systems Uruguay (a company listed on the New Zealand Stock Exchange) is developing over 80,000 acres of grazing dairies in Uruguay. While these investments are relatively large, they represent a very small piece of the global market for grazing technologies and few companies with the ability to deploy these technologies globally.

## **Industry Overview**

## **Origins of the USA Confinement Dairy Industry**

Prior to the 1940s, the United States dairy and beef industries were small, low intensity pastoral-based family farms with less than 20 cows per farm. The main source of animal feed was pasture during the growing season and conserved silage and hay during the winter.

After the great depression in the 1930s, the United States began initiatives to secure its future food and supply by investing in research and development and mechanization to improve the productivity of crop farming. Seed technology, mechanization of farming processes together with cheap fuel and fertilizer saw corn yields increase from 29 bushels/acre in the 1900s to over 130 bushels/acre today. This massive increase in the productivity of all row crops (corn and soybeans), surpassed the demand for human consumption, creating large stockpiles of cheap grain. This cheap grain was used to feed production animals (poultry, pork, beef and dairy) and soon became the core food source which these industries were based.

This fundamental change, over the last 70 years, from a pasture-based diet to a corn/soybean-based diet, required the development of feeding and storage systems, breeding programs designed for animals to milk on grain based diets and the concentrated housing of animals to make the use of processed feeds more efficient. Now over 58% of all corn grown in the United States is fed to animals (5.6 billion bushels), with over 700 million bushels fed to dairy cows. However, as grain feeding became more expensive over the years, animals were forced to produce more milk to cover rising costs, thus diminishing the animal shealth and lifespan. As a result, current animal turnover is often greater than 40% per annum and cows sometimes last less than 2 lactations.

Alternative uses for corn (such as ethanol) and the environmental concerns around confinement farms have pushed this production model to breaking point. The United States industry now relies heavily on government subsidies and support to exist, but such support fails to offset the deficit resulting from the high cost of production.

## **Origins of the New Zealand Pasture Based Dairy Industry**

The New Zealand dairy industry originated on small family pasture-based farms and the subsequent intensification and productivity improvements led New Zealand s industry to continue to utilize pasture as the primary feed source. New Zealand s mild climate generally lends itself to a year round grazing system and has been a major incentive to promote year-round grazing as the production method of choice. In addition, New Zealand did not have the mechanization, land availability and cheap fertilizer to create animal industries based on grain crops. In fact, New Zealand imports grain products for its own human consumption making the feeding of animals with these products inefficient and expensive.

New Zealand was driven to improve its animal productivity by the fact that a country with a small population, it is an export nation that relies on the ability to produce large amounts of quality product to the world at the lowest possible cost. This is in direct contrast to the United States industry, which is largely a domestic market with less focus on competing in world markets.

The greatest spur to New Zealand s improvement in farm productivity and profit was the removal of all government subsidies to agriculture in 1984. At that time, it was predicted that a large number of New Zealand farmers would go out of business. To resolve the problem, there was a massive public and private sector investment in research and development to design efficient and cost-effective pasture-based animals that is currently implemented in New Zealand. The use of sustainable and renewable pasture crops to produce milk and meat enables New Zealand to export these products into the European Union and the United States at a fraction of the cost and carbon footprint compared to the same domestically produced items.

In further contrast to the United States confinement-based dairy industry, New Zealand cows are smaller, more efficient at turning feed (grass) into milk, have higher milk solids and enjoy better health, longevity and reproductive performance Dairy Industry. A New Zealand cow can live for 5-6 lactations and is generally culled for herd improvement reasons rather than health and reproductive reasons. A negative feature of this pasture-based system is that the New Zealand cows have lower overall milk volume. However, management believes the numerous positive aspects addressed above outweigh a lower overall production of milk.

## **United States Dairy Industry**

The United States dairy farming industry is made up of approximately 57,127 operations averaging 163 cows per farm, with 72% of farms having less than 100 cows. The top 10% of farms account for almost 65% of the total cow numbers. Dairy production in the United States is almost exclusively based on the confinement feeding of cows in freestall barns, with a minority of production coming from farms that use pasture as their primary feed source. Over the last 30 years, there has been some aggregation in the production base as many of the smaller operators exit the business and are replaced by larger corporate farms. There has also been a geographical shift in the production base with the larger corporate farms moving west (California, New Mexico and Arizona) and the smaller operators largely exiting the business in the east, creating a large imbalance in regional supply and demand (Source: (Farms, Land in Farms, and Livestock Operations 2008 Summary, Agrcultural Statistics Board USDA-NASS, February 2009; USDA)

The United States milk processing industry is made up of 196 registered co-ops which market over 86% of all milk. The largest of the co-ops is Dairy Farmers of America, which has a production market share of approximately 20% and is the main national player in the United States marketing 30% of all milk produced. There are a total of 1,475 dairy processors (Source: National Milk Producers Federation, Dairy Producer Highlights , 2008) in the United States, which manufacture all products from fluid milk to cheese and ice cream. The dairy retail market generated approximately \$100bn of revenue in 2008, with fluid milk generating approximately \$25bn billion. (Source: USDA, NASS) As indicated in Chart 2 below, U.S. milk production has steadily increased over the past 8 years.

#### Chart 2: Milk Production in the United States, 2000 to 2008

Source: USDA

The greatest input cost for United States dairy farmers is the high use of processed corn and soybean-based feeds. This high input cost system is vulnerable to milk price declines and has a high level of exposure to fluctuations in the pricing of commodity feed and fuel products; this is illustrated in Chart 3 below. United States confinement dairy farms also require greater fuel, machinery and labor costs as the requirements for each in the shipping, storage and feeding of animals in confinement is greater than grazing-based operations where the cows harvest their own food. The current problems that face the United States dairy industry are attributable to this high input costs system. While average milk prices have increased over the last 30 years, the production costs (feed and fuel) have increased at a greater rate (Source: Farms, Land in Farms, and Livestock Operations 2008 Summary, Agrcultural Statistics Board USDA-NASS, February 2009). Human demand for crop products (corn and soy) competes directly with feed for confinement livestock production and the demand for these products is likely to increase as the world population grows resulting in upward pressure on pricing for feed.

Chart 3: Cost of Production and Price received for Every 100 cwt of Milk Production Using the Confinement Farming Model in the U.S.

Source: Redefining Our Dairy Industry, Randy Mooney, presented at the Missouri Dairy Grazing Conference 2009 (USDA Data)

# **Supply Shortage in the Southeastern United States**

Despite the East Coast of the United States being the largest fluid milk market in the world, 83% of milk is produced in the Central and Western States (37% of Population) with only 17% being produced in the East (62% of the population) (Source: USDA; US Census Bureau). As a result, there is insufficient production on the East Coast to meet local demand. For years, population in the Southeastern United States has outpaced milk production. According to management survey of milk cooperatives, approximately 400 million pounds of milk are transported from the Western and Midwestern United States to the Eastern United States every month. Since 1998, milk production in the Southeast has fallen from approximately 13 billion pounds to approximately 9.5 billion pounds as many of the small farmers in the area left the industry with declining economic viability of their production model (Source: USDA). Recent census data from the United States Department of Agriculture (USDA) indicates this trend will continue as dairy farms and cows leave the Southeast and the industry production base is aggregated into larger corporate confinement farms in the Western United States.

2008 US Dairy Market: Production by Region

Total US Milk Production by Region, June 2009

Chart 4: total U.S. Milk Production by Region, June 2009.

Source: USDA

# **US Milk Pricing**

In order to achieve stability and consistency in milk pricing, the USDA implemented an Agricultural Marketing Order system for milk. This was requested by producer (farmer) vote and is imposed on the handlers of perishable goods (milk) to create uniform marketing practices and prevent excessive power of the handler over the producer.

The United States dairy industry is broken up into ten different Federal Marketing Orders, issued under the Agricultural Marketing Agreement Act of 1937 ( AMAA ), the boundaries of which are determined by local competition for fluid milk supply. The current Marketing Orders are Pacific Northwest, Western, Arizona, Central, Southwest, Northern Midwest, Mideast, Appalachian, Northeast, Southeast and Florida. The AMAA was amended by way of the Milk Regulatory Equity Act of 2005 (signed into law April 11, 2006), which was intended to ensure regulatory fairness between and among dairy farmers and handlers for sale of packaged fluid milk in Federal milk marketing order areas and into certain non-federally regulated milk marketing areas. This amendment essentially required all handlers to pay a Federally ordered minimum price for fluid milk.

Milk is separated and priced in four classes depending on the demand for various dairy products and certain qualitative factors in the milk (primarily butterfat content that must be 3.5% or greater in order for milk to fall into Class I).

#### Milk classes:

Class I = used in fluid products;

Class II = used in manufacture of ice cream and other soft products;

Class III = used in manufacture of cheese; and

Class IV = used in manufacture of butter and milk powder

The price for each class is set as follows:

All milk is priced in dollars (\$) per hundredweight (cwt = 100 lbs); Class III and IV prices are determined by National Agricultural Statistics Service (USDA) survey of cheese (Class III) and butter/milk powder (Class IV) each month; and

Class II milk is then set at Class IV plus \$0.70/cwt

96

US Milk Pricing 183

Class I is set at Class III or Class IV (whichever is greater) plus the Class I differential , which reflects the local demand for fluid product (i.e. the greater the shortage of fluid milk the higher the Class I differential)

As Cheese is the dominant dairy product in the USA market, Class III milk price is more often than not used in the Class I calculation

## **US Dairy Support Prices**

Prior to October 2008, there was a direct price support on manufacturing grade (Class III) milk to prevent it falling below \$9.90/cwt (hence Class I would not fall below \$9.90 plus the Class I differential). Milk prices are now kept from falling below a floor price by utilizing various support prices on specific dairy products (rather than on the milk itself). When dairy product prices fall to the support level, the Federal government purchases products through the Commodity Credit Corporation ( CCC ) to prevent prices from falling any further. In July 2009, the USDA announced an increase in these support level prices.

Chart 5: 2000 to 2009 Class I Milk Price in the Southeastern U.S. (\$ per hundred weight)

Source: USDA; Brian Gould, Agricultural & Applied Economics, UW Madison

#### 2007-2009 Conventional Milk Price Environment

Throughout 2007, the milk industry experienced rapidly increasing and historically high conventional milk prices. In 2008, it continued to experience historically high conventional milk prices (as illustrated on Chart 5). Contributors to the high prices included high feed and fuel costs, as well as global demand for dry powdered milk. This was compounded by significant and sustained increases in diesel fuel and resin costs. In the fourth quarter of 2008, the industry experienced a sharp decline in export demands and favorable movements in energy cost components resulting in declines in conventional milk, diesel and resin prices.

Beginning in the fourth quarter of 2008, the industry experienced a sharp decline in export demands and a strengthening of the dollar against other currencies which resulted in declines in conventional raw milk prices. [As a result of decline in demand and as a result from over production (in 2007/2008), the milk price dropped from [\$18] and the U.S. government pursuant to ccc (above) had to purchase excesss.] Throughout the first six months of 2009, the Class I pricing declined significantly from the average for the fourth quarter of 2008 and remained significantly below comparable 2008 levels.

Based on the cyclical nature of the industry and currently challenged dairy economics, formal herd reductions are expected to lead to reduced supply and the average Class I price is expected to increase gradually in the second half of 2009. Chart 6 below sets forth the latest CME Class I milk price futures which include a \$4.05 price premium paid to dairy produced in Georgia.

#### **Class 1 Milk Price Futures for Georgia**

Chart 6: Class 1 Milk Price Forecasts for Georgia, USA

Source: CME Group as of September 1, 2009

# **Business Strategy**

Cullen Agritech s principle focus will be to improve agricultural yields through forage and animal sciences. Significant time and resources have been invested by the key personnel now with Cullen Agritech in developing the necessary capabilities to deploy Cullen Agritech s pasture based technologies. Further investment will support the on-going development of the unique farming system know-how to be acquired upon completion of the Merger. Cullen Agritech has assembled an experienced group of pastoral scientists and dairy science industry participants to further enhance its position as an innovation and technology company with the ability to bring efficient New Zealand grazing-based production systems to the agricultural community in the U.S.

Cullen Agritech was formed to develop, adapt and implement grazing based farming systems in regions of the world, where the geophysical and climatic conditions are suitable for a pasture based model. However, while the potential for the pasture or grazing model is significant in many of the world s developed and developing economies, the systems are highly specific and require significant adaptation and modification. The construction of a robust management framework is essential to deploy the systems effectively in other regions of the world where unique sets of geophysical, climatic and social conditions exist. Specifically, Chile, China, Uruguay, the United States and parts of Eastern Europe fit this criteria.

In addition to ongoing research and development in the areas of pasture sciences, animal genetics and farm management strategies applicable to the global agricultural industry, Cullen Agritech will offer consultancy services to various agricultural industries in which its proprietary systems and know how will help the agricultural industry benefit from increased yields.

Cullen Agritech has identified the global dairy industry as a primary opportunity in which its systems can be applied to improve yields on land and drive cost-base efficiencies. Upon completion of the merger it will have acquired rights to a proprietary farming system developed through adaptation of generic New Zealand farming sciences, processes, technologies and breeding strategies that can effectively be applied to liquid milk production in the Southeastern United States. It will apply this proprietary farming system to its own dairy farming operations through its wholly owned subsidiary, Natural Dairy.

Upon completion of the merger Natural Dairy will deploy Cullen Agritech s proprietary pasture-based farming system for the development of dairy farms in the United States. Natural Dairy aims to build a large-scale farming operation in the Southeast United States for the production of raw milk for the local consumer market.

# Key Components to Cullen Agritech s Pasture Based Grazing System

The proprietary farming system which Cullen Agritech will own upon completion of the merger is applicable to the United States dairy industry and has the potential to significantly increase yields on land in the Southeastern United States. This system is based on a grazing-based farming model, whereby dairy cows are primarily fed a renewable pasture resource as opposed to a corn-based feed. Cullen Agritech believes that with time, effort and resources the intellectual property can be adapted for implementation in markets beyond the Southeastern United States. Key components to Cullen Agritech s farming system are as follows:

Farm Selection and Design: the farming system requires land with specific characteristics that is converted and designed to enable and promote an optimally efficient farming system.

Pasture Sciences: the farming system incorporates a proprietary pasture production strategy that optimizes annual cow feed supply and milk production through an integrated mix of different pasture crop species and varieties, and management thereof. The different pastures are designed to achieve the desired quantitative and qualitative targets set to develop fertilization and irrigation strategies promote efficient grazing management practices.

Animal Genetics, Breeding & Health: the farming system incorporates animal management systems and genetic criteria to compliment the grazing model. The cows are smaller than those of a confinement model, and bred to live longer and achieve greater reproductive rates. Automated animal management systems are used in the large grazing operations to monitor the animals production and health and identify animals that are ready to be bred.

Farm Management: the farming system utilizes efficient milk harvesting and waste management technologies. These milk harvesting systems allow two people to milk approximately 400 cows per hour, greatly reducing labor requirements and waste production. Effluent management systems recycle waste water back onto the pasture as a source of fertilizer eliminating the need for effluent storage while reducing the risk of environmental contamination.

Systems & Training: highly trained farm managers who are skilled in grazing management, pasture crop production and animal science will be employed. The managers, with the help of scientific staff, will provide training for all subordinate farm staff. Investment will also be made in local and international universities to develop a skilled labor pool to draw.

# Implementation of Cullen Agritech s Farming Systems in the Southeastern US

Natural Dairy has identified over 40,000 effective acres (effective means farmable land, typically 75-80% of the property) suitable for deployment of the pasture-based farming systems over the first 12 months following

Business Strategy 186

consummation of the transactions described in this proxy statement/prospectus. These sites are primarily located in the state of Georgia and have access to an ample supply of high quality water. Engineers and other contractors have been identified to complete the conversion of the identified land as well as manufacturing companies for the installation of sheds and milking systems. Livestock required to stock the farms has also been identified through multiple breeders.

The estimated timing of the conversion of each farm is as follows:

(1) Facilities: Infrastructure and pastures in place within initial 4 months.

#### **TABLE OF CONTENTS**

(2) Initial Livestock: 70% of cows stocked within initial 6 months.
 (3) Remaining Livestock: final 30% of cows stocked within initial 12 months.
 The expected roll-out of the 40,000 identified acres is as follows:

During June 2009, Triplecrown and Natural Dairy entered into an agreement to purchase a 3,670 acres of land in Georgia. Natural Dairy plans to use this land to deploy Cullen Agritech s proprietary farming system on three (3) contiguous 1,000 effective acre farms. The assumptions below give an approximation of the costs and profitability of a single 1,000 effective acre farm:

# **Single Farm Parameters**

The following key assumptions relate to the table below:

Milk Production: 85cwt per cow per annum (improving over time)

Cost of production: Initially \$15.5/cwt (improving to \$10/cwt over time)

Surplus livestock revenue: \$250 per head, per annum

Ţ	Year	One	Two	Three	Four
F	Revenue (\$mn)	\$ 2.7	\$ 4.1	\$ 4.4	\$ 4.6
I	EBITDA (\$mn)	0.9	1.5	1.8	2.1
100					

Below is a financial summary of the 12 month rollout of 40 1,000 effective acre farms. This is based on the availability of over \$400 million in capital. Milk price forecasts are based on Class III forecasts provided by the Food and Agricultural Policy Research Institute (FAPRI), University of Missouri, with the average Georgia Class I differential of \$4.05 added to these Class III forecasts. A net haulage cost of \$0.60 has also been deducted from the expected price.

	2010	2011	2012	2013
Total Cumulative Capital Deployed (\$mn)	\$ 365.8	\$ 411.8	\$ 414.8	\$ 414.8
Total Cumulative Equity Deployed (\$mn)	146.3	164.7	165.9	165.9
Acres Being Farmed	40,000	40,000	40,000	40,000
Number of Cows	23,958	86,802	90,508	90,362
Total Milk Production (mn lbs)	197.7	684.9	818.0	872.0
Milk Price (\$/cwt)	\$ 16.5	\$ 18.2	\$ 18.6	\$ 18.7
Total Revenue (\$mn)	46.6	152.0	185.9	190.8
EBITDA (\$mn)	10.0	51.4	82.1	88.5
EBITDA Margin	21.4%	33.8%	44.2%	46.4%

The following potential future organic growth factors are not included in the forecasts above:

Additional Equity: Currently assuming no additional capital is raised after year one Revenue Growth: Premium Grass Fed milk pricing; premium surplus livestock pricing; Margin Expansion: Improved cow genetics; improved forage yields;

Multiple expansion: Licensing and JVs via Cullen Agritech IP; appreciation of farmland and livestock; and Return on capital expansion: leasing of land; leasing of livestock; additional capital structure efficiencies through the availability of potential financing guarantees and grants from both New Zealand and the US

These statements are based on assumptions and estimates that management believes are reasonable based on currently available information; however, management's assumptions and the Company's future performance are subject to a wide range of business risks and uncertainties, and there is no assurance that these goals and projections can or will be met. Any number of factors could cause actual results to differ materially from expectations.

# **Cullen Agritech s Competitive Strengths**

# Key Cullen Agritech personnel have extensive experience in improving yields through applying pasture based farming techniques

Cullen Agritech will rely on in-house scientific experts, along with an adjunct Advisory Board made up of some of the world s leading scientists, policy analysts and business leaders in the agricultural arena. Additionally, Cullen Agritech s Chief Scientific Officer, Dr. Richard H. Watson, has an extensive background in pastoral science and technology, from the laboratory to commercialization and industry application of technologies.

Cullen Agritech also has quantitative geneticists that have the ability to develop lines of dairy cattle designed to produce milk efficiently on pasture. Cullen Agritech has exclusive access to embryo, semen and genetic screening technologies (SNP-chip) to accelerate genetic improvement and deployment of these lines in the United States dairy industry. Only Cullen Agritech and its customers will have access to these grazing lines.

# Proven Model through research farms developed by Key Cullen Agritech Personnel to provide key cost advantages

The proprietary farming system which Cullen Agritech will own upon completion of the merger was developed and proven on research farms in Girard, Georgia, the first of which was established in late 2007 by

the parties from whom Cullen Agritech will acquire the rights to the farming system. These farms were established to develop and test the proprietary grazing system that will be rolled-out in the Southeastern United States through Natural Dairy. The first research farm began producing milk in March 2008. During 2008, it was used to refine and develop the farming system, the know-how of which will make up Cullen Agritech s intellectual property. This research was focused on the development of a pasture crop system to maximize the production and utilization of grown pasture. Breeding and calving season trials have been conducted to optimize the relationship between feed grown on farm and the feed demand of the herd. During 2009, the farming system was refined and has achieved favorable production cost results, proving the efficiency of the system.

# Forage based system provides lower cost per CWT of dairy production:

The cost of producing milk will vary greatly depending on the region, the exact management practices and quality of farmers. For a majority of dairy farmers, the high dependence on the use of corn-based concentrate as a feedstock results in an cost base. The use of pasture as a replacement for corn-based concentrate in Cullen Agritech s model reduces this expense, creating a much more economically-sustainable cost structure. Cullen Agritech s model will also be less labor intensive and has reduced animal health costs due to healthier and less confined conditions.

The chart below depicts the cost structures of various dairy farm operating models. The New Zealand pasture-based grazing model has generally operated at \$8.00-\$10.00 per hundredweight ( cwt ) cost levels. In comparison, over the first seven months of 2009, the research farm has operated at an average cost of \$13.60 per cwt. In order to provide potential upside to the projections, during the first two years of production, Cullen Agritech has incorporated into its projections an average cost per cwt of production of \$15.20, which is higher than it has achieved on the research farm. In comparison, the traditional United States confinement based model operated at an average of \$19.10 per cwt during 2007 and 2008 (\$20.02 per cwt for the US Southern Seaboard region).

Chart 7 Cost Comparison of Different Farming Models

Source: USDA, Cullen Agritech Management

# Natural Dairy is strategically located in a region of significant demand coupled with a substantial shortage in supply

Natural Dairy s roll-out is focused on the Southeastern United States where there is currently a significant shortage in the supply of fresh liquid milk. In addition, the Eastern Seaboard represents the largest fresh liquid milk market in the world. Natural Dairy is strategically positioned to help fill that supply gap and produce milk for this market, which is currently undersupplied. Management estimates that \$2 billion of capital would be needed to be deployed to satisfy the current shortage.

# Cullen Agritech s systems can achieve high production per cow

Cullen Agritech will utilize a pasture production systems that optimizes seasonal qualitative attributes of the pasture to best match the energy demands of the cow. Under these systems, the cow is provided sufficient nutrients to meet her needs for body maintenance and milk production. Cullen Agritech uses energy balances

to ensure that the cows are fed enough pasture to achieve the highest possible production targets in the most cost effective manner. We believe this level of feed management sets Cullen Agritech apart from other grazing operations in the USA.

# Upon completion of the merger, Cullen Agritech s system results in healthier livestock and increased longevity

The common United States dairy industry cow is the US Holstein. Cullen Agritech s model will be suited to smaller framed livestock such as Jersey/Holstein cross-breeds or a Friesian Holstein. These breeds of livestock generally have longer productive lives than a typical US Holstein, which is further lengthened by the healthier conditions associated with Cullen Agritech s pasture-based farming system which it will own upon completion of the merger. This longevity is expected to result in reduced livestock culling rates resulting in additional revenue from surplus livestock sales. Livestock managed under the system are also likely to have fewer health issues due to increased exercise and exposure to cleaner, less confined living conditions. As a result animals are healthier and the speed at which infection can spread throughout a herd is reduced. This, in turn, results in increased longevity as well as reduced health-related operating expenses on the farms.

#### Reduced labor costs

Cullen Agritech will utilize milking systems which will be custom made to maximize efficiency and minimize labor costs. This technology, combined with a unique and efficient farm design and management strategy will result in reduced labor costs, further reducing the cost of production under the system that Cullen Agritech will be utilizing.

# Potential to achieve higher pricing in the future

Management believes the demand for naturally produced animal products is increasing as the population s concern with how their food is produced increases. General awareness of the animal ethics and human health benefits of the grazing-based production system have also grown. For instance, a USDA survey showed that 48% of United States consumers now recognize Grass-fed as a brand.

Products from animals fed on a pasture dominant diet have been found to contain higher levels of a number of naturally occurring metabolites that have proven human health benefits. The fermentation of the pasture diet in grazing animals by rumen bacteria create higher levels of conjugated linoleic acid (CLA), omega-3 and 6 fatty acids and vitamins A&E in the milk. Production of these in confinement cows is reduced by the heavy-starch grain diet, which reduces the formation of these beneficial fermentation products.

Currently, there is a small but rapidly growing market for grass-fed or pasture-fed beef products. However, the grass-fed milk products are is virtually non-existent due to dominance of the confinement model and the lack of producers who have the technical knowledge to produce milk on pasture year round.

Natural Dairy has the ability to produce grass-fed milk year round and its production levels may in the future result in availability of separate processing, with Natural Dairy s milk as a grass-fed product being processed separate from other milk. However, Natural Dairy milk will be initially sold as standard milk along with milk from confinement production. This means it will not initially receive premium pricing on the basis of its grass-fed product. Accordingly, Natural Dairy has not included any premium pricing due to the grass-fed nature of its product in its financial projections.

# Strategic agreements and relationships allow for efficient large scale rollout of pasture based system

Cullen Agritech entered into a strategic cooperation agreement dated with New Zealand Agritech Inc. ( NZ Agritech ) to promote the interests of NZ Agritech and its members. Cullen Agritech will assist members of NZ Agritech to mitigate barriers of market entry and provide the opportunity to realize potential growth in various markets. This alliance reflects an important connection to participants of New Zealand s agricultural technology industry and enables Cullen Agritech to offer its customers the benefit of advanced technologies.

We believe that this and other potential strategic relationships will help to build Cullen Agritech s business and operations.

## **Products and Services**

Cullen Agritech will provide a range of farm management and technology services that help improve the productivity and profitability of food animal production. While Cullen Agritech will specialize in grazing systems and pasture technologies, products and services can be provided for the traditional confinement-based dairy farm operator to integrate one or more grazing technologies into their operations or just refine their feed and animal management to improve profitability within their confinement system.

Cullen Agritech intends to work with producers, industry leaders and governments to improve the economic and environmental sustainability of food animal production. This will include whole farm management plans and feasibility studies, feed production, storage and feeding strategies, genetic improvement and rearing of replacement stock and engineering waste management solutions.

Natural Dairy will be a producer of Class 1 raw milk for the liquid market in the Southeastern U.S. and it will strive to produce milk that is of the highest quality in conformation with food safety standards. Natural Dairy will use a majority of crossbred cows that have a higher milk solid content (butter fat, protein and lactose) than US Holsteins which we believe will consistently produce milk which will exceed the 3.5% butter fat standard for Class I milk. Natural Dairy will not use the Bovine Growth Hormone (rBST) to produce its milk. Natural Dairy milk should receive Class I (fluid) milk grading and pricing with its high butter fat content, low somatic cell and bacteria counts. Natural Dairy may form one or more subsidiaries to implement its business plan.

# **Intellectual Property**

Cullen Agritech will acquire certain intellectual property pursuant to a Deed of Assignment upon completion of the Merger. Cullen Agritech s intellectual property will consists of a farming system that has been developed through research and development. The research and development has focused on improving yields on agricultural land through development of proprietary farming practices based on specific pasture sciences, livestock genetics and breeding strategies, farm technologies and farm management strategies. These efforts have enabled Cullen Agritech to develop and on a proprietary farming system that is applicable to the United States dairy industry with potential to significantly increase yields on land in the Southeastern United States. This system is based on a grazing-based farming model, whereby dairy cows are primarily fed a renewable pasture resource as opposed to a corn-based feed.

Cullen Agritech believes that with time, effort and resources the intellectual property can be adapted for implementation in markets beyond the Southeastern United States.

The existence of the Advisory Board and thr relationships with the members presents future opportunities for Cullen Agritech s development of intellectual property. Collectively, the membership represent the world s preeminent agricultural science universities and comprises individuals widely regarded as experts within their chosen field.

Cullen Agritech will seek to protect the intellectual property which consists of know-how with respect to a farming system by using a combination of trademark, patent and trade secrets laws, licensing and nondisclosure agreements and other security measures. The intellectual property is itself continually evolving as technological advances are made within the agricultural science community.

# **Joint Venture Opportunities**

Products and Services 195

Cullen Agritech s business encompasses a broad exposure to third parties operating within the agricultural science industry, including those which have developed or that otherwise promote products and/or technologies that compliment Cullen Agritech s business objectives. A number of these third parties are seeking to expand into markets in which Cullen Agritech will undertake business activities. Management believes that Cullen Agritech is positioned to partner with such third parties to assist with market entry and joint venture opportunities exist in respect of product and technology adaptation services, in addition to potential marketing arrangements.

As noted above, Cullen Agritech has a strategic alliance with NZ Agritech, New Zealand s national representative body for agricultural technology companies operating in New Zealand, an internationally respected market for efficient farming practices and methodology and where dairy farming is exclusively pasture-based. The relationship with NZ Agritech presents Cullen Agritech with the opportunity to enter joint

ventures and strategic alliances with New Zealand companies offering innovative products and technologies which promote efficient farming systems, including those seeking assistance with adaptation to the Southeastern, United States. The strategic alliance with NZ Agritech recognizes Cullen Agritech as the preferred partner for the exporting of innovative products and technology from the New Zealand grazing farming system to the markets in which Cullen Agritech is operating.

Cullen Agritech s Advisory Board presents on-going opportunities for joint ventures with market participants and developers of related intellectual property.

Within the United States, Cullen Agritech s know-how in relation to its efficient pasture-based farming model will present the opportunity for joint ventures with federal and state departments and businesses including dairy co-operatives, universities and training institutions and farmers.

# **Customers/Sales and Marketing**

Cullen Agritech intends to partner with and provide services to some of the worlds largest agricultural companies, including producer cooperatives, corporate farmers and investment funds and agricultural technology providers.

Cullen Agritech s expertise can be applied across a range of global regions and production systems that utilize pasture systems and technologies for animal food production and can assist industry and government organizations in adapting these technologies to their region and production requirements. Natural Dairy s customer base will be predominately milk cooperatives that supply processing facilities.

# Competition

Potential competitors to Cullen Agritech are large agricultural technology and service providers that develop a globally focused consultancy capacity that is focused on the grazing model and technologies. To Cullen Agritech s knowledge, there are currently no other entities operating in the global grazing technology space, provided, however companies could potentially develop this capability. These companies include PGG Wrightson (NZ), Livestock Improvement Corporation (NZ), New Zealand Farming System Uruguay (NZ), Grasslands Consultancy LLC (Mo, USA), Manuka Farming (Chile) and Fonterra. To managements s knowledge, none of these companies currently provide agricultural consultancy services of significance outside their country of incorporation and may have limited capacity to move to other regions as a technology provider.

Natural Dairy will compete with other large scale milk producers in the Southeastern U.S.

# **Legal Proceedings**

Neither Cullen Agritech nor Natural Dairy is party to any legal proceedings and no proceedings are pending against or involving Cullen Agritech or Natural Dairy.

# **Properties**

Natural Dairy and Triplecrown entered into a land purchase contract with Grimsley LLC on June 27, 2009, as amended, to purchase 3,670 acres of farmland in the State of Georgia. A deposit in the aggregate amount of \$1.7 million has been made and the parties intend to close on the purchase contract on September 15, 2009.

# **Government Regulations**

## **Public Health**

As a manufacturer of food products, Cullen Agritech will be subject to a number of food-related regulations, including the Federal Food, Drug and Cosmetic Act and regulations promulgated thereunder by the United States Food and Drug Administration (FDA). This comprehensive regulatory framework governs the manufacture (including composition and ingredients), labeling, packaging and safety of food in the United States. The FDA:

regulates manufacturing practices for foods through its current good manufacturing practices regulations; specifies the standards of identity for certain foods; and

prescribes the format and content of certain information required to appear on food product labels. In addition, the FDA enforces the Public Health Service Act and regulations issued thereunder, which authorizes regulatory activity necessary to prevent the introduction, transmission or spread of communicable diseases. These regulations require, for example, pasteurization of milk and milk products. Cullen Agritech will be subject to numerous other federal, state and local regulations involving such matters as the licensing and registration of manufacturing facilities, enforcement by government health agencies of standards for its products, inspection of its facilities and regulation of its trade practices in connection with the sale of its products.

Product quality and freshness will be essential to receiving the best possible price for Cullen Agritech s milk. To monitor product quality, milk samples will be taken at each milk pick up from each farm and tested in quality control labs. Cullen Agritech believes its milk harvesting and temporary storage facilities will be in material compliance with all government regulations applicable to its business and are subject to regular monthly inspection from the state Department of Agriculture.

# **Employee Safety Regulations**

Cullen Agritech will be subject to certain safety regulations, including regulations issued pursuant to the U.S. Occupational Safety and Health Act. These regulations will require Cullen Agritech to comply with certain manufacturing safety standards to protect its employees from accidents.

## **Environmental Regulations**

Natural Dairy will maintain small above-ground petroleum storage tanks at many of its facilities. Cullen Agritech will periodically inspect these tanks to determine whether they are in compliance with applicable regulations and, as a result of such inspections, are required to make expenditures from time to time in order to ensure that these tanks remain in compliance. In addition, upon removal of the tanks, Natural Dairy may be required to make expenditures to restore the site in accordance with applicable environmental laws.

# Milk Industry Regulation

The federal government establishes minimum prices that Cullen Agritech receives in federally regulated areas for raw milk. Raw milk primarily contains raw skim milk, in addition to a small percentage of butterfat. The raw milk that will be produced by Cullen Agritech s facilities will be tested to determine the percentage of butterfat and other milk components and will receive payment for the raw milk based on the results of these tests.

The federal government s minimum prices vary depending on the processor s geographic location or sales area and the type of product manufactured. Federal minimum prices change monthly. Class I butterfat and raw skim milk prices (which are the minimum prices we receive for raw milk that is processed into Class I products) and Class II raw milk prices (which are the prices we receive for raw milk that is processed into Class II products such as cottage cheese, creams, creamers, ice cream and sour cream) for each month are announced by the federal government the immediately preceding month.

Some states have established their own rules for determining minimum prices for raw milk. In addition to the federal or state minimum prices, Cullen Agritech may receive producer premiums, procurement costs and other related charges.

Public Health 199

# **Employees**

Cullen Agritech currently has 1 full-time employee and Natural Dairy has 1 full-time employees. Cullen Agritech and Natural Dairy consider their employee relations to be good.

Natural Dairy has an employment agreement with Richard Watson, dated August 31, 2009. Pursuant to the Employment Agreement, Mr. Watson will be employed as Natural Dairy's Chief Scientific Officer, with a base salary of \$100,000. Mr. Watson may be entitled to receive a bonus of up to 50% of the base salary subject to the sole discretion of the Board of Directors. The initial term is for three (3) years, and shall automatically renew for one (1) year periods thereafter.

106

Employees 200

# CULLEN AGRITECH S MANAGEMENT S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion and analysis should be read in conjunction with the financial statements and accompanying notes of Cullen Agritech appearing elsewhere in this proxy statement/prospectus.

# **Overview**

Cullen Agritech is a newly formed company formed in June 2009 committed to the development and commercialization of advanced agricultural technologies. Cullen Agritech will offer advisory services associated with the development and implementation of efficient farming techniques. This will promote a methodology that incorporates components of New Zealand s pasture-based farming system, scientifically adapted for use in local markets. Cullen Agritech s principle focus will be to improve agricultural yields through pasture and animal sciences.

Cullen Agritech will acquired rights with respect to a proprietary, pasture-based, farming system for the production of raw milk, primarily in the Southeastern United States upon completion of the merger. This farming system has been proven through application on research farms in the state of Georgia in the US. The resulting know-how associated with its farming system, including the constituent components incorporated within the holistic system, is proprietary information owned by Cullen Agritech. Natural Dairy Inc. (Natural Dairy or NDI) a wholly owned subsidiary of Cullen Agritech has been formed to rollout a long term scalable dairy farming operation utilizing the Cullen Agritech farming system in the Southeastern United States.

Natural Dairy will implement Cullen Agritech s dairy farming techniques and systems following the purchase of land in the United States. Natural Dairy intends to acquire farms in the Southeastern United States for the implementation of Cullen Agritech s farming systems. That region has been strategically selected as being suitable for Cullen Agritech s pasture-based farming system.

Cullen Agritech s operations are subject to the risks and uncertainties detailed under the section titled Risk Factors appearing elsewhere in this proxy statement/prospectus.

# Critical Accounting Policies and Use of Estimates

# **Principles of Consolidation**

The consolidated financial statements include the accounts of Cullen Agritech and its wholly-owned subsidiary (collectively referred to in this section as the Company or Cullen Agritech ). All significant inter-company accounts and transactions have been eliminated.

#### **Use of Estimates**

The consolidated financial statements include the use of estimates, which management believes are reasonable. The process of preparing financial statements in conformity with generally accepted accounting principles requires the use of estimates and assumptions regarding certain types of assets, liabilities, revenues, and expenses. Such estimates primarily relate to unsettled transactions and events as of the date of the financial statements. Accordingly, upon settlement, actual results may differ from estimated amounts.

## **Other Comprehensive Income**

Total comprehensive income includes changes in equity that are excluded from the consolidated statements of operations and are recorded directly into a separate section of consolidated statements of comprehensive income. The Company s accumulated other comprehensive income shown on the consolidated balance sheet consist of unrealized gains and losses on investment holdings and the fair-value of hedging instruments.

# **Cash and Cash Equivalents**

Cash and cash equivalents represent cash on hand, demand deposits and highly liquid investments placed with banks or other financial institutions, which have original maturities of less than three months. The carrying amounts reported approximate their fair value.

107

Use of Estimates 202

#### **Accounts Receivable**

Accounts receivable consist of trade receivables recorded at original invoice amount, less an estimated allowance for uncollectible amounts. Trade credit is generally extended on a short term basis; thus trade receivables do not bear interest, although a finance charge may be applied to receivables that are past due. Trade receivables are periodically evaluated for collectibility based on past credit history with customers and their current financial condition. Changes in the estimate collectibility of trade receivables are recorded in the results of operations for the period in which the estimated is revised. Trade receivables that are deemed uncollectible are offset against the allowance for uncollectible accounts. The Company generally does not require collateral for trade receivables.

#### **Inventories**

Inventories are stated at the lower of cost or market. Product returns are recorded in inventory when they are received at the lower of their original cost or market, as appropriate. Obsolete inventory is written off and its value is removed from inventory at the time its obsolescence is determined.

# **Property and Equipment and Depreciation and Amortization**

Property and equipment are stated at cost. The Company charges to expense repairs and maintenance items, while major improvements and betterments are capitalized. When the Company sells or otherwise disposes of property and equipment items, the cost and related accumulated depreciation are removed from the respective accounts and any gain or loss is included in earnings.

Depreciation and amortization are provided on the straight-line method over the following estimated useful lives or lease terms of the assets:

Buildings and plant Machinery and equipment Office equipment 20 33 years5 14 years5 years

# **Intangible Assets**

Intangible assets are stated at cost. Patents and other Intangible Assets are amortized on the straight-line method over their remaining useful life. Such intangible assets are reviewed for potential impairment whenever events or circumstances indicate that carrying amounts may not be recoverable.

# **Biological Assets**

#### **Immature Biological Assets**

Biological assets consist of dairy cows held in pastures for milking purposes. Immature biological assets are recorded at cost, including two years depreciation from its parent, acquisition costs, transportation costs, insurance expenses, interest costs and feeding costs, incurred in bringing the asset to its intended productive state. Once the asset reaches productive state, the cost of the immature biological asset is transferred to mature biological assets using the weighted average cost method.

Interest costs associated with immature biological assets are capitalized in the period they are incurred. Interest is no longer capitalized when the asset reaches its productive state.

#### **Mature Biological Assets**

Mature biological assets are recorded at cost. When mature biological assets are retired or otherwise disposed of, the cost and accumulated depreciation are removed from the accounts and any resulting gain or loss is included in the results of operations for the respective period.

Depreciation is provided over the estimated useful live of the mature biological assets of 5 years using the straight-line method worth a residual value of zero.

Feeding and management costs incurred on mature biological assets are included as costs of goods sold on the consolidated statements of operations and comprehensive income.

Cullen Agritech will review the carrying value of biological assets for impairment whenever events and circumstances indicate that the carrying value of an asset may not be recoverable from the estimated future

cash flows expected to result from its use and eventual disposition. In cases where undiscounted expected future cash flows are less than the carrying value, an impairment loss is recognized equal to an amount by which the carrying value exceeds the fair value of assets. The factors considered by management in performing this assessment include current health status of the asset and production capacity.

#### **Income Taxes**

Income tax expense includes federal and state taxes currently payable and deferred taxes arising from temporary differences between income for financial reporting and income tax purposes.

#### **Earnings Per Common Share**

The Company adopted Statement of Financial Accounting Standards (SFAS) No. 128, Earnings Per Share . Basic earnings per share is calculated using the average number of shares of common stock outstanding during the year. Diluted earnings per share is computed on the basis of the average number of common shares outstanding plus the effect of outstanding stock options using the treasury stock method and convertible debentures using the if-converted method. Common stock equivalents consist of stock options.

#### **Revenue Recognition**

The Company recognizes sales upon shipment of merchandise. Net sales comprise gross revenues less expected returns, trade discounts, customer allowances and various sales incentives. Although no legal right of return exists between the customer and the Company, returns are accepted if it is in the best interests of the Company s relationship with the customer.

#### **Cost of Goods Sold**

Cost of goods sold primarily consists of direct and indirect manufacturing costs, including production overhead costs, shipping and handling costs for the products sold.

#### **Hedging Instruments**

The Company uses derivative financial instruments principally to offset exposure to market risks arising from changes in commodity prices. The Company enters into futures and forward contracts, which are designated as hedges of specific volumes of commodities (corn and milk). These derivative financial instruments are recognized in the Consolidated Balance Sheets at fair value.

On the date a derivative contract is entered into, the Company designates the derivative as a hedge of certain forecasted purchases or sales of corn or milk, respectively. For all hedging relationships, the Company formally documents the hedging relationships and its risk-management objective and strategy for undertaking the hedge transactions, the hedging instrument, the item, the nature of the risk being hedged, how the hedging instrument s effectiveness in offsetting the hedged risk will be assessed, and a description of the method of measuring ineffectiveness. The Company also formally assesses, both at the hedge s inception and on an ongoing basis, whether the derivatives that are used in hedging transactions are highly effective in offsetting changes in cash flows or fair values of hedged items. When it is determined that a derivative is not highly effective as a hedge or that it has ceased to be a highly effective hedge, the Company discontinues hedge accounting prospectively.

The Company discontinues hedge accounting prospectively when it is determined that the derivative is no longer effective in offsetting changes in the cash flows or fair value of the hedged item, the derivative expires or is sold, terminated or exercised, the derivative is de-designated as a hedging instrument because it is unlikely that a forecasted transaction will occur, or management determines that designation of the derivative as a hedging instrument is no longer appropriate. When hedge accounting is discontinued because it is probable that a forecasted transaction will not occur, the Company continues to carry the derivative on the Consolidated Balance Sheet at its fair value, and gains and losses that were accumulated in other comprehensive income (loss) are recognized immediately in earnings. When hedge accounting is discontinued because it is determined that the derivative no longer qualifies as an effective fair-value hedge, the Company continues to carry the derivative on the Consolidated Balance Sheet at its fair value and no longer adjusts the hedged asset or liability for changes in fair value. The adjustment of the carrying amount of the hedged asset or liability is accounted for in the same manner as other components of the carrying amount of that asset or liability. In all other situations in which hedge accounting is discontinued, the Company continues to carry the derivative at its fair value on the Consolidated Balance Sheet and recognizes any changes in its fair value in earnings.

109

Hedging Instruments 206

# **Recent Accounting Pronouncements**

In December 2007, the FASB issued Statement of Financial Accounting Standards No. 141 (revised 2007), Business Combinations (SFAS 141R). SFAS 141R changes accounting for acquisitions that close beginning in 2009 in a number of areas including the treatment of contingent consideration, contingencies, acquisition costs, In-process research & development and restructuring costs. More transactions and events will qualify as business combinations and will be accounted for at fair value under the new standard. SFAS 141R promotes greater use of fair values in financial reporting. In addition, under SFAS 141R, changes in deferred tax asset valuation allowances and acquired income tax uncertainties in a business combination after the measurement period will impact income tax expense. Some of the changes will introduce more volatility into earnings. SFAS 141R is effective for fiscal years beginning on or after December 15, 2008. SFAS 141R will have an impact on accounting for any business acquired after the effective date of this pronouncement.

In December 2007, the FASB issued Statement of Financial Accounting Standards No. 160, Noncontrolling Interests in Consolidated Financial Statements, an amendment of ARB No. 51 (SFAS 160). SFAS 160 will change the accounting and reporting for minority interests, which will be recharacterized as noncontrolling interests (NCI) and classified as a component of equity. This new consolidation method will significantly change the accounting for transactions with minority interest holders. SFAS 160 is effective for fiscal years beginning after December 15, 2008. SFAS 160 would have an impact on the presentation and disclosure of the noncontrolling interests of any non-wholly owned business acquired in the future.

In April 2009, the FASB issued FASB Staff Position No. 141(R)-1 (FSP FAS 141(R)-1) which provides additional clarification on the initial recognition and measurement of assets acquired and liabilities assumed in a business combination that arise from contingencies. FSP FAS 141(R)-1 is effective for all fiscal years beginning on or after December 15, 2008. FSP FAS 141(R)-1 will have an impact on the accounting for any business acquired in the future.

In May 2009, the FASB issued SFAS No. 165, Subsequent Events (SFAS No. 165). The statement is to establish general standards of accounting for and disclosure of events that occur after the balance sheet date but before financial statements are issued. SFAS No. 165 became effective June 15, 2009 for all subsequent reporting periods. The adoption of SFAS No. 165 did not have a material impact on the Company s combined financial statements.

In June 2009, the FASB issued SFAS No. 167, Amendments to FASB Interpretation No. 46(R) (SFAS 167). SFAS 167 eliminates Interpretation 46(R) is exceptions to consolidating qualifying special-purpose entities, contains new criteria for determining the primary beneficiary, and increases the frequency of required reassessments to determine whether a company is the primary beneficiary of a variable interest entity. SFAS 167 also contains a new requirement that any term, transaction, or arrangement that does not have a substantive effect on an entity is status as a variable interest entity, a company is power over a variable interest entity, or a company is obligation to absorb losses or its right to receive benefits of an entity must be disregarded in applying Interpretation 46(R) is provisions. The elimination of the qualifying special-purpose entity concept and its consolidation exceptions means more entities will be subject to consolidation assessments and reassessments. SFAS 167 will be effective January 1, 2010. The adoption of SFAS 167 is not expected to have an impact on the Company is combined financial statements or results of operations.

In July 2009, the FASB issued SFAS No. 168, The FASB Accounting Standards Codification and the Hierarchy of Generally Accepted Accounting Principles a replacement of FASB Statement No. 162 (SFAS 168). SFAS 168 establishes the FASB Accounting Standards Codification to become the source of authoritative U.S. GAAP recognized by the FASB to be applied by non-governmental entities. Rules and interpretive releases of the Securities and Exchange Commission (SEC) under authority of federal securities laws are also sources of authoritative GAAP

for SEC registrants. On the effective date of this Statement, the Codification will supercede all then-existing non-SEC accounting and reporting standards. All other nongrandfathered non-SEC accounting literature not included in the Codification will become nonauthoritative. SFAS 168, which changes the referencing of financial standards, is effective for interim or annual financial periods ending after September 15, 2009. The adoption of SFAS 168 is not expected to materially impact the Company s combined financial statements or results of operations.

Management does not believe that any other recently issued, but not yet effective, accounting standards if currently adopted would have a material effect on the combined financial statements.

# **Results of Operations**

As of the date of this proxy statement/prospectus, Cullen Agritech has not yet commenced any significant operations. Cullen Agritech is not aware of any material trends or uncertainties, other than national economic conditions that affect agricultural markets, that may reasonably be expected to have a material impact, favorable or unfavorable, on revenues or income from Cullen Agritech s agricultural products, other then those mentioned in this statement/prospectus.

# **Liquidity and Capital Resources**

Liquidity is a measure of Cullen Agritech s ability to meet potential cash requirements, including ongoing commitments to repay borrowings, fund and maintain Cullen Agritech s assets and operations, and other general business needs. Cullen Agritech will need significant cash to purchase their target assets, repay principal and interest on Cullen Agritech s borrowings, and fund Cullen Agritech s operations. Cullen Agritech s primary source of cash will consist of the funds to be released to it from Triplecrown s trust account upon consummation of the merger, cash generated from Cullen Agritech s operations, and proceeds that may be received from the exercise of Triplecrown s warrants. Should the merger not be consummated, Cullen Agritech would be dependant on its parent company, Cullen Inc. Holding, Ltd., for funding its operations during the development stage.

# **Contractual Obligations and Commitments**

On June 27, 2009, Cullen Agritech and Triplecrown entered into an agreement with a third party to purchase a certain parcel of land (the Property ). The total purchase price of the Property is \$8,662,500. Triplecrown paid an initial deposit of \$866,250 (the Deposit ) on the land. On August 10, 2009, Cullen Agritech, Triplecrown, and the seller of the land entered into an extension agreement (the Extension ) to extend the closing date for the land to September 15, 2009. In exchange for the Extension, Triplecrown paid an additional deposit on the land of \$833,750, interest of \$48,070 and a leasing fee for Cullen Agritech to use the land of \$3,518, for a total of \$885,338 (the Extension Payment ). On August 10, 2009, Cullen Agritech entered into an agreement with Triplecrown in which Cullen Agritech, and Eric J. Watson, agreed to be jointly and severally liable to assume the obligation for the remaining purchase price of the Property of \$6,962,500 if the merger is not consumated.

On August 11, 2009, Cullen Agritech entered into a strategic cooperation agreement with NZ Agritech to promote the interests of NZ Agritech and its members. Cullen Agritech will assist members of NZ Agritech to mitigate barriers of market entry and provide the opportunity to realize the potential growth in various markets. NZ Agritech will actively promote to its members its alliance with Cullen Agritech. Cullen Agritech is obligated to pay a fee to NZ Agritech annually, in arrears, based on its dealings with NZ Agritech s members. The fee is to be negotiated on a case by case basis

On August 14, 2009, Cullen Agritech entered into an agreement with AHA Advisors Ltd to act as a member of Cullen Agritech s advisory board and to work exclusively with Cullen Agritech in the development of a genetically advanced dairy cow milking herd for Natural Dairy Inc. AHA Advisors Ltd will assist in the discovery of a herd uniquely capable of meeting the production and performance goals of Cullen Agritech.

Results of Operations 209

# **Off Balance Sheet Arrangements**

Cullen Agritech does not currently have, nor has it had in the last three years, any relationships with unconsolidated entities or financial partnerships, such as entities referred to as structured finance or special purpose entities, which would have been established for the purpose of facilitating off-balance sheet arrangements or other contractually narrow or limited purposes.

# Quantitative and Qualitative Disclosures About Market Risk Commodity Risk

In the ordinary course of business, we purchase and sell commodities which are subject to market risk resulting from changes in prices. We monitor our positions for all commodities and utilize derivative instruments, primarily contracts, offered through regulated commodity exchanges to reduce exposure to changes in

#### TABLE OF CONTENTS

commodity prices. Certain commodities cannot be hedged with futures or option contracts because such contracts are not offered for these commodities by regulated commodity exchanges.

Inventories and purchase contracts for those commodities are hedged with forward contracts to the extent practical so as to arrive at a net commodity position within the formal position limits set by us and deemed prudent for each of those commodities. Commodities for which futures and options contracts are available are also typically hedged first in this manner, with futures and options used to hedge within position limits that portion not covered by forward contracts.

Cullen Agritech enters into futures and options contract derivatives to reduce risk on the market value of inventory and fixed or partially fixed purchase and sale contracts. The notional or contractual amount of derivatives provides an indication of the extent of the Company s involvement in such instruments at that time but does not represent exposure to market risk or future cash requirements under certain of these instruments.

#### **Interest Rate Risk**

Cullen Agritech can be exposed to market risk from fluctuations in interest rates. Due to the fact that commercial bank accounts, which Cullen Agritech may utilize after the consummation of the merger, are available for withdrawals on a daily basis and traditionally the investments are of a short term duration, an immediate 10% change in interest rates would not have a material effect on the fair market value of Cullen Agritech s accounts. Therefore, Cullen Agritech would not expect operating results or cash flows to be affected to any significant degree by the effect of a sudden change in market interest rates on Cullen Agritech s accounts.

#### **Inflation Risk**

Inflation is not expected to have a significant impact on our business, financial condition or results of operations. Cullen Agritech generally will be able to offset the impact of inflation through a combination of productivity improvements and price increases.

112

Commodity Risk 211

# MANAGEMENT OF CAH FOLLOWING THE MERGER

# General

Pursuant to the merger, (i) Triplecrown will merge with and into CAH with CAH surviving the merger and becoming the new publicly-traded corporation of which the present holders of Triplecrown securities will be security holders and (ii) Triplecrown Merger Sub will merge with and into Cullen Agritech with Cullen Agritech surviving the merger and becoming a subsidiary of CAH.

Upon consummation of the merger, CAH s board of directors will consist of seven directors, four of whom have been designated by Cullen Agritech, three of whom have been designated by Triplecrown. The directors of CAH will be classified as follows:

Class A Directors: and;
Class B Directors: and; and
Class C Directors: Eric J. Watson, and.

The following sets forth certain information concerning the persons who are expected to serve as CAH s directors and executive officers following the consummation of the merger:

Name Age Position with CAH

Eric J. Watson 49

# Information About the Directors and Executive Officers

Eric J. Watson has been Triplecrown s chairman of the board and treasurer since its inception and has also been the chief executive officer of Cullen Agritech since its inception in June 2009 and chief executive officer of CAH since its inception in August 2009. See Other Information Related to Triplecrown Directors and Executive Officers for complete biographical information of Mr. Watson.

# **Independence of Directors**

As a result of its securities being listed on the NYSE Amex, Triplecrown adheres to the rules of that exchange in determining whether a director is independent. Upon the consummation of the merger, it is anticipated that CAH s securities will be listed on either the New York Stock Exchange or NASDAQ. As a result, the board of directors of CAH will consult with its counsel to ensure that the board s determinations are consistent with those rules and all relevant securities and other laws and regulations regarding the independence of directors. The New York Stock Exchange and NASDAQ each requires that a majority of the board must be composed of independent directors, which is defined generally as a person other than an officer or employee of the company or its subsidiaries or any other individual having a relationship, which, in the opinion of the company s board of directors would interfere with the director s exercise of independent judgment in carrying out the responsibilities of a director. Consistent with these considerations, both the board of directors of Triplecrown and CAH have affirmatively determined that, upon the closing of the merger, , , and will be the independent directors of CAH.

# **CAH Board Committees**

Upon consummation of the merger, CAH s board of directors will form an audit committee, a compensation committee and a nominating and corporate governance committee and adopt charters for each of these committees. Each of these committees will have three directors and will be composed exclusively of independent directors, as defined by the listing standards of the New York Stock Exchange and NASDAQ. Moreover, the compensation committee will be composed exclusively of individuals intended to be, to the extent required by Rule 16b-3 of the Exchange Act, non-employee directors and will, at such times as CAH is subject to Section 162(m) of the Internal Revenue Code, qualify as outside directors for purposes of Section 162(m) of the Internal Revenue Code.

# **Audit Committee Information**

Effective October 2007, Triplecrown established an audit committee of the board of directors, which currently consists of Edward J. Mathias, as chairman, Jay H. Nussbaum and Richard Y. Roberts, each of

113

CAH Board Committees 213

whom is an independent director under the NYSE Amex listing standards. The audit committee has met four times in 2008 and three times in 2009. Upon the consummation of the merger, CAH will establish an audit committee of the board of directors and adopt a charter having material provisions described below. The initial members of CAH s audit committee will be, and, with serving as chairman. Each is an independent director under the New York Stock Exchange and NASDAQ listing standards. The audit committee will be responsible for engaging independent certified public accountants, preparing audit committee reports, reviewing with the independent certified public accountants the plans and results of the audit engagement, approving professional services provided by the independent certified public accountants, reviewing the independence of the independent certified public accountants, considering the range of audit and non-audit fees and reviewing the adequacy of CAH s internal accounting controls.

# **Financial Experts on Audit Committee**

The audit committee will at all times be composed exclusively of independent directors who are financially literate as defined under New York Stock Exchange or NASDAQ listing standards. The definition of financially literate generally means being able to read and understand fundamental financial statements, including a company s balance sheet, income statement and cash flow statement.

In addition, a listed company must certify to the exchange that the committee will have at least one member who has past employment experience in finance or accounting, requisite professional certification in accounting, or other comparable experience or background that results in the individual s financial sophistication. The boards of directors of both Triplecrown and CAH have determined that satisfies the definition of financial sophistication and also qualifies as an audit committee financial expert, as defined under rules and regulations of the SEC.

# **Nominating and Corporate Governance Committee**

Effective October 2007, Triplecrown established a nominating committee of the board of directors, which consists of Kerry Kennedy, as chairman, and Jimmie Lee Solomon, Jr., each of whom is an independent director under the NASDAQ listing standards. Upon consummation of the merger, the nominating and corporate governance committee will consist of and, each of whom will be an independent director. will chair CAH is nominating and corporate governance committee will be responsible for seeking, considering and recommending to the board qualified candidates for election as directors and will approve and recommend to the full board of directors the appointment of each of CAH is executive officers. It also will periodically prepare and submit to the board of directors for adoption the committee is selection criteria for director nominees. It will review and make recommendations on matters involving the general operation of the board and CAH is corporate governance, and will annually recommend to the board nominees for each committee of the board. In addition, the committee will annually facilitate the assessment of the board of directors performance as a whole and of the individual directors and report thereon to the board.

# **Compensation Committee**

The compensation committee will consist of , and , each of whom will be an independent director. will chair CAH s compensation committee. The principal functions of the compensation committee will be to:

evaluate the performance of CAH s officers, review any compensation payable to CAH s directors and officers, prepare compensation committee reports, and

administer the issuance of any common stock or other equity awards issued to CAH s officers and directors.

# **CAH Director Compensation**

The policies of CAH with respect to the compensation of its executive officers and following the merger will be administered by CAH s board in consultation with its compensation committee (as described above). The compensation policies followed by CAH will be intended to provide for compensation that is sufficient to

attract, motivate and retain executives of outstanding ability and potential and to establish an appropriate relationship between executive compensation and the creation of shareholder value. To meet these goals, the compensation committee will be charged with recommending executive compensation packages to CAH s board of directors.

It is anticipated that performance-based and equity-based compensation will be an important foundation in executive compensation packages as CAH and Cullen Agritech believe it is important to maintain a strong link between executive incentives and the creation of shareholder value. CAH and Cullen Agritech believe that performance and equity-based compensation can be an important component of the total executive compensation package for maximizing shareholder value while, at the same time, attracting, motivating and retaining high-quality executives. The adoption of the proposed incentive compensation plan and the management incentive compensation plan reflect and will reflect what CAH and Cullen Agritech believe is a focus on performance- and equity-based compensation. Since CAH will not have a compensation committee until completion of the merger, it has not yet adopted any formal guidelines for allocating total compensation between equity compensation and cash compensation for executives hired in the future.

# **Compensation Discussion and Analysis**

Upon consummation of the merger, CAH will seek to provide total compensation packages that are competitive in terms of potential value to its executives, and which are tailored to the unique characteristics and needs of CAH within its industry in order to create an executive compensation program that will adequately reward its executives for their roles in creating value for CAH stockholders. CAH intends to be competitive with other similarly situated companies in its industry following completion of the merger.

The compensation decisions regarding CAH s executives will be based on CAH s need to attract individuals with the skills necessary for CAH to achieve its business plan, to reward those individuals fairly over time, and to retain those individuals who continue to perform at or above CAH s expectations.

It is anticipated that CAH s executives compensation will have three primary components salary, cash incentive bonus and stock-based awards. CAH will view the three components of executive compensation as related but distinct. Although CAH s compensation committee will review total compensation, CAH does not believe that significant compensation derived from one component of compensation should negate or reduce compensation from other components. CAH anticipates determining the appropriate level for each compensation component based in part, but not exclusively, on its view of internal equity and consistency, individual performance and other information deemed relevant and timely. Since CAH s compensation committee will not be formed until consummation of the merger, CAH has not adopted any formal or informal policies or guidelines for allocating compensation between long-term and currently paid out compensation, between cash and non-cash compensation, or among different forms of compensation.

In addition to the guidance provided by its compensation committee, CAH may utilize the services of third parties from time to time in connection with the hiring and compensation awarded to executive employees. This could include subscriptions to executive compensation surveys and other databases.

CAH s compensation committee will be charged with performing an annual review of CAH s executive officers cash compensation and equity holdings to determine whether they provide adequate incentives and motivation to executive officers and whether they adequately compensate the executive officers relative to comparable officers in other companies.

## **Benchmarking of Cash and Equity Compensation**

CAH and Cullen Agritech both believe it is important when making compensation-related decisions to be informed as to current practices of similarly situated publicly held companies in the similar field. It is expected that the compensation committee will stay apprised of the cash and equity compensation practices of publicly held companies in the precision steel production and related industries through the review of such companies public reports and through other resources. It is also expected that any companies chosen for inclusion in any benchmarking group would have business characteristics comparable to CAH, including revenues, financial growth metrics, stage of development, employee headcount and market capitalization. While benchmarking may not always be appropriate as a stand-alone tool for setting compensation due to the

aspects of CAH post-reorganization business and objectives that may be unique to CAH, CAH generally believes that gathering this information will be an important part of its compensation-related decision-making process.

## **Compensation Components**

Base Salary. Generally, CAH, working with the compensation committee, anticipates setting executive base salaries at levels comparable with those of executives in similar positions and with similar responsibilities at comparable companies. CAH will seek to maintain base salary amounts at or near the industry norms, while avoiding paying amounts in excess of what it believes is necessary to motivate executives to meet corporate goals. It is anticipated base salaries will generally be reviewed annually, subject to terms of employment agreements, and that the compensation committee and board will seek to adjust base salary amounts to realign such salaries with industry norms after taking into account individual responsibilities, performance and experience.

Annual Bonuses. CAH intends to utilize cash incentive bonuses for executives to focus them on achieving key operational and financial objectives within a yearly time horizon. Near the beginning of each year, the board, upon the recommendation of the compensation committee and subject to any applicable employment agreements, will determine performance parameters for appropriate executives. At the end of each year, the board and compensation committee will determine the level of achievement for each corporate goal.

Equity Awards. CAH also will use share options and other share-based awards to reward long-term performance. It believes that providing a meaningful portion of its executives—total compensation package in share options and other share-based awards will align the incentives of its executives with the interests of CAH—s shareholders and with CAH—s long-term success. The compensation committee and board will develop their equity award determinations based on their judgments as to whether the complete compensation packages provided to CAH—s executives, including prior equity awards, are sufficient to retain, motivate and adequately award the executives.

Equity awards will be granted through CAH s incentive compensation plan. All of CAH s employees, directors, officers and consultants will be eligible to participate in the incentive compensation plan and all CAH s senior management will be eligible to participate in the management incentive compensation plan.

CAH will account for any equity compensation expense under the rules of SFAS 123R, which requires a company to estimate and record an expense for each award of equity compensation over the service period of the award. Accounting rules also will require CAH to record cash compensation as an expense at the time the obligation is accrued.

Severance Benefit. CAH currently has no severance benefits plan. CAH may consider the adoption of a severance plan for executive officers and other employee in the future.

Other Compensation. CAH will establish and maintain various employee benefit plans, including medical and retirement insurance plans. These plans will be available to all qualified employees.

Director and Consultant Compensation. CAH currently does not have a definitive compensation plan for its future directors or consultants. CAH, working with the compensation committee, anticipates setting director and consultant compensation at a level comparable with those directors and consultants with similar positions at comparable companies. It is currently anticipated that such compensation will be based on cash and/or equity compensation under CAH s incentive compensation plan and management incentive compensation plan.

## **CAH 2009 Long-Term Incentive Equity Plan**

CAH s 2009 Long-Term Incentive Equity Plan is designed to enable CAH to offer its employees, officers, directors and consultants whose past, present and/or potential contributions to CAH have been, are or will be important to the success of CAH, an opportunity to acquire a proprietary interest in CAH. The various types of incentive awards that may be provided under the plan are intended to enable CAH to respond to changes in compensation practices, tax laws, accounting regulations and the size and diversity of its business. The plan reserves a number of shares of CAH common stock for issuance in accordance with the plan s terms equal to

12.5% of the number of shares to be outstanding after the merger, after giving effect to purchases of Public Shares and holders exercising conversion rights.

All officers, directors and employees of CAH, Triplecrown and Cullen Agritech will be eligible to be granted awards under the plan. An incentive stock option may be granted under the plan only to a person who, at the time of the grant, is an employee of CAH, Triplecrown or Cullen Agritech or a related corporation. No allocations of shares that may be subject to awards have been made in respect of the executive officers or any other group. All awards will be subject to the recommendations of the compensation committee and approval by the board of directors or the compensation committee.

A summary of the principal features of the plan is provided below, but is qualified in its entirety by reference to the full text of the plan, which is attached to this proxy statement/prospectus as Annex H.

## **Administration**

The plan is administered by the board of directors of CAH or its compensation committee. Subject to the provisions of the plan, the committee determines, among other things, the persons to whom from time to time awards may be granted, the specific type of awards to be granted, the number of shares subject to each award, share prices, any restrictions or limitations on the awards, and any vesting, exchange, deferral, surrender, cancellation, acceleration, termination, exercise or forfeiture provisions related to the awards.

# Stock Subject to the Plan

Shares of stock subject to other awards that are forfeited or terminated will be available for future award grants under the plan. If a holder pays the exercise price of a stock option by surrendering any previously owned shares of common stock or arranges to have the appropriate number of shares otherwise issuable upon exercise withheld to cover the withholding tax liability associated with the stock option exercise, then the in the board s or committee s discretion, the number of shares available under the plan may be increased by the lesser of the number of such surrendered shares and shares used to pay taxes and the number of shares purchased under the stock option.

Under the plan, on a change in the number of shares of common stock as a result of a dividend on shares of common stock payable in shares of common stock, common stock forward split or reverse split or other extraordinary or unusual event that results in a change in the shares of common stock as a whole, the terms of the outstanding award will be proportionately adjusted.

# **Eligibility**

Awards may be granted under the plan to employees, officers, directors and consultants who are deemed to have rendered, or to be able to render, significant services to CAH and who are deemed to have contributed, or to have the potential to contribute, to our success.

# **Types of Awards**

Options. The plan provides both for incentive stock options as defined in Section 422 of the Code, and for options not qualifying as incentive options, both of which may be granted with any other stock based award under the plan. The board or committee determines the exercise price per share of common stock purchasable under an incentive or non-qualified stock option, which may not be less than 100% of the fair market value on the day of the grant or, if

Administration 220

greater, the par value of a share of common stock. However, the exercise price of an incentive stock option granted to a person possessing more than 10% of the total combined voting power of all classes of stock may not be less than 110% of the fair market value on the date of grant. The aggregate fair market value of all shares of common stock with respect to which incentive stock options are exercisable by a participant for the first time during any calendar year (under all of our plans), measured at the date of the grant, may not exceed \$100,000 or such other amount as may be subsequently specified under the Code or the regulations thereunder.

An incentive stock option may only be granted within a ten-year period from the date of the consummation of the merger and may only be exercised within ten years from the date of the grant, or within five years in the case of an incentive stock option granted to a person who, at the time of the grant, owns common stock possessing more than 10% of the total combined voting power of all classes of our stock. Subject to any

117

## **TABLE OF CONTENTS**

limitations or conditions the board or committee may impose, stock options may be exercised, in whole or in part, at any time during the term of the stock option by giving written notice of exercise to us specifying the number of shares of common stock to be purchased. The notice must be accompanied by payment in full of the purchase price, either in cash or, if provided in the agreement, in our securities or in combination of the two.

Generally, stock options granted under the plan may not be transferred other than by will or by the laws of descent and distribution and all stock options are exercisable during the holder s lifetime, or in the event of legal incapacity or incompetency, the holder s guardian or legal representative. However, a holder, with the approval of the board or committee, may transfer a non-qualified stock option by gift to a family member of the holder, by domestic relations order to a family member of the holder or by transfer to an entity in which more than 50% of the voting interests are owned by family members of the holder or the holder, in exchange for an interest in that entity.

Generally, if the holder is an employee, no stock options granted under the plan may be exercised by the holder unless he or she is employed by CAH or a subsidiary of its at the time of the exercise and has been so employed continuously from the time the stock options were granted. However, in the event the holder s employment is terminated due to disability, the holder may still exercise his or her vested stock options for a period of 12 months or such other greater or lesser period as the board or committee may determine, from the date of termination or until the expiration of the stated term of the stock option, whichever period is shorter. Similarly, should a holder die while employed by CAH or a subsidiary, his or her legal representative or legatee under his or her will may exercise the decedent holder s vested stock options for a period of 12 months from the date of his or her death, or such other greater or lesser period as the board or committee may determine or until the expiration of the stated term of the stock option, whichever period is shorter. If the holder s employment is terminated due to normal retirement, the holder may still exercise his or her vested stock options for a period of 12 months from the date of termination or until the expiration of the stated term of the stock option, whichever period is shorter. If the holder s employment is terminated for any reason other than death, disability or normal retirement, the stock option will automatically terminate, except that if the holder s employment is terminated without cause, then the portion of any stock option that is vested on the date of termination may be exercised for the lesser of three months after termination of employment, or such other greater or lesser period as the board or committee may determine but not beyond the balance of the stock option s term.

Stock Appreciation Rights. Under the plan, stock appreciation rights may be granted to participants who have been, or are being, granted stock options under the plan as a means of allowing the participants to exercise their stock options without the need to pay the exercise price in cash. In conjunction with non-qualified stock options, stock appreciation rights may be granted either at or after the time of the grant of the non-qualified stock options. In conjunction with incentive stock options, stock appreciation rights may be granted only at the time of the grant of the incentive stock options. A stock appreciation right entitles the holder to receive a number of shares of common stock having a fair market value equal to the excess fair market value of one share of common stock over the exercise price of the related stock option, multiplied by the number of shares subject to the stock appreciation rights. The granting of a stock appreciation right will not affect the number of shares of common stock available for awards under the plan. The number of shares available for awards under the plan will, however, be reduced by the number of shares of common stock acquirable upon exercise of the stock option to which the stock appreciation right relates.

Restricted Stock. Under the plan, shares of restricted stock may be awarded either alone or in addition to other awards granted under the plan. The board or committee determines the persons to whom grants of restricted stock are made, the number of shares to be awarded, the price if any to be paid for the restricted stock by the person receiving the stock from us, the time or times within which awards of restricted stock may be subject to forfeiture, the vesting schedule and rights to acceleration thereof, and all other terms and conditions of the restricted stock awards. Each share that is the subject of a restricted stock award will be deemed to be shares for purposes of calculating the number of shares remaining available for awards from the shares authorized under the plan.

## **TABLE OF CONTENTS**

Restricted stock awarded under the plan may not be sold, exchanged, assigned, transferred, pledged, encumbered or otherwise disposed of, other than to us, during the applicable restriction period. In order to enforce these restrictions, the plan requires that all shares of restricted stock awarded to the holder remain in our physical custody until the restrictions have terminated and all vesting requirements with respect to the restricted stock have been fulfilled. Other than regular cash dividends and other cash equivalent distributions as we may designate, pay or distribute, we will retain custody of all distributions made or declared with respect to the restricted stock during the restriction period. A breach of any restriction regarding the restricted stock will cause a forfeiture of the restricted stock and any retained distributions. Except for the foregoing restrictions, the holder will, even during the restriction period, have all of the rights of a stockholder, including the right to receive and retain all regular cash dividends and other cash equivalent distributions as we may designate, pay or distribute on the restricted stock and the right to vote the shares.

Other Stock-Based Awards. Under the plan, other stock-based awards may be granted, subject to limitations under applicable law, that are denominated or payable in, valued in whole or in part by reference to, or otherwise based on, or related to, shares of common stock, as deemed consistent with the purposes of the plan. These other stock-based awards may be in the form of purchase rights, shares of common stock awarded that are not subject to any restrictions or conditions, convertible or exchangeable debentures or other rights convertible into shares of common stock and awards valued by reference to the value of securities of, or the performance of, one of CAH s subsidiaries. These other stock-based awards may include performance shares or options, whose award is tied to specific performance criteria. These other stock-based awards may be awarded either alone, in addition to, or in tandem with any other awards under the plan or any of our other plans.

Accelerated Vesting and Exercisability. If any one person, or more than one person acting as a group, acquires the ownership of stock of the company that, together with the stock held by such person or group, constitutes more than 50% of the total fair market value or combined voting power of the stock of CAH, and its board of directors does not authorize or otherwise approve such acquisition, then the vesting periods of any and all stock options and other awards granted and outstanding under the plan shall be accelerated and all such stock options and awards will immediately and entirely vest, and the respective holders thereof will have the immediate right to purchase and/or receive any and all common stock subject to such stock options and awards on the terms set forth in the plan and the respective agreements respecting such stock options and awards. An increase in the percentage of stock owned by any one person, or persons acting as a group, as a result of a transaction in which the company acquires its stock in exchange for property is not treated as an acquisition of stock.

The committee may, in the event of an acquisition by any one person, or more than one person acting as a group, together with acquisitions during the 12-month period ending on the date of the most recent acquisition by such person or persons, of assets from the company that have a total gross fair market value equal to or more than 50% of the total gross fair market value of all of the assets of CAH immediately before such acquisition or acquisitions, or if any one person, or more than one person acting as a group, acquires the ownership of stock of the company that, together with the stock held by such person or group, constitutes more than 50% of the total fair market value or combined voting power of the stock of the company, which has been approved by its board of directors, (i) accelerate the vesting of any and all stock options and other awards granted and outstanding under the plan, or (ii) require a holder of any award granted under the plan to relinquish such award to the company upon the tender by the company to the holder of cash in an amount equal to the repurchase value of such award. For this purpose, gross fair market value means the value of the assets of the company, or the value of the assets being disposed of, determined without regard to any liabilities associated with such assets.

Notwithstanding any provisions of the plan or any award granted thereunder to the contrary, no acceleration shall occur with respect to any award to the extent such acceleration would cause the plan or an award granted thereunder to fail to comply with Section 409A of the Code.

Repurchases. Unless otherwise provided in the grant of an award, the board or committee may, in the event of a corporate transaction that has been approved by CAH s board of directors, require a holder of any

119

award granted under the plan to relinquish the award to us upon payment by us to the holder of cash in an amount equal to the fair market value of the award.

Award Limitation. No participant may be granted awards for more than shares in any calendar year.

Other Limitations. The board or committee may not modify or amend any outstanding option or stock appreciation right to reduce the exercise price of such option or stock appreciation right, as applicable, below the exercise price as of the date of grant of such option or stock appreciation right. In addition, no option or stock appreciation right may be granted in exchange for, or in connection with, the cancellation or surrender of an option or stock appreciation right or other award having a higher exercise price.

# **Compensation Committee Interlocks and Insider Participation**

None of the persons designated as directors of CAH currently serves on the compensation committee of any other company on which any other director designee of CAH or any officer or director of Triplecrown or CAH is currently a member.

# COMPARISON OF RIGHTS OF TRIPLECROWN AND CAH

The rights of Triplecrown s stockholders are currently governed by the Delaware General Corporation Law and Triplecrown s amended and restated certificate of incorporation and bylaws. Under the merger agreement, at the closing of the merger, the stockholders of Triplecrown will be entitled to receive shares of CAH common stock. Accordingly, after the merger, the rights of any former stockholder of Triplecrown who receives shares of CAH common stock will be governed by CAH s certificate of incorporation and bylaws.

The following discussion identifies material differences between current rights of Triplecrown stockholders and those of the stockholders of CAH following the transactions. The following discussions are summaries only. They do not give you a complete description of the differences that may affect you. You should also refer to the Delaware General Corporation Law, as well as CAH s certificate of incorporation and byalws and Triplecrown s amended and restated certificate of incorporation, copies of which are annexed hereto as Annexes B, C and G, respectively. For a more detailed discussion of your rights as stockholders of CAH, you should also see *Description of Triplecrown and CAH Securities*.

Authorized Capital Stock	Current Triplecrown Stockholder Rights The authorized capital stock of Triplecrown currently consists of 160,000,000 shares of common stock, par value \$0.0001 per share, and 1,000,000 shares of preferred stock, par value \$0.0001 per share. Each share of common stock is entitled	CAH Stockholder Rights  The authorized capital stock of CAH will consist of 200,000,000 shares of common stock, par value \$.0001 per share, and 1,000,000 preferred shares, par value \$0.0001 per share.
Voting Rights	to one vote on all matters before the stockholders of Triplecrown.  Triplecrown shall submit any business combination to its stockholders for approval regardless of whether the business combination is of a type which normally would require such stockholder approval under the	Same.
	Delaware General Corporation Law. Triplecrown shall not consummate any business combination if holders of 30% or more of the shares of common stock issued in Triplecrown s initial public offering exercise their conversion rights as described below.	No comparable provision.
Conversion Rights	Holders of common stock issued in Triplecrown s initial public offering, other than Triplecrown s initial stockholders, who vote against a	No comparable provision.

business combination may demand Triplecrown convert their shares of

common stock into cash.

If a business combination is not consummated prior to the termination

Liquidation if No **Business Combination**  date set forth in Triplecrown s amended No comparable provision.

and certificate of incorporation, Triplecrown shall liquidate and the

holders of common stock issued in

Current Triplecrown CAH
Stockholder Rights Stockholder Rights
Triplecrown s initial public offering shall
receive a pro rata distribution from
Triplecrown s trust account.

# BENEFICIAL OWNERSHIP OF SECURITIES

The following table sets forth information regarding the beneficial ownership of Triplecrown s common stock as of, 2009 (Pre-Merger) and, immediately following consummation of the merger (Post-Merger), ownership of shares of CAH common stock, by:

each person known by Triplecrown to be the beneficial owner of more than 5% of Triplecrown s outstanding shares of common stock either on , 2009 (Pre-Merger) or of shares of CAH common stock outstanding after the consummation of the merger (Post-Merger);

each of Triplecrown s current executive officers and directors;
each person who will become an executive officer or director of CAH upon consummation of the merger;
all of Triplecrown s current executive officers and directors as a group; and
all of the executive officers and directors of CAH as a group after the consummation of the merger.
Information (Pre-Merger) does not reflect beneficial ownership of Triplecrown s outstanding warrants as these
warrants are not currently exercisable and will not become exercisable until consummation of the merger. Information
(Post-Merger) assumes that no Public Shares vote against the merger proposal and seek conversion and gives effect to
the forfeiture of shares by the Triplecrown Founders as described elsewhere in this proxy statement/prospectus.

At any time prior to the special meeting, during a period when they are not then aware of any material nonpublic information regarding Triplecrown or its securities, Triplecrown, the Triplecrown Founders, Cullen Agritech and their respective affiliates may enter into a written plan to purchase Triplecrown securities pursuant to Rule 10b5-1 of the Exchange Act, and may engage in other public market purchases, as well as private purchases, of securities at anytime prior to the special meeting of stockholders. The ownership percentages listed below do not include any such shares that may be purchased after , 2009.

Name and Address of Beneficial Owner <sup>(1)</sup>	Pre-Merger Amount and Nature of Beneficial Ownership		Post-Merger Amount and Nature of Beneficial Ownership	Percent of Class
Jonathan J. Ledecky	6,630,000 (2)	9.6 %	4,500,000 <sup>(2)(3)</sup>	
Eric J. Watson	6,630,000 (4)	9.6 %	18,381,148(3)(5)	
Jay H. Nussbaum	60,000	*	60,000 (6)	
Kerry Kennedy <sup>(7)</sup>	60,000	*	60,000 (6)	
Robert B. Hersov <sup>(8)</sup>	60,000	*	60,000 (6)	
Edward J. Mathias <sup>(9)</sup>	60,000	*	60,000 (6)	
Richard Y. Roberts <sup>(10)</sup>	60,000	*	60,000 (6)	
Jimmie Lee Solomon, Jr. (11)	60,000	*	60,000 (6)	
Jim Gray <sup>(12)</sup>	60,000	*	60,000 (6)	
Richard A. Stein <sup>(13)</sup>	60,000	*	60,000 (6)	
Edward Hanson <sup>(14)</sup>	60,000	*	60,000 (6)	
Fir Tree, Inc. (15)	6,341,200 (16)	9.2 %	6,341,200 (16)	
Israel A. Englander <sup>(17)</sup>	6,275,127 (18)	9.1 %	6,275,127 (18)	
QVT Financial LP <sup>(19)</sup>	6,520,108 (20)	9.5 %	6,520,108 (20)	
HBK Investments L.P. <sup>(21)</sup>	4,062,656 (22)	5.9 %	4,062,656 (22)	
(23)	0	0		
	0	0		
	0	0		
	0	0		
All Pre-Merger directors and executive officers as a group (11 individuals) All Post-Merger directors and executive officers as a group (individuals)	13,800,000	20.0 %	22,770,000(24)	

Less than 1%.

- Unless otherwise indicated, the business address of each of the individuals is 970 West Broadway, PMB 402, Jackson, Wyoming 83001.
- (2) Includes 600,000 Founders Units and underlying securities held by Hat Tricks LLC, an affiliate of Mr. Ledecky. Includes 2,500,000 shares of common stock issuable upon exercise of Sponsors Warrants that will become
- (3) exercisable upon consummation of the merger. Does not include 6,630,000 shares of common stock issuable upon exercise of Founders Warrants that will not become exercisable within 60 days of the consummation of the merger. (4) Represents shares of common stock held by Summit Trust, a trust established for the benefit of Mr. Watson and his beneficiaries.
  - (5)Represents shares of common stock held by Summit Trust and Cullen Holdings.
- (6) Does not include 60,000 shares of common stock issuable upon exercise of Founders Warrants that will not become exercisable within 60 days of the consummation of the merger.
- Ms. Kennedy s business address is c/o Robert F. Kennedy Center, 1367 Connecticut Avenue N.W., Suite 200, (7) Washington, D.C. 20036.
- Mr. Hersov s business address is NetJets Europe, Ltd., Grundstrasse 12, 6343 Rotkreuz, Switzerland. (9) Mr. Mathias business address is c/o The Carlyle Group, 1001 Pennsylvania Avenue, NW, Washington, DC 20004. (10) Mr. Roberts business address is Roberts, Raheb & Gradler, 805 15th Street, NW, Suite 1101, Washington, DC 20005.

(11) Mr. Solomon s address is P.O. Box 265, New York, New York 10163.
(12) Mr. Gray s business address is P.O. Box 774, Pacific Palisades, California 90272.

- (13) Mr. Stein s business address is 435 S. Washington Street, Falls Church, Virginia 22046.
- (14) Mr. Hanson s business address is c/o Babcock & Brown Limited, 53 Davies Street, London WIK 5JH.
- (15) The business address of Fir Tree, Inc. is 505 Fifth Avenue, 23rd Floor, New York, New York 10017. Represents 3,906,258 shares of common stock held by Fir Tree SPAC Holdings 1, LLC (SPAC Holdings 1), 1,641,138 shares of common stock held by Fir Tree SPAC Holdings 2, LLC (SPAC Holdings 2) and 793,804
- (16) shares of common stock held by Fir Tree Value Master Fund, L.P. Fir Tree, Inc. is the investment manager for each of these entities. The foregoing information was derived from a Schedule 13G/A filed with the SEC on February 11, 2009.
  - (17) Mr. Englander s business address is 666 Fifth Avenue, New York, New York 10103.

    Represents 4,030,806 shares of common stock held by Integrated Core Strategies (US) LLC ( ICS ) and 2,244,321 shares of common stock held by Millenco LLC ( Millenco ). Israel A. Englander is the managing member of Millenco and the general methors of Integrated Helding.
- (18) Millenium Management LLC, which is the manager of Millenco and the general partner of Integrated Holding Group LP ( IHG ). IHG is the managing member of ICS. Does not include 7,260,493 shares issuable on the exercise of warrants held by ICS and 646,200 shares issuable upon the exercise of warrants held by Millenco. The foregoing information was derived from a Schedule 13G/A filed with the SEC on February 12, 2009.
- The business address of QVT Financial LP is 1177 Avenue of the Americas, 9th Floor, New York, New York 10036.
  - (20) Represents 5,500,988 shares of common stock owned by QVT Fund LP, 587,613 shares of common stock owned by Quintessence Fund L.P. and 431,507 shares held for a third party account. QVT Financial LP is the investment manager of each of these funds. The foregoing information was derived a Schedule 13G/A filed with the SEC on February 6, 2009.
  - (21) The business address of HBK Investments L.P. is 300 Crescent Court, Suite 700, Dallas, Texas 75201.

    HBK Investments L.P. has delegated discretion to vote and dispose of its securities to HBK Services LLC

    ( Services ). Services may, from time to time, delegate discretion to vote and dispose of certain of the securities to HBK New York LLC, a Delaware limited liability company, HBK Virginia LLC, a Delaware limited liability
- (22) company, and/or HBK Europe Management LLP, a limited liability partnership organized under the laws of the United Kingdom (collectively, the Subadvisors). Each of Services and the Subadvisors is under common control with HBK Investments L.P. Does not include 75,000 shares of common stock issuable upon exercise of warrants that are not exercisable and will not become exercisable within 60 days. The foregoing information was derived from a Schedule 13G/A filed with the SEC on February 9, 2009.
- (23) The business address of such individual is 22 Barnett Shoals Road, Watkinsville, Georgia 30677. Includes 5,000,000 shares of common stock issuable upon exercise of Sponsors Warrants that will become exercisable upon consummation of the merger. Does not include 13,800,000 shares of common stock issuable
- upon exercise of Founders Warrants that will not become exercisable within 60 days of the consummation of the merger.

# CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS

# **Code of Ethics and Related Person Policy**

Triplecrown s Code of Ethics requires it to avoid, wherever possible, all related party transactions that could result in actual or potential conflicts of interest, except under guidelines approved by the board of directors (or the audit committee). Related-party transactions are defined under SEC rules as transactions in which (1) the aggregate amount involved will or may be expected to exceed the lesser of \$120,000 or one percent of the average of the company s total assets at year end for the last two completed years, (2) Triplecrown or any of its subsidiaries is a participant, and (3) any (a) executive officer, director or nominee for election as a director, (b) greater than 5 percent beneficial owner of Triplecrown s common stock, or (c) immediate family member, of the persons referred to in clauses (a) and (b), has or will have a direct or indirect material interest (other than solely as a result of being a director or a less than 10 percent beneficial owner of another entity). A conflict of interest situation can arise when a person takes actions or has interests that may make it difficult to perform his or her work objectively and effectively. Conflicts of interest may also arise if a person, or a member of his or her family, receives improper personal benefits as a result of his or her position. These procedures are intended to determine whether any such related party transaction impairs the independence of a director or presents a conflict of interest on the part of a director, employee or officer.

Upon the consummation of the merger, CAH s board of directors will establish a code of business conduct and ethics that applies to CAH s officers and directors. Among other matters, CAH s code of business conduct and ethics will be designed to detect and deter wrongdoing and to promote:

honest and ethical conduct, including the ethical handling of actual or apparent conflicts of interest between personal and professional relationships;

full, fair, accurate, timely and understandable disclosure in CAH s SEC reports and other public communications; compliance with applicable governmental laws, rules and regulations;

prompt internal reporting of violations of the code to appropriate persons identified in the code; and accountability for adherence to the code.

Any waiver of the code of business conduct and ethics for CAH s executive officers or directors may be made only by CAH s board of directors or one of CAH s board committees and will be promptly disclosed as required by law or stock exchange regulations.

# **Triplecrown Related Person Transactions**

#### Prior Issuances

In June 2007, Triplecornw issued 11,500,000 shares of common stock to the individuals and entities set forth below for \$25,000 in cash, at a purchase price of \$0.002 share. After giving retroactive effect to reflect the dividends described below, Triplecrown effectively issued 13,800,000 units to such individuals, as follows:

Stockholders

Summit Trust

Number of Shares

6,630,000

Edgar Filing: Triplecrown Acquisition Corp. - Form PREM14A

Jonathan J. Ledecky	6,030,000
Hat Tricks LLC	600,000
Kerry Kennedy	60,000
Robert B. Hersov	60,000
Edward J. Mathias	60,000
Richard Y. Roberts	60,000
Jimmie Lee Solomon, Jr.	60,000
Jay H. Nussbaum	60,000
Jim Gray	60,000
Richard A. Stein	60,000
Edward Hanson	60,000

125

Prior Issuances 234

On September 19, 2007, Triplecrown s board of directors authorized a dividend to the Founders of one warrant to purchase one share of common stock for each outstanding share of common stock. Each warrant entitles the registered holder to purchase one share of common stock at a price of \$7.50 per share and expires on October 21, 2012. On September 21, 2007, Mr. Hanson transferred his units to Coronet Group Limited, an entity he controls, at their initial price of approximately \$0.002 per share. On October 22, 2007, Triplecrown s board of directors authorized a dividend to the Founders of 0.2 units, each unit consisting of one share of common stock and one warrant to purchase one share of common stock, for each outstanding unit. Such securities were issued in connection with Triplecrown s organization pursuant to the exemption from registration contained in Section 4(2) of the Securities Act as they were sold to accredited individuals or entities.

## Insider Stockholder Escrow

All of the Founders Units have been placed in escrow with Continental Stock Transfer & Trust Company, as escrow agent, until one year after the consummation of a business combination, except as described below. The Founders Units may be released from escrow earlier than this date if, within the first year after Triplecrown consummates a business combination, (i) Triplecrown s units has a last sales price equal to or exceeding \$13.75 per unit for any 20 trading days within any 30-trading day period or (ii) Triplecrown consummates a subsequent liquidation, merger, stock exchange or other similar transaction which results in all of its stockholders having the right to exchange their shares of common stock for cash, securities or other property. In connection with the merger, the Triplecrown Founders have agreed to have cancelled an aggregate of 11,260,000 shares of common stock of CAH they will receive in exchange for 11,260,000 Founders Shares upon consummation of the merger. The remaining 2,540,000 shares, as well as all of the warrants of CAH that the Triplecrown Founders will receive in exchange for their Founders Warrants, will continue to be held in escrow pursuant to the original terms of the escrow agreement and will be released one year after the consummation of the merger.

# **Initial Stockholder Warrant Purchase**

In connection with the closing of the IPO, Triplecrown sold 2,500,000 Sponsors Warrants to each of Eric J. Watson and Jonathan J. Ledecky at a purchase price of \$1.00 per warrant. These purchases took place on a private placement basis simultaneously with the consummation of the IPO. The Sponsors Warrants are identical to the warrants underlying the units sold in the IPO except that the Sponsors Warrants are not transferable or salable by Messrs. Watson or Ledecky (except in certain limited circumstances such as to relatives and trusts for estate planning purposes, providing the transferee agrees to be bound by the transfer restrictions) until Triplecrown completes a business combination, and if Triplecrown calls the warrants for redemption, the Sponsors Warrants are not redeemable so long as such warrants are held by Messrs. Watson, Ledecky or their affiliates, including any permitted transferees.

# **Registration Rights**

The holders of the majority of the Founders Units and the holders of the majority the Sponsors Warrants (or underlying shares) each will be entitled to make up to two demands that Triplecrown register such units or warrants (or underlying shares) pursuant to a registration rights agreement entered into with Triplecrown in connection with the IPO. The holders of the majority of the Founders Units can elect to exercise these registration rights at any time commencing nine months after the consummation of a business combination. The holders of a majority of the Sponsors Warrants (or underlying shares) can elect to exercise these registration rights at any time after Triplecrown consummates the merger. In addition, these holders have certain piggy-back registration rights on registration statements filed subsequent to such date. Triplecrown will bear the expenses incurred in connection with the filing of

any such registration statements. As a result of the merger, the Founders Warrants and Sponsors Warrants will become warrants to purchase shares of CAH common stock and the obligations of Triplecrown will become similar obligations of CAH.

## **Other Transactions**

Prior to the IPO, Jonathan J. Ledecky and Eric J. Watson advanced an aggregate of \$112,500 to Triplecrown, on a non-interest bearing basis, for payment of offering expenses on its behalf. These amounts were repaid out of proceeds of the IPO.

In connection with the proposed business combination with CAH, Triplecrown and Natural Dairy entered into a contract to purchase a certain piece of land to be used by CAH following consummation of the merger.

126

Registration Rights 236

The total purchase price of the land is \$8,662,500 on which Triplecrown paid a deposit of approximately \$1.8 million. If Triplecrown is unable to complete the proposed business combination by the date it is required to liquidate, Mr. Watson and Natural Dairy have jointly and severally agreed to assume Triplecrown s remaining obligations under the purchase contract (approximately \$6.9 million) and Triplecrown will have no further obligations or liability whatsoever under the purchase contract.

Triplecrown reimburses its officers and directors for any reasonable out-of-pocket business expenses incurred by them in connection with certain activities on Triplecrown s behalf such as identifying and investigating possible target businesses and business combinations. There is no limit on the amount of out-of-pocket expenses reimbursable by Triplecrown, which are reviewed only by its board and audit committee or a court of competent jurisdiction if such reimbursement is challenged, provided that no proceeds held in the trust account will be used to reimburse out-of-pocket expenses prior to the merger.

Other than reimbursable out-of-pocket expenses payable to the officers and directors of Triplecrown, no compensation or fees of any kind, including finder s fees, consulting fees or other similar compensation, has been or will be paid to any of the Triplecrown Founders, including the officers and directors of Triplecrown, or to any of their respective affiliates, prior to or with respect to the business combination (regardless of the type of transaction that it is).

Triplecrown requires that all ongoing and future transactions between itself and any of its officers and directors or their respective affiliates, including loans by its officers and directors, will be on terms that Triplecrown believes to be no less favorable to it than are available from unaffiliated third parties. Such transactions or loans, including any forgiveness of loans, require prior approval by a majority of Triplecrown s uninterested independent directors or the members of its board who do not have an interest in the transaction, in either case who had access, at Triplecrown s expense, to Triplecrown s attorneys or independent legal counsel. Triplecrown will not enter into any such transaction unless its disinterested directors determine that the terms of such transaction are no less favorable to Triplecrown than those that would be available to Triplecrown with respect to such a transaction from unaffiliated third parties. CAH will continue such policy after the merger.

# **Cullen Agritech Related Person Transactions**

Cullen Agritech s Chief Executive Officer, Executive Chairman and Treasurer is a member of the Board of Directors of Cullen Holdings, Cullen Agritech s parent. As of August 18, 2009, Cullen Holdings has been funding the operations of both Cullen Agritech and Natural Dairy and is owed a combined total of \$35,484.

Eric J. Watson, Cullen Agritech s Chief Executive Officer, is the Chairman of the Board and Treasurer of Triplecrown. Natural Dairy and Triplecrown entered into an agreement to purchase land from a third party, as fully described above.

# Section 16(a) Beneficial Ownership Reporting Compliance

Section 16(a) of the Exchange Act requires Triplecrown s directors and officers and persons owning more than 10% of Triplecrown common stock to file reports of ownership and changes of ownership with the SEC. Based on Triplecrown s review of the copies of such reports furnished to it, or representations from certain reporting persons that no other reports were required, Triplecrown believes that all applicable filing requirements were complied with during the fiscal year ended December 31, 2008.

Other Transactions 237

# DESCRIPTION OF TRIPLECROWN S AND CAH S SECURITIES

## General

On October 25, 2007, Triplecrown closed its IPO of 55,200,000 units, including 7,200,000 units subject to the underwriters over-allotment option, with each unit consisting of one share of its common stock and one warrant, each to purchase one share of its common stock at an exercise price of \$7.50 per share. The units (including those subject to the over-allotment option) were sold at an offering price of \$10.00 per unit, generating total gross proceeds of \$552,000,000. Triplecrown s units, common stock and warrants are listed on the NYSE Amex under the symbols TCW.U, TCW and TCW.WS, respectively. The closing bid price for each unit, share of common stock, warrant of Triplecrown on September 3, 2009, the last trading day before announcement of the execution of the merger agreement, was \$, \$ and \$, respectively.

The amended and restated certificate of incorporation of Triplecrown authorizes the issuance of 160,000,000 shares of common stock, par value \$0.0001 per share, and 1,000,000 shares of preferred stock, par value \$0.0001 per share. As of the record date, 69,000,000 shares of common stock were outstanding and no shares of preferred stock were outstanding. CAH s certificate of incorporation authorizes it to issue up to 200,000,000 shares of common stock, par value \$0.0001 per share, and 1,000,000 shares of preferred stock, par value \$0.0001 per share.

# Triplecrown Common Stock and CAH Shares of Common Stock

Triplecrown s stockholders of record are entitled to one vote for each share held on all matters to be voted on by stockholders. In connection with the vote required for any business combination, all of the Triplecrown Founders, including all of its officers and directors, have agreed to vote the Founders Shares owned by them in accordance with the majority of the Public Shares and have agreed to vote any shares purchased following the IPO in the open market in favor of the merger. The Triplecrown Founders will vote all of their shares in any manner they determine, in their sole discretion, with respect to any other items that come before a vote of Triplecrown s stockholders.

Triplecrown will proceed with the business combination only if a majority of the Public Shares is voted in favor of the business combination and holders owning less than 30% of the Public Shares both exercise their conversion rights discussed below and vote against the business combination.

The board of directors of Triplecrown is divided into three classes, each of which will generally serve for a term of three years with only one class of directors being elected in each year. There is no cumulative voting with respect to the election of directors, with the result that the holders of more than 50% of the shares eligible to vote for the election of directors can elect all of the directors. CAH s board of directors is similarly classified.

If Triplecrown is forced to liquidate, the holders of the Public Shares will be entitled to share ratably in the trust fund, including any interest, and any net assets remaining available for distribution to them after payment of liabilities. The Triplecrown Founders have agreed to waive their rights to share in any distribution with respect to the Founders Shares.

Triplecrown s stockholders have no conversion, preemptive or other subscription rights and there are no sinking fund or redemption provisions applicable to the common stock, except that holders of the Public Shares (other than the

Triplecrown Founders) have the right to have their shares of common stock converted to cash equal to their pro rata share of the trust account if they vote against the business combination, properly demand conversion and the business combination is approved and completed. Public stockholders who convert their stock into their share of the trust account still have the right to exercise the warrants that they received as part of the units.

Upon consummation of the merger, the holders of shares of CAH common stock will be entitled to one vote for each share held of record on all matters to be voted on by the stockholders.

Holders of shares of CAH common stock will not have any conversion, preemptive or other subscription rights and there will be no sinking fund or redemption provisions applicable to the shares of common stock.

## **Preferred Stock**

Triplecrown s amended and restated certificate of incorporation authorizes the issuance of 1,000,000 shares of blank check preferred stock with such designation, rights and preferences as may be determined from time to time by its board of directors. No shares of preferred stock are being issued or registered in the merger. Accordingly, the board of directors is empowered, without stockholder approval, to issue preferred stock with dividend, liquidation, conversion, voting or other rights which could adversely affect the voting power or other rights of the holders of common stock.

However, the underwriting agreement from its IPO prohibits Triplecrown, prior to a business combination, from issuing preferred stock which participates in any manner in the proceeds of the trust account, or which votes as a class with the common stock on a business combination. Triplecrown may issue some or all of the preferred stock to effect a business combination. In addition, the preferred stock could be utilized as a method of discouraging, delaying or preventing a change in control of Triplecrown. There are no shares of preferred stock outstanding and Triplecrown does not currently intend to issue any shares of preferred stock.

CAH s certificate of incorporation authorizes the issuance of up to 1,000,000 shares of preferred stock with such designations, rights and preferences as may be determined from time to time by CAH s board of directors. Accordingly, CAH s board of directors will be empowered, without stockholder approval, to issue preferred shares with dividend, liquidation, conversion, voting or other rights which could adversely affect the voting power or other rights of the holders of shares of common stock. In addition, the preferred shares could be utilized as a method of discouraging, delaying or preventing a change in control of CAH.

## **Warrants**

Triplecrown currently has 74,000,000 redeemable common stock purchase warrants outstanding, including the Founders Warrants and Sponsors Warrants. Upon the consummation of the merger, each outstanding common stock purchase warrant shall be automatically converted into a warrant of to purchase shares of CAH common stock that will have terms that are substantially similar in all material respects to those of the Triplecrown warrants. Consequently, each CAH warrant issued in exchange for the public warrants and Sponsors Warrants will entitle the registered holder to purchase one share of CAH common stock at a price of \$12.00 per share (assuming the warrant amendments are approved), subject to adjustment as discussed below, at any time commencing after the completion of the merger. The CAH warrants issued in exchange for the Founders Warrants will not become exercisable until the last sales price of CAH s common stock exceeds \$13.75 per share for any 20 trading days within any 30-trading day period beginning 90 days after the consummation of the merger. The public warrants will be exercisable only if a registration statement relating to the shares of common stock issuable upon exercise of the warrants is effective and current, as described below. Assuming the warrant amendments are approved, the warrants will expire on October 21, 2013 at 5:00 p.m., New York City time.

After the consummation of the merger, assuming the warrant amendments are approved, CAH may call the warrants for redemption (excluding any Founders Warrants and Sponsors Warrants still held by the original purchasers of such warrants or their affiliates),

in whole and not in part,

at a price of \$0.01 per warrant at any time after the warrants become exercisable, upon not less than 30 days prior written notice of redemption to each warrant holder, and if, and only if, the reported last sale price of the shares of common stock equals or exceeds \$17.00 per share, for any 20 trading days within a 30 trading day period ending on the third business day prior to the notice of redemption to warrant holders.

Preferred Stock 241

The right to exercise will be forfeited unless the warrants are exercised prior to the date specified in the notice of redemption. On and after the redemption date, a record holder of a warrant will have no further rights except to receive the redemption price for such holder s warrant upon surrender of such warrant.

The redemption criteria for the warrants have been established at a price which is intended to provide warrant holders a reasonable premium to the initial exercise price and provide a sufficient differential between

129

Warrants 242

## **TABLE OF CONTENTS**

the then-prevailing common stock price and the warrant exercise price so that if the stock price declines as a result of the redemption call, the redemption will not be expected to cause the stock price to drop below the exercise price of the warrants.

The warrants were issued in registered form under a warrant agreement between Continental Stock Transfer & Trust Company, as warrant agent, and Triplecrown. You should review a copy of the warrant agreement, which has been filed as an exhibit to the Registration Statement on Form S-1 for Triplecrown s IPO (SEC File Nos. 333-144523 and 333-146850), for a complete description of the terms and conditions applicable to the warrants. The warrant agreement provides that the terms of the warrants may be amended without consent of any holder to cure any ambiguity or correct any defective provision, but requires the written consent of the registered holders of a majority of the then outstanding warrants in order to make any change that adversely affects the interests of the registered holders.

The exercise price and number of shares of common stock issuable on exercise of the warrants may be adjusted in certain circumstances including in the event of a stock dividend, recapitalization, reorganization, merger or consolidation. However, the warrants will not be adjusted for issuances of Triplecrown common stock or shares of CAH common stock at a price below their respective exercise prices.

The warrants may be exercised upon surrender of the warrant certificate on or prior to the expiration date at the offices of the warrant agent, with the exercise form on the reverse side of the warrant certificate completed and executed as indicated, accompanied by full payment of the exercise price, by certified or official bank check payable to CAH, for the number of warrants being exercised. The warrant holders do not have the rights or privileges of holders of common stock and any voting rights until they exercise their warrants and receive shares of common stock. After the issuance of shares of common stock upon exercise of the warrants, each holder will be entitled to one vote for each share held of record on all matters to be voted on by stockholders.

No public warrants will be exercisable and CAH will not be obligated to issue shares of common stock unless at the time a holder seeks to exercise such warrant, a prospectus relating to the shares of common stock issuable upon exercise of the warrants is current and the shares of common stock have been registered or qualified or deemed to be exempt under the securities laws of the state of residence of the holder of the warrants. Under the terms of the warrant agreement, CAH will use its best efforts to meet these conditions and to maintain a current prospectus relating to the shares of common stock issuable upon exercise of the warrants until the expiration of the warrants. However, CAH cannot assure you that it will be able to do so and, if it does not maintain a current prospectus relating to the shares of common stock issuable upon exercise of the warrants, holders will be unable to exercise their warrants and CAH will not be required to settle any such warrant exercise. If the prospectus relating to the shares of common stock issuable upon the exercise of the warrants is not current or if the shares of common stock are not qualified or exempt from qualification in the jurisdictions in which the holders of the warrants reside, CAH will not be required to net cash settle or cash settle the warrant exercise, the warrants may have no value, the market for the warrants may be limited and the warrants may expire worthless.

No fractional shares will be issued upon exercise of the warrants. If, upon exercise of the warrants, a holder would be entitled to receive a fractional interest in a share, CAH will, upon exercise, round up or down to the nearest whole number of shares of common stock.

The Founders Warrants are identical to the public warrants except that they will become exercisable after the merger if and when the last sales price of CAH s common stock exceeds \$13.75 per share for any 20 trading days within any 30-trading day period beginning 90 days after the consummation of the merger and the Founders Warrants will be exercisable on a cashless basis and will not be redeemable by CAH, in each case, as long as they are held by the Founders or their permitted transferees. In addition, they may be exercised for unregistered shares if a registration

Warrants 243

statement relating to the common stock issuable upon exercise of the warrants is not effective and current and commencing on the date such warrants become exercisable, the Founders Warrants and the underlying common stock are entitled to registration rights.

The Sponsors Warrants are identical to the public warrants except that they are not transferable or salable by Messrs. Watson or Ledecky (except in certain limited circumstances such as to relatives and trusts for

130

Warrants 244

estate planning purposes, providing the transferee agrees to be bound by the transfer restrictions) until Triplecrown completes a business combination and if Triplecrown calls the warrants for redemption, the Sponsors Warrants will not be redeemable so long as such warrants are held by Messrs. Watson, Ledecky or their affiliates, including any permitted transferees. In addition, they may be exercised for unregistered shares if a registration statement relating to the common stock issuable upon exercise of the warrants is not effective and current and commencing on the date such warrants become exercisable, the Sponsors Warrants and the underlying common stock are entitled to registration rights.

# **Transfer Agent and Warrant Agent**

The transfer agent for Triplecrown s securities and warrant agent is Continental Stock Transfer & Trust Company, 17 Battery Place, New York, New York 10004. After the merger, CAH s transfer agent and warrant agent will be Continental Stock Transfer & Trust Company.

# PRICE RANGE OF SECURITIES AND DIVIDENDS

Triplecrown s units, common stock and warrants are listed on the NYSE Amex under the symbols TCW.U, TCW and TCW.WS, respectively. The following table sets forth the range of high and low closing bid prices for the units, common stock and warrants for the periods indicated since such units commenced public trading on October 25, 2007 and since such common stock and warrants commenced public trading on October 31, 2007.

	Units		Common Stock		Warrants	
	High	Low	High	Low	High	Low
2009:						
Second Quarter*						
First Quarter						
2008:						
Fourth Quarter	9.45	8.85	9.05	8.60	0.30	0.06
Third Quarter	9.70	8.95	9.30	8.85	0.51	0.25
Second Quarter	9.95	9.45	9.29	8.99	0.70	0.41
First Quarter	10.2	9.65	9.20	8.97	1.09	0.54
2007:						
Fourth Quarter (commencing October 2007)	10.19	9.85	9.14	8.90	1.20	1.00

Through, 2009

The closing bid price for each share of common stock, warrant and unit of Triplecrown on September 3, 2009, the last trading day before announcement of the execution of the merger agreement, was \$, \$ and \$, respectively. As of , 2009, the record date for the Triplecrown special meeting, the closing bid price for each share of common stock, warrant and unit of Triplecrown was \$, \$ and \$, respectively.

Holders of Triplecrown common stock, warrants and units should obtain current market quotations for their securities. The market price of Triplecrown common stock, warrants and units could vary at any time before the merger.

Historical market price information for CAH s securities is not presented because there is no public market for CAH s securities.

## **Holders**

As of , 2009, there was one holder of record of Triplecrown units, 13 holders of record of Triplecrown common stock and 14 holders of record of Triplecrown warrants. Triplecrown believes that the aggregate number of beneficial holders of its units, common stock and warrants is in excess of persons.

As of , 2009, Triplecrown was the only holder of record of CAH s common stock.

# **Dividends**

Triplecrown has not paid any cash dividends on its common stock to date and does not intend to pay cash dividends prior to the completion of the merger. The payment of any cash dividends subsequent to the merger will be within the discretion of Triplecrown s then board of directors, subject to the relevant provision of Delaware law. The payment of

dividends subsequent to the merger will be contingent upon CAH s revenues and earnings, if any, capital requirements and general financial condition subsequent to completion of the merger, as well as contractual restrictions and other considerations deemed relevant by CAH s then board of directors.

# **Stock Price Performance Graph**

The following graph compares the cumulative total return for Triplecrown s common stock from October 31, 2007, the date its common stock first became separately tradable, through December 31, 2008

132

Dividends 247

with the comparable cumulative return of two indices, the S&P 500 Index and the Amex Composite Index (formerly the Amex Market Value Index). The graph assumes \$100 invested on October 31, 2007 in Triplecrown s common stock and \$100 invested at that same time in each of the two listed indices. The stock price performance shown on the graph is not necessarily indicative of future performance.

# STOCKHOLDER PROPOSALS

The CAH 2010 annual meeting of stockholders will be held on or about , 2010 unless the date is changed by the board of directors. If you are a stockholder of CAH and you want to include a proposal in the proxy statement for the year 2010 annual meeting, you need to provide it to by no later than , 2010. You should direct any proposals to CAH s secretary at CAH s principal office which will be located at . If you want to present a matter of business to be considered at the year 2010 annual meeting, under CAH s bylaws you must give timely notice of the matter, in writing, to its secretary. To be timely, the notice has to be given between , 2010 and , 2010.

# **LEGAL MATTERS**

Graubard Miller, The Chrysler Building, 405 Lexington Avenue, New York, New York 10174, is acting as counsel for Triplecrown and will pass upon the validity of the common stock issued in connection with the merger and certain other legal matters related to this proxy statement/prospectus. Richards, Layton & Finger, P.A. is acting as special counsel for Triplecrown and will pass upon certain legal matters related to Delaware corporate law. Ellenoff Grossman & Schole LLP, 150 East 42<sup>nd</sup> Street, New York, New York 10017, is acting as counsel for Cullen Agritech.

# **EXPERTS**

The audited financial statements of CAH for the period from August 27, 2009 (inception) through August 31, 2009 included in this proxy statement/prospectus have been so included in the reliance on a report of Marcum LLP, an independent registered public accounting firm, appearing elsewhere herein given on the authority of said firm, as experts in auditing and accounting.

The audited financial statements of Cullen Agritech for the period from June 3, 2009 (inception) through August 18, 2009 included in this proxy statement/prospectus have been so included in the reliance on a report of Marcum LLP, an independent registered public accounting firm, appearing elsewhere herein given on the authority of said firm, as experts in auditing and accounting.

133

EXPERTS 248

The audited financial statements of Triplecrown Acquisition Corp. as of December 31, 2008 and 2007, and for the periods from June 8, 2007 (inception) through December 31, 2007 and 2008, included in this proxy statement/prospectus have been so included in the reliance on a report of Marcum LLP (formerly Marcum & Kliegman LLP), an independent registered public accounting firm, appearing elsewhere herein given on the authority of said firm, as experts in auditing and accounting.

Representatives of Marcum LLP will be present at the stockholder meeting or will be available by telephone with the opportunity to make statements and to respond to appropriate questions.

# DELIVERY OF DOCUMENTS TO STOCKHOLDERS

Pursuant to the rules of the SEC, Triplecrown and services that it employs to deliver communications to its stockholders are permitted to deliver to two or more stockholders sharing the same address a single copy of each of Triplecrown s annual report to stockholders and Triplecrown s proxy statement/prospectus. Upon written or oral request, Triplecrown will deliver a separate copy of the annual report to stockholder and/or proxy statement/prospectus to any stockholder at a shared address to which a single copy of each document was delivered and who wishes to receive separate copies of such documents in the future. Stockholders receiving multiple copies of such documents may likewise request that Triplecrown deliver single copies of such documents in the future. Stockholders may notify Triplecrown of their requests by calling or writing Triplecrown at its principal executive offices at 970 West Broadway, PMB 402, Jackson, Wyoming 83001 (307) 633-2831.

# WHERE YOU CAN FIND MORE INFORMATION

Triplecrown files reports, proxy statements and other information with the SEC as required by the Exchange Act. You may read and copy reports, proxy statements and other information filed by Triplecrown with the SEC at the SEC public reference room located at 100 F Street, N.E., Washington, D.C. 20549. You may obtain information on the operation of the Public Reference Room by calling the SEC at 1-800-SEC-0330. You may also obtain copies of the materials described above at prescribed rates by writing to the SEC, Public Reference Section, 100 F Street, N.E., Washington, D.C. 20549. You may access information on Triplecrown at the SEC web site containing reports, proxy statement/prospectus and other information at: <a href="http://www.sec.gov">http://www.sec.gov</a>.

Information and statements contained in this proxy statement/prospectus are qualified in all respects by reference to the copy of the relevant contract or other document included as an annex to this proxy statement/prospectus.

If you would like additional copies of this proxy statement/prospectus or if you have questions about the merger, you should contact via phone or in writing:

Mr. Jonathan J. Ledecky
Secretary
Triplecrown Acquisition Corp.
c/o Paul Vassilakos

590 Madison Avenue, 21st Floor New York, New York 10022 Tel: (212) 521-5396

or

Morrow & Co., LLC 470 West Avenue Stamford, Connecticut 06902 Telephone: (800) 662-5200

If you are a stockholder or warrantholder of Triplecrown and would like to request documents, please do so by , 2009 to receive them before the Triplecrown special meetings. If you request any documents from Triplecrown, Triplecrown will mail them to you by first class mail, or another equally prompt means.

CAH has filed with the SEC a registration statement on Form S-4, including exhibits and schedules filed with the registration statement of which this prospectus is a part, under the Securities Act with respect to the CAH s common stock and warrants to be issued upon consummation of the merger, as well as the common stock underlying the warrants. This Proxy Statement/Prospectus does not contain all of the information set forth in the registration statement and exhibits and schedules to the registration statement. For further information with respect to CAH and its common stock and warrants to be issued upon consummation of the merger, reference is made to the registration statement, including the exhibits and schedules to the registration statement. Copies of the registration statement, including the exhibits and schedules to the registration statement, may be examined without charge at the public reference room of the SEC, 100 F Street, N.E., Room 1580, Washington, D.C. 20549. Information about the operation of the public reference room may be obtained by calling the SEC at 1-800-SEC-0300. Copies of all or a portion of the registration statement may be obtained from the public reference room of the SEC upon payment of prescribed fees. CAH s SEC filings, including its registration statement, are also available to you, free of charge, on the SEC s website at www.sec.gov.

As a result of the consummation of the merger, CAH will become subject to the information and reporting requirements of the Exchange Act and will file periodic reports, proxy statements and will make available to CAH s stockholders annual reports containing audited financial information for each year and quarterly reports for the first three quarters of each fiscal year containing unaudited interim financial information.

All information contained or incorporated by reference in this proxy statement/prospectus relating to Triplecrown and CAH has been supplied by Triplecrown, and all such information relating to Cullen Agritech has been supplied by Cullen Agritech. Information provided by either Triplecrown or Cullen Agritech does not constitute any representation, estimate or projection of the other party.

This document is a prospectus of CAH and proxy statement of Triplecrown for the Special Meetings of Triplecrown s stockholders and warrantholders. Neither CAH nor Triplecrown has authorized anyone to give any information or make any representation about the mergers, CAH, Cullen Agritech or Triplecrown that is different from, or in addition to, that contained in this proxy statement/prospectus or in any of the materials that Triplecrown has incorporated by reference into this proxy statement/prospectus. Therefore, if anyone does give you information of this sort, you should not rely on it. The information contained in this document speaks only as of the date of this document unless the information specifically indicates that another date applies.

# **INDEX TO FINANCIAL STATEMENTS**

	Page
Cullen Agricultural Technologies Inc.	
Report of Independent Registered Public Accounting Firm	<u>F-2</u>
Combined Balance Sheet at August 18, 2009	<u>F-3</u>
Combined Statement of Operations for the Period from June 3, 2009 (inception) through August	<u>F-4</u>
<u>18, 2009</u>	14
Combined Statement of Changes in Stockholder s Deficit for the Period from June 3, 2009	F-5
(inception) through August 18, 2009	<u>1-5</u>
Combined Statement of Cash Flows for the Period from June 3, 2009 (inception) through August	F-6
<u>18, 2009</u>	
Notes to Combined Financial Statements	<u>F-7</u>
Cullen Agricultural Holding Corp.	
Report of Independent Registered Public Accounting Firm	<u>F-12</u>
Combined Balance Sheet at August 31, 2009	<u>F-13</u>
Combined Statement of Operations for the Period from August 27, 2009 (inception) through	F-14
<u>August 31, 2009</u>	1 17
Combined Statement of Changes in Stockholder s Deficit for the Period from August 27, 2009	F-15
(inception) through August 31, 2009	1 13
Combined Statement of Cash Flows for the Period from August 27, 2009 (inception) through	F-15
<u>August 31, 2009</u>	
Notes to Combined Financial Statements	<u>F-17</u>
Triplecrown Acquisition Corp.	
Report of Independent Registered Public Accounting Firm	<u>F-21</u>
Balance Sheets as of December 31, 2008 and December 31, 2007	<u>F-22</u>
Statements of Income for the year ended December 31, 2008, and for the period from June 8,	
2007 (inception) through December 31, 2007, and for the period from June 8, 2007 (inception)	<u>F-23</u>
through December 31, 2008	
Statement of Changes in Stockholders Equity for the period from June 8, 2007 (inception)	F-24
through December 31, 2008	<del></del>
Statements of Cash Flows for the year ended December 31, 2008, and for the period from June 8,	
2007 (inception) through December 31, 2007, and for the period from June 8, 2007 (inception)	<u>F-25</u>
through December 31, 2008	
Notes to Financial Statements	<u>F-26</u>
Condensed Balance Sheets at June 30, 2009 (Unaudited) and December 31, 2008	<u>F-36</u>
Condensed Statements of Operations (Unaudited) for the three and six months ended June 30.	F-37
2009 and 2008, and for the period June 8, 2007 (inception) through June 30, 2009	
Condensed Statement of Changes in Stockholders Equity (Unaudited) for the period from June 8,	F-38
2007 (inception) through June 30, 2009	
Condensed Statements of Cash Flows (Unaudited) for the three and six months ended June 30,	F-39
2009, and for the period June 8, 2007 (inception) through June 30, 2009	
Notes to Unaudited Condensed Financial Statements	<u>F-40</u>

F-1

### REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Stockholder and Board of Directors of Cullen Agricultural Technologies Inc. and Affiliate

We have audited the accompanying combined balance sheet of Cullen Agricultural Technologies Inc. and Affiliate (a development stage company) (the Company), as of August 18, 2009 and the related combined statements of operations, changes in stockholder is deficit and cash flows for the period from June 3, 2009 (inception) through August 18, 2009. These combined financial statements are the responsibility of the Company is management. Our responsibility is to express an opinion on these combined financial statements based on our audit.

We conducted our audit in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. Our audit included consideration of internal control over financial reporting as a basis for designing audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion. An audit also includes examining, on a test basis, evidence supporting the amounts and disclosures in the combined financial statements, assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audit provide a reasonable basis for our opinion.

The accompanying financial statements have been prepared assuming that the Company will continue as a going concern. As discussed in Note 1 to the combined financial statements, the Company s operations have been funded by its parent company, Cullen Inc. Holding Ltd., a New Zealand company (the Parent ). Although the Parent continues to fund the operations as deemed necessary, these factors raise substantial doubt about its ability to continue as a going concern. Management s plans regarding those matters are described in Note 1. The combined financial statements do not include any adjustments that might result from the outcome of this uncertainty.

In our opinion, the combined financial statements referred to above present fairly, in all material respects, the financial position of Cullen Agricultural Technologies Inc. and Affiliate (a development stage company) as of August 18, 2009, and the results of its operations and cash flows for the period from June 3, 2009 (inception) through August 18, 2009, in conformity with accounting principles generally accepted in the United States of America.

/s/ Marcum LLP

September 4, 2009 Melville, NY

### CULLEN AGRICULTURAL TECHNOLOGIES INC. AND AFFILIATE

(A Development Stage Company)

### COMBINED BALANCE SHEET August 18, 2009

Current assets	
Prepaid expenses	\$19,241
Intangible assets	3,327
Total assets	\$22,568
LIABILITIES AND STOCKHOLDER S DEFICIT	
Current liabilities:	
Accrued expenses	\$9,246
Due to parent company	35,484
Total current liabilities	44,730
Stockholder s deficit:	

Common stock no par value; authorized 200,000 shares; issued and outstanding 200 shares

Additional paid-in capital 200

Deficit accumulated during development stage (22,362)

Total stockholder s deficit (22,162)

Total liabilities and stockholder s deficit \$22,568

The accompanying notes are an integral part of these combined financial statements.

**ASSETS** 

### CULLEN AGRICULTURAL TECHNOLOGIES INC. AND AFFILIATE

(A Development Stage Company)

# COMBINED STATEMENT OF OPERATIONS For the Period from June 3, 2009 (Inception) Through August 18, 2009

Revenue	\$
Operating expenses	
Organization costs	6,200
Professional fees	15,013
Travel	753
Website costs	396
Total operating expenses	22,362
Net loss	\$ (22,362)

The accompanying notes are an integral part of these combined financial statements.

### CULLEN AGRICULTURAL TECHNOLOGIES INC. AND AFFILIATE

(A Development Stage Company)

#### COMBINED STATEMENT OF CHANGES IN STOCKHOLDER S DEFICIT For the Period from June 3, 2009 (Inception) Through August 18, 2009

			Deficit	
	AdditionalAccumulated Total			
	Common	Paid-In	During	Stockholder s
	Stock	Capital	Developmen	nt Deficit
	SharesAm	ount	Stage	
Balance, June 3, 2009 (inception)	\$	\$	\$	\$
Issuance of stock to initial stockholder	200	200		200
Net loss for the period from June 3, 2009 (inception)			(22,362)	(22.362)
through August 18, 2009			(22,302 )	(22,502 )
Balance, August 18, 2009	200 \$	\$ 200	\$(22,362)	\$(22,162)

The accompanying notes are an integral part of these combined financial statements.

### CULLEN AGRICULTURAL TECHNOLOGIES INC. AND AFFILIATE

(A Development Stage Company)

#### COMBINED STATEMENT OF CASH FLOWS For the Period from June 3, 2009 (Inception) Through August 18, 2009

Cash Flows from Operating Activities:		
Net loss	\$ (22,362	)
Adjustments To Reconcile Net Income To Net		
Cash Provided By Operating Activities: Cash Provided By Operating Activities:		
Changes in Operating Assets and Liabilities:		
Prepaid expenses	(19,241	)
Accrued expenses	9,246	
Net Cash Provided by Operating Activities	(32,357	)
Cash Flows From Investing Activities:		
Acquisition of intangible assets	(3,327	)
Net Cash (Used In) Investing Activities	(3,327	)
Cash Flows From Financing Activities:		
Due to Parent Company	35,484	
Issuance of Common Stock	200	
Net Cash Provided by Financing Activities	35,684	
Net Increase (Decrease) in Cash		
Cash at the Beginning of the Period		
Cash at the End of the Period	\$	

#### **Supplemental Disclosure**

As fully described in Note 1, the Company s operations have been funded by its stockholder, Cullen Holding, Ltd. during the period from June 3, 2009 (inception) through August 18, 2009.

The accompanying notes are an integral part of these combined financial statements.

# CULLEN AGRICULTURAL TECHNOLOGIES INC. AND AFFILIATE

(A Development Stage Company)

#### NOTES TO COMBINED FINANCIAL STATEMENTS

### Note 1 Organization, Business Operations, Significant Accounting Policies and Going Concern Consideration

#### **Organization and Nature of Operations**

Cullen Agricultural Technologies Inc. and Natural Dairy Inc., (collectively the Company or Cullen Agritech), were incorporated under the laws of the state of Georgia on June 3, 2009. Cullen Agricultural Technologies Inc. (Cullen) was formed to be a research and development company committed to the development and commercialization of advanced agricultural technologies. Cullen will provide advisory services associated with the implementation of more efficient farming techniques and promotes a methodology that incorporates components of New Zealand s pasture-based farming system. Natural Dairy Inc. (Natural Dairy) was formed to deploy Cullen s farming techniques and systems, through the purchase of land to be used as a dairy farm. The Company has selected December 31 as its year end.

#### **Principles of Combination**

The combined financial statements include the accounts of Cullen Agricultural Technologies Inc. and Natural Dairy Inc., both of which are under common control and ownership of Cullen Inc Holdings Ltd., a New Zealand company (the Parent ). All significant intercompany balances and transactions have been eliminated in combination.

#### **Going Concern Consideration**

The accompanying combined financial statements have been prepared on a going concern basis, which contemplates the realization of assets and the settlement of liabilities and commitments in the normal course of business. As reflected in the accompanying combined financial statements, the Company had a net loss of \$22,362 for the period from June 3, 2009 (inception) through August 18, 2009, and has no cash as of August 18, 2009. To date, the Company s operations have been funded by the Parent. Although the Parent plans to continue to fund the operations as deemed necessary, these factors, among others, raise substantial doubt about the Company s ability to continue as a going concern. The Company s ability to continue as a going concern is dependent on its ability to finance operations through debt and equity financing. The Company may have difficulty in obtaining financing on favorable terms, if at all. The combined financial statements do not include any adjustments that might be necessary if it is unable to continue as a going concern.

#### **Cash and Cash Equivalents**

The Company considers all highly liquid investments purchased with an original maturity of three months or less to be cash equivalents. At August 18, 2009 the Company had no cash or cash equivalents.

#### **Intangible Assets**

Intangible assets consist of costs related to the design and development of the Company s web site. These costs were capitalized in accordance with the Financial Accounting Standards Board (FASB) Emerging Issues Task Force (EITF) No. 00-2, Accounting for Web Site Development Costs. No amortization expense has been recorded for the period from June 3, 2009 (inception) through August 18, 2009.

#### **Use of Estimates**

The preparation of combined financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the combined financial statements and the reported amounts of expenses during the reporting period. Actual results could differ from those estimates.

## CULLEN AGRICULTURAL TECHNOLOGIES INC. AND AFFILIATE

(A Development Stage Company)

#### NOTES TO COMBINED FINANCIAL STATEMENTS

# Note 1 Organization, Business Operations, Significant Accounting Policies and Going Concern Consideration (continued)

#### **Organization Costs**

Organization costs of \$6,200 consist of legal and incorporation fees, incurred through the balance sheet date that are related to the formation of the corporations and were expensed as incurred in accordance with Statement of Position 98-5, Reporting on the Cost of Start-up Activities.

#### **Income Taxes**

The Company uses the asset and liability method of accounting for income taxes in accordance with FAS 109, Accounting for Income Taxes . Under the asset and liability method, deferred tax assets and liabilities are recognized for the future tax consequences attributable to differences between the financial statement carrying amounts and the tax bases of existing assets and liabilities. A valuation allowance against deferred tax assets is provided when it is more likely than not that the deferred tax asset will not be fully realized. As of August 18, 2009, the Company has a net loss of \$22,362, which if realized would have a tax benefit of \$7,603. The Company has determined that this deferred tax benefit has no value at this time, as the Company does not believe that it will utilize these losses in the future, and accordingly has not been recorded as a deferred tax asset.

The Company adopted the provisions of FASB Interpretation No. 48 (FIN 48), on June 3, 2009 (inception). FIN 48 clarifies the accounting for uncertainty in income taxes recognized in an enterprise s financial statements in accordance with SFAS No. 109, Accounting for Income Taxes, and prescribes a recognition threshold and measurement process for financial statement recognition and measurement of a tax position taken or expected to be taken in a income tax return. FIN 48 also provides guidance on derecognition, classification, interest and penalties, accounting in interim periods, disclosure and transition. In accordance with FASB Staff Position (FSP) FIN 48-3, the Company is not required to reflect the adoption of FIN 48 in its combined financial statements until the issuance of its combined financial statements for the period ending December 31, 2009. The adoption of FIN 48 has not had a material effect on the Company s combined financial position or results of operations. The Company classifies interest and penalties, if any, associated with our uncertain tax positions as a component of income tax expense. No interest and penalties related to uncertain tax positions were accrued at August 18, 2009.

During the period from June 3, 2009 (inception) through August 18, 2009, the Company recognized no adjustments for uncertain tax benefits. The Company is subject to U.S. federal and state examinations by tax authorities since its

#### Edgar Filing: Triplecrown Acquisition Corp. - Form PREM14A

inception. The Company does not expect any significant changes to its unrecognized tax positions during the next 12 months.

#### **Recent Accounting Pronouncements**

In December 2007, the FASB issued Statement of Financial Accounting Standards No. 141 (revised 2007), Business Combinations (SFAS 141R). SFAS 141R changes accounting for acquisitions that close beginning in 2009 in a number of areas including the treatment of contingent consideration, contingencies, acquisition costs, In-process research & development and restructuring costs. More transactions and events will qualify as business combinations and will be accounted for at fair value under the new standard. SFAS 141R promotes greater use of fair values in financial reporting. In addition, under SFAS 141R, changes in deferred tax asset valuation allowances and acquired income tax uncertainties in a business combination after the measurement period will impact income tax expense. Some of the changes will introduce more volatility into earnings. SFAS 141R is effective for fiscal years beginning on or after December 15, 2008. SFAS 141R will have an impact on accounting for any business acquired after the effective date of this pronouncement.

In December 2007, the FASB issued Statement of Financial Accounting Standards No. 160, Noncontrolling Interests in Consolidated Financial Statements, an amendment of ARB No. 51 (SFAS 160). SFAS 160

F-8

Income Taxes 261

# CULLEN AGRICULTURAL TECHNOLOGIES INC. AND AFFILIATE

(A Development Stage Company)

#### NOTES TO COMBINED FINANCIAL STATEMENTS

# Note 1 Organization, Business Operations, Significant Accounting Policies and Going Concern Consideration (continued)

will change the accounting and reporting for minority interests, which will be recharacterized as noncontrolling interests (NCI) and classified as a component of equity. This new consolidation method will significantly change the accounting for transactions with minority interest holders. SFAS 160 is effective for fiscal years beginning after December 15, 2008. SFAS 160 would have an impact on the presentation and disclosure of the noncontrolling interests of any non-wholly owned business acquired in the future.

In April 2009, the FASB issued FASB Staff Position No. 141(R)-1 (FSP FAS 141(R)-1) which provides additional clarification on the initial recognition and measurement of assets acquired and liabilities assumed in a business combination that arise from contingencies. FSP FAS 141(R)-1 is effective for all fiscal years beginning on or after December 15, 2008. FSP FAS 141(R)-1 will have an impact on the accounting for any business acquired in the future.

In May 2009, the FASB issued SFAS No. 165, Subsequent Events (SFAS No. 165). The statement is to establish general standards of accounting for and disclosure of events that occur after the balance sheet date but before financial statements are issued. SFAS No. 165 became effective June 15, 2009 for all subsequent reporting periods. The adoption of SFAS No. 165 did not have a material impact on the Company s combined financial statements.

In June 2009, the FASB issued SFAS No. 167, Amendments to FASB Interpretation No. 46(R) (SFAS 167). SFAS 167 eliminates Interpretation 46(R) s exceptions to consolidating qualifying special-purpose entities, contains new criteria for determining the primary beneficiary, and increases the frequency of required reassessments to determine whether a company is the primary beneficiary of a variable interest entity. SFAS 167 also contains a new requirement that any term, transaction, or arrangement that does not have a substantive effect on an entity s status as a variable interest entity, a company s power over a variable interest entity, or a company s obligation to absorb losses or its right to receive benefits of an entity must be disregarded in applying Interpretation 46(R) s provisions. The elimination of the qualifying special-purpose entity concept and its consolidation exceptions means more entities will be subject to consolidation assessments and reassessments. SFAS 167 will be effective January 1, 2010. The adoption of SFAS 167 is not expected to have an impact on the Company s combined financial statements or results of operations.

In July 2009, the FASB issued SFAS No. 168, The FASB Accounting Standards Codification and the Hierarchy of Generally Accepted Accounting Principles a replacement of FASB Statement No. 162 (SFAS 168). SFAS 168 establishes the FASB Accounting Standards Codification to become the source of authoritative U.S. GAAP recognized by the FASB to be applied by non-governmental entities. Rules and interpretive releases of the Securities and Exchange Commission (SEC) under authority of federal securities laws are also sources of authoritative GAAP

#### Edgar Filing: Triplecrown Acquisition Corp. - Form PREM14A

for SEC registrants. On the effective date of this Statement, the Codification will supercede all then-existing non-SEC accounting and reporting standards. All other nongrandfathered non-SEC accounting literature not included in the Codification will become nonauthoritative. SFAS 168, which changes the referencing of financial standards, is effective for interim or annual financial periods ending after September 15, 2009. The adoption of SFAS 168 is not expected to materially impact the Company s combined financial statements or results of operations.

Management does not believe that any other recently issued, but not yet effective, accounting standards if currently adopted would have a material effect on the combined financial statements.

#### Note 2 Related Parties

The Company s Chief Executive Officer, Executive Chairman and Treasurer (the CEO) is a member of the Board of Directors of the Company s Parent. As of August 18, 2009, the Parent has been funding the operations of the Company and is owed a combined total of \$35,484.

# CULLEN AGRICULTURAL TECHNOLOGIES INC. AND AFFILIATE

(A Development Stage Company)

#### NOTES TO COMBINED FINANCIAL STATEMENTS

Note 2 Related Parties (continued)

The Company s CEO is the Chairman of the Board and Treasurer of Triplecrown Acquisition Corp. (the Related Party or Triplecrown). The Company and the Related Party entered into an agreement to purchase land from a third party, as fully described in Note 3. On September 4, 2009, the Company entered into an Agreement and Plan of Reorganization with the Related Party (the Merger) as fully described in Note 4.

#### Note 3 Commitments and Contingencies

On June 27, 2009, the Company and the Related Party entered into an agreement with a third party (the Seller ) to purchase a certain parcel of land (the Property ). The total purchase price of the Property is \$8,662,500. The Related Party paid an initial deposit of \$866,250 (the Deposit ) on the land. On August 10, 2009, the Company, the Related Party, and the Seller entered into an extension agreement (the Extension ) to extend the closing date for the land to September 15, 2009. In exchange for the Extension, the Related Party paid an additional deposit on the land of \$833,750, interest of \$48,070 and a leasing fee for the Company to use the land of \$3,518, for a total of \$885,338 (the Extension Payment ). On August 10, 2009, the Company entered into an agreement with the Related Party in which the Company, and the Company s CEO, agreed to be jointly and severally liable to assume the obligation for the remaining purchase price of the Property of \$6,962,500 if the Merger is not consumated.

On August 11, 2009, Cullen entered into a strategic cooperation agreement with New Zealand Agritech Inc ( Agritech ) to promote the interests of Agritech and its members. Cullen will assist members of Aritech to mitigate barriers of market entry and provide the opportunity to realize the potential growth in various markets. Agritech will actively promote to its members its alliance with Cullen. Cullen is obligated to pay a fee to Agritech annually, in arrears, based on its dealings with Agritech s members. The fee is to be negotiated on a case by case basis.

On August 1, 2009, the Company obtained a nonbinding letter of intent from Struve Technologies LLC (Struve) whereby Struve has agreed to supply livestock and breeding services to Cullen. Struve offered Cullen a preferred supplier agreement for a period of not less than five years covering the supply of livestock related to the Company s business. For 2010, Struve has agreed that it has the ability to supply the Company up to 60,000 dairy cows meeting the Company s specifications.

On August 13, 2009, Cullen obtained a nonbinding letter of intent from Discus Enterprises Limited granting the Company the exclusive license to use the EpiManager software in the United States. EpiManager is a web based information management system used in tracking non-feral animals, including food animals and pets. Cullen will incur the associated costs to adapt the EpiManager system to their specific needs and other dairy operators in the United States.

#### Edgar Filing: Triplecrown Acquisition Corp. - Form PREM14A

On August 14, 2009, the Company entered into a nonbinding letter of intent with Farmax to explore the development of a dairy farm management software model for the United States (the US Model). Cullen will pay for the development and reconfiguration the Farmax Dairy Pro software to the US model. Cullen will own the intellectual property related to the US model and will have exclusive rights to use and market the model. Cullen will have to pay a royalty to Farmax based on the sales revenue generated in the United States from the sale of the US model.

On August 14, 2009, Cullen entered into an agreement with AHA Advisors Ltd to act as a member of the Company s advisory board and to work exclusively with the Company in the development of a genetically advanced dairy cow milking herd for Natural Dairy Inc. AHA Advisors Ltd will assist in the discovery of a herd uniquely capable of meeting the production and performance goals of the Company.

During August 2009, REPOROA Engineering 1982 Limited (REPOROA) and the Company entered into a letter of intent whereby REPOROA will design and construct milking parlors in the United States for

# CULLEN AGRICULTURAL TECHNOLOGIES INC. AND AFFILIATE

(A Development Stage Company)

#### NOTES TO COMBINED FINANCIAL STATEMENTS

Note 3 Commitments and Contingencies (continued)

the Company. REPOROA will initially design and construct three milking parlors and can potentially provide twenty parlors during 2010. REPOROA is currently working on an estimate and finance package for Cullen.

#### Note 4 Subsequent Events

On August 30, 2009, pursuant to an agreement of a plan of contribution of shares, the Parent contributed 100 shares of Natural Dairy to Cullen Agricultrural Technologies, Inc. As a result of the contribution, Natural Dairy is a now a wholly-owned subsidiary of Cullen Agricultural Technologies, Inc.

On August 31, 2009, Natural Dairy entered into an employment agreement with Richard Watson (Watson) to be the Chief Scientific Officer with a term commencing on September 1, 2009 and expiring on August 31, 2012. The agreement calls for a base salary of \$100,000 and a discretionary bonus of up to 50% of the base salary.

On September 3, 2009, the Company entered into an agreement with Cullen Investments Limited (CIL), Hart Acquisitions, LLC (Hart), and Watson, whereby CIL, Hart and Watson will assign the rights to certain intellectual property including a proprietary farming system to the Company, should the Merger be completed. The Company will have the sole right to use the proprietary farming system, including assigning rights for the intellectual property to third parties, as of the effective date of the Merger.

Pursuant to the Agreement and Plan of Reorganization dated September 4, 2009, executed by Cullen Agritech, Triplecrown, Cullen Agricultural Holdings, Inc. (Holdco), CAT Merger Sub, Inc. (MergerSub), and Cullen Inc Holdings Ltd (CIL), the parties to the agreement propose to merge Triplecrown into Holdco, with Holdco being the surviving public entity. Cullen Agritech will merge into MergerSub, a wholly-owned subsidiary of Holdco, and MergerSub shall cease to exist, with Cullen Agritech as the surviving subsidiary of Holdco. As a result of the merger, the holders of common stock and warrants of Triplecrown will receive like securities of Holdco, on a one-to-one basis, in exchange for their existing securities, except that 11,260,000 shares of the 13,800,000 shares of common stock owned by Triplecrown s founding shareholders will be cancelled. Cullen Agritech shares of Cullen Agritech s common stock, Holdco will receive 100 shares of Cullen Agritech s common stock. As a result of the merger, Cullen Agritech will be a wholly-owned subsidiary of Holdco.

Completion of the merger is subject to and conditioned upon, among other things, approval by the stockholders of Triplecrown and agreement by the holders of Triplecrown s warrants to agree to amend the warrant agreement increase the strike price of the warrants from \$7.50 per share to \$12.00 per share, to extend the expiration date to October 22, 2013. the Triplecrown s trust account and to increase the price at which the stock must trade for the warrants to be

#### Edgar Filing: Triplecrown Acquisition Corp. - Form PREM14A

called for redemption from \$13.75 per share to \$17.00 per share. In order to improve the probability that the merger is approved, Triplecrown may enter into agreements to purchase shares of common stock from shareholders who indicate their intention to vote against the merger, paying for the shares with funds from the trust account immediately after consummation of the merger. The funds available to Holdco following consummation of the merger, therefore, will be reduced based upon the number of stockholders who exercise their conversion rights or arrange to sell their shares to Triplecrown upon consummation of the merger.

In the event the proposed merger is not consummated, the parties to the merger agreement will bear their own legal and other costs.

The Company has evaluated subsequent events that occurred subsequent to August 18, 2009 through September 4, 2009, the date on which these financial statements were issued. Except as disclosed, management concluded that no other events required potential adjustment to or disclosure in these combined financial statements.

### REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Stockholder and Board of Directors of Cullen Agricultural Holding Corp. and Subsidiary

We have audited the accompanying consolidated balance sheet of Cullen Agricultural Holding Corp. and Subsidiary (a development stage company) (the Company), a wholly owned subsidiary of Triplecrown Acquisition Corp. as of August 31, 2009 and the related consolidated statements of operations, changes in stockholder s deficit and cash flows for the period from August 27, 2009 (inception) through August 31, 2009. These consolidated financial statements are the responsibility of the Company s management. Our responsibility is to express an opinion on these consolidated financial statements based on our audit.

We conducted our audit in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the consolidated financial statements are free of material misstatement. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. Our audit included consideration of internal control over financial reporting as a basis for designing audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company s internal control over financial reporting. Accordingly, we express no such opinion. An audit also includes examining, on a test basis, evidence supporting the amounts and disclosures in the consolidated financial statements, assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audit provide a reasonable basis for our opinion.

The accompanying consolidated financial statements have been prepared assuming the Company will continue as a going concern. As discussed in Note 1 to the consolidated financial statements, the Company s funds may not be sufficient to maintain the Company. In addition, the certificate of incorporation of Triplecrown Acquisition Corp. (the Parent ) provides for mandatory liquidation of the Parent in the event that the Parent does not consummate a business combination prior to October 22, 2009. These factors raise substantial doubt about the Company s ability to continue as a going concern. Management s plans regarding those matters are described in Note 1. The consolidated financial statements do not include any adjustments that might result from the outcome of this uncertainty.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the financial position of Cullen Agricultural Holding Corp. and Subsidiary (a development stage company) as of August 31, 2009, and the results of its operations and cash flows for the period from August 27, 2009 (inception) through August 31, 2009, in conformity with accounting principles generally accepted in the United States of America.

/s/ Marcum LLP

September 4, 2009 Melville, NY

### CULLEN AGRICULTURAL HOLDING CORP. AND SUBSIDIARY

(A Development Stage Company)
(A Wholly Owned Subsidiary of Triplecrown Acquisition Corp.)

### CONSOLIDATED BALANCE SHEET August 31, 2009

ASSETS	
Total assets	\$
LIABILITIES AND STOCKHOLDER S DEFICIT	
Current liabilities:	
Accrued expenses	1,000
Total current liabilities	1,000
Stockholder s deficit:	
Preferred stock \$0.0001 par value; authorized 1,000,000 shares; no shares issued and outstanding	
Common stock \$0.0001 par value; authorized 160,000,000 shares; issued and outstanding	10
shares	
Deficit accumulated during development stage	(1,000)
Total stockholder s deficit	(1,000)
Total liabilities and stockholder s deficit	\$

The accompanying notes are an integral part of these consolidated financial statements.

### CULLEN AGRICULTURAL HOLDING CORP. AND SUBSIDIARY

(A Development Stage Company)
(A Wholly Owned Subsidiary of Triplecrown Acquisition Corp.)

#### CONSOLIDATED STATEMENT OF OPERATIONS For the Period from August 27, 2009 (Inception) Through August 31, 2009

Revenue	\$	
Operating expenses		
Organization costs	1,000	
Total operating expenses	1,000	
Net loss	\$ (1,000	)
Weighted average number of common shares outstanding basic and diluted	10	
Basic and diluted net loss per share	\$ (100	)

The accompanying notes are an integral part of these consolidated financial statements.

### CULLEN AGRICULTURAL HOLDING CORP. AND SUBSIDIARY

(A Development Stage Company)
(A Wholly Owned Subsidiary of Triplecrown Acquisition Corp.)

# CONSOLIDATED STATEMENT OF CHANGES IN STOCKHOLDER S DEFICIT For the Period from August 27, 2009 (Inception) Through August 31, 2009

					Deficit				
		AdditionAccumulated Total							
		Com		Paid-	In During		Stockhole	der s	
		Common Stock		Capit	al DevelopmentDe		ntDeficit	Deficit	
			x esAmoı	ınt	Stage				
Balance, August 27, 2009 (inception)		Silar	\$	\$	\$		\$		
Issuance of stock to initial stockholder (Parent)	10		Ψ	Ψ	Ψ		Ψ		
shares at \$0.001 per share		10							
Net (loss) for the period from August 27, 2009 (inception) through August 31, 2009					(1,000	)	(1,000	)	
Balance, August 31, 2009		10	\$	\$	\$ (1,000	)	\$(1,000	)	

The accompanying notes are an integral part of these consolidated financial statements.

### CULLEN AGRICULTURAL HOLDING CORP. AND SUBSIDIARY

(A Development Stage Company)
(A Wholly Owned Subsidiary of Triplecrown Acquisition Corp.)

#### CONSOLIDATED STATEMENT OF CASH FLOWS For the Period from August 27, 2009 (Inception) Through August 31, 2009

Cash flows from operating activities:

Net (loss) \$(1,000 )

Adjustments to reconcile net (loss) to net cash provided by operating activities: Cash

Provided By Operating Activities:

Changes in operating assets and liabilities:

Accrued expenses 1,000

Net cash provided by operating activities

Net increase (decrease) in cash

Cash at the beginning of the period

Cash at the end of the period \$

The accompanying notes are an integral part of these consolidated financial statements.

### CULLEN AGRICULTURAL HOLDING CORP. AND SUBSIDIARY

(A Development Stage Company)
(A Wholly Owned Subsidiary of Triplecrown Acquisition Corp.)

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

### Note 1 Organization, Business Operations, Significant Accounting Policies and Going Concern Consideration

#### **Organization and Nature of Operations**

Cullen Agricultural Holding Corp. (a development stage company) was incorporated under the laws of the state of Delaware on August 27, 2009. Cullen Agricultural Holding Corp. was formed as a wholly-owned subsidiary of Triplecrown Acquisition Corp. (the Parent or Triplecrown). CAT Merger Sub, Inc., a Georgia Corporation, was incorporated as a wholly-owned subsidiary of Cullen Agricultural Holding Corp. on August 31, 2009. Cullen Agricultural Holding Corp. and CAT Merger Sub, Inc. (collectively, the Company) were formed effect the transactions contemplated by the Merger Agreement described below in Note 2. The company chose December 31 as its fiscal year end.

#### **Principles of Consolidation**

The consolidated financial statements include the accounts of Cullen Agricultural Holding Corp. and CAT Merger Sub, Inc. All intercompany accounts and transactions have been eliminated upon consolidation.

#### **Going Concern Consideration**

The accompanying consolidated financial statements have been prepared on a going concern basis, which contemplates the realization of assets and the settlement of liabilities and commitments in the normal course of business. As reflected in the accompanying consolidated financial statements, the Company had a net loss of \$1,000 for the period from August 27, 2009 (inception) through August 31, 2009, and no cash as of August 31, 2009. The Company s funds may not be sufficient to maintain the Company. In addition, the Parent s certificate of incorporation provides for mandatory liquidation of the Parent in the event that the Parent des not complete a business combination prior to October 22, 2009. These factors raise substantial doubt about the Company s ability to continue as a going concern. The consolidated financial statements do not include any adjustments that might be necessary if it is unable to continue as a going concern.

#### **Cash and Cash Equivalents**

The Company considers all highly liquid investments purchased with an original maturity of three months or less to be cash equivalents. At August 31, 2009, the Company had no cash or cash equivalents.

#### **Use of Estimates**

The preparation of consolidated financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of expenses during the reporting period. Actual results could differ from those estimates.

#### **Organization Costs**

Organization costs of \$1,000 consist of legal and incorporation fees, incurred through the balance sheet date that are related to the formation of the corporations and were expensed as incurred in accordance with Statement of Position 98-5, Reporting on the Cost of Start-up Activities.

#### **Earnings per Share**

The Company follows the provisions of SFAS No. 128, Earnings Per Share . In accordance with SFAS No. 128, earnings per common share amounts ( Basic EPS ) are computed by dividing earnings by the weighted average number of common shares outstanding for the period. Earnings per common share amounts, assuming dilution ( Diluted EPS ), gives effect to dilutive options, warrants, and other potential common stock outstanding during the period. SFAS No.128 requires the presentation of both Basic EPS and Diluted EPS on the face of the statements of operations.

### CULLEN AGRICULTURAL HOLDING CORP. AND SUBSIDIARY

(A Development Stage Company)
(A Wholly Owned Subsidiary of Triplecrown Acquisition Corp.)

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

# Note 1 Organization, Business Operations, Significant Accounting Policies and Going Concern Consideration (continued)

#### **Income Taxes**

The Company uses the asset and liability method of accounting for income taxes in accordance with FAS 109, Accounting for Income Taxes . Under the asset and liability method, deferred tax assets and liabilities are recognized for the future tax consequences attributable to differences between the financial statement carrying amounts and the tax bases of existing assets and liabilities. A valuation allowance against deferred tax assets is provided when it is more likely than not that the deferred tax asset will not be fully realized. As of August 31, 2009, the Company has a net loss of \$1,000, which if realized would have a tax benefit of \$340. The Company has determined that this deferred tax benefit has no value at this time, as the Company does not believe that it will utilize these losses in the future, and accordingly has not been recorded as a deferred tax asset.

The Company adopted the provisions of FASB Interpretation No. 48 (FIN 48), on June 3, 2009 (inception). FIN 48 clarifies the accounting for uncertainty in income taxes recognized in an enterprise s financial statements in accordance with SFAS No. 109, Accounting for Income Taxes, and prescribes a recognition threshold and measurement process for financial statement recognition and measurement of a tax position taken or expected to be taken in a income tax return. FIN 48 also provides guidance on derecognition, classification, interest and penalties, accounting in interim periods, disclosure and transition. In accordance with FASB Staff Position (FSP) FIN 48-3, the Company is not required to reflect the adoption of FIN 48 in its combined financial statements until the issuance of its combined financial statements for the period ending December 31, 2009. The adoption of FIN 48 has not had a material effect on the Company s combined financial position or results of operations. The Company classifies interest and penalties, if any, associated with our uncertain tax positions as a component of income tax expense. No interest and penalties related to uncertain tax positions were accrued at August 31, 2009.

During the period from August 27, 2009 (inception) through August 31, 2009, the Company recognized no adjustments for uncertain tax benefits. The Company is subject to U.S. federal and state examinations by tax authorities since its inception. The Company does not expect any significant changes to its unrecognized tax positions during the next 12 months.

#### **Recent Accounting Pronouncements**

In December 2007, the FASB issued Statement of Financial Accounting Standards (SFAS) No. 141 (revised 2007), Business Combinations (SFAS 141R). SFAS 141R changes accounting for acquisitions that close beginning in 2009 in a number of areas including the treatment of contingent consideration, contingencies, acquisition costs, In-process research & development and restructuring costs. More transactions and events will qualify as business combinations and will be accounted for at fair value under the new standard. SFAS 141R promotes greater use of fair values in financial reporting. In addition, under SFAS 141R, changes in deferred tax asset valuation allowances and acquired income tax uncertainties in a business combination after the measurement period will impact income tax expense. Some of the changes will introduce more volatility into earnings. SFAS 141R is effective for fiscal years beginning on or after December 15, 2008. SFAS 141R will have an impact on accounting for any business acquired after the effective date of this pronouncement.

In December 2007, the FASB issued Statement of Financial Accounting Standards No. 160, Noncontrolling Interests in Consolidated Financial Statements, an amendment of ARB No. 51 (SFAS 160). SFAS 160 will change the accounting and reporting for minority interests, which will be recharacterized as noncontrolling interests (NCI) and classified as a component of equity. This new consolidation method will significantly change the accounting for transactions with minority interest holders. SFAS 160 is effective for

### CULLEN AGRICULTURAL HOLDING CORP. AND SUBSIDIARY

(A Development Stage Company)
(A Wholly Owned Subsidiary of Triplecrown Acquisition Corp.)

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

# Note 1 Organization, Business Operations, Significant Accounting Policies and Going Concern Consideration (continued)

fiscal years beginning after December 15, 2008. SFAS 160 would have an impact on the presentation and disclosure of the noncontrolling interests of any non-wholly owned business acquired in the future.

In April 2009, the FASB issued FASB Staff Position No. 141(R)-1 (FSP FAS 141(R)-1) which provides additional clarification on the initial recognition and measurement of assets acquired and liabilities assumed in a business combination that arise from contingencies. FSP FAS 141(R)-1 is effective for all fiscal years beginning on or after December 15, 2008. FSP FAS 141(R)-1 will have an impact on the accounting for any business acquired in the future.

In May 2009, the FASB issued SFAS No. 165, Subsequent Events (SFAS No. 165). The statement is to establish general standards of accounting for and disclosure of events that occur after the balance sheet date but before financial statements are issued. SFAS No. 165 became effective June 15, 2009 for all subsequent reporting periods. The adoption of SFAS No. 165 did not have a material impact on the Company s consolidated financial statements.

In June 2009, the FASB issued SFAS No. 167, Amendments to FASB Interpretation No. 46(R) (SFAS 167). SFAS 167 eliminates Interpretation 46(R) is exceptions to consolidating qualifying special-purpose entities, contains new criteria for determining the primary beneficiary, and increases the frequency of required reassessments to determine whether a company is the primary beneficiary of a variable interest entity. SFAS 167 also contains a new requirement that any term, transaction, or arrangement that does not have a substantive effect on an entity is status as a variable interest entity, a company is power over a variable interest entity, or a company is obligation to absorb losses or its right to receive benefits of an entity must be disregarded in applying Interpretation 46(R) is provisions. The elimination of the qualifying special-purpose entity concept and its consolidation exceptions means more entities will be subject to consolidation assessments and reassessments. SFAS 167 will be effective January 1, 2010. The adoption of SFAS 167 is not expected to have an impact on the Company is consolidated financial statements or results of operations.

In July 2009, the FASB issued SFAS No. 168, The FASB Accounting Standards Codification and the Hierarchy of Generally Accepted Accounting Principles a replacement of FASB Statement No. 162 (SFAS 168). SFAS 168 establishes the FASB Accounting Standards Codification to become the source of authoritative U.S. GAAP recognized by the FASB to be applied by non-governmental entities. Rules and interpretive releases of the Securities

#### Edgar Filing: Triplecrown Acquisition Corp. - Form PREM14A

and Exchange Commission (SEC) under authority of federal securities laws are also sources of authoritative GAAP for SEC registrants. On the effective date of this Statement, the Codification will supercede all then-existing non-SEC accounting and reporting standards. All other nongrandfathered non-SEC accounting literature not included in the Codification will become nonauthoritative. SFAS 168, which changes the referencing of financial standards, is effective for interim or annual financial periods ending after September 15, 2009. The adoption of SFAS 168 is not expected to materially impact the Company s consolidated financial statements or results of operations.

Management does not believe that any other recently issued, but not yet effective, accounting standards if currently adopted would have a material effect on the consolidated financial statements.

#### Note 2 Subsequent Events

Pursuant to the Agreement and Plan of Reorganization dated September 4, 2009, executed by Cullen Agricultural Technologies, Inc. ( Cullen Agritech ), Triplecrown, Cullen Agricultural Holdings, Inc. ( Holdco ), CAT Merger Sub, Inc. ( MergerSub ), and Cullen Inc Holdings Ltd ( CIL ), the parties to the agreement propose to merge Triplecrown into Holdco, with Holdco being the surviving public entity. Cullen Agritech will merge into MergerSub, a wholly-owned subsidiary of Holdco, and MergerSub shall cease to exist, with Cullen Agritech as the surviving subsidiary of Holdco.

As a result of the merger, the holders of

### CULLEN AGRICULTURAL HOLDING CORP. AND SUBSIDIARY

(A Development Stage Company)
(A Wholly Owned Subsidiary of Triplecrown Acquisition Corp.)

#### NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

#### Note 2 Subsequent Events (continued)

common stock and warrants of Triplecrown will receive like securities of Holdco, on a one-to-one basis, in exchange for their existing securities, except that 11,260,000 shares of the 13,800,000 shares of common stock owned by Triplecrown s founding shareholders will be cancelled. Cullen Agritech shares of common stock will be converted into 15,881,148 shares of Holdco s common stock. Upon conversion of the shares of Cullen Agritech s common stock, Holdco will receive 100 shares of Cullen Agritech s common stock. As a result of the merger, Cullen Agritech will be a wholly-owned subsidiary of Holdco.

Completion of the merger is subject to and conditioned upon, among other things, approval by the stockholders of Triplecrown and agreement by the holders of Triplecrown s warrants to increase the strike price of the warrants from \$7.50 per share to \$12.00 per share, to extend the expiration date to October 22, 2013 and to increase the price at which the stock must trade for the warrants to be called for redemption from \$13.75 per share to \$17.00 per share.

The assets of Holdco after completion of the merger will consist primarily of funds distributed from a trust account in which a substantial portion of the net proceeds of Triplecrown s initial public offering are held. Stockholders who vote against the merger and demand conversion of their shares will have the right to convert their shares into a pro rata portion of the funds held in the Triplecrown s trust account. In order to improve the probability that the merger is approved, Triplecrown may enter into agreements to purchase shares of common stock from shareholders who indicate their intention to vote against the merger, paying for the shares with funds from the trust account immediately after consummation of the merger. The funds available to Holdco following consummation of the merger, therefore, will be reduced based upon the number of stockholders who exercise their conversion rights or arrange to sell their shares to Triplecrown upon consummation of the merger.

In the event the proposed merger is not consummated, the parties to the merger agreement will bear their own legal and other costs.

Management has evaluated subsequent events to determine if events or transactions occurring through September 3, 2009, require potential adjustment or disclosure in the consolidated financial statements. Except as disclosed above, management concluded that no other events required disclosure in these consolidated financial statements.

Edgar Filing: Triplecrown Acquisition Corp. - Form PREM14A

### REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Audit Committee of the Board of Directors and Stockholders of Triplecrown Acquisition Corp.

We have audited the accompanying balance sheets of Triplecrown Acquisition Corp. (a development stage enterprise) (the Company ) as of December 31, 2008 and 2007, and the related statements of operations, changes in stockholders equity and cash flows for the year ended December 31, 2008, and for the periods from June 8, 2007 (inception) through December 31, 2008 and December 31, 2007. These financial statements are the responsibility of the Company s management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit also includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audit provides a reasonable basis for our opinion.

The accompanying financial statements have been prepared assuming that the Company will continue as a going concern. As discussed in Note 1 to the financial statements, the Company s certificate of incorporation provides for mandatory liquidation of the Company in the event that the Company does not consummate a business combination (as defined) prior to October 22, 2009. These factors raise substantial doubt about its ability to continue as a going concern. Management s plans regarding those matters are described in Note 1. The financial statements do not include any adjustments that might result from the outcome of this uncertainty.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of Triplecrown Acquisition Corp. (a development stage enterprise) as of December 31, 2008 and 2007, and the results of its operations and its cash flows for the year ended December 31, 2008 and for the periods from June 8, 2007 (inception) through December 31, 2008 and 2007, in conformity with United States generally accepted accounting principles.

We have also audited, in accordance with the standards of the Public Company Accounting Oversight Board (Unites States), Triplecrown Acquisition Corp s internal control over financial reporting as of December 31, 2008, based on the criteria established in Internal Control-Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission and our report dated March 12, 2009 expressed an unqualified opinion on the effectiveness of the Company s internal control over financial reporting.

/s/ Marcum & Kliegman LLP

Marcum & Kliegman LLP Melville, New York March 12, 2009

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

#### **BALANCE SHEETS**

ASSETS	December 31, 2008	December 31, 2007
Current Assets		
Cash and cash equivalents	\$111,606	\$296,913
Cash held in trust account, interest and dividends available for working	7,000	+ -> -> ->
capital and taxes (including accrued interest receivable of \$1,539,038 as	4,707,238	3,178,633
of December 31, 2007)		
Other current assets	91,067	21,370
Total current assets	4,909,911	3,496,916
Trust account, restricted		
Cash held in trust account	538,747,464	536,930,000
Accrued interest receivable, trust account	129,192	
Total trust account, restricted	538,876,656	536,930,000
Total assets	\$543,786,567	\$540,426,916
LIABILITIES AND STOCKHOLDERS EQUITY		
Current Liabilities		
Accounts payable and accrued expenses	\$404,682	\$135,400
Income taxes payable		1,568,439
Total liabilities	404,682	1,703,839
Common Stock, subject to possible conversion, 16,559,999 shares at conversion value	161,662,987	161,078,990
Commitment and Contingencies		
Stockholders Equity		
Preferred stock, \$.0001 par value, authorized 1,000,000 shares; none		
issued		
Common stock, \$.0001 par value, authorized 160,000,000 shares; total		
issued and outstanding 69,000,000 shares, (less 16,559,999 shares,	5,244	5,244
subject to possible conversion)		
Additional paid-in capital	375,230,449	375,814,446
Income accumulated during development stage	6,483,205	1,824,397
Total stockholders equity	381,718,898	377,644,087
Total liabilities and stockholders equity	\$543,786,567	\$540,426,916

The accompanying notes are an integral part of these financial statements

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

#### STATEMENTS OF INCOME

		For the	For the
	For the	Period from	Period from
	Year Ended	June 8, 2007	June 8, 2007
		(Inception)	(Inception)
	December 31, 2008	Through	Through
	2008	December 31,	December 31,
		2007	2008
Revenue	\$	\$	\$
Formation and operating costs	1,484,138	180,350	1,664,488
Loss from operations	(1,484,138)	(180,350)	(1,664,488)
Interest and dividend income	8,443,326	3,573,186	12,016,512
Income before provision for income taxes	6,959,188	3,392,836	10,352,024
Provision for income taxes	(2,300,380)	(1,568,439)	(3,868,819)
Net income	4,658,808	1,824,397	6,483,205
Accretion of trust account income relating to common stock subject to possible conversion	(583,997)		(583,997)
Net income attributable to other common stockholders	\$4,074,811	\$1,824,397	\$5,899,208
Weighted average number of common shares outstanding			
excluding shares subject to possible conversion basic and	52,440,001	26,554,952	
diluted			
Basic and diluted net income per share attributable to other common stockholders	\$0.08	\$0.07	

The accompanying notes are an integral part of these financial statements

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

### STATEMENT OF CHANGES IN STOCKHOLDERS EQUITY

### For the Period from June 8, 2007 (Inception) Through December 31, 2008

	Common Sto	ock	Additional Paid-In	Income Accumulated During	Stockholders
	Shares	Amount	Capital	Developmer Stage	ntEquity
Balance, June 8, 2007 (inception)		\$	\$	\$	\$
Issuance of stock to initial stockholders at \$0.002 per share Sale of 55,200,000 units, net of	13,800,000	1,380	23,620		25,000
underwriters discount and offering expenses of \$20,126,320 (includes 16,559,999 shares subject to possible conversion)	55,200,000	5,520	531,868,160		531,873,680
Proceeds subject to possible conversion of 16,559,999 shares		(1,656)	(161,077,334)		(161,078,990)
Proceeds from issuance of sponsor warrants			5,000,000		5,000,000
Net income for the period June 8, 2007 (inception) through December 31, 2007				1,824,397	1,824,397
Balance, December 31, 2007	69,000,000	5,244	375,814,446	1,824,397	377,644,087
Accretion of trust account income relating to common stock subject to possible conversion for the year ended December 31, 2008			(583,997 )		(583,997 )
Net income for the year ended December 31, 2008				4,658,808	4,658,808
Balance, December 31, 2008	69,000,000	\$5,244	\$375,230,449	\$6,483,205	\$381,718,898

The accompanying notes are an integral part of these financial statements



### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

#### STATEMENT OF CASH FLOWS

	For Year Ended December 31, 2008	For the Period from June 8, 2007 (Inception) Through December 31, 2007	For the Period from June 8, 2007 (Inception) Through December 31, 2008
Cash flows from operating activities			
Net income	\$4,658,808	\$1,824,397	\$6,483,205
Adjustments to reconcile net income to cash provided			
by operating activities			
Changes in operating assets and liabilities:	(60 60 <b>=</b> )	(21.250	(0.1.0.CT
Other current assets	(69,697)	(21,370 )	(91,067)
Income taxes payable	(1,568,439)	1,568,439	40.4.602
Accounts payable and accrued expenses	269,282	135,400	404,682
Net cash provided by operating activities	3,289,954	3,506,866	6,796,820
Cash flows from investing activities Cash held in trust account, restricted	(2.075.040)	(536,930,000)	(539,005,848)
Cash held in trust account, restricted  Cash held in trust account interest and dividend	(2,075,848)	(330,930,000)	(339,003,646)
income available for working capital and taxes	(1,399,413)	(3,178,633 )	(4,578,046 )
Net cash used in investing activities	(3,475,261)	(540,108,633)	(543,583,894)
Cash flows from financing activities	(, , , ,	, , , ,	
Proceed from issuance of stock to initial stockholders		25,000	25,000
Proceeds from notes payable, stockholders		112,500	112,500
Repayments of notes payable, stockholders		(112,500)	(112,500)
Gross proceeds from issuance of sponsors warrants		5,000,000	5,000,000
Gross proceeds from initial public offering and		552,000,000	552,000,000
over-allotments options		332,000,000	332,000,000
Payment of offering costs		(20,126,320)	(20,126,320)
Net cash provided by financing activities		536,898,680	536,898,680
Net (decrease) increase in cash	(185,307)	296,913	111,606
Cash at beginning of the period	296,913		
Cash at end of the period	\$111,606	\$296,913	\$111,606
Supplemental disclosure of cash flow information			
Cash paid during the period for:			
Income taxes	\$3,995,000	\$	\$3,995,000

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

#### NOTES TO FINANCIAL STATEMENTS

### Note 1 Organization, Business Operations, Significant Accounting Policies and Going Concern Consideration

Triplecrown Acquisition Corp. (the Company) was incorporated in Delaware on June 8, 2007 as a blank check company to serve as a vehicle to effect a merger, capital stock exchange, asset acquisition, stock purchase, reorganization or other similar business combination with one or more operating businesses in the financial services industry (Business Combination).

All activity from June 8, 2007 (inception) through October 25, 2007 relates to the Company s formation and the Company s initial public offering, (Offering) described below. Since October 26, 2007, the Company has been searching for a target business to acquire. The Company has selected December 31 as its fiscal year end.

The registration statement for the Company s initial public offering (Offering) was declared effective October 22, 2007. The Company consummated the Offering on October 25, 2007 and received gross proceeds of \$552,000,000 and \$5,000,000 from the sale of Sponsors Warrants on a private placement basis (see Note 2). The Company s management has broad discretion with respect to the specific application of the net proceeds of this Offering, although substantially all of the net proceeds of the Offering are intended to be generally applied toward consummating a Business Combination. The Business Combination must be with a target business that has a fair market value of at least 80% of the balance in the trust account (excluding deferred underwriting discounts and commissions) at the time of such acquisition and operate in the financial services industry. Furthermore, there is no assurance that the Company will be able to successfully effect a Business Combination. An amount of \$536,930,000 (or approximately \$9.73 per unit) of the net proceeds of the Offering and the sale of the Sponsors Warrants (see Note 2) is being held in a trust account ( Trust Account ) and is invested in United States government securities within the meaning of Section 2(a) (16) of the Investment Company Act of 1940 having a maturity of 180 days or less or in money market funds meeting certain conditions under Rule 2a-7 promulgated under the Investment Company Act of 1940 until the earlier of (i) the consummation of its initial Business Combination or (ii) liquidation of the Company. The placing of funds in the Trust Account may not protect those funds from third party claims against the Company. Although the Company will seek to have all vendors, prospective target businesses or other entities it engages, execute agreements with the Company waiving any right, title, interest or claim of any kind in or to any monies held in the Trust Account, there is no guarantee that they will execute such agreements.

The Company s officers have agreed that they will be personally liable under certain circumstances to ensure that the proceeds in the Trust Account are not reduced by the claims of target businesses or vendors or other entities that are owed money by the Company for services rendered, contracted for or products sold to the Company. However, there can be no assurance that they will be able to satisfy those obligations. The remaining net proceeds (not held in the Trust Account) may be used to pay for business, legal and accounting due diligence on prospective acquisitions and continuing general and administrative expenses. Except with respect to interest income that may be released to the Company of (i) up to \$6,000,000 of interest accrued on the amount held in the Trust Account (plus taxes, if any,

payable by the Company with respect to such interest) will be released by the Company in monthly installments to fund expenses related to investigating and selecting a target business and the Company s other working capital requirements and (ii) any additional amounts needed to pay income or other tax obligations. The proceeds held in trust will not be released from the trust account until the earlier of the completion of a Business Combination or the Company s liquidation.

The Company, after signing a definitive agreement for a Business Combination with a target business or businesses, is required to submit such transaction for stockholder approval. In the event that the stockholders owning 30% or more of the shares sold in the Offering vote against the Business Combination and exercise their conversion rights described below, the Business Combination will not be consummated. All of the Company s stockholders prior to the Offering, including all of the officers and directors of the Company (Initial Stockholders) have agreed to vote all of their founding shares of common stock in accordance with the vote

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

#### NOTES TO FINANCIAL STATEMENTS

# Note 1 Organization, Business Operations, Significant Accounting Policies and Going Concern Consideration (continued)

of the majority in interest of all other stockholders of the Company ( Public Stockholders ) with respect to any Business Combination. After consummation of a Business Combination, these voting safeguards will no longer apply.

With respect to a Business Combination which is approved and consummated, any Public Stockholder who voted against the Business Combination may demand that the Company convert his or her shares into cash from the Trust Account. The per share conversion price will equal the amount in the Trust Account, calculated as of two business days prior to the consummation of the proposed Business Combination, divided by the number of shares of common stock held by Public Stockholders at the consummation of the Offering. Accordingly, Public Stockholders holding 30% less one share of the aggregate number of shares owned by all Public Stockholders may seek conversion of their shares in the event of a Business Combination. Such Public Stockholders are entitled to receive their per share interest in the Trust Account computed without regard to the shares held by Initial Stockholders.

The Company's Certificate of Incorporation was amended on October 22, 2007 to provide that the Company will continue in existence only until 24 months from the effective date of the registration statement relating to the Offering (Effective Date), or October 22, 2009. If the Company has not completed a Business Combination by such date, its corporate existence will cease except for the purposes of liquidating and winding up its affairs. In the event of liquidation, it is possible that the per share value of the residual assets remaining available for distribution (including Trust Account assets) will be less than the initial public offering price per Unit in the Offering.

#### Going Concern and Management s Plan and Intentions

As of December 31, 2008, excluding Cash held in the Trust Account-restricted of \$538,876,656, the Company had working capital of \$4,505,229. The Company s only source of income, to enable it to continue to fund its search for an acquisition candidate, is the interest and dividends it earns on its cash held in the Trust Account. These funds may not be sufficient to maintain the Company until a Business Combination is consummated. Pursuant to its Certificate of Incorporation, if the Company is unable to consummate a timely Business Combination prior to October 22, 2009, the Company would have to liquidate pursuant and return the funds held in the Trust Account to the holders of shares issued in the Offering. There can be no assurance that the Company will enter into a Business Combination prior to October 22, 2009. These factors raise substantial doubt about the Company s ability to continue as a going concern. The audited financial statements do not include any adjustments that might result from the outcome of these uncertainties.

#### **Cash and Cash Equivalents**

The Company considers all highly liquid instruments with maturities of three months or less to be cash equivalents.

#### **Earnings Per Share**

The Company follows the provisions of Statement of Financial Accounting Standards (SFAS) No. 128, Earnings Per Share. In accordance with SFAS No. 128, earnings per common share amounts (Basic EPS) were computed by dividing earnings by the weighted average number of common shares outstanding for the period. Common shares subject to possible conversion of 16,559,999 have been excluded from the calculation of basic earnings per share since such shares, if redeemed, only participate in their pro rata shares of the trust earnings, such earnings as applicable are accrued to the value of the common stock subject to conversion. Earnings per common share amounts, assuming dilution (Diluted EPS), gives effect to dilutive options, warrants, and other potential common stock outstanding during the period. SFAS No. 128 requires the presentation of both Basic EPS and Diluted EPS on the face of the statements of operations. The effect of the 74,000,000 outstanding Warrants issued in connection with the Public Offering and the Private Placement

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

#### NOTES TO FINANCIAL STATEMENTS

# Note 1 Organization, Business Operations, Significant Accounting Policies and Going Concern Consideration (continued)

described in Note 2 has not been considered in the diluted earnings per share calculation since the exercise of the Warrants are contingent upon the occurrence of future events, and therefore, is not includable in the calculation of diluted earnings per share in accordance with SFAS 128.

#### **Concentration of Credit Risk**

SFAS No. 105, Disclosure of Information about Financial Instruments with Off-Balance Sheet Risk and Financial Instruments with Concentration of Credit Risk, requires disclosure of significant concentrations of credit risks regardless of the degree of risk. At December 31, 2008, financial instruments that potentially expose the Company to credit risk consist of cash. The Company maintains its cash balances in a U.S. Treasury-only money market fund at JPMorgan Private Bank. At times, the Company s cash and cash equivalents may be uninsured or in deposit accounts that exceed the Federal Deposit Insurance Corporation and the Securities Investor Protection Corporation insurances

#### **Fair Value of Financial Instruments**

The carrying value of cash, cash held in Trust Account and accrued expenses are reasonable estimates of the fair values due to their short-term maturity.

#### Cash Held in Trust Account-Restricted

The Company considers the restricted portion of the funds held in the Trust Account as being a non-current asset. A current asset is one that is reasonably expected to be used to pay current liabilities, such as accounts payable or short-term debt or to pay current operating expenses, or will be used to acquire other current assets. Since the acquisition of a business is principally considered to be a long-term purpose, with long-term assets such as property and intangibles typically being a major part of the acquired assets, the Company has reported the funds anticipated to be used in the acquisition as a non-current asset. In addition, the Company has recorded interest receivables of \$129,192 as a non-current asset as December 31, 2008. The Trust Account has earned more than \$6 million, which is the maximum amount allowance to be used for working capital purposes.

#### Accretion of Trust Account Relating to Common Stock Subject to Possible Conversion

The Company records accretion of the income earned in the Trust Account relating to the common stock subject to possible conversion based on the excess of the earnings for the period over the amount which is available to be used for working capital and taxes. Since 30% (less one share) of the shares issued in the Offering are subject to possible conversion, the portion of the excess earnings related to those shares are reflected on the balance sheet as part of Common stock subject to possible conversion and is deducted from Additional paid-in capital. The portion of the excess earnings is also presented as a deduction from net income on the Statements of Operations to appropriately reflect the amount of net income which would remain available to the common stockholders who did not elect to convert their shares to cash.

#### Reclassifications

Certain prior year amounts have been reclassified to conform to the current year presentation.

#### **Use of Estimates**

The preparation of financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions that affect the reported amounts of assets, liabilities and disclosures of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

#### NOTES TO FINANCIAL STATEMENTS

# Note 1 Organization, Business Operations, Significant Accounting Policies and Going Concern Consideration (continued)

#### **Income Taxes**

Deferred income taxes, if applicable, are provided for the differences between the basis of assets and liabilities for financial reporting and income tax purposes. A valuation allowance is established when necessary to reduce deferred tax assets to the amount expected to be realized. Management did not record the impact of deferred income taxes as they were deemed immaterial.

#### **Recently Issued Accounting Pronouncements**

In September 2006, the Financial Accounting Standards Board (FASB) issued Statement of Financial Accounting Standards (SFAS) No.157, Fair Value Measurements, which is effective for fiscal years beginning after November 15, 2007. The Statement defines fair value, establishes a frame work for measuring fair value in accordance with Generally Accepted Accounting Principles, and expands disclosures about fair value measurements. The Statement codifies the definition of fair value as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. The standard clarifies the principle that fair value should be based on the assumptions market participants would use when pricing the asset or liability and establishes a fair value hierarchy that prioritizes the information used to develop those assumptions. The adoption did not have a material impact on its financial position and results of operations.

In February 2007, the FASB issued SFAS No. 159 The Fair Value Opinion for Financial Assets and Financial Liabilities Including an amendment of FASB Statement No. 115 (SFAS No. 159), which permits entities to choose to measure many financial instruments and certain other items at fair value. The fair value option established by this Statement permits all entities too choose to measure eligible items at fair value at specified election dates. A business entity shall report unrealized gains and losses on items for which the fair value option has been elected in earnings at each subsequent reporting date. Adoption is required for fiscal years beginning after November 15, 2007. Early adoption is permitted as of the beginning of a fiscal year that begins on or before November 15, 2007, provided the entity also elects to apply the provisions of SFAS No. 159. The adoption did not have a material impact on its financial position and results of operations.

In December 2007, the FASB issued SFAS No. 141 (revised 2007) Business Combinations (SFAS 141R). SFAS 141R changes accounting for acquisitions that close beginning in 2009 in a number of areas including the treatment of contingent consideration, contingencies, acquisition costs, in-process research and development and restructuring costs. More transactions and events will qualify as business combinations and will be accounted for at fair value under

the new standard. SFAS 141R promotes greater use of fair values in financial reporting. In addition, under SFAS 141R, changes in deferred tax asset valuation allowances and acquired income tax uncertainties in a business combination after the measurement period will impact income tax expense. Some of the changes will introduce more volatility into earnings. SFAS 141R is effective for fiscal years beginning on or after December 15, 2008. SFAS 141R will have an impact on accounting for any business acquired in the future.

In December 2007, the FASB issued SFAS No. 160 Noncontrolling Interests in Consolidated Financial Statements, an amendment of ARB No. 51 (SFAS 160). SFAS 160 will change the accounting and reporting for minority interests, which will be recharacterized as noncontrolling interests (NCI) and classified as a component of equity. This new consolidation method will significantly change the accounting for transactions with minority interest holders. SFAS 160 is effective for fiscal years beginning after December 15, 2008. SFAS 160 would have an impact on the presentation and disclosure of the noncontrolling interests of any non-wholly owned business acquired in the future.

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

#### NOTES TO FINANCIAL STATEMENTS

# Note 1 Organization, Business Operations, Significant Accounting Policies and Going Concern Consideration (continued)

In December 2008, the FASB issued FSP SFAS No. 140-4 and FASB Interpretation Number (FIN) No. 46R-8, Disclosures by Public Entities (Enterprises) about Transfers of Financial Assets and Interests in Variable Interest Entities (FSP SFAS No. 140-4 and FIN No. 46R-8). This statement increases the disclosure requirements regarding continuing involvement with financial assets that have been transferred, as well as the company s involvement with variable interest entities. The FSP is effective for financial statements issued for interim periods ending after Dec. 15, 2008. Accordingly, The Company had adopted FSP SFAS 140-4 and FIN No. 46R-8. The adoption of this pronouncement is not expected to have a significant effect on its financial statements.

In February 2008, the FASB issued FASB Staff Positions (FSP) No. 157-1 Application of FASB Statement No. 157 to FASB Statement No. 13 and Other Pronouncements That Address Fair Value Measurements for Purposes of Lease Classification or Measurement Under Statement 13, and No. 157-2 ( Effective Date of FASB Statement No. 157, which remove leasing transactions from the scope of SFAS No. 157 and defer its effective date for one year relative to certain nonfinancial assets and liabilities. As a result, the application of the definition of fair value and related disclosures of SFAS No. 157 (as impacted by these two FSP s) was effective for the Company beginning January 1, 2008 on a prospective basis with respect to fair value measurements of (a) nonfinancial assets and liabilities that are recognized or disclosed at fair value in the Company s financial statements on a recurring basis (at least annually) and (b) all financial assets and liabilities. This adoption did not have a material impact on the Company s results of operations or financial condition. The remaining aspects of SFAS No. 157 for which the effective date was deferred under FSP No. 157-2 are currently being evaluated by the company. Areas impacted by the deferral relate to nonfinancial assets and liabilities that are measured at fair value, but are recognized or disclosed at fair value on a nonrecurring basis. This deferral applies to such items as nonfinancial assets and liabilities initially measured at fair value in a business combination (but not measured at fair value in subsequent periods) or nonfinancial long-lived asset groups measured at fair value for an impairment assessment. The effects of these remaining aspects of SFAS No. 157 are to be applied to fair value measurements prospectively beginning January 1, 2009. The Company does not expect them to have a material impact on the Company s results of operations or financial condition. In October 2008, the FASB issued FSP No. 157-3, Determining the Fair Value of a Financial Asset When the Market for That Asset is Not Active (FSP 157-3). FSP 157-3 clarifies the application of SFAS 157, which the Company adopted as of January 1, 2008, in cases where a market is not active. The Company has considered FSP 157-3 in its determination of estimated fair values as of December 31, 2008, and the impact was not material.

In May 2008, the FASB issued SFAS No. 162, The Hierarchy of Generally Accepted Accounting Principles. The statement is intended to improve financial reporting by identifying a consistent hierarchy for selecting accounting principles to be used in preparing financial statements that are prepared in conformance with generally accepted accounting principles. Unlike Statement on Auditing Standards (SAS) No. 69, The Meaning of Present Fairly in

Conformity With GAAP, SFAS No. 162 is directed to the entity rather than the auditor. The statement is effective 60 days following the SEC s approval of the Public Company Accounting Oversight Board (PCAOB) amendments to AU Section 411, The Meaning of Present Fairly in Conformity with GAAP, and is not expected to have any impact on the Company s results of operations, financial condition or liquidity.

In June 2008, FASB issued FSP Emerging Issues Task Force (EITF) No. 03-6-1, Determining Whether Instruments Granted in Share-Based Payment Transactions Are Participating Securities. Under the FSP, unvested share-based payment awards that contain rights to receive nonforfeitable dividends (whether paid or unpaid) are participating securities, and should be included in the two-class method of computing EPS. The

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

#### NOTES TO FINANCIAL STATEMENTS

# Note 1 Organization, Business Operations, Significant Accounting Policies and Going Concern Consideration (continued)

FSP is effective for fiscal years beginning after December 15, 2008, and interim periods within those years, and is not expected to have a significant impact on the Company s results of operations, financial condition or liquidity.

Management does not believe that any other recently issued, but not yet effective, accounting standards if currently adopted would have a material effect on the accompanying financial statements.

#### Note 2 Initial Public Offering

On October 25, 2007, the Company sold 55,200,000 Units, including 7,200,000 units from the exercise of the underwriters over-allotment option, at a offering price of \$10.00 per Unit. Each Unit consists of one share of the Company s common stock, \$.0001 par value, and one Redeemable Common Stock Purchase Warrant ( Warrant ). Each Warrant entitles the holder to purchase from the Company one share of common stock at an exercise price of \$7.50 commencing the later of the completion of a Business Combination or October 22, 2008 and expiring October 22, 2012. The Company may redeem the Warrants, at a price of \$0.01 per Warrant upon 30 days notice after the Warrants become exercisable, only in the event that the last sale price of the common stock is at least \$13.75 per share for any 20 trading days within a 30 trading day period ending on the third day prior to the date on which the notice of redemption is given. In accordance with the warrant agreement relating to the Warrants sold and issued in the Offering, the Company is only required to use its best efforts to maintain the effectiveness of the registration statement covering the Warrants. The Company will not be obligated to deliver securities, and there are no contractual penalties for failure to deliver securities, if a registration statement is not effective at the time of exercise. Additionally, in the event that a registration is not effective at the time of exercise, the holder of such Warrant shall not be entitled to exercise such Warrant and in no event (whether in the case of a registration statement not being effective or otherwise) will the Company be required to settle the warrant exercise, whether by net cash settlement or otherwise. Consequently, the Warrants may expire unexercised, unredeemed and worthless, and an investor in the Offering may effectively pay the full Unit price solely for the shares of common stock included in the units (since the Warrants may expire worthless).

On October 25, 2007, pursuant to Subscription Agreements, dated July 10, 2007, Eric J. Watson, the Company s Chairman of the Board and Treasurer and Jonathan J. Ledecky, the Company s President and Secretary purchased from the Company, in the aggregate, 5,000,000 warrants for \$5,000,000 (the Sponsors Warrants ). The purchase and issuance of the Sponsors Warrants occurred simultaneously with the consummation of the Offering on a private placement basis. The Company believes the purchase price of these warrants approximates the fair value of such warrants. All of the proceeds the Company received from these purchases were placed in the Trust Account. The

Sponsors Warrants are identical to the Warrants underlying the Units in the Offering except that if the Company calls the Warrants for redemption, the Sponsors Warrants will be exercisable on a cashless basis and will not be redeemable by the Company so long as they are still held by the original purchasers or their affiliates. If the Company does not complete a Business Combination then the \$5,000,000 will be part of the liquidating distribution to the Company s public stockholders, and the Sponsors Warrants will expire worthless. The purchasers of the Sponsors Warrants have agreed that the Sponsors Warrants will not be sold or transferred by them until after the Company has completed a business combination.

The Company s Initial Stockholders have waived their rights to participate in any liquidation distribution, but only with respect to those shares of common stock owned by them prior to the Offering; they will participate in any liquidation distribution with respect to any shares of common stock acquired in connection with or following the Offering.

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

#### NOTES TO FINANCIAL STATEMENTS

#### Note 3 Offering Costs

Offering costs incurred through October 25, 2007 were charged to capital at the time of the closing of the Offering.

#### Note 4 Notes Payable, Stockholders

The Company issued two unsecured promissory notes for \$62,500 and \$50,000 (a total of \$112,500) to two Initial Stockholders, who are also officers and directors of the Company, as of June 18, 2007. The notes are non-interest bearing and became payable upon the consummation of the Offering. These notes were repaid as of November 20, 2007.

#### Note 5 Commitments and Contingencies

The Company entered into an agreement with the underwriters of the Offering (the Underwriting Agreement ). The Underwriting Agreement requires the Company to pay 3.5% of the gross proceeds of the Offering as an underwriting discount plus an additional 3.5% of the gross proceeds only upon consummation of a Business Combination. The Company paid an underwriting discount of 3.5% of the gross proceeds (\$19,320,000) in connection with the consummation of the Offering and has placed 3.5% of the gross proceeds (\$19,320,000) in the Trust Account. The Company did not pay any discount related to the Sponsors Warrants sold in the private placement. The underwriters have waived their right to receive payment of the 3.5% of the gross proceeds upon the Company s liquidation if it is unable to complete a Business Combination.

The Initial Stockholders and holders of the Sponsors Warrants (or underlying securities) will be entitled to registration rights with respect to their Founders Units or Sponsors Warrants (or underlying securities), as the case may be, pursuant to an agreement dated October 22, 2007. The holders of the majority of the Founders Units (and underlying securities) are entitled to demand that the Company register the resale of these securities at any time commencing nine months after the consummation of a Business Combination. The holders of the Sponsors Warrants (or underlying securities) are entitled to demand that the Company register such securities at any time after the Company consummates a Business Combination. The sponsors warrants are not transferable or salable by the purchasers (subject to limited exceptions including the transferee agreeing to be bound to such transfer restrictions) until the Company completes a business combination, and are exercisable on a cashless basis and will be non-redeemable by the Company so long as they are held by the purchasers or their affiliates. In addition, the Initial Stockholders and holders of the Sponsors Warrants (or underlying securities) have certain piggy-back registration rights on registration statements filed after the Company s consummation of a Business Combination.

The Company hired a consultant on March 20, 2008 to perform services through the earlier of March 20, 2009 or completion of a Business Combination. The Company will pay the consultant a base salary of £33,000 (approximately \$50,000) per year. The Company will also issue 25,000 shares of the Company s common stock to this consultant upon

the closing of the initial Business Combination by the Company.

The Company engaged Petrina Advisors, Inc. on May 19, 2008 to perform consulting services through the earlier of October 22, 2009 or upon the closing of a Business Combination. The Company will pay this consultancy firm a base fee of \$192,000 per year and the Company will issue to them 100,000 shares of the Company s common stock upon the closing of the initial Business Combination by the Company.

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

#### NOTES TO FINANCIAL STATEMENTS

#### Note 6 Preferred Stock

The Company is authorized to issue 1,000,000 shares of preferred stock with such designations, voting and other rights and preferences as may be determined from time to time by the Board of Directors.

The underwriting agreement prohibits the Company, prior to a Business Combination, from issuing preferred stock which participates in the proceeds of the Trust Account or which votes as a class with the common stock on a Business Combination.

#### Note 7 Common Stock

On September 19, 2007, the Company s Board of Directors authorized a dividend of one warrant to purchase one share of common stock for each outstanding share of founders common stock. As a result, the Initial Stockholders effectively hold units (Founders Units).

Effective October 22, 2007, the Company s Board of Directors authorized a unit dividend of 0.2 units for each outstanding unit. On October 22, 2007, the Company s Board of Directors authorized an amendment to the Company s Certificate of Incorporation to increase the authorized shares of common stock from 75,000,000 shares of common stock with a par value of \$.0001 to 160,000,000 shares of common stock with a par value of \$.0001. All references in the accompanying financial statements as of December 31, 2008 and for the period June 8, 2007 (inception) to December 31, 2008 to the number of shares of common stock have been retroactively restated to reflect this transaction.

As of December 31, 2008, there were 74,000,000 shares of common stock outstanding for issuance upon exercise of warrants and the Sponsors Warrants.

#### Note 8 Income Taxes

Effective June 8, 2007, the Company adopted the provisions of the Financial Accounting Standards Board (FASB) Interpretation No. 48, Accounting for Uncertainty in Income Taxesem an interpretation of FASB Statement No. 109 (FIN 48). FIN 48 prescribes a recognition threshold and a measurement attribute for the financial statement recognition and measurement of tax positions taken or expected to be taken in a tax return. For those benefits to be recognized, a tax position must be more-likely-than-not to be sustained upon examination by taxing authorities. FIN 48 also provides guidance on derecognition, classification, interest and penalties, accounting in interim period, disclosure and transition.

The Company has identified its federal tax return and its state tax return in Wyoming as its major tax jurisdiction, as defined. Based on the Company s evaluation, it has been concluded that there are no significant uncertain tax positions

requiring recognition in the Company s financial statements. Since the Company was incorporated on June 8, 2007 the evaluation was performed for the 2008 and 2007 tax years. The Company believes that its income tax positions and deductions would be sustained on audit and does not anticipate any adjustments that would result in a material change to its financial position. No liability for unrecognized tax benefits was required to be reported as at December 31, 2008 and 2007.

The Company s policy for recording interest and penalties associated with audits is to record such items as a component of income tax expense. There were no amounts accrued for penalties or interest as of or during the period from June 8, 2007 (inception) through December 31, 2007 and for the year ended December 31, 2008. The Company does not expect its unrecognized tax benefit position to change during the next twelve months. Management is currently unaware of any issues under review that could result in significant payments, accruals or material deviations from its position. The adoption of the provisions of FIN 48 did not have a material impact on the Company s financial position, results of operations and cash flows.

Deferred income taxes reflect the net tax effects of operating losses and other temporary differences between carrying amounts of assets and liabilities for financial reporting purposes and the amounts used for income tax purposes. In assessing the realizability of deferred tax assets, management considers whether it is

F-33

Note 8 Income Taxes 305

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

#### NOTES TO FINANCIAL STATEMENTS

Note 8 Income Taxes (continued)

more likely than not that some portion or all of the deferred tax assets will not be realized. The ultimate realization of deferred tax assets is dependent upon the generation of future taxable income during the periods in which temporary differences representing net future deductible amounts become deductible. SFAS No. 109 requires that a valuation allowance be established when it is more likely than not that all or a portion of deferred tax assets will not be realized. A review of all available positive and negative evidence needs to be considered, including a company s performance, the market environment in which the company operates, the length of carryback and carryforward periods, and expectations of future profits.

Corporate taxes payable as of December 31, consisted of the following:

Federal income tax

2008

\$ 1,568,439

\$ 1,568,439

The provision for income tax consists of the following:

For the Year Ended December 31, 2008 For the Period from June 8, 2007 (Inception) Through December 31,

2007

Current:

Federal \$2,300,380 \$1,568,439

State and Local

Deferred

Federal

State and Local

Total provision for income taxes \$2,300,380 \$1,568,439

A reconciliation between the effective rate for income taxes and the amount computed by applying the statutory Federal income tax rate to income (loss) from continuing operations before provision for income taxes is as follows:

	For the	For the
	Year	Period from
	Ended	June 8, 2007
	December	(Inception)
	31,	Through
	2008	December
		31,
		2007
Tax provision at statutory rate	34 %	34 %
Personal Holding Company taxes (net of federal tax benefit)		10 %
Other	(1)%	2 %
	33 %	46 %

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

#### **NOTES TO FINANCIAL STATEMENTS**

#### Note 9. Unaudited Quarterly Financial Information

Following is a summary of the quarterly results of operations for the year ended December 31, 2008.

	For the Three	For the Three	For the Three	For the Three
	Months Ended	Months Ended	Months Ended	Months Ended
	March 31,	June 30,	September 30,	December 31,
	2008	2008	2008	2008
Revenue	\$	\$	\$	\$
Operating Loss	(362,730 )	(319,209)	(343,403)	(458,796)
Interest Income	3,530,645	1,976,161	2,071,917	864,603
Income (Loss) before Provision for Income	3,167,915	1 656 052	1,728,514	405 907
Taxes	3,107,913	1,656,952	1,728,314	405,807
Provision for Income Taxes	1,002,120	567,744	591,632	138,884
Net Income (Loss)	2,165,795	1,089,208	1,136,882	266,923
Accretion of Trust Account Relating to				
Common Stock subject to Possible			(378,787)	(205,210 )
conversion				
Net Income (Loss) Available (attributable)	\$2,165,795	\$1,089,208	\$758,095	61,713
to Common Stockholders	\$2,105,795	\$1,009,200	\$ 750,095	01,713
Weighted Average Shares Outstanding	52,440,001	52,440,001	52,440,001	52,440,001
Basic and Diluted	32,440,001	32,440,001	32,440,001	32,440,001
Basic and Diluted Net Income (Loss) Per	\$.04	.02	.01	.00
Share	ψ.0Τ	.02	.01	.00

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

#### **CONDENSED BALANCE SHEETS**

	June 30, 2009 (Unaudited)	December 31, 2008
ASSETS		
Current assets		
Cash and cash equivalents	\$	\$111,606
Cash held in trust account, interest and dividend income available for		
working capital and taxes (including prepaid income taxes of \$360,139	2,856,870	4,707,238
and \$126,181 in 2009 and 2008, respectively)		
Deposit on contract	866,250	
Other current assets	366,511	91,067
Total current assets	4,089,631	4,909,911
Trust account, restricted		
Cash held in trust account, restricted	539,259,567	538,747,464
Accrued interest receivable, trust account, restricted	68,962	129,192
Total trust account restricted	539,328,529	538,876,656
Total assets	\$543,418,160	\$543,786,567
LIABILITIES AND STOCKHOLDERS EQUITY		
Current liabilities		
Cash overdraft	\$505,574	\$
Accounts payable and accrued expenses	265,025	404,682
Total current liabilities	770,599	404,682
Common stock, subject to possible conversion, 16,559,999 shares at	161 700 540	161 662 007
conversion value	161,798,549	161,662,987
Commitment and contingencies		
Stockholders equity		
Preferred stock, \$.0001 par value, authorized 1,000,000 shares; none		
issued		
Common stock, \$.0001 par value, authorized 160,000,000 shares; total		
issued and outstanding 69,000,000 shares, (less 16,559,999 shares,	5,244	5,244
subject to possible conversion)		
Additional paid-in capital	375,094,887	375,230,449
Income accumulated during development stage	5,748,881	6,483,205
Total stockholders equity	380,849,012	381,718,898
Total liabilities and stockholders equity	\$543,418,160	\$543,786,567

The accompanying notes are an integral part of these condensed financial statements.

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

### CONDENSED STATEMENTS OF OPERATIONS (Unaudited)

Revenue	Months Ende	ed	For the Three Months Ended June 30, 2008	Months End			For the Period from June 8, 2007 (Inception) Through June 30, 2009 \$
Formation and operating costs	643,309		319,209	1,267,364		681,939	2,931,852
Loss from operations	(643,309	)	(319,209)	(1,267,364	<b>!</b> )	(681,939 )	(2,931,852)
Interest and dividend income	121,006		1,976,161	172,901		5,506,806	12,189,413
(Loss) income before (benefit from) provision for income taxes	(522,303	)	1,656,952	(1,094,463	3)	4,824,867	9,257,561
(Benefit) provision for income taxes	(165,636	)	567,744	(360,139	)	1,569,864	3,508,680
Net (loss) income	(356,667	)	1,089,208	(734,324	)	3,255,003	5,748,881
Accretion of trust account income relating to common stock subject to possible conversion	(73,645	)		(135,562	)		(719,559 )
Net (loss) income attributable to other common stockholders	\$(430,312	)	\$1,089,208	\$(869,886	)	\$3,255,003	\$5,029,322
Weighted average number of common shares outstanding excluding shares subject to possible conversion-basic and diluted	52,440,001		52,440,001	52,440,00	1	52,440,001	
Basic and diluted net (loss) income per share attributable to other common stockholders	\$(0.01	)	\$0.02	\$(0.02	)	\$0.06	

The accompanying notes are an integral part of these condensed financial statements.

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

# CONDENSED STATEMENT OF CHANGES IN STOCKHOLDERS EQUITY (Unaudited) For the Period from June 8, 2007 (Inception) Through June 30, 2009

	Common Sto Shares	ock Amount	Additional Paid-In Capital	Income Accumulated During Development Stage	Stockholders
Balance, June 8, 2007 (inception)		\$	\$	\$	\$
Issuance of stock to initial stockholders at \$0.002 per share	13,800,000	1,380	23,620		25,000
Sale of 55,200,000 units at \$10.00 per unit, net of underwriters discount and offering expenses of \$20,126,320 (includes 16,559,999 shares subject to possible conversion)	55,200,000	5,520	531,868,160		531,873,680
Proceeds subject to possible conversion of 16,559,999 shares		(1,656)	(161,077,334)		(161,078,990)
Proceeds from issuance 5,000,000 of sponsor warrants on October 25, 2007 Net income for the period			5,000,000		5,000,000
from June 8, 2007 (inception)				1,824,397	1,824,397
through December 31, 2007 Balance, December 31, 2007 Accretion of trust account income relating to common stock subject to possible conversion for the year ended	69,000,000	5,244	375,814,446 (583,997 )	1,824,397	377,644,087 (583,997 )

Edgar Filing: Triplecrown Acquisition Corp. - Form PREM14A

December 31, 2008						
Net income for the year				4,658,808	4,658,808	
ended December 31, 2008				4,030,000	4,030,000	
Balance, December 31, 2008	69,000,000	5,244	375,230,449	6,483,205	381,718,898	
Accretion of trust account						
income relating to common						
stock subject to possible			(135,562)		(135,562	`
conversion for the period			(133,302 )		(133,302	,
from January 1, 2009 through						
June 30, 2009						
Net loss for the six months				(734,324)	(734,324	`
ended June 30, 2009				(734,324 )	(734,324	)
Balance, June 30, 2009	69,000,000	\$5.244	\$375,094,887	\$5,748,881	\$380,849,012	
(unaudited)	03,000,000	φ 5,444	φ313,034,001	φ3,740,001	φ300,049,012	

The accompanying notes are an integral part of these condensed financial statements.

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

### CONDENSED STATEMENTS OF CASH FLOWS (Unaudited)

	For the Six Months Ended June 30, 2009	For the Six Months Ended June 30, 2008	For the Period from June 8, 2007 (Inception) Through June 30, 2009
Cash flows from operating activities			
Net (loss) income	\$(734,324)	\$3,255,003	\$5,748,881
Adjustments to reconcile net (loss) income to net cash			
(used in) provided by operating activities			
Changes in operating assets and liabilities			
Other current assets	(275,444 )	(29,724)	(366,511)
Prepaid income taxes	(233,958)	(211,698)	(360,139)
Income taxes payable		(1,568,439)	
Accounts payable and accrued expenses	(139,657)	19,651	265,025
Net cash (used in) provided by operating activities	(1,383,383)	1,464,793	5,287,256
Cash flows from investing activities			
Cash held in trust account, restricted	(451,873)		(539,328,529)
Cash held in trust account, interest and dividend income	2,084,326	(1,760,678)	(2,496,731)
available for working capital and taxes	2,004,320	(1,700,070)	(2,470,731 )
Deposit on contract	(866,250)		(866,250)
Net cash provided by (used in) investing activities	766,203	(1,760,678)	(542,691,510)
Cash flows from financing activities			
Cash overdraft	505,574		505,574
Proceeds from issuance of stock to initial stockholders			25,000
Proceeds from notes payable, stockholders			112,500
Repayments of notes payable, stockholders			(112,500)
Gross proceeds from issuance of sponsors warrants			5,000,000
Gross proceeds from initial public offering and			552,000,000
over-allotments options			332,000,000
Payment of offering costs			(20,126,320)
Net cash provided by financing activities	505,574		537,404,254
Net decrease in cash	(111,606)	(295,885)	
Cash at beginning of the period	111,606	296,913	
Cash at end of the period	\$	\$1,028	\$
Supplemental disclosure of cash flow information:			
Cash paid during the period for:			
Income taxes	\$	\$1,750,000	\$3,868,819

Supplemental disclosure of non-cash transaction: Accretion of trust account income relating to common

Accretion of trust account income relating to common stock subject to possible conversion

\$135,562

\$

\$719,559

The accompanying notes are an integral part of these condensed financial statements.

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

### NOTES TO THE UNAUDITED CONDENSED INTERIM FINANCIAL STATEMENTS

# Note 1 Interim Financial Information, Organization, Business Operations, Going Concern Considerations and Significant Accounting Policies

Triplecrown Acquisition Corp. s (the Company ) (a development stage enterprise) unaudited condensed interim financial statements as of June 30, 2009, and for the three and six month periods ending June 30, 2009 and 2008, and for the period from June 8, 2007 (inception) through June 30, 2009, have been prepared in accordance with accounting principles generally accepted in the United States of America ( GAAP ) for interim financial information and with the instructions to Form 10-Q. Accordingly, they do not include all of the information and footnotes required by GAAP for complete financial statements. In the opinion of management, all adjustments (consisting of normal recurring accruals) considered necessary for a fair presentation have been included. Operating results for the interim period presented are not necessarily indicative of the results to be expected for any other interim period or for the full year. These unaudited condensed interim financial statements should be read in conjunction with the audited financial statements and notes thereto for the fiscal year ended December 31, 2008 included in the Company s Form 10-K filed on March 13, 2009. The accounting policies used in preparing these unaudited condensed financial statements are consistent with those described in the December 31, 2008 audited financial statements.

The Company was incorporated in Delaware on June 8, 2007 as a blank check company to serve as a vehicle to effect a merger, capital stock exchange, asset acquisition, stock purchase, reorganization or other similar business combination with one or more operating businesses in the financial services industry (Business Combination).

All activity from June 8, 2007 (inception) through October 25, 2007 relates to the Company s formation and the Company s initial public offering ( Offering ). Since October 25, 2007, the Company has been searching for a target business to acquire. The Company has selected December 31 as its fiscal year end.

#### Going Concern and Management s Plan and Intentions

The Company s only source of income to enable it to continue to fund its search for an acquisition candidate is the interest and dividends it earns on its cash held in the Trust Account. These funds may not be sufficient to maintain the Company until a Business Combination is consummated. Pursuant to its Certificate of Incorporation, if the Company is unable to consummate a timely Business Combination prior to October 22, 2009, the Company would have to liquidate and return the funds held in the Trust Account to the holders of shares issued in the Offering. There can be no assurance that the Company will complete a Business Combination prior to October 22, 2009. These factors raise substantial doubt about the Company s ability to continue as a going concern. The unaudited condensed interim financial statements do not include any adjustment that might result from the outcome of these uncertainties.

#### **Concentration of Credit Risk**

At June 30, 2009, financial instruments that potentially expose the Company to credit risk consist of cash and cash equivalents held in trust. The Company maintains its cash balances in a U.S. Treasury - only money market fund at JP Morgan Private Bank. At times, the Company s cash and cash held in the trust account may be uninsured or in deposit accounts that exceed the Federal Deposit Insurance Corporation (FDIC) and Securities Investor Protection Corporation (SIPC) insurance limits.

#### Accretion of Trust Account Relating to Common Stock Subject to Possible Conversion

The Company records accretion of the income earned in the Trust Account relating to the common stock subject to possible conversion based on the excess of the earnings for the period over the amount which is available to be used for working capital and taxes. Since 30% (less one share) of the shares issued in the Offering are subject to possible conversion, the portion of the excess earnings related to those shares are reflected on the balance sheet as part of Common stock subject to possible conversion and is deducted from Additional paid-in capital. The portion of the excess earnings is also presented as a deduction from net income on the Statements of Operations to appropriately reflect the amount of net income which would remain available to the common stockholders who did not elect to convert their shares to cash.

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

### NOTES TO THE UNAUDITED CONDENSED INTERIM FINANCIAL STATEMENTS

# Note 1 Interim Financial Information, Organization, Business Operations, Going Concern Considerations and Significant Accounting Policies (continued)

#### **Earnings per Share**

The Company follows the provisions of SFAS No. 128, Earnings Per Share . In accordance with SFAS No. 128, earnings per common share amounts ( Basic EPS ) are computed by dividing earnings by the weighted average number of common shares outstanding for the period. Common shares subject to possible conversion of 16,559,999 have been excluded from the calculation of basic earnings per share since such shares, if redeemed, only participate in their pro rata share of the trust earnings, such earnings as applicable are accrued to the value of the common stock subject to conversion. Earnings per common share amounts, assuming dilution ( Diluted EPS ), gives effect to dilutive options, warrants, and other potential common stock outstanding during the period. SFAS No.128 requires the presentation of both Basic EPS and Diluted EPS on the face of the statements of operations. The effect of the 74,000,000 outstanding warrants issued in connection with the Offering and the Private Placement has not been considered in the Diluted EPS calculation since the exercise of the warrants are contingent upon the occurrence of future events, and therefore, is not includable in the calculation of Diluted EPS in accordance with SFAS 128.

#### **Recently Issued and Adopted Pronouncements**

In June 2009, the FASB issued SFAS No. 167, Amendments to FASB Interpretation No. 46(R) (SFAS 167). SFAS 167 eliminates Interpretation 46(R) is exceptions to consolidating qualifying special-purpose entities, contains new criteria for determining the primary beneficiary, and increases the frequency of required reassessments to determine whether a company is the primary beneficiary of a variable interest entity. SFAS 167 also contains a new requirement that any term, transaction, or arrangement that does not have a substantive effect on an entity is status as a variable interest entity, a company is power over a variable interest entity, or a company is obligation to absorb losses or its right to receive benefits of an entity must be disregarded in applying Interpretation 46(R) is provisions. The elimination of the qualifying special-purpose entity concept and its consolidation exceptions means more entities will be subject to consolidation assessments and reassessments. SFAS 167 will be effective January 1, 2010. The adoption of SFAS 167 is not expected to have an impact on our financial statements or results of operations.

In July 2009, the FASB issued SFAS No. 168, The FASB Accounting Standards Codification and the Hierarchy of Generally Accepted Accounting Principles a replacement of FASB Statement No. 162 (SFAS 168). SFAS 168 establishes the FASB Accounting Standards Codification to become the source of authoritative U.S. GAAP recognized by the FASB to be applied by non-governmental entities. Rules and interpretive releases of the Securities

and Exchange Commission (SEC) under authority of federal securities laws are also sources of authoritative GAAP for SEC registrants. On the effective date of this Statement, the Codification will supercede all then-existing non-SEC accounting and reporting standards. All other nongrandfathered non-SEC accounting literature not included in the Codification will become nonauthoritative. SFAS 168, which changes the referencing of financial standards, is effective for interim or annual financial periods ending after September 15, 2009. The adoption of SFAS 168 is not expected to materially impact our financial statements or results of operations.

In April 2009, the Financial Accounting Standards Board (FASB) issued FASB Staff Position No. 107-1 (FSP FAS 107-1) and APB 28-1 (APB 28-1), which amends FASB Statement No. 107, Disclosures about Fair Value of Financial Instruments and APB Opinion No. 28, Interim Financial Reporting, to require disclosures about the fair value of financial instruments for interim reporting periods. FSP FAS 107-1 and APB 28-1 will be effective for interim reporting periods ending after June 15, 2009. The adoption of this staff position did not have a material impact on the Company s Condensed financial position or results of operation.

In April 2009, the FASB issued FASB Staff Position No. 157-4 (FSP FAS 157-4), which provides additional guidance in accordance with FASB No. 157, Fair Value Measurements, when the volume and level

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

### NOTES TO THE UNAUDITED CONDENSED INTERIM FINANCIAL STATEMENTS

# Note 1 Interim Financial Information, Organization, Business Operations, Going Concern Considerations and Significant Accounting Policies (continued)

of activity for the asset or liability has significantly decreased. FSP FAS 157-4 shall be effective for interim and annual reporting periods ending after June 15, 2009. The adoption of this staff position did not have a material impact on the Company s Condensed financial statements.

In April 2009, the FASB issued FASB Staff Position No. 115-2 (FSP FAS 115-2) and FASB Staff Position No. 124-2 (FSP FAS 124-2), which amends the other-than-temporary impairment guidance for debt and equity securities. FSP FAS 115-2 and FSP FAS 124-2 shall be effective for interim and annual reporting periods ending after June 15, 2009. The adoption of this staff position did not have a material impact on the Company s Condensed financial statements.

In April 2009, the FASB issued FASB Staff Position No. 141(R)-1 (FSP FAS 141(R)-1) which provides additional clarification on the initial recognition and measurement of assets acquired and liabilities assumed in a business combination that arise from contingencies. FSP FAS 141(R)-1 is effective for all fiscal years beginning on or after December 15, 2008. FSP FAS 141(R)-1 will have an impact on the accounting for any business acquired in the future.

In May 2009, the FASB issued Statement of Financial Accounting Standards (SFAS) No. 165, Subsequent Events (SFAS No. 165). The statement is to establish general standards of accounting for and disclosure of events that occur after the balance sheet date but before financial statements are issued. SFAS No. 165 shall become effective June 15, 2009 for all subsequent reporting periods. The adoption of SFAS No. 165 did not have a material impact on the Company s Condensed financial position or results of operation.

In June 2009, the FASB issued SFAS No. 168, The FASB Accounting Standards Codification and the Hierarchy of Generally Accepted Accounting Principles (SFAS No. 168). This statement will become the source of authoritative U.S. generally accepted accounting principles recognized by the FASB to be applied by nongovernmental entities. On the effective date of this statement, the codification will supersede all then-existing non SEC accounting and reporting standards. SFAS No. 168 is effective for financial statements issued for interim and annual periods ending after September 15, 2009. The adoption of this statement is not expected to have a material impact on the Company s financial statements.

Management does not believe that any other recently issued, but not yet effective, accounting standards if currently adopted would have a material effect on the Company s financial statements.

#### Note 2 Commitments and Contingencies

The Company entered into an agreement with the underwriters of the Offering (the Underwriting Agreement ). The Underwriting Agreement requires the Company to pay 3.5% of the gross proceeds of the Offering as an underwriting discount plus an additional 3.5% of the gross proceeds only upon consummation of a Business Combination. The Company paid an underwriting discount of 3.5% of the gross proceeds (\$19,320,000) in connection with the consummation of the Offering and has placed 3.5% of the gross proceeds (\$19,320,000) in the Trust Account. The Company did not pay any discount related to the Sponsors Warrants sold in the private placement that was consummated simultaneously with the consummation of the Offering. The underwriters have waived their right to receive payment of the 3.5% of the gross proceeds upon the Company s liquidation if it is unable to complete a Business Combination.

The Company engaged Petrina Advisors, Inc. on May 19, 2008 to perform consulting services through the earlier of October 22, 2009 or upon the closing of a Business Combination. The Company will pay this consultancy firm a base fee of \$192,000 per year and the Company will issue to them 100,000 shares of the Company s common stock upon the closing of the initial Business Combination by the Company.

### TRIPLECROWN ACQUISITION CORP. (A Development Stage Enterprise)

### NOTES TO THE UNAUDITED CONDENSED INTERIM FINANCIAL STATEMENTS

#### Note 3 Amendment to Bylaws

On May 13, 2009, the Company announced through a press release that its board of directors approved an amendment to the Bylaws fixing the number of directors that shall constitute the Board at eleven and requiring approval by the holders of at least 85% of the outstanding shares of common stock of the Company in order for stockholders to amend this provision at any time prior to consummation by the Company of a Business Combination.

#### Note 4 Income Taxes

During the three and six months ended June 30, 2009, the Company recorded a tax benefit of \$165,636 and \$360,139, respectively. The tax benefit was the result of current operating loss, which the Company expects to utilize.

#### Note 5 Deposit on Contract

On June 27, 2009, in connection with a potential Business Combination, the Company and an entity controlled by the Company s Chairman of the Board and Treasurer (Related Party) entered into a contract to purchase a certain piece of land (the Property). The total purchase price of the Property is \$8,662,500 on which the Company paid a deposit of \$866,250 (the Deposit).

#### Note 6 Subsequent Events

On July 1, 2009, the Company entered into agreements with the Related Party pursuant to which the Related Party agreed to (i) repay the Deposit (see Note 5) to the Company if the Company is not able to enter into a definitive agreement for a Business Combination relating to the Property within a certain period of time and (ii) assume the Company s obligations to pay the remaining purchase price for the Property if the Company is not able to complete the purchase of the Property for any reason whatsoever.

The Company has evaluated events that occurred subsequent to June 30, 2009 through August 10, 2009, the date on which these financial statements were issued. Except as discussed above, management concluded that no other events required disclosure in these financial statements.

#### TABLE OF CONTENTS

	Page
ANNEXES	_
Agreement and Plan of Reorganization	<u>A-1</u>
CAH Amended and Restated Certificate of Incorporation	<u>B-1</u>
<u>CAH Bylaws</u>	<u>C-1</u>
Form of Employment Agreement	<u>D-1</u>
Form of Escrow Agreement	<u>E-1</u>
Tax Opinion of Graubard Miller	<u>F-0</u>
Triplecrown Amended and Restated Certificate of Incorporation	<u>G-1</u>
2009 Long-Term Incentive Equity Plan	<u>H-1</u>
Opinion of Richards, Layton & Finger, P.A.	<u>I-1</u>
Lock-Up Agreement	<u>J-1</u>
Fairness Opinion of Duff & Phelps	<u>K-1</u>
Section 262 of the Delaware General Corporation Law	<u>L-1</u>
F-44	

Annex A

# AGREEMENT AND PLAN OF REORGANIZATION BY AND AMONG TRIPLECROWN ACQUISITION CORP., CULLEN AGRICULTURAL HOLDING CORP., CAT MERGER SUB, INC., CULLEN AGRICULTURAL TECHNOLOGIES, INC. and CULLEN INC HOLDINGS LIMITED DATED AS OF SEPTEMBER 4, 2009

#### AGREEMENT AND PLAN OF REORGANIZATION

THIS AGREEMENT AND PLAN OF REORGANIZATION is made and entered into as of September 4, 2009, by and among Triplecrown Acquisition Corp., a Delaware corporation ( *Parent* ), Cullen Agricultural Holding Corp., a Delaware corporation and wholly owned subsidiary of Parent ( *Holdco* ), CAT Merger Sub, Inc., a Georgia corporation and wholly owned subsidiary of Holdco ( *Merger Sub* ), Cullen Agricultural Technologies, Inc., a Georgia corporation ( *Company* ), and Cullen Inc Holdings Limited, a New Zealand limited company and the sole stockholder of the Company ( *Stockholder* ).

#### **RECITALS**

- A. Upon the terms and subject to the conditions of this Agreement (as defined in Section 1.3) and in accordance with the General Corporation Law of the State of Delaware (the *DGCL*) and the Georgia Corporate Code ( *GCC* and together with the DGCL, the *Applicable Corporate Law*) and other applicable law, Parent and Company intend to enter into a business combination transaction by means of (i) a merger of Parent with and into Holdco, with Holdco surviving the merger and becoming the public company through the automatic conversion of all the issued and outstanding securities of Parent into a like type and number of securities of Holdco as provided herein (the *Parent Merger*) and (ii) a merger of Merger Sub with and into the Company, with the Company surviving the merger and becoming a wholly owned subsidiary of Holdco through an exchange of all the issued and outstanding capital stock of the Company for shares of common stock of Holdco as provided herein (the *Company Merger* and, together with the Parent Merger, the *Mergers*).
- B. The boards of directors of each of Parent, Holdco, Merger Sub, the Company and the Stockholder have determined that the Mergers are fair to, and in the best interests of, their respective companies and their respective stockholders.
- C. The parties intend for the Mergers to qualify as Tax-free transactions pursuant to Section 351, Section 368(a)(1)(A) and Section 368(a)(2)(E) of the Internal Revenue Code of 1986, as amended (the *Code*).

NOW, THEREFORE, in consideration of the covenants, promises and representations set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows (defined terms used in this Agreement are listed alphabetically in Article IX, together with the Section and, if applicable, paragraph number in which the definition of each such term is located):

#### ARTICLE I

#### THE MERGERS

- 1.1 *The Parent Merger*. At the Effective Time (as defined in Section 1.3) and subject to and upon the terms and conditions of this Agreement and the applicable provisions of the DGCL, Parent shall be merged with and into Holdco, the separate corporate existence of Parent shall cease and Holdco shall continue as the public corporation. Holdco as the surviving public corporation after the Parent Merger is hereinafter sometimes referred to as the *Surviving Pubco*.
- 1.2 *The Company Merger*. At the Effective Time and subject to and upon the terms and conditions of this Agreement and the applicable provisions of the GCC, Merger Sub shall be merged with and into the Company, the separate corporate existence of Merger Sub shall cease and the Company shall continue as the surviving entity of the Company Merger. The Company as the surviving entity after the Company Merger is hereinafter sometimes referred to as the *Surviving Subsidiary*.

1.3 Effective Time; Closing. Subject to the conditions of this Agreement, as soon as practicable on or after the Closing Date (as hereinafter defined), the parties hereto shall cause (a) the Parent Merger to be consummated by filing a Certificate of Merger (the Parent Certificate of Merger) with the Secretary of State of the State of Delaware in accordance with the applicable provisions of Delaware law and (b) the Company Merger to be consummated by filing an Articles of Merger with the Secretary of State of the State of Georgia in accordance with the applicable provisions of Georgia law (the Company Articles of Merger) (the time of the later of such filings, or such later time as may be agreed in writing by Company and Parent

and specified in the Parent Certificate of Merger and the Company Articles of Merger, being the *Effective Time*). The term *Agreement* as used herein refers to this Agreement and Plan of Reorganization, as the same may be amended from time to time, and all schedules hereto (including the Company Schedule and the Parent Schedule, as defined in the preambles to Articles II and III hereof, respectively). Unless this Agreement shall have been terminated pursuant to Section 8.1, the consummation of the transactions contemplated by this Agreement (the *Closing*), other than the filing of the Parent Certificate of Merger and the Company Articles of Merger, shall take place at the offices of Graubard Miller, counsel to Parent and Holdco, 405 Lexington Avenue, New York, New York 10174-1901 at a time and date to be specified by the parties, which shall be no later than the second business day after the satisfaction or waiver of the conditions set forth in Article VI, or at such other time, date and location as the parties hereto agree in writing (the *Closing Date*). Closing signatures may be transmitted by facsimile or emailed .pdf file.

1.4 Effect of the Mergers. At the Effective Time, (a) the effect of the Parent Merger shall be as provided in this Agreement and the Applicable Corporate Law and other laws and (b) the effect of the Company Merger shall be as provided in this Agreement and the Applicable Corporate Law and other laws. Without limiting the generality of the foregoing, and subject thereto, at the Effective Time (c) all the property, rights, privileges, powers and franchises of Parent and Holdco shall vest in Surviving Pubco, and all debts, liabilities and duties of Parent and Holdco shall become the debts, liabilities and duties of Surviving Pubco and (d) all the property, rights, privileges, powers and franchises of the Company and Merger Sub shall vest in the Surviving Subsidiary, and all debts, liabilities and duties of the Company and Merger Sub shall become the debts, liabilities and duties of the Surviving Subsidiary.

1.5 Governing Documents. At the Effective Time,

- (a) the Certificate of Incorporation of Holdco in the form of *Exhibit A* hereto shall become the Certificate of Incorporation of Surviving Pubco;
  - (b) the Bylaws of Holdco in the form of *Exhibit B* hereto shall become the Bylaws of Surviving Pubco;
- (c) the Articles of Incorporation of Merger Sub in the form of *Exhibit C* shall become the Articles of Incorporation of the Surviving Subsidiary; and
  - (d) the Bylaws of Merger Sub in the form of *Exhibit D* shall become the Bylaws of the Surviving Subsidiary.
- 1.6 *Effect on Securities*. Subject to the terms and conditions of this Agreement, at the Effective Time, by virtue of the Mergers and this Agreement and without any action on the part of Parent, Holdco or the Company or the holders of any of the securities of Parent or the Company, the following shall occur:
- (a) Conversion of Parent Common Stock. Each share of common stock, par value \$.0001, of Parent ( Parent Common Stock ) issued and outstanding immediately prior to the Effective Time will be automatically converted into one share of common stock, par value \$.0001, of Holdco ( Holdco Common Stock ) and the holders thereof shall cease to have any further rights as holders of capital stock of Parent. Immediately following the Parent Merger, each certificate that evidenced Parent Common Stock immediately prior to the Parent Merger ( Parent Certificate ) shall represent the applicable number of shares of Holdco Common Stock into which the Parent Common Stock represented thereby have been converted.
- (b) Conversion of Parent Warrants. Each outstanding warrant to purchase a share of Parent Common Stock ( Parent Warrant ) issued and outstanding immediately prior to the Effective Time shall remain outstanding immediately following the Effective Time but shall be deemed to have been converted and to represent a warrant ( Holdco Warrants ) to purchase one share of Holdco Common Stock on the same terms existing under the Parent Warrant

immediately prior to the Effective Time, subject to the changes contemplated by Section 5.1(c). Immediately following the Parent Merger, each certificate that evidenced Parent Warrants immediately prior to the Parent Merger ( *Parent Warrant Certificate* ) shall represent the applicable number of Holdco Warrants into which the Parent Warrants represented thereby have been converted.

(c) Cancellation of Holdco Capital Stock. Each share of capital stock of Holdco issued and outstanding immediately prior to the Effective Time will be canceled.

- (d) Conversion of Company Common Stock; Issuance to Holdco. At the Effective Time, all of the outstanding shares of the common stock of the Company, no par value ( Company Common Stock ), shall be automatically converted into an aggregate of 15,881,148 shares of Holdco Common Stock ( Transaction Shares ). Subject to Section 1.12, below, certificates representing the Transaction Shares shall be issued to the Stockholder upon surrender of the certificates representing all of the outstanding Company Common Stock ( Company Certificates ). Immediately following the conversion contemplated by this Section, 100 shares of the Company, as the Surviving Subsidiary, shall be issued to Holdco.
- (e) Cancellation of Merger Sub Capital Stock. At the Effective Time, each share of common stock, par value \$.0001 per share, of Merger Sub (the Merger Sub Common Stock) issued and outstanding immediately prior to the Effective Time shall be cancelled.
- (f) Adjustments to Exchange Ratios. The numbers of shares of Holdco Common Stock that the holders of the Parent Common Stock and the Stockholder are entitled to receive as a result of the Mergers shall be equitably adjusted to reflect appropriately the effect of any stock split, reverse stock split, stock dividend (including any dividend or distribution of securities convertible into Parent Common Stock or Company Common Stock), extraordinary cash dividends, reorganization, recapitalization, reclassification, combination, exchange of shares or other like change with respect to Parent Common Stock or Company Common Stock occurring on or after the date hereof and prior to the Effective Time.

#### 1.7 Surrender of Company Certificates.

- (a) *Exchange Procedures*. Upon surrender of Company Certificates at the Closing, the Stockholder shall receive certificates representing the Transaction Shares into which the Company Common Stock shall be converted at the Effective Time, less the Escrow Shares, and the Company Certificates so surrendered shall forthwith be canceled. Until so surrendered after the Effective Time for the Transaction Shares, outstanding Company Certificates will be deemed, from and after the Effective Time, to evidence no rights.
- (b) Distributions With Respect to Unexchanged Shares. No dividends or other distributions declared or made after the date of this Agreement with respect to Holdco Common Stock with a record date after the Effective Time will be paid to the holders of any unsurrendered Company Certificates with respect to the Transaction Shares to be issued upon surrender thereof until the holders of record of such Company Certificates shall surrender such certificates. Subject to applicable law, following surrender of any such Company Certificates, Holdco shall promptly deliver to the record holders thereof, without interest, the certificates representing the Transaction Shares issued in exchange therefor and the amount of any such dividends or other distributions with a record date after the Effective Time theretofore paid with respect to such Transaction Shares.
- (c) *Transfers of Ownership*. If certificates representing Transaction Shares are to be issued in a name other than that in which the Company Certificates surrendered in exchange therefor are registered, it will be a condition of the issuance thereof that the Company Certificates so surrendered will be properly endorsed and otherwise in proper form for transfer and that the persons requesting such exchange will have paid to Holdco or any agent designated by it any transfer or other Taxes required by reason of the issuance of certificates representing Transaction Shares in any name other than that of the registered holder of the Company Certificates surrendered, or established to the satisfaction of Holdco or any agent designated by it that such Tax has been paid or is not payable.
  - (d) *Required Withholding*. Parent and Holdco shall each be entitled to deduct and withhold from any consideration payable or otherwise deliverable pursuant to this Agreement to any holder or former holder of Company Common Stock such amounts as are required to be deducted or withheld therefrom under the Code or under any provision of

state, local or foreign Tax law or under any other applicable legal requirement. To the extent such amounts are so deducted or withheld, such amounts shall be treated for all purposes under this Agreement as having been paid to the Person to whom such amounts would otherwise have been paid.

(e) No Further Ownership Rights in Company Common Stock. All shares of Holdco Common Stock issued to the Stockholder in accordance with the terms hereof shall be deemed to have been issued in consideration of the Stockholder transferring all ownership rights pertaining to the Company Common Stock and there shall be no further registration of transfers on the records of the Surviving Subsidiary of Company

- Common Stock that were outstanding immediately prior to the Effective Time. If, after the Effective Time, Company Certificates are presented to the Surviving Subsidiary for any reason, they shall be canceled and exchanged as provided in this Article I.
  - 1.8 *No Liability*. Notwithstanding anything to the contrary in this Agreement, no party hereto shall be liable to a holder of shares of Parent Common Stock or Company Common Stock for any amount properly paid to a public official pursuant to any applicable abandoned property, escheat or similar law.
- 1.9 Lost, Stolen or Destroyed Certificates. In the event that any Company Certificates shall have been lost, stolen or destroyed, Holdco shall issue in exchange for such lost, stolen or destroyed Company Certificates, upon the making of an affidavit of that fact by the holder thereof, the certificates representing the Transaction Shares that the Company Common Stock formerly represented by such Company Certificates were converted into and any dividends or distributions payable pursuant to Section 1.7(b); provided, however, that, as a condition precedent to the issuance of such certificates representing Transaction Shares and other distributions, the owner of such lost, stolen or destroyed Company Certificates shall indemnify Holdco against any claim that may be made against the Surviving Pubco or Surviving Subsidiary with respect to the Company Certificates alleged to have been lost, stolen or destroyed.
- 1.10 Tax Consequences. It is intended by the parties hereto that the Mergers shall constitute Tax-free transactions pursuant to Section 351, Section 368(a)(1)(A) and Section 368(a)(2)(E) of the Code and the United States Income Tax Regulations ( *Regulations* ). The parties hereto adopt this Agreement as a plan or reorganization for the Merger within the meaning of Section 1.368-2(g) and 1.368(a) of the Regulations.
- 1.11 *Taking of Necessary Action; Further Action*. If, at any time after the Effective Time, any further action is necessary or desirable to carry out the purposes of this Agreement and to vest the Surviving Subsidiary with full right, title and possession to all assets, property, rights, privileges, powers and franchises of the Company, the Company and Holdco and their respective officers and directors will take all such lawful and necessary action.
- 1.12 Escrow. As the sole remedy for the indemnity obligations set forth in Article VII, at the Closing, the Stockholder shall deposit in escrow 1,588,114 of the Transaction Shares ( Escrow Shares ), all in accordance with the terms and conditions of the Escrow Agreement to be entered into at the Closing between Holdco, the Company, the Stockholder and Continental Stock Transfer & Trust Company ( Continental ), as Escrow Agent, in the form annexed hereto as Exhibit E (the Escrow Agreement ). The Escrow Agreement shall provide that on the date Holdco has filed with the SEC its Annual Report on Form 10-K for the year ending December 31, 2010 but, in any event, no later than April 16, 2011 (the Escrow Termination Date ), the Escrow Agent shall release the Escrow Shares, less that number of Escrow Shares applied in satisfaction of or reserved with respect to indemnification claims made prior to such date, to the Stockholder. Any Escrow Shares held with respect to any unresolved claim for indemnification and not applied as indemnification with respect to such claim upon its resolution shall be delivered to such Persons promptly upon such resolution. While in escrow, the Escrow Shares shall be deemed to be issued and outstanding and the Stockholder shall have the right to (i) vote the Escrow Shares at any meeting of the stockholders of Holdco or (ii) receive dividends or distributions paid or made on the Escrow Shares.
- 1.13 New Zealand Income Tax Act 2007. To reflect that the conversion of the Company Common Stock constitutes a disposal of shares for the purposes of the financial arrangements rules in the Income Tax Act 2007 (New Zealand), the parties agree solely for the purposes of those rules that the consideration received by the Stockholder for that disposal (including, but not limited to, the Transaction Shares as provided in Section 1.6(d) and any Holdco Common Stock issued to the Stockholder pursuant to the Holdco Plan) is the lowest price that the parties would have agreed upon with respect to the Company Common Stock at the time this Agreement was executed on the basis of payment in full at the time at which the first right in the Company Common Stock is to be transferred and on that basis no income or

expenditure arises under those rules.

1.14 *Rule 145*. All shares of Holdco Common Stock issued pursuant to this Agreement will be subject to certain resale restrictions under Rule 145 promulgated under the Securities Act and all certificates representing such shares shall bear an appropriate restrictive legend.

#### 1.15 Stockholder Matters.

- (a) By its execution of this Agreement, the Stockholder, as the holder of all of the outstanding capital stock of the Company entitled to vote, hereby approves the adoption of this Agreement, approves the Mergers and the related transactions and authorizes the Company, its directors and officers to take all actions necessary for the consummation of the Mergers and the other transactions contemplated hereby pursuant to the terms of this Agreement and its Exhibits.
- (b) The Stockholder hereby represents and warrants as follows: (i) all Holdco Common Stock to be acquired by it pursuant to this Agreement will be acquired for its account and not with a view towards distribution thereof other than, with respect to transfers to its stockholders; (ii) it understands that it must bear the economic risk of the investment in the Holdco Common Stock; (iii) it has had both the opportunity to ask questions and receive answers from the officers and directors of Parent and Holdco and all persons acting on Parent s and Holdco s behalf concerning the business and operations of Parent and Holdco and to obtain any additional information to the extent Parent or Holdco possesses or may possess such information or can acquire it without unreasonable effort or expense necessary to verify the accuracy of such information; and (iv) it has had access to the Parent SEC Reports (as defined in Section 3.7) filed prior to the date of this Agreement. The Stockholder acknowledges and represents that it is either (A) an accredited investor—as such term is defined in Rule 501(a) promulgated under the Securities Act or (B) a Person possessing sufficient knowledge and experience in financial and business matters to enable it to evaluate the merits and risks of an investment in Holdco. The Stockholder understands that the certificates representing the Holdco Common Stock to be received by it will bear legends to the effect that the Holdco Common Stock may not be transferred except upon compliance with (x) the registration requirements of the Securities Act of 1933, as amended (the Securities Act), or an exemption therefrom, and (y) the provisions of this Agreement.
- (c) The Stockholder represents and warrants that the execution and delivery of this Agreement by the Stockholder does not, and the performance of its obligations hereunder will not, require any consent, approval, authorization or permit of, or filing with or notification to, any court, administrative agency, commission, governmental or regulatory authority, domestic or foreign (a *Governmental Entity*), except (i) for applicable requirements, if any, of the Securities, the Securities Exchange Act of 1934, as amended (the *Exchange Act*), state securities laws ( *Blue Sky Laws*), and the rules and regulations thereunder, and (ii) where the failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect (as defined in Section 10.2(a)) on the Stockholder or the Company or, after the Closing, the Surviving Pubco or the Surviving Subsidiary, or prevent consummation of the Mergers or otherwise prevent the parties hereto from performing their obligations under this Agreement.
- (d) The Stockholder represents and warrants it owns shares of Company Common Stock listed on Schedule 2.3(a) free and clear of all Liens (as defined in Section 10.2(e)).
- 1.16 Committee for Purposes of Escrow Agreement. Prior to the Closing, the Board of Directors of Parent shall appoint a committee consisting of one or more of its then members to act on behalf of Holdco, as the Surviving Pubco, to take all necessary actions and make all decisions pursuant to the Escrow Agreement regarding Parent s and Holdco s right to indemnification pursuant to Article VII hereof. In the event of a vacancy in such committee, the board of directors of Parent or, after the consummation of the Mergers, the board of directors of Holdco, shall appoint as a successor a Person who was a director of Parent prior to the Closing Date or some other Person who would qualify as an independent director of Holdco and who has not had any relationship with the Company prior to the Closing. Such committee is intended to be the Committee referred to in Article VII hereof and the Escrow Agreement.

1.17 *Sale Restriction*. The Stockholder shall not, prior to the first anniversary of the Closing, sell, transfer or otherwise dispose of an interest in any of the shares of Holdco Common Stock received as a result of the Mergers other than as permitted pursuant to the Lock-Up Agreement in the form of *Exhibit F* hereto to be executed by the Stockholder simultaneously herewith.

1.18 Holdco, as the Surviving Pubco, shall implement an Incentive Stock Option Plan (the *Holdco Plan* ) in the form of *Exhibit G* hereto. The Holdco Plan shall provide that an aggregate of 12.5% of the

shares of Holdco Common Stock issued and outstanding as of the Closing Date (after giving effect to any conversions, surrenders and repurchases of Parent or Holdco Common Stock in connection with the Mergers and the approval by the Parent Stockholder thereof) shall be reserved for issuance pursuant to the Holdco Plan.

#### ARTICLE II

# REPRESENTATIONS AND WARRANTIES REGARDING THE COMPANY AND THE STOCKHOLDER

Subject to the exceptions set forth in *Schedule 2* attached hereto (the *Company Schedule*), the Company and the Stockholder jointly and severally represent and warrant to, and covenant with, Parent, Holdco and Merger Sub as follows (as used in this Article II, and elsewhere in this Agreement, the term Company includes the Subsidiaries, as hereinafter defined, unless the context clearly otherwise indicates):

#### 2.1 Organization and Qualification.

- (a) Each of the Company and the Stockholder is a corporation duly organized, validly existing and in good standing under the law of the jurisdiction of its formation. The Company has the requisite power and authority to own, lease and operate its assets and properties and to carry on its business as it is now being or currently planned by the Company to be conducted. The Company is in possession of all franchises, grants, authorizations, licenses, permits, easements, consents, certificates, approvals and orders ( *Approvals* ) necessary to own, lease and operate the properties it purports to own, operate or lease and to carry on its business as it is now being conducted, except where the failure to have such Approvals could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Company. Complete and correct copies of the articles or certificate of incorporation and bylaws (or other comparable governing instruments with different names) (collectively referred to herein as *Charter Documents* ) of each of the Company and the Stockholder, as amended and currently in effect, have been heretofore made available to Parent s counsel. Neither the Company nor the Stockholder is in violation of any of the material provisions of its Charter Documents.
  - (b) The Company is duly qualified or licensed to do business and is in good standing in each jurisdiction where the character of the properties owned, leased or operated by it or the nature of its activities makes such qualification or licensing necessary, except for such failures to be so duly qualified or licensed and in good standing that could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Company. Each jurisdiction in which the Company is so qualified or licensed is listed in *Schedule 2.1*.
  - (c) Copies of the minute books and records of the Company have been made available to Parent or Parent s counsel.
- (d) Copies of the equity ownership and transfer records of the Company have been made available to Parent or Parent s counsel.

### 2.2 Subsidiaries.

(a) The Company has no direct or indirect subsidiaries or participations in joint ventures or other entities other than those listed in *Schedule 2.2* and, as the context permits, those contemplated to be formed as specifically provided in the business section of the Proxy Statement/Prospectus (as defined hereinafter) (the *Subsidiaries*). The Company owns all of the outstanding equity securities of the Subsidiaries, free and clear of all Liens, either directly or indirectly through one or more Subsidiaries. Except for the Subsidiaries, the Company does not own, directly or indirectly, any ownership, equity, profits or voting interest in any Person or has any agreement or commitment to purchase any such

interest, and has not agreed and is not obligated to make nor is bound by any written, oral or other agreement, contract, subcontract, lease, binding understanding, instrument, note, option, warranty, purchase order, license, sublicense, insurance policy, benefit plan, commitment or undertaking of any nature, as of the date hereof or as may hereafter be in effect under which it may become obligated to make, any future investment in or capital contribution to any other entity.

- (b) Each Subsidiary that is a corporation is duly incorporated, validly existing and in good standing under the laws of its state of incorporation (as listed in *Schedule 2.2*) and has the requisite corporate power and authority to own, lease and operate its assets and properties and to carry on its business as it is now being or currently planned by the Company to be conducted. Each Subsidiary that is a limited liability company is duly organized or formed, validly existing and in good standing under the laws of its state of organization or formation (as listed in *Schedule 2.2*) and has the requisite power and authority to own, lease and operate its assets and properties and to carry on its business as it is now being conducted. Each Subsidiary is in possession of all Approvals necessary to own, lease and operate the properties it purports to own, operate or lease and to carry on its business as it is now being conducted, except where the failure to have such Approvals could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Company or such Subsidiary. Complete and correct copies of the Charter Documents of each Subsidiary, as amended and currently in effect, have been heretofore delivered to Parent or Parent s counsel. No Subsidiary is in violation of any of the material provisions of its Charter Documents.
- (c) Each Subsidiary is duly qualified or licensed to do business as a foreign corporation or foreign limited liability company and is in good standing in each jurisdiction where the character of the properties owned, leased or operated by it or the nature of its activities makes such qualification or licensing necessary, except for such failures to be so duly qualified or licensed and in good standing that could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Company or such Subsidiary. Each jurisdiction in which each Subsidiary is so qualified or licensed is listed in *Schedule 2.2*.
  - (d) Copies of the minute books and records of each Subsidiary have been heretofore made available to Parent or Parent s counsel.

#### 2.3 Capitalization.

- (a) The authorized capital stock of the Company consist of 100,000 shares of Company Common Stock and no preferred stock. As of the date of this Agreement, 100 shares of Company Common Stock are issued and outstanding, all of which are validly issued. The Stockholder is the sole owner of all of the outstanding Company Common Stock. Schedule 2.3(a) hereto contains a list of all Persons who hold options to purchase shares of Company Common Stock, the number of shares purchasable under such option, the vesting schedule relating to such option, the expiration date of such option and the per-share exercise price of such option. Except as set forth in Schedule 2.3(a) hereto, as of the date of this Agreement, no shares of Company Common Stock are reserved for issuance upon the exercise of outstanding options or other rights. There are no commitments or agreements of any character to which the Company is bound obligating the Company to accelerate the vesting of any option to purchase Company Common Stock as a result of the Company Merger. All outstanding shares of Company Common Stock and options to acquire Company Common Stock have been issued and granted in compliance with (x) all applicable securities laws and (in all material respects) other applicable laws and regulations, and (y) all requirements set forth in any applicable Company Contracts (as defined in Section 2.19). The Company has heretofore delivered to Parent or Parent s counsel true and accurate copies of the forms of documents used for the issuance of options to acquire Company Common Stock.
- (b) Except as set forth in *Schedule 2.3(a)* hereto or as set forth in Section 2.3(a) hereof, there are no subscriptions, options, warrants, equity securities, partnership interests or similar ownership interests, calls, rights (including preemptive rights), commitments or agreements of any character to which the Company is a party or by which it is bound obligating the Company to issue, deliver or sell, or cause to be issued, delivered or sold, or repurchase, redeem or otherwise acquire, or cause the repurchase, redemption or acquisition of, any shares of capital stock or other ownership interests of the Company or obligating the Company to grant, extend, accelerate the vesting of or enter into any such subscription, option, warrant, equity security, call, right, commitment or agreement.

(c) Except as contemplated by this Agreement and except as set forth in *Schedule 2.3(c)* hereto, there are no registration rights, and there is no voting trust, proxy, rights plan, antitakeover plan or other agreement or understanding to which the Company is a party or by which the Company is bound with respect to any ownership interest or equity security of any class of the Company.

- (d) No shares of capital stock of the Company are unvested or subject to a repurchase option, risk of forfeiture or other condition under any applicable agreement with the Company.
- (e) The authorized and outstanding capital stock of each Subsidiary is set forth in *Schedule 2.3(d)* hereto. The Company owns all of the outstanding equity securities of each Subsidiary, free and clear of all Liens, either directly or indirectly through one or more other Subsidiaries. There are no outstanding options, warrants or other rights to purchase securities of any Subsidiary.
- 2.4 Authority Relative to this Agreement. Each of the Company and the Stockholder has all necessary power and authority to execute and deliver this Agreement and to perform its obligations hereunder and, to consummate the transactions contemplated hereby (including the Company Merger). The execution and delivery of this Agreement and the consummation by each of the Company and the Stockholder of the transactions contemplated hereby (including the Company Merger) have been duly and validly authorized by all necessary action on the part of the Company and the Stockholder (including the approval by its board of directors and stockholders, subject in all cases to the satisfaction of the terms and conditions of this Agreement, including the conditions set forth in Article VI), and no other proceedings on the part of the Company or the Stockholder or its respective directors, officers or stockholders are necessary to authorize this Agreement or to consummate the transactions contemplated hereby pursuant to Applicable Corporate Law and the terms and conditions of this Agreement. This Agreement has been duly and validly executed and delivered by each of the Company and the Stockholder and, assuming the due authorization, execution and delivery thereof by the other parties hereto, constitutes the legal and binding obligation of each of the Company and the Stockholder, enforceable against the Company and the Stockholder in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization or other similar laws affecting the enforcement of creditors rights generally and by general principles of equity.
  - 2.5 No Conflict; Required Filings and Consents. Except as set forth in Schedule 2.5 hereto:
- (a) The execution and delivery of this Agreement by each of the Company and the Stockholder do not, and the performance of this Agreement by each of the Company and the Stockholder shall not, (i) conflict with or violate the Company s or the Stockholder s Charter Documents, (ii) conflict with or violate any Legal Requirements (as defined in Section 10.2(b)), (iii) result in any breach of or constitute a default (or an event that with notice or lapse of time or both would become a default) under, or materially impair the Company s rights or alter the rights or obligations of any third party under, or give to others any rights of termination, amendment, acceleration or cancellation of, or result in the creation of a Lien on any of the properties or assets of the Company pursuant to any Company Contracts or (iv) except as contemplated by the Farm Purchase Agreement (as defined hereinafter), the IP Transfer Agreement (as defined hereinafter) and the transactions contemplated thereunder, result in the triggering, acceleration or increase of any payment to any Person pursuant to any Company Contract, including any change in control or similar provision of any Company Contract, except, with respect to clauses (ii), (iii) or (iv), for any such conflicts, violations, breaches, defaults, triggerings, accelerations, increases or other occurrences that would not, individually and in the aggregate, have a Material Adverse Effect on the Company.
- (b) The execution and delivery of this Agreement by each of the Company and the Stockholder does not, and the performance of its obligations hereunder will not, require any consent, approval, authorization or permit of, or filing with or notification to, any Governmental Entity or other third party (including, without limitation, lenders and lessors), except (i) for applicable requirements, if any, of the Securities Act, the Exchange Act or Blue Sky Laws, and the rules and regulations thereunder, and appropriate documents received from or filed with the relevant authorities of other jurisdictions in which the Company is licensed or qualified to do business, (ii) for the filing of any notifications required under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the *HSR Act*), and the expiration of the required waiting period thereunder,(iii) the consents, approvals, authorizations and permits described

in *Schedule 2.5(b)* ( *Governmental and Third Party Consents* ) and (iv) where the failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Company or the Stockholder or, after the Closing, the Surviving Pubco or Surviving Subsidiary, or prevent consummation of the Company Merger or otherwise prevent the parties hereto from performing their obligations under this Agreement.

2.6 Compliance. The Company has all necessary licenses and registrations in each state and other jurisdiction where the Company s current business activities require such licensing and registration and is in good standing with respect to such licenses and registrations except where the failure, individually or in the aggregate, is not reasonably likely to have a Material Adverse Effect on the Company. The Company has complied with and is not in violation of any Legal Requirements with respect to the conduct of its business as currently conducted except for failures to comply or violations which, individually or in the aggregate, have not had and are not reasonably likely to have a Material Adverse Effect on the Company. The businesses and activities of the Company have not been and are not being conducted in violation of any Legal Requirements except where the failure, individually or in the aggregate, is not reasonably likely to have a Material Adverse Effect on the Company. Except as set forth in *Schedule 2.6*, no written notice of non-compliance with any Legal Requirements has been received by the Company (and neither the Company nor the Stockholder has any knowledge of any such notice delivered to any other Person).

#### 2.7 Financial Statements.

(a) The Company has provided to Parent a correct and complete copies of the audited consolidatedfinancial statements (including any related notes thereto) of the Company and its Subsidiaries for the period from the Company s inception through and including August 18, 2009 (the *Audited Financial Statements*). The Audited Financial Statements were prepared in accordance with the published rules and regulations of any applicable Governmental Entity and with generally accepted accounting principles of the United States ( *U.S. GAAP*) applied on a consistent basis throughout the periods involved (except as may be indicated in the notes thereto), and each fairly presents in all material respects the financial position of the Company and its Subsidiaries at the respective dates thereof and the results of their operations and cash flows for the periods indicated.

#### (b) [Reserved]

- (c) The books of account, minute books and transfer ledgers and other similar books and records of the Company and its Subsidiaries have been maintained in accordance with good business practice, are complete and accurate in all material respects and there have been no material transactions that are required to be set forth therein and which have not been so set forth.
- (d) Except as otherwise noted in the Audited Financial Statements, the accounts and notes receivable of the Company and its Subsidiaries reflected on the balance sheets included in the Audited Financial Statements: (i) arose from bona fide sales transactions in the ordinary course of business and are payable on ordinary trade terms, (ii) are legal, valid and binding obligations of the respective debtors enforceable in accordance with their terms, except as such may be limited by bankruptcy, insolvency, reorganization, or other similar laws affecting creditors—rights generally, and by general equitable principles, (iii) are not subject to any valid set-off or counterclaim except to the extent set forth in such balance sheet contained therein, (iv) are collectible in the ordinary course of business in the aggregate recorded amounts thereof, net of any applicable reserve reflected in such balance sheet referenced above, and (v) are not the subject of any actions or proceedings brought by or on behalf of the Company or any of its Subsidiaries.
- 2.8 No Undisclosed Liabilities. Except as set forth in Schedule 2.8 hereto and as contemplated by the Farm Purchase Agreement and the transactions contemplated thereunder, the Company and its Subsidiaries have no liabilities (absolute, accrued, contingent or otherwise) of a nature required to be disclosed on a balance sheet or in the related notes to financial statements that are, individually or in the aggregate, material to the business, results of operations or financial condition of the Company and its Subsidiaries, except: (i) liabilities provided for in or otherwise disclosed in the balance sheet included in the Audited Financial Statements or in the notes thereto and (ii) liabilities incurred in a commercially reasonable manner and arising since September 1, 2009, none of which, individually or in the aggregate, would have a Material Adverse Effect on the Company.

2.9 Absence of Certain Changes or Events. Since September 1, 2009, there has not been: (i) any Material Adverse Effect on the Company, (ii) any declaration, setting aside or payment of any dividend on, or other distribution (whether in cash, stock or property) in respect of, any of the Company s equity interests, or any purchase, redemption or other acquisition by the Company of any of the Company s equity interests or

any other securities or any options, warrants, calls or rights to acquire any such shares or other securities, (iii) any split, combination or reclassification of any of the Company s equity interests, (iv) any granting by the Company of any increase in compensation or fringe benefits, or any payment by the Company of any bonus, or any granting by the Company of any increase in severance or termination pay or any entry by the Company into any currently effective employment, severance, termination or indemnification agreement or any agreement the benefits of which are contingent or the terms of which are materially altered upon the occurrence of a transaction involving the Company of the nature contemplated hereby, (v) except as contemplated by the IP Transfer Agreement and the transactions contemplated thereunder, entry by the Company into any licensing or other agreement with regard to the acquisition or disposition of any Intellectual Property (as defined in Section 2.18 hereof) or consent with respect to any licensing agreement filed or required to be filed by the Company with respect to any Governmental Entity, (vi) any material change by the Company in its accounting methods, principles or practices except as required by concurrent changes in U.S. GAAP, (vii) any change in the auditors of the Company, (viii) except for the Capital Raise (as defined hereinafter), any issuance of equity interests of the Company, (ix) any revaluation by the Company of any of its assets, including, without limitation, writing down the value of capitalized inventory or writing off notes or accounts receivable or any sale of assets of the Company that is not commercially reasonable, or (x) any agreement, whether written or oral, to do any of the foregoing. To the knowledge of the Stockholder, no event has occurred and no circumstances exist as of the date of this Agreement that would reasonably be deemed likely to prevent or prohibit the Stockholder from consummating the transactions contemplated by this Agreement.

2.10 *Litigation*. There are no claims, suits, actions or proceedings pending or, to the knowledge of the Company or the Stockholder, threatened against the Company before any court, governmental department, commission, agency, instrumentality or authority, or any arbitrator. There are no claims, suits, actions or proceedings pending or, to the knowledge of the Company or the Stockholder, threatened against the Stockholder before any court, governmental department, commission, agency, instrumentality or authority, or any arbitrator that would be reasonably likely to prevent or prohibit the Stockholder from consummating the transactions contemplated by this Agreement.

#### 2.11 Employee Benefit Plans.

(a) Schedule 2.11(a) lists all employee compensation, incentive, fringe or benefit plans, programs, policies, commitments or other arrangements (whether or not set forth in a written document) covering any active or former employee, director or consultant of the Company, or any trade or business (whether or not incorporated) which is under common control with the Company, with respect to which the Company has liability (individually, a *Plan*, and, collectively, the Plans ). All Plans have been maintained and administered in all material respects in compliance with their respective terms and with the requirements prescribed by any and all statutes, orders, rules and regulations which are applicable to such Plans, and all liabilities with respect to the Plans have been properly reflected in the financial statements and records of the Company. No suit, action or other litigation (excluding claims for benefits incurred in the ordinary course of Plan activities) has been brought, or, to the knowledge of the Company, is threatened, against or with respect to any Plan. There are no audits, inquiries or proceedings pending or, to the knowledge of the Company or the Stockholder, threatened by any governmental agency with respect to any Plan. All contributions, reserves or premium payments required to be made or accrued as of the date hereof to the Plans have been timely made or accrued. The Company does not have any plan or commitment to establish any new Plan, to modify any Plan (except to the extent required by law or to conform any such Plan to the requirements of any applicable law, in each case as previously disclosed to Parent in writing, or as required by this Agreement), or to enter into any new Plan. Except as disclosed in Schedule 2.11(a), each Plan can be amended, terminated or otherwise discontinued after the Closing in accordance with its terms, without liability to Surviving Pubco or Surviving Subsidiary (other than ordinary administration expenses and expenses for benefits accrued but not yet paid).

(b) Except as disclosed in *Schedule 2.11(b)* hereto, neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated hereby will (i) result in any payment (including severance, unemployment compensation, golden parachute, bonus or otherwise) becoming due to any stockholder, director or employee of the Company under any Plan or otherwise, (ii) materially increase any benefits otherwise payable under any Plan, or (iii) result in the acceleration of the time of payment or vesting of any such benefits.

#### 2.12 Labor Matters.

- (a) Except as set forth on *Schedule 2.12(a)*, the Company is not a party to any collective bargaining agreement or other labor union contract applicable to persons employed by the Company nor does the Company or the Stockholder know of any activities or proceedings of any labor union to organize any such employees. There are no pending grievances or similar proceedings involving the Company and any of its employees subject to a collective bargaining agreement or other labor union contract and there are no continuing obligations of the Company pursuant to the resolution of any such proceeding that is no longer pending.
- (b) Except as provided for in the collective bargaining agreements and labor union contracts set forth on *Schedule 2.12(b)*, each employee and consultant of the Company is terminable at will subject to applicable notice periods as set forth by law or in the employment agreement and there are no agreements or understandings between the Company and any of its employees or consultants that their employment or services will be for any particular period. The Company is not aware that any of its officers or key employees intends to terminate his or her employment with the Company. The Company is in compliance in all material respects and, to the Company s and the Stockholder s knowledge, each of its employees and consultants is in compliance in all material respects, with the terms of the respective employment and consulting agreements between the Company and such individuals. There are not, and there have not been, any oral or informal arrangements, commitments or promises between the Company and any employees or consultants of the Company that have not been documented as part of the formal written agreements between any such individuals and the Company that have been made available to Parent.
- (c) The Company is in compliance in all material respects with all Legal Requirements applicable to its employees, respecting employment, employment practices, terms and conditions of employment and wages and hours and is not liable for any arrears of wages or penalties with respect thereto. The Company s obligations to provide statutory severance pay to its employees are fully funded or accrued on the Audited Financial Statements and neither the Company nor the Stockholder has any knowledge of any circumstance that could give rise to any valid claim by a current or former employee of the Company for compensation on termination of employment (beyond the statutory severance pay to which employees are entitled). All amounts that the Company is legally or contractually required either (x) to deduct from its employees—salaries or to transfer to such employees—pension or provident, life insurance, incapacity insurance, continuing education fund or other similar funds or (y) to withhold from its employees—salaries and benefits and to pay to any Governmental Entity as required by applicable Legal Requirements have, in each case, been duly deducted, transferred, withheld and paid, and the Company does not have any outstanding obligation to make any such deduction, transfer, withholding or payment. There are no pending, or to the Company—s and the Stockholder—s knowledge, threatened or reasonably anticipated claims or actions against the Company by any employee in connection with such employee—s employment or termination of employment by the Company.
- (d) No employee or former employee of the Company is owed any wages, benefits or other compensation for past services (other than wages, benefits and compensation accrued in the ordinary course of business during the current pay period and any accrued benefits for services, which by their terms or under applicable law, are payable in the future, such as accrued vacation, recreation leave and severance pay).
- 2.13 Restrictions on Business Activities. Except as disclosed in Schedule 2.13 hereto, there is no agreement, commitment, judgment, injunction, order or decree binding upon the Company or its assets or to which the Company is a party which has or could reasonably be expected to have the effect of prohibiting or materially impairing any current business practice or operation of the Company, any acquisition of property by the Company, or the conduct of business by the Company as currently conducted, other than such effects, individually or in the aggregate, which have not had and could not reasonably be expected to have a Material Adverse Effect on the Company.

2.14 Title to Property.

(a) All real property owned by the Company (including improvements and fixtures thereon, easements and rights of way) is shown or reflected on the balance sheet of the Company included in the Audited Financial Statements. The Company has good, valid and marketable fee simple title to the real property owned by it, and except as set forth in the Audited Financial Statements or on *Schedule 2.14(a)* hereto, all of such

real property is held free and clear of (i) all leases, licenses and other rights to occupy or use such real property and (ii) all Liens, rights of way, easements, restrictions, exceptions, variances, reservations, covenants or other title defects or limitations of any kind, other than liens for Taxes not yet due and payable and such liens or other imperfections of title, if any, as do not materially detract from the value of or materially interfere with the present use of the property affected thereby. *Schedule 2.14(a)* hereto also contains a list of all options or other contracts under which the Company has a right to acquire or the obligation to sell any interest in real property.

- (b) All leases of real property held by the Company, and all personal property and other property and assets of the Company owned, used or held for use in connection with the business of the Company (the *Personal Property*) are shown or reflected on the balance sheet included in the Audited Financial Statements to the extent required by U.S. GAAP, as of the dates of such Audited Financial Statements, other than those entered into or acquired on or after the date of the Audited Financial Statements in a commercially reasonable manner. *Schedule 2.14(b)* hereto contains a list of all leases of real property and leases of Personal Property held by the Company. The Company has good and marketable title to the Personal Property owned by it, and all such Personal Property is in each case held free and clear of all Liens, except for Liens disclosed in the Audited Financial Statements or in *Schedule 2.14(b)* hereto, none of which Liens is reasonably expected to have, individually or in the aggregate, a Material Adverse Effect on such property or on the present or contemplated use of such property in the businesses of the Company.
- (c) All leases pursuant to which the Company leases from others are valid and effective in accordance with their respective terms, and there is not, under any of such leases, any existing material default or event of default of the Company or, to the Company s knowledge, any other party (or any event which with notice or lapse of time, or both, would constitute a material default), except where the lack of such validity and effectiveness or the existence of such default or event of default could not reasonably be expected to have a Material Adverse Effect on the Company.
- (d) The Company is in possession of, or has valid and effective rights to, all properties, assets and rights (including Intellectual Property) required, in all material respects for the effective conduct of its business, as it is currently operated.

#### 2.15 *Taxes*.

- (a) *Definition of Taxes*. For the purposes of this Agreement, *Tax* or *Taxes* refers to any and all federal, state, local and foreign taxes, including, without limitation, gross receipts, income, profits, sales, use, occupation, value added, ad valorem, transfer, franchise, withholding, payroll, recapture, employment, excise and property taxes, assessments, governmental charges and duties together with all interest, penalties and additions imposed with respect to any such amounts and any obligations under any agreements or arrangements with any other Person with respect to any such amounts and including any liability of a predecessor entity for any such amounts.
  - (b) Tax Returns and Audits. Except as set forth in Schedule 2.15 hereto:
- (i) The Company has timely filed all federal, state, local and foreign returns, estimates, information statements and reports relating to Taxes ( *Returns* ) required to be filed by the Company with any Tax authority prior to the date hereof, except such Returns that are not material to the Company. All such Returns are true, correct and complete in all material respects. The Company has paid all Taxes shown to be due and payable on such Returns.
- (ii) All Taxes that the Company is required by law to withhold or collect have been duly withheld or collected, and have been timely paid over to the proper governmental authorities to the extent due and payable.

(iii) The Company has not been delinquent in the payment of any Tax (except for Taxes being contested in good faith for which adequate reserves have been established, all of which are listed on *Schedule 2.15(b)* hereto) nor is there any Tax deficiency outstanding, proposed or assessed against the Company, nor has the Company executed any unexpired waiver of any statute of limitations on or extending the period for the assessment or collection of any Tax. The Company has complied with all

Legal Requirements with respect to payments made to third parties and the withholding of any payment of withheld Taxes and has timely withheld from employee wages and other payments and timely paid over in full to the proper Taxing authorities all amounts required to be so withheld and paid over for all periods.

- (iv) To the knowledge of the Company and the Stockholder, no audit or other examination of any Return of the Company by any Tax authority is presently in progress, nor has the Company been notified of any request for such an audit or other examination.
- (v) No adjustment relating to any Returns filed by the Company has been proposed in writing, formally or informally, by any Tax authority to the Company or any representative thereof.
- (vi) The Company has no liability for any unpaid Taxes which have not been accrued for or reserved on the Company s balance sheets included in the Audited Financial Statements, whether asserted or unasserted, contingent or otherwise, other than any liability for unpaid Taxes that may have accrued since the end of the most recent fiscal year (or the Company s inception, if more recent) in connection with the operation of the business of the Company in a commercially reasonable manner.
- (vii) Neither the Company nor the Stockholder has taken any action and does not know of any fact, agreement, plan or other circumstance that is reasonably likely to prevent the Mergers from qualifying as a reorganization within the meaning of Section 368(a) of the Code.

#### 2.16 Environmental Matters.

- (a) Except as disclosed in *Schedule 2.16* hereto: (i) the Company has complied with all applicable Environmental Laws (as defined below); (ii) the properties currently operated by the Company (including soils, groundwater, surface water, air, buildings or other structures) and, to the knowledge of the Stockholder and the Company, the property that is subject to the Farm Purchase Agreement (as defined in Section 4.2) are not contaminated with any Hazardous Substances (as defined below); (iii) the properties formerly owned, operated or constructed by the Company were not contaminated with Hazardous Substances during the period of ownership, operation or construction by the Company or, to the Company s knowledge, during any prior period; (iv) the Company is not subject to liability for any Hazardous Substance disposal or contamination on any third party or public property (whether above, on or below ground or in the atmosphere or water); (v) the Company has not been associated with any release or threat of release of any Hazardous Substance; (vi) the Company has not received any notice, demand, letter, claim or request for information alleging that the Company may be in violation of or liable under any Environmental Law; and (vii) the Company is not subject to any orders, decrees, injunctions or other arrangements with any Governmental Entity or subject to any indemnity or other agreement with any third party relating to liability under any Environmental Law or relating to Hazardous Substances.
- (b) As used in this Agreement, the term *Environmental Law* means any federal, state, local or foreign law, regulation, order, decree, permit, authorization, opinion, common law or agency requirement relating to: (A) the protection, investigation or restoration of the environment, health and safety, or natural resources; (B) the handling, use, presence, disposal, release or threatened release of any Hazardous Substance or (C) noise, odor, wetlands, pollution, contamination or any injury or threat of injury to persons or property.
- (c) As used in this Agreement, the term *Hazardous Substance* means any substance that is: (i) listed, classified or regulated pursuant to any Environmental Law; (ii) any petroleum product or by-product, asbestos-containing material, lead-containing paint or plumbing, polychlorinated biphenyls, radioactive materials or radon; or (iii) any other substance which is the subject of regulatory action by any Governmental Entity pursuant to any Environmental Law.

(d) *Schedule 2.16(d)* sets forth all environmental studies and investigations completed with respect to the Company and/or its Subsidiaries or their respective properties, assets or operations. All such written reports and material documentation relating to any such study or investigation has been provided by the Company to Parent.

2.17 Brokers; Third Party Expenses. Except as set forth in Schedule 2.17 hereto, neither the Company nor the Stockholder has incurred, nor will it incur, directly or indirectly, any liability for brokerage, finders fees, agent s commissions or any similar charges in connection with this Agreement or any transactions contemplated hereby.

#### 2.18 Intellectual Property.

- (a) Schedule 2.18 hereto contains a complete descriptive overview of the material Intellectual Property necessary in all material respects for the operation of the Company s business as currently proposed in the business section of the Proxy Statement/Prospectus. To the Company s and the Stockholder s knowledge, all Company Intellectual Property is (i) owned solely by the Company or (ii) sole ownership will be irrevocably and unconditionally assigned and transferred to the Company at Closing pursuant to the Deed of Assignment Relating to Intellectual Property ( IP Transfer Agreement ) in the form of Exhibit H hereto (which is being executed concurrently with this Agreement). The IP Transfer Agreement has been duly authorized by the Company and has been duly and validly executed and delivered by each party thereto and, to the knowledge of the Stockholder and the Company, comprises the binding obligation of each party thereto. No director, officer, consultant, agent, employee, equity holder or other party affiliated with the Company or any affiliate thereof (collectively, the Company Related Parties ) has any right or claim to such Company Intellectual Property. Except as described on Schedule 2.18, none of the Company Intellectual Property was developed under the terms of any agreement with any third party or using the assets or resources of any other party that would give such third party any right or claim to any ownership or economic or pecuniary interest in such Company Intellectual Property ( Third Party IP Rights ), including any agreement or understanding with any existing or prior employer or client of any of the Company Related Parties or any government or educational agency, institution or other body. All Third Party IP Rights are being waived or transferred to the Company, effective as of Closing, pursuant to the IP Transfer Agreement.
  - (b) For the purposes of this Agreement, the following terms have the following definitions:
- (i) *Intellectual Property* shall mean any or all of the following and all worldwide common law and statutory rights in, arising out of, or associated therewith: (i) patents and applications therefor and all reissues, divisions, renewals, extensions, provisionals, continuations and continuations-in-part thereof ( *Patents* ); (ii) inventions (whether patentable or not), invention disclosures, improvements, trade secrets, proprietary information, know how, systems, processes, plans, procedures, approaches, technology, technical data and customer lists, and all documentation relating to any of the foregoing; (iii) copyrights, copyright registrations and applications therefor, and all other rights corresponding thereto throughout the world ( *Copyrights* ); (iv) software and software programs; (v) the registered domain names naturaldairy.com and cullenagritech.com; (vi) industrial designs and any registrations and applications therefor; (vii) trade names, logos, common law trademarks and service marks, trademark and service mark registrations and applications therefor (collectively, *Trademarks* ); (viii) all databases and data collections and all rights therein; (ix) all moral and economic rights of authors and inventors, however denominated; and (x) any similar or equivalent rights to any of the foregoing (as applicable).
  - (ii) Company Intellectual Property shall mean the Intellectual Property known to the Company as necessary to implement the subject matter which is described in Schedule 2.18 hereto for the conduct of the Company s business and operations as currently conducted or as otherwise proposed in the business section of the Proxy Statement/Prospectus and such Intellectual Property as otherwise owned by the Company.
  - (iii) Registered Intellectual Property means all Intellectual Property that is the subject of an application, certificate, filing, registration or other document issued, filed with, or recorded by any government or other legal authority.

- (iv) Company Registered Intellectual Property means all of the Registered Intellectual Property owned by, or filed in the name of, the Company.
  - (v) Company Products means all current products or service offerings of the Company.

- (c) Except as contemplated by the IP Transfer Agreement and the transactions contemplated thereunder, no Company Intellectual Property or Company Product is subject to any material proceeding or outstanding decree, order, judgment, contract, license, agreement or stipulation restricting in any manner the use, transfer or licensing thereof by the Company, or which may affect the validity, use or enforceability of such Company Intellectual Property or Company Product, which in any such case could reasonably be expected to have a Material Adverse Effect on the Company.
- (d) Giving effect to the IP Transfer Agreement, the Company owns and has good and exclusive title to each material item of Company Intellectual Property owned by it free and clear of any Liens and the Company is the exclusive owner of all material registered Trademarks and Copyrights used in connection with the operation or conduct of the business of the Company including the sale of any products or the provision of any services by the Company.
- (e) To the Company s and the Stockholder s knowledge, the Company Intellectual Property does not currently infringe or misappropriate the Intellectual Property of any third party or constitute unfair competition or trade practices under the laws of any jurisdiction. The Company has not received any claims or threats from third parties alleging any such infringement, misappropriation or unfair competition or trade practices.

#### 2.19 Agreements, Contracts and Commitments.

- (a) Schedule 2.19 hereto sets forth a complete and accurate list of all Material Company Contracts (as hereinafter defined), specifying the parties thereto. For purposes of this Agreement, (i) the term Company Contracts shall mean all contracts, agreements, leases, mortgages, indentures, notes, bonds, licenses, permits, franchises, purchase orders, sales orders, and other understandings, commitments and obligations (including, without limitation, outstanding offers and proposals) of any kind, whether written or oral, to which the Company is a party or by or to which any of the properties or assets of the Company may be bound, subject or affected (including without limitation notes or other instruments payable to the Company) and (ii) the term Material Company Contracts shall mean (x) each Company Contract (A) providing for payments (present or future) to the Company in excess of \$100,000 in the aggregate, (B) any Company Contract relating to Company Intellectual Property, (C) any Company Contract relating to land purchasable by the Company or (D) under or in respect of which the Company presently has any liability or obligation of any nature whatsoever (absolute, contingent or otherwise) in excess of \$100,000, (y) each Company Contract that otherwise is or may be material to the businesses, operations, assets, condition (financial or otherwise) or prospects of the Company, and (z) the limitations of subclause (x) and subclause (y) notwithstanding, each of the following Company Contracts:
- (i) any mortgage, indenture, note, installment obligation or other instrument, agreement or arrangement for or relating to any borrowing of money by or from the Company and by or to any officer, director, shareholder or holder of derivative securities of the Company ( *Insider* );
- (ii) any mortgage, indenture, note, installment obligation or other instrument, agreement or arrangement for or relating to any borrowing of money from an Insider by the Company;
  - (iii) any guaranty, direct or indirect, by the Company, a Subsidiary or any Insider of the Company of any obligation for borrowings, or otherwise, excluding endorsements made for collection in a commercially reasonable manner;
    - (iv) any Company Contract of employment or management;
- (v) any Company Contract made other than in a commercially reasonable manner or (x) providing for the grant of any preferential rights to purchase or lease any asset of the Company or (y) providing for any right (exclusive or

non-exclusive) to sell or distribute, or otherwise relating to the sale or distribution of, any product or service of the Company;

(vi) any obligation to register any shares of the capital stock or other securities of the Company with any Governmental Entity;

(vii) any obligation to make payments, contingent or otherwise, arising out of the prior acquisition of the business, assets or stock of other Persons;

#### TABLE OF CONTENTS

(viii) any collective bargaining agreement with any labor union;

- (ix) any lease or similar arrangement for the use by the Company of real property or Personal Property where the annual lease payments are greater than \$100,000 (other than any lease of vehicles, office equipment or operating equipment made in a commercially reasonable manner);
- (x) any Company Contract granting or purporting to grant, or otherwise in any way relating to, any mineral rights or any other interest (including, without limitation, a leasehold interest) in real property;
- (xi) any Company Contract to which any Insider of the Company, or any entity owned or controlled by an Insider, is a party; and
  - (xii) any offer or proposal which, if accepted, would constitute any of the foregoing.
- (b) Each Material Company Contract was entered into at arms length and in a commercially reasonable manner, is in full force and effect and, to the Company s knowledge, is valid and binding upon and enforceable against each of the parties thereto, except insofar as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditors rights generally or by principles governing the availability of equitable remedies. To the Company s and the Stockholder s knowledge, no other party to a Material Company Contract is the subject of a bankruptcy or insolvency proceeding. True, correct and complete copies of all Material Company Contracts (or written summaries in the case of oral Material Company Contracts) have been heretofore delivered to Parent or Parent s counsel.
- (c) Except as set forth in *Schedule 2.19*, neither the Company nor, to the Company s knowledge, any other party thereto is in breach of or in default under, and no event has occurred which with notice or lapse of time or both would become a breach of or default under, any Material Company Contract, and no party to any Material Company Contract has given any written notice of any claim of any such breach, default or event, which, individually or in the aggregate, are reasonably likely to have a Material Adverse Effect on the Company. Each Material Company Contract that has not expired by its terms is in full force and effect.
- 2.20 Insurance. Schedule 2.20 sets forth the Company s insurance policies and fidelity and surety bonds covering the assets, business, equipment, properties, operations, employees, officers and directors (collectively, the Insurance Policies). The insurances provided by such Insurance Policies are adequate in amount and scope as required to be maintained by Material Company Contracts.

#### 2.21 Governmental Actions/Filings.

(a) Except as set forth in *Schedule 2.21(a)* and as contemplated by the Farm Purchase Agreement and the transactions contemplated thereunder, the Company has been granted and holds, and has made, all Governmental Actions/Filings (as defined below) (including, without limitation, Governmental Actions/Filings required for emission or discharge of effluents and pollutants into the air and the water) necessary to the conduct by the Company of its business as presently conducted or used or held for use by the Company, and true, complete and correct copies of which have heretofore been delivered to Parent. Each such Governmental Action/Filing is in full force and effect and, except as disclosed in *Schedule 2.21(a)* hereto, will not expire prior to December 31, 2009 and the Company is in material compliance with all of its obligations with respect thereto. No event has occurred and is continuing which requires or permits, or after notice or lapse of time or both would require or permit, and consummation of the transactions contemplated by this Agreement or any ancillary documents will not require or permit (with or without notice or lapse of time, or both), any modification or termination of any such Governmental Actions/Filings except such events

which, either individually or in the aggregate, would not have a Material Adverse Effect upon the Company. Except as set forth in *Schedule 2.21(a)* and as contemplated by the Farm Purchase Agreement and the transactions contemplated thereunder, no Governmental Action/Filing is necessary to be obtained, secured or made by the Company to enable it to continue to conduct its businesses and operations and use its properties and assets after the Closing in a manner which is consistent with current practice.

(b) Except as set forth in *Schedule 2.21(b)* and as contemplated by the Farm Purchase Agreement and the transactions contemplated thereunder, no Governmental Action/Filing is necessary to be obtained, secured or made by the Company to enable it to continue to conduct its businesses and operations and use its properties after the Closing in a manner which is consistent with current practice.

- (c) For purposes of this Agreement, the term *Governmental Action/Filing* shall mean any franchise, license, certificate of compliance, authorization, consent, order, permit, approval, consent or other action of, or any filing, registration or qualification with, any federal, state, municipal, foreign or other governmental, administrative or judicial body, agency or authority.
- (d) As of the date of this Agreement, no state agency or the Securities and Exchange Commission (SEC) have held that the products or services offered by the Company constitute a security under federal or state securities laws and the Company is not currently a party to any action or proceeding asserting that any product or service offered by the Company is a security and to the Company s knowledge, no such action or proceeding is currently threatened.
- 2.22 Interested Party Transactions. Except as set forth in the Schedule 2.22 hereto, no employee, director, officer or stockholder of the Company or a member of his or her immediate family is indebted to the Company, nor is the Company indebted (or committed to make loans or extend or guarantee credit) to any of such Persons, other than (i) for payment of salary for services rendered, (ii) reimbursement for reasonable expenses incurred on behalf of the Company, and (iii) for other employee benefits made generally available to all employees. Except as set forth in Schedule 2.22, to the Company s and the Stockholder s knowledge, none of such individuals has any direct or indirect ownership interest in any Person with whom the Company is affiliated or with whom the Company has a contractual relationship, or in any Person that competes with the Company, except that each employee, stockholder, officer or director of the Company and members of their respective immediate families may own less than 5% of the outstanding stock in publicly traded companies that may compete with the Company. Except as set forth in Schedule 2.22, to the knowledge of the Company and the Stockholder, no officer, director or stockholder or any member of their immediate families is, directly or indirectly, interested in any Material Company Contract with the Company (other than such contracts as relate to any such Person s ownership of capital stock or other securities of the Company or such Person s employment with the Company).
- 2.23 Board of Director and Certain Stockholder Approvals. The board of directors of the Company (including any required committee or subgroup thereof) and the Stockholder (and the voting shareholders of the Stockholder) have, as of the date of this Agreement, duly approved, this Agreement and the transactions contemplated hereby.
- 2.24 No Illegal or Improper Transactions. To the knowledge of the Company and the Stockholder, neither the Company or the Stockholder, nor any officer, director, stockholder, employee, agent or Affiliate thereof on the Company s or the Stockholder s behalf, has offered, paid or agreed to pay to any Person (including any governmental official) or solicited, received or agreed to receive from any such Person, directly or indirectly, any money or anything of value for the purpose or with the intent of (a) obtaining or maintaining business for the Company, (b) facilitating the purchase or sale of any product or service, or (c) avoiding the imposition of any fine or penalty, in any manner which is in violation of any applicable ordinance, regulation or law, the effect of which, individually or in the aggregate, would reasonably be expected to be materially adverse to the business, assets, prospects or financial condition of the Company, taken as a whole. To the Company s and the Stockholder s knowledge, no employee of the Company has provided or is providing information to any law enforcement agency regarding the commission or possible commission of any crime or the violation or possible violation of any applicable law. Neither the Company, has discharged, demoted, suspended, threatened, harassed or in any other manner discriminated against an employee of the Company in the terms and conditions of employment because of any act of such employee described in 18 U.S.C. § 1514A(a).
- 2.25 *Representations and Warranties Complete*. The representations and warranties of the Company and the Stockholder included in this Agreement and any list, statement, document or information set forth in, or attached to, any Schedule provided pursuant to this Agreement or delivered hereunder, are true and complete in all material

respects and do not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements contained therein not misleading, under the circumstance under which they were made.

#### **TABLE OF CONTENTS**

2.26 *Survival of Representations and Warranties*. The representations and warranties of the Company and the Stockholder set forth in this Agreement shall survive the Closing as set forth in Article VII.

#### ARTICLE III

#### REPRESENTATIONS AND WARRANTIES OF PARENT

Subject to the exceptions set forth in *Schedule 3* attached hereto (the *Parent Schedule* ), Parent represents and warrants to, and covenants with, the Company and the Stockholder as follows:

#### 3.1 Organization and Qualification.

- (a) Parent is a corporation duly incorporated, validly existing and in good standing under the laws of the State of Delaware and has the requisite corporate power and authority to own, lease and operate its assets and properties and to carry on its business as it is now being or currently planned by Parent to be conducted. Parent is in possession of all Approvals necessary to own, lease and operate the properties it purports to own, operate or lease and to carry on its business as it is now being or currently planned by Parent to be conducted, except where the failure to have such Approvals could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on Parent. Complete and correct copies of the Charter Documents of Parent, as amended and currently in effect, have been heretofore delivered to the Company. Parent is not in violation of any of the provisions of Parent s Charter Documents.
- (b) Parent is duly qualified or licensed to do business as a foreign corporation and is in good standing, in each jurisdiction where the character of the properties owned, leased or operated by it or the nature of its activities makes such qualification or licensing necessary, except for such failures to be so duly qualified or licensed and in good standing that could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on Parent.

#### 3.2 Subsidiaries.

- (a) Except for Holdco, which is a wholly owned subsidiary of Parent, and Merger Sub, which is a wholly owned subsidiary of Holdco, Parent has no direct or indirect Subsidiaries and does not own, directly or indirectly, any ownership, equity, profits or voting interest in any Person or have any agreement or commitment to purchase any such interest, and Parent has not agreed and is not obligated to make nor is bound by any written, oral or other agreement, contract, subcontract, lease, binding understanding, instrument, note, option, warranty, purchase order, license, sublicense, insurance policy, benefit plan, commitment or undertaking of any nature, as of the date hereof or as may hereafter be in effect under which it may become obligated to make, any future investment in or capital contribution to any other entity.
- (b) Holdco is a corporation duly incorporated, validly existing and in good standing under the law of the State of Delaware and has the requisite corporate power and authority to own, lease and operate its assets and properties and to carry on its business as it is now being or currently planned by Parent to be conducted. Complete and correct copies of the Charter Documents of Holdco, as amended and currently in effect, have been delivered to the Company or its counsel. Holdco is not in violation of any of the material provisions of its Charter Documents.
- (c) Merger Sub is a corporation duly incorporated, validly existing and in good standing under the law of the State of Georgia and has the requisite corporate power and authority to own, lease and operate its assets and properties and to carry on its business as it is now being or currently planned by Parent to be conducted. Complete and correct copies of

- the Charter Documents of Merger Sub, as amended and currently in effect, have been delivered to the Company or its counsel. Merger Sub is not in violation of any of the material provisions of its Charter Documents.
- (d) Each of Holdco and Merger Sub (together, the *Parent Subsidiaries*) has no assets or properties of any kind, does not now conduct and has never conducted any business, and has and will have at the Closing no obligations or liabilities of any nature whatsoever except such obligations and liabilities as are imposed under this Agreement.

#### 3.3 Capitalization.

- (a) As of the date of this Agreement, the authorized capital stock of Parent consists of 160,000,000 shares of Parent Common Stock and 1,000,000 shares of preferred stock, par value \$.0001 per share ( *Parent Preferred Stock* ), of which 69,000,000 shares of Parent Common Stock and no shares of Parent Preferred Stock are issued and outstanding, all of which are validly issued, fully paid and nonassessable.
- (b) Except as set forth in *Schedule 3.3(b)*, (i) no shares of Parent Common Stock or Parent Preferred Stock are reserved for issuance upon the exercise of outstanding options to purchase Parent Common Stock or Parent Preferred Stock granted to employees of Parent or other parties ( *Parent Stock Options* ) and there are no outstanding Parent Stock Options; (ii) no shares of Parent Common Stock or Parent Preferred Stock are reserved for issuance upon the exercise of outstanding warrants to purchase Parent Common Stock or Parent Preferred Stock; and (iii) no shares of Parent Common Stock or Parent Preferred Stock are reserved for issuance upon the conversion of the Parent Preferred Stock or any outstanding convertible notes, debentures or securities ( *Parent Convertible Securities* ). All shares of Parent Common Stock and Parent Preferred Stock subject to issuance as aforesaid, upon issuance on the terms and conditions specified in the instrument pursuant to which they are issuable, will be duly authorized, validly issued, fully paid and nonassessable. All outstanding shares of Parent Common Stock and all outstanding Parent Warrants have been issued and granted in compliance with (x) all applicable securities laws and (in all material respects) other applicable laws and regulations, and (y) all requirements set forth in any applicable Parent Contracts (as defined in Section 3.19). Parent has heretofore delivered to the Company true, complete and accurate copies of the Parent Warrants, including any and all documents and agreements relating thereto.
  - (c) The shares of Holdco Common Stock to be issued by Holdco in connection with the Mergers, upon issuance in accordance with the terms of this Agreement, will be duly authorized and validly issued and such shares of Holdco Common Stock will be fully paid and nonassessable.
- (d) Except as set forth in *Schedule 3.3(d)* or as contemplated by this Agreement or the Parent SEC Reports (as defined in Section 3.7), there are no registrations rights, and there is no voting trust, proxy, rights plan, antitakeover plan or other agreements or understandings to which the Parent or Holdco is a party or by which the Parent or Holdco is bound with respect to any equity security of any class of the Parent or Holdco.
- (e) Except as set forth on *Schedule 3.3(e)*, there are no subscriptions, options, warrants, equity securities, partnership interests or similar ownership interests, calls, rights (including preemptive rights), commitments or agreements of any character to which Holdco is a party or by which it is bound obligating Holdco to issue, deliver or sell, or cause to be issued, delivered or sold, or repurchase, redeem or otherwise acquire, or cause the repurchase, redemption or acquisition of, any shares of capital stock or ownership interests of Holdco or obligating Holdco to grant, extend, accelerate the vesting of or enter into any such subscription, option, warrant, equity security, call, right, commitment or agreement.
- (f) Except as provided for in this Agreement or as set forth in *Schedule 3.3(f)*, as a result of the consummation of the transactions contemplated hereby, no shares of capital stock, warrants, options or other securities of the Parent or Holdco are issuable and no rights in connection with any shares, warrants, options or other securities of the Parent or Holdco accelerate or otherwise become triggered (whether as to vesting, exercisability, convertibility or otherwise).
- 3.4 *Authority Relative to this Agreement*. Each of Parent and the Parent Subsidiaries has full corporate power or other and authority to: (i) execute, deliver and perform this Agreement, and each ancillary document that Parent or Parent Subsidiaries have executed or delivered or is to execute or deliver pursuant to this Agreement, and (ii) carry out Parent s and the Parent Subsidiaries obligations hereunder and thereunder and, to consummate the transactions

contemplated hereby (including the Parent Merger). The execution and delivery of this Agreement by Parent and Parent Subsidiaries and the consummation by Parent and the Parent Subsidiaries of the transactions contemplated hereby (including the Merger) have been duly and validly authorized by all necessary corporate or other action on the part of Parent and the Parent Subsidiaries (including the approval by their respective boards of directors or similar body), and no other corporate or other proceedings on the part of Parent or the Parent Subsidiaries are necessary to authorize this Agreement or to consummate the transactions contemplated hereby, other than the Parent Stockholder Approval (as defined in Section

5.1(a)). This Agreement has been duly and validly executed and delivered by Parent and each of the Parent Subsidiaries and, assuming the due authorization, execution and delivery thereof by the other parties hereto, constitutes the legal and binding obligation of each of Parent and the Parent Subsidiaries, enforceable against Parent and the Parent Subsidiaries in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization or other similar laws affecting the enforcement of creditors rights generally and by general principles of equity.

3.5 No Conflict; Required Filings and Consents.

- (a) The execution and delivery of this Agreement by Parent and the Parent Subsidiaries do not, and the performance of this Agreement by Parent and the Parent Subsidiaries shall not: (i) conflict with or violate Parent s or Parent Subsidiaries Charter Documents, (ii) conflict with or violate any Legal Requirements, (iii) result in any breach of or constitute a default (or an event that with notice or lapse of time or both would become a default) under, or materially impair Parent s or the Parent Subsidiaries rights or alter the rights or obligations of any third party under, or give to others any rights of termination, amendment, acceleration or cancellation of, or result in the creation of a Lien on any of the properties or assets of Parent pursuant to, any Parent Contracts or (iv) result in the triggering, acceleration or increase of any payment to any Person pursuant to any Parent Contract, including any change in control or similar provision of any Parent Contracts except, with respect to clauses (ii), (iii) or (iv), for any such conflicts, violations, breaches, defaults or other occurrences that would not, individually and in the aggregate, have a Material Adverse Effect on Parent.
- (b) The execution and delivery of this Agreement by Parent and the Parent Subsidiaries do not, and the performance of their respective obligations hereunder will not, require any consent, approval, authorization or permit of, or filing with or notification to, any Governmental Entity or third party, except (i) for applicable requirements, if any, of the Securities Act, the Exchange Act, Blue Sky Laws, and the rules and regulations thereunder, and appropriate documents with the relevant authorities of other jurisdictions in which Parent or any Parent Subsidiary is qualified to do business, (ii) for the filing of any notifications required under the HSR Act and the expiration of the required waiting period thereunder, (iii) the qualification of Holdco as a foreign corporation in those jurisdictions in which the business of Holdco makes such qualification necessary, and (iv) where the failure to obtain such consents, approvals, authorizations or permits, or to make such filings or notifications, would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on Parent, or after the Closing, the Surviving Pubco or the Surviving Subsidiary, or prevent consummation of the Parent Merger or otherwise prevent the parties hereto from performing their obligations under this Agreement.
- 3.6 Compliance. Each of Parent and the Parent Subsidiaries has complied with, and is not in violation of, any Legal Requirements with respect to the conduct of its business, or the ownership or operation of its business, except where the failure, individually or in the aggregate, is not reasonably likely to have a Material Adverse Effect on Parent or the Parent Subsidiaries. The business and activities of Parent and the Parent Subsidiaries have not been and are not being conducted in violation of any Legal Requirements, except where the failure, individually or in the aggregate, is not reasonably likely to have a Material Adverse Effect on Parent or the Parent Subsidiaries. Neither Parent nor any Parent Subsidiary is in default or violation of any term, condition or provision of its Charter Documents. No written notice of non-compliance with any Legal Requirements has been received by Parent or any Parent Subsidiary.

3.7 SEC Filings; Financial Statements.

(a) Parent has made available to the Company and the Stockholder a correct and complete copy of each report, registration statement and definitive proxy statement filed by Parent with the SEC (the *Parent SEC Reports*), which are all the forms, reports and documents required to be filed by Parent with the SEC prior to the date of this

Agreement. All Parent SEC Reports required to be filed by Parent since its inception were filed in a timely manner. As of their respective dates the Parent SEC Reports: (i) were prepared in accordance and complied in all material respects with the requirements of the Securities Act or the Exchange Act, as the case may be, and the rules and regulations of the SEC thereunder applicable to such Parent SEC Reports, and (ii) did not at the time they were filed (and if amended or superseded by a filing prior to the date of this Agreement then on the date of such filing and as so amended or superseded) contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the

statements therein, in light of the circumstances under which they were made, not misleading. Except to the extent set forth in the preceding sentence, Parent makes no representation or warranty whatsoever concerning any Parent SEC Report as of any time other than the date or period with respect to which it was filed.

(b) Except as set forth in *Schedule 3.7(b)*, each set of financial statements (including, in each case, any related notes thereto) contained in Parent SEC Reports, including each Parent SEC Report filed after the date hereof until the Closing, complied or will comply as to form in all material respects with the published rules and regulations of the SEC with respect thereto, was or will be prepared in accordance with U.S. GAAP applied on a consistent basis throughout the periods involved (except as may be indicated in the notes thereto or, in the case of unaudited statements, do not contain footnotes as permitted by Form 10-Q of the Exchange Act) and each fairly presents or will fairly present in all material respects the financial position of Parent at the respective dates thereof and the results of its operations and cash flows for the periods indicated, except that the unaudited interim financial statements were, are or will be subject to normal adjustments which were not or are not expected to have a Material Adverse Effect on Parent taken as a whole.

3.8 No Undisclosed Liabilities. Parent has no liabilities (absolute, accrued, contingent or otherwise) of a nature required to be disclosed on a balance sheet or in the related notes to the financial statements included in Parent SEC Reports that are, individually or in the aggregate, material to the business, results of operations or financial condition of Parent, except (i) liabilities provided for in or otherwise disclosed in Parent SEC Reports filed prior to the date hereof, and (ii) liabilities incurred since June 30, 2009 in the ordinary course of business, none of which would have a Material Adverse Effect on Parent.

3.9 Absence of Certain Changes or Events. Except as set forth in Parent SEC Reports filed prior to the date of this Agreement, and except as contemplated by this Agreement, since June 30, 2009, there has not been: (i) any Material Adverse Effect on Parent, (ii) any declaration, setting aside or payment of any dividend on, or other distribution (whether in cash, stock or property) in respect of, any of Parent s capital stock, or any purchase, redemption or other acquisition by Parent of any of Parent s capital stock or any other securities of Parent or any options, warrants, calls or rights to acquire any such shares or other securities, (iii) any split, combination or reclassification of any of Parent s capital stock, (iv) any granting by Parent of any increase in compensation or fringe benefits, except for normal increases of cash compensation in the ordinary course of business consistent with past practice, or any payment by Parent of any bonus, except for bonuses made in the ordinary course of business consistent with past practice, or any granting by Parent of any increase in severance or termination pay or any entry by Parent into any currently effective employment, severance, termination or indemnification agreement or any agreement the benefits of which are contingent or the terms of which are materially altered upon the occurrence of a transaction involving Parent of the nature contemplated hereby, (v) entry by Parent into any licensing or other agreement with regard to the acquisition or disposition of any Intellectual Property other than licenses in the ordinary course of business consistent with past practice or any amendment or consent with respect to any licensing agreement filed or required to be filed by Parent with respect to any Governmental Entity, (vi) any material change by Parent in its accounting methods, principles or practices, except as required by concurrent changes in U.S. GAAP, (vii) any change in the auditors of Parent, (viii) any issuance of capital stock of Parent, (ix) any revaluation by Parent of any of its assets, including, without limitation, writing down the value of capitalized inventory or writing off notes or accounts receivable or any sale of assets of Parent other than in the ordinary course of business, or (x) any agreement, whether written or oral, to do any of the foregoing.

3.10 *Litigation*. There are no claims, suits, actions or proceedings pending or to Parent s knowledge, threatened against Parent, before any court, governmental department, commission, agency, instrumentality or authority, or any arbitrator.

3.11 Employee Benefit Plans. Except as may be contemplated by the Holdco Plan (as defined in Section 5.1(a)), Parent does not maintain, and has no liability under, any employee compensation, incentive, fringe or benefit plans, programs, policies, commitments or other arrangements (whether or not set forth in a written document) covering any active or former employee, director or consultant of Parent, or any trade or business (whether or not incorporated) which is under common control with Parent, with respect to which Parent has liability, and neither the execution and delivery of this Agreement nor the consummation of the

transactions contemplated hereby will (i) result in any payment (including severance, unemployment compensation, golden parachute, bonus or otherwise) becoming due to any stockholder, director or employee of Parent, or (ii) result in the acceleration of the time of payment or vesting of any such benefits.

- 3.12 *Labor Matters*. Parent is not a party to any collective bargaining agreement or other labor union contract applicable to persons employed by Parent and Parent does not know of any activities or proceedings of any labor union to organize any such employees. Parent has no employees and has had no employees since inception.
- 3.13 Business Activities. Since its organization, Parent has not conducted any business activities other than activities directed toward the accomplishment of a business combination. Except as set forth in the Parent Charter Documents, there is no agreement, commitment, judgment, injunction, order or decree binding upon Parent or to which Parent is a party which has or could reasonably be expected to have the effect of prohibiting or materially impairing any business practice of Parent, any acquisition of property by Parent or the conduct of business by Parent as currently conducted other than such effects, individually or in the aggregate, which have not had and could not reasonably be expected to have, a Material Adverse Effect on Parent.
- 3.14 *Title to Property*. Parent does not own or lease any real property or personal property. Except as set forth in *Schedule 3.14*, there are no options or other contracts under which Parent has a right or obligation to acquire or lease any interest in real property or personal property.
  - 3.15 Taxes. Except as set forth in Schedule 3.15 hereto:
- (a) Parent has timely filed all Returns required to be filed by Parent with any Tax authority prior to the date hereof, except such Returns which are not material to Parent. All such Returns are true, correct and complete in all material respects. Parent has paid all Taxes shown to be due on such Returns.
- (b) All Taxes that Parent is required by law to withhold or collect have been duly withheld or collected, and have been timely paid over to the proper governmental authorities to the extent due and payable.
- (c) Parent has not been delinquent in the payment of any Tax that has not been accrued for in Parent s books and records of account for the period for which such Tax relates nor is there any Tax deficiency outstanding, proposed or assessed against Parent, nor has Parent executed any unexpired waiver of any statute of limitations on or extending the period for the assessment or collection of any Tax.
- (d) No audit or other examination of any Return of Parent by any Tax authority is presently in progress, nor has Parent been notified of any request for such an audit or other examination.
- (e) No adjustment relating to any Returns filed by Parent has been proposed in writing, formally or informally, by any Tax authority to Parent or any representative thereof.
- (f) Parent has no liability for any unpaid Taxes which have not been accrued for or reserved on Parent s balance sheets included in the audited financial statements for the most recent fiscal year ended, whether asserted or unasserted, contingent or otherwise, other than any liability for unpaid Taxes that may have accrued since the end of the most recent fiscal year in connection with the operation of the business of Parent in the ordinary course of business, none of which is material to the business, results of operations or financial condition of Parent.
- (g) Parent and the Parent Subsidiaries have not taken any action and do not know of any fact, agreement, plan or other circumstances that is reasonably likely to prevent the Mergers from qualifying as a reorganization within the meaning

of Section 368(a) of the Code.

3.16 Environmental Matters. Except for such matters that, individually or in the aggregate, are not reasonably likely to have a Material Adverse Effect: (i) Parent has complied with all applicable Environmental Laws; (ii) Parent is not subject to liability for any Hazardous Substance disposal or contamination on any third party property; (iii) Parent has not been associated with any release or threat of release of any Hazardous Substance; (iv) Parent has not received any notice, demand, letter, claim or request for information alleging that Parent may be in violation of or liable under any Environmental Law; and (v) Parent is not subject to any

orders, decrees, injunctions or other arrangements with any Governmental Entity or subject to any indemnity or other agreement with any third party relating to liability under any Environmental Law or relating to Hazardous Substances.

- 3.17 *Brokers*. Except as set forth in *Schedule 3.17*, Parent has not incurred, nor will it incur, directly or indirectly, any liability for brokerage or finders fees or agent s commissions or any similar charges in connection with this Agreement or any transaction contemplated hereby.
- 3.18 *Intellectual Property*. Parent does not own, license or otherwise have any right, title or interest in any material Intellectual Property or Registered Intellectual Property except non-exclusive rights to the name Triplecrown.
  - 3.19 Agreements, Contracts and Commitments.
- (a) Except as set forth in the Parent SEC Reports filed prior to the date of this Agreement, other than confidentiality and non-disclosure agreements, there are no contracts, agreements, leases, mortgages, indentures, notes, bonds, liens, license, permit, franchise, purchase orders, sales orders or other understandings, commitments or obligations (including without limitation outstanding offers or proposals) of any kind, whether written or oral, to which Parent is a party or by or to which any of the properties or assets of Parent may be bound, subject or affected, which either (a) creates or imposes a liability greater than \$100,000, or (b) may not be cancelled by Parent on less than 30 days or less prior notice ( *Parent Contracts* ). All Parent Contracts are listed in *Schedule 3.19* other than those that are exhibits to the Parent SEC Reports.
- (b) Except as set forth in the Parent SEC Reports filed prior to the date of this Agreement, each Parent Contract was entered into at arms—length and in the ordinary course, is in full force and effect and is valid and binding upon and enforceable against each of the parties thereto. True, correct and complete copies of all Parent Contracts (or written summaries in the case of oral Parent Contracts) and of all outstanding offers or proposals of Parent have been heretofore made available to the Company.
- (c) Neither Parent nor, to the knowledge of Parent, any other party thereto is in breach of or in default under, and no event has occurred which with notice or lapse of time or both would become a breach of or default under, any Parent Contract, and no party to any Parent Contract has given any written notice of any claim of any such breach, default or event, which, individually or in the aggregate, are reasonably likely to have a Material Adverse Effect on Parent. Each agreement, contract or commitment to which Parent is a party or by which it is bound that has not expired by its terms is in full force and effect, except where such failure to be in full force and effect is not reasonably likely to have a Material Adverse Effect on Parent.
- 3.20 *Insurance*. Except for directors and officers liability insurance, Parent does not maintain any Insurance Policies.
- 3.21 *Interested Party Transactions*. Except as set forth in the Parent SEC Reports filed prior to the date of this Agreement: (a) no employee, officer, director or stockholder of Parent or a member of his or her immediate family is indebted to Parent nor is Parent indebted (or committed to make loans or extend or guarantee credit) to any of them, other than reimbursement for reasonable expenses incurred on behalf of Parent; (b) to Parent s knowledge, none of such individuals has any direct or indirect ownership interest in any Person with whom Parent is affiliated or with whom Parent has a material contractual relationship, or any Person that competes with Parent, except that each employee, stockholder, officer or director of Parent and members of their respective immediate families may own less than 5% of the outstanding stock in publicly traded companies that may compete with Parent; and (c) to Parent s knowledge, no officer, director or stockholder or any member of their immediate families is, directly or indirectly, interested in any material contract with Parent (other than such contracts as relate to any such individual ownership of capital stock or other securities of Parent).

- 3.22 Indebtedness. Parent has no indebtedness for borrowed money, except as set forth on Schedule 3.22.
- 3.23 NYSE Amex Listing. Parent Common Stock is currently listed on the NYSE Amex. Except as set forth on *Schedule 3.23*, there is no action or proceeding pending or, to Parent's knowledge, threatened against

Parent by the NYSE Amex or the Financial Industry Regulatory Authority, Inc. with respect to any intention by such entities to prohibit or terminate the listing of Parent Common Stock on the NYSE Amex.

- 3.24 Board Approval. The uninterested members of the Board of Directors of Parent and each Parent Subsidiary (including any required committee or subgroup of the Board of Directors of Parent and each Parent Subsidiary) has, as of the date of this Agreement, unanimously (i) declared the advisability of the Mergers and approved this Agreement and the transactions contemplated hereby, and (ii) determined that the Mergers are in the best interests of the stockholders of Parent and each Parent Subsidiary.
- 3.25 Trust Fund. As of the date hereof, Parent has no less than \$536,930,000 invested in United States Government securities in a trust account administered by Continental (the Trust Fund ). At Closing, Holdco and/or Parent will be required to pay monies from the Trust Fund (i) as deferred underwriters compensation in connection with Parent s initial public offering, (ii) for income Tax or other Tax obligations generated by the Trust Fund prior to Closing, (iii) reimbursement of expenses to directors, officers and founding stockholders of Parent, (iv) in connection with the actions, agreements and financings undertaken under Section 4.3, (v) to third parties (e.g., professionals, printers, etc.) who have rendered services to Parent in connection with its operations and efforts to effect a business combination, including the Mergers and (vi) for the cost and expenses incurred by the Stockholder, the Company or any Subsidiary, or on behalf of the Stockholder, the Company or any Subsidiary, in the connection with the preparation, negotiation, execution and performance of this Agreement and the transactions contemplated hereunder as set forth in Schedule 5.20. The balance of the funds will be released to Holdco to pay Parent stockholders who properly exercise their conversion rights and for working capital and general corporate purposes of Holdco and the Company. To the extent funds remain after satisfaction of the foregoing items (i) through (vi) and the payment to converting stockholders of Parent, such funds will be used to make the reimbursements as contemplated by Section 4.2(b).
- 3.26 Governmental Filings. Except as set forth in Schedule 3.26, Parent has been granted and holds, and has made, all Governmental Actions/Filings necessary to the conduct by Parent of its business (as presently conducted) or used or held for use by Parent, and true, complete and correct copies of which have heretofore been delivered to the Company. Each such Governmental Action/Filing is in full force and effect and, except as disclosed in Schedule 3.26, will not expire prior to December 31, 2009, and Parent is in material compliance with all of its obligations with respect thereto. No event has occurred and is continuing which requires or permits, or after notice or lapse of time or both would require or permit, and consummation of the transactions contemplated by this Agreement or any ancillary documents will not require or permit (with or without notice or lapse of time, or both), any modification or termination of any such Governmental Actions/Filings except such events which, either individually or in the aggregate, would not have a Material Adverse Effect upon Parent.
  - 3.27 Representations and Warranties Complete. The representations and warranties of Parent included in this Agreement and any list, statement, document or information set forth in, or attached to, any Schedule provided pursuant to this Agreement or delivered hereunder, are true and complete in all material respects and do not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements contained therein not misleading, under the circumstance under which they were made.
    - 3.28 *Survival of Representations and Warranties*. The representations and warranties of Parent set forth in this Agreement shall survive until the Closing.
- 3.29 Regulatory Agreements; Investment Company Act. There are no written agreements, memoranda of understanding, commitment letters, or cease and desist orders, to which Parent or any Parent Subsidiary is a party, on the one hand, and any Governmental Entity is a party or addressee, on the other hand. Parent is not an investment company or a Person directly or indirectly controlled by or acting on behalf of an investment company, in each case

within the meaning of the Investment Company Act of 1940, as amended.

#### ARTICLE IV

#### CONDUCT PRIOR TO THE EFFECTIVE TIME

- 4.1 Conduct of Business by the Company and Parent. Subject to the other Sections of this Article IV, during the period from the date of this Agreement and continuing until the earlier of the termination of this Agreement pursuant to its terms or the Closing, each of the Company, Parent and the Parent Subsidiaries shall, except to the extent that the other party shall otherwise consent in writing, carry on its business in the usual, regular and ordinary course consistent with past practices, in the case of Parent and the Parent Subsidiaries, or in a commercially reasonable manner, in the case of the Company, and in compliance with all applicable laws and regulations (except where noncompliance would not have a Material Adverse Effect), pay its debts and Taxes when due subject to good faith disputes over such debts or Taxes, pay or perform other material obligations when due, and use its commercially reasonable best efforts consistent with past practices and policies, in the case of Parent and the Parent Subsidiaries, or in a commercially reasonable manner, in the case of the Company, to (i) preserve substantially intact its present business organization, (ii) keep available the services of its present officers and employees and (iii) preserve its relationships with customers, suppliers, distributors, licensors, licensees, and others with which it has significant business dealings. In addition, except as required or permitted by the terms of this Agreement, without the prior written consent of the other party, during the period from the date of this Agreement and continuing until the earlier of the termination of this Agreement pursuant to its terms or the Closing, each of the Company, Parent and the Parent Subsidiaries shall not do any of the following (and the Stockholder shall cause the Company not to do any of the following):
- (a) Waive any stock repurchase rights, accelerate, amend or (except as specifically provided for herein) change the period of exercisability of options or restricted stock, or reprice options granted under any employee, consultant, director or other stock plans or authorize cash payments in exchange for any options granted under any of such plans;
- (b) Grant any severance or termination pay to any officer or employee except pursuant to applicable law, written agreements outstanding, or policies existing on the date hereof and as previously or concurrently disclosed in writing or made available to the other party, or adopt any new severance plan, or amend or modify or alter in any manner any severance plan, agreement or arrangement existing on the date hereof;
- (c) Except as contemplated by the IP Transfer Agreement and the transactions contemplated thereunder, transfer or license to any Person or otherwise extend, amend or modify any material rights to any Intellectual Property of the Company or Parent, as applicable, or enter into grants to transfer or license to any Person future patent rights, other than in a commercially reasonable manner provided that in no event shall the Company or Parent license on an exclusive basis or sell any Intellectual Property of the Company or Parent as applicable;
  - (d) Declare, set aside or pay any dividends on or make any other distributions (whether in cash, stock, equity securities or property) in respect of any capital stock or split, combine or reclassify any capital stock or issue or authorize the issuance of any other securities in respect of, in lieu of or in substitution for any capital stock;
- (e) Purchase, redeem or otherwise acquire, directly or indirectly, any shares of capital stock or other equity securities or ownership interests of such party including repurchases of unvested shares at cost in connection with the termination of the relationship with any employee or consultant pursuant to agreements in effect on the date hereof;
- (f) Except for the offering on commercially reasonable terms by the Company of securities, debt or other equity of the Company to raise capital in furtherance of its business plan as described in the business section of the Proxy Statement/Prospectus (the *Capital Raise*), issue, deliver, sell, authorize, pledge or otherwise encumber, or agree to any of the foregoing with respect to, any shares of capital stock or other equity securities or ownership interests or any

securities convertible into or exchangeable for shares of capital stock or other equity securities or ownership interests, or subscriptions, rights, warrants or options to acquire any shares of capital stock or other equity securities or ownership interests or any securities convertible into or exchangeable for shares of capital stock or other equity securities or other ownership interests, or enter into

other agreements or commitments of any character obligating it to issue any such shares, equity securities or other ownership interests or convertible or exchangeable securities;

#### (g) Amend its Charter Documents;

- (h) Except as contemplated by the Farm Purchase Agreement and the transactions contemplated thereunder, acquire or agree to acquire by merging or consolidating with, or by purchasing any equity interest in or a portion of the assets of, or by any other manner, any business or any corporation, partnership, association or other business organization or division thereof, or otherwise acquire or agree to acquire any assets which are material, individually or in the aggregate, to the business of Parent or the Company as applicable, or enter into any joint ventures, strategic partnerships or alliances or other arrangements that provide for exclusivity of territory or otherwise restrict such party s ability to compete or to offer or sell any products or services. For purposes of this paragraph, material includes the requirement that, as a result of such transaction, financial statements of the acquired, merged or consolidated entity be included in the business section of the Proxy Statement/Prospectus);
  - (i) Except as contemplated by the IP Transfer Agreement and the transactions contemplated thereunder, sell, lease, license, encumber or otherwise dispose of any properties or assets, except (A) sales of inventory in a commercially reasonable manner, and (B) the sale, lease or disposition (other than through licensing) of property or assets that are not material, individually or in the aggregate, to the business of such party;
- (j) Except as contemplated by the Capital Raise, incur any indebtedness for borrowed money in excess of \$200,000 in the aggregate or guarantee any such indebtedness of another Person or Persons in excess of \$200,000 in the aggregate, issue or sell any debt securities or options, warrants, calls or other rights to acquire any debt securities of such party, enter into any keep well or other agreement to maintain any financial statement condition or enter into any arrangement having the economic effect of any of the foregoing;
- (k) Adopt or amend any employee benefit plan, policy or arrangement, any employee stock purchase or employee stock option plan, or enter into any employment contract or collective bargaining agreement (other than offer letters and letter agreements entered into in a commercially reasonable manner with employees who are terminable at will), pay any special bonus or special remuneration to any director or employee, or increase the salaries or wage rates or fringe benefits (including rights to severance or indemnification) of its directors, officers, employees or consultants, except in a commercially reasonable manner;
- (l) Except as contemplated by the Farm Purchase Agreement and the transactions contemplated thereunder, pay, discharge, settle or satisfy any claims, liabilities or obligations (absolute, accrued, asserted or unasserted, contingent or otherwise), or litigation (whether or not commenced prior to the date of this Agreement) other than the payment, discharge, settlement or satisfaction, in a commercially reasonable manner or in accordance with their terms, or liabilities recognized or disclosed in the Audited Financial Statements or in the most recent financial statements included in the Parent SEC Reports filed prior to the date of this Agreement, as applicable, or incurred since the date of such financial statements, or waive the benefits of, agree to modify in any manner, terminate, release any Person from or knowingly fail to enforce any confidentiality or similar agreement to which the Company is a party or of which the Company is a beneficiary, as applicable;
- (m) Except in a commercially reasonable manner, modify, amend or terminate any Material Company Contract or Parent Contract, as applicable, or waive, delay the exercise of, release or assign any material rights or claims thereunder;

- (n) Except as required by U.S. GAAP, revalue any of its assets or make any change in accounting methods, principles or practices;
- (o) Except as contemplated by the Farm Purchase Agreement and the transactions contemplated thereunder and except in a commercially reasonable manner, incur or enter into any agreement, contract or commitment requiring such party to pay in excess of \$200,000 in any 12 month period;
  - (p) Engage in any action that could reasonably be expected to cause the Mergers to fail to qualify as a Tax-free transactions pursuant to Section 351, Section 368(a)(1)(A) and Section 368(a)(2)(E) of the Code;

- (q) Settle any litigation where the consideration given is other than monetary or to which an Insider, or any officer, director, shareholder or holder of derivative securities of Parent, as applicable, is a party;
- (r) Make or rescind any Tax elections that, individually or in the aggregate, could be reasonably likely to adversely affect in any material respect the Tax liability or Tax attributes of such party, settle or compromise any material income Tax liability or, except as required by applicable law, materially change any method of accounting for Tax purposes or prepare or file any Return in a manner inconsistent with past practice;
  - (s) Form, establish or acquire any subsidiary except as contemplated by this Agreement and/or as specifically described in the business section of the Proxy Statement/Prospectus;
- (t) Permit any Person to exercise any of its discretionary rights under any Plan to provide for the automatic acceleration of any outstanding options, the termination of any outstanding repurchase rights or the termination of any cancellation rights issued pursuant to such plans;
  - (u) Except as contemplated by the Farm Purchase Agreement and the transactions contemplated thereunder, make capital expenditures except in accordance with prudent business and operational practices;
    - (v) Make or omit to take any action which would be reasonably anticipated to have a Material Adverse Effect;
  - (w) Enter into any transaction with or distribute or advance any assets or property to any of its officers, directors, partners, stockholders, managers, members or other Affiliates other than the payment of salary and benefits in a commercially reasonable manner; or
  - (x) Agree in writing or otherwise agree, commit or resolve to take any of the actions described in Section 4.1 (a) through (w) above.

In addition, the Stockholder shall not do any of the foregoing with respect to itself to the extent it prevents the consummation of the transactions contemplated hereby. Notwithstanding the foregoing, no Capital Raise shall be consummated during the period commencing on the date of effectiveness of the Proxy Statement/Prospectus and ending on the earlier of the date of consummation of the Mergers or the termination of this Agreement.

### 4.2 Exercise of Certain Contracts.

- (a) On or prior to September 15, 2009, the Parent, the Stockholder and the Company shall use commercially reasonable efforts to obtain an extension of the closing date of the agreement by and among the Parent, the Company's Subsidiary, Natural Diary, Inc., and Grimsley LLC, dated as of June 27, 2009 (the *Farm Purchase Agreement*), to purchase certain land and other assets (collectively, the *Grimsley Farm*), to October 22, 2009.
- (b) If the parties are unable to obtain such an extension, the Company and/or the Stockholder shall cause the Company s Subsidiary, Natural Dairy, Inc. (or a subsidiary thereof), to exercise the right to purchase the Grimsley Farm on or prior to September 15, 2009 and shall supply all amounts then due under the Farm Purchase Agreement ( Purchase Cost ). The Company shall cause Natural Dairy, Inc. (or a subsidiary thereof) to hold and maintain the Grimsley Farm on behalf of the Surviving Company pending the Closing of the Company Merger. At the time of the consummation of the Company Merger, the party or parties that funded the amount of the Purchase Cost shall have such amount repaid to it from amounts in the Trust Fund, if any, or if no such amounts are available at Closing after giving effect to all other items required to be paid as described in Section 5.20, such amounts shall be repaid as soon as practicable by Holdco (and in no event later than 90 days after the issuance of such note) and Holdco shall issue to

such parties a promissory note evidencing such obligation and paying simple interest at 8% per annum.

(c) If the Mergers are not consummated for any reason, the Company shall repay to Parent all amounts paid by Parent to Grimsley LLC pursuant to the Farm Purchase Agreement on or prior to the date of final distribution of the Trust Fund.

- (d) Except as set forth herein, from and after the date hereof, the Farm Purchase Agreement shall not be amended or modified without the prior written consent of each of the Company and Parent. Upon consummation of the transactions contemplated by the Farm Purchase Agreement, Parent shall have all inspection and information rights reasonably necessary to ensure the Grimsley Farm is properly maintained and operated prior to the consummation of the Company Merger.
- 4.3 Purchases of Parent Common Stock and Warrants. Notwithstanding anything to the contrary contained in this Agreement, from and after the date of this Agreement until the Closing, for purposes of achieving approval of the Mergers and the related transactions by the stockholders of Parent, Parent and its Affiliates may, and shall be permitted to, seek, negotiate and enter into arrangements for the purchase, redemption, tender or assignment of currently outstanding shares of Parent Common Stock and/or Parent Warrants, declare lawful dividends to be effective immediately following the Effective Time or enter into similar or related arrangements and engage in related undertakings that would be consummated at, prior to or immediately following Closing and which may utilize a portion of the funds held in the Trust Fund and/or released from the Trust Fund upon consummation of the transactions contemplated hereunder. The parties to this Agreement and/or their Affiliates may secure a bridge loan or loans or similar credit facilities to finance the foregoing activities, in which case, the obligations to the lenders under such facilities shall be repaid at Closing from amounts in the Trust Fund or from amounts held by Holdco.

#### 4.4 Exclusivity.

- (a) From the date hereof until the earlier of (x) the termination of this Agreement pursuant to Section 8.1 or (y) October 22, 2009, the Company and the Stockholder shall not, and the Company and the Stockholder shall each use commercially reasonable best efforts to cause each of its officers, directors, employees, stockholders, representatives and agents not to, directly or indirectly, (i) encourage, solicit, initiate, engage or participate in negotiations with any Person (other than the Parent) concerning any Acquisition Transaction or (ii) take any other action intended or designed to facilitate the efforts of any Person (other than Parent) relating to a possible Acquisition Transaction. For purposes of this Agreement, the term *Acquisition Transaction* shall mean any of the following involving the Company or any Subsidiary: (i) any merger, consolidation, share exchange, business combination or other similar transaction (excluding the transactions contemplated by this Agreement); or (ii) any sale, lease, exchange, transfer or other disposition of any of the assets of the Company or Subsidiaries (other than in a commercially reasonable manner) or any shares of the capital stock of the Company or any Subsidiary in a single transaction or series of transactions (excluding the Capital Raise and the transactions contemplated by this Agreement).
  - (b) In the event that there is an unsolicited proposal for or an unsolicited indication of a serious interest in entering into, an Acquisition Transaction, communicated to the Company or the Stockholder (or any of the Stockholder s officers, directors, employees or stockholders) or any of their respective representatives or agents, such party shall promptly give written notice of same to the Parent.

#### ARTICLE V

#### ADDITIONAL AGREEMENTS

5.1 Proxy Statement/Prospectus; Special Meeting.

(a) As soon as is reasonably practicable after receipt by Parent from the Company of all financial and other information relating to the Company as Parent may reasonably request for its preparation, Parent shall prepare and file with the SEC under the Securities Act, and with all other applicable regulatory bodies, a registration statement on Form S-4 with respect to the shares of Holdco Common Stock into which Parent soutstanding common stock will be

converted in connection with the Mergers (the \*Registration Statement\*), which shall include proxy materials for the purpose of soliciting proxies from holders of Parent Common Stock to vote, at a meeting of the holders of Parent Common Stock to be called for such purpose (the \*Special Meeting\*), in favor of (i) the amendments to Parent s amended and restated certificate of incorporation necessary to allow Parent to complete the Mergers as contemplated hereby and in compliance with the DGCL and GCL, (ii) the adoption of this Agreement and the approval of the Mergers (\*Parent Stockholder Approval\*), (iii) increasing the authorized number of shares of capital stock of Surviving Pubco to 200,000,000 shares of common stock and 1,000,000 shares of preferred stock, (iv) the effective modifications

to the public corporation s certificate of incorporation resulting from the adoption of Holdco s certificate of incorporation as the certificate of incorporation of the Surviving Pubco, and (v) an adjournment proposal, if necessary, to adjourn the Special Meeting if, based on the tabulated vote count, Parent and Holdco are not authorized to proceed with the Mergers. The Registration Statement shall also include such information as necessary for its use as contemplated by Section 5.1(c) below. Such proxy materials shall be in the form of a proxy statement/prospectus to be used for the purpose of soliciting proxies from holders of Parent Common Stock for the matters to be acted upon at the Special Meeting and also for the purpose of issuing Holdco Common Stock as contemplated hereby and permitted by Law (the *Proxy Statement/Prospectus*). The Company shall furnish to Parent all information concerning the Company (and its Subsidiaries and the Stockholder) as Parent may reasonably request in connection with the preparation of the Registration Statement. The Company and its counsel shall be given an opportunity to review and comment on the Registration Statement prior to its filing with the SEC. Parent, with the assistance of the Company and its counsel, shall promptly respond to any SEC comments on the Registration Statement and shall otherwise use commercially reasonable best efforts to cause the Registration Statement to be declared effective by the SEC as promptly as practicable. Parent shall also take any and all actions required to satisfy the requirements of the Securities Act and the Exchange Act.

- (b) As soon as practicable following the declaration of effectiveness of the Registration Statement by the SEC, Parent shall distribute the Proxy Statement/Prospectus to the holders of Parent Common Stock and, pursuant thereto, shall call the Special Meeting in accordance with the DGCL and, subject to the other provisions of this Agreement, solicit proxies from such holders to vote in favor of the adoption of this Agreement and the approval of the Merger and the other matters presented to the stockholders of Parent for approval or adoption at the Special Meeting, including, without limitation, the matters described in Section 5.1(a).
- (c) As soon as practicable following the declaration of effectiveness of the Registration Statement by the SEC, Parent shall distribute the Proxy Statement/Prospectus to the holders of Parent Warrants and, pursuant thereto, shall call a special meeting of the holders of the Parent Warrants in accordance with the DGCL ( *Warrantholders Meeting* ) and, subject to the other provisions of this Agreement, solicit proxies from such holders of the Parent Warrants to vote in favor of the adoption ( *Warrantholders Approval* ) of the Warrant Amendment Agreement between Holdco, as the Surviving Pubco, Parent and Continental Stock Transfer & Trust Company in the form of *Exhibit I* hereto ( *Warrant Amendment Agreement* ), which, if adopted, shall increase the exercise price of the Holdco Warrants to \$12.00 per share, extend the duration of each Holdco Warrant by one additional year and increase the redemption price of the Holdco Warrants to \$17.00.
- (d) Parent shall comply with all applicable provisions of and rules under the Securities Act, the Exchange Act and all applicable provisions of the DGCL in the preparation, filing and distribution of the Registration Statement and the Proxy Statement/Prospectus, the solicitation of proxies thereunder, and the calling and holding of the Special Meeting and Warrantholders Meeting. Without limiting the foregoing, Parent shall ensure that the Proxy Statement/Prospectus does not, as of the date on which the Registration Statement is declared effective, and as of the date of the Special Meeting and Warrantholders Meeting, contain any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements made, in light of the circumstances under which they were made, not misleading (provided that Parent shall not be responsible for the accuracy or completeness of any information relating to the Company or any other information furnished by the Company for inclusion in the Proxy Statement/Prospectus). Each of the Company and the Stockholder represents and warrants that the information relating to the Company and the Stockholder supplied by the Company or the Stockholder for inclusion in the Proxy Statement/Prospectus will not as of the date on which the Proxy Statement/Prospectus (or any amendment or supplement thereto) is first distributed to stockholders of Parent or at the time of the Special Meeting or Warrantholders Meeting contain any statement which, at such time and in light of the circumstances under which it is made, is false or misleading with respect to any material fact, or omits to state any material fact required to be stated therein or necessary in order to make the

statement therein not false or misleading.

(e) Parent, acting through its board of directors, shall include in the Proxy Statement/Prospectus the recommendation of its board of directors that the holders of Parent Common Stock vote in favor of the adoption of this Agreement and the approval of the Mergers and that the holders of the Parent Warrants vote in

favor of the adoption of the Warrant Amendment Agreement and shall otherwise use commercially reasonable best efforts to obtain the Parent Stockholder Approval and Warrantholders Approval.

- 5.2 Directors and Officers of Surviving Pubco and Surviving Subsidiary After Merger. The Parties shall take all necessary action so that the persons listed in Schedule 5.2 are elected to the positions of officers and directors of Surviving Pubco and Surviving Subsidiary, as set forth therein, to serve in such positions effective immediately after the Closing. If any Person listed in Schedule 5.2 is unable to serve, the party appointing such Person shall designate a successor; provided that, if such designation is to be made after the Closing, any successor to a Person designated by Parent shall be made by the person serving as Chief Executive Officer of Parent immediately prior to Closing.
- 5.3 HSR Act. If required pursuant to the HSR Act, as promptly as practicable after the date of this Agreement, Parent and the Company shall each prepare and file the notification required of it thereunder in connection with the transactions contemplated by this Agreement and shall promptly and in good faith respond to all information requested of it by the Federal Trade Commission and Department of Justice in connection with such notification and otherwise cooperate in good faith with each other and such Governmental Entities. Parent and the Company shall (a) promptly inform the other of any communication to or from the Federal Trade Commission, the Department of Justice or any other Governmental Entity regarding the transactions contemplated by this Agreement, (b) give the other prompt notice of the commencement of any action, suit, litigation, arbitration, proceeding or investigation by or before any Governmental Entity with respect to such transactions and (c) keep the other reasonably informed as to the status of any such action, suit, litigation, arbitration, proceeding or investigation. Filing fees with respect to the notifications required under the HSR Act shall be paid by the Company.

#### 5.4 Other Actions.

- (a) As promptly as practicable after execution of this Agreement, Parent will prepare and file a Current Report on Form 8-K pursuant to the Exchange Act to report the execution of this Agreement ( Signing Form 8-K ), which the Company shall have the right to review and comment upon prior to filing. Any language included in the Signing Form 8-K that reflects the Company s comments, as well as any text as to which the Company has not commented upon being given a reasonable opportunity to comment, shall, notwithstanding the provisions of Sections 5.6 and 5.7, be deemed to have been approved by the Company and the Stockholder and may henceforth be used by Parent in other filings made by it with the SEC and in other documents distributed by Parent in connection with the transactions contemplated by this Agreement without further review or consent of the Stockholder or the Company. Promptly after the execution of this Agreement, Parent and the Company shall also issue a mutually agreed upon press release pursuant to Section 5.7 announcing the execution of this Agreement (the Signing Press Release).
- (b) At least five (5) days prior to Closing, Parent shall prepare a draft Form 8-K announcing the Closing, together with, or incorporating by reference, the financial statements prepared by the Company and its accountant, and such other information that may be required to be disclosed with respect to the Mergers in any report or form to be filed with the SEC ( *Closing Form 8-K* ), which shall be in a form reasonably acceptable to the Company. Prior to Closing, Parent and the Company shall prepare a press release announcing the consummation of the Mergers hereunder ( *Closing Press Release* ). Concurrently with the Closing, Surviving Pubco shall distribute the Closing Form 8-K with the Commission.
- 5.5 Required Information. In connection with the preparation of the Signing Form 8-K, the Signing Press Release, the Registration Statement, the Closing Form 8-K and the Closing Press Release, or any other statement, filing notice or application made by or on behalf of Parent, Surviving Pubco and/or the Company to any Government Entity or other third party in connection with Mergers and the other transactions contemplated hereby, and for such other reasonable

purposes, the Company and the Stockholder, on the one hand, and Parent, on the other hand, shall, upon request by the other, furnish the other with all information concerning themselves, their respective directors, officers, managers, members and stockholders (including the directors and officers of Surviving Pubco and Surviving Subsidiary to be appointed or elected effective as of the Closing pursuant to Section 5.2 hereof) and such other matters as may be reasonably necessary or advisable in connection with the Mergers. Each party warrants and represents to the other party that all such

information shall be true and correct in all material respects and will not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements contained therein, in light of the circumstances under which they were made, not misleading.

5.6 Confidentiality; Access to Information.

(a) *Confidentiality*. Any confidentiality agreement previously executed by the parties shall be superseded in its entirety by the provisions of this Agreement. Each party agrees to maintain in confidence any non-public information received from the other party, and to use such non-public information only for purposes of consummating the transactions contemplated by this Agreement. Such confidentiality obligations will not apply to (i) information which was known to the one party or their respective agents prior to receipt from the other party; (ii) information which is or becomes generally known to the public through no fault of the party receiving the information; (iii) information acquired by a party or their respective agents from a third party who was not bound to an obligation of confidentiality; and (iv) disclosure required by law or at the order of any Governmental Entity having jurisdiction over such party (including, without limitation, any rules or regulations of the SEC or the Financial Industry Regulatory Authority). In the event this Agreement is terminated as provided in Article VIII hereof, each party (i) will destroy or return or cause to be destroyed or returned to the other all documents and other material obtained from the other in connection with the Mergers contemplated hereby, and (ii) will use its commercially reasonable best efforts to delete from its computer systems all documents and other material obtained from the other in connection with the Mergers contemplated hereby.

#### (b) Access to Information.

- (i) The Company and the Stockholder will afford Parent and its financial advisors, accountants, counsel and other representatives reasonable access during normal business hours, upon reasonable notice, to the properties, books, records and personnel of the Company during the period prior to the Closing to obtain all information concerning the business, including proposed business and operations and the status of business development efforts, properties, results of operations and personnel of the Company, as Parent may reasonably request. No information or knowledge obtained by Parent in any investigation pursuant to this Section 5.6(b) will affect or be deemed to modify any representation or warranty contained herein or the conditions to the obligations of the parties to consummate the Mergers.
- (ii) Parent will afford the Company, the Stockholder and their financial advisors, underwriters, accountants, counsel and other representatives reasonable access during normal business hours, upon reasonable notice, to the properties, books, records and personnel of Parent during the period prior to the Closing to obtain all information concerning the business, including properties, results of operations and personnel of Parent, as the Company or the Stockholder may reasonably request. No information or knowledge obtained by the Company in any investigation pursuant to this Section 5.6(b) will affect or be deemed to modify any representation or warranty contained herein or the conditions to the obligations of the parties to consummate the Mergers.
- 5.7 *Public Disclosure*. From the date of this Agreement until Closing or termination, the parties shall cooperate in good faith to jointly prepare all press releases and public announcements pertaining to this Agreement and the transactions governed by it, and no party shall issue or otherwise make any public announcement or communication pertaining to this Agreement or the transaction without the prior consent of Parent (in the case of the Company and the Stockholder) or the Company or the Stockholder (in the case of Parent), except as required by any legal requirement or by the rules and regulations of, or pursuant to any agreement of a stock exchange or trading system. Each party will not unreasonably withhold approval from the others with respect to any press release or public announcement. If any party determines with the advice of counsel that it is required to make this Agreement and the terms of the transaction

public or otherwise issue a press release or make public disclosure with respect thereto, it shall, at a reasonable time before making any public disclosure, consult with the other party regarding such disclosure, seek such confidential treatment for such terms or portions of this Agreement or the transaction as may be reasonably requested by the other party and disclose only such information as is legally compelled to be disclosed. This provision will not apply to communications by any party to its counsel, accountants and other professional advisors.

5.8 Commercially Reasonable Best Efforts. Upon the terms and subject to the conditions set forth in this Agreement, each of the parties agrees to use its commercially reasonable best efforts to take, or cause to be taken, all actions, and to do, or cause to be done, and to assist and cooperate with the other parties in doing, all things necessary, proper or advisable to consummate and make effective, in the most expeditious manner practicable, the Mergers and the other transactions contemplated by this Agreement, including using commercially reasonable best efforts to accomplish the following: (i) the taking of all reasonable acts necessary to cause the conditions precedent set forth in Article VI to be satisfied, (ii) the obtaining of all necessary actions, waivers, consents, approvals, orders and authorizations from Governmental Entities and third parties and the making of all necessary registrations, declarations and filings (including registrations, declarations and filings with Governmental Entities, if any) and the taking of all reasonable steps as may be necessary to avoid any suit, claim, action, investigation or proceeding by any Governmental Entity, (iii) the obtaining of all consents, approvals or waivers from third parties required as a result of the transactions contemplated in this Agreement, including the consents referred to in Schedule 2.5 of the Company Schedule, (iv) the defending of any suits, claims, actions, investigations or proceedings, whether judicial or administrative, challenging this Agreement or the consummation of the transactions contemplated hereby, including seeking to have any stay or temporary restraining order entered by any court or other Governmental Entity vacated or reversed and (v) the execution or delivery of any additional instruments reasonably necessary to consummate the transactions contemplated by, and to fully carry out the purposes of, this Agreement. In connection with and without limiting the foregoing, Parent and its board of directors and the Company and its board of directors and the Stockholder shall, if any state takeover statute or similar statute or regulation is or becomes applicable to the Mergers, this Agreement or any of the transactions contemplated by this Agreement, use its commercially reasonable best efforts to enable the Mergers and the other transactions contemplated by this Agreement to be consummated as promptly as practicable on the terms contemplated by this Agreement. Notwithstanding anything herein to the contrary, nothing in this Agreement shall be deemed to require Parent, the Company or the Stockholder to agree to any divestiture by itself or any of its Affiliates of shares of capital stock or of any business, assets or property, or the imposition of any material limitation on the ability of any of them to conduct their business or to own or exercise control of such assets, properties and stock.

5.9 *Tax-Free Transactions*. Neither Parent nor the Company nor the Members shall take any action prior to or following the Mergers that could reasonably be expected to cause the Mergers to fail to qualify as Tax-free transactions pursuant to Section 351, Section 368(a)(1)(A) and Section 368(a)(2)(E) of the Code and the Regulations thereunder.

5.10 *Certain Claims*. As additional consideration for the issuance of Holdco Common Stock pursuant to this Agreement, the Stockholder hereby releases and forever discharges, effective as of the Closing Date, the Company and its directors, officers, managers, employees and agents, from any and all rights, claims, demands, judgments, obligations, liabilities and damages, whether accrued or unaccrued, asserted or unasserted, and whether known or unknown arising out of or resulting from the Stockholder s (i) status as a holder of an equity interest in the Company; and (ii) service, management, consulting or other similar agreement entered into with the Company prior to Closing to the extent that the basis for claims under any such agreement that survives the Closing arise prior to the Closing, provided, however, the foregoing shall not release any obligations of Surviving Pubco set forth in this Agreement or any of the other documents executed in connection with the transactions contemplated hereby.

5.11 *No Securities Transactions*. Neither the Company or the Stockholder nor any of their Affiliates, directly or indirectly, shall engage in any transactions involving the securities of Parent prior to the time of the making of a public announcement of the transactions contemplated by this Agreement. Each of the Company and the Stockholder shall use its commercially reasonable best efforts to require each of its officers, directors, stockholder and employees, and shall use commercially reasonable best efforts to require each of its agents, advisors, contractors, associates, clients, customers and representatives, to comply with the foregoing requirement.

5.12 *No Claim Against Trust Fund*. Notwithstanding anything else in this Agreement, the Company and the Stockholder acknowledge that they have read Parent s final prospectus dated October 22, 2007 and understand that Parent has established the Trust Fund for the benefit of Parent s public stockholders and that Parent may disburse monies from the Trust Fund only (a) to Parent in limited amounts for its working capital

requirements and Tax obligations, (b) to Parent s public stockholders in the event they elect to convert their shares into cash in accordance with Parent s Charter Documents and/or the liquidation of Parent or (c) to Parent concurrently with, or after it consummates, a business combination. The Company and the Stockholder further acknowledge that, if the transactions contemplated by this Agreement, or, upon termination of this Agreement, another business combination, are not consummated by October 22, 2009, Parent will be obligated to return to its stockholders the amounts being held in the Trust Fund. Accordingly, the Company and the Stockholder, for themselves and their subsidiaries, affiliated entities, directors, officers, managers, members, employees, stockholders, representatives, advisors and all other associates and Affiliates, hereby waive all rights, title, interest or claim of any kind against Parent to collect from the Trust Fund any monies that may be owed to them by Parent for any reason whatsoever, including but not limited to a breach of this Agreement by Parent or any negotiations, agreements or understandings with Parent (whether in the past, present or future), and will not seek recourse against the Trust Fund at any time for any reason whatsoever. This section will survive this Agreement and will not expire and will not be altered in any way without the express written consent of Parent, the Company and the Stockholder.

5.13 Disclosure of Certain Matters. Each of Parent, on the one hand, and the Company and the Stockholder, on the other hand, will provide the other with prompt written notice of any event, development or condition that (a) would cause any of such party s representations and warranties to become untrue or misleading or which may affect its ability to consummate the transactions contemplated by this Agreement, (b) had it existed or been known on the date hereof would have been required to be disclosed under this Agreement, (c) gives such party any reason to believe that any of the conditions set forth in Article VI will not be satisfied, (d) is of a nature that is or may be materially adverse to the operations, prospects or condition (financial or otherwise) of the Company or Parent, as applicable, or (e) would require any amendment or supplement to the Proxy Statement/Prospectus. Notwithstanding any such disclosure after the date hereof, the representations and warranties of the parties shall be as made on the date of execution of this Agreement and with reference to the Disclosure Schedule as they exist at the time of execution of this Agreement, subject to such anticipated changes as expressly contemplated by this Agreement as of the date of execution.

5.14 Securities Listing. Parent and the Company shall use their commercially reasonable best efforts to obtain the listing for trading on the New York Stock Exchange or, if such listing is not obtainable, on the Nasdaq, of the Holdco Common Stock and the Holdco Warrants. If any such listing is not obtained by the Closing, the parties shall continue to use their commercially reasonable best efforts after the Closing to obtain such listing.

#### 5.15 Further Actions.

- (a) The Company shall use its commercially reasonable best efforts to take such actions as are necessary to fulfill its obligations under this Agreement.
  - (b) Parent shall use its commercially reasonable best efforts to take such actions as are necessary to fulfill its obligations under this Agreement.
    - 5.16 Charter Protections; Directors and Officers Liability Insurance.
- (a) All rights to indemnification for acts or omissions occurring through the Closing Date now existing in favor of the current directors and officers of Parent as provided in the Charter Documents of Parent or in any indemnification agreements shall survive the Mergers and shall continue in full force and effect in accordance with their terms.
  - (b) For a period of six (6) years after the Closing Date, Holdco shall cause to be maintained in effect the current policies of directors' and officers' liability insurance maintained by Parent (or policies of at least the same coverage and amounts containing terms and conditions which are no less advantageous), with respect to claims arising from

facts and events that occurred prior to the Closing Date.

(c) If Parent or any of its successors or assigns (i) consolidates with or merges into any other Person and shall not be the continuing or surviving entity of such consolidation or merger, or (ii) transfers or conveys all

or substantially all of its properties and assets to any Person, then, in each such case, to the extent necessary, proper provision shall be made so that the successors and assigns of Parent assume the obligations set forth in this Section 5.16.

- (d) The provisions of this Section 5.16 are intended to be for the benefit of, and shall be enforceable by, each Person who will have been a director or officer of Parent for all periods ending on or before the Closing Date and may not be changed without the consent of Committee referred to in Section 1.16.
- 5.17 *Insider Loans; Equity Ownership in Subsidiaries*. The Stockholder and each officer, director, employee and consultant of the Company and each similar person, at or prior to Closing, shall (i) repay to the Company any loan by the Company to such person and any other amount owed by such person to the Company; (ii) cause any guaranty or similar arrangement pursuant to which the Company has guaranteed the payment or performance of any obligations of such person to a third party to be terminated; and (iii) cease to own any direct equity interests in any Subsidiary of the Company or in any other Person that utilizes the name Natural Dairy or Cullen Agricultural or derivatives thereof.
- 5.18 Certain Financial Information. Within fifteen (15) business days after the end of each calendar month between the date hereof and the earlier of the Closing Date and the date on which this Agreement is terminated, the Company shall deliver to Parent unaudited consolidated financial statements of the Company for such month, including a balance sheet, statement of operations, statement of cash flows and statement of stockholders—equity, that are certified as correct and complete by a suitably authorized officer of the Company, prepared in accordance with the U.S. GAAP applied on a consistent basis to prior periods (except as may be indicated in the notes thereto) and fairly present in all material respects the financial position of the Company at the date thereof and the results of its operations and cash flows for the period indicated, except that such statements need not contain notes and may be subject to normal adjustments that are not expected to have a Material Adverse Effect on the Company.
- 5.19 Access to Financial Information. The Company will, and will cause its auditors to (a) continue to provide Parent and its advisors full access to all of the Company s financial information used in the preparation of its Audited Financial Statements and the financial information furnished pursuant to Section 5.18 hereof and (b) cooperate fully with any reviews performed by Parent or its advisors of any such financial statements or information.
- 5.20 *Trust Fund Disbursement*. Parent shall cause the Trust Fund to be dispersed at Closing to pay the amounts payable (i) as deferred underwriters—compensation in connection with Parent—s initial public offering, (ii) for income Tax or other Tax obligations generated by the Trust Fund prior to Closing, (iii) reimbursement of expenses to directors, officers and founding stockholders of Parent, (iv) in connection with the actions, agreements and financings undertaken under Section 4.3, (v) to third parties (e.g., professionals, printers, etc.) who have rendered services to Parent in connection with its operations and efforts to effect a business combination, including the Mergers and (vi) for the cost and expenses incurred by the Stockholder, the Company or any Subsidiary, or on behalf of the Stockholder, the Company or any Subsidiary, in the connection with the preparation, negotiation, execution and performance of this Agreement and the transactions contemplated hereunder as set forth in *Schedule 5.20*. The balance of the funds will be released to Holdco to pay Parent stockholders who properly exercise their conversion rights and for working capital and general corporate purposes of Holdco and the Company. To the extent funds remain after satisfaction of the foregoing items (i) through (vi) and the payment to converting stockholders of Parent, such funds will be used to make the reimbursements as contemplated by Section 4.2(b).
- 5.21 Section 16 Matters. Prior to the Effective Time, Parent and the Company shall take all reasonable steps as may be required or permitted to cause any dispositions of the Parent Common Stock and Parent Warrants that occur or are deemed to occur by reason of or pursuant to the transactions contemplated by this Agreement by each individual who is or will be subject to the reporting requirements of Section 16(a) of the Exchange Act with respect to Parent to be

exempt under Rule 16b-3 promulgated under the Exchange Act, including by taking steps in accordance with the No-Action Letter, dated January 12, 1999, issued by the SEC regarding such matters.

### **TABLE OF CONTENTS**

5.22 Surrender of Founders Shares. Concurrently with Closing, the 4,630,000 shares of Holdco Common Stock deemed issued upon conversion of the Parent Common Stock beneficially held by Jonathan Ledecky and the 6,630,000 shares of Holdco Common Stock deemed issued upon conversion of the Parent Common Stock beneficially held by Eric Watson, shall be transferred to Holdco and immediately and automatically cancelled by Holdco.

#### 5.23 [Reserved.]

5.24 Consulting Agreement Amendments. The consulting agreements between Parent and each of Paul Vassilakos (or his entity as applicable) and Miles Leahy shall be amended at Closing in the form of Exhibit J and Exhibit K hereto (together the Consulting Agreement Amendments).

#### 5.25 Noncompete.

- (a) For a period of three years from the date of the Closing, neither the Stockholder nor any director or officer or stockholder of the Stockholder shall, directly or indirectly, individually or as an employee, partner, officer, director or shareholder or in any other capacity whatsoever of or for any Person other than Surviving Pubco, Surviving Subsidiary or their respective Subsidiaries or Affiliates (and other as passive investors of less than 5% of the voting stock of a company listed on a national exchange):
- (i) Own, manage, operate, sell, control or participate in the ownership, management, operation, sales or control of or be connected in any manner, including as an employee, advisor or consultant or similar role, with any business engaged, in the geographical areas referred to in Section 5.25(b) below, in the design, research, development, marketing, sale, branding or licensing of products or services that are substantially similar to or competitive with the business of Surviving Subsidiary, including such operations as contemplated to be undertaken by the Proxy Statement/Prospectus; or
- (ii) Recruit, attempt to hire, solicit, or assist others in recruiting or hiring, in or with respect to the geographical areas referred to in Section 5.25(b) below, any person who is an employee of Surviving Pubco or Surviving Subsidiary or any of their respective Subsidiaries or Affiliates or induce or attempt to induce any such employee to terminate his employment with Surviving Pubco, Surviving Subsidiary or any of their respective Subsidiaries or Affiliates; provided, however, a general solicitation of employment placed by the Stockholder or any of its directors, officers or stockholders to which employees of Surviving Pubco or Surviving Subsidiary or any of their respective subsidiaries or Affiliates respond shall not be deemed a breach of this sub Section.
- (b) The geographical areas in which the restrictions provided for in this Section apply include all cities, counties and states of the United States, and all other countries in which Surviving Pubco, Surviving Subsidiary (or any of their respective Subsidiaries or Affiliates) are conducting business at the time in question, whether or not any of Surviving Pubco or Surviving Subsidiary (or such Subsidiary or Affiliate) has an actual physical presence in such location. The Stockholder acknowledges that the obligations and restrictions provided for herein are an integral part of the consideration motivating Parent to enter into this Agreement.
- (c) In addition to, and not in limitation of, the non-competition covenants set forth above in this Section, the Stockholder agrees that it will not, either for itself or for any other Person, directly or indirectly (other than for Holdco, Parent, the Company and any of their respective Subsidiaries or Affiliates and other as a passive investor of less than 5% of the voting stock of a company listed on a national exchange), solicit business away from, or attempt to sell, license or provide the same or similar products or services as are then provided by Surviving Pubco or Surviving Subsidiary or any Subsidiary or Affiliate thereof to any customer thereof.

- (d) Within ten business days of the date of this Agreement, the Company and the Stockholder shall cause the persons listed on *Schedule 5.25* to execute a letter ( *Cullen Noncompete Letters* ) agreeing to be bound by the provisions of this Section 5.25. The Company and the Stockholder hereby represent and warrant that the persons listed on *Schedule 5.25* comprise all of the persons described in Section 5.25(a), above.
  - (e) It is the intent of the parties that the provisions of this Section will be enforced to the fullest extent permissible under applicable law. If any particular provision or portion of this Section is adjudicated to be invalid or unenforceable, the Agreement will be deemed amended to revise that provision or portion to the

minimum extent necessary to render it enforceable. Such amendment will apply only with respect to the operation of this Section in the particular jurisdiction in which such adjudication was made.

#### ARTICLE VI

#### CONDITIONS TO THE TRANSACTION

- 6.1 *Conditions to Obligations of Each Party to Effect the Mergers*. The respective obligations of each party to this Agreement to effect the Mergers shall be subject to the satisfaction at or prior to the Closing Date of the following conditions:
- (a) *Parent Stockholder Approval*. The Parent Stockholder Approval shall have been obtained by the requisite vote under the laws of the State of Delaware and the Parent Charter Documents.
- (b) Parent Common Stock. Holders of thirty percent (30%) or more of the shares of Parent Common Stock issued in Parent s initial public offering of securities and outstanding as of the record date for the Special Meeting shall not have exercised their rights to convert their shares into a pro rata share of the Trust Fund in accordance with Parent s certificate of incorporation.
- (c) Effective Registration Statement. The Registration Statement shall have been declared effective by the SEC and no stop order suspending the effectiveness of the Registration Statement shall be in effect and no proceedings for that purpose shall be pending before or threatened by the SEC. All necessary permits and authorizations under state securities or Blue Sky laws, the Securities Act and the Exchange Act relating to the issuance and trading of the Holdco Common Stock to be issued hereunder shall have been obtained and shall be in effect.
- (d) *HSR Act; No Order*. All specified waiting periods under the HSR Act shall have expired and no Governmental Entity shall have enacted, issued, promulgated, enforced or entered any statute, rule, regulation, executive order, decree, injunction or other order (whether temporary, preliminary or permanent) which is in effect and which has the effect of making the Mergers illegal or otherwise prohibiting consummation of the Mergers, substantially on the terms contemplated by this Agreement.
- (e) *Opinion of Counsels*. The Company shall have received from Graubard Miller, counsel to Parent, and Parent shall have received from Ellenoff Grossman & Schole LLP, counsel to the Company, an opinion of counsel in the form mutually agreed upon by the parties.
- 6.2 Additional Conditions to Obligations of the Company and the Stockholder. The obligations of the Company and the Stockholder to consummate and effect the Mergers shall be subject to the satisfaction at or prior to the Closing Date of each of the following conditions, any of which may be waived, in writing, exclusively by the Company.
- (a) Representations and Warranties. Each representation and warranty of Parent contained in this Agreement that is (i) qualified as to materiality shall have been true and correct (A) as of the date of this Agreement and (B) subject to the provision of the last sentence of Section 5.13, on and as of the Closing Date with the same force and effect as if made on the Closing Date, and (ii) not qualified as to materiality shall have been true and correct (A) as of the date of this Agreement and (B) subject to the provision of the last sentence of Section 5.13, on and as of the Closing Date in all material respects with the same force and effect as if made on the Closing Date. The Company shall have received a certificate with respect to the foregoing signed on behalf of Parent by an authorized officer of Parent ( Parent Closing Certificate ).

- (b) *Agreements and Covenants*. Parent and the Parent Subsidiaries shall have materially performed or complied with all agreements and covenants required by this Agreement to be performed or complied with by them on or prior to the Closing Date, and the Parent Closing Certificate shall include a provision to such effect.
- (c) *No Litigation*. No action, suit or proceeding shall be pending or threatened before any Governmental Entity or by any third party against Parent or any Affiliate thereof which, in the reasonable opinion of Company counsel, is likely to occur and (i) prevent consummation of any of the transactions contemplated by this Agreement, (ii) cause any of the transactions contemplated by this Agreement to be rescinded following consummation or (iii) affect materially and adversely the right of the Surviving Pubco or Surviving Subsidiary

#### **TABLE OF CONTENTS**

to own, operate or control any of its respective assets following the Mergers and no order, judgment, decree, stipulation or injunction to any such effect shall be in effect..

- (d) *Consents*. Parent shall have obtained all material consents, waivers and approvals required to be obtained by Parent in connection with the consummation of the transactions contemplated hereby, and the Parent Closing Certificate shall include a provision to such effect.
- (e) *Material Adverse Effect*. No Material Adverse Effect with respect to Parent shall have occurred since the date of this Agreement.
- (f) SEC Compliance. Immediately prior to Closing, Parent shall be in compliance with the reporting requirements under the Securities Act and Exchange Act applicable to Parent.

## (g) [Reserved]

- (h) *Other Deliveries*. At or prior to Closing, Parent shall have delivered to the Company (i) copies of resolutions and actions taken by Parent's and each Parent Subsidiary s board of directors and stockholders in connection with the approval of this Agreement and the transactions contemplated hereunder, as certified by the respective corporate secretary of each of Parent and the Parent Subsidiaries and (ii) such other documents or certificates as shall reasonably be required by the Company and its counsel in order to consummate the transactions contemplated hereunder.
- (i) *Resignations*. The persons listed in *Schedule 6.2(i)* shall have resigned from all of their positions and offices with Parent.
- (j) *Trust Fund*. Parent shall have made appropriate arrangements to have the Trust Fund disbursed in accordance with *Section 5.20*.
  - (k) *Special Meeting*. The requisite approval from the holders of Parent Common Stock and Parent Warrants at the Special Meeting shall have been obtained such that those proposals and actions (other than those regarding adjournment) described in Section 5.1 shall have been adopted in accordance with the DGCL and Parent s Charter.
  - (1) Warrant Amendment Proposal. The Warrant Amendment Proposal shall have been approved by the necessary warrantholders to adopt and effect the Warrant Amendment.
- (m) *The Consulting Agreement Amendments*. The Consulting Agreement Amendment shall be executed and delivered by Parent.
  - (n) *Ledecky Founder Share Surrender*. The shares of Holdco Common Stock owned by Mr. Ledecky to be surrendered under Section 5.22 shall have been so surrendered by Mr. Ledecky.
- (o) *Opinion of Delaware Counsel*. Parent shall have received an opinion of Richards, Layton & Finger, P.A., in form and substance reasonably satisfactory to both Parent and the Company, that the amendments to Parent s amended and restated certificate of incorporation necessary to allow Parent to complete the Mergers as contemplated hereby would be valid and effective under the DGCL when filed with the Secretary of State of Delaware and such amended and restated certificate of incorporation of Parent shall have been filed with the Secretary of State of Delaware and shall be in full force and effect at the Closing.

(p) [Reserved]

6.3 Additional Conditions to the Obligations of Parent. The obligations of Parent and the Parent Subsidiaries to consummate and effect the Mergers shall be subject to the satisfaction at or prior to the Closing Date of each of the following conditions, any of which may be waived, in writing, exclusively by Parent:

(a) Representations and Warranties. Each representation and warranty of the Company and the Stockholder contained in this Agreement that is (i) qualified as to materiality shall have been true and correct (A) as of the date of this Agreement and (B) subject to the provisions of the last sentence of Section 5.13, on and as of the Closing Date with the same force and effect as if made on the Closing Date, and (ii) not qualified as to materiality shall have been true and correct (A) as of the date of this Agreement and (B) subject to the provisions of the last sentence of Section 5.13, on and as of the Closing Date in all material respects with the same

force and effect as if made on the Closing Date. Parent shall have received a certificate with respect to the foregoing signed on behalf of each of the Company and the Stockholder by an authorized officer of each ( *Company Closing Certificate* ).

- (b) Agreements and Covenants. The Company and the Stockholder shall have performed or complied with all agreements and covenants required by this Agreement to be performed or complied with by them at or prior to the Closing Date except to the extent that any failure to perform or comply (other than a willful failure to perform or comply or failure to perform or comply with an agreement or covenant reasonably within the control of the Company) does not, or will not, constitute a Material Adverse Effect on the Company, and the Company Closing Certificate shall include a provision to such effect.
- (c) *No Litigation*. No action, suit or proceeding shall be pending or threatened before any Governmental Entity or by any third party against the Company or the Stockholder which, in the reasonable opinion of Parent counsel, is likely to occur and (i) prevent consummation of any of the transactions contemplated by this Agreement, (ii) cause any of the transactions contemplated by this Agreement to be rescinded following consummation or (iii) affect materially and adversely the right of the Surviving Pubco or Surviving Subsidiary to own, operate or control any of its respective assets following the Mergers and no order, judgment, decree, stipulation or injunction to any such effect shall be in effect.
- (d) *Consents*. The Company and the Stockholder shall have obtained all material consents, waivers, permits and approvals required to be obtained by the Company and the Stockholder in connection with the consummation of the transactions contemplated hereby, including all Governmental and Third Party Consents, and the Company Closing Certificate shall include a provision to such effect.
- (e) *Material Adverse Effect*. No Material Adverse Effect with respect to the Company shall have occurred since the date of this Agreement.
  - (f) Lock-Up Agreements. The Lock-Up Agreements in the form of Exhibit F shall be in full force and effect.
- (g) *Employment Agreements*. Employment Agreements between Natural Dairy, Inc. and Richard Watson in the form of *Exhibit L* shall be executed and delivered to Parent.
  - (h) [Reserved]
  - (i) [Reserved]
- (j) *Cullen Noncompete Letters*. Parent shall have received executed Cullen Noncompete Letters from each of the persons required under Section 5.25 above and each such letter shall have been duly and validly executed and shall be in full force and effect.
- (k) *Watson Founder Share Surrender*. The surrender of the shares of Holdco Common Stock owned by Mr. Watson to be surrendered under Section 5.22 shall have been so surrendered by Mr. Watson.
- (l) Farm Purchase. Natural Dairy, Inc. (or a subsidiary thereof) has performed its obligations under the Farm Purchase Agreement and, if the Farm has been purchased, Natural Dairy, Inc. (or a subsidiary thereof) shall have held and maintained the Farm on behalf of the Surviving Pubco and Surviving Company as contemplated by Section 4.2(b).

## (m) [Reserved]

- (n) *The Consulting Agreement Amendments*. The Consulting Agreement Amendment shall be executed and delivered by each of Messrs. Vassilakos (or his entity as applicable) and Leahy.
  - (o) *IP Transfer Agreement*. The IP Transfer Agreement shall be in full force and effect and the transactions contemplated thereby shall have been consummated or shall be consummated concurrently with the Closing.
- (p) Other Deliveries. At or prior to Closing, the Company and the Stockholder shall have delivered to Parent: (i) copies of resolutions and actions taken by the Company's board of directors and the Stockholder s board of directors and the Stockholder s stockholders in connection with the approval of this Agreement and

the transactions contemplated hereunder, and (ii) such other documents or certificates as shall reasonably be required by Parent and its counsel in order to consummate the transactions contemplated hereunder.

(q) *Resignations*. The persons listed in *Schedule 6.3(q)* shall have resigned from their positions and offices with the Company.

#### ARTICLE VII

#### **INDEMNIFICATION**

#### 7.1 Indemnification.

- (a) *Indemnification of Parent*. Subject to the terms and conditions of this Article VII (including without limitation the limitations set forth in Section 7.4), Parent, the Surviving Pubco, the Surviving Subsidiary and their respective representatives, successors and permitted assigns (the *Parent Indemnitees*) shall be indemnified, defended and held harmless by the Stockholder, but, except as provided for in this Article VII, only to the extent of the Escrow Shares, from and against all Losses asserted against, resulting to, imposed upon, or incurred by any Parent Indemnitee by reason of, arising out of or resulting from:
- (i) the inaccuracy or breach of any representation or warranty of the Company contained in or made pursuant to this Agreement, any Schedule or any certificate delivered by the Company to Parent pursuant to this Agreement with respect hereto or thereto in connection with the Closing; and
  - (ii) the non-fulfillment or breach of any covenant or agreement of the Company contained in this Agreement.
- (b) *Losses*. As used in this Article VII, the term *Losses* shall include all losses, liabilities, damages, judgments, awards, orders, penalties, settlements, costs and expenses (including, without limitation, interest, penalties, court costs and reasonable legal fees and expenses) including those arising from any demands, claims, suits, actions, costs of investigation, notices of violation or noncompliance, causes of action, proceedings and assessments whether or not made by third parties or whether or not ultimately determined to be valid.
- 7.2 *Indemnification of Third Party Claims*. The indemnification obligations and liabilities under this Article VII with respect to actions, proceedings, lawsuits, investigations, demands or other claims brought against a party under an obligation to indemnify pursuant to Section 7.1 (the *Indemnifying Party*) by a Person other than a party entitled to indemnification pursuant to Section 7.1 (the *Indemnified Party*) (a *Third Party Claim*) shall be subject to the following terms and conditions:
- (a) *Notice of Claim*. The Indemnified Party will give the Indemnifying Party prompt written notice after receiving written notice of any Third Party Claim or discovering the liability, obligation or facts giving rise to such Third Party Claim (a *Notice of Claim*) which Notice of Third Party Claim shall set forth (i) a brief description of the nature of the Third Party Claim, (ii) the total amount of the actual out-of-pocket Loss or the anticipated potential Loss (including any costs or expenses which have been or may be reasonably incurred in connection therewith), and (iii) whether such Loss may be covered (in whole or in part) under any insurance and the estimated amount of such Loss which may be covered under such insurance, and the Indemnifying Party shall be entitled to participate in the defense of Third Party Claim at its expense.
  - (b) *Defense*. The Indemnifying Party shall have the right, at its option (subject to the limitations set forth in subsection 7.2(c) below) and at its own expense, by written notice to the Indemnified Party, to assume the entire

control of, subject to the right of the Indemnified Party to participate (at its expense and with counsel of its choice) in, the defense, compromise or settlement of the Third Party Claim as to which such Notice of Claim has been given, and shall be entitled to appoint a recognized and reputable counsel reasonably acceptable to the Indemnified Party to be the lead counsel in connection with such defense. If the Indemnifying Party is permitted and elects to assume the defense of a Third Party Claim:

(i) the Indemnifying Party shall diligently and in good faith defend such Third Party Claim and shall keep the Indemnified Party reasonably informed of the status of such defense; provided, however, that the Indemnified Party shall have the right to approve any settlement, which approval will not be unreasonably withheld, delayed or conditioned; and

- (ii) the Indemnified Party shall cooperate fully in all respects with the Indemnifying Party in any such defense, compromise or settlement thereof, including, without limitation, the selection of counsel, and the Indemnified Party shall make available to the Indemnifying Party all pertinent information and documents under its control.
- (c) *Limitations of Right to Assume Defense*. The Indemnifying Party shall not be entitled to assume control of such defense and shall pay the fees and expenses of counsel retained by the Indemnified Party if (i) the Third Party Claim relates to or arises in connection with any criminal proceeding, action, indictment, allegation or investigation; (ii) the Third Party Claim seeks an injunction or equitable relief against an Indemnified Party; or (iii) there is a reasonable probability that a Third Party Claim may materially and adversely affect the Indemnified Party or its subsidiaries other than as a result of money damages or other money payments.
- (d) *Other Limitations*. Failure to give prompt Notice of Claim or to provide copies of relevant available documents or to furnish relevant available data shall not constitute a defense (in whole or in part) to any Third Party Claim by an Indemnified Party against the Indemnifying Party and shall not affect the Indemnifying Party s duty or obligations under this Article VII, except to the extent (and only to the extent that) such failure shall have adversely affected the ability of the Indemnifying Party to defend against or reduce its liability or caused or increased such liability or otherwise caused the damages for which the Indemnifying Party is obligated to be greater than such damages would have been had the Indemnified Party given the Indemnifying Party prompt notice hereunder. So long as the Indemnifying Party is defending any such action actively and in good faith, the Indemnified Party shall not settle such action. The Indemnified Party shall make available to the Indemnifying Party all relevant records and other relevant materials required by them and in the possession or under the control of the Indemnified Party, for the use of the Indemnifying Party and its representatives in defending any such action, and shall in other respects give reasonable cooperation in such defense.
- (e) Failure to Defend. If the Indemnifying Party, promptly after receiving a Notice of Claim, fails to defend such Third Party Claim actively and in good faith, the Indemnified Party, at the reasonable cost and expense of the Indemnifying Party, will (upon further written notice) have the right to undertake the defense, compromise or settlement of such Third Party Claim as it may determine in its reasonable discretion, provided that the Indemnifying Party shall have the right to approve any settlement, which approval will not be unreasonably withheld, delayed or conditioned.
- (f) *Indemnified Party Rights*. Anything in this Section 7.2 to the contrary notwithstanding, the Indemnifying Party shall not settle or compromise any action or consent to the entry of any judgment which does not include as an unconditional term thereof the giving by the claimant or the plaintiff to each Indemnified Party of a full and unconditional release from all liability and obligation in respect of such action without any payment by any Indemnified Party.
- (g) *Indemnifying Party Consent*. Unless the Indemnifying Party has consented to a settlement of a Third Party Claim, the amount of the settlement shall not be a binding determination of the amount of the Loss and such amount shall be determined in accordance with the provisions of the Escrow Agreement.
- 7.3 *Insurance Effect*. To the extent that any Losses that are subject to indemnification pursuant to this Article VII are covered by insurance paid for by an Indemnified Party prior to the Closing, the Indemnified Party shall use commercially reasonable best efforts to obtain the maximum recovery under such insurance; provided that the Indemnified Party shall nevertheless be entitled to bring a claim for indemnification under this Article VII in respect of such Losses and the time limitations set forth in Section 7.4 hereof for bringing a claim of indemnification under this Agreement shall be tolled during the pendency of such insurance claim. The existence of a claim by the Indemnified Party for monies from an insurer or against a third party in respect of any Loss shall not, however, delay

any payment pursuant to the indemnification provisions contained herein and otherwise determined to be due and owing by the Indemnifying Party. If the Indemnified Party has received the payment required by this Agreement from the Indemnifying Party in respect of any Loss and later receives proceeds from insurance or other amounts in respect of such Loss, then it shall hold such proceeds or other amounts in trust for the benefit of the Indemnifying Party and shall pay to the Indemnifying Party, as promptly as practicable after receipt, a sum equal to the amount of such proceeds or other

amount received, up to the aggregate amount of any payments received from the Indemnifying Party pursuant to this Agreement in respect of such Loss. Notwithstanding any other provisions of this Agreement, it is the intention of the parties that no insurer or any other third party shall be (i) entitled to a benefit it would not be entitled to receive in the absence of the foregoing indemnification provisions, or (ii) relieved of the responsibility to pay any claims for which it is obligated.

#### 7.4 Limitations on Indemnification.

- (a) *Survival; Time Limitation*. The representations, warranties, covenants and agreements in this Agreement or in any writing delivered by the Company or Stockholder to Parent in connection with this Agreement (including the certificate required to be delivered by the Company pursuant to Section 6.3(a)) shall survive the Closing for the period that ends on the Escrow Termination Date (the *Survival Period*) except that claims for the breach of the representations and warranties in Sections 1.15(d), 2.3, and 2.4 shall survive without limitation as to time (but subject to applicable statutes of limitation).
- (b) Any indemnification claim made by an Indemnified Party prior to the termination of the Survival Period shall be preserved despite the subsequent termination of the Survival Period and any claim set forth in a Notice of Claim sent prior to the expiration of such Survival Period shall survive until final resolution thereof. Except as set forth in the immediately preceding sentence, no claim for indemnification under this Article VII shall be brought after the end of the Survival Period.
- (c) *Deductible*. No amount shall be payable under Article VII unless and until the aggregate amount of all indemnifiable Losses otherwise payable exceeds \$500,000 (the *Deductible*), in which event the amount payable shall the full amount of all Indemnifiable Losses from the first dollar thereof.
- (d) *Aggregate Amount Limitation*. An Indemnifying Party s aggregate liability for Losses pursuant to Section 7.1shall not in any event exceed the Escrow Shares and the Indemnified Party shall have no claim against the Indemnifying Party other than for such Escrow Shares (and any proceeds of the shares or distributions with respect to the Escrow Shares).
- 7.5 Exclusive Remedy. The Indemnified Party hereby acknowledges and agrees that, from and after the Closing, the sole remedy of the Indemnified Party with respect to any and all claims for money damages arising out of or relating to this Agreement shall be pursuant and subject to the requirements of the indemnification provisions set forth in this Article VII. Notwithstanding any of the foregoing, nothing contained in this Article VII shall in any way impair, modify or otherwise limit an Indemnified Party s right to bring any claim, demand or suit against the Indemnifying Party based upon the Indemnifying Party s actual fraud or intentional or willful misrepresentation or omission, it being understood that a mere breach of a representation and warranty, without intentional or willful misrepresentation or omission, does not constitute fraud.
- 7.6 Adjustment to Merger Consideration. Amounts paid by the Stockholder for indemnification under Article VII shall be deemed to be an adjustment to the value of the shares of Holdco Common Stock issued by Holdco as a result of the Company Merger, except as otherwise required by Law.
- 7.7 Application of Escrow Shares. The Escrow Agent, pursuant to the Escrow Agreement after the Closing, may apply all or a portion of the Escrow Shares to satisfy any claim for indemnification pursuant to this Article VII. The Escrow Agent will hold the remaining portion of the Escrow Shares until final resolution of all claims for indemnification or disputes relating thereto.

#### ARTICLE VIII

#### **TERMINATION**

- 8.1 *Termination*. This Agreement may be terminated at any time prior to the Closing:
  - (a) by mutual written agreement of Parent and the Company at any time;
- (b) by either Parent or the Company if the Mergers shall not have been consummated by October 22, 2009 for any reason; provided, however, that the right to terminate this Agreement under this Section 8.1(b) shall not be available to any party whose action or failure to act has been a principal cause of or resulted in the failure of the Mergers to occur on or before such date and such action or failure to act constitutes a breach of this Agreement;
- (c) by either Parent or the Company if a Governmental Entity shall have issued an order, decree, judgment or ruling or taken any other action, in any case having the effect of permanently restraining, enjoining or otherwise prohibiting the Mergers, which order, decree, ruling or other action is final and nonappealable;
- (d) by the Company, upon a breach of any representation, warranty, covenant or agreement on the part of Parent set forth in this Agreement, or if any representation or warranty of Parent shall have become untrue, in either case such that the conditions set forth in Article VI would not be satisfied as of the time of such breach or as of the time such representation or warranty shall have become untrue, provided, that if such breach by Parent is curable by Parent prior to the Closing Date, then the Company may not terminate this Agreement under this Section 8.1(d) for thirty (30) days after delivery of written notice from the Company to Parent of such breach, provided Parent continues to exercise commercially reasonable best efforts to cure such breach (it being understood that the Company may not terminate this Agreement pursuant to this Section 8.1(d) if it shall have materially breached this Agreement or if such breach by Parent is cured during such thirty (30)-day period);
- (e) by Parent, upon a breach of any representation, warranty, covenant or agreement on the part of the Company or the Stockholder set forth in this Agreement, or if any representation or warranty of the Company or the Stockholder shall have become untrue, in either case such that the conditions set forth in Article VI would not be satisfied as of the time of such breach or as of the time such representation or warranty shall have become untrue, provided, that if such breach is curable by the Company or the Stockholder prior to the Closing Date, then Parent may not terminate this Agreement under this Section 8.1(e) for thirty (30) days after delivery of written notice from Parent to the Company or the Stockholder, as applicable, of such breach, provided the Company or the Stockholder, as applicable, continue to exercise commercially reasonable best efforts to cure such breach (it being understood that Parent may not terminate this Agreement pursuant to this Section 8.1(e) if it shall have materially breached this Agreement or if such breach by the Company or the Stockholder is cured during such thirty (30) day period); or
- (f) by either Parent or the Company, if, at the Special Meeting (including any adjournments thereof), this Agreement and the transactions contemplated thereby shall fail to be approved and adopted by the affirmative vote of the holders of Parent Common Stock required under Parent s certificate of incorporation, or the holders of 30% or more of the number of shares of Parent Common Stock issued in Parent s initial public offering and outstanding as of the record date for the Special Meeting exercise their rights to convert the shares of Parent Common Stock held by them into cash in accordance with Parent s certificate of incorporation.
- 8.2 *Notice of Termination; Effect of Termination*. Any termination of this Agreement under Section 8.1 above will be effective immediately upon (or, if the termination is pursuant to Section 8.1(d) or Section 8.1(e) and the proviso therein is applicable, thirty (30) days after) the delivery of written notice of the terminating party to the other parties

hereto. In the event of the termination of this Agreement as provided in Section 8.1, this Agreement shall be of no further force or effect and the Mergers shall be abandoned, except for and subject to the following: (i) Sections 4.2, 5.6, 5.12, 8.2 and 8.3 and Article X (General Provisions) shall survive the termination of this Agreement, and (ii) nothing herein shall relieve any party from liability for any breach of this Agreement, including a breach by a party electing to terminate this Agreement pursuant to Section 8.1(b) caused by the action or failure to act of such party constituting a principal cause of or resulting in the failure of the Mergers to occur on or before the date stated therein.

8.3 Fees and Expenses. Except as set forth in Sections 3.25 and 5.20, all fees and expenses incurred in connection with this Agreement and the transactions contemplated hereby shall be paid by the party incurring such expenses whether or not the Mergers are consummated. Notwithstanding the foregoing, in the event the Mergers are consummated, Parent and Holdco hereby agree to pay the reasonable fees and expenses incurred by the Stockholder, the Company or any Subsidiary, or on behalf of the Stockholder, the Company or any Subsidiary, in the connection with the preparation, negotiation, execution and performance of this Agreement and the transactions contemplated hereby as further described in *Schedule 5.20*.

#### ARTICLE IX

#### **DEFINED TERMS**

Certain terms defined in this Agreement are organized alphabetically as follows, together with the Section and, where applicable, paragraph, number in which definition of each such term is located:

AAA	Section 10.12
Acquisition Transaction	Section 4.4(a)
Affiliate	Section 10.2(f)
Agreement	Section 1.3
Applicable Corporate Law	Recital A
Approvals	Section 2.1(a)
Audited Financial Statements	Section 2.7(a)
Blue Sky Laws	Section1.15(c)
Charter Documents	Section 2.1(a)
Closing	Section 1.3
Closing Date	Section 1.3
Closing Form 8-K	Section 5.1(b)
Closing Press Release	Section 5.1(b)
Code	Recital C
Committee	Section 1.16
Company	Heading
Company Articles of Merger	Section 1.3
Company Certificates	Section 1.6(d)
Company Closing Certificate	Section 6.3(a)
Company Common Stock	Section 5.26(e)
Company Contracts	Section 2.19(a)
Company Intellectual Property	Section 2.18(a)(ii)
Company Merger	Recital A
Company Products	Section 2.18(a)(v)

Company Products

Company Registered Intellectual Property

Company Related Parties

Company Schedule

Continental

Section 2.18(a)(v)

Section 2.18(a)(iv)

Section 2.18(a)

Article II Preamble

Section 1.12

Continental Section 1.12
Copyrights Section 2.18(a)(i)
Deductible Section 7.4(c)
DGCL Recital A
Disclosure Schedules Section 5.13

Effective Time Section 1.3
Environmental Law Section 2.16(b)
Escrow Agreement Section 1.12

Facus Chanca	Castian 1 10
Escrow Shares	Section 1.12
Escrow Termination Date	Section 1.12
Exchange Act	Section 1.15(c)
Farm Purchase Agreement	Section 4.2
GCC	Recital A
Governmental Action/Filing	Section 2.21(c)
Governmental and Third Party Consents	Section 2.5(b)
Governmental Entity	Section 1.15(c)
Grimsley Farm	Section 4.2
Hazardous Substance	Section 2.16(c)
Holdco	Heading
Holdco Common Stock	Section 1.6(a)
Holdco Plan	Section 5.1(a)
Holdco Warrants	Section 1.6(b)
HSR Act	Section 2.5(b)
Insider	Section 2.19(a)(i)
Insurance Policies	Section 2.20
Intellectual Property	Section 2.18(a)
IP Transfer Agreement	Section 2.18(a)
knowledge	Section 10.2(d)
Last Reported Sale Price	Section 1.13(c)(iv)
Legal Requirements	Section 10.2(b)
Lien	Section 10.2(e)
Losses	Section 7.1(b)
Material Adverse Effect	Section 10.2(a)
Material Company Contracts	Section 2.19(a)
Mergers	Recital A
Merger Sub	Heading
Merger Sub Common Stock	Section 1.6(e)
Notice of Claim	Section 7.2(a)
Parent	Heading
Parent Certificate	Section 1.6(a)
Parent Certificate of Merger	Section 1.3
Parent Closing Certificate	Section 6.2(a)
Parent Common Stock	Section 1.6(a)
Parent Contracts	Section 3.19(a)
Parent Convertible Securities	Section 3.17(a) Section 3.3(b)
Parent Indemnitees	Section 7.1(a)
Parent Merger	Recital A
Parent Preferred Stock	
Parent Schedule	Section 3.3(a)
	Article III Preamble
Parent SEC Reports	Section 3.7(a)
Parent Stockholder Approval	Section 5.1(a)
Parent Stock Options	Section 3.3(b)
Parent Subsidiaries	Section 3.2(d)
Parent Warrant	Section 1.6(b)

Parent Warrant Certificate	Section 1.6(b)
Patents	Section 2.18(a)(i)
Person	Section 10.2(c)
Personal Property	Section 2.14(b)
Plan/Plans	Section 2.11(a)
Proxy Statement/Prospectus	Section 5.1(a)
Purchase Cost	Section 4.2
Registered Intellectual Property	Section 2.18(a)(iii)
Registration Rights Agreement	Section 1.18
Registration Statement	Section 5.1(a)
Representative	Section 1.18
Returns	Section 2.15(b)(i)
SEC	Section 2.21(d)
Securities Act	Section 1.15(b)
Signing Form 8-K	Section 5.4(a)
Signing Press Release	Section 5.4(a)
Special Meeting	Section 5.1(a)
Stockholder	Heading
Subsidiaries	Section 2.2(a)
Survival Period	Section 7.4(a)
Surviving Pubco	Section 1.1
Surviving Subsidiary	Section 1.2
Tax/Taxes	Section 2.15(a)
Third Party Claim	Section 7.2
Trademarks	Section 2.18(a)(i)
Trading Day	Section $1.13(c)(v)$
Transaction Shares	Section 1.6(d)
Trust Fund	Section 3.25
U.S. GAAP	Section 2.7(a)
Warrant Amendment Agreement	Section 5.1(c)
Warrantholders Approval	Section 5.1(c)
Warrantholders Meeting	Section 5.1(c)

#### ARTICLE X

#### **GENERAL PROVISIONS**

10.1 *Notices*. All notices and other communications hereunder shall be in writing and shall be deemed given if delivered personally or by commercial delivery service, or sent via telecopy (receipt confirmed) to the parties at the following addresses or telecopy numbers (or at such other address or telecopy numbers for a party as shall be specified by like notice):

if to Parent, to:

Triplecrown Acquisition Corp 970 West Broadway PMB 402 Jackson, Wyoming 83001 Attention: Chief Executive Officer Telephone: 307-633-2831 Telecopy:

with a copy to:

David Alan Miller, Esq.
Graubard Miller
405 Lexington Avenue
New York, New York 10174-1901
Telephone: 212-818-8661
Telecopy: 212-818-8881

if to the Company or the Stockholder, to:

Cullen Inc Holdings Limited Level 9 Walker Wayland Centre 68 Shortland Street Auckland, New Zealand Post Office Box 91269 Attention: The Directors Telephone: 64 9 363 1200 Telecopy: 64 9 363 1201

with a copy to:

Douglas S. Ellenoff, Esq. Ellenoff Grossman & Schole LLP 150 East 42<sup>nd</sup> Street New York, New York 10017 Telephone: 212-370-1300

Telecopy: 212-370-7889

10.2 Interpretation. The definitions of the terms herein shall apply equally to the singular and plural forms of the terms defined. Whenever the context shall require, any pronoun shall include the corresponding masculine, feminine and neuter forms. When a reference is made in this Agreement to an Exhibit or Schedule, such reference shall be to an Exhibit or Schedule to this Agreement unless otherwise indicated. When a reference is made in this Agreement to Sections or subsections, such reference shall be to a Section or subsection of this Agreement. Unless otherwise indicated the words include, includes and including when used herein shall be deemed in each case to be followed by the words without limitation. The table of contents and headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. When reference is made herein to the business of an entity, such reference shall be deemed to include the business of all direct and indirect subsidiaries of such entity.

Reference to the subsidiaries of an entity shall be deemed to include all direct and indirect subsidiaries of such entity.

For purposes of this Agreement:

- (a) the term *Material Adverse Effect* when used in connection with an entity means any change, event, violation, inaccuracy, circumstance or effect, individually or when aggregated with other changes, events, violations, inaccuracies, circumstances or effects, that is materially adverse to the business, assets (including intangible assets), revenues, financial condition, prospects or results of operations of such entity, it being understood that none of the following alone or in combination shall be deemed, in and of itself, to constitute a Material Adverse Effect: (i) changes attributable to the public announcement or pendency of the transactions contemplated hereby, (ii) changes in general national or regional political, economic or industry-wide conditions (except to the extent the party suffering such event is affected in a disproportionate manner relative to other companies in the industry in which such party conducts business), (iii) any SEC rulemaking requiring enhanced disclosure of transactions with a public shell, or (iv) any failure to meet any financial or other projections;
- (b) the term *Legal Requirements* means any federal, state, local, municipal, foreign or other law, statute, constitution, principle of common law, resolution, ordinance, code, edict, decree, rule, regulation, ruling or requirement issued, enacted, adopted, promulgated, implemented or otherwise put into effect by or under the authority of any Governmental Entity and all requirements set forth in applicable Material Company Contracts or Parent Contracts;
- (c) the term *Person* shall mean any individual, corporation (including any non-profit corporation), general partnership, limited partnership, limited liability partnership, joint venture, estate, trust, company (including any limited liability company or joint stock company), firm or other enterprise, association, organization, entity or Governmental Entity;
- (d) the term *knowledge* means actual knowledge or awareness, after due inquiry, as to a specified fact or event of a Person that is an individual or of an executive officer or director of a Person that is a corporation or of a Person in a similar capacity of an entity other than a corporation;
- (e) the term *Lien* means any mortgage, pledge, security interest, encumbrance, lien, restriction or charge of any kind (including, without limitation, any conditional sale or other title retention agreement or lease in the nature thereof, any sale with recourse against the seller or any Affiliate of the seller, or any agreement to give any security interest);
- (f) the term *Affiliate* means, as applied to any Person, any other Person directly or indirectly controlling, controlled by or under direct or indirect common control with, such Person. For purposes of this definition, control (including with correlative meanings, the terms controlling, controlled by and under common control with ), as applied to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract or otherwise; and
  - (g) all monetary amounts set forth herein are referenced in United States dollars, unless otherwise noted.
- 10.3 Counterparts; Electronic Delivery. This Agreement and each other document executed in connection with the transactions contemplated hereby, and the consummation thereof, may be executed in one or more counterparts, all of which shall be considered one and the same document and shall become effective when one or more counterparts have been signed by each of the parties and delivered to the other party, it being understood that all parties need not sign the same counterpart. Delivery by facsimile or electronic transmission to counsel for the other party of a counterpart executed by a party shall be deemed to meet the requirements of the previous sentence.
  - 10.4 Entire Agreement; Third Party Beneficiaries. This Agreement and the documents and instruments and other agreements among the parties hereto as contemplated by or referred to herein, including the Exhibits and Schedules

hereto (a) constitute the entire agreement among the parties with respect to the subject matter hereof and supersede all prior agreements and understandings, both written and oral, among the parties with respect to the subject matter hereof, and (b) are not intended to confer upon any other Person any rights or remedies hereunder (except as specifically provided in this Agreement).

10.5 Severability. In the event that any provision of this Agreement, or the application thereof, becomes or is declared by a court of competent jurisdiction to be illegal, void or unenforceable, the remainder of this Agreement will continue in full force and effect and the application of such provision to other persons or circumstances will be interpreted so as reasonably to effect the intent of the parties hereto. The parties further agree to replace such void or unenforceable provision of this Agreement with a valid and enforceable provision that will achieve, to the extent possible, the economic, business and other purposes of such void or unenforceable provision.

10.6 Other Remedies; Specific Performance. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a party will be deemed cumulative with and not exclusive of any other remedy conferred hereby, or by law or equity upon such party, and the exercise by a party of any one remedy will not preclude the exercise of any other remedy. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to seek an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court of the United States or any state having jurisdiction, this being in addition to any other remedy to which they are entitled at law or in equity.

10.7 *Governing Law*. This Agreement shall be governed by and construed in accordance with the law of the State of New York regardless of the law that might otherwise govern under applicable principles of conflicts of law thereof.

10.8 *Rules of Construction*. The parties hereto agree that they have been represented by counsel during the negotiation and execution of this Agreement and, therefore, waive the application of any law, regulation, holding or rule of construction providing that ambiguities in an agreement or other document will be construed against the party drafting such agreement or document.

10.9 Assignment. No party may assign either this Agreement or any of its rights, interests, or obligations hereunder without the prior written approval of the other parties. Subject to the first sentence of this Section 10.9, this Agreement shall be binding upon and shall inure to the benefit of the parties hereto and their respective successors and permitted assigns.

10.10 *Amendment*. This Agreement may be amended by the parties hereto at any time by execution of an instrument in writing signed on behalf of each of the parties.

10.11 Extension; Waiver. At any time prior to the Closing, any party hereto may, to the extent legally allowed, (i) extend the time for the performance of any of the obligations or other acts of the other parties hereto, (ii) waive any inaccuracies in the representations and warranties made to such party contained herein or in any document delivered pursuant hereto and (iii) waive compliance with any of the agreements or conditions for the benefit of such party contained herein. Any agreement on the part of a party hereto to any such extension or waiver shall be valid only if set forth in an instrument in writing signed on behalf of such party. Delay in exercising any right under this Agreement shall not constitute a waiver of such right.

10.12 *Arbitration*. Any disputes or claims arising under or in connection with this Agreement or the transactions contemplated hereunder shall be resolved by binding arbitration. Notice of a demand to arbitrate a dispute by either party shall be given in writing to the other at their last known address. Arbitration shall be commenced by the filing by a party of an arbitration demand with the American Arbitration Association ( *AAA* ) in its office in New York City, New York. The arbitration and resolution of the dispute shall be resolved by a single arbitrator appointed by the AAA pursuant to AAA rules. The arbitration shall in all respects be governed and conducted by applicable AAA rules, and any award and/or decision shall be conclusive and binding on the parties. The arbitration shall be conducted in New

York City or such other place as the parties shall agree. The arbitrator shall supply a written opinion supporting any award, and judgment may be entered on the award in any court of competent jurisdiction. Each party shall pay its own fees and expenses for the arbitration, except that any costs and charges imposed by the AAA and any fees of the arbitrator for his services shall be assessed against the losing party by the arbitrator. In the event that preliminary or permanent injunctive relief is necessary or desirable in order to prevent a party from acting contrary to this Agreement or to prevent irreparable harm prior to a confirmation of an arbitration award, then either party

is authorized and entitled to commence a lawsuit solely to obtain equitable relief against the other pending the completion of the arbitration in a court having jurisdiction over the parties. Each party hereby consents to the exclusive jurisdiction of the federal and state courts located in the State of New York, New York County, for such purpose. All rights and remedies of the parties shall be cumulative and in addition to any other rights and remedies obtainable from arbitration.

10.13 *Waiver of Jury Trial*. Each of the parties hereto hereby waives any right it may have to a trial by jury with respect to any action or dispute arising out of or in connection with this Agreement or the transactions contemplated hereby.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed as of the date first written above.

TRIPLECROWN ACQUISTION CORP.
By:
Name: Title:
CULLEN AGRICULTURAL HOLDING CORP. By:
Name: Title:
CAT MERGER SUB, INC. By:
Name: Title:
CULLEN AGRICULTURAL TECHNOLOGIES, INC. By:
Name: Title:

CULLEN INC HOLDINGS LIMITED
By:

Name: Title:

#### Annex B

# **AMENDED AND RESTATED**

# CERTIFICATE OF INCORPORATION

# **OF**

# CULLEN AGRICULTURAL HOLDING CORP.

# Pursuant to Section 245 of the Delaware General Corporation Law

CULLEN AGRICULTURAL HOLDING CORP., a corporation existing under the laws of the State of Delaware (the Corporation ), by its Chief Executive Officer, hereby certifies as follows:

- 1. The name of the Corporation is Cullen Agricultural Holding Corp.
- 2. The Corporation s Certificate of Incorporation was filed in the office of the Secretary of State of the State of Delaware on August 27, 2009.
- 3. This Amended Restated Certificate of Incorporation restates, integrates and amends the Certificate of Incorporation of the Corporation.
- 4. This Amended and Restated Certificate of Incorporation was duly adopted by joint written consent of the directors and stockholders of the Corporation in accordance with the applicable provisions of Sections 141(f), 228, 242 and 245 of the General Corporation Law of the State of Delaware (GCL).
  - 5. The text of the Certificate of Incorporation of the Corporation is hereby amended and restated to read in full as follows:
    - I, the undersigned, in order to form a corporation for the purposes hereinafter stated, under and pursuant to the provisions of the General Corporation Law of the State of Delaware (the GCL), do hereby certify as follows:

FIRST: The name of the corporation is Cullen Agricultural Holding Corp. (hereinafter sometimes referred to as the Corporation ).

SECOND: The registered office of the Corporation is to be located at 615 S. DuPont Hwy., Kent County, Dover, Delaware. The name of its registered agent at that address is National Corporate Research, Ltd.

THIRD: The purpose of the Corporation shall be to engage in any lawful act or activity for which corporations may be organized under the GCL.

FOURTH: The total number of shares of all classes of capital stock which the Corporation shall have authority to issue is 201,000,000 of which 200,000,000 shares shall be Common Stock of the par value of \$.0001 per share and 1,000,000 shall be Preferred Stock of the par value of \$.0001 per share.

A. *Preferred Stock*. The Board of Directors is expressly granted authority to issue shares of the Preferred Stock, in one or more series, and to fix for each such series such voting powers, full or limited, and such designations, preferences and relative, participating, optional or other special rights and such qualifications, limitations or restrictions thereof as shall be stated and expressed in the resolution or resolutions adopted by the Board of Directors providing for the issue of such series (a Preferred Stock Designation ) and as may be permitted by the GCL. The number of authorized shares of Preferred Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the voting power of all of the then outstanding shares of the capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class, without a

separate vote of the holders of the Preferred Stock, or any series thereof, unless a vote of any such holders is required pursuant to any Preferred Stock Designation.

B. Common Stock. Except as otherwise required by law or as otherwise provided in any Preferred Stock Designation, the holders of the Common Stock shall exclusively possess all voting power and each share of Common Stock shall have one vote.

FIFTH: The name and mailing address of the sole incorporator of the Corporation are as follows:

Address Name

Graubard Miller

The Chrysler Building Jeffrey M. Gallant 405 Lexington Avenue

New York, New York 10174

SIXTH: The Board of Directors shall be divided into three classes: Class A, Class B and Class C. The number of directors in each class shall be as nearly equal as possible. At the first election of directors by the incorporator, the incorporator shall elect a Class C director for a term expiring at the Corporation s third Annual Meeting of Stockholders. The Class C director shall then appoint additional Class A, Class B and Class C directors, as necessary. The directors in Class A shall be elected for a term expiring at the first Annual Meeting of Stockholders, the directors in Class B shall be elected for a term expiring at the second Annual Meeting of Stockholders and the directors in Class C shall be elected for a term expiring at the third Annual Meeting of Stockholders. Commencing at the first Annual Meeting of Stockholders, and at each annual meeting thereafter, directors elected to succeed those directors whose terms expire shall be elected for a term of office to expire at the third succeeding annual meeting of stockholders after their election. Except as the GCL may otherwise require, in the interim between annual meetings of stockholders or special meetings of stockholders called for the election of directors and/or the removal of one or more directors and the filling of any vacancy in that connection, newly created directorships and any vacancies in the Board of Directors, including unfilled vacancies resulting from the removal of directors for cause, may be filled by the vote of a majority of the remaining directors then in office, although less than a quorum (as defined in the Corporation s Bylaws), or by the sole remaining director. All directors shall hold office until the expiration of their respective terms of office and until their successors shall have been elected and qualified. A director elected to fill a vacancy resulting from the death, resignation or removal of a director shall serve for the remainder of the full term of the director whose death, resignation or removal shall have created such vacancy and until his successor shall have been elected and qualified.

SEVENTH: The following provisions are inserted for the management of the business and for the conduct of the affairs of the Corporation, and for further definition, limitation and regulation of the powers of the Corporation and of its directors and stockholders:

- A. Election of directors need not be by ballot unless the Bylaws of the Corporation so provide.
- B. The Board of Directors shall have the power, without the assent or vote of the stockholders, to make, alter, amend, change, add to or repeal the Bylaws of the Corporation as provided in the Bylaws of the Corporation.
- C. The directors in their discretion may submit any contract or act for approval or ratification at any annual meeting of the stockholders or at any meeting of the stockholders called for the purpose of considering any such act or contract, and any contract or act that shall be approved or be ratified by the vote of the holders of a majority of the stock of the Corporation which is represented in person or by proxy at such meeting and entitled to vote thereat (provided that a lawful quorum of stockholders be there represented in person or by proxy) shall be as valid and binding upon the

Corporation and upon all the stockholders as though it had been approved or ratified by every stockholder of the Corporation, whether or not the contract or act would otherwise be open to legal attack because of directors interests, or for any other reason.

D. In addition to the powers and authorities hereinbefore or by statute expressly conferred upon them, the directors are hereby empowered to exercise all such powers and do all such acts and things as may be exercised or done by the Corporation; subject, nevertheless, to the provisions of the statutes of Delaware, of this

B-2

Certificate of Incorporation, and to any Bylaws from time to time made by the stockholders; provided, however, that no Bylaw so made shall invalidate any prior act of the directors which would have been valid if such Bylaw had not been made.

EIGHTH: A. A director of the Corporation shall not be personally liable to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, except for liability (i) for any breach of the director s duty of loyalty to the Corporation or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) under Section 174 of the GCL, or (iv) for any transaction from which the director derived an improper personal benefit. If the GCL is amended to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of a director of the Corporation shall be eliminated or limited to the fullest extent permitted by the GCL, as so amended. Any repeal or modification of this paragraph A by the stockholders of the Corporation shall not adversely affect any right or protection of a director of the Corporation with respect to events occurring prior to the time of such repeal or modification.

B. The Corporation, to the full extent permitted by Section 145 of the GCL, as amended from time to time, shall indemnify all persons whom it may indemnify pursuant thereto. Expenses (including attorneys fees) incurred by an officer or director in defending any civil, criminal, administrative, or investigative action, suit or proceeding for which such officer or director may be entitled to indemnification hereunder shall be paid by the Corporation in advance of the final disposition of such action, suit or proceeding upon receipt of an undertaking by or on behalf of such director or officer to repay such amount if it shall ultimately be determined that he is not entitled to be indemnified by the Corporation as authorized hereby.

NINTH: Whenever a compromise or arrangement is proposed between this Corporation and its creditors or any class of them and/or between this Corporation and its stockholders or any class of them, any court of equitable jurisdiction within the State of Delaware may, on the application in a summary way of this Corporation or of any creditor or stockholder thereof or on the application of any receiver or receivers appointed for this Corporation under Section 291 of Title 8 of the Delaware Code or on the application of trustees in dissolution or of any receiver or receivers appointed for this Corporation under Section 279 of Title 8 of the Delaware Code order a meeting of the creditors or class of creditors, and/or of the stockholders or class of stockholders of this Corporation, as the case may be, to be summoned in such manner as the said court directs. If a majority in number representing three fourths in value of the creditors or class of creditors, and/or of the stockholders or class of stockholders of this Corporation, as the case may be, agree to any compromise or arrangement and to any reorganization of this Corporation as a consequence of such compromise or arrangement, the said compromise or arrangement and the said reorganization shall, if sanctioned by the court to which the said application has been made, be binding on all the creditors or class of creditors, and/or on all the stockholders or class of stockholders, of this Corporation, as the case may be, and also on this Corporation.

IN WITNESS WHEREOF, the Corporation has caused this Amended and Restated Certificate of Incorporation to be signed by Eric J. Watson, its Chief Executive Officer, as of the day of , 2009.

Eric J. Watson, Chief Executive Officer

Annex C

Adopted as of August 27, 2009

# BY LAWS

# OF

# CULLEN AGRICULTURAL HOLDING CORP.

# ARTICLE I OFFICES

- 1.1 *Registered Office*. The registered office of Cullen Agricultural Holding Corp. (the Corporation ) in the State of Delaware shall be established and maintained at 615 S. DuPont Highway, Kent County, Dover, Delaware and National Corporate Research, Ltd. shall be the registered agent of the corporation in charge thereof.
- 1.2 Other Offices. The Corporation may also have offices at such other places both within and without the State of Delaware as the board of directors of the Corporation (the Board of Directors ) may from time to time determine or the business of the Corporation may require.

# ARTICLE II MEETINGS OF STOCKHOLDERS

- 2.1 *Place of Meetings*. All meetings of the stockholders shall be held at such time and place, either within or without the State of Delaware, as shall be designated from time to time by the Board of Directors and stated in the notice of the meeting or in a duly executed waiver of notice thereof.
- 2.2 Annual Meetings. The annual meeting of stockholders shall be held on such date and at such time as may be fixed by the Board of Directors and stated in the notice of the meeting, for the purpose of electing directors and for the transaction of only such other business as is properly brought before the meeting in accordance with these Bylaws (the Bylaws).
- Written notice of an annual meeting stating the place, date and hour of the meeting, shall be given to each stockholder entitled to vote at such meeting not less than ten (10) nor more than sixty (60) days before the date of the annual meeting.
- To be properly brought before the annual meeting, business must be either (i) specified in the notice of annual meeting (or any supplement or amendment thereto) given by or at the direction of the Board of Directors, (ii) otherwise brought before the annual meeting by or at the direction of the Board of Directors, or (iii) otherwise properly brought

before the annual meeting by a stockholder. In addition to any other applicable requirements, for business to be properly brought before an annual meeting by a stockholder, the stockholder must have given timely notice thereof in writing to the Secretary of the Corporation. To be timely, a stockholder s notice must be delivered to or mailed and received at the principal executive offices of the Corporation not less than ninety (90) days nor more than one hundred and twenty (120) days prior to the meeting; provided, however, that in the event that less than ninety (90) days notice or prior public disclosure of the date of the annual meeting is given or made to stockholders, notice by a stockholder, to be timely, must be received no later than the close of business on the tenth (10th) day following the day on which such notice of the date of the annual meeting was mailed or such public disclosure was made, whichever first occurs. A stockholder s notice to the Secretary shall set forth (a) as to each matter the stockholder proposes to bring before the annual meeting (i) a brief description of the business desired to be brought before the annual meeting and the reasons for conducting such business at the annual meeting, and (ii) any material interest of the stockholder in such business, and (b) as to the stockholder giving the notice (i) the name and record address of the stockholder and (ii) the class, series and number of shares of capital stock of the Corporation which are beneficially owned by the stockholder. Notwithstanding anything in these Bylaws to the contrary, no business shall be conducted at the annual meeting except in accordance with the procedures set forth in this Article II, Section 2. The officer of the Corporation presiding at an annual meeting shall, if the facts warrant, determine and declare to the annual meeting that business was not properly brought before the annual meeting

C-1

in accordance with the provisions of this Article II, Section 2, and if such officer should so determine, such officer shall so declare to the annual meeting and any such business not properly brought before the meeting shall not be transacted.

2.3 Special Meetings. Special meetings of the stockholders, for any purpose or purposes, unless otherwise prescribed by statute or by the Certificate of Incorporation of the Corporation (the Certificate of Incorporation ), may only be called by a majority of the entire Board of Directors, or the President or the Chairman, and shall be called by the Secretary at the request in writing of stockholders owning a majority in amount of the entire capital stock of the corporation issued and outstanding and entitled to vote. Such request shall state the purpose or purposes of the proposed meeting.

Unless otherwise provided by law, written notice of a special meeting of stockholders, stating the time, place and purpose or purposes thereof, shall be given to each stockholder entitled to vote at such meeting, not less than ten (10) or more than sixty (60) days before the date fixed for the meeting. Business transacted at any special meeting of stockholders shall be limited to the purposes stated in the notice.

2.4 *Quorum*. The holders of a majority of the capital stock issued and outstanding and entitled to vote thereat, present in person or represented by proxy, shall constitute a quorum at all meetings of the stockholders for the transaction of business except as otherwise provided by statute or by the Certificate of Incorporation. If, however, such quorum shall not be present or represented at any meeting of the stockholders, the holders of a majority of the votes entitled to be cast by the stockholders entitled to vote thereat, present in person or represented by proxy, shall have power to adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum shall be present or represented. At such adjourned meeting at which a quorum shall be present or represented, any business may be transacted which might have been transacted at the meeting as originally noticed. If the adjournment is for more than thirty (30) days, or if after the adjournment a new record date is fixed for the adjourned meeting, a notice of the adjourned meeting shall be given to each stockholder entitled to vote at the meeting.

2.5 *Organization*. The Chairman of the Board of Directors shall act as chairman of meetings of the stockholders. The Board of Directors may designate any other officer or director of the Corporation to act as chairman of any meeting in the absence of the Chairman of the Board of Directors, and the Board of Directors may further provide for determining who shall act as chairman of any stockholders meeting in the absence of the Chairman of the Board of Directors and such designee.

The Secretary of the Corporation shall act as secretary of all meetings of the stockholders, but in the absence of the Secretary the presiding officer may appoint any other person to act as secretary of any meeting.

2.6 *Voting*. Unless otherwise required by law, the Certificate of Incorporation or these Bylaws, any question (other than the election of directors) brought before any meeting of stockholders shall be decided by the vote of the holders of a majority of the stock represented and entitled to vote thereat. At all meetings of stockholders for the election of directors, a plurality of the votes cast shall be sufficient to elect. Each stockholder represented at a meeting of stockholders shall be entitled to cast one vote for each share of the capital stock entitled to vote thereat held by such stockholder, unless otherwise provided by the Certificate of Incorporation. Each stockholder entitled to vote at a meeting of stockholders or to express consent or dissent to corporate action in writing without a meeting may authorize any person or persons to act for him by proxy. All proxies shall be executed in writing and shall be filed with the Secretary of the Corporation not later than the day on which exercised. No proxy shall be voted or acted upon after three (3) years from its date, unless the proxy provides for a longer period. The Board of Directors, in its discretion, or the officer of the Corporation presiding at a meeting of stockholders, in his discretion, may require that any votes cast at such meeting shall be cast by written ballot.

2.7 Action of Shareholders Without Meeting. Unless otherwise provided by the Certificate of Incorporation, any action required to be taken at any annual or special meeting of stockholders, or any action which may be taken at any annual or special meeting of such stockholders, may be taken without a meeting, without prior notice and without a vote, if a consent in writing, setting forth the action so taken, shall be signed by the holders of outstanding stock having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted,

C-2

and shall be delivered to the Corporation by delivery to its registered office in the State of Delaware, its principal place of business, or an officer or agent of the Corporation having custody of the book in which proceedings of meetings of stockholders are recorded. Delivery made to the Corporation s registered office shall be by hand or by certified or registered mail, return receipt requested. Prompt notice of the taking of the corporate action without a meeting by less than unanimous written consent shall be given to those stockholders who have not consented in writing.

- 2.8 *Voting List*. The officer who has charge of the stock ledger of the corporation shall prepare and make, at least ten (10) days before every meeting of stockholders, a complete list of the stockholders entitled to vote at the meeting, arranged in alphabetical order, showing the address of each stockholder and the number of shares registered in the name of each stockholder. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting, during ordinary business hours, for a period of at least ten (10) days prior to the election, either at a place within the city, town or village where the election is to be held, which place shall be specified in the notice of the meeting, or, if not specified, at the place where said meeting is to be held. The list shall be produced and kept at the time and place of election during the whole time thereof, and may be inspected by any stockholder of the Corporation who is present.
- 2.9 *Stock Ledger*. The stock ledger of the Corporation shall be the only evidence as to who are the stockholders entitled to examine the stock ledger, the list required by Section 8 of this Article II or the books of the Corporation, or to vote in person or by proxy at any meeting of stockholders.
  - 2.10 *Adjournment*. Any meeting of the stockholders, including one at which directors are to be elected, may be adjourned for such periods as the presiding officer of the meeting or the stockholders present in person or by proxy and entitled to vote shall direct.
- 2.11 *Ratification*. Any transaction questioned in any stockholders—derivative suit, or any other suit to enforce alleged rights of the Corporation or any of its stockholders, on the ground of lack of authority, defective or irregular execution, adverse interest of any director, officer or stockholder, nondisclosure, miscomputation or the application of improper principles or practices of accounting may be approved, ratified and confirmed before or after judgment by the Board of Directors or by the holders of Common Stock and, if so approved, ratified or confirmed, shall have the same force and effect as if the questioned transaction had been originally duly authorized, and said approval, ratification or confirmation shall be binding upon the Corporation and all of its stockholders and shall constitute a bar to any claim or execution of any judgment in respect of such questioned transaction.
- 2.12 *Inspectors*. The election of directors and any other vote by ballot at any meeting of the stockholders shall be supervised by at least one inspector. Such inspectors shall be appointed by the Board of Directors in advance of the meeting. If the inspector so appointed shall refuse to serve or shall not be present, such appointment shall be made by the officer presiding at the meeting.

# ARTICLE III DIRECTORS

3.1 *Powers; Number; Qualifications.* The business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors, except as may be otherwise provided by law or in the Certificate of Incorporation. The number of directors which shall constitute the Board of Directors shall be not less than one (1) nor more than nine (9). The exact number of directors shall be fixed from time to time, within the limits specified in this Article III

ARTICLE III DIRECTORS 432

Section 1 or in the Certificate of Incorporation, by the Board of Directors. Directors need not be stockholders of the Corporation. The Board may be divided into Classes as more fully described in the Certificate of Incorporation.

3.2 Election; Term of Office; Resignation; Removal; Vacancies. Each director shall hold office until the next annual meeting of stockholders at which his Class stands for election or until such director s earlier resignation, removal from office, death or incapacity. Unless otherwise provided in the Certificate of Incorporation, vacancies and newly created directorships resulting from any increase in the authorized number of directors or from any other cause may be filled by a majority of the directors then in office, although less than a quorum, or by a sole remaining director and each director so chosen shall hold office until the next annual

meeting and until such director s successor shall be duly elected and shall qualify, or until such director s earlier resignation, removal from office, death or incapacity.

- 3.3 Nominations. Nominations of persons for election to the Board of Directors of the Corporation at a meeting of stockholders of the Corporation may be made at such meeting by or at the direction of the Board of Directors, by any committee or persons appointed by the Board of Directors or by any stockholder of the Corporation entitled to vote for the election of directors at the meeting who complies with the notice procedures set forth in this Article III, Section 3. Such nominations by any stockholder shall be made pursuant to timely notice in writing to the Secretary of the Corporation. To be timely, a stockholder s notice shall be delivered to or mailed and received at the principal executive offices of the Corporation not less than sixty (60) days nor more than ninety (90) days prior to the meeting; provided however, that in the event that less than seventy (70) days notice or prior public disclosure of the date of the meeting is given or made to stockholders, notice by the stockholder, to be timely, must be received no later than the close of business on the tenth (10th) day following the day on which such notice of the date of the meeting was mailed or such public disclosure was made, whichever first occurs, Such stockholder s notice to the Secretary shall set forth (i) as to each person whom the stockholder proposes to nominate for election or reelection as a director, (a) the name, age, business address and residence address of the person, (b) the principal occupation or employment of the person, (c) the class and number of shares of capital stock of the Corporation which are beneficially owned by the person, and (d) any other information relating to the person that is required to be disclosed in solicitations for proxies for election of directors pursuant to the Rules and Regulations of the Securities and Exchange Commission under Section 14 of the Securities Exchange Act of 1934, as amended, and (ii) as to the stockholder giving the notice (a) the name and record address of the stockholder and (b) the class and number of shares of capital stock of the Corporation which are beneficially owned by the stockholder. The Corporation may require any proposed nominee to furnish such other information as may reasonably be required by the Corporation to determine the eligibility of such proposed nominee to serve as a director of the Corporation. No person shall be eligible for election as a director of the Corporation unless nominated in accordance with the procedures set forth herein. The officer of the Corporation presiding at an annual meeting shall, if the facts warrant, determine and declare to the meeting that a nomination was not made in accordance with the foregoing procedure, and if he should so determine, he shall so declare to the meeting and the defective nomination shall be disregarded.
- 3.4 *Meetings*. The Board of Directors of the Corporation may hold meetings, both regular and special, either within or without the State of Delaware. The first meeting of each newly elected Board of Directors shall be held immediately after and at the same place as the meeting of the stockholders at which it is elected and no notice of such meeting shall be necessary to the newly elected directors in order to legally constitute the meeting, provided a quorum shall be present. Regular meetings of the Board of Directors may be held without notice at such time and place as shall from time to time be determined by the Board of Directors. Special meetings of the Board of Directors may be called by the President or a majority of the entire Board of Directors. Notice thereof stating the place, date and hour of the meeting shall be given to each director either by mail not less than forty-eight (48) hours before the date of the meeting, by telephone, facsimile, telegram or e-mail on twenty-four (24) hours notice, or on such shorter notice as the person or persons calling such meeting may deem necessary or appropriate in the circumstances.
- 3.5 *Quorum*. Except as may be otherwise specifically provided by law, the Certificate of Incorporation or these Bylaws, at all meetings of the Board of Directors or any committee thereof, a majority of the entire Board of Directors or such committee, as the case may be, shall constitute a quorum for the transaction of business and the act of a majority of the directors present at any meeting at which there is a quorum shall be the act of the Board of Directors. If a quorum shall not be present at any meeting of the Board of Directors or of any committee thereof, a majority of the directors present thereat may adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum shall be present.

3.6 Organization of Meetings. The Board of Directors shall elect one of its members to be Chairman of the Board of Directors. The Chairman of the Board of Directors shall lead the Board of Directors in fulfilling its responsibilities as set forth in these By-Laws, including its responsibility to oversee the performance of the Corporation, and shall determine the agenda and perform all other duties and exercise all other powers which are or from time to time may be delegated to him or her by the Board of Directors.

# **TABLE OF CONTENTS**

Meetings of the Board of Directors shall be presided over by the Chairman of the Board of Directors, or in his or her absence, by the President, or in the absence of the Chairman of the Board of Directors and the President by such other person as the Board of Directors may designate or the members present may select.

- 3.7 Actions of Board of Directors Without Meeting. Unless otherwise restricted by the Certificate of Incorporation or these Bylaws, any action required or permitted to be taken at any meeting of the Board of Directors or of any committee thereof may be taken without a meeting, if all members of the Board of Directors or of such committee, as the case may be, consent thereto in writing, and the writing or writings are filled with the minutes of proceedings of the Board of Directors or committee.
- 3.8 *Removal of Directors by Stockholders*. The entire Board of Directors or any individual Director may be removed from office with or without cause by a majority vote of the holders of the outstanding shares then entitled to vote at an election of directors. In case the Board of Directors or any one or more Directors be so removed, new Directors may be elected at the same time for the unexpired portion of the full term of the Director or Directors so removed.
- 3.9 *Resignations*. Any Director may resign at any time by submitting his written resignation to the Board of Directors or Secretary of the Corporation. Such resignation shall take effect at the time of its receipt by the Corporation unless another time be fixed in the resignation, in which case it shall become effective at the time so fixed. The acceptance of a resignation shall not be required to make it effective.
- 3.10 Committees. The Board of Directors may designate one or more committees, each committee to consist of one or more of the directors of the Corporation. In the absence or disqualification of a member of a committee, the member or members thereof present at any meeting and not disqualified from voting, whether or not he or they constitute a quorum, may unanimously appoint another member of the Board of Directors to act at the meeting in the place of any such absent or disqualified member. Any such committee, to the extent provided by law and in the resolution of the Board of Directors establishing such committee, shall have and may exercise all the powers and authority of the Board of Directors in the management of the business and affairs of the Corporation, and may authorize the seal of the Corporation to be affixed to all papers which may require it; but no such committee shall have the power or authority in reference to amending the Certificate of Incorporation, adopting an agreement of merger or consolidation, recommending to the stockholders the sale, lease or exchange of all or substantially all of the Corporation s property and assets, recommending to the stockholders a dissolution of the Corporation or a revocation of a dissolution or amending the Bylaws of the Corporation; and, unless the resolution expressly so provides, no such committee shall have the power or authority to declare a dividend or to authorize the issuance of stock or to adopt a certificate of ownership and merger. Each committee shall keep regular minutes of its meetings and report the same to the Board of Directors when required.
  - 3.11 *Compensation*. The directors may be paid their expenses, if any, of attendance at each meeting of the Board of Directors and may be paid a fixed amount (in cash or other form of consideration) for attendance at each meeting of the Board of Directors or a stated salary as director. No such payment shall preclude any director from serving the Corporation in any other capacity and receiving compensation therefor. Members of special or standing committees may be allowed like compensation for attending committee meetings.
- 3.12 Interested Directors. No contract or transaction between the Corporation and one or more of its directors or officers, or between the Corporation and any other corporation, partnership, association, or other organization in which one or more of its directors or officers are directors or officers, or have a financial interest, shall be void or voidable solely for this reason, or solely because the director or officer is present at or participates in the meeting of the Board of Directors or committee thereof which authorizes the contract or transaction, or solely because his or their votes are counted for such purpose, if (i) the material facts as to his or their relationship or interest and as to the

contract or transaction are disclosed or are known to the Board of Directors or the committee, and the Board of Directors or committee in good faith authorizes the contract or transaction by the affirmative votes of a majority of the disinterested directors, even though the disinterested directors be less than a quorum; or (ii) the material facts as to his or their relationship or interest and as to the contract or transaction are disclosed or are known to the stockholders entitled to vote thereon, and the contract or transaction is specifically approved in good faith by vote of the stockholders; or (iii) the contract or transaction is fair as to the Corporation as of the time it is authorized, approved or ratified, by the Board of

Directors, a committee thereof or the stockholders. Common or interested directors may be counted in determining the presence of a quorum at a meeting of the Board of Directors or of a committee which authorizes the contract or transaction.

3.13 *Meetings by Means of Conference Telephone*. Members of the Board of Directors or any committee designed by the Board of Directors may participate in a meeting of the Board of Directors or of a committee of the Board of Directors by means of conference telephone or similar communications equipment by means of which all persons participating in the meeting can hear each other, and participation in a meeting pursuant to this subsection shall constitute presence in person at such meeting.

# ARTICLE IV OFFICERS

- 4.1 *General*. The officers of the Corporation shall be elected by the Board of Directors and may consist of: a Chairman of the Board, Vice Chairman of the Board, President, Chief Financial Officer, Secretary and Treasurer. The Board of Directors, in its discretion, may also elect one or more Vice Presidents (including Executive Vice Presidents and Senior Vice Presidents), Assistant Secretaries, Assistant Treasurers, a Controller and such other officers as in the judgment of the Board of Directors may be necessary or desirable. Any number of offices may be held by the same person and more than one person may hold the same office, unless otherwise prohibited by law, the Certificate of Incorporation or these Bylaws. The officers of the Corporation need not be stockholders of the Corporation, nor need such officers be directors of the Corporation.
- 4.2 *Election*. The Board of Directors at its first meeting held after each annual meeting of stockholders shall elect the officers of the Corporation who shall hold their offices for such terms and shall exercise such powers and perform such duties as shall be determined from time to time by the Board of Directors; and all officers of the Corporation shall hold office until their successors are chosen and qualified, or until their earlier resignation or removal. Except as otherwise provided in this Article IV, any officer elected by the Board of Directors may be removed at any time by the affirmative vote of a majority of the Board of Directors. Any vacancy occurring in any office of the Corporation shall be filled by the Board of Directors. The salaries of all officers who are directors of the Corporation shall be fixed by the Board of Directors.
- 4.3 *Voting Securities Owned by the Corporation*. Powers of attorney, proxies, waivers of notice of meeting, consents and other instruments relating to securities owned by the Corporation may be executed in the name of and on behalf of the Corporation by the President or any Vice President, and any such officer may, in the name and on behalf of the Corporation, take all such action as any such officer may deem advisable to vote in person or by proxy at any meeting of security holders of any corporation in which the Corporation may own securities and at any such meeting shall possess and may exercise any and all rights and powers incident to the ownership of such securities and which, as the owner thereof, the Corporation might have exercised and possessed if present. The Board of Directors may, by resolution, from time to time confer like powers upon any other person or persons.
- 4.4 *President*. Subject to the provisions of these Bylaws and to the direction of the Board of Directors, the President shall have ultimate authority for decisions relating to the general management and control of the affairs and business of the Corporation and shall perform such other duties and exercise such other powers which are or from time to time may be delegated to him or her by the Board of Directors or these Bylaws, all in accordance with basic policies as established by and subject to the oversight of the Board of Directors.

4.5 *Chief Financial Officer*. The Chief Financial Officer shall have general supervision, direction and control of the financial affairs of the Corporation and shall perform such other duties and exercise such other powers which are or from time to time may be delegated to him or her by the Board of Directors or these Bylaws, all in accordance with basic policies as established by and subject to the oversight of the Board of Directors. In the absence of a named Treasurer, the Chief Financial Officer shall also have the powers and duties of the Treasurer as hereinafter set forth and shall be authorized and empowered to sign as Treasurer in any case where such officer s signature is required.

4.6 *Vice Presidents*. At the request of the President or in the absence of the President, or in the event of his or her inability or refusal to act, the Vice President or the Vice Presidents if there is more than one (in the order designated by the Board of Directors) shall perform the duties of the President, and when so acting,

C-6

#### TABLE OF CONTENTS

shall have all the powers of and be subject to all the restrictions upon such office. Each Vice President shall perform such other duties and have such other powers as the Board of Directors from time to time may prescribe. If there be no Vice President, the Board of Directors shall designate the officer of the Corporation who, in the absence of the President or in the event of the inability or refusal of such officer to act, shall perform the duties of such office, and when so acting, shall have all the powers of and be subject to all the restrictions upon such office.

4.7 Secretary. The Secretary shall attend all meetings of the Board of Directors and all meetings of stockholders and record all the proceedings thereat in a book or books to be kept for that purpose; the Secretary shall also perform like duties for the standing committees when required. The Secretary shall give, or cause to be given, notice of all meetings of the stockholders and special meetings of the Board of Directors, and shall perform such other duties as may be prescribed by the Board of Directors or the President, under whose supervision the Secretary shall be. If the Secretary shall be unable or shall refuse to cause to be given notice of all meetings of the stockholders and special meetings of the Board of Directors, then any Assistant Secretary shall perform such actions. If there be no Assistant Secretary, then the Board of Directors or the President may choose another officer to cause such notice to be given. The Secretary shall have custody of the seal of the Corporation and the Secretary or any Assistant Secretary, if there be one, shall have authority to affix the same to any instrument requiring it and when so affixed, it may be attested by the signature of the Secretary or by the signature of any such Assistant Secretary. The Board of Directors may give general authority to any other officer to affix the seal of the Corporation and to attest the affixing by his signature. The Secretary shall see that all books, reports, statements, certificates and other documents and records required by law to be kept or filed are properly kept or filed, as the case may be.

4.8 *Treasurer*. The Treasurer shall have the custody of the corporate funds and securities and shall keep full and accurate accounts of receipts and disbursements in books belonging to the Corporation and shall deposit all moneys and other valuable effects in the name and to the credit of the Corporation in such depositories as may be designated by the Board of Directors. The Treasurer shall disburse the funds of the Corporation as may be ordered by the Board of Directors, taking proper vouchers for such disbursements, and shall render to the President and the Board of Directors, at its regular meetings, or when the Board of Directors so requires, an account of all his transactions as Treasurer and of the financial condition of the Corporation. If required by the Board of Directors, the Treasurer shall give the Corporation a bond in such sum and with such surety or sureties as shall be satisfactory to the Board of Directors for the faithful performance of the duties of his office and for the restoration to the Corporation, in case of his death, resignation, retirement or removal from office, of all books, papers, vouchers, money and other property of whatever kind in his possession or under his control belonging to the Corporation.

4.9 Assistant Secretaries. Except as may be otherwise provided in these Bylaws, Assistant Secretaries, if there be any, shall perform such duties and have such powers as from time to time may be assigned to them by the Board of Directors, the President, any Vice President, if there be one, or the Secretary, and in the absence of the Secretary or in the event of his disability or refusal to act, shall perform the duties of the Secretary, and when so acting, shall have all the powers of and be subject to all the restrictions upon the Secretary.

4.10 *Assistant Treasurers*. Assistant Treasurers, if there be any, shall perform such duties and have such powers as from time to time may be assigned to them by the Board of Directors, the President, any Vice President, if there be one, or the Treasurer, and in the absence of the Treasurer or in the event of his disability or refusal to act, shall perform the duties of the Treasurer, and when so acting, shall have all the powers of and be subject to all the restrictions upon the Treasurer. If required by the Board of Directors, an Assistant Treasurer shall give the Corporation a bond in such sum and with such surety or sureties as shall be satisfactory to the Board of Directors for the faithful performance of the duties of his office and for the restoration to the Corporation, in case of his death, resignation, retirement or removal from office, of all books, papers, vouchers, money and other property of whatever kind in his possession or under his control belonging to the Corporation.

4.11 *Controller*. The Controller shall establish and maintain the accounting records of the Corporation in accordance with generally accepted accounting principles applied on a consistent basis, maintain proper

C-7

internal control of the assets of the Corporation and shall perform such other duties as the Board of Directors, the President or any Vice President of the Corporation may prescribe.

- 4.12 *Other Officers*. Such other officers as the Board of Directors may choose shall perform such duties and have such powers as from time to time may be assigned to them by the Board of Directors. The Board of Directors may delegate to any other officer of the Corporation the power to choose such other officers and to prescribe their respective duties and powers.
  - 4.13 *Vacancies*. The Board of Directors shall have the power to fill any vacancies in any office occurring from whatever reason.
- 4.14 *Resignations*. Any officer may resign at any time by submitting his written resignation to the Corporation. Such resignation shall take effect at the time of its receipt by the Corporation, unless another time be fixed in the resignation, in which case it shall become effective at the time so fixed. The acceptance of a resignation shall not be required to make it effective.
  - 4.15 *Removal*. Subject to the provisions of any employment agreement approved by the Board of Directors, any officer of the Corporation may be removed at any time, with or without cause, by the Board of Directors.

# ARTICLE V CAPITAL STOCK

- 5.1 Form of Certificates. The shares of stock in the Corporation shall be represented by certificates, provided that the Board of Directors may provide by resolution or resolutions that some or all of any or all classes or series of the Corporation s stock shall be in uncertificated form. Stock certificates shall be in such forms as the Board of Directors may prescribe and signed by the Chairman of the Board, President or a Vice President and by the Treasurer or an Assistant Treasurer, or the Secretary or an Assistant Secretary of the Corporation.
- 5.2 Signatures. Any or all of the signatures on a stock certificate may be a facsimile, including, but not limited to, signatures of officers of the Corporation and countersignatures of a transfer agent or registrar. In case an officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such officer, transfer agent or registrar before such certificate is issued, it may be issued by the Corporation with the same effect as if such person were such officer, transfer agent or registrar at the date of issue.
- 5.3 Lost Certificates. The Board of Directors may direct a new stock certificate or certificates to be issued in place of any stock certificate or certificates theretofore issued by the Corporation alleged to have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming the certificate of stock to be lost, stolen or destroyed. When authorizing such issue of a new stock certificate, the Board of Directors may, in its discretion and as a condition precedent to the issuance thereof, require the owner of such lost, stolen or destroyed certificate, or his legal representative, to advertise the same in such manner as the Board of Directors shall require and/or to give the Corporation a bond in such sum as it may direct as indemnity against any claim that may be made against the Corporation with respect to the certificate alleged to have been lost, stolen or destroyed.
- 5.4 *Transfers*. Stock of the Corporation shall be transferable in the manner prescribed by law and in these Bylaws. Transfers of certificated stock shall be made on the books of the Corporation only by the person named in the certificate or by such person s attorney lawfully constituted in writing and upon the surrender of the certificate therefor,

which shall be canceled before a new certificate shall be issued. Transfers of uncertificated stock shall be made on the books of the Corporation only by the person then registered on the books of the Corporation as the owner of such shares or by such person's attorney lawfully constituted in writing and written instruction to the Corporation containing such information as the Corporation or its agents may prescribe. No transfer of uncertificated stock shall be valid as against the Corporation for any purpose until it shall have been entered in the stock records of the Corporation by an entry showing from and to whom transferred. The Corporation shall have no duty to inquire into adverse claims with respect to any stock transfer unless (a) the Corporation has received a written notification of an adverse claim at a time and in a manner which affords the Corporation a reasonable opportunity to act on it prior to the issuance of a new, reissued or

re-registered share certificate, in the case of certificated stock, or entry in the stock record books of the Corporation, in the case of uncertificated stock, and the notification identifies the claimant, the registered owner and the issue of which the share or shares is a part and provides an address for communications directed to the claimant; or (b) the Corporation has required and obtained, with respect to a fiduciary, a copy of a will, trust, indenture, articles of co-partnership, Bylaws or other controlling instruments, for a purpose other than to obtain appropriate evidence of the appointment or incumbency of the fiduciary, and such documents indicate, upon reasonable inspection, the existence of an adverse claim. The Corporation may discharge any duty of inquiry by any reasonable means, including notifying an adverse claimant by registered or certified mail at the address furnished by him or, if there be no such address, at his residence or regular place of business that the security has been presented for registration of transfer by a named person, and that the transfer will be registered unless within thirty days from the date of mailing the notification, either (a) an appropriate restraining order, injunction or other process issues from a court of competent jurisdiction; or (b) an indemnity bond, sufficient in the Corporation s judgment to protect the Corporation and any transfer agent, registrar or other agent of the Corporation involved from any loss which it or they may suffer by complying with the adverse claim, is filed with the Corporation.

- 5.5 Fixing Record Date. In order that the Corporation may determine the stockholders entitled to notice or to vote at any meeting of stockholders or any adjournment thereof, or to express consent to corporate action in writing without a meeting, or entitled to receive payment of any dividend or other distribution or allotment of any rights, or entitled to exercise any rights in respect of any change, conversion or exchange of stock or for the purpose of any other lawful action, the Board of Directors may fix a record date, which record date shall not precede the date upon which the resolution fixing the record is adopted by the Board of Directors, and which record date shall not be more than sixty (60) nor less than ten (10) days before the date of such meeting, nor more than ten (10) days after the date upon which the resolution fixing the record date of action with a meeting is adopted by the Board of Directors, nor more than sixty (60) days prior to any other action. If no record date is fixed:
- (a) The record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which the meeting is held.
- (b) The record date for determining stockholders entitled to express consent to corporate action in writing without a meeting, when no prior action by the Board of Directors is necessary, shall be the first date on which a signed written consent is delivered to the Corporation.
- (c) The record date for determining stockholders for any other purpose shall be at the close of business on the day on which the Board of Directors adopts the resolution relating thereto.

A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board of Directors may fix a new record date for the adjourned meeting.

5.6 Registered Stockholders. Prior to due presentment for transfer of any share or shares, the Corporation shall treat the registered owner thereof as the person exclusively entitled to vote, to receive notifications and to all other benefits of ownership with respect to such share or shares, and shall not be bound to recognize any equitable or other claim to or interest in such share or shares on the part of any other person, whether or not it shall have express or other notice thereof, except as otherwise provided by the laws of the State Delaware.

# ARTICLE VI

6.1 *Form of Notice*. Notices to directors and stockholders other than notices to directors of special meetings of the board of Directors which may be given by any means stated in Article III, Section 4, shall be in writing and delivered personally or mailed to the directors or stockholders at their addresses appearing on the books of the corporation. Notice by mail shall be deemed to be given at the time when the same shall be mailed. Notice to directors may also be given by telegram.

C-9

ARTICLE VI NOTICES 445

6.2 Waiver of Notice. Whenever any notice is required to be given under the provisions of law or the Certificate of Incorporation or by these Bylaws of the Corporation, a written waiver, signed by the person or persons entitled to notice, whether before or after the time stated therein, shall be deemed equivalent to notice. Attendance of a person at a meeting shall constitute a waiver of notice of such meeting, except when the person attends a meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened. Neither the business to be transacted at, nor the purpose of, any regular, or special meeting of the stockholders, Directors, or members of a committee of Directors need be specified in any written waiver of notice unless so required by the Certificate of Incorporation.

# ARTICLE VII INDEMNIFICATION OF DIRECTORS AND OFFICERS

7.1 The Corporation shall indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (other than an action by or in the right of the Corporation) by reason of the fact that he is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise, against expenses (including attorneys fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by him in connection with such action, suit or proceeding if he acted in good faith and in a manner he reasonably believed to be in or not opposed to the best interests of the Corporation, and, with respect to any criminal action or proceeding, had no reasonable cause to believe his conduct was unlawful. The termination of any action, suit or proceeding by judgment, order, settlement, conviction, or upon a plea of nolo contendere or its equivalent, shall not, of itself, create a presumption that the person did not act in good faith and in a manner which he reasonably believed to be in or not opposed to the best interests of the Corporation, and, with respect to any criminal action or proceeding, had reasonable cause to believe that his conduct was unlawful.

7.2 The Corporation shall indemnify any person who was or is a party, or is threatened to be made a party to any threatened, pending or completed action or suit by or in the right of the Corporation to procure a judgment in its favor by reason of the fact that he is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another Corporation, partnership, joint venture, trust or other enterprise against expenses (including attorneys fees) actually and reasonably incurred by him in connection with the defense or settlement of such action or suit if he acted in good faith and in a manner he reasonably believed to be in or not opposed to the best interests of the Corporation and except that no indemnification shall be made in respect of any claim, issue or matter as to which such person shall have been adjudged to be liable to the Corporation unless and only to the extent that the Court of Chancery or the court in which such action or suit was brought shall determine upon application that, despite the adjudication of liability but in view of all the circumstances of the case, such person is fairly and reasonably entitled to indemnity for such expenses which the Court of Chancery or such other court shall deem proper.

7.3 To the extent that a director, officer, employee or agent of the Corporation has been successful on the merits or otherwise in defense of any action, suit or proceeding referred to in Sections 1 or 2 of this Article, or in defense of any claim, issue or matter therein, he shall be indemnified against expenses (including attorneys fees) actually and reasonably incurred by him or her in connection therewith.

7.4 Any indemnification under sections 1 or 2 of this Article (unless ordered by a court) shall be made by the Corporation only as authorized in the specific case upon a determination that indemnification of the director, officer,

employee or agent is proper in the circumstances because he has met the applicable standard of conduct set forth in such section. Such determination shall be made:

- (a) By the Board of Directors by a majority vote of a quorum consisting of directors who were not parties to such action, suit or proceeding, or
  - (b) If such a quorum is not obtainable, or, even if obtainable a quorum of disinterested directors so directs, by independent legal counsel in a written opinion, or

# **TABLE OF CONTENTS**

- (c) By the stockholders.
- 7.5 Expenses (including attorneys fees) incurred by an officer or director in defending any civil, criminal, administrative or investigative action, suit or proceeding may be paid by the Corporation in advance of the final disposition of such action, suit or proceeding upon receipt of an undertaking by or on behalf of such director or officer to repay such amount if it shall ultimately be determined that he is not entitled to be indemnified by the Corporation as authorized in this Section. Such expenses (including attorneys fees) incurred by other employees and agents may be so paid upon such terms and conditions, if any, as the Board of Directors deems appropriate.
  - 7.6 The indemnification and advancement of expenses provided by, or granted pursuant to the other sections of this Article shall not be deemed exclusive of any other rights to which those seeking indemnification or advancement of expenses may be entitled under any bylaw, agreement, vote of stockholders or disinterested directors or otherwise, both as to action in his official capacity and as to action in another capacity while holding such office.
- 7.7 The Corporation shall have power to purchase and maintain insurance on behalf of any person who is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another Corporation, partnership, joint venture, trust or other enterprise against any liability asserted against him and incurred by him in any such capacity, or arising out of his status as such, whether or not the Corporation would have the power to indemnify him against such liability under the provisions of this Article.
- 7.8 For purposes of this Article, references to the Corporation shall include, in addition to the resulting Corporation, any constituent Corporation (including any constituent of a constituent) absorbed in a consolidation or merger which, if its separate existence had continued, would have had power and authority to indemnify its directors, officers, and employees or agents, so that any person who is or was a director, officer employee or agent of such constituent Corporation, or is or was serving at the request of such constituent Corporation as a director, officer, employee or agent of another Corporation, partnership, joint venture, trust or other enterprise, shall stand in the same position under this Article with respect to the resulting or surviving Corporation as he would have with respect to such constituent Corporation of its separate existence had continued.
- 7.9 For purposes of this Article, references to other enterprises shall include employee benefit plans; references to fines shall include any excise taxes assessed on a person with respect to any employee benefit plan; and references to serving at the request of the Corporation shall include any service as a director, officer, employee or agent of the Corporation which imposes duties on, or involves services by, such director, officer, employee, or agent with respect to an employee benefit plan, its participants or beneficiaries; and a person who acted in good faith and in a manner he reasonably believed to be in the interest of the participants and beneficiaries of an employee benefit plan shall be deemed to have acted in a manner not opposed to the best interests of the Corporation as referred to in this Article.
- 7.10 The indemnification and advancement of expenses provided by, or granted pursuant to, this Article shall, unless otherwise provided when authorized or ratified, continue as to a person who has ceased to be a director, officer, employee or agent and shall inure to the benefit of the heirs, executors and administrators of such a person.
- 7.11 No director or officer of the Corporation shall be personally liable to the Corporation or to any stockholder of the Corporation for monetary damages for breach of fiduciary duty as a director or officer, provided that this provision shall not limit the liability of a director or officer (i) for any breach of the director s or the officer s duty of loyalty to the Corporation or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) under Section 174 of the General Corporation Law of Delaware, or (iv) for any transaction from which the director or officer derived an improper personal benefit.

# ARTICLE VIII GENERAL PROVISIONS

- 8.1 *Reliance on Books and Records*. Each Director, each member of any committee designated by the Board of Directors, and each officer of the Corporation, shall, in the performance of his duties, be fully protected in relying in good faith upon the books of account or other records of the Corporation, including reports made to the Corporation by any of its officers, by an independent certified public accountant, or by an appraiser selected with reasonable care.
- 8.2 *Maintenance and Inspection of Records*. The Corporation shall, either at its principal executive office or at such place or places as designated by the Board of Directors, keep a record of its stockholders listing their names and addresses and the number and class of shares held by each stockholder, a copy of these by-laws, as may be amended to date, minute books, accounting books and other records.

Any such records maintained by the Corporation may be kept on, or by means of, or be in the form of, any information storage device or method, provided that the records so kept can be converted into clearly legible paper form within a reasonable time. The Corporation shall so convert any records so kept upon the request of any person entitled to inspect such records pursuant to the provisions of the Delaware General Corporation Law. When records are kept in such manner, a clearly legible paper form produced from or by means of the information storage device or method shall be admissible in evidence, and accepted for all other purposes, to the same extent as an original paper form accurately portrays the record.

Any stockholder of record, in person or by attorney or other agent, shall, upon written demand under oath stating the purpose thereof, have the right during the usual hours for business to inspect for any proper purpose the Corporation s stock ledger, a list of its stockholders, and its other books and records and to make copies or extracts therefrom. A proper purpose shall mean a purpose reasonably related to such person s interest as a stockholder. In every instance where an attorney or other agent is the person who seeks the right to inspection, the demand under oath shall be accompanied by a power of attorney or such other writing that authorizes the attorney or other agent to so act on behalf of the stockholder. The demand under oath shall be directed to the Corporation at its registered office in Delaware or at its principal executive office.

- 8.3 *Inspection by Directors*. Any director shall have the right to examine the Corporation s stock ledger, a list of its stockholders, and its other books and records for a purpose reasonably related to his or her position as a director.
- 8.4 *Dividends*. Subject to the provisions of the Certificate of Incorporation, if any, dividends upon the capital stock of the Corporation may be declared by the Board of Directors at any regular or special meeting, pursuant to law. Dividends may be paid in cash, in property, or in shares of the capital stock, subject to the provisions of the Certificate of Incorporation. Before payment of any dividend, there may be set aside out of any funds of the Corporation available for dividends such sum or sums as the Directors from time to time, in their absolute discretion, think proper as a reserve or reserves to meet contingencies, or for equalizing dividends, or for repairing or maintaining any property of the Corporation, or for such other purpose as the Directors shall think conducive to the interest of the Corporation, and the Directors may modify or abolish any such reserve in the manner in which it was created.
- 8.5 Annual Statement. The Board of Directors shall present at each annual meeting, and at any special meeting of the stockholders when called for by vote of the stockholders, a full and clear statement of the business and condition of the Corporation.

- 8.6 *Checks*. All checks or demands for money and notes of the Corporation shall be signed by such officer or officers or such other persons as the Board of Directors may from time to time designate.
- 8.7 *Fiscal Year*. The fiscal year of the Corporation shall be as determined by the Board of Directors. If the Board of Directors shall fail to do so, the President shall fix the fiscal year.
- 8.8 *Seal*. The corporate seal shall have inscribed thereon the name of the Corporation, the year of its organization and the words Corporate Seal, Delaware . The seal may be used by causing it or a facsimile thereof to be impressed or affixed or in any manner reproduced.

8.9 Amendments. The original or other Bylaws may be adopted, amended or repealed by the stockholders entitled to vote thereon at any regular or special meeting or, if the Certificate of Incorporation so provides, by the Board of Directors. The fact that such power has been so conferred upon the Board of Directors shall not divest the stockholders of the power nor limit their power to adopt, amend or repeal Bylaws.

8.10 *Interpretation of Bylaws*. All words, terms and provisions of these Bylaws shall be interpreted and defined by and in accordance with the General Corporation Law of the State of Delaware, as amended, and as amended from time to time hereafter.

#### Annex D

# **EMPLOYMENT AGREEMENT**

THIS EMPLOYMENT AGREEMENT ( Agreement ) is made and entered into as of the 3 day of August, 2009, by and between Natural Dairy, Inc. ( NDI or the Company ), and Richard Watson ( Employee ).

# WITNESSETH

**WHEREAS**, Employee desires to be employed by the Company and the Company desires to employ Employee on the terms and conditions hereinafter provided.

**NOW THEREFORE**, for and in consideration of the mutual promises and covenants contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged and accepted, the parties hereto agree as follows:

. POSITION

The Company shall employ Employee in the capacity of Chief Scientific Officer of the Company. In his role as Chief Scientific Officer, Employee will report directly to the Chairman of the Company or as otherwise determined from time to time by the Board of Directors. So long as he is employed hereunder, Employee agrees to devote sufficient and proper time and energy to the business and affairs of the Company, so as to perform his duties hereunder to the best of his ability and at a level of competency consistent with the position occupied, to act on all matters in a manner he reasonably believes to be in and not opposed to the best interests of the Company, to use his best efforts, skill and ability to promote the profitable growth of the Company, and to perform such duties as are customarily performed by one holding a position of comparable seniority at a company comparable to the Company. Employee acknowledges that he has a duty of loyalty to the Company and shall not, either directly or indirectly, engage in any outside business or other activity unrelated to serving the Company s interest, during the term of this Agreement, without the prior approval of the Board of Directors.

2. TERM

The Employee's employment will commence on 1st September, 2009( Commencement Date ) and shall continue until 31st August, 2012 unless terminated earlier in accordance with Section 11 of this Agreement. Thereafter, this Agreement shall be automatically renewed for successive one year periods.

. REMUNERATION

The Employee shall receive the following forms of compensation from the Company in exchange for the services provided by the Employee pursuant to this Agreement.

- 3.1. *Base Salary*. The Company shall pay the Employee an annual base salary of \$100,000.00 ( Base Salary ), less all applicable deductions and withholdings required by law. The Base Salary shall be paid in accordance with the Company s then-current payroll schedule.
- 3.2. *Bonus*. The Employee shall be eligible to receive, subject to the sole discretion of the Board of Directors, a bonus of up to 50% of the Employee s Base Salary. If awarded, such bonus shall be paid on a quarterly basis.

- 3.3. *BlackBerry*. The Company shall provide the Employee with a mobile phone/BlackBerry for business use. Unless otherwise agreed between the Company and Employee, the Company will directly meet the costs of providing the device, maintaining a network connection and the costs arising from its use.
- 3.4. Vacation and Holidays. The Employee shall receive paid time off for any holidays designated by the Company. In addition, the Employee shall be entitled to 20 days of paid vacation in each calendar year, prorated for any partial year of employment. Vacation days must be used in the year in which they accrue, and may not be carried over from year to year without the express permission of the Company. Any unused vacation days shall be forfeited in the event of the termination of the Employee s employment with the Company. The Employee shall consult with the Company regarding mutually convenient times for the Employee to

D-1

take vacation, but failing agreement, the Company may direct when vacation days must be taken. The Employee acknowledges that the Company may close down all or part of its operations during the Christmas and New Year period. When this occurs, the Employee may be required to take leave during the period of the close down, even where it requires the Employee to take leave without pay. The Company will provide the Employee with at least 14 days advance notice of the close down.

- 3.5. Sick and Bereavement Leave. The Employee shall be entitled to take up to 5 days of paid sick leave per calendar year, prorated for any partial year of employment. The Employee may use such sick leave in the event the Employee is sick or injured, or if the Employee is required to stay home to care for: (a) the employee's spouse or partner; or (b) a person who depends on the Employee for care, either because they are sick or injured, or in the case of a dependant, their caregiver is sick or injured. Sick leave must be used in the year in which it was accrued, and may not be carried over from year to year. Any unused sick leave shall be forfeited in the event of the termination of the Employee s employment with the Company. In addition, the Employee shall be entitled to take up to three days of paid bereavement leave in the event that a member of the Employee's immediate family dies (i.e., the Employee's spouse or partner, parent, child, sibling, grandparent, grandchild, or a parent of the Employee's spouse or partner). The Employee will also be entitled to one day of paid leave where the Company otherwise accepts that the Employee has suffered a bereavement. In taking sick or bereavement leave, the Employee must follow all notice and documentation requirements as may be established by the Company from time to time.
  - 3.6. *Business Expenses*. The Company will reimburse the Employee for (or, at the Company's option, pay) all business travel and other out-of-pocket expenses reasonably incurred by the Employee in the performance of the Employee's duties. All reimbursable expenses must be appropriately documented together with receipts by the Employee on submission of any request for reimbursement.
- 3.7. Benefits and Perquisites. If offered by the Company, the Employee shall be eligible to participate in each additional benefit plan or program sponsored or maintained by the Company, whether now existing or established hereafter, to the extent that the Employee is eligible and/or invited to participate in any such plan under the applicable provisions thereof. With respect to any such plan, the Employee shall receive credit for his previous service as an employee of Cullen Investments Limited. The Company may, in its discretion, amend or terminate any such plan in accordance with the terms thereof. All plan terms shall control the entitlement over any benefits provided pursuant to each plan and moreover shall control over anything to the contrary herein.

#### . ANNUAL REVIEW

The Company shall review the Employee's performance annually, with the first such review to take place no later than the first anniversary of the Commencement Date. The review may take into account certain business objectives and performance indicators specified by the Company to the Employee. During the annual review, the Company will also review the Employee's salary and other benefits. The Employee is not guaranteed an increase in salary or improvement in benefits following the review.

5. DUTIES

The Employee shall perform the duties described in Schedule 1, as well as such other duties as may be assigned to him by the Company's Board of Directors from time to time. The Employee shall perform these duties primarily at the Company s facility in Georgia, USA, but the Company reserves the right to have the Employee perform services in additional locations at the Company s sole discretion. This may include locations within the USA and at any other location worldwide.

5.1. *Duties*: The Employee must, in carrying out the Employee's duties and responsibilities as outlined in the job description set out in Schedule 1:

- (a) act in a professional manner at all times;
- (b) promptly and fully comply with, and carry out, all policies, rules and directions given by the Company from time to time;
- (c) subject to Section 1, devote the whole of the Employee's efforts during work hours exclusively to D-2

the discharge of those duties and responsibilities, and the achievement of business objectives and performance indicators specified by the Company;

- perform such other reasonable duties and responsibilities in addition to those specified in this agreement as the Company may require the Employee to perform taking into account the Employee's skills and capabilities;
- render services away from the location specified above from time to time as the proper performance of his duties may require; and
- diligently and faithfully serve the Company, Eric Watson and the Company's associated and related businesses (f)including, without limitation, Cullen Agricultural Technologies, Inc., ( NDI Group ) using the Employee's best endeavors to promote and protect the interests of the Company, Eric Watson and the NDI Group.
  - 5.2. Measures. The measures and standards of performance include the following:
  - (a) effectively managing responsibilities and reporting personnel;
     (b) achievement of business objectives and performance indicators specified by the Company;
  - (c) contribution to overall efficiency of cost effectiveness and profitability of operations;
    - (d) application and development of technical skills;
      (e) high standard of work;
      (f) high standard of appearance;
      (g) ability to communicate;
      (h) internal/external feedback;
      (i) ability to back up other employees;
    - (j) friendly professional manner, even when under pressure.
    - 5.3. Working Relationships: In performing the Employee's duties the Employee will:
- (a) build good working relationships with the Company s affiliates, advisers and suppliers; ensure that the Employee has an honest and professional working relationship with management and other employees;
  - (c) be committed, flexible, trustworthy and reliable at all times; and
    (d) be able to submit ideas and recommendations on better ways to improve operations.

    6. POLICIES

The Employee will be bound by the Company's policies as may be issued from time to time, copies of which will be made available to the Employee. The Company reserves the right to amend its policies from time to time. The terms of this Agreement will override the Company's policies if they expressly conflict.

- 7. TRADE SECRETS AND CONFIDENTIAL INFORMATION; RESTRICTIVE COVENANTS 7.1. *Definitions*. For the purposes of this Agreement, the following definitions shall apply:
- (a) Trade Secrets shall mean information not generally known about the Business of the Company that is the subject of efforts that are reasonable under the circumstances to maintain its secrecy or confidentiality and from which the Company derives economic value from the fact that the information is not generally known to other persons who can obtain economic value from its disclosure or use. Trade Secrets include, but are not limited to, technical or non-technical data, compilations, programs, methods and systems, techniques, drawings, processes, financial data, lists of actual customers and potential customers, customer route books, cards or lists containing the names, addresses, buying habits and business locations of past, present and prospective customers, sales reports, price lists, product formulae, methods and procedures relating to services.

D-3

# **TABLE OF CONTENTS**

- (b) Confidential Information means other business information of the Company not generally known to the public and which the Company desires and makes reasonable efforts to keep confidential.
  - (c) Business of the Company means agricultural operations involving the implementation of proprietary systems, including those relevant to dairy farming.
- 7.2. Confidentiality. Employee agrees that he will not at any such time, other than in the performance of his duties for the Company, both during and after his employment by the Company, communicate or disclose to any person or entity, or use for his benefit or for the benefit of any other person or entity, directly or indirectly, any of the Company's Trade Secrets and/or Confidential Information. For the purposes of this Agreement, the prohibition against the disclosure of Confidential Information *only* shall end 10 years after the Termination Date. The disclosure of Trade Secrets is prohibited in perpetuity, or until the trade secret information becomes publicly available through no fault of Employee. The rights set forth in this paragraph are in addition to, and not in lieu of, any other rights of the Company whether statutory, at common law or otherwise.
- 7.3. Agreement Not to Solicit Customers. Employee covenants and agrees that during his employment by the Company and for a period of 36 months following the termination, for any reason, of such employment, whether voluntary or involuntary, he will not, whether directly or indirectly, on his own behalf or in the service or on behalf of another person or entity, solicit or attempt to solicit any customer or active prospect of the Company, or any representative of any customer or active prospect of the Company, with which Employee had personal contact for the purpose of furthering a business relationship on behalf of the Company, with a view to selling or providing any product, equipment or service competitive with the Business of the Company.
- 7.4. Covenant Not to Compete. Employee covenants and agrees that, unless expressly agreed otherwise by the Board of Directors, during the term of Employee s employment with the Company and for a period of 24 months following the termination, for any reason whether voluntary or involuntary, Employee shall not, within the USA and any other location in which the Company conducts, or plans to conduct within the following 12 months, business activities at the termination (which Employee expressly agrees is the primary geographic area to which his duties for the Company extend), either himself or on behalf of a business in competition with the Business of the Company, either directly or indirectly (whether through affiliates, subsidiaries or otherwise), perform any duties that are the same or similar to those that he presently performs for the Company.
- 7.5. Non-solicitation of Employees. Employee covenants and agrees that, during his employment by the Company and for a period of 36 months following the termination of such employment, for any reason, he will not, either directly or indirectly, on his own behalf or in the service or on behalf of another person or entity, recruit, solicit, induce or attempt to recruit, solicit or induce any employee of the Company with whom he had contact during his employment, to leave or cease his or her employment with the Company in order to undertake employment with or provide services to any business in competition with the Business of the Company.
- 7.6. Remedies. Employee acknowledges and agrees that by virtue of the duties and responsibilities attendant to his employment by the Company and the special knowledge of the Company s affairs, business, clients and operations that he has and will have as a consequence of such employment, irreparable loss and damage will be suffered by the Company if Employee should breach or violate this Section 7; and Employee further acknowledges and agrees that these covenants are reasonably necessary to preserve the Business of the Company. Employee, therefore, agrees and consents that, in addition to any other remedies available to it, the Company shall be entitled to an injunction to prevent a breach or contemplated breach of this Section 7.

INTELLECTUAL PROPERTY

WITNESSETH 458

8.

8.1. *NDI's Property*. The Employee acknowledges and agrees that any work the Employee produces in the course of the Employee's employment, and all intellectual property rights and interests (including common law rights and interests) everywhere in the world in or arising from that work, will belong exclusively to the Company or the relevant member of NDI Group: and the Employee assigns to the Company absolutely and waives all rights whatsoever (including moral rights) in any of that work and intellectual property.

D-4

#### **TABLE OF CONTENTS**

8.2. Assistance. The Employee must:

(a) without additional consideration, promptly sign all documents and do all things required by the NDI Group to give effect to this clause; and

not remove any of that work or intellectual property (or anything comprising that intellectual property or copies of (b)it) from NDI Group business premises, other than in the ordinary and proper course of the Employee's duties or with NDI's express consent.

9. OTHER BUSINESS INTERESTS/CONFLICTS

9.1. [Intentionally omitted]

- 9.2. *Relatives*. The Employee must advise the Company if an immediate family member, partner, friend, or close business associate is or becomes employed by one of the Company's competitors to avoid disclosing any commercially sensitive information to the Company's competitors, even inadvertently.
- 9.3. *Gifts*. The Employee may not accept or seek, in connection with NDI Group business any type of compensation, fees, commission, gifts, or any other personal favor or preferential treatment which exceeds a nominal value or goes beyond common courtesy associated with accepted business practice. This policy applies to consideration from other employees, prospective employees, customers, competitors, suppliers and any other persons or entities with whom NDI has or is likely to have any business relationship.
- 9.4. *Conflict of Interest*. The Employee must refer to the Company all business contacts, ideas and opportunities of which the Employee becomes aware which relate to NDI's business. The Employee must not use the Employee's position to benefit any business or person other than NDI or its affiliates. The Employee will avoid and will declare to NDI any conflict of interest as soon as it arises.

10. [Intentionally Omitted]11. SERIOUS MISCONDUCT

- 11.1. *Instant Dismissal*. Nothing in this Agreement prevents the Company from terminating the Employee's employment without notice if it considers the Employee is guilty of serious misconduct as set out in Schedule 2, which justifies instant dismissal.
- 11.2. Suspension. Where the Company considers that an act of serious misconduct may have occurred which warrants further investigation, it will be entitled to suspend the Employee on full pay, while an investigation into the matter is carried out. If, for reasons beyond its control, NDI is unable to complete its investigation within two weeks of the suspension being effected, any continuation of the suspension will be without pay. Otherwise, the Employee will remain on full pay during the period of suspension.

12. REDUNDANCY

- 12.1 *Definition*. Redundancy means a situation in which termination of the Employee's employment is attributable wholly or mainly to the fact that the position filled by the Employee has or will become superfluous to the existing requirements/needs of NDI.
- 12.2. *Notice*. If the Employee's employment is terminated for reason of redundancy, the Employee will be given three months' notice. NDI may require the Employee not to work out the full period of notice or part of it, and make a payment in lieu of notice of the Base Salary the Employee would have received had the full notice or part of it been worked.
  - 12.3. Compensation. No additional compensation for the loss of employment will be paid.

- 12.4. *Alternative Employment*. In the event that the Employee obtains alternative employment during the notice period applicable under clause 12.2, the Employee may, with NDI's consent (which will not be unreasonably withheld) leave without forfeiting the entitlement to be paid for the remainder of the notice period.
- 12.5 *Termination on External Restructure*. Where the termination of the Employee's employment is attributable wholly or mainly to an external restructure or sale, transfer or disposition of all or a material part

D-5

#### TABLE OF CONTENTS

of the NDI Group or NDI's business, no notice provisions will apply. NDI will advise the Employee of the Employee's entitlements (if any) on termination. For the avoidance of doubt, this clause does not impose an obligation on NDI to pay redundancy or any other form of compensation for the loss of the Employee's employment.

#### 13. EMPLOYEE'S OBLIGATIONS ON TERMINATION

On termination of employment the Employee must immediately, and before receiving the Employee's final pay, return to NDI, in a satisfactory condition, all records, documents, letters, books, computer disks, keys, security cards and other materials of every description (including copies), which are within the Employee's possession or control and which belong to or came from NDI or the NDI Group. If the Employee owes NDI Group any money, NDI may deduct the amount owed from the Employee's final pay and Employee hereby consents to such deduction.

#### 4. REPRESENTATIONS

In appointing the Employee to his position, NDI has relied on the Employee's representations as to the Employee's qualifications and experience. The Employee confirms that those representations are true and correct and that the Employee has disclosed everything, which if disclosed, would or may have been material to NDI's decision to appoint the Employee to the position. The Employee also acknowledges that NDI may take disciplinary action, including dismissal, if the Employee's representations were misleading or incorrect.

15. PRIVACY
The Employee consents to NDI:

- (a) collecting personal information about the Employee from time to time for NDI's personnel administration and insurance purposes;
- (b) disclosing this information fairly and reasonably to third parties when providing references or similar information; and
  - (c) disclosing this information where reasonably required for business reasons.

    16. GENERAL
  - 16.1 Smoke Free Environment. The Employee agrees to comply with NDI's non-smoking policy.
- 16.2 *Waiver of Breach.* No provision of this Agreement shall be modified, waived or discharged unless the modification, waiver or discharge is agreed to in writing and signed by Employee and by an authorized officer of the Company (other than Employee). No waiver by either party hereto of any breach of, or of compliance with, any condition or provision of this Agreement by the other party shall be considered a waiver of any other condition or provision or of the same condition or provision at another time.
- 16.3 Entire Agreement: Cancellation of Prior Agreements. This Agreement contains the entire agreement of the parties with respect to the subject matter hereof. It may not be changed orally, but only by an amendment in writing signed by the parties hereto. All prior agreements or understandings concerning Employee's employment by the Company are hereby canceled and superseded by this Agreement.
- 16.4. *Severability*. The invalidity or unenforceability of any provision of this Agreement shall not affect the validity or enforceability of the remainder of this Agreement.
- 16.5 *Headings*. The headings contained in this Agreement are for convenience only and shall not be deemed a part of this Agreement in construing or interpreting the provisions hereof.
- 16.6. *Governing Law*. This Agreement shall be governed by and construed in accordance with the laws of the State of New York. In connection with the enforcement of this Agreement, the parties consent to jurisdiction in the courts of

### New York.

- 16.7. *Costs*. Should the Company successfully enforce its rights against Employee under this Agreement, Company shall be entitled to its costs of such enforcement, including reasonable attorney s fees.
- 16.8. *Assignment*. Any successor to the Company (whether direct or indirect and whether by purchase, lease, merger, consolidation, liquidation or otherwise) to all or substantially all of the Company's business

D-6

and/or assets shall be entitled to assume the rights and obligations of the Company under this Agreement. This is a contract for personal services by Employee. Neither Employee s duties nor rights hereunder may be assigned by Employee in whole or part.

**IN WITNESS WHEREOF**, the parties have executed this Agreement as of the day, month and year first above written.

Company **NATURAL DAIRY, INC.** By:

Employee

Richard Watson

D-7

# SCHEDULE 1 JOB DESCRIPTION

Position Description: Chief Scientific Officer

Scope: The Chief Scientific Officer will take a lead role in developing and implementing the NDI Group's Agricultural Investment Strategy (AIS), including helping set the primary investment objectives as part of a comprehensive business plan. The Chief Scientific Officer, with the support of other executive staff, will be responsible for implementation of the AIS with the goal of meeting all primary investment objectives. The Chief Scientific Officer together with other executive staff shall be responsible for establishing and maintaining relationships with industry partners and service providers, government and regulatory bodies, banking and finance institutions, media, and public and private research and development organizations. The Chief Scientific Officer, with the help of senior farm management staff, will be responsible for overseeing general farming operations, with the key objective of maximizing returns to company shareholders.

#### Specific Responsibilities/Duties:

1. Agricultural Investment Strategy

- a. Identify properties/opportunities for investment of sufficient number and quality to achieve the objectives as detailed in the business plan.
- b. Prepare initial financial and physical feasibility studies for each property as part of an investment proposal for submission and consideration by appropriate executives within the NDI Group.
- Upon approval, manage the property acquisition process and due diligence activities, including but not limited to c.negotiations with vendors, appraisals, compliance and permitting requirements, project planning, securing financing and activities relevant to the closing of acquisitions.
- Manage farm conversion process with reference to the objectives of the business plan, including ensuring the most appropriate use of land prior to conversion, the appointment and oversight of project managers and contractors, development of a comprehensive project budget and timeline, and ensuring that the budgetary and timeline milestones are met.
- e. Prepare and submit monthly project reports, summary reports at frequencies reasonably determined by the NDI Group and a final project report to the NDI Group.
  - Farming Operations
  - a. Provide oversight to all farm management and technical staff.
- b. Identification, recruitment, and evaluation of all farm management, technical, and clerical staff. Work with farm managers to develop and implement long term (5 yrs) Farm Management Plans (FMP), which include, but are not limited to, pasture production and renovation, supplemental/grain feed, staffing and succession c. planning, animal breeding, management and production targets (including milk production and quality, stocking rate, culling rate, reproductive performance and body condition). FMP will be reviewed annually at the end of each production year.
- d. Collate individual FMPs and fit into a master management framework that enables synergies/efficiencies between farming units within a regional cluster in terms of expenditure, staffing requirements, management and production.

  e. Assist with professional development of management and subordinate staff, including the provision of training for new technologies and management practices as part of the FMP implementation.
  - f. Use FMP targets to prepare annual operating budgets for each farm, including any expected capital expenditure. Develop an expenditure policy, including spending limits for staff and delegated authority on spending. Coordinate company accountant, clerical staff, and farm managers in compilation of a quarterly budget reports and annual financial report for each farm.

Build individual FMPs and budgets into consolidated quarterly reports, and annual financial report to be provided to appropriate executives within the NDI Group.

D-8

# **TABLE OF CONTENTS**

h. Ensure company accountant uses appropriate accounting practice, and that all taxes and levies are paid as appropriate.

- i. Maintain insurance policies that are appropriate for the nature of farming activities in the relevant jurisdiction.
  - j. Ensure that at all times farms are in compliance with state and federal regulations.

k. Coordinate R&D activities and IP protection.

D-9

# SCHEDULE 2 SERIOUS MISCONDUCT

NDI may dismiss the Employee without notice (and without pay for the notice period) for serious misconduct, which includes (but is not limited to) the following types of behaviors:

(a) Dishonesty.
(b) Theft.
(c) Gross negligence.
(d) Unreasonable refusal to carry out duties.
(e) Physical violence against any person on NDI's premises.
(f) Intoxication at work or during work time.

(g) Being convicted of any crime involving dishonesty or any crime for which the Employee is sentenced to any term of periodic detention or imprisonment.

- (h) Loss of or inability to hold a visa necessary to be located at a particular location and/or perform applicable employment duties at a required location.
- (i) Supplying any false or misleading information in support of the Employee's application for employment.
  - (j) Unlawful harassment of or discrimination against any employee or customer.
  - (k) Use or possession of illegal drugs while on company property or during work time.
  - (1) Breach of the confidentiality provisions contained in the employment agreement.
- (m) Unauthorized absence arising upon absence from work for a continuous period of 3 consecutive working days without NDI's consent and without good reason.
- Abandonment arising upon absence from work for a continuous period of 3 consecutive working days without any notification to NDI and where NDI has made reasonable efforts to contact the Employee.

D-10

#### Exhibit E

# **ESCROW AGREEMENT**

ESCROW AGREEMENT ( Agreement ) dated [Closing Date] by and among CULLEN INC. HOLDINGS LTD. ( Stockholder ), CULLEN AGRICULTURAL HOLDING CORP. ( CAH ), and CONTINENTAL STOCK TRANSFER & TRUST COMPANY, as escrow agent (the Escrow Agent ).

Stockholder, CAH, Triplecrown Acquisition Corp. ( Triplecrown ), the parent of CAH, Cullen Agricultural Technologies Inc., a wholly owned subsidiary of Stockholder ( Cullen Agritech ), and CAT Merger Sub, Inc., a wholly owned subsidiary of CAH ( Merger Sub ), are the parties to a Agreement and Plan of Reorganization dated as of September , 2009 (the Merger Agreement ) pursuant to which Triplecrown has merged with and into CAH and Merger Sub has merged with and into Cullen Agritech. Capitalized terms used herein that are not otherwise defined herein shall have the meanings ascribed to them in the Merger Agreement. Pursuant to the Merger Agreement, CAH is to be indemnified in certain respects. The parties desire to establish an escrow fund as collateral security for the indemnification obligations under the Merger Agreement that is to consist of [1,588,115] shares of CAH Common Stock to be deposited by Stockholder (the Escrow Shares ).

#### The parties agree as follows:

- 1. (a) Concurrently with the execution hereof, Stockholder is delivering to the Escrow Agent, to be held in escrow pursuant to the terms of this Agreement, share certificates issued in the name of Stockholder representing the Escrow Shares, together with five (5) share transfer instruments separate from such certificate executed in blank by Stockholder, with medallion signature guaranties. The shares of CAH Common Stock represented by the share certificates so delivered by Stockholder to the Escrow Agent are herein referred to in the aggregate as the Escrow Fund.
- (b) The Escrow Agent hereby agrees to act as escrow agent and to hold, safeguard and disburse the Escrow Fund pursuant to the terms and conditions hereof. It shall treat the Escrow Fund as a trust fund in accordance with the terms of this Agreement and not as the property of CAH. The Escrow Agent s duties hereunder shall terminate upon its distribution of the entire Escrow Fund in accordance with this Agreement.
- (c) Except as herein provided, Stockholder and all Permitted Transferees (as hereinafter defined and, together with Stockholder, the Owners ) shall retain all of their rights as stockholders of CAH with respect to the CAH Common Stock constituting the Escrow Fund during the period any portion of the Escrow Fund is held by the Escrow Agent (the Escrow Period ), including, without limitation, the right to vote its CAH Common Stock included in the Escrow Fund.
- (d) During the Escrow Period, all dividends payable in cash with respect to the CAH Common Stock included in the Escrow Fund shall be paid to the Owners, but all dividends payable in stock or other non-cash property (Non-Cash Dividends) shall be delivered to the Escrow Agent to hold in accordance with the terms hereof. As used herein, the term Escrow Fund shall be deemed to include the Non-Cash Dividends distributed thereon, if any.
- (e) During the Escrow Period, no sale, transfer or other disposition may be made of any or all of the CAH Common Stock in the Escrow Fund except (i) to a Permitted Transferee (as hereinafter defined), (ii) by virtue of the laws of descent and distribution upon death of any Owner, or (iii) pursuant to a qualified domestic relations order; provided, however, that such permissive transfers may be implemented only upon the respective transferee s written agreement

to be bound by the terms and conditions of this Agreement. As used in this Agreement, the term Permitted Transferee shall include an entity in which (A) Stockholder beneficially owns 100% of such entity s voting and non-voting equity securities, or (B) Stockholder is a general partner and in which Stockholder beneficially owns 100% of all capital accounts of such entity. In connection with and as a condition to each permitted transfer, the Permitted Transferee shall deliver to the Escrow Agent an assignment separate from certificate executed by Stockholder, with medallion signature guaranty, or where applicable, an order of a court of competent jurisdiction, evidencing the transfer of shares to the Permitted Transferee, together with two (2) assignments separate from certificate executed in blank by the Permitted Transferee, with medallion signature guaranties, with respect to the shares transferred to the Permitted Transferee. Upon receipt of such documents, the Escrow Agent shall deliver to CAH s transfer agent the original

E-1

# **TABLE OF CONTENTS**

share certificate out of which the assigned shares are to be transferred, together with the executed share transfer instrument separate from certificate executed by Stockholder, or a copy of the applicable court order, and shall request that CAH issue new certificates representing (m) the number of shares, if any, that continue to be owned by the transferor, and (n) the number of shares owned by the Permitted Transferee as the result of such transfer. Stockholder, CAH and the Permitted Transferee shall cooperate in all respects with the Escrow Agent in documenting each such transfer and in effectuating the result intended to be accomplished thereby. During the Escrow Period, no Owner shall pledge or grant a security interest in such Owner s CAH Common Stock included in the Escrow Fund or grant a security interest in such Owner s rights under this Agreement.

#### 2. Indemnification Claims Procedure.

- (a) CAH, acting through the current or former member or members of CAH s Board of Directors who has or have been appointed by CAH to take all necessary actions and make all decisions on behalf of CAH with respect to its rights to indemnification under Article VII of the Merger Agreement (the Committee), may make a claim for indemnification pursuant to the Merger Agreement (Indemnification Claim) against the Escrow Shares in the Escrow Fund by giving notice (a Notice) to Stockholder (with a copy to the Escrow Agent) specifying (i) the covenant, representation, warranty, agreement, undertaking or obligation contained in the Merger Agreement which it asserts has been breached or otherwise entitles CAH to indemnification and (ii) in reasonable detail, the nature and dollar amount of any Indemnification Claim. The Committee also shall deliver to the Escrow Agent (with a copy to Stockholder), concurrently with its delivery to the Escrow Agent of the Notice, a certification as to the date on which the Notice was delivered to Stockholder.
- (b) If Stockholder shall give a notice to the Committee (with a copy to the Escrow Agent) (a Counter Notice), within 30 days following the date of receipt (as specified in the Committee s certification) by Stockholder of a copy of the Notice, disputing whether the Indemnification Claim is indemnifiable under the Merger Agreement, the Committee and Stockholder shall attempt to resolve such dispute by voluntary settlement as provided in Section 2(c) below. If no Counter Notice with respect to an Indemnification Claim is received by the Escrow Agent from Stockholder within such 30-day period, the Indemnification Claim shall be deemed to be an Established Claim (as hereinafter defined) for purposes of this Agreement.
- (c) If Stockholder delivers a Counter Notice to the Escrow Agent, the Committee and Stockholder shall, during the period of 60 days following the delivery of such Counter Notice or such greater period of time as the parties may agree to in writing (with a copy to the Escrow Agent), attempt to resolve the dispute with respect to which the Counter Notice was given. If the Committee and Stockholder shall reach a settlement with respect to any such dispute, they shall jointly deliver written notice of such settlement to the Escrow Agent specifying the terms thereof. If the Committee and Stockholder shall be unable to reach a settlement with respect to a dispute, such dispute shall be resolved by arbitration pursuant to Section 2(d) below.
  - (d) If the Committee and Stockholder cannot resolve a dispute prior to expiration of the 60-day period referred to in Section 2(c) above (or such longer period as the parties may have agreed to in writing), then such dispute shall be submitted (and either party may submit such dispute) to arbitration as set forth in Section 7 hereof.
- (e) As used in this Agreement, Established Claim means any (i) Indemnification Claim deemed established pursuant to the last sentence of Section 2(b) above, (ii) Indemnification Claim resolved in favor of CAH by settlement pursuant to Section 2(c) above, resulting in a dollar award to CAH, (iii) Indemnification Claim established by the decision of an arbitrator pursuant to Section 2(d) above, resulting in a dollar award to CAH, (iv) Third Party Claim that has been sustained by a final determination (after exhaustion of any appeals) of a court of competent jurisdiction, or (v) Third Party Claim that the Committee and Stockholder have jointly notified the Escrow Agent has been settled in

accordance with the provisions of the Merger Agreement.

E-2

of the Established Claim (or, if at such time there remains in the Escrow Fund less than the full amount so payable, the full amount remaining in the Escrow Fund).

- (ii) Payment of an Established Claim shall be made from Escrow Shares pro rata from the account maintained on behalf of each Owner. For purposes of each payment, such shares shall be valued at the Fair Market Value (as defined below). However, in no event shall the Escrow Agent be required to calculate Fair Market Value or make a determination of the number of shares to be delivered to Parent in satisfaction of any Established Claim; rather, such calculation shall be included in and made part of the Joint Notice. The Escrow Agent shall transfer to CAH out of the Escrow Fund that number of CAH Common Stock necessary to satisfy each Established Claim, as set out in the Joint Notice. Any dispute between the Committee and Stockholder concerning the calculation of Fair Market Value or the number of shares necessary to satisfy any Established Claim, or any other dispute regarding a Joint Notice, shall be resolved between the Committee and Stockholder in accordance with the procedures specified in Section 2(d) above, and shall not involve the Escrow Agent. Each transfer of shares in satisfaction of an Established Claim shall be made by the Escrow Agent delivering to CAH one or more share certificates held in each Owner s account evidencing not less than such Owner s pro rata portion of the aggregate number of shares specified in the Joint Notice, together with share transfer instruments separate from certificate executed in blank by such Owner and completed by the Escrow Agent in accordance with instructions included in the Joint Notice. Upon receipt of the share certificates and share transfer assignments, CAH shall deliver to the Escrow Agent new certificates representing the number of shares owned by each Owner after such payment. The parties hereto (other than the Escrow Agent) agree that the foregoing right to make payments of Established Claims in CAH Common Stock may be made notwithstanding any other agreements restricting or limiting the ability of any Owner to sell any shares of CAH Common Stock or otherwise. The Committee and Stockholder shall be required to exercise utmost good faith in all matters relating to the preparation and delivery of each Joint Notice. As used herein, Fair Market Value means the average reported closing price for the CAH Common Stock for the ten trading days ending on the last trading day prior to (x) the day the Established Claim is paid with respect to Indemnification Claims paid on or before the Escrow Termination Date, and (y) the Escrow Termination Date with respect to shares constituting the Pending Claims Reserve (as hereinafter defined) on the Escrow Termination Date.
- (iii) Notwithstanding anything herein to the contrary, at such time as an Indemnification Claim has become an Established Claim, Stockholder shall have the right to substitute for the Escrow Shares that otherwise would be paid in satisfaction of such claim (the Claim Shares), cash in an amount equal to the Fair Market Value of the Claim Shares (Substituted Cash). In such event (i) the Joint Notice shall include a statement describing the substitution of Substituted Cash for the Claim Shares, and (ii) substantially contemporaneously with the delivery of such Joint Notice, Stockholder shall cause currently available funds to be delivered to the Escrow Agent in an amount equal to the Substituted Cash. Upon receipt of such Joint Notice and Substituted Cash, the Escrow Agent shall (y) in payment of the Established Claim described in the Joint Notice, deliver the Substituted Cash to Stockholder in lieu of the Claim Shares, and (z) cause the Claim Shares to be returned to Stockholder.
- (g) On the first Business Day after the Escrow Termination Date, upon receipt of a Joint Notice, the Escrow Agent shall distribute and deliver to each Owner certificates representing Escrow Shares then in such Owner s account in the Escrow Fund less that number of shares in such Owner s account equal to the sum of (i) the number of shares applied in satisfaction of Indemnification Claims made prior to that date and (ii) the number of shares in the Pending Claims Reserve allocated to such Owner s account, as provided in the following sentence. If, at such time, there are any Indemnification Claims with respect to which Notices have been received but which have not been resolved pursuant to Section 2 hereof or in respect of which the Escrow Agent has not been notified of, and received a copy of, a final determination (after exhaustion of any appeals) by a court of competent jurisdiction, as the case may be (in either case, Pending Claims ), and which, if resolved or finally determined in favor of CAH, would result in a payment to CAH, the Escrow Agent shall retain in the Pending Claims Reserve that number of CAH Common Stock having a Fair

Market Value equal to the dollar amount for which indemnification is sought in such Indemnification Claim, allocated pro rata from the account maintained on behalf of each Owner. The Committee shall certify to the Escrow Agent the Fair Market Value to be used in calculating the Pending Claims Reserve and the number of CAH Common Stock to be retained therefor. Thereafter, if any Pending Claim becomes an Established Claim, the

E-3

# **TABLE OF CONTENTS**

Committee and Stockholder shall deliver to the Escrow Agent a Joint Notice directing the Escrow Agent to deliver to Parent the number of shares in the Pending Claims Reserve in respect thereof determined in accordance with Section 2(f) above and to deliver to each Owner the remaining shares in the Pending Claims Reserve allocated to such Pending Claim, all as specified in a Joint Notice. If any Pending Claim is resolved against Parent, the Committee and Stockholder shall deliver to the Escrow Agent a Joint Notice directing the Escrow Agent to pay to each Owner its pro rata portion of the number of shares allocated to such Pending Claim in the Pending Claims Reserve.

- (h) As used herein, the Pending Claims Reserve shall mean, at the time any such determination is made, that number of Escrow Shares in the Escrow Fund having a Fair Market Value equal to the sum of the aggregate dollar amounts claimed to be due with respect to all Pending Claims (as shown in the Notices of such Claims).
- 3. The Escrow Agent, the Committee and Stockholder shall cooperate in all respects with one another in the calculation of any amounts determined to be payable to Parent, Stockholder and the Owners in accordance with this Agreement and in implementing the procedures necessary to effect such payments.
- 4. (a) The Escrow Agent undertakes to perform only such duties as are expressly set forth herein. It is understood that the Escrow Agent is not a trustee or fiduciary and is acting hereunder merely in a ministerial capacity.
- (b) The Escrow Agent shall not be liable for any action taken or omitted by it in good faith and in the exercise of its own best judgment, and may rely conclusively and shall be protected in acting upon any order, notice, demand, certificate, opinion or advice of counsel (including counsel chosen by the Escrow Agent), statement, instrument, report or other paper or document (not only as to its due execution and the validity and effectiveness of its provisions, but also as to the truth and acceptability of any information therein contained) which is believed by the Escrow Agent to be genuine and to be signed or presented by the proper person or persons. The Escrow Agent shall not be bound by any notice or demand, or any waiver, modification, termination or rescission of this Agreement unless evidenced by a writing delivered to the Escrow Agent signed by the proper party or parties and, if the duties or rights of the Escrow Agent are affected, unless it shall have given its prior written consent thereto.
- (c) The Escrow Agent s sole responsibility upon receipt of any notice requiring any payment to Stockholder pursuant to the terms of this Agreement or, if such notice is disputed by the Committee or Stockholder, the settlement with respect to any such dispute, whether by virtue of joint resolution, arbitration or determination of a court of competent jurisdiction, is to pay to Parent the amount specified in such notice, and the Escrow Agent shall have no duty to determine the validity, authenticity or enforceability of any specification or certification made in such notice.
- (d) The Escrow Agent shall not be liable for any action taken by it in good faith and believed by it to be authorized or within the rights or powers conferred upon it by this Agreement, and may consult with counsel of its own choice and shall have full and complete authorization and indemnification under Section 4(g), below, for any action taken or suffered by it hereunder in good faith and in accordance with the opinion of such counsel.
- (e) The Escrow Agent may resign at any time and be discharged from its duties as escrow agent hereunder by its giving the other parties hereto written notice and such resignation shall become effective as hereinafter provided. Such resignation shall become effective at such time that the Escrow Agent shall turn over the Escrow Fund to a successor escrow agent appointed jointly by the Committee and Stockholder. If no new escrow agent is so appointed within the 60 day period following the giving of such notice of resignation, the Escrow Agent may deposit the Escrow Fund with any court it reasonably deems appropriate.
  - (f) The Escrow Agent shall be indemnified and held harmless by Stockholder and CAH, jointly and severally, from and against any expenses, including counsel fees and disbursements, or loss suffered by the Escrow Agent in

connection with any action, suit or other proceeding involving any claim which in any way, directly or indirectly, arises out of or relates to this Agreement, the services of the Escrow Agent hereunder, or the Escrow Fund held by it hereunder, other than expenses or losses arising from the gross negligence or willful misconduct of the Escrow Agent.

Promptly after the receipt by the Escrow Agent of notice of any

E-4

# **TABLE OF CONTENTS**

demand or claim or the commencement of any action, suit or proceeding, the Escrow Agent shall notify the other parties hereto in writing. In the event of the receipt of such notice, the Escrow Agent, in its sole discretion, may commence an action in the nature of interpleader in the United States District Court for the Southern District of New York.

- (g) The Escrow Agent shall be entitled to reasonable compensation from CAH for all services rendered by it hereunder. The Escrow Agent shall also be entitled to reimbursement from CAH for all expenses paid or incurred by it in the administration of its duties hereunder including, but not limited to, all counsel, advisors and agents fees and disbursements and all taxes or other governmental charges.
  - (h) From time to time on and after the date hereof, the Committee and Stockholder shall deliver or cause to be delivered to the Escrow Agent such further documents and instruments and shall do or cause to be done such further acts as the Escrow Agent shall reasonably request to carry out more effectively the provisions and purposes of this Agreement, to evidence compliance herewith or to assure itself that it is protected in acting hereunder.
  - (i) Notwithstanding anything herein to the contrary, the Escrow Agent shall not be relieved from liability hereunder for its own gross negligence or its own willful misconduct.
- 5. This Agreement expressly sets forth all the duties of the Escrow Agent with respect to any and all matters pertinent hereto. No implied duties or obligations shall be read into this Agreement against the Escrow Agent. The Escrow Agent shall not be bound by the provisions of any agreement among the parties hereto except this Agreement and shall have no duty to inquire into the terms and conditions of any agreement made or entered into in connection with this Agreement, including, without limitation, the Merger Agreement.
- 6. This Agreement shall inure to the benefit of and be binding upon the parties and their respective heirs, successors, assigns and legal representatives, shall be governed by and construed in accordance with the law of New York applicable to contracts made and to be performed therein. This Agreement cannot be changed or terminated except by a writing signed by the Committee, Stockholder and the Escrow Agent.
- 7. All disputes arising under this Agreement between the Committee and Stockholder, including a dispute arising from a party s failure or refusal to sign a Joint Notice, shall be submitted to arbitration to the American Arbitration Association in New York City. The Committee and Stockholder each hereby consents to the exclusive jurisdiction of the federal and state courts sitting in New York County, New York, with respect to any claim or controversy arising out of this Agreement. Service of process in any action or proceeding brought against the Committee or Stockholder in respect of any such claim or controversy may be made upon it by registered mail, postage prepaid, return receipt requested, at the address specified in Section 8, with copies delivered by nationally recognized overnight carrier to Graubard Miller, The Chrysler Building, 405 Lexington Avenue, New York, N.Y. 10174-1901, Attention: David Alan Miller, Esq., and to Ellenoff Grossman & Schole LLP, 150 East 42nd Street, New York, NY 10017, Attention: Douglas S. Ellenoff, Esq.
- 8. All notices and other communications under this Agreement shall be in writing and shall be deemed given if given by hand or delivered by nationally recognized overnight carrier, or if given by telecopier and confirmed by mail (registered or certified mail, postage prepaid, return receipt requested), to the respective parties as follows:

A. If to the Committee, to it at: with a copy to:

Graubard Miller
The Chrysler Building
405 Lexington Avenue
New York, New York 10174-1901
Attention: David Alan Miller, Esq.

Telecopier No.: 212-818-8881

E-5

B. If to Stockholder, to it at:
Cullen Inc. Holdings Ltd.
[Address]
Attention: Eric J. Watson
Telecopier No.:

with a copy to:

Ellenoff Grossman & Schole LLP 150 East 42nd Street New York, NY 10017 Attention: Douglas S. Ellenoff, Esq. Telecopier No.: 212-370-7889

C. If to the Escrow Agent, to it at:
Continental Stock Transfer & Trust Company
2 Broadway
New York, New York 10004
Attention: Steven G. Nelson
Telecopier No.: 212-509-5150

or to such other person or address as any of the parties hereto shall specify by notice in writing to all the other parties hereto.

- 9. (a) If this Agreement requires a party to deliver any notice or other document, and such party refuses to do so, the matter shall be submitted to arbitration as provided for herein.
- (b) All notices delivered to the Escrow Agent shall refer to the provision of this Agreement under which such notice is being delivered and, if applicable, shall clearly specify the aggregate dollar amount due and payable to Parent.
- (c) This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original instrument and all of which together shall constitute a single agreement.

IN WITNESS WHEREOF, each of the parties hereto has duly executed this Agreement on the date first above written.

# [Signatures are on following page]

E-6

# [Signature Page to Escrow Agreement]

CULLEN INC. HOLDINGS LTD. By:
Name:
Title:
CULLEN AGRICULTURAL HOLDING CORP. By:
Name:
Title:
ESCROW AGENT CONTINENTAL STOCK TRANSFER & TRUST COMPANY By:
Name:
Title:

#### Annex G

# AMENDED AND RESTATED CERTIFICATE OF INCORPORATION OF TRIPLECROWN ACQUISITION CORP.

# Pursuant to Section 245 of the Delaware General Corporation Law

TRIPLECROWN ACQUISITION CORP., a corporation existing under the laws of the State of Delaware (the Corporation ), by its President, hereby certifies as follows:

- 1. The name of the Corporation is Triplecrown Acquisition Corp.
- 2. The Corporation s Certificate of Incorporation was filed in the office of the Secretary of State of the State of Delaware on June 8, 2007.
- 3. This Amended Restated Certificate of Incorporation restates, integrates and amends the Certificate of Incorporation of the Corporation.
- 4. This Amended and Restated Certificate of Incorporation was duly adopted by joint written consent of the directors and stockholders of the Corporation in accordance with the applicable provisions of Sections 141(f), 228, 242 and 245 of the General Corporation Law of the State of Delaware (GCL).
  - 5. The text of the Certificate of Incorporation of the Corporation is hereby amended and restated to read in full as follows:
  - *FIRST*: The name of the corporation is Triplecrown Acquisition Corp. (hereinafter sometimes referred to as the Corporation ).
  - SECOND: The registered office of the Corporation is to be located at 615 S. DuPont Hwy., Kent County, Dover, Delaware. The name of its registered agent at that address is National Corporate Research, Ltd.
- *THIRD*: The purpose of the Corporation shall be to engage in any lawful act or activity for which corporations may be organized under the GCL.

*FOURTH*: The total number of shares of all classes of capital stock which the Corporation shall have authority to issue is 161,000,000 of which 160,000,000 shares shall be Common Stock of the par value of \$.0001 per share and 1,000,000 shares shall be Preferred Stock of the par value of \$.0001 per share.

A. *Preferred Stock.* Subject to paragraph F of Article SEVENTH, the Board of Directors is expressly granted authority to issue shares of the Preferred Stock, in one or more series, and to fix for each such series such voting powers, full or limited, and such designations, preferences and relative, participating, optional or other special rights and such qualifications, limitations or restrictions thereof as shall be stated and expressed in the resolution or resolutions adopted by the Board of Directors providing for the issue of such series (a Preferred Stock Designation ) and as may be permitted by the GCL. The number of authorized shares of Preferred Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the voting power of all of the then outstanding shares of the capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class, without a separate vote of the holders of the Preferred Stock, or any series thereof, unless a vote of any such holders is required pursuant to any Preferred Stock Designation.

B. *Common Stock*. Except as otherwise required by law or as otherwise provided in any Preferred Stock Designation, the holders of the Common Stock shall exclusively possess all voting power and each share of Common Stock shall have one vote.

FIFTH: Intentionally Omitted.

G-1

SIXTH: The Corporation s existence shall terminate on October 22, 2009 (the Termination Date ). This provision may only be amended in connection with, and become effective upon, the consummation of a Business Combination (defined below). A proposal to so amend this section shall be submitted to stockholders in connection with any proposed Business Combination pursuant to Article Seventh (A) below.

SEVENTH: The following provisions (A) through (E) shall apply during the period commencing upon the filing of this Certificate of Incorporation and terminating upon the consummation of any Business Combination, and may not be amended during the Target Business Acquisition Period. A Business Combination shall mean any merger, capital stock exchange, asset, stock purchase, reorganization or other similar business combination between the Corporation and one or more entities or assets ( Target Business or Target Businesses ). Any acquisition of multiple Target Businesses shall occur simultaneously. The Target Business Acquisition Period shall mean the period from the effectiveness of the registration statement filed in connection with the Corporation s initial public offering ( IPO ) up to and including the first to occur of (a) a Business Combination or (b) the Termination Date.

A. Prior to the consummation of any Business Combination, the Corporation shall submit such Business Combination to its stockholders for approval regardless of whether the Business Combination is of a type which normally would require such stockholder approval under the GCL. In the event that a majority of the IPO Shares (defined below) present and entitled to vote at the meeting to approve the Business Combination are voted for the approval of such Business Combination, the Corporation shall be authorized to consummate the Business Combination; provided that the Corporation shall not consummate any Business Combination if the holders of 30% or more of the IPO Shares exercise their conversion rights described in paragraph B below. The Corporation shall not seek to consummate a Business Combination unless stockholders owning up to 30% of the IPO Shares are able to elect conversion pursuant to the provisions of paragraph B below.

B. In the event that a Business Combination is approved in accordance with the above paragraph (A) and is consummated by the Corporation, any stockholder of the Corporation holding shares of Common Stock issued in the IPO ( IPO Shares ) who voted against the Business Combination may, contemporaneously with such vote, demand that the Corporation convert his IPO Shares into cash. If so demanded, the Corporation shall, promptly after consummation of the Business Combination, convert such shares into cash at a per share conversion price equal to the quotient determined by dividing (i) the amount in the Trust Fund (as defined below), inclusive of any interest thereon, calculated as of two business days prior to the consummation of the Business Combination, by (ii) the total number of IPO Shares. Trust Fund shall mean the trust account established by the Corporation at the consummation of its IPO and into which a certain amount of the net proceeds of the IPO is deposited.

C. In the event that the Corporation does not consummate a Business Combination by the Termination Date, the officers of the Corporation shall take all such action necessary to liquidate the Corporation as soon as reasonably possible. In the event that the Corporation is so liquidated, only the holders of IPO Shares shall be entitled to receive pro rata liquidating distributions and the Corporation shall pay no liquidating distributions with respect to any other shares of capital stock of the Corporation.

D. A holder of IPO Shares shall be entitled to receive distributions from the Trust Fund only in the event of a liquidation of the Corporation or in the event he demands conversion of his shares in accordance with paragraph B above. In no other circumstances shall a holder of IPO Shares have any right or interest of any kind in or to the Trust Fund.

E. Unless and until the Corporation has consummated a Business Combination as permitted under this Article SEVENTH, the Corporation may not consummate any other business combination, whether by merger, capital stock exchange, asset acquisition, stock purchase, reorganization or other similar business combination or transaction or

F. The Corporation shall not, and no employee of the Corporation shall, disburse or cause to be disbursed any of the proceeds held in the Trust Account except (i) for the payment of the Corporation s tax liability, (ii) for the release of interest income of up to \$6.0 million to the Corporation to fund the Corporation s working capital requirements, (iii) in connection with a Business Combination or thereafter, including the payment

G-2

of any Deferred Underwriting Compensation in accordance with the terms of the Underwriting Agreement, (iv) upon the Corporation s liquidation or (v) as otherwise set forth herein.

- G. The Audit Committee of the Corporation s Board of Directors will review and approve all payments made by the Corporation to its initial stockholders, sponsors, officers, directors and their or the Corporation s affiliates. Any payment made to a member of the Audit Committee will be reviewed and approved by the Board of Directors, with any member of the Board of Directors that has a financial interest in such payment abstaining from such review and approval. In no event will the Corporation pay any of its initial stockholders, sponsors, officers or directors or any entity with which they or it are affiliated, any finder s fee or other compensation for services rendered to it prior to or in connection with the consummation of a Business Combination; provided that the Corporation s initial stockholders, sponsors, officers, directors and its and their affiliates shall be entitled to reimbursement from the Corporation for their out-of-pocket expenses incurred in connection with investigating and consummating a Business Combination from the amounts not held in the Trust Account and interest income of up to \$5.0 million which may be released to the Corporation from the Trust Account. Repayment of advances of up to \$112,500 in the aggregate made to us by Eric J. Watson and Jonathan J. Ledecky to cover offering expenses shall not be subject to the provisions of this paragraph
- H. The members of the Audit Committee shall review the requirements of this Article SEVENTH at each quarterly meeting of the Audit Committee to determine compliance by the Corporation with the requirements hereof. In addition, the members of the Corporation s Audit Committee shall review the terms of all agreements (the IPO Agreements) between the Corporation and any of its officers or directors included as exhibits to the Registration Statement filed by the Corporation with the Securities and Exchange Commission to register the IPO Shares at each quarterly meeting of the Audit Committee to determine whether the parties to each IPO Agreement are in compliance. If any noncompliance is identified, then the Audit Committee shall promptly take all action necessary to rectify such noncompliance or otherwise cause compliance with the requirements of this Article SEVENTH or the terms and provisions of each IPO Agreement.
- I. The Board of Directors may not in any event issue any securities convertible, exercisable or redeemable into Common Stock, shares of Common Stock or Preferred Stock prior to a Business Combination that participates in or is otherwise entitled in any manner to any of the proceeds in the Trust Account or votes as a class with the Common Stock on a Business Combination.
- J. The Board of Directors shall be divided into three classes: Class A, Class B and Class C. The number of directors in each class shall be as nearly equal as possible. At the first election of directors by the incorporator, the incorporator shall elect a Class C director for a term expiring at the Corporation s third Annual Meeting of Stockholders. The Class C director shall then appoint additional Class A, Class B and Class C directors, as necessary. The directors in Class A shall be elected for a term expiring at the first Annual Meeting of Stockholders, the directors in Class B shall be elected for a term expiring at the second Annual Meeting of Stockholders and the directors in Class C shall be elected for a term expiring at the third Annual Meeting of Stockholders. Commencing at the first Annual Meeting of Stockholders, and at each annual meeting thereafter, directors elected to succeed those directors whose terms expire shall be elected for a term of office to expire at the third succeeding annual meeting of stockholders after their election. Except as the GCL may otherwise require, in the interim between annual meetings of stockholders or special meetings of stockholders called for the election of directors and/or the removal of one or more directors and the filling of any vacancy in that connection, newly created directorships and any vacancies in the Board of Directors, including unfilled vacancies resulting from the removal of directors for cause, may be filled by the vote of a majority of the remaining directors then in office, although less than a quorum (as defined in the Corporation s Bylaws), or by the sole remaining director. All directors shall hold office until the expiration of their respective terms of office and until their successors shall have been elected and qualified. A director elected to fill a vacancy resulting from the death,

resignation or removal of a director shall serve for the remainder of the full term of the director whose death, resignation or removal shall have created such vacancy and until his successor shall have been elected and qualified.

*EIGHTH*: The following provisions are inserted for the management of the business and for the conduct of the affairs of the Corporation, and for further definition, limitation and regulation of the powers of the Corporation and of its directors and stockholders:

G-3

- A. Election of directors need not be by ballot unless the by-laws of the Corporation so provide.
- B. The Board of Directors shall have the power, without the assent or vote of the stockholders, to make, alter, amend, change, add to or repeal the by-laws of the Corporation as provided in the by-laws of the Corporation, subject to the power of stockholders to alter or repeal any by-law whether adopted by them or otherwise.
- C. The directors in their discretion may submit any contract or act for approval or ratification at any annual meeting of the stockholders or at any meeting of the stockholders called for the purpose of considering any such act or contract, and any contract or act that shall be approved or be ratified by the vote of the holders of a majority of the stock of the Corporation which is represented in person or by proxy at such meeting and entitled to vote thereat (provided that a lawful quorum of stockholders be there represented in person or by proxy), unless a higher vote is required by applicable law, shall be as valid and binding upon the Corporation and upon all the stockholders as though it had been approved or ratified by every stockholder of the Corporation, whether or not the contract or act would otherwise be open to legal attack because of directors interests, or for any other reason.
- D. In addition to the powers and authorities hereinbefore or by statute expressly conferred upon them, the directors are hereby empowered to exercise all such powers and do all such acts and things as may be exercised or done by the Corporation; subject, nevertheless, to the provisions of the statutes of Delaware, of this Certificate of Incorporation, and to any by-laws from time to time made by the stockholders; provided, however, that no by-law so made shall invalidate any prior act of the directors which would have been valid if such by-law had not been made.
- NINTH: A. A director of the Corporation shall not be personally liable to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, except for liability (i) for any breach of the director s duty of loyalty to the Corporation or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) under Section 174 of the GCL, or (iv) for any transaction from which the director derived an improper personal benefit. If the GCL is amended to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of a director of the Corporation shall be eliminated or limited to the fullest extent permitted by the GCL, as so amended. Any repeal or modification of this paragraph A by the stockholders of the Corporation shall not adversely affect any right or protection of a director of the Corporation with respect to events occurring prior to the time of such repeal or modification.
- B. The Corporation, to the full extent permitted by Section 145 of the GCL, as amended from time to time, shall indemnify all persons whom it may indemnify pursuant thereto. Expenses (including attorneys fees) incurred by an officer or director in defending any civil, criminal, administrative, or investigative action, suit or proceeding for which such officer or director may be entitled to indemnification hereunder shall be paid by the Corporation in advance of the final disposition of such action, suit or proceeding upon receipt of an undertaking by or on behalf of such director or officer to repay such amount if it shall ultimately be determined that he is not entitled to be indemnified by the Corporation as authorized hereby.
- TENTH: Whenever a compromise or arrangement is proposed between this Corporation and its creditors or any class of them and/or between this Corporation and its stockholders or any class of them, any court of equitable jurisdiction within the State of Delaware may, on the application in a summary way of this Corporation or of any creditor or stockholder thereof or on the application of any receiver or receivers appointed for this Corporation under Section 291 of Title 8 of the Delaware Code or on the application of trustees in dissolution or of any receiver or receivers appointed for this Corporation under Section 279 of Title 8 of the Delaware Code order a meeting of the creditors or class of creditors, and/or of the stockholders or class of stockholders of this Corporation, as the case may be, to be summoned in such manner as the said court directs. If a majority in number representing three fourths in value of the creditors or class of creditors, and/or of the stockholders or class of stockholders of this Corporation, as the case may

be, agree to any compromise or arrangement and to any reorganization of this Corporation as a consequence of such compromise or arrangement, the said compromise or arrangement and the said reorganization shall, if sanctioned by the court

G-4

to which the said application has been made, be binding on all the creditors or class of creditors, and/or on all the stockholders or class of stockholders, of this Corporation, as the case may be, and also on this Corporation.

ELEVENTH: The Corporation hereby elects not to be governed by Section 203 of the GCL.

IN WITNESS WHEREOF, the Corporation has caused this Amended and Restated Certificate of Incorporation to be signed by Jonathan J. Ledecky, its President, as of the day of , 2009.

/s/ Jonathan J. Ledecky

Jonathan J. Ledecky, President

G-5

#### Annex H

# CULLEN AGRICULTURAL HOLDING CORP.

# 2009 Long-Term Incentive Equity Plan

# Section 1. Purpose; Definitions.

- 1.1. *Purpose*. The purpose of the 2009 Long-Term Incentive Equity Plan (Plan) is to enable the Company to offer to its employees, officers, directors and consultants whose past, present and/or potential contributions to the Company and its Subsidiaries have been, are or will be important to the success of the Company, an opportunity to acquire a proprietary interest in the Company. The various types of long-term incentive awards that may be provided under the Plan will enable the Company to respond to changes in compensation practices, tax laws, accounting regulations and the size and diversity of its businesses.
  - 1.2. Definitions. For purposes of the Plan, the following terms shall be defined as set forth below:
  - (a) Agreement means the agreement between the Company and the Holder, or such other document as may be determined by the Committee, setting forth the terms and conditions of an award under the Plan.
    - (b) Board means the Board of Directors of the Company.
    - (c) Code means the Internal Revenue Code of 1986, as amended from time to time.
  - (d) Committee means the committee of the Board designated to administer the Plan as provided in Section 2.1.
    - (e) Common Stock means the Common Stock of the Company, par value \$.0001 per share.
  - (f) Company means Cullen Agricultural Holding Corp., a corporation organized under the law of the State of Delaware.
    - (g) [Intentionally omitted.]
- (h) Disability means physical or mental impairment as determined under procedures established by the Committee for purposes of the Plan.
  - (i) Effective Date means the date determined pursuant to Section 11.1.
- (j) Fair Market Value, unless otherwise required by any applicable provision of the Code or any regulations issued thereunder, means, as of any given date: (i) if the Common Stock is listed on a national securities exchange or the Nasdaq Stock Market, the last sale price of the Common Stock in the principal trading market for the Common Stock on such date, as reported by the exchange or Nasdaq, as the case may be; (ii) if the Common Stock is not listed on a national securities exchange or the Nasdaq Stock Market, but is traded in the over-the-counter market, the closing bid price for the Common Stock on such date, as reported by the OTC Bulletin Board or Pink Sheets, LLC or similar publisher of such quotations; and (iii) if the fair market value of the Common Stock cannot be determined pursuant to clause (i) or (ii) above, such price as the Committee shall determine, in good faith.

- (k) Holder means a person who has received an award under the Plan.
- (l) Incentive Stock Option means any Stock Option intended to be and designated as an incentive stock option within the meaning of Section 422 of the Code.
  - (m) Non-qualified Stock Option means any Stock Option that is not an Incentive Stock Option.
- (n) Normal Retirement means retirement from active employment with the Company or any Subsidiary on or after such age which may be designated by the Committee as retirement age for any particular Holder. If no age is designated, it shall be 65.
- (o) Other Stock-Based Award means an award under Section 8 that is valued in whole or in part by reference to, or is otherwise based upon, Common Stock.

H-1

#### TABLE OF CONTENTS

- (p) Parent means any present or future parent corporation of the Company, as such term is defined in Section 424(e) of the Code.
- (q) Plan means the Cullen Agricultural Holding Corp. 2009 Long-Term Incentive Equity Plan, as hereinafter amended from time to time.
- (r) Repurchase Value shall mean the Fair Market Value if the award to be settled under Section 2.2(e) or repurchased under Section 9.2 is comprised of shares of Common Stock and the difference between Fair Market Value and the Exercise Price (if lower than Fair Market Value) if the award is a Stock Option or Stock Appreciation Right; in each case, multiplied by the number of shares subject to the award.
  - (s) Restricted Stock means Common Stock received under an award made pursuant to Section 7 that is subject to restrictions under Section 7.
- (t) SAR Value means the excess of the Fair Market Value (on the exercise date) over the exercise price that the participant would have otherwise had to pay to exercise the related Stock Option, multiplied by the number of shares for which the Stock Appreciation Right is exercised.
- (u) Stock Appreciation Right means the right to receive from the Company, on surrender of all or part of the related Stock Option, without a cash payment to the Company, a number of shares of Common Stock equal to the SAR Value divided by the Fair Market Value (on the exercise date).
- (v) Stock Option or Option means any option to purchase shares of Common Stock which is granted pursuant to the Plan.
- (w) Subsidiary means any present or future subsidiary corporation of the Company, as such term is defined in Section 424(f) of the Code.
  - (x) Vest means to become exercisable or to otherwise obtain ownership rights in an award.

# Section 2. Administration.

- 2.1. *Committee Membership.* The Plan shall be administered by a Committee of the Board of at least two directors, all of whom are outside directors within the meaning of the regulations issued under Section 162(m) of the Code. Committee members shall serve for such term as the Board may in each case determine and shall be subject to removal at any time by the Board.
- 2.2. *Powers of Committee*. The Committee shall have full authority to award, pursuant to the terms of the Plan: (i) Stock Options, (ii) Stock Appreciation Rights, (iii) Restricted Stock, and/or (iv) Other Stock-Based Awards. For purposes of illustration and not of limitation, the Committee shall have the authority (subject to the express provisions of this Plan):
  - (a) to select the officers, employees, directors and consultants of the Company or any Subsidiary to whom Stock Options, Stock Appreciation Rights, Restricted Stock and/or Other Stock-Based Awards may from time to time be awarded hereunder.

- (b) to determine the terms and conditions, not inconsistent with the terms of the Plan, of any award granted hereunder (including, but not limited to, number of shares, share exercise price or types of consideration paid upon exercise of such options, such as other securities of the Company or other property, any restrictions or limitations, and any vesting, exchange, surrender, cancellation, acceleration, termination, exercise or forfeiture provisions, as the Committee shall determine);
  - (c) to determine any specified performance goals or such other factors or criteria which need to be attained for the vesting of an award granted hereunder;
- (d) to determine the terms and conditions under which awards granted hereunder are to operate on a tandem basis and/or in conjunction with or apart from other equity awarded under this Plan and cash and non-cash awards made by the Company or any Subsidiary outside of this Plan; and
  - (e) to make payments and distributions with respect to awards (*i.e.*, to settle awards) through cash payments in an amount equal to the Repurchase Value.

H-2

The Committee may not modify or amend any outstanding Option or Stock Appreciation Right to reduce the exercise price of such Option or Stock Appreciation Right, as applicable, below the exercise price as of the date of grant of such Option or Stock Appreciation Right. In addition, no Option or Stock Appreciation Right may be granted in exchange for, or in connection with, the cancellation or surrender of an Option or Stock Appreciation Right or other award having a higher exercise price.

Notwithstanding anything to the contrary, the Committee shall not grant to any one Holder in any one calendar year awards for more than shares in the aggregate.

#### 2.3. *Interpretation of Plan*.

- (a) *Committee Authority*. Subject to Section 12, the Committee shall have the authority to adopt, alter and repeal such administrative rules, guidelines and practices governing the Plan as it shall from time to time deem advisable to interpret the terms and provisions of the Plan and any award issued under the Plan (and to determine the form and substance of all agreements relating thereto), and to otherwise supervise the administration of the Plan. Subject to Section 12, all decisions made by the Committee pursuant to the provisions of the Plan shall be made in the Committee s sole discretion and shall be final and binding upon all persons, including the Company, its Subsidiaries and Holders.
- (b) *Incentive Stock Options*. Anything in the Plan to the contrary notwithstanding, no term or provision of the Plan relating to Incentive Stock Options (including but not limited to Stock Appreciation rights granted in conjunction with an Incentive Stock Option) or any Agreement providing for Incentive Stock Options shall be interpreted, amended or altered, nor shall any discretion or authority granted under the Plan be so exercised, so as to disqualify the Plan under Section 422 of the Code or, without the consent of the Holder(s) affected, to disqualify any Incentive Stock Option under such Section 422.

# Section 3. Stock Subject to Plan.

- 3.1. *Number of Shares*. Subject to the last sentence of Section 7.1, the total number of shares of Common Stock reserved and available for issuance under the Plan shall be shares. Shares of Common Stock under the Plan (Shares) may consist, in whole or in part, of authorized and unissued shares or treasury shares. If any shares of Common Stock that have been granted pursuant to a Stock Option cease to be subject to a Stock Option, or if any shares of Common Stock that are subject to any Stock Appreciation Right, Restricted Stock award or Other Stock-Based Award granted hereunder are forfeited or any such award otherwise terminates without a payment being made to the Holder in the form of Common Stock, such shares shall again be available for distribution in connection with future grants and awards under the Plan. If a Holder pays the exercise price of a Stock Option by surrendering any previously owned shares and/or arranges to have the appropriate number of shares otherwise issuable upon exercise withheld to cover the withholding tax liability associated with the Stock Option exercise, then, in the Committee s discretion, the number of shares available under the Plan may be increased by the lesser of (i) the number of such surrendered shares and shares used to pay taxes; and (ii) the number of shares purchased under such Stock Option.
- 3.2. Adjustment Upon Changes in Capitalization, Etc. In the event of any common stock dividend payable on shares of Common Stock, Common Stock split or reverse split, combination or exchange of shares of Common Stock, or other extraordinary or unusual event which results in a change in the shares of Common Stock of the Company as a whole, the Committee shall determine, in its sole discretion, whether such change equitably requires an adjustment in the terms of any award in order to prevent dilution or enlargement of the benefits available under the Plan (including number of shares subject to the award and the exercise price) or the aggregate number of shares reserved for issuance

under the Plan. Any such adjustments will be made by the Committee, whose determination will be final, binding and conclusive.

# Section 4. Eligibility.

Awards may be made or granted to employees, officers, directors and consultants who are deemed to have rendered or to be able to render significant services to the Company or its Subsidiaries and who are deemed to have contributed or to have the potential to contribute to the success of the Company and which recipients are qualified to receive options under the regulations governing Form S-8 registration statements under the Securities Act of 1933, as amended (Securities Act). No Incentive Stock Option shall be granted to any person who is not an employee of the Company or an employee of a Subsidiary at the time of grant or

H-3

Section 4. Eligibility. 496

so qualified as set forth in the immediately preceding sentence. Notwithstanding the foregoing, an award may also be made or granted to a person in connection with his hiring or retention, or at any time on or after the date he reaches an agreement (oral or written) with the Company with respect to such hiring or retention, even though it may be prior to the date the person first performs services for the Company or its Subsidiaries; provided, however, that no portion of any such award shall vest prior to the date the person first performs such services and the date of grant shall be the date hiring or retention commences.

# Section 5. Stock Options.

- 5.1. Grant and Exercise. Stock Options granted under the Plan may be of two types: (i) Incentive Stock Options and (ii) Non-qualified Stock Options. Any Stock Option granted under the Plan shall contain such terms, not inconsistent with this Plan, or with respect to Incentive Stock Options, not inconsistent with the Plan and the Code, as the Committee may from time to time approve. The Committee shall have the authority to grant Incentive Stock Options or Non-qualified Stock Options, or both types of Stock Options which may be granted alone or in addition to other awards granted under the Plan. To the extent that any Stock Option intended to qualify as an Incentive Stock Option does not so qualify, it shall constitute a separate Non-qualified Stock Option.
  - 5.2. *Terms and Conditions*. Stock Options granted under the Plan shall be subject to the following terms and conditions:
- (a) *Option Term*. The term of each Stock Option shall be fixed by the Committee; provided, however, that an Incentive Stock Option may be granted only within the ten-year period commencing from the Effective Date and may only be exercised within ten years of the date of grant (or five years in the case of an Incentive Stock Option granted to an optionee who, at the time of grant, owns Common Stock possessing more than 10% of the total combined voting power of all classes of voting stock of the Company (10% Stockholder)).
- (b) Exercise Price. The exercise price per share of Common Stock purchasable under a Stock Option shall be determined by the Committee at the time of grant and may not be less than 100% of the Fair Market Value on the date of grant (or, if greater, the par value of a share of Common Stock); provided, however, that the exercise price of an Incentive Stock Option granted to a 10% Stockholder will not be less than 110% of the Fair Market Value on the date of grant.
- (c) Exercisability. Stock Options shall be exercisable at such time or times and subject to such terms and conditions as shall be determined by the Committee and as set forth in Section 10. The Committee intends generally to provide that Stock Options be exercisable only in installments, i.e., that they vest over time, typically over a three-year period. The Committee may waive such installment exercise provisions at any time at or after the time of grant in whole or in part, based upon such factors as the Committee determines.
- (d) *Method of Exercise*. Subject to whatever installment, exercise and waiting period provisions are applicable in a particular case, Stock Options may be exercised in whole or in part at any time during the term of the Option by giving written notice of exercise to the Company specifying the number of shares of Common Stock to be purchased. Such notice shall be accompanied by payment in full of the purchase price, which shall be in cash or, if provided in the Agreement, either in shares of Common Stock (including Restricted Stock and other contingent awards under this Plan) or partly in cash and partly in such Common Stock, or such other means which the Committee determines are consistent with the Plan s purpose and applicable law. Cash payments shall be made by wire transfer, certified or bank check or personal check, in each case payable to the order of the Company; provided, however, that the Company shall not be required to deliver certificates for shares of Common Stock with respect to which an Option is exercised

until the Company has confirmed the receipt of good and available funds in payment of the purchase price thereof (except that, in the case of an exercise arrangement approved by the Committee and described in the last sentence of this paragraph, payment may be made as soon as practicable after the exercise). The Committee may permit a Holder to elect to pay the Exercise Price upon the exercise of a Stock Option by irrevocably authorizing a third party to sell shares of Common Stock (or a sufficient

H-4

portion of the shares) acquired upon exercise of the Stock Option and remit to the Company a sufficient portion of the sale proceeds to pay the entire Exercise Price and any tax withholding resulting from such exercise.

- (e) *Stock Payments*. Payments in the form of Common Stock shall be valued at the Fair Market Value on the date of exercise. Such payments shall be made by delivery of stock certificates in negotiable form that are effective to transfer good and valid title thereto to the Company, free of any liens or encumbrances. A Holder shall have none of the rights of a Stockholder with respect to the shares subject to the Option until such shares shall be transferred to the Holder upon the exercise of the Option.
- (f) Transferability. Except as may be set forth in the next sentence of this Section or in the Agreement, no Stock Option shall be transferable by the Holder other than by will or by the laws of descent and distribution, and all Stock Options shall be exercisable, during the Holder s lifetime, only by the Holder (or, to the extent of legal incapacity or incompetency, the Holder s guardian or legal representative). Notwithstanding the foregoing, a Holder, with the approval of the Committee, may transfer a Non-Qualified Stock Option (i) (A) by gift, for no consideration, or (B) pursuant to a domestic relations order, in either case, to or for the benefit of the Holder s Immediate Family (as defined below), or (ii) to an entity in which the Holder and/or members of Holder s Immediate Family own more than fifty percent of the voting interest, in exchange for an interest in that entity, subject to such limits as the Committee may establish and the execution of such documents as the Committee may require, and the transferee shall remain subject to all the terms and conditions applicable to the Non-Qualified Stock Option prior to such transfer. The term Immediate Family shall mean any child, stepchild, grandchild, parent, stepparent, grandparent, spouse, former spouse, sibling, niece, nephew, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother-in-law or sister-in-law, including adoptive relationships, any person sharing the Holder s household (other than a tenant or employee), a trust in which these persons have more than fifty percent beneficial interest, and a foundation in which these persons (or the Holder) control the management of the assets. The Committee may, in its sole discretion, permit transfer of an Incentive Stock Option in a manner consistent with applicable tax and securities law upon the Holder s request.
- (g) *Termination by Reason of Death*. If a Holder s employment by, or association with, the Company or a Subsidiary terminates by reason of death, any Stock Option held by such Holder, unless otherwise determined by the Committee and set forth in the Agreement, shall thereupon automatically terminate, except that the portion of such Stock Option that has vested on the date of death may thereafter be exercised by the legal representative of the estate or by the legatee of the Holder under the will of the Holder, for a period of one year (or such other greater or lesser period as the Committee may specify in the Agreement) from the date of such death or until the expiration of the stated term of such Stock Option, whichever period is shorter.
- (h) *Termination by Reason of Disability*. If a Holder s employment by, or association with, the Company or any Subsidiary terminates by reason of Disability, any Stock Option held by such Holder, unless otherwise determined by the Committee and set forth in the Agreement, shall thereupon automatically terminate, except that the portion of such Stock Option that has vested on the date of termination may thereafter be exercised by the Holder for a period of one year (or such other greater or lesser period as the Committee may specify in the Agreement) from the date of such termination or until the expiration of the stated term of such Stock Option, whichever period is shorter.
- (i) *Termination by Reason of Normal Retirement*. Subject to the provisions of Section 12.3, if such Holder s employment by, or association with, the Company or any Subsidiary terminates due to Normal Retirement, any Stock Option held by such Holder, unless otherwise determined by the Committee and set forth in the Agreement, shall thereupon automatically terminate, except that the portion of such Stock Option that has vested on the date of termination may thereafter be exercised by the Holder for a period of one year (or such other greater or lesser period as the Committee may specify in the Agreement) from the date of such termination or until the expiration of the stated term of such Stock Option, whichever period is shorter.

- (j) Other Termination. Subject to the provisions of Section 12.3, if such Holder s employment by, or association with, the Company or any Subsidiary terminates for any reason other than death, Disability or Normal Retirement, any Stock Option held by such Holder, unless otherwise determined by the Committee and set forth in the Agreement, shall thereupon automatically terminate, except that, if the Holder s employment is terminated by the Company or a Subsidiary without cause, the portion of such Stock Option that has vested on the date of termination may thereafter be exercised by the Holder for a period of three months (or such other greater or lesser period as the Committee may specify in the Agreement) from the date of such termination or until the expiration of the stated term of such Stock Option, whichever period is shorter.
- (k) Additional Incentive Stock Option Limitation. In the case of an Incentive Stock Option, the aggregate Fair Market Value (on the date of grant of the Option) with respect to which Incentive Stock Options become exercisable for the first time by a Holder during any calendar year (under all such plans of the Company and its Parent and Subsidiaries) shall not exceed \$100,000.
  - (1) *Buyout and Settlement Provisions*. The Committee may at any time, in its sole discretion, offer to repurchase a Stock Option previously granted, based upon such terms and conditions as the Committee shall establish and communicate to the Holder at the time that such offer is made.

# **Section 6. Stock Appreciation Rights.**

- 6.1. *Grant and Exercise*. Subject to the terms and conditions of the Plan, the Committee may grant Stock Appreciation Rights in tandem with an Option or alone and unrelated to an Option. The Committee may grant Stock Appreciation Rights to participants who have been or are being granted Stock Options under the Plan as a means of allowing such participants to exercise their Stock Options without the need to pay the exercise price in cash. In the case of a Non-qualified Stock Option, a Stock Appreciation Right may be granted either at or after the time of the grant of such Non-qualified Stock Option. In the case of an Incentive Stock Option, a Stock Appreciation Right may be granted only at the time of the grant of such Incentive Stock Option.
  - 6.2. Terms and Conditions. Stock Appreciation Rights shall be subject to the following terms and conditions:
- (a) *Exercisability*. Stock Appreciation Rights shall be exercisable as shall be determined by the Committee and set forth in the Agreement, subject to the limitations, if any, imposed by the Code with respect to related Incentive Stock Options.
- (b) *Termination*. A Stock Appreciation Right shall terminate and shall no longer be exercisable upon the termination or after the exercise of the related Stock Option.
- (c) *Method of Exercise*. Stock Appreciation Rights shall be exercisable upon such terms and conditions as shall be determined by the Committee and set forth in the Agreement and by surrendering the applicable portion of the related Stock Option. Upon such exercise and surrender, the Holder shall be entitled to receive a number of shares of Common Stock equal to the SAR Value divided by the Fair Market Value on the date the Stock Appreciation Right is exercised.
- (d) Shares Affected Upon Plan. The granting of a Stock Appreciation Right shall not affect the number of shares of Common Stock available for awards under the Plan. The number of shares available for awards under the Plan will, however, be reduced by the number of shares of Common Stock acquirable upon exercise of the Stock Option to which such Stock Appreciation Right relates.

# Section 7. Restricted Stock.

7.1. *Grant.* Shares of Restricted Stock may be awarded either alone or in addition to other awards granted under the Plan. The Committee shall determine the eligible persons to whom, and the time or times at which, grants of Restricted Stock will be awarded, the number of shares to be awarded, the price (if any) to be paid by the Holder, the time or times within which such awards may be subject to forfeiture (Restriction Period), the vesting schedule and rights to acceleration thereof and all other terms and conditions of the awards. Notwithstanding anything to the contrary elsewhere in this Plan, for purposes of determining the

H-6

number of Shares available for awards pursuant to Section 3.1, each share of Common Stock subject to a Restricted Stock award shall be deemed to be Shares.

- 7.2. Terms and Conditions. Each Restricted Stock award shall be subject to the following terms and conditions:
- (a) *Certificates*. Restricted Stock, when issued, will be represented by a stock certificate or certificates registered in the name of the Holder to whom such Restricted Stock shall have been awarded. During the Restriction Period, certificates representing the Restricted Stock and any securities constituting Retained Distributions (as defined below) shall bear a legend to the effect that ownership of the Restricted Stock (and such Retained Distributions) and the enjoyment of all rights appurtenant thereto are subject to the restrictions, terms and conditions provided in the Plan and the Agreement. Such certificates shall be deposited by the Holder with the Company, together with stock powers or other instruments of assignment, each endorsed in blank, which will permit transfer to the Company of all or any portion of the Restricted Stock and any securities constituting Retained Distributions that shall be forfeited or that shall not become vested in accordance with the Plan and the Agreement.
- (b) *Rights of Holder*. Restricted Stock shall constitute issued and outstanding shares of Common Stock for all corporate purposes. The Holder will have the right to vote such Restricted Stock and to exercise all other rights, powers and privileges of a holder of Common Stock with respect to such Restricted Stock, with the exceptions that (i) the Holder will not be entitled to delivery of the stock certificate or certificates representing such Restricted Stock until the Restriction Period shall have expired and unless all other vesting requirements with respect thereto shall have been fulfilled; (ii) the Company will retain custody of the stock certificate or certificates representing the Restricted Stock during the Restriction Period; (iii) the Company will retain custody of all dividends and distributions ( Retained Distributions ) made, paid or declared with respect to the Restricted Stock (and such Retained Distributions will be subject to the same restrictions, terms and conditions as are applicable to the Restricted Stock) until such time, if ever, as the Restricted Stock with respect to which such Retained Distributions shall have been made, paid or declared shall have become vested and with respect to which the Restriction Period shall have expired; (iv) a breach of any of the restrictions, terms or conditions contained in this Plan or the Agreement or otherwise established by the Committee with respect to any Restricted Stock or Retained Distributions will cause a forfeiture of such Restricted Stock and any Retained Distributions with respect thereto.
- (c) *Vesting; Forfeiture*. Upon the expiration of the Restriction Period with respect to each award of Restricted Stock and the satisfaction of any other applicable restrictions, terms and conditions (i) all or part of such Restricted Stock shall become vested in accordance with the terms of the Agreement, and (ii) any Retained Distributions with respect to such Restricted Stock shall become vested to the extent that the Restricted Stock related thereto shall have become vested. Any such Restricted Stock and Retained Distributions that do not vest shall be forfeited to the Company and the Holder shall not thereafter have any rights with respect to such Restricted Stock and Retained Distributions that shall have been so forfeited.

# Section 8. Other Stock-Based Awards.

Other Stock-Based Awards may be awarded, subject to limitations under applicable law, that are denominated or payable in, valued in whole or in part by reference to, or otherwise based on or related to, shares of Common Stock, as deemed by the Committee to be consistent with the purposes of the Plan, including, without limitation, purchase rights, shares of Common Stock awarded which are not subject to any restrictions or conditions, convertible or exchangeable debentures, or other rights convertible into shares of Common Stock and awards valued by reference to the value of securities of or the performance of specified Subsidiaries. These other stock-based awards may include performance shares or options, whose award is tied to specific performance criteria. Other Stock-Based Awards may

be awarded either alone or in addition to or in tandem with any other awards under this Plan or any other plan of the Company. Each other Stock-Based Award shall be subject to such terms and conditions as may be determined by the Committee.

H-7

# Section 9. Accelerated Vesting and Exercisability.

- 9.1. Non-Approved Transactions. If any one person, or more than one person acting as a group, acquires the ownership of stock of the Company that, together with the stock held by such person or group, constitutes more than 50% of the total fair market value or combined voting power of the stock of the Company, and the Board does not authorize or otherwise approve such acquisition, then the vesting periods of any and all Stock Options and other awards granted and outstanding under the Plan shall be accelerated and all such Stock Options and awards will immediately and entirely vest, and the respective holders thereof will have the immediate right to purchase and/or receive any and all Common Stock subject to such Stock Options and awards on the terms set forth in this Plan and the respective agreements respecting such Stock Options and awards. An increase in the percentage of stock owned by any one person, or persons acting as a group, as a result of a transaction in which the Company acquires its stock in exchange for property is not treated as an acquisition of stock.
- 9.2. Approved Transactions. The Committee may, in the event of an acquisition by any one person, or more than one person acting as a group, together with acquisitions during the 12-month period ending on the date of the most recent acquisition by such person or persons, of assets from the Company that have a total gross fair market value equal to or more than 50% of the total gross fair market value of all of the assets of the Company immediately before such acquisition or acquisitions, or if any one person, or more than one person acting as a group, acquires the ownership of stock of the Company that, together with the stock held by such person or group, constitutes more than 50% of the total fair market value or combined voting power of the stock of the Company, which has been approved by the Company s Board of Directors, (i) accelerate the vesting of any and all Stock Options and other awards granted and outstanding under the Plan, or (ii) require a Holder of any award granted under this Plan to relinquish such award to the Company upon the tender by the Company to Holder of cash in an amount equal to the Repurchase Value of such award. For this purpose, gross fair market value means the value of the assets of the Company, or the value of the assets being disposed of, determined without regard to any liabilities associated with such assets.
- 9.3. *Code Section 409A*. Notwithstanding any provisions of this Plan or any award granted hereunder to the contrary, no acceleration shall occur with respect to any award to the extent such acceleration would cause the Plan or an award granted hereunder to fail to comply with Code Section 409A.

# Section 10. Amendment and Termination.

The Board may at any time, and from time to time, amend alter, suspend or discontinue any of the provisions of the Plan, but no amendment, alteration, suspension or discontinuance shall be made that would impair the rights of a Holder under any Agreement theretofore entered into hereunder, without the Holder s consent, except as set forth in this Plan.

# Section 11. Term of Plan.

- 11.1. *Effective Date*. The Plan shall be effective as of the date of the closing of the merger of Primoris Corporation, a Nevada corporation, into the Company.
- 11.2. *Termination Date*. Unless terminated by the Board, this Plan shall continue to remain effective until such time as no further awards may be granted and all awards granted under the Plan are no longer outstanding. Notwithstanding the foregoing, grants of Incentive Stock Options may be made only during the ten-year period beginning on the Effective Date.

# Section 12. General Provisions.

12.1. Written Agreements. Each award granted under the Plan shall be confirmed by, and shall be subject to the terms of, the Agreement executed by the Company and the Holder, or such other document as may be determined by the Committee. The Committee may terminate any award made under the Plan if the Agreement relating thereto is not executed and returned to the Company within 10 days after the Agreement has been delivered to the Holder for his or her execution.

H-8

12.2. *Unfunded Status of Plan*. The Plan is intended to constitute an unfunded plan for incentive and deferred compensation. With respect to any payments not yet made to a Holder by the Company, nothing contained herein shall give any such Holder any rights that are greater than those of a general creditor of the Company.

#### 12.3. Employees.

- (a) Engaging in Competition With the Company; Solicitation of Customers and Employees; Disclosure of Confidential Information. If a Holder s employment with the Company or a Subsidiary is terminated for any reason whatsoever, and within 12 months after the date thereof such Holder either (i) accepts employment with any competitor of, or otherwise engages in competition with, the Company or any of its Subsidiaries, (ii) solicits any customers or employees of the Company or any of its Subsidiaries to do business with or render services to the Holder or any business with which the Holder becomes affiliated or to which the Holder renders services or (iii) uses or discloses to anyone outside the Company any confidential information or material of the Company or any of its Subsidiaries in violation of the Company s policies or any agreement between the Holder and the Company or any of its Subsidiaries, the Committee, in its sole discretion, may require such Holder to return to the Company the economic value of any award (profit) that was realized or obtained by such Holder at any time during the period beginning on the date that is six months prior to the date such Holder s employment with the Company is terminated. In such event, Holder agrees to remit to the Company, in cash, an amount equal to the difference between the Fair Market Value of the Shares on the date of termination (or the sales price of such Shares if the Shares were sold during such six month period) and the price the Holder paid the Company for such Shares.
- (b) *Termination for Cause*. If a Holder s employment with the Company or a Subsidiary is terminated for cause, the Committee may, in its sole discretion, require such Holder to return to the Company the economic value of any award (profit) that was realized or obtained by such Holder at any time during the period beginning on that date that is six months prior to the date such Holder s employment with the Company is terminated. In such event, Holder agrees to remit to the Company, in cash, an amount equal to the difference between the Fair Market Value of the Shares on the date of termination (or the sales price of such Shares if the Shares were sold during such six month period) and the price the Holder paid the Company for such Shares.
- (c) *No Right of Employment*. Nothing contained in the Plan or in any award hereunder shall be deemed to confer upon any Holder who is an employee of the Company or any Subsidiary any right to continued employment with the Company or any Subsidiary, nor shall it interfere in any way with the right of the Company or any Subsidiary to terminate the employment of any Holder who is an employee at any time.
- 12.4. *Investment Representations; Company Policy*. The Committee may require each person acquiring shares of Common Stock pursuant to a Stock Option or other award under the Plan to represent to and agree with the Company in writing that the Holder is acquiring the shares for investment without a view to distribution thereof. Each person acquiring shares of Common Stock pursuant to a Stock Option or other award under the Plan shall be required to abide by all policies of the Company in effect at the time of such acquisition and thereafter with respect to the ownership and trading of the Company s securities.
- 12.5. Additional Incentive Arrangements. Nothing contained in the Plan shall prevent the Board from adopting such other or additional incentive arrangements as it may deem desirable, including, but not limited to, the granting of Stock Options and the awarding of Common Stock and cash otherwise than under the Plan; and such arrangements may be either generally applicable or applicable only in specific cases.
- 12.6. Withholding Taxes. Not later than the date as of which an amount must first be included in the gross income of the Holder for Federal income tax purposes with respect to any Stock Option or other award under the Plan, the

Holder shall pay to the Company, or make arrangements satisfactory to the Committee regarding the payment of, any Federal, state and local taxes of any kind required by law to be withheld or paid with respect to such amount. If permitted by the Committee, tax withholding or payment obligations may be settled with Common Stock, including Common Stock that is part of the award that gives rise to the

H-9

withholding requirement. The obligations of the Company under the Plan shall be conditioned upon such payment or arrangements and the Company or the Holder s employer (if not the Company) shall, to the extent permitted by law, have the right to deduct any such taxes from any payment of any kind otherwise due to the Holder from the Company or any Subsidiary.

- 12.7. *Governing Law*. The Plan and all awards made and actions taken thereunder shall be governed by and construed in accordance with the law of the State of Delaware (without regard to choice of law provisions).
- 12.8. Other Benefit Plans. Any award granted under the Plan shall not be deemed compensation for purposes of computing benefits under any retirement plan of the Company or any Subsidiary and shall not affect any benefits under any other benefit plan now or subsequently in effect under which the availability or amount of benefits is related to the level of compensation (unless required by specific reference in any such other plan to awards under this Plan).
  - 12.9. *Non-Transferability*. Except as otherwise expressly provided in the Plan or the Agreement, no right or benefit under the Plan may be alienated, sold, assigned, hypothecated, pledged, exchanged, transferred, encumbranced or charged, and any attempt to alienate, sell, assign, hypothecate, pledge, exchange, transfer, encumber or charge the same shall be void.
- 12.10. *Applicable Laws*. The obligations of the Company with respect to all Stock Options and awards under the Plan shall be subject to (i) all applicable laws, rules and regulations and such approvals by any governmental agencies as may be required, including, without limitation, the Securities Act, and (ii) the rules and regulations of any securities exchange on which the Common Stock may be listed.
- 12.11. *Conflicts*. If any of the terms or provisions of the Plan or an Agreement conflict with the requirements of Section 422 of the Code, then such terms or provisions shall be deemed inoperative to the extent they so conflict with such requirements. Additionally, if this Plan or any Agreement does not contain any provision required to be included herein under Section 422 of the Code, such provision shall be deemed to be incorporated herein and therein with the same force and effect as if such provision had been set out at length herein and therein. If any of the terms or provisions of any Agreement conflict with any terms or provisions of the Plan, then such terms or provisions shall be deemed inoperative to the extent they so conflict with the requirements of the Plan. Additionally, if any Agreement does not contain any provision required to be included therein under the Plan, such provision shall be deemed to be incorporated therein with the same force and effect as if such provision had been set out at length therein.
- 12.12. Certain Awards Deferring or Accelerating the Receipt of Compensation. To the extent applicable, all awards granted, and all agreements entered into, under the Plan are intended to comply with Section 409A of the Code, which was added by the American Jobs Creation Act of 2004 and relates to deferred compensation under nonqualified deferred compensation plans. The Committee, in administering the Plan, intends, and the parties entering into any agreement intend, to restrict provisions of any awards that may constitute deferred receipt of compensation subject to Code Section 409A requirements to those consistent with this Section. The Board may amend the Plan to comply with Code Section 409A in the future.
- 12.13. *Non-Registered Stock*. The shares of Common Stock to be distributed under this Plan have not been, as of the Effective Date, registered under the Securities Act or any applicable state or foreign securities laws and the Company has no obligation to any Holder to register the Common Stock or to assist the Holder in obtaining an exemption from the various registration requirements, or to list the Common Stock on a national securities exchange or any other trading or quotation system, including the Nasdaq Stock Market.

#### Annex I

September 4, 2009

Triplecrown Acquisition Corp. 970 West Broadway, PMB 402 Jackson, Wyoming 83001

#### Ladies and Gentlemen:

We have acted as special Delaware counsel to Triplecrown Acquisition Corp., a Delaware corporation (the Company), in connection with the proposed amendments to the certificate of incorporation of the Company. In this connection, you have requested our opinion as to certain matters under the General Corporation Law of the State of Delaware (the General Corporation Law).

For the purpose of rendering our opinion as expressed herein, we have been furnished and have reviewed the following documents:

- (i) the Amended and Restated Certificate of Incorporation of the Company, as filed with the Secretary of State of the State of Delaware (the Secretary of State) on October 22, 2007, as corrected by the Certificate of Correction of Amended and Restated Certificate of Incorporation of the Company, as filed with the Secretary of State on October 22, 2007 (collectively, the Certificate of Incorporation);
  - (ii) the Bylaws of the Company, adopted as of June 8, 2007 (the Bylaws );
- (iii) the form of the Amended and Restated Certificate of Incorporation of the Company (the Restated Certificate ) (a copy of which is attached hereto as Exhibit A);
  - (iv) the Form S-1/A of the Company (the Registration Statement ), as filed with the Securities and Exchange Commission (the SEC ) on October 5, 2007 in connection with the Company s initial public offering ( IPO );
  - (v) the Proxy Statement proposed to be filed with the SEC in connection with, among other things, the Restated Certificate (the Proxy Statement ); and
- (vi) the Agreement and Plan of Reorganization (the Agreement and Plan of Reorganization ), dated as of September \_\_\_, 2009, by and among the Company, Cullen Agricultural Holding Corp. ( Cullen Holdco ), CAT Merger Sub, Inc. ( Merger Sub ), Cullen Agricultural Technologies, Inc. ( Cullen Agritech ) and Cullen Inc Holdings Ltd ( Cullen Holdings ).

With respect to the foregoing documents, we have assumed: (a) the genuineness of all signatures, and the incumbency, authority, legal right and power and legal capacity under all applicable laws and regulations, of each of the officers and other persons and entities signing or whose signatures appear upon each of said documents as or on behalf of the parties thereto; (b) the conformity to authentic originals of all documents submitted to us as certified, conformed, photostatic, electronic or other copies; and (c) that the foregoing documents, in the forms submitted to us for our review, have not been and will not be altered or amended in any respect material to our opinion as expressed herein. For the purpose of rendering our opinion as expressed herein, we have not reviewed any document other than the documents set forth above, and, except as set forth in this opinion, we assume there exists no provision of any such other document that bears upon or is inconsistent with our opinion as expressed herein. We have conducted no

independent factual investigation of our own, but rather have relied as to factual matters solely upon the foregoing documents, the statements and information set forth therein, and the additional matters recited or assumed herein, all of which we assume to be true, complete and accurate in all material respects.

Triplecrown Acquisition Corp. September 4, 2009 Page 2

# **BACKGROUND**

We have been advised, and accordingly assume for purposes of our opinion as expressed herein, that (i) the Company has entered into the Agreement and Plan of Reorganization, which provides for (a) the merger of the Company with and into Cullen Holdco, with Cullen Holdco surviving the merger and becoming a public company through the automatic conversion of all the issued and outstanding securities of the Company into a like type and number of securities of Cullen Holdco, and (b) the merger of Merger Sub with and into Cullen Agritech, with Cullen Agritech surviving the merger and becoming a wholly owned subsidiary of Cullen Holdco through an exchange of all the issued and outstanding capital stock of Cullen Agritech for shares of common stock of Cullen Holdco (collectively, the Proposed Business Combination ); and (ii) the Proposed Business Combination does not constitute a Business Combination within the meaning of Article Seventh of the Certificate of Incorporation because (a) the sole stockholder of Cullen Agritech, the entity that will become Cullen Holdco s operating business following the Proposed Business Combination, is Cullen Holdings, an entity controlled by Eric J. Watson, the Company s chairman of the board and treasurer, (b) the Company will ultimately be acquired by Cullen Holdco and therefore the Company will not acquire at least 50% of the voting securities of the target business, (c) Cullen Agritech is not an operating business in the financial services industry, and (d) the fair market value of Cullen Agritech on the date of the Proposed Business Combination is less than 80% of the balance in the Company s Trust Fund (as defined in Article Seventh of the Certificate of Incorporation set forth below).

We have also been advised, and accordingly assume for purposes of our opinion as expressed herein, that (i) the Company considered and analyzed numerous companies and acquisition opportunities in its search for an attractive business combination candidate, none of which were believed to be as attractive to the Company s public stockholders as the Proposed Business Combination, and (ii) under Articles Sixth and Seventh of the Certificate of Incorporation, a failure to consummate a Business Combination (as defined in Article Seventh of the Certificate of Incorporation set forth below) by October 22, 2009 (the Termination Date ) will result in the dissolution and liquidation of the Company.

Accordingly, in order to consummate a Business Combination prior to the Termination Date, the Company is proposing to amend the Certificate of Incorporation as provided in the Restated Certificate to (i) delete all references to fair market value, (ii) delete the second sentence of Section E of Article Seventh relating to the prohibition on the Company consummating a business combination with an entity affiliated with any of the Company's officers, directors, initial stockholders or sponsors, and (iii) revise the definition of Business Combination to read in its entirety as follows:

A Business Combination shall mean any merger, capital stock exchange, asset, stock purchase, reorganization or other similar business combination between the Corporation and one or more entities or assets ( Target Business or Target Businesses ).

I-2

#### TABLE OF CONTENTS

Triplecrown Acquisition Corp. September 4, 2009 Page 3

In addition, we have been advised, and accordingly assume for purposes of our opinion as expressed herein, that (i) any of the Company s stockholders who affirmatively vote their IPO Shares against the Proposed Business Combination and demand that such shares be converted into cash will be entitled to receive a *pro rata* portion of the Trust Fund, calculated in accordance with the Certificate of Incorporation, if the Proposed Business Combination is consummated (the Conversion Rights); (ii) the stockholders vote on any proposal other than the proposal to approve the Proposed Business Combination will not adversely affect the stockholders Conversion Rights; and (iii) if the Proposed Business Combination is consummated, the Company s officers, directors and stockholders prior to the IPO (the Company Founders) have agreed to have cancelled an aggregate of 12,030,000 shares of common stock of Cullen Holdco they will receive in exchange for 12,030,000 shares of common stock acquired by them prior to the IPO (the Founders Shares) upon consummation of the Proposed Business Combination, and the remaining 1,770,000 Founders Shares, as well as all of the warrants of Cullen Holdco that the Company Founders will receive in exchange for their warrants, will continue to be held in escrow pursuant to the original terms of the escrow agreement and will be released one year after consummation of the Proposed Business Combination.

Article Seventh of the Certificate of Incorporation provides, in relevant part, as follows:

The following provisions (A) through (E) shall apply during the period commencing upon the filing of this Certificate of Incorporation and terminating upon the consummation of any Business Combination, and may not be amended during the Target Business Acquisition Period. A Business Combination shall mean the initial acquisition by the Corporation, whether by merger, capital stock exchange, asset or stock acquisition or other similar type of transaction, of one or more operating businesses in the financial services industry ( Target Businesses ), whose collective fair market value is equal to at least 80% of the balance in the Trust Fund (as defined below), excluding deferred underwriting discounts and commissions, and resulting in the ownership by the Corporation of not less than 50% of the voting securities of the Target Business or Businesses. Any acquisition of multiple Target Businesses shall occur simultaneously. The Target Business Acquisition Period shall mean the period from the effectiveness of the registration statement filed in connection with the Corporation s initial public offering ( IPO ) up to and including the first to occur of (a) a Business Combination or (b) the Termination Date.

Fair market value for purposes of this Article SEVENTH shall be determined by the Board of Directors of the Corporation based upon financial standards generally accepted by the financial community, such as actual and potential gross margins, the values of comparable businesses, earnings and cash flow, and book value. If the Corporation s Board of Directors is not able to determine independently that the Target Business or Businesses has a sufficient fair market value to meet the threshold criterion, it will obtain an opinion in that regard from an unaffiliated, independent investment banking firm that is a member of the Financial Industry Regulatory Authority and any successor organization thereto. The Corporation is not required to obtain an opinion from an investment banking firm as to the fair market value of the Target Business or Businesses if its Board of Directors independently determines that the Target Business or Businesses have sufficient fair market value to meet the threshold criterion.

We understand that the shares of common stock of the Company issued in the IPO constitute the IPO Shares. Thus, the IPO Shares consist of all of the Company's common stock except those shares of common stock acquired by the Company Founders prior to the IPO. We understand that the Company Founders do not have Conversion Rights with respect to their pre-IPO shares of the Company's common stock (*i.e.*, the Founders Shares).

BACKGROUND 514

#### TABLE OF CONTENTS

Triplecrown Acquisition Corp. September 4, 2009 Page 4

A. Prior to the consummation of any Business Combination, the Corporation shall submit such Business Combination to its stockholders for approval regardless of whether the Business Combination is of a type which normally would require such stockholder approval under the GCL. In the event that a majority of the IPO Shares (defined below) present and entitled to vote at the meeting to approve the Business Combination are voted for the approval of such Business Combination, the Corporation shall be authorized to consummate the Business Combination; provided that the Corporation shall not consummate any Business Combination if the holders of 30% or more of the IPO Shares exercise their conversion rights described in paragraph B below. The Corporation shall not seek to consummate a Business Combination unless stockholders owning up to 30% of the IPO Shares are able to elect conversion pursuant to the provisions of paragraph B below.

B. In the event that a Business Combination is approved in accordance with the above paragraph (A) and is consummated by the Corporation, any stockholder of the Corporation holding shares of Common Stock issued in the IPO (IPO Shares) who voted against the Business Combination may, contemporaneously with such vote, demand that the Corporation convert his IPO Shares into cash. If so demanded, the Corporation shall, promptly after consummation of the Business Combination, convert such shares into cash at a per share conversion price equal to the quotient determined by dividing (i) the amount in the Trust Fund (as defined below), inclusive of any interest thereon, calculated as of two business days prior to the consummation of the Business Combination, by (ii) the total number of IPO Shares. Trust Fund shall mean the trust account established by the Corporation at the consummation of its IPO and into which a certain amount of the net proceeds of the IPO is deposited.

C. In the event that the Corporation does not consummate a Business Combination by the Termination Date, the officers of the Corporation shall take all such action necessary to liquidate the Corporation as soon as reasonably possible. In the event that the Corporation is so liquidated, only the holders of IPO Shares shall be entitled to receive pro rata liquidating distributions and the Corporation shall pay no liquidating distributions with respect to any other shares of capital stock of the Corporation.

\*\*\*

E. Unless and until the Corporation has consummated a Business Combination as permitted under this Article SEVENTH, the Corporation may not consummate any other business combination, whether by merger, capital stock exchange, asset acquisition, stock purchase, reorganization or other similar business combination or transaction or otherwise. The Corporation will not enter into a Business Combination with any entity which is affiliated with any of its officers, directors, initial stockholders or sponsors or any entity that has received a material financial investment from its initial stockholders or sponsors or any entity affiliated with our officers, directors, initial stockholders or sponsors.

Thus, the underlined language in the first sentence of Article Seventh of the Certificate of Incorporation purports to eliminate the Company s (and, consequently, the Company s directors and stockholders) power to amend sections A through E of Article Seventh during the period commencing upon the effectiveness of the Registration Statement and terminating upon the first to occur of a Business Combination or the Termination Date (the Target Business Acquisition Period).

I-4

Triplecrown Acquisition Corp. September 4, 2009 Page 5

# DISCUSSION

You have asked our opinion as to whether Article Seventh may be amended as provided in the Restated Certificate. For the reasons set forth below, in our opinion, the proviso in the first sentence of Article Seventh of the Certificate of Incorporation, which purports to eliminate the Company s statutory power to amend sections A through E of Article Seventh during the Target Business Acquisition Period, is not a valid certificate of incorporation provision under the General Corporation Law.<sup>(2)</sup> Thus, Article Seventh may be amended as provided in the Restated Certificate, subject to compliance with the amendatory procedures set forth in Section 242(b) of the General Corporation Law.<sup>(3)</sup>

Section 242(a) of the General Corporation Law provides:

After a corporation has received payment for any of its capital stock, it may amend its certificate of incorporation, from time to time, in any and as many respects as may be desired, so long as its certificate of incorporation as amended would contain only such provisions as it would be lawful and proper to insert in an original certificate of incorporation filed at the time of the filing of the amendment....

8 Del. C. § 242(a). In addition, Section 242(b) of the General Corporation Law provides:

Every amendment [to the Certificate of Incorporation] ... *shall* be made and effected in the following manner: (1) [i]f the corporation has capital stock, its board of directors *shall* adopt a resolution setting forth the amendment proposed, declaring its advisability, and either calling a special meeting of the stockholders entitled to vote in respect thereof for the consideration of such amendment or directing that the amendment proposed be considered at the next annual meeting of the stockholders.... If a majority of the outstanding stock entitled to vote thereon, and a majority of the outstanding stock of each class entitled to vote thereon as a class has been voted in favor of the amendment, a certificate setting forth the amendment and certifying that such amendment has been duly adopted in accordance with this section *shall* be executed, acknowledged and filed and *shall* become effective in accordance with § 103 of this title.

Technically, of the amendments proposed, only the amendment to delete the second sentence of Section E of Article Seventh relating to the prohibition on the Company consummating a business combination with an entity affiliated with any of the Company s officers, directors, initial stockholders or sponsors implicates the bar on amendments to sections A through E of Article Seventh. The other amendments proposed are amendments to the introductory paragraphs of Article Seventh of the Certificate of Incorporation, and therefore the language purporting to prohibit amendments to sections A through E is not implicated.

Section 245(a) of the General Corporation Law provides that a corporation may, whenever desired, integrate into a single instrument all of the provisions of its certificate of incorporation that are then in effect, and it may at the same time also further amend its certificate of incorporation by adopting a restated certificate of incorporation. Section 245(b) provides that if the restated certificate of incorporation restates and integrates and also further amends in any respect the certificate of incorporation, it shall be proposed by the directors and adopted by the stockholders in the manner and by the vote prescribed by Section 242 of the General Corporation Law.

DISCUSSION 518

DISCUSSION 519

#### TABLE OF CONTENTS

Triplecrown Acquisition Corp. September 4, 2009 Page 6

8 *Del. C.* § 242(b) (emphasis added). Thus, Section 242(a) grants Delaware corporations broad statutory power to amend their certificates of incorporation to the extent permitted under Delaware law, including to the extent contemplated by the Restated Certificate, subject to compliance with the amendatory procedures set forth in Section 242(b). Implicit in the language of Section 242 is that the power to amend the certificate of incorporation is a fundamental power of Delaware corporations vested in directors and stockholders of a corporation. Nothing in Section 242 suggests that this statutory power may be entirely eliminated by a provision of the certificate of incorporation with respect to certain provisions thereof. Indeed, the mandatory language in Section 242(b) supports the proposition that the corporation s broad power to amend the certificate of incorporation cannot be eliminated. Section 242(b) mandates that, absent a provision permitting the board to abandon a proposed amendment, a certificate setting forth the amendment ....] *shall* be executed, acknowledged and filed and *shall* become effective upon obtaining the requisite board and stockholder approvals. 8 *Del. C.* § 242(b)(1) (emphasis added).

In our opinion, the proviso in Article Seventh of the Certificate of Incorporation that purports to eliminate the statutory power of the Company (and, consequently, of the directors and stockholders of the Company) to amend sections A through E of Article Seventh of the Certificate of Incorporation is contrary to the laws of the State of Delaware and, therefore, is invalid pursuant to Section 102(b)(1) of the General Corporation Law. Section 102(b)(1) provides that a certificate of incorporation may contain

[a]ny provision for the management of the business and for the conduct of the affairs of the corporation, and any provision creating, defining, limiting and regulating the powers of the corporation, the directors, and the stockholders, or any class of the stockholders...; if such provisions are not contrary to the laws of [the State of Delaware].

- 8 *Del. C.* § 102(b)(1) (emphasis added). Thus, the ability to curtail the powers of the corporation, the directors and the stockholders through the certificate of incorporation is not without limitation. Any provision in the certificate of incorporation that is contrary to Delaware law is invalid. *See Lions Gate Entm t Corp. v. Image Entm t Inc.*, 2006 WL 1668051, at \*7 (Del. Ch. June 5, 2006) (footnote omitted) (noting that a charter provision purport[ing] to give the Image board the power to amend the charter unilaterally without a shareholder vote after the corporation had received payment for its stock contravenes Delaware law [*i.e.*, Section 242 of the General Corporation Law] and is invalid ). In *Sterling v. Mayflower Hotel Corp.*, 93 A.2d 107, 118 (Del. 1952), the Court found that a charter provision is contrary to the laws of [Delaware] if it transgresses a statutory enactment or a public policy settled by the common law or implicit in the General Corporation Law itself. The Court in *Loew s Theatres, Inc. v. Commercial Credit Co.*, 243 A.2d 78, 81 (Del. Ch. 1968), adopted this view, noting that a charter provision which seeks to waive a statutory right or requirement is unenforceable.<sup>(4)</sup>
- (4) We note that Section 102(b)(4) of the General Corporation Law expressly permits a Delaware corporation to include in its certificate of incorporation provisions that modify the voting rights of directors and stockholders set forth in other provisions of the General Corporation Law. 8 *Del. C.* § 102(b)(4) ( the certificate of incorporation may also contain ... [p]rovisions requiring for any corporate action, the vote of a larger portion of the stock ... or a larger number of the directors, than is required by this chapter ). While Section 102(b)(4) permits certificate of incorporation provisions to require a greater vote of directors or stockholders than is otherwise required by the General Corporation Law, in our view, nothing in Section 102(b)(4) purports to authorize a certificate of incorporation provision that entirely eliminates the power of directors and stockholders to amend the certificate of incorporation (with respect to certain provisions thereof or otherwise) as expressly permitted by Section 242. *See*

DISCUSSION 520

also Sellers v. Joseph Bancroft & Sons Co., 2 A.2d 108, 114 (Del. Ch. 1938) (where the court questioned the validity of a certificate of incorporation provision requiring the vote or consent of 100% of the preferred stockholders to amend the certificate of incorporation in any manner that reduced the pecuniary rights of the preferred stock because the 100% vote requirement made such provision practically irrepealable ). I-6

DISCUSSION 521

Triplecrown Acquisition Corp. September 4, 2009 Page 7

That the statutory power to amend the certificate of incorporation is a fundamental power of Delaware corporations is supported by Delaware case law. Delaware courts have repeatedly held that a reservation of the right to amend the certificate of incorporation is a part of any certificate of incorporation, whether or not such reservation is expressly included therein. (5) See, e.g., Maddock v. Vorclone Corp., 147 A. 255 (Del. Ch. 1929); Coyne v. Park & Tilford Distillers Corp., 154 A.2d 893 (Del. 1959); Weinberg v. Baltimore Brick Co., 114 A.2d 812, 814 (Del. 1955); Morris v. American Public Utilities Co., 122 A. 696, 701 (Del. Ch. 1923). See also 2 David A. Drexler, Lewis S. Black, Jr. & A. Gilchrist Sparks, III, Delaware Corporation Law & Practice, § 32.02 (2005) ( No case has ever questioned the fundamental right of corporations to amend their certificates of incorporation in accordance with statutory procedures. From the earliest decisions, it has been held that every corporate charter implicitly contains as a constituent part thereof every pertinent provision of the corporation law, including the provisions authorizing charter amendments. ); 1 R. Franklin Balotti & Jesse A. Finkelstein, The Delaware Law of Corporations & Business Organizations § 8.1 (2007) Supp.) ( The power of a corporation to amend its certificate of incorporation was granted by the original General Corporation Law and has continued to this day. ) (footnotes omitted); 1 Edward P. Welch, Andrew J. Turezyn & Robert S. Saunders, Folk on the Delaware General Corporation Law § 242.2.2, GCL-VIII-13 (2007-1 Supp.) ( A corporation may ... do anything that section 242 authorizes because the grant of amendment power contained in section 242 and its predecessors is itself a part of the charter. ) (citing Goldman v. Postal Tel., Inc., 52 F.Supp. 763, 769 (D. Del. 1943); Davis v. Louisville Gas & Electric Co., 142 A. 654, 656-58 (Del. Ch. 1928); Morris, 122 A. at 701; Peters v. United States Mortgage Co., 114 A. 598, 600 (Del. Ch. 1921)); Peters, 114 A. at 600 (There is impliedly written into every corporate charter in this state, as a constituent part thereof, every pertinent provision of our Constitution and statutes. The corporation in this case was created under the General Corporation Law ... That law clearly reserves to this corporation the right to amend its certificate in the manner proposed. ).

In *Davis v. Louisville Gas & Electric Co.*, 142 A. 654 (Del. Ch. 1928), the Court of Chancery interpreted this reserved right to amend the certificate of incorporation broadly and observed that the legislature, by granting broad powers to the stockholders to amend the certificate of incorporation, recognized the unwisdom of casting in an unchanging mould the corporate powers which it conferred touching these questions so as to leave them fixed for all time. *Id.* at 657. Indeed, the Court queried, [m]ay it not be assumed that the Legislature foresaw that the interests of the corporations created by it might, as experience supplied the material for judgment, be best subserved by an alteration of their intracorporate and in a sense private powers, *i.e.*, by an alteration of the terms of the certificate of incorporation? *Id.* The Court further confirmed the important public policy underlying the reservation of the right to amend the certificate of incorporation:

The very fact that the [General Corporation Law]...deal[s] in great detail with innumerable aspects of the [certificate of incorporation] in what upon a glance would be regarded as relating to its private as distinguished from its public character, has some force to suggest that the state, by dealing with such subjects in the statute rather than by leaving them to be arranged by the corporate membership, has impliedly impressed upon such matters the quality of public interest and concern.

(5) This principle is also codified in Section 394 of the General Corporation Law. See 8 Del. C. § 394.

DISCUSSION 522

#### **TABLE OF CONTENTS**

Triplecrown Acquisition Corp. September 4, 2009 Page 8

### ld.

While there is no definitive case law addressing the enforceability or validity, under Delaware law or otherwise, of a certificate of incorporation provision that attempts to place a blanket prohibition on amendments to certain provisions thereof, in our view such a provision would be invalid. Indeed, in confirming the fundamental importance of a corporation s power to amend the certificate of incorporation, Delaware courts have suggested, in dicta, that such provision might be unenforceable. See, e.g., Jones Apparel Group, Inc. v. Maxwell Shoe Co., 883 A.2d 837 (Del. Ch. 2004) (indicating that the statutory power to recommend to stockholders amendments to the certificate of incorporation is a core duty of directors and noting that a certificate of incorporation provision purporting to eliminate a core duty of the directors would likely contravene Delaware public policy); Triplex Shoe Co. v. Rice & Hutchins, Inc., 152 A. 342, 347, 351 (Del. 1930) (finding that, despite the absence of common stockholders who held the sole power to vote on amendments to the certificate of incorporation, an amendment to the certificate of incorporation nonetheless had been validly approved by the preferred stockholders given that the holders of preferred stock, by the very necessities of the case, had the power to vote where no common stock had been validly issued because the corporation would otherwise be unable to function ); Sellers v. Joseph Bancroft & Sons Co., 2 A.2d 108, 114 (Del. Ch. 1938) (questioning the validity of a certificate of incorporation provision requiring the vote or consent of 100% of the preferred stockholders to amend the certificate of incorporation in any manner that reduced the pecuniary rights of the preferred stock because the 100% vote requirement made such provision practically irrepealable ).

More recently, the Court in *Jones Apparel* suggested that the right of directors to recommend to stockholders amendments to the certificate of incorporation is a core right of fundamental importance under the General Corporation Law. In *Jones Apparel*, the Delaware Court of Chancery examined whether a certificate of incorporation provision eliminating the power of a board of directors to fix record dates was permitted under Section 102(b)(1) of the General Corporation Law. While the Court upheld the validity of the record date provision, it was quick to point out that not all provisions in a certificate of incorporation purporting to eliminate director rights would be enforceable. *Jones Apparel*, 883 A.2d at 848. Rather, the Court suggested that certain statutory rights involving core director duties may not be modified or eliminated through the certificate of incorporation. The *Jones Apparel* Court observed:

[Sections] 242(b)(1) and 251 do not contain the magic words [ unless otherwise provided in the certificate of incorporation ] and they deal respectively with the fundamental subjects of certificate amendments and mergers. Can a certificate provision divest a board of its statutory power to approve a merger? Or to approve a certificate amendment? Without answering those questions, I think it fair to say that those questions inarguably involve far more serious intrusions on core director duties than does [the record date provision at issue]. I also think that the use by our judiciary of a more context- and statute-specific approach to police horribles is preferable to a sweeping rule that denudes § 102(b)(1) of its utility and thereby greatly restricts the room for private ordering under the DGCL.

*Id.* at 852. While the Court in *Jones Apparel* recognized that certain provisions for the regulation of the internal affairs of the corporation may be made subject to modification or elimination through the private ordering system of the certificate of incorporation and bylaws, it suggested that other powers vested in directors—such as the power to amend the certificate of incorporation—are so fundamental to the proper functioning of the corporation that they cannot be so modified or eliminated. *Id.* 

I-8

#### TABLE OF CONTENTS

Triplecrown Acquisition Corp. September 4, 2009 Page 9

As set forth above, the statutory language of Section 242 and Delaware case law confirm that the statutory power to amend the certificate of incorporation is a fundamental power of Delaware corporations as a matter of Delaware public policy. Moreover, Delaware case law also suggests that the fundamental power to amend the certificate of incorporation is a core right of the directors of a Delaware corporation. Because the proviso in the first sentence of Article Seventh of the Certificate of Incorporation purports to eliminate the fundamental power of the Company (and the core right of the Company s directors) to amend sections A through E thereof during the Target Business Acquisition Period, such provision is contrary to the laws of the State of Delaware and therefore is invalid.

Given our conclusion that Article Seventh may be amended as provided in the Restated Certificate, subject to compliance with the amendatory procedures set forth in Section 242(b) of the General Corporation Law, you have asked our opinion as to the vote required for approval of the Restated Certificate. Section 242(b) of the General Corporation Law provides the default voting requirements for an amendment to the certificate of incorporation. Under Section 242(b)(1), the board of directors of the Company (the Board ) would be required to adopt a resolution setting forth the amendments proposed (i.e., the amendments set forth in the Restated Certificate) and declaring their advisability prior to submitting such amendments to the stockholders entitled to vote on amendments to the Certificate of Incorporation. The Board may adopt such resolution by the affirmative vote of a majority of the directors present at a meeting at which a quorum is present or, alternatively, by unanimous written consent of all directors. See 8 Del. C. §§ 141(b), 141(f). After such amendments have been duly approved by the Board, they must then be submitted to the stockholders of the Company for a vote thereon. The affirmative vote (or written consent) of a majority of the outstanding stock entitled to vote thereon would be required for approval of such amendments. See 8 Del. C. §§ 242(b)(1), 228(a). The default voting requirements set forth above may be increased to require a greater vote of the directors or stockholders by a provision in the certificate of incorporation or the bylaws (in the case of the Board). See 8 Del. C. §§ 102(b)(4), 141(b), 216, 242(b)(4). However, any certificate of incorporation or bylaw provision purporting to impose a supermajority or unanimous voting requirement must otherwise be valid under the General Corporation Law. As discussed above, in our opinion, the proviso in the first sentence of Article Seventh of the Certificate of Incorporation, which purports to eliminate the statutory power of the Company (and, consequently, the directors and stockholders of the Company) to amend sections A through E of Article Seventh of the Certificate of Incorporation, is not a valid certificate of incorporation provision under the General Corporation Law. Because there is no valid provision in the Certificate of Incorporation or Bylaws purporting to impose a different or greater vote of the directors or stockholders for the approval of an amendment to the Certificate of Incorporation, in our view, the statutory default voting requirements would apply to the approval of the Restated Certificate by the directors and stockholders of the Company.

In addition, in our view, a Delaware court would not interpret the proviso in Article Seventh of the Certificate of Incorporation that purports to eliminate the power of the Company (and, consequently, the directors and stockholders) to amend sections A through E of Article Seventh as requiring a supermajority or unanimous vote of the directors and/or stockholders to approve the amendments purportedly prohibited thereby. Nothing in the language of Article Seventh suggests that the drafter s intent was to impose a supermajority or unanimous voting requirement on amendments thereto. Rather, the language in Article Seventh purports to entirely eliminate any vote by prohibiting any amendment to sections A through E of Article Seventh during the Target Business Acquisition Period. Moreover, in our view, a Delaware court would not reform the provisions of Article Seventh to provide for a voting requirement not intended by the drafters. See Lions Gate, 2006 WL 1668051, at \*8 (holding that reformation of a certificate of incorporation is unavailable where the proponent fails to demonstrate that all present and past shareholders intended the reformed provision to be included within the certificate) (citing Waggoner v. Laster, 581 A.2d 1127, 1135 (Del.

I-9

Triplecrown Acquisition Corp. September 4, 2009 Page 10

# CONCLUSION

Based upon and subject to the foregoing, and subject to the limitations stated herein, it is our opinion that the Restated Certificate, if duly adopted by the Board of Directors of the Company (by vote of the majority of the directors present at a meeting at which a quorum is present or, alternatively, by unanimous written consent) and duly approved by the holders of a majority of the outstanding stock of the Company entitled to vote thereon, all in accordance with Section 242(b) of the General Corporation Law, would be valid and effective when filed with the Secretary of State in accordance with Sections 103, 242 and 245 of the General Corporation Law.

The foregoing opinion is limited to the General Corporation Law. We have not considered and express no opinion on any other laws or the laws of any other state or jurisdiction, including federal laws regulating securities or any other federal laws, or the rules and regulations of stock exchanges or of any other regulatory body.

We understand that you may furnish a copy of this opinion letter to the SEC in connection with the matters addressed herein. We further understand that you may include this opinion letter as an annex to your Proxy Statement for the special meeting of stockholders of the Company to consider and vote upon the Restated Certificate, and we consent to your doing so. Except as stated in this paragraph, this opinion letter may not be furnished or quoted to, nor may the foregoing opinion be relied upon by, any other person or entity for any purpose without our prior written consent.

Very truly yours,

/s/ Richards, Layton & Finger, P.A.

CSB/TNP

I-10

CONCLUSION 527

# TABLE OF CONTENTS

#### Exhibit A

Amended and Restated Certificate of Incorporation

I-11

CONCLUSION 528

# AMENDED AND RESTATED CERTIFICATE OF INCORPORATION OF

# TRIPLECROWN ACQUISITION CORP.

# Pursuant to Section 245 of the Delaware General Corporation Law

TRIPLECROWN ACQUISITION CORP., a corporation existing under the laws of the State of Delaware (the Corporation ), by its President, hereby certifies as follows:

- 1. The name of the Corporation is Triplecrown Acquisition Corp.
- 2. The Corporation s Certificate of Incorporation was filed in the office of the Secretary of State of the State of Delaware on June 8, 2007.
- 3. This Amended Restated Certificate of Incorporation restates, integrates and amends the Certificate of Incorporation of the Corporation.
- 4. This Amended and Restated Certificate of Incorporation was duly adopted by joint written consent of the directors and stockholders of the Corporation in accordance with the applicable provisions of Sections 141(f), 228, 242 and 245 of the General Corporation Law of the State of Delaware ( GCL ).
  - 5. The text of the Certificate of Incorporation of the Corporation is hereby amended and restated to read in full as follows:
  - FIRST: The name of the corporation is Triplecrown Acquisition Corp. (hereinafter sometimes referred to as the Corporation ).
  - SECOND: The registered office of the Corporation is to be located at 615 S. DuPont Hwy., Kent County, Dover, Delaware. The name of its registered agent at that address is National Corporate Research, Ltd.
- THIRD: The purpose of the Corporation shall be to engage in any lawful act or activity for which corporations may be organized under the GCL.

FOURTH: The total number of shares of all classes of capital stock which the Corporation shall have authority to issue is 161,000,000 of which 160,000,000 shares shall be Common Stock of the par value of \$.0001 per share and 1,000,000 shares shall be Preferred Stock of the par value of \$.0001 per share.

A. *Preferred Stock*. Subject to paragraph F of Article SEVENTH, the Board of Directors is expressly granted authority to issue shares of the Preferred Stock, in one or more series, and to fix for each such series such voting powers, full or limited, and such designations, preferences and relative, participating, optional or other special rights and such qualifications, limitations or restrictions thereof as shall be stated and expressed in the resolution or resolutions adopted by the Board of Directors providing for the issue of such series (a Preferred Stock Designation ) and as may be permitted by the GCL. The number of authorized shares of Preferred Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the voting power of all of the then outstanding shares of the capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class, without a separate vote of the holders of the Preferred Stock, or any series thereof, unless a vote of any such holders is required pursuant to any Preferred Stock Designation.

B. *Common Stock*. Except as otherwise required by law or as otherwise provided in any Preferred Stock Designation, the holders of the Common Stock shall exclusively possess all voting power and each share of Common Stock shall have one vote.

#### TABLE OF CONTENTS

FIFTH: Intentionally Omitted.

SIXTH: The Corporation s existence shall terminate on October 22, 2009 (the Termination Date ). This provision may only be amended in connection with, and become effective upon, the consummation of a Business Combination (defined below). A proposal to so amend this section shall be submitted to stockholders in connection with any proposed Business Combination pursuant to Article Seventh (A) below.

SEVENTH: The following provisions (A) through (E) shall apply during the period commencing upon the filing of this Certificate of Incorporation and terminating upon the consummation of any Business Combination, and may not be amended during the Target Business Acquisition Period. A Business Combination shall mean any merger, capital stock exchange, asset, stock purchase, reorganization or other similar business combination between the Corporation and one or more entities or assets ( Target Business or Target Businesses ). Any acquisition of multiple Target Businesses shall occur simultaneously. The Target Business Acquisition Period shall mean the period from the effectiveness of the registration statement filed in connection with the Corporation s initial public offering ( IPO ) up to and including the first to occur of (a) a Business Combination or (b) the Termination Date.

A. Prior to the consummation of any Business Combination, the Corporation shall submit such Business Combination to its stockholders for approval regardless of whether the Business Combination is of a type which normally would require such stockholder approval under the GCL. In the event that a majority of the IPO Shares (defined below) present and entitled to vote at the meeting to approve the Business Combination are voted for the approval of such Business Combination, the Corporation shall be authorized to consummate the Business Combination; provided that the Corporation shall not consummate any Business Combination if the holders of 30% or more of the IPO Shares exercise their conversion rights described in paragraph B below. The Corporation shall not seek to consummate a Business Combination unless stockholders owning up to 30% of the IPO Shares are able to elect conversion pursuant to the provisions of paragraph B below.

B. In the event that a Business Combination is approved in accordance with the above paragraph (A) and is consummated by the Corporation, any stockholder of the Corporation holding shares of Common Stock issued in the IPO (IPO Shares) who voted against the Business Combination may, contemporaneously with such vote, demand that the Corporation convert his IPO Shares into cash. If so demanded, the Corporation shall, promptly after consummation of the Business Combination, convert such shares into cash at a per share conversion price equal to the quotient determined by dividing (i) the amount in the Trust Fund (as defined below), inclusive of any interest thereon, calculated as of two business days prior to the consummation of the Business Combination, by (ii) the total number of IPO Shares. Trust Fund shall mean the trust account established by the Corporation at the consummation of its IPO and into which a certain amount of the net proceeds of the IPO is deposited.

C. In the event that the Corporation does not consummate a Business Combination by the Termination Date, the officers of the Corporation shall take all such action necessary to liquidate the Corporation as soon as reasonably possible. In the event that the Corporation is so liquidated, only the holders of IPO Shares shall be entitled to receive pro rata liquidating distributions and the Corporation shall pay no liquidating distributions with respect to any other shares of capital stock of the Corporation.

D. A holder of IPO Shares shall be entitled to receive distributions from the Trust Fund only in the event of a liquidation of the Corporation or in the event he demands conversion of his shares in accordance with paragraph B above. In no other circumstances shall a holder of IPO Shares have any right or interest of any kind in or to the Trust Fund.

E. Unless and until the Corporation has consummated a Business Combination as permitted under this Article SEVENTH, the Corporation may not consummate any other business combination, whether by merger, capital stock exchange, asset acquisition, stock purchase, reorganization or other similar business combination or transaction or otherwise.

F. The Corporation shall not, and no employee of the Corporation shall, disburse or cause to be disbursed any of the proceeds held in the Trust Account except (i) for the payment of the Corporation s tax liability, (ii) for the release of interest income of up to \$6.0 million to the Corporation to fund the Corporation s working capital requirements, (iii) in connection with a Business Combination or thereafter, including the payment

of any Deferred Underwriting Compensation in accordance with the terms of the Underwriting Agreement, (iv) upon the Corporation s liquidation or (v) as otherwise set forth herein.

- G. The Audit Committee of the Corporation s Board of Directors will review and approve all payments made by the Corporation to its initial stockholders, sponsors, officers, directors and their or the Corporation s affiliates. Any payment made to a member of the Audit Committee will be reviewed and approved by the Board of Directors, with any member of the Board of Directors that has a financial interest in such payment abstaining from such review and approval. In no event will the Corporation pay any of its initial stockholders, sponsors, officers or directors or any entity with which they or it are affiliated, any finder s fee or other compensation for services rendered to it prior to or in connection with the consummation of a Business Combination; provided that the Corporation s initial stockholders, sponsors, officers, directors and its and their affiliates shall be entitled to reimbursement from the Corporation for their out-of-pocket expenses incurred in connection with investigating and consummating a Business Combination from the amounts not held in the Trust Account and interest income of up to \$5.0 million which may be released to the Corporation from the Trust Account. Repayment of advances of up to \$112,500 in the aggregate made to us by Eric J. Watson and Jonathan J. Ledecky to cover offering expenses shall not be subject to the provisions of this paragraph (G).
- H. The members of the Audit Committee shall review the requirements of this Article SEVENTH at each quarterly meeting of the Audit Committee to determine compliance by the Corporation with the requirements hereof. In addition, the members of the Corporation s Audit Committee shall review the terms of all agreements (the IPO Agreements) between the Corporation and any of its officers or directors included as exhibits to the Registration Statement filed by the Corporation with the Securities and Exchange Commission to register the IPO Shares at each quarterly meeting of the Audit Committee to determine whether the parties to each IPO Agreement are in compliance. If any noncompliance is identified, then the Audit Committee shall promptly take all action necessary to rectify such noncompliance or otherwise cause compliance with the requirements of this Article SEVENTH or the terms and provisions of each IPO Agreement.
- I. The Board of Directors may not in any event issue any securities convertible, exercisable or redeemable into Common Stock, shares of Common Stock or Preferred Stock prior to a Business Combination that participates in or is otherwise entitled in any manner to any of the proceeds in the Trust Account or votes as a class with the Common Stock on a Business Combination.
- J. The Board of Directors shall be divided into three classes: Class A, Class B and Class C. The number of directors in each class shall be as nearly equal as possible. At the first election of directors by the incorporator, the incorporator shall elect a Class C director for a term expiring at the Corporation s third Annual Meeting of Stockholders. The Class C director shall then appoint additional Class A, Class B and Class C directors, as necessary. The directors in Class A shall be elected for a term expiring at the first Annual Meeting of Stockholders, the directors in Class B shall be elected for a term expiring at the second Annual Meeting of Stockholders and the directors in Class C shall be elected for a term expiring at the third Annual Meeting of Stockholders. Commencing at the first Annual Meeting of Stockholders, and at each annual meeting thereafter, directors elected to succeed those directors whose terms expire shall be elected for a term of office to expire at the third succeeding annual meeting of stockholders after their election. Except as the GCL may otherwise require, in the interim between annual meetings of stockholders or special meetings of stockholders called for the election of directors and/or the removal of one or more directors and the filling of any vacancy in that connection, newly created directorships and any vacancies in the Board of Directors, including unfilled vacancies resulting from the removal of directors for cause, may be filled by the vote of a majority of the remaining directors then in office, although less than a quorum (as defined in the Corporation s Bylaws), or by the sole remaining director. All directors shall hold office until the expiration of their respective terms of office and until their successors shall have been elected and qualified. A director elected to fill a vacancy resulting from the death,

resignation or removal of a director shall serve for the remainder of the full term of the director whose death, resignation or removal shall have created such vacancy and until his successor shall have been elected and qualified.

EIGHTH: The following provisions are inserted for the management of the business and for the conduct of the affairs of the Corporation, and for further definition, limitation and regulation of the powers of the Corporation and of its directors and stockholders:

- A. Election of directors need not be by ballot unless the by-laws of the Corporation so provide.
- B. The Board of Directors shall have the power, without the assent or vote of the stockholders, to make, alter, amend, change, add to or repeal the by-laws of the Corporation as provided in the by-laws of the Corporation, subject to the power of stockholders to alter or repeal any by-law whether adopted by them or otherwise.
- C. The directors in their discretion may submit any contract or act for approval or ratification at any annual meeting of the stockholders or at any meeting of the stockholders called for the purpose of considering any such act or contract, and any contract or act that shall be approved or be ratified by the vote of the holders of a majority of the stock of the Corporation which is represented in person or by proxy at such meeting and entitled to vote thereat (provided that a lawful quorum of stockholders be there represented in person or by proxy), unless a higher vote is required by applicable law, shall be as valid and binding upon the Corporation and upon all the stockholders as though it had been approved or ratified by every stockholder of the Corporation, whether or not the contract or act would otherwise be open to legal attack because of directors interests, or for any other reason.
- D. In addition to the powers and authorities hereinbefore or by statute expressly conferred upon them, the directors are hereby empowered to exercise all such powers and do all such acts and things as may be exercised or done by the Corporation; subject, nevertheless, to the provisions of the statutes of Delaware, of this Certificate of Incorporation, and to any by-laws from time to time made by the stockholders; provided, however, that no by-law so made shall invalidate any prior act of the directors which would have been valid if such by-law had not been made.
- NINTH: A. A director of the Corporation shall not be personally liable to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, except for liability (i) for any breach of the director s duty of loyalty to the Corporation or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) under Section 174 of the GCL, or (iv) for any transaction from which the director derived an improper personal benefit. If the GCL is amended to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of a director of the Corporation shall be eliminated or limited to the fullest extent permitted by the GCL, as so amended. Any repeal or modification of this paragraph A by the stockholders of the Corporation shall not adversely affect any right or protection of a director of the Corporation with respect to events occurring prior to the time of such repeal or modification.
- B. The Corporation, to the full extent permitted by Section 145 of the GCL, as amended from time to time, shall indemnify all persons whom it may indemnify pursuant thereto. Expenses (including attorneys fees) incurred by an officer or director in defending any civil, criminal, administrative, or investigative action, suit or proceeding for which such officer or director may be entitled to indemnification hereunder shall be paid by the Corporation in advance of the final disposition of such action, suit or proceeding upon receipt of an undertaking by or on behalf of such director or officer to repay such amount if it shall ultimately be determined that he is not entitled to be indemnified by the Corporation as authorized hereby.
- TENTH: Whenever a compromise or arrangement is proposed between this Corporation and its creditors or any class of them and/or between this Corporation and its stockholders or any class of them, any court of equitable jurisdiction within the State of Delaware may, on the application in a summary way of this Corporation or of any creditor or stockholder thereof or on the application of any receiver or receivers appointed for this Corporation under Section 291 of Title 8 of the Delaware Code or on the application of trustees in dissolution or of any receiver or receivers

appointed for this Corporation under Section 279 of Title 8 of the Delaware Code order a meeting of the creditors or class of creditors, and/or of the stockholders or class of stockholders of this Corporation, as the case may be, to be summoned in such manner as the said court directs. If a majority in number representing three fourths in value of the creditors or class of creditors, and/or of the stockholders or class of stockholders of this Corporation, as the case may be, agree to any compromise

or arrangement and to any reorganization of this Corporation as a consequence of such compromise or arrangement, the said compromise or arrangement and the said reorganization shall, if sanctioned by the court to which the said application has been made, be binding on all the creditors or class of creditors, and/or on all the stockholders or class of stockholders, of this Corporation, as the case may be, and also on this Corporation.

#### Annex J

# LOCK-UP AGREEMENT

, 2009

Cullen Agricultural Holding Corp. 22 Barnett Shoals Road Watkinsville, GA 30677

Ladies and Gentlemen:

In connection with the Agreement and Plan of Reorganization, dated as of September , 2009, among Triplecrown Acquisition Corp. ( Parent ), Cullen Agricultural Holding Corp. ( CAH ), CAT Merger Sub, Inc. ( Triplecrown Merger Sub ), a wholly-owned subsidiary of CAH, Cullen Agricultural Technologies Inc. and Cullen Inc. Holdings Ltd. (the Merger Agreement ), to induce Parent to consummate the transactions contemplated by the Merger Agreement, the undersigned agrees to, neither directly nor indirectly, during the Restricted Period (as hereinafter defined):

sell or offer or contract to sell or offer, grant any option or warrant for the sale of, assign, transfer, pledge, hypothecate, or otherwise encumber or dispose of (all being referred to as a Transfer ) any legal or beneficial (1) interest in any shares of Holdco Common Stock (as defined in the Merger Agreement) issued to the undersigned in connection with the transactions contemplated by the Merger Agreement or otherwise acquired by the undersigned on or prior to the Closing Date (the Restricted Securities ), or enter into any swap or any other agreement or any transaction that transfers, in whole or in part, directly or indirectly, the economic consequence of ownership of any of the Restricted Securities, whether such swap transaction is to be settled by delivery of any Restricted Securities or other securities of any person, in cash or otherwise.

As used herein, Restricted Period means the period commencing on the Closing Date (as defined in the Merger Agreement) and ending on the day preceding the day that is one year after the Closing Date.

Any of the Restricted Securities subject to this Lock-Up Agreement may be released in whole or part from the terms hereof only upon the approval of the Board of Directors of Parent and a majority of the persons serving as directors of Parent immediately prior to the Closing Date (if any of such persons shall serve as directors).

The undersigned hereby authorizes Parent s transfer agent to apply to any certificates representing Restricted Securities issued to the undersigned the appropriate legend to reflect the existence and general terms of this Lock-up Agreement.

This Lock-up Agreement will be legally binding on the undersigned and on the undersigned s successors and permitted assigns, and is executed as an instrument governed by the law of New York.

Very truly yours,
CULLEN INC. HOLDINGS LTD.
By:
Title:

J-1

LOCK-UP AGREEMENT 538

LOCK-UP AGREEMENT 539

#### Annex K

August 31, 2009

# Confidential

Board of Directors
Triplecrown Acquisition Corp.
970 West Broadway
PMB 420
Jackson, WY 83001

To the Board of Directors:

The Board of Directors of Triplecrown Acquisition Corp. (the Company ) has engaged Duff & Phelps, LLC ( Duff & Phelps ) as its independent financial advisor to provide an opinion (the Opinion ) as of the date hereof as to the fairness, from a financial point of view to the Company, of the valuation of Cullen Agricultural Technologies Inc. ( Target ) negotiated among the parties in connection with the contemplated transaction described below (the Proposed Transaction ).

# **Description of the Proposed Transaction**

The Proposed Transaction involves the acquisition by the Company of Target based upon an agreed valuation of Target of \$155 million. Consideration consisting of newly issued shares of common stock (Shares) of a subsidiary of the Company which will be the new public company upon consummation of the Proposed Transaction would be paid to Target in the following manner:

Shares in the amount of \$155 million (15,881,148 Shares at an agreed value of \$9.76 per Share) will be issued to Target at closing. These Shares shall be locked up and may not be transferred until the first anniversary of the closing date of the Proposed Transaction. Release of the Shares in escrow is subject only to the passage of time.

Upon consummation of the Proposed Transaction, the Company intends to deploy Target s intellectual property in concert with the Company s own capital to make additional investments in land, infrastructure, livestock and working capital in the furtherance of its U.S. based dairy business plan involving the establishment of 40 new dairy farms operating within the Southeast United States (the U.S. Dairy Business Plan ). The investment thesis for Target is directly linked to the economics the Company believes it will realize as a result of successfully implementing its U.S. Dairy Business Plan.

# **Scope of Analysis**

In connection with this Opinion, Duff & Phelps has made such reviews, analyses and inquiries as we have deemed necessary and appropriate under the circumstances. Duff & Phelps also took into account its assessment of general economic, market and financial conditions, as well as its experience in securities and business valuation, in general, and with respect to similar transactions, in particular. Duff & Phelps procedures, investigations, and financial analysis with respect to the preparation of our Opinion included, but were not limited to, the items summarized below:

Confidential 540

- 1. Discussed the operations, financial conditions future prospects and projected operations and performance of the Company and Target with the managements of the Company and Target;
- Discussed the Proposed Transaction with the managements of the Company and Target;
   Reviewed certain publicly available financial statements and other business and financial information of the
   Company and proprietary information regarding Target and publicly available data for certain operating companies within the industry sector in which Target operates or will operate;

K-1

Scope of Analysis 541

Reviewed certain internal financial statements and other financial and operating data concerning the Company and 4. Target, which each of the Company and Target have identified as being the most current financial statements available;

- 5. Reviewed certain operating information for two test farms in Georgia owned by an affiliate of Target provided by the management of Target;
- 6. Reviewed certain financial forecasts prepared by the managements of the Company and Target, respectively, reflecting base case forecasts for financial and operating results under the U.S. Dairy Business Plan;
  - 7. Reviewed the Cullen Agricultural Technologies investor presentation dated August 2009; Reviewed the Cullen Agricultural Technologies presentation entitled Reconciliation of Actual CAT Research Farm Performance to Forecasts;
    - 9. Reviewed historical and forecasted domestic (U.S.) milk price information;
- 10. Reviewed U.S. milk supply and demand data prepared by the U.S. Department of Agriculture; Reviewed farm operating (e.g., herd stocking rates) and operating cost data for various farming formats (e.g., confinement, pastoral) for farms based domestically and abroad;
- 12. Reviewed a draft of the definitive merger agreement dated August 28, 2009 (the Agreement );
  13. Reviewed the historical trading price and trading volume of the publicly traded securities of certain other companies that we deemed relevant;
- 14. Compared the financial performance of Target with financial performance metrics of certain other publicly traded companies that we deemed relevant;
- 15. Compared certain financial terms of the Proposed Transaction to financial terms, to the extent publicly available, of certain other business combination transactions that we deemed relevant; and
  - 16. Conducted such other analyses and considered such other factors as we deemed appropriate.

# **Assumptions, Qualifications and Limiting Conditions**

In performing its analyses and rendering this Opinion with respect to the Proposed Transaction, Duff & Phelps, with your consent:

- Relied upon the accuracy, completeness, and fair presentation of all information, data, advice, opinions and
- 1.representations obtained from public sources or provided to it from private sources, including Company and Target management, and did not independently verify such information;
  - Relied upon managements characterization of the proprietary nature of Target s intellectual property and the fact that while such technology is not currently patent protected, the trade practices developed by Target in the areas of
- 2. forage sciences, animal genetics and health, farm management, and systems and training, have been uniquely adapted for conditions in the target market in Georgia and could not be readily imitated or adopted by current or future market participants;
  - Assumed that any estimates, evaluations, forecasts and projections furnished to Duff & Phelps were reasonably
- 3. prepared and based upon the best currently available information and good faith judgment of the person furnishing the same;
- 4. Assumed that the final versions of all documents reviewed by Duff & Phelps in draft form conform in all material respects to the drafts reviewed;
  - Assumed that information supplied to Duff & Phelps and representations and warranties made in the Agreement are substantially accurate;

Assumed that all of the conditions required to implement the Proposed Transaction will be satisfied and that the 6. Proposed Transaction will be completed in accordance with the Agreement without any amendments thereto or any waivers of any terms or conditions thereof;

K-2

8.

- 7 Assumed that all of the capital required to implement the proposed U.S. Dairy Business Plan will be available in the amounts required as and when needed as reflected in the U.S. Dairy Business Plan;
  - Relied upon the fact that the Board of Directors and the Company have been advised by counsel as to all legal
- 8. matters with respect to the Proposed Transaction, including whether all procedures required by law to be taken in connection with the Proposed Transaction have been duly, validly and timely taken; and
  - Assumed that all governmental, regulatory or other consents and approvals necessary for the consummation of the
- 9. Proposed Transaction will be obtained without any adverse effect on the Company, Target, the U.S. Dairy Business Plan or the contemplated benefits expected to be derived in the Proposed Transaction.

In our analysis and in connection with the preparation of this Opinion, Duff & Phelps has made numerous assumptions with respect to industry performance, regulatory and environmental (e.g., weather), general business, market and economic conditions and other matters, many of which are beyond the control of any party involved in the Proposed Transaction. To the extent that any of the foregoing assumptions or any of the facts on which this Opinion is based prove to be untrue in any material respect, this Opinion cannot and should not be relied upon.

Duff & Phelps did not make any independent evaluation, appraisal or physical inspection of the Company s solvency or of any specific assets or liabilities (contingent or otherwise). This Opinion should not be construed as a valuation opinion, credit rating, solvency opinion, an analysis of the Company s credit worthiness, as tax advice, or as accounting advice. Duff & Phelps has not been requested to, and did not, (a) initiate any discussions with, or solicit any indications of interest from, third parties with respect to the Proposed Transaction, Target or the Company s assets, businesses or operations of the Company, or any alternatives to the Proposed Transaction, (b) negotiate the terms of the Proposed Transaction, and therefore, Duff & Phelps has assumed that such terms are the most beneficial terms, from the Company s perspective, that could, under the circumstances, be negotiated among the parties to the Agreement and the Proposed Transaction, or (c) advise the Board of Directors or any other party with respect to alternatives to the Proposed Transaction. In addition, Duff & Phelps is not expressing any opinion as to the market price or value of the Shares or Share equivalents either as part of the Proposed Transaction or after announcement of the Proposed Transaction. Duff & Phelps has not made, and assumes no responsibility to make, any representation, or render any opinion, as to any legal matter.

In rendering this Opinion, Duff & Phelps is not expressing any opinion with respect to the amount or nature of any compensation to any of the Company s officers, directors, or employees, or any class of such persons, relative to the consideration to be received by the public shareholders of the Company in the Proposed Transaction, or with respect to the fairness of any such compensation.

The basis and methodology for this Opinion have been designed specifically for the Board of Directors to analyze the Proposed Transaction and may not translate to any other purposes. This Opinion (a) does not address the merits of the underlying business decision to enter into the Proposed Transaction versus any alternative strategy or transaction, including, without limitation, to liquidate the Company; (b) is not a recommendation as to how the Board of Directors or any stockholder should vote or act with respect to any matters relating to the Proposed Transaction (including, without limitation, with respect to the exercise of rights to convert shares of the Company s common stock into cash) or whether to proceed with the Proposed Transaction or any related transaction, and (c) does not indicate that the valuation of Target negotiated by the parties to the Proposed Transaction is the best possibly attainable under any circumstances; instead, it merely states whether the valuation of the Target in the Proposed Transaction is within a range suggested by certain financial analyses. The decision as to whether to proceed with the Proposed Transaction or any related transaction may depend on an assessment of factors unrelated to the financial analysis on which this Opinion is based. This letter should not be construed as creating any fiduciary duty on the part of Duff & Phelps to any party.

Duff & Phelps has prepared this Opinion effective as of the date hereof. This Opinion is necessarily based upon market, economic, financial and other conditions as they exist and can be evaluated as of the date

K-3

hereof, and Duff & Phelps disclaims any undertaking or obligation to advise any person of any change in any fact or matter affecting this Opinion which may come or be brought to the attention of Duff & Phelps after the date hereof.

This Opinion may be included in its entirety in any proxy statement distributed to stockholders of the Company in connection with the Proposed Transaction or other document required by law or regulation to be filed with the Securities and Exchange Commission, and you may summarize or otherwise reference the existence of this Opinion in such documents.

Duff & Phelps has acted as financial advisor to the Board of Directors of the Company and will receive a fee for its services. No portion of Duff & Phelps fee is contingent upon either the conclusion expressed in the Opinion or whether or not the Proposed Transaction is successfully consummated. Pursuant to the terms of the engagement letter between the Company and Duff & Phelps, a portion of Duff & Phelps fee is payable upon Duff & Phelps stating to the Board of Directors that it is prepared to deliver its Opinion. Other than this engagement, during the two years preceding the date of this Opinion, Duff & Phelps has not had any material relationship with any party to the Proposed Transaction for which compensation has been received or is intended to be received, nor is any such material relationship or related compensation mutually understood to be contemplated. As of the date hereof, the Target is considering utilizing the services of Duff & Phelps Securities, LLC, an affiliate of Duff & Phelps, LLC, in connection with capital raising efforts the Target may undertake if the Proposed Transaction does not occur. Duff & Phelps Securities, LLC is qualified to perform such services. The professional fees Duff & Phelps Securities, LLC would receive if it is awarded this additional project, some of which may be contingent upon the amount and type of capital raised, would not be material to the financial and operating results of Duff & Phelps.

# Conclusion

Based upon and subject to the foregoing, Duff & Phelps is of the opinion that as of the date hereof the valuation of Target negotiated between the parties in connection with the Proposed Transaction is fair, from a financial point of view, to the Company.

This Opinion has been approved by the internal opinion committee of Duff & Phelps.

Respectfully submitted,

Duff & Phelps, llc

K-4

Conclusion 546

#### Annex L

# SECTION 262 OF THE DELAWARE GENERAL CORPORATION LAW

- 262 APPRAISAL RIGHTS. (a) Any stockholder of a corporation of this State who holds shares of stock on the date of the making of a demand pursuant to subsection (d) of this section with respect to such shares, who continuously holds such shares through the effective date of the merger or consolidation, who has otherwise complied with subsection (d) of this section and who has neither voted in favor of the merger or consolidation nor consented thereto in writing pursuant to § 228 of this title shall be entitled to an appraisal by the Court of Chancery of the fair value of the stockholder s shares of stock under the circumstances described in subsections (b) and (c) of this section. As used in this section, the word stockholder means a holder of record of stock in a stock corporation and also a member of record of a nonstock corporation; the words stock and share mean and include what is ordinarily meant by those words and also membership or membership interest of a member of a nonstock corporation; and the words depository receipt mean a receipt or other instrument issued by a depository representing an interest in one or more shares, or fractions thereof, solely of stock of a corporation, which stock is deposited with the depository.
  - (b) Appraisal rights shall be available for the shares of any class or series of stock of a constituent corporation in a merger or consolidation to be effected pursuant to § 251 (other than a merger effected pursuant to § 251(g) of this title), § 252, § 254, § 257, § 258, § 263 or § 264 of this title:
- (1) Provided, however, that no appraisal rights under this section shall be available for the shares of any class or series of stock, which stock, or depository receipts in respect thereof, at the record date fixed to determine the stockholders entitled to receive notice of and to vote at the meeting of stockholders to act upon the agreement of merger or consolidation, were either (i) listed on a national securities exchange or designated as a national market system security on an interdealer quotation system by the National Association of Securities Dealers, Inc. or (ii) held of record by more than 2,000 holders; and further provided that no appraisal rights shall be available for any shares of stock of the constituent corporation surviving a merger if the merger did not require for its approval the vote of the stockholders of the surviving corporation as provided in subsection (f) of § 251 of this title.
- (2) Notwithstanding paragraph (1) of this subsection, appraisal rights under this section shall be available for the shares of any class or series of stock of a constituent corporation if the holders thereof are required by the terms of an agreement of merger or consolidation pursuant to §§ 251, 252, 254, 257, 258, 263 and 264 of this title to accept for such stock anything except:
- a. Shares of stock of the corporation surviving or resulting from such merger or consolidation, or depository receipts in respect thereof;
- b. Shares of stock of any other corporation, or depository receipts in respect thereof, which shares of stock (or depository receipts in respect thereof) or depository receipts at the effective date of the merger or consolidation will be either listed on a national securities exchange or designated as a national market system security on an interdealer quotation system by the National Association of Securities Dealers, Inc. or held of record by more than 2,000 holders;
- c. Cash in lieu of fractional shares or fractional depository receipts described in the foregoing subparagraphs a. and b. of this paragraph; or

- d. Any combination of the shares of stock, depository receipts and cash in lieu of fractional shares or fractional depository receipts described in the foregoing subparagraphs a., b. and c. of this paragraph.
- (3) In the event all of the stock of a subsidiary Delaware corporation party to a merger effected under § 253 of this title is not owned by the parent corporation immediately prior to the merger, appraisal rights shall be available for the shares of the subsidiary Delaware corporation.
- (c) Any corporation may provide in its certificate of incorporation that appraisal rights under this section shall be available for the shares of any class or series of its stock as a result of an amendment to its certificate of incorporation, any merger or consolidation in which the corporation is a constituent corporation or the sale

of all or substantially all of the assets of the corporation. If the certificate of incorporation contains such a provision, the procedures of this section, including those set forth in subsections (d) and (e) of this section, shall apply as nearly as is practicable.

### (d) Appraisal rights shall be perfected as follows:

(1) If a proposed merger or consolidation for which appraisal rights are provided under this section is to be submitted for approval at a meeting of stockholders, the corporation, not less than 20 days prior to the meeting, shall notify each of its stockholders who was such on the record date for such meeting with respect to shares for which appraisal rights are available pursuant to subsection (b) or (c) hereof that appraisal rights are available for any or all of the shares of the constituent corporations, and shall include in such notice a copy of this section. Each stockholder electing to demand the appraisal of such stockholder s shares shall deliver to the corporation, before the taking of the vote on the merger or consolidation, a written demand for appraisal of such stockholder s shares. Such demand will be sufficient if it reasonably informs the corporation of the identity of the stockholder and that the stockholder intends thereby to demand the appraisal of such stockholder s shares. A proxy or vote against the merger or consolidation shall not constitute such a demand. A stockholder electing to take such action must do so by a separate written demand as herein provided. Within 10 days after the effective date of such merger or consolidation, the surviving or resulting corporation shall notify each stockholder of each constituent corporation who has complied with this subsection and has not voted in favor of or consented to the merger or consolidation of the date that the merger or consolidation has become effective; or

(2) If the merger or consolidation was approved pursuant to § 228 or § 253 of this title, then, either constituent corporation before the effective date of the merger or consolidation, or the surviving or resulting corporation within ten days thereafter, shall notify each of the holders of any class or series of stock of such constituent corporation who are entitled to appraisal rights of the approval of the merger or consolidation and that appraisal rights are available for any or all shares of such class or series of stock of such constituent corporation, and shall include in such notice a copy of this section. Such notice may, and, if given on or after the effective date of the merger or consolidation, shall, also notify such stockholders of the effective date of the merger or consolidation. Any stockholder entitled to appraisal rights may, within 20 days after the date of mailing of such notice, demand in writing from the surviving or resulting corporation the appraisal of such holder s shares. Such demand will be sufficient if it reasonably informs the corporation of the identity of the stockholder and that the stockholder intends thereby to demand the appraisal of such holder s shares. If such notice did not notify stockholders of the effective date of the merger or consolidation, either (i) each such constituent corporation shall send a second notice before the effective date of the merger or consolidation notifying each of the holders of any class or series of stock of such constituent corporation that are entitled to appraisal rights of the effective date of the merger or consolidation or (ii) the surviving or resulting corporation shall send such a second notice to all such holders on or within 10 days after such effective date; provided, however, that if such second notice is sent more than 20 days following the sending of the first notice, such second notice need only be sent to each stockholder who is entitled to appraisal rights and who has demanded appraisal of such holder s shares in accordance with this subsection. An affidavit of the secretary or assistant secretary or of the transfer agent of the corporation that is required to give either notice that such notice has been given shall, in the absence of fraud, be prima facie evidence of the facts stated therein. For purposes of determining the stockholders entitled to receive either notice, each constituent corporation may fix, in advance, a record date that shall be not more than 10 days prior to the date the notice is given, provided, that if the notice is given on or after the effective date of the merger or consolidation, the record date shall be such effective date. If no record date is fixed and the notice is given prior to the effective date, the record date shall be the close of business on the day next preceding the day on which the notice is given.

(e) Within 120 days after the effective date of the merger or consolidation, the surviving or resulting corporation or any stockholder who has complied with subsections (a) and (d) hereof and who is otherwise entitled to appraisal rights, may file a petition in the Court of Chancery demanding a determination of the value of the stock of all such stockholders. Notwithstanding the foregoing, at any time within 60 days after

the effective date of the merger or consolidation, any stockholder shall have the right to withdraw such stockholder s demand for appraisal and to accept the terms offered upon the merger or consolidation. Within 120 days after the effective date of the merger or consolidation, any stockholder who has complied with the requirements of subsections (a) and (d) hereof, upon written request, shall be entitled to receive from the corporation surviving the merger or resulting from the consolidation a statement setting forth the aggregate number of shares not voted in favor of the merger or consolidation and with respect to which demands for appraisal have been received and the aggregate number of holders of such shares. Such written statement shall be mailed to the stockholder within 10 days after such stockholder s written request for such a statement is received by the surviving or resulting corporation or within 10 days after expiration of the period for delivery of demands for appraisal under subsection (d) hereof, whichever is

- (f) Upon the filing of any such petition by a stockholder, service of a copy thereof shall be made upon the surviving or resulting corporation, which shall within 20 days after such service file in the office of the Register in Chancery in which the petition was filed a duly verified list containing the names and addresses of all stockholders who have demanded payment for their shares and with whom agreements as to the value of their shares have not been reached by the surviving or resulting corporation. If the petition shall be filed by the surviving or resulting corporation, the petition shall be accompanied by such a duly verified list. The Register in Chancery, if so ordered by the Court, shall give notice of the time and place fixed for the hearing of such petition by registered or certified mail to the surviving or resulting corporation and to the stockholders shown on the list at the addresses therein stated. Such notice shall also be given by one or more publications at least one week before the day of the hearing, in a newspaper of general circulation published in the City of Wilmington, Delaware or such publication as the Court deems advisable. The forms of the notices by mail and by publication shall be approved by the Court, and the costs thereof shall be borne by the surviving or resulting corporation.
- (g) At the hearing on such petition, the Court shall determine the stockholders who have complied with this section and who have become entitled to appraisal rights. The Court may require the stockholders who have demanded an appraisal for their shares and who hold stock represented by certificates to submit their certificates of stock to the Register in Chancery for notation thereon of the pendency of the appraisal proceedings; and if any stockholder fails to comply with such direction, the Court may dismiss the proceedings as to such stockholder.
- (h) After determining the stockholders entitled to an appraisal, the Court shall appraise the shares, determining their fair value exclusive of any element of value arising from the accomplishment or expectation of the merger or consolidation, together with a fair rate of interest, if any, to be paid upon the amount determined to be the fair value. In determining such fair value, the Court shall take into account all relevant factors. In determining the fair rate of interest, the Court may consider all relevant factors, including the rate of interest which the surviving or resulting corporation would have had to pay to borrow money during the pendency of the proceeding. Upon application by the surviving or resulting corporation or by any stockholder entitled to participate in the appraisal proceeding, the Court may, in its discretion, permit discovery or other pretrial proceedings and may proceed to trial upon the appraisal prior to the final determination of the stockholder entitled to an appraisal. Any stockholder whose name appears on the list filed by the surviving or resulting corporation pursuant to subsection (f) of this section and who has submitted such stockholder s certificates of stock to the Register in Chancery, if such is required, may participate fully in all proceedings until it is finally determined that such stockholder is not entitled to appraisal rights under this section.
- (i) The Court shall direct the payment of the fair value of the shares, together with interest, if any, by the surviving or resulting corporation to the stockholders entitled thereto. Interest may be simple or compound, as the Court may direct. Payment shall be so made to each such stockholder, in the case of holders of uncertificated stock forthwith, and the case of holders of shares represented by certificates upon the surrender to the corporation of the certificates representing such stock. The Court s decree may be enforced as other decrees in the Court of Chancery may be

enforced, whether such surviving or resulting corporation be a corporation of this State or of any state.

- (j) The costs of the proceeding may be determined by the Court and taxed upon the parties as the Court deems equitable in the circumstances. Upon application of a stockholder, the Court may order all or a portion of the expenses incurred by any stockholder in connection with the appraisal proceeding, including, without limitation, reasonable attorney s fees and the fees and expenses of experts, to be charged pro rata against the value of all the shares entitled to an appraisal.
- (k) From and after the effective date of the merger or consolidation, no stockholder who has demanded appraisal rights as provided in subsection (d) of this section shall be entitled to vote such stock for any purpose or to receive payment of dividends or other distributions on the stock (except dividends or other distributions payable to stockholders of record at a date which is prior to the effective date of the merger or consolidation); provided, however, that if no petition for an appraisal shall be filed within the time provided in subsection (e) of this section, or if such stockholder shall deliver to the surviving or resulting corporation a written withdrawal of such stockholder s demand for an appraisal and an acceptance of the merger or consolidation, either within 60 days after the effective date of the merger or consolidation as provided in subsection (e) of this section or thereafter with the written approval of the corporation, then the right of such stockholder to an appraisal shall cease. Notwithstanding the foregoing, no appraisal proceeding in the Court of Chancery shall be dismissed as to any stockholder without the approval of the Court, and such approval may be conditioned upon such terms as the Court deems just.
- (l) The shares of the surviving or resulting corporation to which the shares of such objecting stockholders would have been converted had they assented to the merger or consolidation shall have the status of authorized and unissued shares of the surviving or resulting corporation.